## NUREG-0945 Vol. 2

# Final Environmental Impact Statement on 10 CFR Part 61 "Licensing Requirements for Land Disposal of Radioactive Waste"

Appendices A-B

## U.S. Nuclear Regulatory Commission

## Office of Nuclear Material Safety and Safeguards

November 1982



Salety and Saleguards

en anti-service and a service of the the service of the se the service of the

(a) A subject of the second s second seco

Reprinted January 1986

## NOTICE

### Availability of Reference Materials Cited in NRC Publications

Most documents cited in NRC publications will be available from one of the following sources:

- 1. The NRC Public Document Room, 1717 H Street, N.W. Washington, DC 20555
- 2. The Superintendent of Documents, U.S. Government Printing Office, Post Office Box 37082, Washington, DC 20013-7082
- 3. The National Technical Information Service, Springfield, VA 22161

Although the listing that follows represents the majority of documents cited in NRC publications, it is not intended to be exhaustive.

Referenced documents available for inspection and copying for a fee from the NRC Public Document Room include NRC correspondence and internal NRC memoranda; NRC Office of Inspection and Enforcement bulletins, circulars, information notices, inspection and investigation notices; Licensee Event Reports; vendor reports and correspondence; Commission papers; and applicant and licensee documents and correspondence.

The following documents in the NUREG series are available for purchase from the GPO Sales Program: formal NRC staff and contractor reports, NRC-sponsored conference proceedings, and NRC booklets and brochures. Also available are Regulatory Guides, NRC regulations in the Code of Federal Regulations, and Nuclear Regulatory Commission Issuances.

Documents available from the National Technical Information Service include NUREG series reports and technical reports prepared by other federal agencies and reports prepared by the Atomic Energy Commission, forerunner agency to the Nuclear Regulatory Commission.

Documents available from public and special technical libraries include all open literature items, such as books, journal and periodical articles, and transactions. *Federal Register* notices, federal and state legislation, and congressional reports can usually be obtained from these libraries.

Documents such as theses, dissertations, foreign reports and translations, and non-NRC conference proceedings are available for purchase from the organization sponsoring the publication cited.

Single copies of NRC draft reports are available free, to the extent of supply, upon written request to the Division of Technical Information and Document Control, U.S. Nuclear Regulatory Commission, Washington, DC 20555.

Copies of industry codes and standards used in a substantive manner in the NRC regulatory process are maintained at the NRC Library, 7920 Norfolk Avenue, Bethesda, Maryland, and are available there for reference use by the public. Codes and standards are usually copyrighted and may be purchased from the originating organization or, if they are American National Standards, from the American National Standards Institute, 1430 Broadway, New York, NY 10018.

NUREG-0945 Vol. 2

## **Final Environmental Impact Statement** on 10 CFR Part 61 "Licensing Requirements for Land Disposal of Radioactive Waste"

**Appendices A-B** 

U.S. Nuclear Regulatory Commission

Office of Nuclear Material Safety and Safeguards

November 1982



## ABSTRACT

The three-volume final environmental impact statement (FEIS) is prepared to guide and support publication of a final regulation, 10 CFR Part 61, for the land disposal of low-level radioactive waste. The FEIS is prepared in response to public comments received on the draft environmental impact statement (DEIS) on the proposed Part 61 regulation. The DEIS was published in September 1981 as NUREG-0782. Public comments received on the proposed Part 61 regulation separate from the DEIS are also considered in the FEIS. The FEIS is not a rewritten version of the DEIS, which contains an exhaustive and detailed analysis of alternatives, but rather references the DEIS and presents the final decision bases and conclusions (costs and impacts) which are reflected in the Part 61 requirements. Four cases are specifically considered in the FEIS representing the following: past disposal practice, existing disposal practice, Part 61 requirements, and an upper bound example.

The Summary and Main Report are contained in Volume 1. Volume 2 consists of Appendices A - Staff Analysis of Public Comments on the DEIS for 10 CFR Part 61, and Appendices B - Staff Analysis of Public Comments on Proposed 10 CFR Part 61 Rulemaking. Volume 3 contains Appendices C-F, entitled as follows: Appendix C - Revisions to Impact Analysis Methodology, Appendix D - Computer Codes Used for FEIS Calculations, Appendix E - Errata for the DEIS for 10 CFR Part 61 and last, Appendix F - Final Rule and Supplementary Information.

.

÷.,

### FOREWARD

In September 1981, NRC published the Draft Environmental Impact Statement on 10 CFR Part 61: "Licensing Requirements for Land Disposal of Radioactive Waste" (NUREG-0782). This draft environmental impact statement (EIS) contains an exhaustive and detailed analysis of a wide range of alternatives. Based upon NRC analysis of public comments on both the draft EIS and upon the proposed Part 61 regulation itself (Federal Register Notice 46 FR 38081, July 24, 1981), no new alternatives or principles were identified which required analysis. No major changes were required for several requirements of the Part 61 regulation, including the overall performance objectives which should be achieved in the land disposal of low-level radioactive waste, administrative and procedural requirements for licensing a land disposal facility, and the requirements for financial assurance. Many clarifying and explanatory changes were, however, required with respect to specific rule provisions.

Given this conclusion and public comments suggesting that the number of alternatives considered in the EIS be reduced to a smaller, more understandable number, NRC has chosen not to republish the extensive analysis of alternatives as presented in the draft EIS. Rather, NRC has refined the EIS impact analysis methodology based upon public comments and has grouped the alternatives analyzed onto four major alternatives which present the basis for decisions made regarding the Part 61 requirements.

This final EIS is therefore <u>not</u> a revision of the draft EIS but a stand-alone statement which <u>uses</u> the draft EIS as a resource and reference document. Refinements made to the draft EIS assumptions and impact analysis methodology are noted and used in the final EIS. NRC hopes that in this way, the final EIS will be of a more managable size and the alternatives analyzed and conclusions reached presented in more of a concise, understandable manner.

## List of Preparers

The overall responsibility for the preparation of this draft environmental impact statement was assigned to the Low-Level Waste Licensing Branch, Division of Waste Management, Office of Nuclear Material Safety and Safeguards, U.S. Nuclear Regulatory Commission. The statement was prepared with technical assistance from the firm of Dames and Moore, White Plains, New York.

### Nuclear Regulatory Commission Staff

### Preparers

R. Dale Smith, Chief Low-Level Waste Licensing Branch Kitty S. Dragonette Paul H. Lohaus George C. Pangburn G. W. Roles

#### Contributors

Timothy C. Johnson Mary Jo Seemann James A. Shaffner David L. Siefken Derek Widmayer

#### Special Contributors

Willste CRESS Central Dictation Unit Carole Finan Jeannette Kiminas Pauline Rock Charline Simon Irene Suissa

Cathy S. Bromberg, Secretary, Low-Level Waste Licensing Branch Robert Fonner, Attorney, Office of the Executive Legal Director

## Dames and Moore Staff

Others contributing significantly to this environmental impact statement included personnel of the firm of Dames and Moore, Inc. (D&M) of White Plains, New York. The D&M contribution was directed by Dr. Oktay I. Oztunali.

**OPTIONR & GRWATRR Codes** 

Oktay I. Oztunali C. Joseph Pitt

. ·

.

Input on I-129 & C-14

Oktay I. Oztunali Leslie Skoski Kim D. Petschek APPENDIX A STAFF ANALYSIS OF PUBLIC COMMENTS ON THE DRAFT ENVIRONMENTAL IMPACT STATEMENT FOR 10 CFR PART 61

•

## APPENDIX A

STAFF ANALYSIS OF PUBLIC COMMENTS ON THE DRAFT ENVIRONMENTAL IMPACT STATEMENT FOR 10 CFR PART 61

The draft Environmental Impact Statement (DEIS) for 10 CFR 61 was issued in September 1981 as NUREG-0782. The public comment period for the DEIS ended on January 14, 1982, and during this period 50 commenters provided written comments to NRC. Of the 50 comments received by the Commission, 8 contained no reference to the DEIS but were limited instead to comments on 10 CFR 61.

In this appendix the staff has assembled and organized the comments received and the staff's responses to them. The comment letters and staff responses have been placed in order of receipt by reference to the docket number assigned to each letter. Each letter was reviewed by the staff to identify items which required a response for clarification, additional information, etc. Each such item was bracketed in the margin of the letter and assigned a number. Therefore, response items are identified by their item number within the letter and the docketed comment number of the letter: e.g., Item 6, Comment 25.

Response items were assigned to appropriate staff members for preparation and were then organized for each specific comment letter. In the pages that follow, each docketed comment letter is reproduced in its entirety and adjoining each letter are the staff's responses to specific items within the letter. (The 8 letters mentioned earlier which do not comment upon the DEIS have not been reproduced in this appendix. However, they were considered by the staff in the rulemaking and are available for public review at the NRC Public Document Room (PDR), 1717 H Street NW., Washington, D.C. and are included in the analysis of rule comments in Appendix B.)

A-1

## APPENDIX A LIST OF COMMENTERS ON THE DEIS

.

•

.

Docketed Comment	
Number	Commenter
_	
1	State of New Jersey
2	State of Georgia
3	Department of Housing and Urban Development
4	State of Hawaii
5	State of Missouri
6	State of South Dakota
7	State of New Jersey
8	Georgia Institute of Technology
9	Stock Equipment Company
10	State of Rhode Island
11	Argonne National Laboratory
12	Massachusetts Natural Heritage Program
13	State of Delaware
14	State of North Dakota
15	US Department of Health & Human Services
16	Arizona State Clearinghouse
17	State of Iowa
18	Minnesota Department of Economic Development
19	State of Oregon
20	Dow Chemical, U.S.A.
21	Northeast Utilities
22	Township of Lower Alloways Creek
23	General Electric
24	Amy Hubbard
25	Los Alamos National Laboratory
26	State of California
27	Duke Power Company
28	Arkansas Power and Light

Number	Commenter
29	Stone and Webster
30	Health Physics Society
31	Betty Johnson
32	Atomic Industrial Forum
33	U.S. Ecology
34	State of Washington
35	American Institute of Chemical Engineers
36	Conference of Radiation Control Program
	Directors
37	(Not assigned)
38	New England Nuclear
39	New York State Department of Environmenta
· · · · ·	Conservation
40	Commonwealth of Virginia
41	U.S. Department of the Interior
· 42	Utility Nuclear Waste Management Group
43	Tennessee Valley Authority
44	Chem-Nuclear Systems, Inc.
45	Atomic Energy of Canada, Ltd.
. 46	State of New Mexico
47	Texas Energy and Natural Resources Advisor
	Council
48	Texas Department of Water Resources
49	Virgnia Electric and Power Company
50	Argonne National Laboratory
51	U.S. Environmental Protection Agency
. •	

•

.

.

·. .

.

•

•

· · .

-----

A-3

:.\*



The New Jersey State Clearinghouse has received and is processing your Project Notification as required by the provisions of the U.S. Office of Management and Budget Circular A-95 Revised and Chapter 85, New Jersey Laws of 1944. This project has been assigned the State Application Identifier Number NJ8111021968.

Effective with the date of this letter, the Clearinghouse has assigned a 30 day review period, which is consistent with our internal procedures and Federal regulations relevant to your program. The appropriate review agencies have been requested to comment on your application, and the Clearinghouse will perform its own review. If comments are received and any conflicts or issues arise, the Clearinghouse will notify you. It may be necessary to request additional information and/or schedule a conference in order to resolve any issues prior to clearance. Otherwise, you are cleared at the end of the review period to forward your final application to the Federal funding agency, accompanied by a copy of this letter. As an applicant, it is your responsibility to include any comments with your final application submission to the Federal agency.

If you encounter any problems or have any questions, at any time during the review process, please refer to the enclosed brochure for information, contact people, and telephone numbers.

Very truly yours, State Review/Coordinator

Attachment

NOTE: is extremely important that you put your State Application Identifier Number on all forms and correspondence (especially SUMMARY FORM 424), prior to final submission to the Federal funding agency.



#### NEW JERSEY DEPARTMENT OF COMMUNITY AFFAIRS

MEMORANDUM

TO Applicant

FROM New Jersey State Clearinghouse DATE October, 1981

SUBJECT Regional Review

As a result of budgetary constraints, the Delaware Valley Regional Planning Commission (DVRPC) has been forced to reduce its review activities as mandated by Federal A-95 Project Notification and Review System requirements. In order to insure that the regional perspective continues to be represented, the New Jersey State Clearinghouse has assumed the responsibility of circulating applications to the county planning board having jurisdiction. A copy of this application has ensure to the appropriate county planning board for its review.



Docketed Comment Number: 1

<u>Commenter</u>: State of New Jersey, Department of Community Affairs, Division of Planning

<u>Responses(s)</u>: This comment was docketed and reviewed by the staff. No items were found in the comment which require a response.

and a second second

• • • • • • •

· ·

. . .

4

.

.

.

• • • • • • • •

Office of Planning and Budget

.....

Clark T. Stevens Director

## GEORGIA STATE CLEARINGHOUSE MEMORANDUM

T0:

Dale Smith, Chief Low Level Waste Licensing Branch Divison of Waste Management Office of Nuclear Material U.S. Nuclear Regulatory Commission Washington, D.C. 20355

- FROM: Georgia State Clearinghouse Office of Planning and Budget
- DATE: November 6, 1981
- SUBJECT: RESULTS OF STATE LEVEL REVIEW
  - Applicant: Nuclear Regulatory Commission
  - Project: Draft EIS on 10 CFR 61 "Licensing Requirements for Land Disposal of Radioactive Waste" State Application Identifier: GA 81-10-14-004/NUREG - 0782

The State of Georgia is pleased to review this document and to see what administrative and procedural requirements are being proposed for licensing a facility for the land disposal of radioactive wasta. However, the state is opposed to any disposal program that could impact its resources such as bedrock storage. This document as presented addresses two phases of land disposal: 1) Short-term operational and 2) Long-term operational also known as after operations phase. In either phase, the concern is the protection of the environment, and the health and safety orniderations. Some parts of this rulemaking would serve to strengthen EPD rules and regulations involving disposal of radioactive waste. The State urges NRC and other federal agencies to continue and take a lead role in trying to resolve the waste dilemnas confronting our state and society. The State will continue to work with other agencies towards improving waste management.

The following State agenices have been offered the opportunity to review and comment on this project: DNR/EPD

OPB/Physical and Ec. Dev.

CHB/1r cc: Jim Benson, EPD

270 Mashington St., S. M. . Atlanta, Georgia 30334 4-NCC

#### Docketed Comment Number: 2

<u>Commenter</u>: State of Georgia, Office of Planning and Budget, Executive Department

<u>Response(s)</u>: <u>Item 1</u> - The State notes its opposition "... to any disposal program that could impact its resources..." The proposed 10 CFR Part 61 and its supportive Environmental Impact Statement (NUREG-0782) do not constitute a disposal program, but rather a comprehensive set of regulations to be applied to the land disposal of radioactive waste. NRC will not be acting as a proponent or supporter of any specific waste disposal site or sites, but will review and take licensing action on any disposal facility application brought to it by commercial entities, individual states or regional compacts of states. Moreover, the siting requirements in § 61.50 of 10 CFR Part 61 require avoidance of areas having known natrual resources which, if exploited, would result in failure of the site to meet the performance objectives.

. .

. . . . 

DEPARTMENT OF HOUSING AND URBAN DEVELOPMENT REGIONAL/AREA OFFICE EXECUTIVE TOWER - 1405 CURTIS STREET DENVER, COLORADO 80202

REGION VIII

November 10; 8981 NOV 27 A10:31

BRANCH

£. 1....

IN REPLY REFER TO: 850Q-590d

54 B

(3)

51776

ⓓ

Low Level Waste Licensing Branch Office of Nuclear Material Safety and Safeguards Nuclear Regulatory Commission Washington, D.C. 20555

Gentlemen:

Thank you for the opportunity to review and comment on the Draft Environmental Impact Statement (EIS) on 10 CFR Part 61. "Licensing Requirements for Land Disposal of Radioactive Waste."

Your draft has been reviewed with specific consideration for the areas of responsibility assigned to the Department of Housing and Urban Development (HUD). This review considered the proposal's compatibility with local and regional comprehensive planning and impacts on urbanized areas. Since this Draft EIS did not attempt to address site-specific locations for disposal, we would request that impacts be considered on an individual site basis prior to any actual site selection.

If you have any questions regarding these comments, please contact Mr. Carroll F. Goodwin, Area Environmental Officer, at FTS 327-3102.

Sincerely,

Raymond D. Mckinney mul

Director Program Planning and Evaluation

> 14 AREA OFFICE DENVER, COLORADO

#### Docketed Comment Number: 3

Commenter: Department of Housing and Urban Development, Region VIII Office

Response(s): Item\_1 - Impacts will be considered on a site-specific basis. Site selection will, however, be conducted by a potential applicant prior to submittal of an application to NRC. In the NRC environmental review of the application, the staff will evaluate impacts of the proposed site relative to alternative sites considered by the applicant.

• ...

----

· . · . .



• • .

Subject: Draft Environmental Impact Statement on 10 CFR Part 61 "Licensing Requirements for Land Disposal of Radioactive Wastes"

Thank you for the opportunity to review the subject draft EIS. Inasmuch as the proposed regulations provide ample opportunity for the expression of State concerns prior to the licensing of a disposal site, we feel that any Coastal Zone Management (CZM) program concerns can be addressed at that time. Moreover, should a disposal site be proposed within the CZM areas of the State, the federal consistency provisions of the National CZM Act require that the licensing be subject to State review for consistency with Hawaii's federally approved CZM program.

• •

Sincerely Hideto Kono

#### Docketed Comment Number: 4

<u>Commenter</u>: State of Hawaii, Department of Planning and Economic Development

<u>Response(s):</u> Item 1 - The commentor's observation regarding the Coastal Zone Management Act of 1972 (PL92-583, as amended by PL94-370) is correct. Section 307 (c)(3) of the act states that "... any applicant for a required Federal license or permit to conduct an activity affecting land or water uses in the coastal zone of that state shall provide to the licensing or permitting agency a certification that the proposed activity complies with the state's approved program and that such activity will be conducted in a manner consistent with the program." Requirements contained in Part 61 (specifically §61.50) would generally preclude siting of a low-level waste disposal facility in the coastal zone. §61.50 specifically prohibits waste disposal in a "... coastal high-hazard area or wetland." NRC will, however, work closely with the states in the licensing process of a new low-level waste disposal facility to assure that the requirements of this act and other applicable legislation are complied with.

Acrows Stilled by card 12/3/81 9 Mg







State of Missouri OFFICE OF ADMINISTRATION P.O. Box 809 Jefferson City 65102

Alden Shields, Director Division of Budget and Planning

November 23, 1961

Director Low-Level Waste Licensing Branch Division of Waste Management Office of Nuclear Material Safety and Safeguards Nuclear Regulatory Commission Washington, D. C. 20555

Dear Sir:

Christopher S. Bond

Governor

Subject: 81110011 - Draft Environmental Impact Statement on 10 CFR Part 61 "Licensing Requirements for Land Disposal of Radioactive Waste"

The State Clearinghouse, in cooperation with state agencies interested or possibly affected, has completed the A-95 review on the above project application.

None of the state agencies involved in the review had comments or recommendations to offer at this time. This concludes the State Clearinghouse's review.

A copy of this letter is to be attached to the application as evidence of compliance with the A-95 requirements.

-

. .

Sincerely,

٠.

Agia Cope

Lois Pohl Chief, Grants Coordination

. . .

e stands

LP:fgm

Docketed Comment Number: 5

Commenter: State of Missouri, Office of Administration

## <u>Response(s)</u>: This comment was docketed and reviewed by the staff. No items were found in the comment which require a response.

Υ.

A-9

STATE PLANNING BUREAULOUTH State Capitol ALASSAL Office of 605/773-3661 Executive Management Pierre, South Dakota 57501 6 \*81 DEC -9 December 2. 1981 40,51,61 Low-Level Waste Licensing Branch Division of Waste Management Office of Nuclear Material Safety and Safeguards SCORET AUCCER Nuclear Regulatory Commission Washington, DC 20555 PROPOSED RULP RE: SAI# SD811020-E15, Volumes 1-4 DRAFT EIS on 10 CFR Part 61 "Licensing Requirements for Land Disposi of Radioactive Waste"

·•\*\*\*\*

Dear Sirs:

Thank you for the opportunity to review and comment on your draft environmental impact statement on 10 CFR Part 61 "Licensing Requirements for Land Disposal of Radioactive Waste."

Attached are some notations made by the South Dakota Department of Water and Natural Resources during the course of their review. You may want to take their comments into consideration when you prepare your final report.

Overall, South Dakota agencies felt your impact statement deserved recommendation for approval.



Tony C. Merry Commissioner STATE PLANNING BUREAU

#### COMMENTS: 10 CFR 61 PROPOSED RULES

The South Dakota Department of Water and Natural Resource's Radiation Program has reviewed the summary volume of DEIS 10 CFR 61, and we recommend approval. The following are comments relative to minor discrepancies.

1. Volume 1, page 5, section 2.1, paragraph 3.

The word "generated" should be replaced by "produced" because it may create confusion regarding fuel cycle power "generation", and the "generation" of LLW. Therefore, the last sentence should read: Institutional LLW production will account for about 19% of the non-fuel sources.

2. Volume 1, page 9, paragraph 3, line 12.

"The most important . . .". The use of "geometric means" should be justified 2

(3)

(4)

3. Volume 1, page 19, Table S.5.

The column heading "Costs & Impacts" should define units in column (i.e. dollars, etc.)

4. Volume 1, page 27, paragraph 2, line 8.

The use of the term "daughters" should be replaced by "progeny" likewise throughout the proposal.

The following comments pertain to Volume 1, Attachment A, Proposed Rule 10 CFR 61: Licensing Requirements for Land Disposal of Radioactive Waste.

- Volume 1, Attachment A, page 38082, column 1, paragraph 3, line 6. The word "numbr" should read "number".
- Column 2, paragraph 1, line 11. "nonradio-active" should read "nonradioactive".
- 3. Volume 1. Attachment A, page 38084, Column 1, paragraph 1, item(5) "Stability - " The statement appears too generalized and vague. It should read: Stability - Stability of the disposal site over the long term (100 years) is mandatory to prevent loss of site integrity; The potential for migration and transport of wastes to offsite areas should be virtually eliminated. (As an example).
- Volume 1. Attachment A, page 38084, Column 2, line 24. The word "if" should read "it".
- 5. Volume 1, Attachment A, page 38085. Column 1. The classification scheme presented covers present waste streams but contains no provisions for future waste stream developments. If a "miscellaneous" class "C" which is characterized by activity can be included it should provide a time-buffer for future amendment needs.

12/15/81 egr

#### Docketed\_Comment\_Number: 6

<u>Commenter</u>: State of South Dakota, Office of Executive Management, State Planning Bureau

·•. •

<u>Response(s): Items 1</u>: The word "generated" is widely used and generally accepted in reference to radioactive waste. Use of the word "produced" would carry with it the connotation of an end result or product, which is not appropriate to waste.

Item 2: On page D-35 of Appendix D, Low-Level Waste Sources and Processing Options, the use of geometric means in lieu of arithmetic means is discussed. The rationale is that geometric means allow representative estimates to be made from sets of data that contain a few concentrations that are several orders of magnitude greater than the majority in the set and that would dominate the average if arithmetic means were used.

<u>Item 3</u>: The commenter's observation is accurate and tables in the FEIS will be revised to clarify costs and impacts.

.

Item 4: The term "daughter" has been used routinely to describe a nuclide formed by the radioactive decay of another nuclide. The staff sees no advantage to replacing the term "daughters" with the term "progeny" in this EIS.

.

•. •••

-1

. .



Mr. Samuel J. Chilk Secretary of the Commission Low-Level Waste Licensing Branch Division of Waste Management ' Office of Nuclear Material Safety and Safeguards Nuclear Regulatory Commission Washington, D.C. 20555

RE: NJ8111021968

Draft Environmental Impact Statement on 10 CFR Part 61 "Licensing Requirements for Land Disposal of Radioactive Waste"

(flo FR 5/1776)

Dear Mr. Chilk:

This Letter of Clearance is to certify that your application, with the State Identifier Number NJ8111021968, has met the Project Notification and Review System requirements of the U.S. Office of Management and Budget's Circular A-95 Revised and Chapter 85 of the New Jersey Laws of 1944.

The New Jersey State Clearinghouse has circulated the application to the appropriate state agencies and has received comments from one (1) agency relative to its final review. Based upon these comments, which appear below, the Clearinghouse recommends that the application be:

X Approved

Approved with conditions

\_\_\_\_Disapproved

The New Jersey Department of Energy has made the following comment:

"The EIS is a positive step toward the land disposal of radioactive wastes being generated by the nuclear powered generating stations.

Current activities of the State departments involved with the national waste disposal, as noted on the attached memo, are



NEW JERSEY IS AN EQUAL OPPORTUNITY EMPLOYER

Mr. Samuel J. Chilk

-2-

December 3, 1981

A. .

being directed at mutual agreements between adjacent states for multi-state disposal rather than national repositories. I believe that Fennsylvania has taken the lead in this effort".

As an applicant, it is your responsibility to include a copy of this Latter of Clearance when you submit your formal application to the Federal funding agency. Also, if you should change your formal application by submitting a request that differs substantially from this one, then you will have to resubmit your final application to this office for review.

If you have any questions, please call Vincent Amico of my staff at 609-292-2963.

erv truly yours. State Review Coordin

RAG:cp Attachment

A-12

## STATE OF NEW JERSEY DEPARTMENT OF ENERGY INTER-COMMUNICATION DATE: November 24, 1981

TO: Dr. Bharat C. Patel, Administrator Office of Resource Application

FROM: Anthony Rizzolo

#### **PHONE: 2403**

SUBJECT: Comments On NRC Draft, Environmental Impact Statement: 10 CRF Part 61 "Licensing Requirements for Land Discosal Of Radioactive Waste", NUREG-0782, Vols 1 thru 4.

#### ORA #103-81

The U.S. Nuclear Regulatory Commission, Office of Nuclear Material Safety and Safeguards submitted this EIS. The EIS was submitted by NRC as part of its responsibility under the National Environmental Policy Act of 1969 and to demonstrate the decision process applied in the development of Part 61.

After a cursory review of the materials contained in HUREG-0782, Vols 1 thru 4, nothing obvious could be found requiring comments either positive or negative. The ultimate responsibility in the State for waste disposal siting and control will rest with N.J. Department of Environmental Protection. The documents reviewed were NRC's evaluation of the environmental impact of land disposal of radioactive waste.

State Departments involved in this area are the Department of Environmental Protection, National Governors Association and the National Council of State Legislators. New Jersey Department of Energy is not actively involved.

#### BCP/par

#### cc: Edward Linky

here . Conver News ofte Avanuase. Age

#### Docketed Comment Number: 7

Commenter: State of New Jersey, Department of Energy

<u>Response(s)</u>: This comment was docketed and reviewed by the staff. No items were found in the comment which require a response.

۰.





#### Docketed Comment Number: 8

Commenter: Georgia Institute of Technology

<u>Response(s)</u>: This comment was inadvertently docketed for both the rule and the EIS, although the comment addressed only the rule. The commenter's concerns were reviewed and responded to in the development of the final rule.

.

· · · · ·

.

Docketed Comment\_Number: 9

Commenter: Stock Equipment Company

<u>Response(s)</u>: This comment was inadvertently docketed for both the rule and the EIS, although the comment addressed only the rule. The commenter's concerns were reviewed and responded to in the development of the final rule.

Here is a second state of the gradient of the gradient Here is a second state of the Here is a second state of the second st

A-15

: • ...

.

.

.

## 46 PR 51776

(1)

, TEU

81 DEC 23 P1:39

A ST

STATE OF RHODE ISLAND AND PROVIDENCE PLANTATIONS

Department of Administration STATEWIDE PLANNING PROGRAM 265 Melrose Street Providence, Rhode Island 02907

Hr. R. Dale Smith, Chief Low-level Waste Licensing Branch Division of Waste Management Office of Nuclear Material Safety and Safeguards U.S. Nuclear Regulatory Commission Washington, D.C. 20555 HOPORED RULE PR-GI

December 15; 1981

Dear Mr. Smith:

This office, in the capacity of clearinghouse designate under OME Circular Number A-95, Part II, has reviewed the Draft Environmental Impact Statement for the Licensing Requirements for Land Disposal of Radioactive Waste.

The Technical Committee of the Statewide Planning Program was presented the staff findings as a result of the review at its meeting of December 4, 1981. The Technical Committee recommends the following:

"The section of the Draft EIS pertaining to reuse of closed disposal sites is very brief and sketchy. Recreational uses, such as a golf course, might involve excavation to construct or reconstruct the course. The concept of reuse of these sites should be studied in more detail so that effective legal protections will be required."

Comments from the R.I. Historical Preservation Commission and the R.I. Coastal Resources Management Council are attached.

We thank you for the opportunity to review this draft EIS.

Rene Fontaine A-95 Clearinghouse Coordinator

RJF/XFR/sic

Reference File: EIS-81-12

STATE OF RHODE ISLAND AND PROVIDENCE PLANTATIONS INTER-OFFICE MEMO

TO . Mr. Daniel W. Varin, Chief

DEFT : Statewide Planning Program

HROM : Mr. John A. Lyons, Chairman

DEPT : Coastal Resources Management Council

SUBJECT: Draft Environmental Impact Statement "Licensing Requirments for Land Disposal of Radioactive Waste" File No. EIS-81-12.

No comment on the DEIS. Any proposed disposal project in Rhode Island will require CRMC review and approval.

JAL/drc

FEDEIVED

. . .

DATE: 23 November 1981

NOV 24 1981 AU 71819101111213141516 A 

STATE OF RHODE ISLAND AND PROVIDENCE PLANTATIONS

HISTORICAL PRESERVATION COMMISSION Old State House **150 Benefit Street** Providence, R.I. 02903 (401) 277-2678

11

December 3, 1981

#### RE: EIS-81-12

Mr. Daniel W. Varin, Chief Rhode Island Statewide Planning Program 265 Melrose Street Providence, RI 02907

Dear Mr. Varin:

This office has reviewed the above-referenced DEIS for licensing land disposal of radioactive waste.

..... The DEIS is deficient in that it does not even mention impacts to cultural properties (or materials, for that matter). The DEIS should be revised to assess impacts to cultural properties, since the proposed activity may affect such resources.

> nchrely. ent felder Deputy State Historic Preservation Officer

Docketed Comment Number: 10

Commenter: State of Rhode Island, Department of Administration, Statewide Planning Program

1 1

Response(s): Item 1 - The staff received several comments on the draft rule with respect to control of site access and productive land use during the active institutional control period. The staff considered the issue of reuse of waste disposal facility land and determined that the government landowner administering the active institutional control program should have flexibility in controlling site access. This flexibility may include allowance of productive uses of the land provided the integrity and long-term performance of the site are not affected (emphasis added). Any productive use of the land during the institutional control period will require prior review and approval by NRC through the licensing process and would specifically consider the potential effects on site integrity and long-term performance."

and the second second Item 2 - The staff recognizes that construction and operation of land disposal sites for radioactive waste disposal may result in impacts to cultural resources. In preparing the draft EIS, the staff felt that these impacts were site-specific in nature and could not be adequately assessed in the absence of a specific site proposal. In the review and licensing process of a proposed disposal facility. NRC will operate under the requirements of Federal laws and regulations for the protection of cultural resources. Among other things, these requirements include coordination with the State Historic Preservation Office, conduct of a pre-construction cultural resources survey and the identification of mitigating measures to protect any known or encountered resources.

. .

RECEIVED P.I. STATEWIDE FLAMUNG PROGRAM DEC 4 1981 789,10,10,12,1,2,3,4,5,6 ļ.

(ຊ)

A-17

US Department of Every	
ARCONNENATIONAL LABORATORY (1997)	
9700 South Cass Avenue Arconne Ilinois 60439 Edebhone 312/	172-6677
- 81 DEC 23 P3:05	
(GB) December 14	, 1981
DOCKET NUMBER PR-2, 19, 20, 21, 30, 40, 51,	
EROPOSED BULLET TO THE FILL OF	
61, 70, 73, 170 (46 FR 38081) I MORE NUMBER DD - (1)	
ADDATED BULE FIL-COT	
(116 FR 51776)	
Mr. R. Dale Smith, Chief	
Low-Level Waste Licensing Branch	
Division of Waste Management	
U. S. Nuclear Regulatory Commission	

Dear Mr. Smith:

Washington, D. C. 20555

Subject: Comments by Argonne National Laboratory on NRC Proposed Licensing Requirements for Land Disposal of Radioactive Waste (10 CFR 61), and Supporting Environmental Impact Statement (NUREG-0782)

Argonne National Laboratory has reviewed the Proposed Licensing Requirements for Land Disposal of Radioactive Waste (10 CFR 61) and the supporting Environmental Impact Statement (NUREG-0782). Our comments are attached.

We believe that the proposed 10 CFR 61 rule will provide a workable regulatory framework for licensing and operating new low-level radioactive disposal sites. The site requirements and criteria, operating and closure practices, and standards are conservative but in our opinion are generally practicable.

Very truly yours,

J. Howard Kittel, Manager Office of Waste Management Programs

JHK:sfn Enclosure

Acknowledged by card. 5 23 81 mdy

Comments on Proposed Licensing

•

Requirements for Land Disposal of Radioactive

Waste, 10 CFR Part 61, and on Supporting Draft

Environmental Impact Statement, NUREG-0782

Argonne National Laboratory

December 10, 1981

I. 10 CFR 61

#### A. General Comment

Our general impression of the proposed rule 10 CFR 61 is that it is a good document. It should provide a workable regulatory framework for the successful licensing - and operation - of new low-level waste disposal sites. We do not find any serious flaws. It proposes reasonable site requirements and criteria, operating and closure practices, and standards. It implicitly and explicitly states, by virtue of its performance standards, that zero release or zero migration is not expected.

#### B. Definitions (61.2)\*

The addition of definitions and discussions of several terms which have been omitted from Section 61.2 (Definitions) might eliminate some ambiguities in interpreting the regulations. The suggested additions and the reasons for adding them are outlined below.

1. "Long-Term" In Supplementary Information, Section V.B, "long-term" is defined as the time after operations cease (presumably the post-closure period). It is not clear that this is the intended definition to be used in the many references to "long-term" in the regulations. If so, further subdivision of the time following cessation of operations may be appropriate because the impacts and problems for different intervals of time beyond closure are quite different. For example, the problems during the period that one can rely on "passive" institutional controls (deeds, records, etc., that allow the owner and potential user to be aware of past use) are different from the problems beyond that period, and also from the problems in the period of active institutional control. A claim [Section 61.7(b)(3)] that is reasonable for a period of the order of 1000 years is that future occupation and use of the site is unlikely; it is less reasonable for a period of the order of  $10^4$ years or longer. It has not been established that the allowed concentrations of very long-lived radioisotopes are low enough to permit unrestricted use of the site (which must be considered probable after all records are lost), and there is nothing in the regulations that limits the period of concern for public health and safety.

\* Numbers in ( ) refer to Section Nos. in 10 CFR 61.

The University of Chicago

ARCONNE UNVERSITIES ASSOCIATION

-2-

2. <u>"Disposal"</u> The word "disposal is commonly interpreted to mean "permanent disposition of". If this is the intended definition, it should be so stated and noted that mear-surface disposal is not necessarily a permanent means of disposition. Over a time period of the order of 10<sup>5</sup> years or longer, one cannot exclude the possibility (or even the likelihood) that the waste will be dispersed into the environment. The definition of "disposal" raises a legacy problem, and the implications of this for the hazards of waste with the limiting uranium and TRU concentrations need to be addressed, or at lease acknowledged, in the regulations.

3. <u>"Stability"</u> It is not clear whether the word "stability" is meant to be volume stability, so that the waste will not degrade, slump or collapse after burial, or also shape and physical stability, so that an intruder would clearly distinguish it from soil. If the former definition is allowed, then FUSRAP and similar waste is stable; if the latter definition is intended, it is not. If volume, shape, physical stability are required, some time limits may be needed; it might be difficult to ensure shape and physical stability for 10<sup>4</sup> years or longer unless rather expensive means, such as those proposed for high-level wastes, were used.

#### C. Protection of General Population from Releases of Radioactivity (61.41)

1. The performance objectives are given in terms of radiation dose. Since chemically-toxic, in addition to radiotoxic, substances may also be present in the waste, we believe that a general statement, at least, be included to the effect that releases of chemically-toxic substances shall not exceed any local or Federal standards that exist.

2. Two sets of radiation standards have been specified - one in terms of annual dose to any member of the public (25 mrem whole body and any organ except thyroid) and one in terms of drinking water concentration. The latter standard is based on 4 mrem/year for man-made radionuclides. Although it is recognized that the former is for <u>individuals</u> and the latter is for <u>populations</u>, it appears there are two different sets of standards. It is conceivable that releases to the general environment may cause exposures to as many individuals as contamination of the nearest public drinking water supply.

3. Regarding the statement "...at the nearest public drinking water supply...," this supply may not be the one most likely to be affected by the disposal site. The intent of this performance objective is certainly meant to apply to any water supply contaminated by waste migration, and this should be so stated.

4. It is possible that the last sentence in this paragraph might be misinterpreted by some to mean that the national drinking water standards are being applied to groundwater in general and not only to public drinking water supplies. We suggest that this sentence be reworded in somewhat this manner: "The waste disposal site shall not cause the National Primary Drinking Water Standards to be exceeded in any public drinking water supply." Additional clarification is needed to make the first and second sentences more compatible in terms of allowable dose, since in the first sentence drinking water could

··· · · ·

yield a dose of 25 mrem to the whole body and still be in compliance, while in the second sentence it would not.

. . . .....

\*\*\*\*

£3.1

-:1

5. The evaluation of an annual dose to the individual requires a model which allows one to calculate dose from an environmental radioactivity concentration or source term. This model can, of course, not be given in the proposed rule, but it is presumed that guidance in this area will be provided later in Regulatory Guides. The rule could give some indication as to how this performance objective is to be met.

6. There is typographical error in the spelling of "radioactive" in the second sentence of the paragraph.

7. This Section is a general statement on standards, although not specifically directed at these. Standards are fixed absolute numbers, regardless of the uncertainties in the data on which they are based. Measurements and calculations made to assess performance against these standards are subject to uncertainties and to analytical and statistical errors. Thus, if the standard is 5 pCi/l, is a measurement of  $5.1 \pm 0.2$  pCi/l in violation? Probably yes, but is a measurement of  $4.9 \pm 0.2$  pCi/l in violation? Probably no, but the two measurements do not significantly differ. It would be reasonable and useful if the standard could address this problem in some way. We do not have a clear answer at this time, but it is a technical rather than a legal question, and this may make it difficult to resolve. Possibilities are (1) specify a dose standard, e.g., 25 mrem/year, and the probability of delivering that dose, (2) specify a concentration, e.g., 5 pCi/l, and the standard deviation tolerated in a measurement to meat this standard and the method by which it was calculated.

#### D. Protection of Individuals from Inadvertent Intrusion (61.42)

1. It is our belief that the inadvertent intruder scenario is given too much weight and leads to some unreasonably low concentrations in Table 1, for example, in the case of  $9^{4}$ Nb (0.002 µCi/g). This may not cause any impact on waste disposal, since  $9^{4}$ Nb is not an abundant radionuclide, but this does establish a precedent that could be unnecessarily troublesome.

2. The inadvertent intruder scenario is tenuous at best - it requires predicting some far distant future event for which the uncertainty is large - and should not be the limiting or driving force in determining the hazards.

#### . Disposal Site Suitability Requirements for Land Disposal (61.50)

1. We believe that the intent of this requirement is that the water table shall not cyclically rise into and fall beneath the buried waste. Burial beneath the water table could be satisfactory, if diffusion is the controlling rate (as stated in this paragraph), if the travel time is very slow, if the performance objectives can still be met, and if the water table never drops below the buried waste.

POPS Delow the outres resour

#### F. Environmental Monitoring (61.53)

1. It is not clearly stated in this section that the radiological and/or nonradiological (chemical and biological) characteristics of the environment should be determined to establish baseline concentrations.

2. Should there not be a reporting requirement to demonstrate compliance with applicable standards and discuss results? This is implicitly covered in 61.80 (h) (l).

G. Waste Classifications (61.55)

1. The proposed 10 CFR 61 specifically mentions two waste categories although they are outside its intended scope. These categories are: (1) wastes with radioisotope concentrations that exceed the limits in column (3) of Table I [Part 61.550]; and (2) wastes that might be exempted from the regulations (Supplementary Information, last paragraph of Section V.C). On the other hand, no mention is made in the current proposed regulations of the category referred to as "low-activity bulk solid waste" although it was included in the preliminary draft of 10 CFR 61 (issued November 5, 1979). Waste from the Formerly Utilized Sites Remedial Action Program (FUSRAP) would, presumably, fall into this category. FUSRAP waste is within the scope of 10 CFR 61, but it is unclear whether this was intended or incidental. It is of considerable interest why the low-activity.bulk solid waste category was eliminated and whether it may be re-introduced at some future time.

2. FUSRAP waste meets the requirements of all of the 10 CFR 61 waste classifications (except possibly with regard to dimensional stability -- see below); it is mainly soil contaminated with very long-lived radioisotopes (mostly uranium and thorium ores and processing residues) at average concentrations that are smaller than the uranium and TRU limits in Table I by a factor of 100 or more. Waste-specific requirements for Class A, B, and C wastes may not be appropriate for such wastes.

3. In raising this question regarding the fate of the low-activity bulk solid waste category, we are aware of the recent published Branch Technical Position on Disposal or Onsite Storage of Thorium or Uranium Wastes from Past Operations (46 FR 52061). The question concerns the waste identified in category 5 of the position paper, i.e., waste for which long-term disposal at a site other than a licensed disposal site will not normally be a vianle option.

H. . Labeling (61.57), and Tests at Land Disposal Facilities (61.81)

1. It is not clear where the primary responsibility lies for verifying the character of a waste shipment. Is it the responsibility of the generator (Section 61.57), or the site operator, or the Commission (Section 61.81)? What means will be adopted to provide guality assurance?

-5-

• . •

(3)

(5)

(6)

#### II. NUREG - 0782

#### A. Federal and State Responsibilities (1.2.3)\*

1. The proposed <u>differences</u> (if any) <u>between the responsibilities of</u> <u>agreement states and those of nonagreement states</u> with respect to the proposed rules are not clearly identified. For example, in the case of nonagreement state-owned disposal facility, is the state considered acceptable to provide surveillance during the site operational, closure, and institutional control phases?

 If the site is owned by a state, the proposed rules should permit transfer to federal ownership during site operation or after closure. Such action could become desirable, although unforeseen at the time the license was issued.

#### B. Other Issues Regarding Classifications (2.4.3)

1. The EIS alludes to potential nonradiological hazards in LLW, but notes that NRC <u>does not</u> plan to <u>address the total hazard</u> of LLW. Nevertheless, it is desirable that the EIS or 10 CFR 61 note that the licensing applicant must take into account possible effects from biological or chemical hazards in the LLW and from any adjacent or colocated hazardous waste disposal site.

2. A "<u>de-minimus</u>" classification should be provided for LLW that is near or below background levels. The need for such a classification was noted in the 1980 regional workshops held to review the preliminary draft regulation (see App. C, Section 6.1.3). Support for a "de-minimus" or comparable classification has also been expressed by informed study groups including the Low-Level Waste Strategy Task Force (Ref. 1), the Conservation Foundation Dialogue Group on Low-Level Radioactive Waste Management (Ref.2), and the State Planning Council on Radioactive Waste Management (Ref.3).

#### C. Reference Disposal Facility Costs (3.6.5)

1. The <u>direct operation cost for environmental monitoring</u> (about \$26,700 per year) shown in Table 3.6 (s. believed to be inadequate. We estimate that the cost of only the radiochemical analyses listed in Appendix E, page E-55 is about \$40,000 per year. In addition, the cost for sample collection, sample preparation, quality assurance, and other factors might increase this cost by a factor of two.

#### D. Alternatives to the Base Case (5.2.4)

1. The EIS mentions use of high-integrity containers, but defines "high-integrity" only in subjective terms: LLW shippers and site operators will need a tightening of the definition of "high-integrity", if the use of such containers is specified as meeting NRC technical criteria for disposal. Will NRC provide a quantitative definition of "high-integrity container". or will this be left to others, such as state authorities or the private sector?

\* Numbers in ( ) refer to Sections in NUREG-0782, Vol. 2, unless stated otherwise.

#### -6-

#### E. Classification of New Requirements (5.5.2)

1. Subsidence has proved to be a problem at LLW disposal sites, particularly in humid areas. The proposed approach of requiring structural stability for high-activity waste therefore has merit. Of greater importance, from a site operational standpoint, is the decontainerized disposal of low-activity waste, briefly discussed on page 5-113. Inis option should be available to-waste generators and site operators for low-activity waste such as building rubble, machinery and other metal objects, biological waste, and compressible trash\_\_\_Airbone activity release from dusting during emptying of containers Can be minimized by use of dust control procedures.

F. Potential Public Impacts from Small Spills During Normal Operation (6.2.1)

1. "Th-238" in Tables 6-3 is a typographical error.

. . .

G. Background Irradiation (Appendix E, 3.2.7)

The pre-operational tritium concentraion of 350 pCi/l is about three times greater than it is in our (northern Illinois) area. The gross alpha and beta concentrations are reasonable.

- н. References
  - 1. "Managing Low-Level Wastes: A Proposed Approach," EGLG Idaho, LLWMP-1 (August 1980)

4: 2. \* "Toward a National Policy for Managing Low-Level Radioactive Waste," The Conservation Foundation (June 1981). د Light & Same

3. "Low-Level Radioactive Waste Management: An Economic Assessment," State Planning Council on Radioactive Waste Management (July 1981).

. . . 1

#### Docketed Comment Number: 11

#### Commenter: Argonne National Laboratory

Response(s): Item 1 - The responsibilities of Agreement and non-Agreement States would be different with respect to licensing of a near-surface disposal facility. In the case of a facility located in an Agreement State, the state would be responsible for licensing and regulatory control of the site. In the case of a non-Agreement State, the U.S. NRC would have licensing and regulatory jurisdiction. With respect to surveillance, monitoring, institutional and other land ownership responsibilities, however, both Agreement and non-Agreement States would have the same responsibilities as landowners and NRC believes both can administer acceptable programs. (The only difference would occur during the institutional control period where, in the case of an Agreement State, the Agreement State regulatory agency would license the state custodial agency. In a non-Agreement State, however, NRC would be responsible for such licensing.) NRC thus considers states (both Agreement and non-Agreement States) as well as the Federal government acceptable for providing land ownership, surveillance and monitoring during the institutional control period.

Item 2 - Part 61 does not preclude a state from transferring ownership of a site to the Federal government. Present laws, however, contain no specific provisions for such transfers (e.g., how they would take place and what Federal agency would assume ownership responsibility). As such, no specific provisions were included in Part 61 addressing such transfers. Each would need to be worked out on a case-by-case basis.

كمشعفن

.1

.... - 7

4

Item 3 - Several commenters on the proposed Part 61 regulations suggested that NRC adopt a total hazard approach to waste classification. That is, both the radiological hazard and nonradiological hazard would be considered and related in some manner so as to arrive at a combined hazard index. This combined hazard index would then be used to set different disposal requirements for different types and forms of waste. This approach, at the moment, presents a number of practical problems--the principal problem being

the second se · . . . الاستان المراجع 

. .

ଚ

0

that to NRC's knowledge there is no accepted consistent way to numerically compare radiological and nonradiological hazards. This was the conclusion, for example, of a study which NRC commissioned to directly investigate the two types of hazard and try to compare the two numerically. (Reference 1) There are currently over 600 known radioisotopes which may emit three types of ionizing radiation: alpha, beta, and gamma. The effects of contact with radioactive material (e.g., through ingestion, inhalation or direct contact resulting in whole-body irradiation) can be quantified in terms of dose equivalents (e.g., so many millirems) which can be in turn related to an estimate of risk (e.g., a certain probability of an additional health effect). Thus, a "hazard index" (dose equivalent) which can be used to numerically compare radiological hazards exists and is generally accepted. A comparable hazard index, however, has not been generically developed for nonradiological hazards. There are hundreds of thousands of different chemicals in existence, and the level of knowledge of the effects of these chemicals on the human body is much less understood than the effects of radioactive material. Tests to determine whether a particular chemical may be a potential carcinogen are often accomplished by administering massive quantities of the chemical to laboratory animals. Here, it is difficult to relate the quantity of chemical uptake to a probability of a health effect,

. . . . .

Thus, NRC did not adopt a "total hazard" approach to waste classification for purely practical reasons. Nonetheless, NRC has not totally ignored potential nonradiological hazards associated with low-level waste. For example, proposed paragraph 61.51 (a)(7) states that only wastes containing radioactive materials shall be disposed of at the disposal site. This requirement is meant to preclude comingling of radioactive waste and nonradioactive hazardous waste. Siting a hazardous waste disposal site adjacent to a low-level waste disposal site could be allowed, however, as long as there was no interaction between the two facilities. [See §61.50(a)(11).]

Paragraph 61.56 (a)(8) requires that wastes containing biological, pathogenic, or infectious material must be treated to reduce the potential hazard. In addition, disposal facility licenses contain conditions which state that the chemical hazard in waste should not exceed the radiological hazard. Paragraph 61.56 (a)(2) prohibits waste being received at the site in cardboard or fiberboard boxes. This was intended in part to help ensure worker safety from possible harm from biological or infectious material. Finally, as part of reviewing a license for siting and operating a low-level waste disposal facility, NRC staff will review the applicant's nonradiological safety program.

Furthermore, the Commission believes that the technical provisions of Part 61 generally meet or exceed those expected in the Environmental Protection Agency's rules for the disposal of hazardous wastes. Although it is not the Commission's intent to allow disposal of hazardous wastes in a radioactive waste disposal facility, as is noted in the regulation, the Commission recognizes that such wastes may be present in low-level radioactive wastes. It is the Commission's view that disposal of these combined wastes in accordance with the requirements of Part 61 will adequately protect the public health and safety. Such hazardous wastes are expected to be such a small percentage of the total volume that dilution by other wastes would greatly minimize any risks. The Commission intends to work closely with the Environmental Protection Agency to assure continued compatibility. Further, EPA in its response to a resolution of the Conference of Radiation Control Program Directors indicated their willingness to work with other Federal agencies to deal with this problem.

Beyond this, the shipment manifest discussed in the proposed new paragraph 20.311 requires that the principal chemicals contained in the waste be identified. This is to allow identification of the presence of toxic or hazardous chemicals in specific waste streams. This will improve NRC's data base on the nonradio-logical hazard of LLW and allow consideration of any additional disposal requirements that may be required on a case-by-case basis.

Item 4 - NRC agrees that providing levels and other requirements for disposal of waste by less restrictive means (setting levels of "de minimis" waste disposal) is a very important issue. Setting such "de minimis" levels would accomplish at least three objectives:

 It would reduce costs of disposal to licensees, particularly small entities.  It would help to conserve valuable disposal space in disposal facilities for waste which truly needs to be disposed according to the Part 61 requirements.

. . . . .

.

 It would improve overall disposal site stability and thus help to reduce impacts from ground water migration and other long-term environmental releases as well as reduce long-term care costs.

Rather than delay the final EIS and promulgation of the final Part 61 rule, NRC staff have not included de minimis levels for radioactive wastes in the final EIS and rule. However, NRC intends to accelerate its schedule for development of de minimis levels. NRC believes that the fastest way to arrive at meaningful results in this matter is to first examine disposal of some specific waste streams by less restrictive means. From this experience, it is possible that generic levels may be developed which apply to all waste. In this regard, NRC is prepared to accept applications for licensees for declaring certain waste streams to be of no regulatory concern.

<u>Item 5</u> - The commenter was contacted regarding the basis for the comment and provided estimates of the costs of various radiochemical analyses. Based upon this and other data, the environmental monitoring costs for the reference disposal facility (operational and post-operational) were recomputed. The costs for the alternative improved monitoring system discussed in Appendix F of the DEIS were also recomputed.

4. . . . . . .

These new cost estimates have been set forth in Appendix C of this volume. These revised estimates do not change the overall conclusions reached in the DEIS but are included in the FEIS for the sake of accuracy and completeness.

<u>Item 6</u> - NRC has already provided specific guidance on the criteria for containers which include, where possible, quantitative data. The criteria have been set out in the draft Low-Level Waste Licensing Branch Technical Position on Waste Forms. Flexibility is being maintained to allow for a range of container designs and uses to meet individual waste generator needs.

. . . .

<u>Item 7</u> - NRC has attempted to maintain flexibility in meeting the performance, objectives of Part 61. The option of decontainerized disposal or use of sanitary landfilling techniques for low activity compressible wastes was examined by NRC and is not precluded from use under final Part 61. NRC staff recognizes that this method of disposal may be one way of achieving greater site stability for low activity compressible wastes. Of concern during licensing would be the applicant's proposed methods to maintain operational exposures and potential airborne releases to low levels.

<u>Item 8</u> - The listing in question should be for Th-228. See Errata section of this volume.

Item 9 - The comment refers to estimated background levels of tritium in surface water and ground water in the environs of the reference disposal facility. These estimates were included in Appendix B of the DEIS for illustration and completeness and are not meant to be representative of the Northern Illinois area. Background levels of tritium and other radionuclides in the environment vary from one region or area of the country to the next.

. .



Mr. R. Dale Smith Division of Waste Management Office of Nuclear Material Safety and Safeguards Nuclear Regulatory Commission Washington, DC 20555

Re: DEIS on 10 CFR Part 61

Program Coordinator

UOCKET HUMBER

PROPOSED RULE

(46 FR 51776)

Dear Mr. Smith:

The Massachusetts Natural Heritage Program has reviewed the above referenced document and would like to compliment your agency on its careful and thorough documentation of a complex and controversial issue. As our particular concerns regard the maintenance and protection of rare species populations and other ecologically important natural features, our review focussed on the considerations given these natural environmental resources during the licensing process.

While the licensing process must comply with the National Environmental Policy Act and, thereby, with appurtenant federal laws and regulations dealing with rare spacies and significant habitats, we were pleased to acknowledge the DEIS's reference to state laws and regulations governing rare species. We would like to point out, though, that the vast majority of recognized rare species and, particularly, ecologically important habitat areas are not covered by either state or federal regulations and are not, therefore, subject to automatic consideration. So as to avoid impacting these unregulated natural features, we suggest that provision for this consideration be incorporated into the licensing procedure, particularly as it pertains to site selection. Incorporation of this concern into the site selection process could contribute to locating disposal facilities in the most environmentally sound manner, thereby avoiding later conflicts.

As you may know, nearly 30 states have Natural Heritage Programs similar to ours which map and maintain extensive records on the location and status of rare plant and animal populations and other ecologically significant features. We, and the other Natural Heritage Programs, look forward to participating in the site selection and assessment for low-level radioactive waste disposal facilities.

We appreciate this opportunity to be of service. Please feel free to contact us for additional clarification or with any questions you may have.

9112290376 811208 61 46FR51776 PDR

cc: Curtis Danforth, A-95 Clearinghouse

Department of Environmental Management 100 Cambridge Street, Boston, Mass. 02202 16171727-3188

#### Docketed Comment Number: 12

Commenter: Massachusetts Natural Heritage Program

<u>Response(s)</u>: <u>Item 1</u> - Potential applicants are encouraged by NRC to meet with the staff early in the planning process to discuss site selection criteria and procedures. These discussions will include recommendations by the staff for the applicant to consult as fully as possible with local and state agencies and resource centers having information on site-specific features such as rare species and important habitat areas which may or may not be protected by state or federal regulations. NRC will also work closely with state, county, municipal, and other agencies in the licensing process.

## BARAN STREET

182 JAN -4 All 149 STATE OF DELAWARE EXECUTIVE DEPARTMENT OFFICE OF THE BUDGET DOVER DELAWARE 19901 TELEMONE (302) 7364205 December 8, 1981 December

Low-Level Waste Licensing branch Division of Waste Management Office of Nuclear Material Safety & Safeguards Nuclear Regulatory Commission Washington, DC 20555

. .

Dear Sir:

RE: Draft Divironmental Impact Statement on 10 CFR Part 61 "Licensing Requirements for Land Disposal of Radioactive Waste" (Volumes I, II, III & IV)

The Office of the Budget, in its function as the State Clearinghouse, has reviewed the above listed Draft EIS and has no negative comments to offer at this time.

,

Sincerely Henry James becker State Bodget Director

Achnowic data by card.

.

HJD:FB:jm

## Docketed Comment Number: 13

Commenter: State of Delaware, Executive Department, Office of the Budget

· ·

<u>Response(s)</u>: This comment was docketed and reviewed by the staff. No items were found in the comment which require a response.

۸

· · · ·



Subject: Draft Environmental Impact Statement for the Licensing Requirements for Land Disposal of Radioactive Waste.

This Draft EIS was received in this office on November 2, 1981.

Thank you for submitting your draft environmental impact statement for review and comment through the North Dakota State Intergovernmental Clearinghouse.

Your draft was referred to the appropriate agencies, and no comments were received to this date.

Please send copies of the final environmental impact statement and any supplemental impact statements to the North Dakota agencies that have commented on the draft and to this office. The opportunity to review your draft is appreciated, and if this office as Clearinghouse can be of further assistance with this project, please let me know.

Sincerely yours,

nnu U

Mrs. Leonard E. Banks Coordinator State Intergovernmental Clearinghouse

BAB/gd

Acknowledged by card .. 1 4 82 mdv.

224-2447

State Intergovernmental Clearinghouse 224-2095

State & Local Planning 224-2818

Esonemic Opportunity Energy Management & Conservation 224-2258

#### Docketed Comment Number: 14

Commenter: State of North Dakota, Federal Aid Coordinator Office

<u>Response(s)</u>: This comment was docketed and reviewed by the staff. No items were found in the comment which require a response.

· -



46 MR 57.110

DEPARTMENT OF HEALTH & HUMAN SERVICES

Food and Drug Administration Rockville MD 20857

**Public Health Service** 

182 JAN -4 AII :49 DEC 24 1981

<u>.</u>

Mr. R. Dale Smith, Chief Low-Level Waste Licensing Branch Office of Material Safety and Safeguards U.S. Nuclear Regulatory Commission Washington, D.C. 20555

COCCES BUKER PR-GI (1 ENDERDED BUKE PR-GI (1 (46 FR 51776

Dear Mr. Smith:

1 1 1 1 1 1

• <u>•</u> • • • •

The Bureau of Radiological Health staff have reviewed the Draft Environmental Impact Statement (DEIS) on 10 CFR 61, <u>Licensing Requirements for Land Disposal of</u> Radioactive Waste, NUREG-0782, dated September 1981. In reviewing this DEIS, we have limited our comments to the public health and safety impacts associated with the proposed regulations and have the following comments to offer:

1. In commenting on NUREG-0782, it is recognized that it is not a generic DEIS on disposal of low-level radioactive waste, but is a decision document which will provide a basis for decisions on the performance objectives and technical and financial criteria set forth in the proposed 10 CFR 61. Basic performance objectives are to (1) protect the inadvertent intruder, (2) assure long-term stability, (3) protect public health and safety over the long-term, and (4) assure safety during the short-term operational phase.

Radiation protection standards are considered a part of setting the performance objectives. An annual exposure limit of 25 mrem whole body, 75 mrem thyroid, and 25 mrem to any other organ of maximally exposed individuals at the site boundary (40 CFR 190), and en annual population limit of 4 mrem at the nearest public drinking water supply (40 CFR 141) are considered to be appropriate performance objectives for land disposal of low-level radioactive waste.

Because of the wide range of potential hazards, a waste classification system needs to be developed based on the methods or requirements that should be applied for disposal. These requirements can be defined by (1) waste characteristics, (2) containment and isolation capabilities of the method of disposal, and (3) social commitment controls. The implementation of this objective would require developing a set of potential exposure events at model waste disposal facilities and determining limiting concentrations of radionuclides in the waste such that any postulated event would not result in population exposure greater than the present radiation protection guidelines. These requirements would likely assure safe disposal of the radioactive waste and would, to the maximum extent possible, provide for long-term protection of the public health and safety.

Acknowledged by card. 1 4. 82. md.

#### Page 2 - Mr. R. Dale Smith

2. The envronmental pathways identified in Volume 2, Chapter 5, and analyzed in Appendix G, Section 2, appear to cover the major emission pathways (Figures G.1 and G.2) through which radionuclides contained in the low-level wastes may be transported through the environment and impact on the population. The dose computational methodology and computer codes used to calculate individual and population exposure have resulted in reasonable estimates of doses resulting from assumed regional facility operations. The summary of the short-term and long-term environmental impacts for the base case and regional case studies shown in Volume 1, Tables 5.5, 5.6, 5.7, and 5.9 are within current radiation protection standards.

. .

(1)

The release/transport/pathway and waste classification scenarios involve complex interacting parameters and diverse mechanisms through which radionuclides may be released and transported through the environment. Thus, it should be recognized that the dose calculations are based on many assumed parameters and environmental characteristics. On this basis, the DEIS should contain a section in Volume 2 on the uncertainties in the data base by expanding Section 3.3, Development of Data Bases for the Analysis. It is particularly important to address this issue since the doses are estimated for 100 and 500 years following closure of the disposal site. In Volume 1, page 20, it is stated that the exposures are calculated in a conservative manner indicating that the doses are on the high side. Because of the uncertainties discussed above and the time frame involved, it would be more appropriate to round off the doses and show the most likely range of population and individual doses. Such an approach would make the dose data more credible and would lead to an improved public understanding of the public health and environmental impact of land disposal of radioactive waste. 195-

3. Potential public impact from operational accidents is discussed in Volume 2, Section 6.2.2. Potential releases of radioactive material to the environment could occur from rupturing of a waste container or from a fire on-site that might consume waste packages with a subsequent release of a portion of the radioactivity in the waste. It is likely that the consequences of the accident would be confined to the site, and measures to mitigate the accident would be the responsibility of the onsite radiation safety personnel. Appendix E, Section 5.2.5.4 addresses abnormal or emergency situations and cites existing Federal regulations on notification of Federal authorities. In our view, this section, perhaps, should be expanded to include emergency planning requirements and the need for coordination with State and local authorities.

The analyses of an accidental fire and of a ruptured dropped container are based on accident scenarios and methodology described in Appendix G. This analysis resulted in (1) stream-by-stream (mode) impacts to the whole body and bone from a fire accident (Table 6.3), and (2) stream-by-stream impacts to the whole body and lungs from a dropped container accident (Table 6.4). The calculated doses appear to be reasonable for the accident situation described. It is noted that any action at the facility to reduce potential long-term impacts from ground water intrusion or inadvertent human intrusion would have an additional benefit of reducing the short-term impact (i.e., individual exposure) from potential accidents.

#### Page 3 - Mr. R. Dale Smith

4. An environmental monitoring program that will be carried out at the referenced disposal facility is described in Appendix E, Section 5.2.6; and Appendix F, Section 2.3.1, and is summarized in Table E.10. This program appears to provide adequate sampling frequency and analysis for specific radionuclides in critical exposure pathways. It is considered sufficiently inclusive to measure potential emissions from short-term operational releases. Over the long-term, the ground water monitoring system is likely to detect ground water migration at the onsite and offsite sampling wells. The water sampling program should be extended to the nearest public drinking water supply to assure that the dose equivalent to the total body or any internal organ is not greater than 4 mrcm/ yr. (40 CFR 141.16). The facility operator should be on the altert to detect the potential repid movement of radioactive material through fractured or jointed geological formations and showing up in the test wells.

In our view the monitoring program must be capable of supplying information on the performance of the site, and the data must be interpreted in such a timely manner that actions to mitigate any unusual release can be initiated.

5. Volume 2, Chapter 10, identified, evaluated and quantified the effects of the proposed regulation 10 CFR 61 on management of low-level radioactive waste disposal sites. Section 10.3.3 contains the long-term and short-term radiological impacts based on the regional analysis. There is, however, no discussion in this DEIS of the risk associated with operation of a low-level waste disposal facility. In our view, this section should be expanded to quantify the risks in relation to a referenced regional facility.

Thank you for the opportunity to review and comment on this Draft Environmental Impact Statement.

Sincerely yours, Fin W/UFH Pohn C. Villforth Director Bureau of Radiological Health

#### Docketed Comment Number: 15

<u>Commenter</u>: U.S. Department of Health and Human Services, Public Health Service, Food and Drug Administration

<u>Response(s)</u>: <u>Item 1</u> - NRC agrees that uncertainties in the data bases should be included in the analyses. Due to the lack of specific data in some cases and the wide range in specific data points for many waste streams, however, NRC found that uncertainties could not be quantified and when quantified led to extremely large ranges in some cases. As such, and given the uncertain nature of accurately predicting many of the exposure pathways, NRC chose to point out the uncertainties in the data bases and pathways in the text and to emphasize that the doses reported were conservative. NRC also pointed out that potential exposures from disposal of waste at an actual site would not exceed these doses and would be much lower than those reported.

Item 2 - Potential public impacts from operational accidents were addressed in a conservative manner in the draft EIS, principally to investigate the effects that NRC's requirements addressing intrusion, environmental monitoring, and disposal facility stability would have on operational safety. It was concluded in the DEIS that such requirements generally helped to improve operational safety. NRC did not perform a detailed analysis, however, of all aspects of site operational safety. Such an analysis would have most use as part of an individual licensing action for a specific site.

In any case, NRC staff believes that the best approach would be to maintain a high degree of flexibility in possible approaches to achieving and improving operational safety. Additional safety requirements to those in the Part 61 rule would be imposed for different sites, disposal methods, or waste forms on a case by case basis. Reviewing and updating a licensee's operational safety program would also receive detailed consideration during periodic license renewal activities.

As the commenter has stated, measures to mitigate an accident would be primarily carried out by onsite radiation safety personnel. Sites licensed for radioactive waste disposal must have procedures in place for handling unusual or

(3
potentially hazardous occurrences. These procedures are supplemented by training and drills. Site procedures to handle abnormal or emergency situations would be reviewed as part of licensing the disposal site as well as during license renewals. These procedures would include personnel training and drills, use of emergency equipment, and coordination with state and local authorities (police, hospitals, fire, etc.). This latter point is addressed in Section 5.2.5 in Appendix E, but may be clarified by adding the following to the end of the second complete paragraph on Page E-54: "This planning' includes dialog and coordination with State and local authorities and emergency groups such as police, fire, and hospitals." This change has been incorporated into the Errata section of this volume.

Item 3: Water sampling will be part of an overall environmental monitoring program for a new LLW disposal facility which will included as a part of each application. The number and location of water monitoring locations is a critically site-specific consideration. NRC will review each applicant's monitoring program to assure that it will adequately assess site performance. The program may or may not include the nearest public drinking water supply.

With respect to the comment on rapid movement of radioactive material through fractured or jointed geological formations, the criteria for determining site suitability in §61.50 of 10 CFR Part 61 are intended to ensure that a disposal site is located in geologic media having predictable transport characteristics. The inability to monitor and predict site performance is one of the reasons for avoiding such formations. Although the site operator should be alert to any unusual monitoring results, the staff believes that any future licensed facilities would not be located in fractured or jointed geological formations.

. . . . . .

Item 4 - NRC's review of the applicant's proposed environmental monitoring program will be based upon the ability of the program to supply information on site performance and the applicant's procedures to collect, interpret and take appropriate action on monitoring results.

Item 5 - In the draft EIS, NRC expressed radiological impacts associated with operation of a near-surface disposal facility in terms of exposures to individuals and populations. - NRC did not convert or express these exposures in terms of risk because of the difficulty of accurately assessing risks of exposures to future populations and the small number of individuals involved who could receive a potential exposure. The staff reconsidered its decision on this issue, but has not changed its position. Expressing exposure in terms of risk would involve new work and time which is not warranted given the urgent need for Part 61 and the limited additional information which would be provided. In the DEIS, NRC compared calculated doses on a common basis to existing standards which are expressed in terms of dose equivalent. NRC has, however, attempted to express the overall impacts of Part 61 in the FEIS in a clearer manner such that comparison of alternatives and unmitigated impacts are easier to discern understand.

In addition, in response to this comment and to place in perspective the potential risk associated with the doses calculated in this FEIS. NRC has included a section in the summary which provides dose response relationships as set forth in International Commission on Radiation Protection Publication 26. The reader can use these relationships to estimate the level of risk associated with doses calculated for various alternatives.

3

	major Marti putrison - 61 (16)							
_	1997 D 53	<u>"PK (4</u>	6FR	51776	) 5		OMB Aupford No. 20- R0216	
	FEDERA	ASSISTAN	CE	2. Applicant's	s. Number	3, White all Nur application AZ	81-80-006ວ	
	1. Type Of Preapplication Action Application /Mark expropriate Notification Of Intent (Opt.) box Report Of Federal Action			D Date 19 Year Month Asy	A BZILLIA	lane A11 15881 10 28		
			Leave DEC 11 1981		Getra	Octra chein tento		
	4. Legal Applicant/Recipient					5. Lideral Employer Hentification No.		
	b. Organization Unit : Off. of Nuclea			ar Material Safety & Safeguards		6. Program (From 8. Nur	mber 17 17 10 19 19 19	
	d. City	Washi	ngton	e. Cour	ty :	Federal b. Tit Catalog)	"Unknown	
ş	h. Contact Person : R. Dale Smith			1 (301) 427-4433		Nuclear Regulatory Conum		
Ĩ	7. Title and description of applicant's project Draft Environmental					8. Type of applican	E/recipionent	
/ Recipi	Impact Statement on 10 CFR Part 61 "Licensing Requirements for Land Disposal of Radioactive Waste" - NUREG-0782					Angene G-Sotta Purper Daniel Bernungen Anton Agner C-Sonner Detroit In Highe Faumberg D-County Higher Faumberg S-City Jacks Traje		
licam	This EIS posal of r	analyzes r adioactive	equirer waste	nents for and spec	the land dis- ifically, near	(Specify): Fed.	Agency Enter exprovemente letter K	
1- 40	surface disposal. The proposed action will provide licensing proce-					9. Type of atsistence A-Basic Grant D-Insurance		
ction	dures, pe ments for	dures, performance objectives, & tech. require- ments for issuance of licenses for the land (over)					B-Supplemental Grant E-Other C-Loan Enter appropriate letter(s) []]	
Ä	Statewide, Arizona				of persons baneliting	A-New C-Re B-Renewal D-Co	vision '- E-Augmentation Intrustion Enter appropriate letter	
	13. Proposed Fund	13. Proposed Funding 14. Congressional Distr			is Of:	15. Type of change For 12c or 12e		
	a. Federal S	00 a. Applicant			D. Project	8-Decrease Dollars C-Increase Duratio	n P=Otierspectyr:	
	c. State	.00 16. Projec		Start ear month day	17. Project Duration	D-Decrease Duratic E-Cancellation	Enter expro-	
	e. Other				Year month date	19. Existing federal identification number		
	1. Total S CO to federal agency 19 20. Federal agency to receive request (Name, city, state, sip code)					1	21. Remarks added	
-	22. a. To t	22. a. To the best of my knowledge and b. If required by OMB Circular A-95 this application was submitted. No Response						
Icatio	The application are true and correct, the all responses are attached all responses are attached all responses are attached. Application correct authorized (1) A sing one State Classifier houses and correct (1) State Classifier houses are stateched attached (1) State Classifier houses are stateched (1) State Classifier hous						nouses and response attached	
- Certi	That Can's with t	e governing body of the sopie and the sopicant will comply the stached siturances if the accession oved (3)(Region II, IV will be				aringhouses forwarded)		
	23. a. Typ Cartifying	a. Typed name and title			b. Signature	•	c. Date signed Year month day	
š	tetwe					19		
	24. Agency name						25. Year month day Application received 19	
	26. Organizational Unit				27. Administrative office		28. Federal application identification	
y Act	23. Address						30, Federal grant identification	
Agene	31. Action taken	32. Funding		Year monsh dag		34. Year month day Starting		
- In a	Da. Awarded	b. Applicant	<b>•</b>	.00	35. Contect for addition	nal information	36. Year month day	
1	C. Returned for	c. State		.00	(Nome and telephon	e number)	date 19	
i	d. Deferred	d. Local		.00			Jr. numerka edded	
note	e. Withdrawn	I. Total	5	.00				
÷,	38. a. In taking above act Federal agency A-95 action of Part 1, OMS Circu		re action, an indered. If Circular A-9	ny comments received from clearing- agency response is due under provisions 35, it has been or is bring mece.		b. Feceral Agency A-95 Official (Name and telephone number)		
	474-101	<u> </u>				<u> </u>	tandard Form 424 Page 1 (10-75)	

#### Prescribed by GSA, Federal Management Circular 74-7

Docketed Comment Number: 16

٠.

Commenter: Arizona State Clearing House

<u>Response(s)</u>: This comment was docketed and reviewed by the staff. No items were found in the comment which require a response.

.

.

. . .

×..

ļ

.

<u>ᠵ</u> ᢩᢇᡕᡛᠯᢧᢦᢐᢑᠮ᠋ᠬᡯᢣ	LOCARE MUMBER PR- 61	i	46 FR 51776		
	UG FR 51776) STATE OF IOWA	, tegy a tegy		Doc	:keted Commen
	Office for Plan	ming-and Pr	ogramming	Con	<u>menter</u> : Sta
RCSERT D. RAY	523 East 12th Street, Des Moines, STATE CI	LEARINGHOUSE	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	Res	<u>sponse(s)</u> : T re found in t
EDWARD J. STANEK, PhD Director Date Assigned: November 2	O, 1981 STATE	APPLICATION IDENTIFI	<u>F</u> er: IA821130-347		
<del></del>	Review	Completed:	December 31, 1981		
APPLICANT PROJECT TITLE: )raft EIS, Requirements for APPLICANT AGENCY: U.S. Nuc Address Office C Usebiant	Land Disposal of Radioacti Tear Regulatory Commission of Nuclear Material Safety a con. D. 2055	ive Waste NUREG-078 and Safeguards	2		
FEDERAL PROGRAM TITLE, AGEN AND CATALOG NUMBER:	CY U.S. Nuclear Regulaton	ry Commission			
AMOUNT OF FUNDS REQUESTED:	NA				
PROJECT DESCRIPTION: Draft Environmental Impact Disposal of Radioactive Was NUREG-0782 Volumes 1 - 4	Statement on 10 CFR Part 6 te" 1.	1 "Licensing Require	ments for Land		
The State Clearinghouse ma	kes the following disposition	on concerning this a	pplication:		
X No Comment Nece: the Clearinghous review has been	ssary. The application mus se with this form attached performed.	t be submitted as re as evidence that the	cceived by required		
Comments are At: plus the attach performed.	tached: The application mu ed comments as evidence that	st be submitted with t the required revie	this form w has been		
STATE CLEARINGHOUSE COMMEN	IS:				
•	32 JAN -7 P3:02			•	
Ci C.	00% . 114 & SE- ERALCI				. <del>.</del>
				<b>.</b>	
CH-14 Acknowledged by	caro. 1 17 82 mdy	<u>A. Mono</u>	) MAILARAS	·	

## Docketed Comment Number: 17

.

•

Commenter: State of Iowa, Office for Planning and Programming

!

<u>Response(s)</u>: This comment was docketed and reviewed by the staff. No items were found in the comment which require a response.

•

•

A-31



Minnesota Department of Economic Development

182 JAN -7 P3:23

DOCKET

001-671

December 18, 1981

U. S. Nuclear Regulatory Commission Washington D. C. 20555

RE: Draft Environmental Impact Statement on 10 CFR Part 61 Licensing Requirements for Land Disposal of Radio Active Waste SCH #8110402 NUSEG - 078-2

HONE BULE PR-G

Dear Sir:

This is to certify that the Minnesota State Clearinghouse has, in accordance with the procedures established by Office of Management and Budget (OMB) Circular A-95, reviewed the above project. State agencies which may be interested in or affected have been informed of the proposed project by this office.

This letter is to inform you that no state agency had any negative comment regarding your proposed project, and that you are therefore authorized to proceed with the application process. Your funding agency may want to know either the State Clearinghouse number or to see a copy of this letter in order to verify that you have complied with the requirements of OMB Circular A-95.

Sincerely

Richard Woodbury, Administrator State Clearinghouse 612/296-2289

RW:pas



480 Cedar Street, St. Paul, Minnesota 55101 612/296-2753

And speaks against and the set 

Docketed Comment Number: 18

. • .

Commenter: Minnesota Department of Economic Development

<u>Response(s)</u>: This comment was docketed and reviewed by the staff. No items were found in the comment which require a response.

:

. .



Docketed Comment Number: 19

Commenter: State of Oregon, Executive Department

<u>Response(s)</u>: This comment was docketed and reviewed by the staff. No items were found in the comment which require a response.

. . . . . . . .



#### DOW CHEMICAL U.S.A.

January 12, 1982 LARKIN LABORATORY 122 2315 9420 1691 N. SWEDE RD. mf MIDLAND, MICHIGAN 48640 20 • • • • PROPOSED RULE PR-41 Secretary of the Commission U. S. Nuclear Regulatory Commission (46 FR 517) Washington, D.C. 20555 OCCUET DESCRIPTION Attention: Docketing and Service Branch PERMISSION SULL

Subject: DOW COMMENTS, PROPOSED RULEMAKING 10 CFR 61 (46 FR 3808

The subject proposed rulemaking was published in the Federal Register on July 24, 1981, with the comment period to expire on October 22, 1981. NUREG-0782, a draft environmental impact statement, was referenced to provide guidance and support to 10 CFR 61, however, NUREG-0782 had not been published. Dow comments on proposed rulemaking 10 CFR 61 are dated September 18, 1981, and were submitted to the Commission with a cover letter dated October 12, 1981. The cover letter states that additional comments will be submitted as necessary and as opportunities arise.

The Commission has extended the comment period for 10 CFR bl to January 14, 1982. Dow received NUREG-0782 on October 19, 1981, and a draft Branch Technical Position (BTP) on Waste Form dated October 30, 1981. Additional Dow comments are as follows:

#### NUREG-0782

On pages 4, 5, 6, and 7 of the comments dated September 18, 1981, Dow stated concern that NUREG-0782 may be placing too much emphasis on pathway analysis and overlooking the concepts of ALARA and best available technology and ignoring the needs and objectives of assuring protection of the workers, the general population, and the environment during the operation of the disposal facility.

Review of NUREG-0782 has confirmed that the concern is valid and justified.

#### BTP on Waste Form

Dow conments dated November 25, 1981, were submitted to Mr. Robert E. Browning, Deputy Director Division of Waste Management. Copies of the BTP and Dow comments are attached. Please consider them as part of the Dow comments on 10 CFR bl.

AN OPERATING UNIT OF THE DOW CHEMICAL COMPANY

Dow appreciates the opportunity to comment on 10 CFR 61. The Commission now has sufficient technical information and experience to justify establishment and enforcement of regulatory standards and technical criteria for the proper disposal of radioactive wastes. Dow encourages the Commission to take immediate actions on 10 CFR 61.

-2-

Sincerely,

(1. B. Oven Group Leader Nuclear & Solidification Services 517-636-3388

fo

Attachment

January 12, 1982

. . •

(1)

Docketed Comment Number: 20

· 17: Commenter: .. Dow Chemical, U.S.A.

Response(s): Item 1 - With respect to the application of ALARA to a nearsurface disposal (NSD) facility, NRC intends that ALARA apply to the performance objectives addressing releases of radioactivity to the environment and safety during operation. Changes have been made to 10 CFR 61 to reflect this intent. With respect to individual technical requirements based on ALARA, NRC made no change to the rule. Part 61 sets out minimum requirements that should be met in all cases. The choice of an individual licensee in meeting any given requirement(s) would be done on an individual basis considering all aspects of ALARA (e.g., occupational exposures during operations, effluent releases, cost, etc.). (NRC also addressed the issue of application of ALARA to an NSD facility and development of requirements based on ALARA in response to specific comments filed on proposed 10 CFR 61. The reader is referred to in the second Appendix B.)

•

A CAR SHOW ال بالان المحمل المحمل أن المان المحمل العالم العام العام العام المحمل المحمل المحمل المحمل المحمل المحمل الم المحمد المحم . . . . NRC has considered in its EIS the use of "best available technology" for the design, operation, closure and form of waste disposed of at a near-surface disposal facility. NRC does not believe, in its efforts to analyze the potential long-term costs and impacts of disposal that it has ignored the needs and objectives of assuring protection of the workers, the general population and the environment during the operations of the disposal facility. As discussed in Chapter 6 of the DEIS, the improvements reflected in Part 61 serve to reduce any potential operational impacts. NRC believes that flexibility is needed in methods to achieve and improve operational safety and that possible further improvements in operational safety may best be addressed on a case-by-case

basis. 1 . . . 3. . . A-35

. . . .

1. . 

· · · · · . .

.



Reference: Proposed Changes to 10 CFR Parts 2, 11, 20, 21, 30, 40, 51, 61, 70, 73 and 170, Federal Register pages 38, 081-38,105, dated July 24, 1981.

Dear Sir:

#### NRC Proposed Rule on Licensing Requirements for Land Disposal of Radioactive Waste

Northeast Utilities Service Company, on behalf of Northeast Nuclear Energy Company and Connecticut Yankee Atomic Power Company commends the Commission on its work to date regarding land disposal of radioactive Wastes. While the Notice of Proposed Rulensking represents a significant improvement over earlier drafts, the present version still requires additional modifications to ensure that disposal of low-level waste is accomplished in a fair and equitable manner.

Northeast Utilities' subsidiary companies, responsible for the operation of three nuclear power plants and the part owner of five others that are either operating or under construction, has a vital interest in providing for the safe and efficient disposal of nuclear waste. As such, we believe that all rules must have a sound basis and that arbitrary, capricious rules have no place in federal regulations. With these thoughts in mind, we offer our comments on the proposed rule and the accompanying "Draft Environmental Impact Statement" (NUREG-0782) for the Commission's consideration in development of the final rule for land disposal of radioactive wastes.

#### SPECIFIC COMMENTS

1. Waste Stability Requirement

Paragraph 61.7(b)(2):

As it is not possible to reduce water access to zero the phrase "eliminated or" (line 6) should be deleted. Furthermore, "stability of the waste and the disposal site" needs to be clarified as to whather stability of the disposal site refers to its operational phase or the stabilization for site closure, the latter of which, according to paragraph 61.7(c)(2), would not be required until

disposal operations are about to cease. 1/22/82 cmp

8201240457 820115 PDR PR 2 46FR38081 PDF PDR

05,10 1/1 ADD: R.D.M.H

Job Dorroghus P. G. d bozz D. Nusshowner

115 94:15

"...institutional control is relied on for periods up to 100 years to control access to the closed site. This permits the disposal of Class A segregated and Class B stable waste without special provisions for intrusion protection, since these classes of waste contain types and quantities of radioisotopes that will decay during the 100-year period to levels that do not pose a danger to public health and safety ... " (emphasis added)

This paragraph appears to support the numerical values of maximum permissible concentrations listed in Table 1. However, paragraph 61.56(b) Statest

"... The requirements in this section are intended to provide stability of the waste for at least 150 years. Stability is intended to assure that the waste does not degrade and promote slumping, collapse, or other failure of the disposal unit and thereby lead to water infiltration. Stability is also a factor in limiting exposure to an inadvertant intruder, since it provides a recognizable and nondispersible waste .... "

There is no justification for providing "stability" for 150 years when the waste does not pose a danger to public health after 100 years. Therefore, we recommend that the waste stability requirement of paragraph 61.56(b) be changed from 150 to 100 years to be consistent with paragraph 61.7(b)(4).

#### Paragraph 61.7(b)(5)

This paragraph needs to clarify whether the High Integrity Container (HIC) alone will meet the stability requirments for Class C wastes. (1.e. 500 year stability requirement)

#### Paragraph 61.44

Burial trenches that contain only Class A waste, which according to paragraph 61.55(a) are not required to be stable, should be excluded from long-term stability requirements of paragraph 61.44.

#### Paragraph 61.50(a)(5)

The terms "coastal high-hazard area" and "wetland" should be defined.

#### Paragraph 61.52(a)(3)

The term "cover" should be clarified as to whether it includes an impervious cap.

-2-

#### 2. License Renewal

#### Paragraph 61.7(c)(2) states:

"... Periodically, the suthority to conduct the above surface operations and receive waste will be subject to a license renewal, at which time the operating history will be reviewed and a decision made to permit or deny continued operation ... "

-3-

We understand the above requirement to mean that the disposal facility operating license must be renewed periodically, at five year intervals. This is not appropriate since the operation of the disposal facility is viewed by NRC and industry as a long-term activity. Therefore, the license should be for the operational lifetime of the facility as is the current licensing practice for operating plants. This would require a long-term commitment from the facility operator while at the same time prevent a possible periodic disruption of service. . . . .

Furthermore, periodic license renewals are unnecessary as NRC has authority to perform inspections under Paragraph 61.82 and to take approprists action in instances of violation under Paragraph 61.24 and 61.83 of this rule, including revoking or suspending any license. (Additional comments on public hearings associated with license renewals are pro-vided under item 5.) 

3. Transuranic Limit Paragraph V.B. states: "... The Commission is applying a 500 mrem/yr maximum individual 2010

- exposure limit for this unusual case. (intrusion) This limit is based on ICRP recommendations for dose limits to individuals and is
- a level that is recognized as providing adequate protection. Since only one, or at most a few, persons would be involved, it is not
- necessary to consider a population dose. This limit is then used
- "It to determine the allovable concentrations of nuclides in each class
- of waste. (See Paragraph 61.42) ...." . . . · · · · ·

As stated above, Paragraph 61.42 provides a 500 mrem/yr accidental instrusion dose limit for the isotopes listed in Table 1, including transuranics. Yet this stated criteria (500 mrem/yr) was apparently ignored in the development of the maximum allowable concentration for alpha-emitting transuranic isotopes for Class C waste. Rather, Paragraph V.C. states:

• • • • • • • 

"...For most of the alpha emitting transuranic puclides, the maximum allowable concentrations were calculated to be in the range of 10 nanocuries per gram currently imposed by disposal facilities. These calculations were conservatively based. in the that did not allow credit for dilution by other wastes. If this factor were changed, the values would increase somewhat. A decision was made " not to recalculate in order to come up with higher values. This ... decision is based on two factors. 'First, in the spirit of the " ALARA (as Low as Reasonably Achievable) concept, the lower value of 10 nCi/g has been demonstrated as an achievable concentration to ... control the disposal of transuranic nuclides. This value has been imposed by the Department of Energy for some eleven years and by most of the commerical disposal site operators for nearly that long. The last commercial site imposed the 10 nCi/g restriction in 1981; Thus, there is no need to increase the limit from the stand-1981; Inus, there is no new to react the point of achievability..."

. .

77

1

16

One of the major problems the NRC needs to address is the development of consistency throughout its regulations. There is no technical justification for arbitrarily lowering the limits on the maximum allowable concentration of Class C transuranic waste to 10 nCi/gm from that value necessary to limit intruder dose to 500 mrem/yr. Establishment of a limit simply because it is believed to be achievable does not consititute a valid or rational basis in the absence of a cost/benefit analysis.

An EPRI study described in NP-1494, "Activity Levels of Transuranic Nuclides in Low-Level Solid Waste from U. S. Power Reactors" presents the results of isotopic analysis of various waste forms from 25 nuclear plants. Although for transuranic isotopes, the median values were within the 10 nCi/g proposed limit, there were dozens of analyses of Pu-239. Am-241. Cu-242 and Cu-244 which contained concentrations in the 10-100 nCi/g range. These results clearly show that the 10 nCi/g limit on transuranic elements is not readily achievable in all cases and could cause needless hardship and expense. The reference to ALARA, therefore. as justification for establishment of this arbitrary limit is a blatant misuse of this concept." The "Reasonably Achievable" concept within ALARA is just as important as the concept of "As low'as..." ( See 199 Content content of the second of the second s Les Seletter We also call your attention to the fact that the House Science Committee, during a recent mark-up session on HR 5016, voted to expand the definition of transuranic waste from 10 to 100 pCi/g: · ·

Paragraph 61.55 - Table 1

At present the burial sites in South Carolina and Nevada will not accept any waste that has transuranics above 10 nCi/g. However, alpha-emitting transuranics that are found in nuclear power plant radioactive wasts are not the isotopes of major concern. Therefore, we recommend that Columns 1 and 2 of Table 1 permit burial of alpha-emitting transuranic isotopes of up to 100 nCi/g when the isotopes have been identified and rationed to specific gamma emitting isotopes. We also recommend that class A and B waste limits should also be established for Pu-241.

المانية المحرية بالمركزية (الرابي). محالج المركزية (مركزية الرابي)

Furthermore, as the intruder pathway (i.e. long-term potential for hazard) is the basis for the TRU limit, Cm-242, which has a 163 day half-life should be exempted from whatever limit is ultimately established for TRU wastes.

Paragraph 61.55(d)

The paragraph indicates that radiocctive wastes with concentrations that exceed the values shown in column 3 are not generally acceptable for near-surface disposal and shall not be disposed of without spacific Commission approval pursuant to subsection 61.58. This requirement would clearly create problems for spant resin shipments from our nuclear facilities. As such, we recommend that the final rule spacify the criteria the Commission intends to use in authorizing disposal of wastes which exceed the limits for class C wastes spacified in column 3 of Table 1.

#### 5. De Minimus Concentrations

#### Paragraph V.C. states:

. . .

"...The Commission recognizes the need for a "de minimis" classification of wastes, wastes that would be exempt from Part 61 and would be considered of no regulatory concern. The Commission believes, however, as the Federal Radiation Folicy Council has recommended, that such exemptions should be determined on a specific waste basis. In this regard, a recent rulemaking (46 FR 16230) established such an exemption in a new S 20,306 for certain levels of tritium and carbon-14 contained in liquid scintillation and animal carcass waste. Other wastes may also readily lend themselves to treatment in this menner. The Commission will be working over the next 2 years to define these wastes and provide for additional exemptions as appropriate. Thus, Part 61 will not establish a generic "de minimis" category for wastes..."

Northeast Utilities supports the "de minimus" concept and encourages the prompt establishment of the necessary criteria. It appears to us that broader use of the "de minimus" classification would result in conservation of scarce disposal site area while maintaining protection for the health and safety of the public. The AIP's National Environmental Studies Project has issued a report entitled "De Minimus Concentrations of Radionuclides in Solid Wastes" which should be reviewed by the Commission. -6-

. . .

#### 5. Public Bearings

÷

#### Paragraph V.G. states:

"...The life of a typical facility can be broken into 5 phases: presperational, operational, closure, postclosure observation, and institutional control...at intervals specified in the license, (the normal term for materials licenses is currently 5 years) the licensee would be required to submit a license remeval application (S 61.27). At this time, the disposal site closure plan and funding requirements would be updated and financial arrangements for asgurance of adequate funding reviewed. A public hearing would be offered..."

Northeast Utilities has reviewed the five phases that make up the life cycle of the disposal facility. Within these five phases there are provisions for multiple public hearings.

- 1. The first public hearing is provided for subsequent to docketing the license application.
- During operation, public hearings are provided for at each of the 5 year license renevals. For a typical facility with a 25 year life, public hearings would be held at 5, 10, 15 and 20 years after initial operation.
- A public bearing is provided for at the time of site closure, i.e., the 25th year.
- 4. The final hearing is provided for at the time of license transfer, i.e., about 5 years after closure or the 30th year.

Thus, for a disposal facility with a 25 year operating life, the regulations provide for a total of <u>seven</u> public hearings.

Although Northeast Utilities supports the concept of public participation, and believes that all partiment issues should be addressed prior to commencement of construction, we have also observed the obstruc-" tionist tactics some parties have utilized at public hearings and the disastrous influence these have had on cost and schedule of a project.

The provisions for a multitude of public hearings with their permicious effect on orderly and predictable construction, operation, closure and transfer of license processes makes us deeply concerned that no private entity will be willing to subject themselves to this degree of regulatory uncertainty.

· · · · ·

. r

-7-

Northeast Utilities believes that the conditions for facility construction, operation, closure and license transfer must be specified and agreed upon at the outset by the licenses. A public hearing will be held at that time, if requested, for public input to the process. Subsequent to license issuance, the NRC should monitor and inspect the activities at the facility to ensure they are in conformance with the license. Unforessem events can be accommodated by amending the facility license. Further participation by the public is not necessary to ensure the facility is being operated properly. The prospect of repeated public hearings every five years to consider continued operation of the facility is totally unnecessary and must be eliminated from the final rule. A more appropriate and effective oversight would be provided by assigning a full time NRC inspector to a disposal site.

· · · · ·

#### Additional Comments on 10CTR61

#### Title of Proposed Rule:

international de la construction d

÷.,

It is essential for the public to begin to distinguish between highlevel nuclear wastes which will require disposal in geologic repositories and low-level waste which, under 10 CFR 61; will be permitted to be disposed of in shallow land burial sites. Furthermore, the act P.L.96-573, which gives authority to states to establish regional compacts and for which 10 CFR 61 will be a major guideline is, is titled the <u>"Low Level</u> Radioactive Waste Policy Act" (emphases added). Therefore, we believe that it is emirely appropriate to change the title of the proposed rule to "Licensing Requirements for Land Disposal of Low-Level Radioactive Wastes."

#### Paragraph 61.59(g):

The 100 year institutional control period should be extended for as long as the governing body exists. This would extend the surveillance period and protect against site intrusion until the governing body determines the site could be reopened to the public.

.

Paragraph 61.62(g):

"Pay as you go" surety requirements for closure should be permitted, as opposed to surety bonding for an entire site.

General Comment on Draft Environmental Impact Statement (NUREG-0782)

1.

The Commission should indicate how the adequacy of quality-scaling factors (used to estimate nuclides not readily identifiable) will be determined.

1

#### General Comments on 10 CFR 20 Proposed Changes

Notice should be provided as to when the Regulatory Guide on classification of radioactive waste will be available. In the interim, guidance should be provided regarding classification of wastes as required by 20.311(c).

Paragraph 20.311(d)(3) and (f)(5):

The degree of implementation and criteria for the quality assurance programs, required under these paragraphs should be indicated. We would also recommend that the term, "quality assurance" be changed to "quality control" so as not to be confused with the quality assurance requirements of 10 CFR 50, Appendix B.

Should you have any questions regarding our comments, please feel free to contact us.

By:

Very truly yours,

NORTHEAST UTILITIES SERVICE COMPANY

: .

W. G. Counsil

Senior Vice President

Chenetta

Vice President Suclear and Environmental Engineering

•••••••

## 

## Docketed Comment Number: 21

#### Commenter: Northeast Utilities

Response(s): Item 1 - In Chapter 7 of Volume 2 (Main Report) of the draft EIS, NRC presented an example of the possible use of scaling factors to estimate the presence of trace radionuclides based upon measurements of radionuclides which are easier to measure. The example used the isotopes Co-60 and Cs-137 as indicator isotopes. The isotope Ce-144 has also been used by licensees to estimate the presence of transuranic isotopes. The scaling factors were developed based on reported concentrations obtained from a number of studies involving measurements of radionuclide concentrations in reactor wastes. One of the intents of the example was to inform the public that NRC staff recognized the difficulties that would result from a requirement to strictly measure every radionuclide listed in Table 1 in every waste package. NRC staff recognized that compliance with the waste classification requirement would be of concern and used the example as a means of helping to focus input on the subject. Specific factors that might be applied at a particular facility would be determined based on measurements of radionuclide concentrations in waste generated at that facility. The accuracy of such factors would be confirmed through periodic specific measurements. · . .

In the final Part 61 rule, NRC intends to help clarify its intent regarding waste classification by allowing indirect methods to determine radionuclide concentrations and waste classes. Further, the NRC Low Level Waste Licensing Branch of the Division of Waste Management has prepared a draft branch technical position (BTP) on waste classification and has made it available for public review. This BTP outlines acceptable methods by which a licensee may comply with waste classification, including the use of scaling factors, and will eventually be used as a basis for a regulatory guide on waste classification.

.

.

A-40

Docketed Comment Number: 22

.

Commenter: Township of Lower Alloways Creek

<u>Response(s)</u>: This comment was inadvertently docketed for both the rule and the EIS, although the comment addressed only the rule. The commentor's concerns were reviewed and responded to in the development of the final rule.

A-41

1

1

.

Docketed Comment Number: 23

Commenter: General Electric, Nuclear Energy Products Division

<u>Response(s)</u>: This comment was inadvertently docketed for both the rule and the EIS, although the comment addressed only the rule. The commentor's concerns were reviewed and responded to in the development of the final rule.

•

. .

· · :

.





Secretary of the Commission SZ 20011 U.S. Mucleur Regulatory Commission SZ 20011 Mashington, D.C. 20555 ATTM: Docketing and Service Branch of 1115

Deur Sir/Ms.:

I would like to comment on the proposed rule, "Licensin, 46 FR 5/776 Requirements for Land Disposal of Radicactive Waste" (45 FR 38081), and on the Draft Invironmental Impact Statement, HUREG-0782, supporting that proposed rule.

In light of the troubled history of land disposal of redicactive waste, this proposed rule is an udmirable attempt to regulate such facilities. The provisions which would subject these facilities to the MKC hearing process, thereby encouraging public decute, are especially commandable. However, I feel that insufficient consideration has been given to the long-term consequences of shallow land burial of redicactive wastes.

A realistic extrapolation of past and present societal trends would reveal some flaws in the analysis presented in NURIG-0782. Adjoactive wastes remain dangerous for centurias; their safe disposal is dependent upon a stable social structure for at least 500 years. Alstory (and current events) clearly shows that such stability chunch be predicted, much less insured. It is questionable whether the United States, as we now know it, will exist 500 years from now. The stability of the financial world is even more uncertain. The financial assurances proposed in NURE-0782 are not depression-proof. Should this Mation be intact in the future, its economy nost surely would not have remained static. The effects of war may end been considered; even a conventional war fought near a maste disposal site would release large quantities of radioactivity to the environment.

The DEIS analysis is particularly deficient in its assessment of future land use. Present trends in population growth, soil erosion, and water resources indicate that in the future land and sater will be extremely precious commodities. There is already a water shortage in many parts of the country. As the population increases and more and more farmland is lost to ercsion and urgan growth, wrable land will be in short supply. It may be that the "intruder scenarios" proposed in the DEIS will not be inaivertunt, as is now assumed; the need for land and food may DS/OI ,/0 be so severe that the use of waste ourial sites for agriculture ADD! may be permitted. This is especially likely since the site characteristics (topography, soil permeability, meteorology) most 20,50. favorable for waste disposal are also the best for farming or D. Harston residential purposes. The water crisis may be so serious that 7.6 d bez the use of contaminated wells will be allowed.

in example of an analysis in the DLIS that is deficient Jos Danes Kare even in the context of today's society is that of exposure calculations for the intruder-agriculture scenaric, appendix 3, Section

8201260473 820104 PDR PR 2 46FR38081 PDF PDR

Actinowise and by card 122 82 'energy

3.4.2. The intruder is assumed to be an individual who lives in a house on the waste ourial site and consumes food grown in a garden on the site. The intruder is assumed to work at a regular job during the day and spend only about half his time at home. This scenario does not consider a more plausible family-intruder, in which the following factors would alter the risk analysis: (1) some members of the family may not work or may work in the home, thus spending much more time at the site; (2) some family for a pregnant women, who are more sensitive to rediation. This analysis is inadequate for present-day society; its applicability to the future is even more uncertain.

(mit)

(4)

(5)

The DEIS also neglects the vulnerability of shallow land ourial to acts of terrorism or sanotage.

I fear that the proposed rule, if adopted, would encourage the proliferation of wuste disposal sites. This is undesirable for the reasons detuiled above. Instead, the problem of rudio-active waste disposal should be adjressed from another perspective, that of waste reduction. Hather than continuing to produce wustes when there is no sutisfactory method to dispose of them, & moratorium on waste production from those waste sources for which there exist alternative means of obtaining the same benefits is appropriate. The use of nuclear fission to generate electricity is an obvious candidate. Mable alternatives exist for the production of electricity: solar, wind, coul, magnetohydrodynamics, etc. The costs, of which waste generation is one, far cutweigh the benefits of nuclear pover plants. An innediate simidown of all operating plants will decrease the volume of waste to be aisposed of or about 65% (according to figures on p. D-65 of SUREG-0782). This would permit the use of existing maste facilities while alternatives can be developed for institutional and industrial uses of radioactive materials. This alternative, which was not considered in the DIIS, is clearly the most reasonable solution to the problem of radioactive waste disposal.

Sincerely, Aur S. Hubbard

· . ·

Box 5636 Cleveland, Chio 44101

. .

A-43

(1)

. .

#### Docketed Comment Number: 24

#### Commenter: Amy Hubbard

<u>Response(s)</u>: <u>Item 1</u> - The safe disposal of radioactive waste at a near-surface disposal facility is not solely dependent on a stable social structure for at least 500 years as noted in the comment. The approach NRC has followed in defining requirements for safe disposal of LLW is to establish controls on each of the principal components of a "disposal system"--the waste form and package, site characteristics, facility design and engineering and institutional controls. Complete reliance is not placed on any one component (e.g., institutional controls), but each acts with the others to collectively ensure safe disposal over the long term. Thus, Part 61 does not assume total and complete reliance on institutional controls to prevent disturbance of the waste. Rather, Part 61 assumes reliance on active institutional controls for a limited time frame (100 years) after which the waste form, site characteristics, and facility design and operations continue to provide the necessary control.

If the social structure were to change radically or if a major war were to be fought, as noted in the comment, the impacts from such changes would probably be far more significant than the radioactivity which might be released from a disposal facility.

Item 2 - The approach NRC has followed in Part 61 is to ensure that if someone unknowingly intrudes into a disposal facility after the end of active institutional controls, that individual or individuals would not receive an unacceptable dose. Based on NRC's analysis, exposures to such individuals, assuming reasonable activities would result in doses only a few hundred millirem at 100 years and a few millirem at 500 years.

Part 61 would allow productive uses of the site provided such uses would not affect site integrity or lead to disturbance of the disposed waste. If at some future time it is decided to use the site for productive purposes (such as farming as noted in the comment) the potential impacts of doing so would have to be weighed and balanced against the benefits.

Item 3 - For purposes of analysis NRC considered 3 intrusion events. These were selected based on evaluation of the broadest range of events possible, those potential events considered by other investigators and the likelihood of occurrence. The 3 events can be characterized as intruderconstruction (exposure to workers constructing a house at the site), intruder-agriculture (exposure to individuals living in the house constructed and consuming food grown onsite), and intruder-discovery (exposure to an individual who digs into the waste, realizes that something is wrong and ceases his excavation activities). NRC assumed that only a few individuals would be exposed through such activities based on the number of people normally required to construct and live in a house. NRC could have used much more conservative events and assumptions regarding the types of individuals involved and time spent at home. Given the unlikely nature of the assumed event, and the conservative nature of many assumptions in the analysis, NRC did not make such assumptions. NRC has generally tried to consider a more realistic set of likely individual actions rather than a less realistic worst-case approach.

• : •

Item 4 - NRC did not consider the effects of terrorism or sabotage. Besides the unlikely nature of such events, their consequences would generally be limited and involve only onsite effects.

Item 5 - The proposed rule is not intended to encourage proliferation of waste disposal sites. Rather the purpose of the rule is to establish comprehensive national standards and technical criteria for siting, licensing, operation, closure and institutional care to ensure the safe disposal of LLW.

We concur that waste reduction is a laudable goal. The staff has encouraged waste generators to use available technology and administrative procedures to reduce the volume of shipments. The disposal site operators and state governments have also encouraged volume reduction.

With respect to the suggestion for immediate shutdown of all operating nuclear power plants as a means of waste reduction, this action would result in immediate and severe impacts to society as a whole without corresponding



clearly-defined benefits. At the same time this would not eliminate the generation of waste from the shutdown plants or from nonfuel-cycle sources. Moreover, in the absence of compelling public health, safety and/or environmental reasons, NRC is not empowered to take such an action. • . • · · . ۰.

. . . . . .

. . . . .

, ÷ , . .

.

. . .

*·*·••

٠. ۰. .

1 . . .. .

• .

.

- -

..... .: ,

- ( ) - C ••.• J. ., . ~**%**p •. . . . .-

.

• • .

. .

、 ・

. • . · · 

· . ·

. 

•

۰.۰

.

A-45

Secretary of the Commission

ATTN: Docketing and Service Branch U.S. Nuclear Regulatory Commission

Washington, D. C. 20555

Dear Sir:

.....

PROFOSED BULLE

22 JUL15 25:02

Enclosed are comments on the NRC draft EIS on regulations for shallow-land

burial of radioactive waste (10 CFR 61). Questions on these comments should

be directed to either Dr. John C. Podgers or Dr. Betty A. Perkins, Group LS-6.

Environmental Science Group, Los Alemos National Laboratory, FTS 843-3167.

Sincerely,

Robert T. Lowrey

Director, Waste Management and

# Rodyno/ Parkin Million Kard by card. 1/7.7/82, good Lo Glamo Testino Lentry Surger 25-6-72-10, Trio Stop 455 Lo Glamo, True Traco 87545

#### **GENERAL COMMENTS ON THE DRAFT EIS FOR INCFR61 (NUREG-0782)**

(1

3

Environmental scientists at the Los Alamos National Laboratory have made a preliminary review of the draft EIS for the proposed NRC regulations 10CFR61.

The following comments, grouped as general and technical, are offered as brief summary statements.

#### General Issues

1) This EIS is to be commended for trying to bring a voluminous amount of data into one set of documents that can be used as a beginning for open discussion on the Issue of disposal of radioactive wastes. Moreover the presentation of the concept of intruder scenarios is important in determining possible pathways for mobilization of radionuclides. The documents also suggest the necessity for de minimus, shallow land burlal, and deeper confinement classification types of wastes, which is important.

2) The proposed regulations purport to provide a generic waste classification system which is an "umbrella" (Main Report 7.1) under which the disposal of all types of non-high level wastes can be regulated. This umbrella approach, in the manner in which the regulations are formulated, is flawed because in some cases the linking of the waste classification system with the specific disposal regulrements for near surface land disposal NSD (Main Report 7.1) has the result of imposing technological fixes (such as waste form, and layering) on classes of wastes which might have otherwise been candidates for disposal by some form of greater confinement disposal (GSD) system. These technological "fixes" are questionable because their long-term containment properties are poorly understood and largely untested under the expected geophysical conditions of shallow land burial and possible future land use.

3) The regulations, in classifying wastes as concentration as a function of volume, appear to encourage dilution as a means of allowing NSD for some types of wastes. The draft EIS should discuss whether this type of "technical fix" is desirable or whether greater confinement facilities should be the preferred mode of disposal.

4) The regulations in classifying wasta as a function of radionuclide contant place an extremcly large burden on the accurate measurement of these radionuclides. In many cases the use of scaling factors will not be satisfactory because of the variation of radionuclide distribution as a function of time, operating parameters, specific events during operation, and initial conditions. These accurate measurement requirements (many of which are technically difficult to make) in turn place a large burden on the waste generator, disposal site operator, and finally on the governmental manager (who must accept that the wastes have been accurately classified).

5) There are many uncertainties in the data base used to develop the proposed classification system. The draft EIS itself contains numerous references to the highly uncertain and often extremely variable nature of much of the data used in determining the concentration, inventory, and hazard potentials of low-level wastes (Note in particular the discussion of uncertainties in source characterization in section 3 of Appendix D, of intrusion pathway characterization in the Main Report Ch. 4, and especially the problems of defining the uncertainties in the proposed methods and parameters used to assign hazard reduction credits for supposedly atable, nondis-persible, low-leaching waste forms in the Main Report, section 4.3.4). The propagation of these uncertainties through the system to the formulation of waste classification liself is nowhere explicitly evaluated, and incorporated in these regulations. The order of magnitude increase of waste classification limits for Cs-137 over calculated concentration limits (Main Report, Section 7.2.5) is particularly questionable, in light of these uncertainties.

Although ALARA considerations are mentioned in the EIS (Main Report, sections 7.2.5, 7.2.6, and elsowhere) it is always with the qualifiers "in the interests of" or "in the spirit of" ALARA on the part of NRC, rather than where it is perhaps most needed as part of per- $\bigcirc$ formance objectives for waste generators and site operators (as was the case in cartler versions of these regulations). Thus, the requirement to meet or exceed (in the sense of ALARA) the performance objectives of these regulations has been improperly lifted from the requirements of this Part.

6

A

8

**?**} An oversight noted is that NRC has not stated its support for regulation of the hazardous, non-radioactive components in low-level wastes, or how these regulations might be integrated with the requirements of this Part.

£) Neither the documentation of the data bases for waste stream characterization (references 5 and 45, App. D), nor documentation of the data and methodology for the pathway analyses references I, 8, and 12, App. C) have been available for review due to a failure by NRC to have them published and available with the release of the EIS. This severely limits a thorough assessment of the suitability of the EIS on the proposed action.

9) While the draft EIS covers burial sites that may be licensed in the future, the draft does not discuss the impact of the regulations on sites that are presently in use or have been used. Some type of Appendix is needed to indicate how the proposed regulations may affect these existing sites. . .

#### Technical Issues

ي وروند المعالية المعالية الم As mentioned in the general comments, there are uncertainties and technical difficulties in determining radionuclide content, yet the regulations are very specific about permitted concontrations in the three waste categories. The responsibility for correctly segregating and properly certifying that wastes meet the requirements of this Part rests solely on waste generators." The NRC recognizes that waste generators face severe operational difficulties and economic penalities meeting the requirements by direct measurement, and are prepared to compromise in terms of 'scaling factors' (Main Report, section 7.5). The stated (Main Report, Ch. 7 and App. G) examples of how this might work are fraught with many technical difficultics, including two in particular: 1.) Because the data on radionuclide concentration in most waste streams are highly variable (App. G), or completely unknown and must be guessed at (Main Report, Ch. 7), resort has been made to dubious techniques such as forming products of geometric averages of basic data with geometric averages of ratios of known to unknown concentrations to obtain what are claimed to be 'reasonable' scaled estimates of unknown radionuclide concentrations. These very data bases lead one to wonder if requisite correlations between known and unknown radionuclides can be established with sufficient reliability to make the scaling factor approach acceptable, as attractive as it may be from an operational view-. point; 2.) in application, (e.g., Main Report Table 7.5), the scaling factor concept seems to be applied as though there were no other radionuclides present in a given waste stream except the pair of measured and target nuclides, and that the sum of fractions rule (Main Report, section 7.4.2) for mixtures docs not apply. For these and other reasons the concept presents a disturbing prospect for quality assurance and enforcement. Ultimately it may force disposal site operators to prepare their own verification system in self-defense, as has been seen in some recent temporary site closures in Nevada and Washington. From a larger perspective this issue can be seen to be the result of having drawn a box around the problem of low-level waste disposal and then attempting to force the solution to fit the box. The box is the linking of waste classification and the requirements of waste form and disposal by shallow land burial. 

The solution then only seems to require some means for the generator to practically and economically segregate, identify specific radionuclide content, and modify waste form or package so as to meet the requirements of disposal in the near surface environment. But of course another solution is to reformulate the problem in terms of the constructive role greater confinement disposal technologies can play alongside NSD, and thereby redefine the requirements for disposal, taking into account the many advantages GCD offers with respect to contaminant migration and human reuse of a site, and thus, considerably modify and ease the burdens of measurement and waste form modification for the generators. Such a solution would greatly enhance the prospects for quality assurance (QA) and enforcement as well.

6nto

U

ू..... इ.ट.

÷.,

The NRC should directly address the QA and enforcement issues of these proposed regulations in the EIS, and not leave them to a proposed Regulatory Guide (Main Report section 7.5). The NRC has itself identified elsewhere many problems with QA programs in other aspects of the nuclear incustry, including unqualified workers and QA inspectors, faisified records, lack of authority, lack of communication, inadequate corrective action systems, lack of superlack of authority, lack of communication, indequate corrective action systems, lack of super-vision, and poor to non-existent procedures. The proposed scheme to implement waste clas-sification minimally sketched out in the EIS (Main Report Ch. 7) could too easily suffer these sorts of QA deficiencies and should be carefully reconsidered. Preferably such reconsideration would be done in a context that would make it possible to compare the overall QA and en-forcement potential of the preferred alternative (linking waste classification and NSD requirements, plus adding scaling factors to make the system practicable), with that of a system that decouples waste classification and disposal requirements to an extent that permits GCD technology to play a constructive role. Then the possible institutional, economic, and QA penalties of the proposed action of the EIS can be more directly evaluated.

The presumption that any NSD facility will be a man-made artifact whose hazard potential (perticularly due to various forms of inadvertent human reuse) might well outlive institutional control measures is a common feature of many governmental and private industry studies of shallow land burial regulation (Main Report, section 4.2). NRC ass commendably followed this lead. However, NRC has significantly limited the intruder scenario. This limitation should not be justified on the grounds that intrusion is "only hypothetical" (Main Report, section 4.2). Artificial restrictions in the basic intruder scenarios, which have been used to set waste classification limits include: 1.) the intruder who builds a house cannot live in it (Main Report, 4.2.2.1); 2.) the intruder who lives in a house cannot drink water drawn from a well onsite or nearby (Main Report, 4.2.2.2); 3.) the agricultural intruder cannot grow deep rooted plants that would contact the wastes through remaining trench covers (App. G, 3.4.2); 4.) the agricultural intruder cannot dig stockponds, soptic tanks, drain fields or utility trenches (App. G, 3.4.2); 5.) the agricultural intruder cannot work or normally stay at home (e.g., be a farmer, housewife, or child) (App. G, 3.4.2); 6.) and the agricultural intruder cannot occupy the basement he is vapors, C-14 labeled gases, etc.) (App. G, 3.4.2). The treatment of intruder contact with artifacts or relatively stable waste forms is unsatisfactory and unconvincing (Main Report, 4.3.4). The only justification given in this EIS for a lack of attention to the consequences of burial of stable waste forms containing high surface contamination or large concentrations of long-lived radionuclides is simply that "it is not credible" that extensive human reuse of a NSD site or of extended contact or recovery of persistent waste forms buried in it would occur (Main Report, 4.3.4.3). The EIS should discuss the alternative of limiting NSD disposal to those wastes that by nature of decay and dilution in trench materials will not present a hazard to any inadvertent human reuse following loss of institutional controls, and disposing of higher activity. and/or longer lived non-high level waste in GSD systems.

NRC has gone only part way toward bringing their radiological dose assessment methodologies up to date by switching from those used in Regulatory Guide 1.109 to the Task Group Lung Model. To the extent that ingestion doses continue to be based on this Guide (App. G,

# and the second states the second

## 

A-47

(1) The second s Second s Second sec second sec 

2.4.2), they may be based on outdated data and assumptions. For example, current revisions in uranium ingestion dosimetery for environmental sources have resulted in EPA limiting uranium concentration in public drinking water to 10 pcl/1 (Adopted in these proposed Part 61 regular lions in section 61.41). These are not reflected in the pathway dose conversion factors. These factors also need to be taken into account in considering the disposal of natural and depleted uranium (which is in the proposed regulations up to the natural specific activity con-

•

5) In attempts to reproduce the NRC intruder scenario computation of concentration limits, it was found that in the agricultural scenarios plant uptake was based on the leaching of waste to the interstillial water and that only the fractions of radionuclides transferred from waste to water was assumed accessible to roots (Appendix G. 3.4.2). The calculational procedures used in the NRC calculation do not have a clear basis in the literature and moreover considerably underestimate plant uptake. The NRC calculations should be changed to reflect these considerations.

6) If the pathway assumptions and models given in the draft EIS for the arid site are used, the calculated arid site concentration limits for Pu are more restrictive than the generic site by an order of magnitude (data from App. J. Table J.S. methodology from App. G. Ch.3). There is no technical basis for the NRC conclusion (Main Report, 1.2.4) that this consequence is accquately offset by consideration of differences in intruder behavior at arid sites. The NRC should set generic concentration limits based on the most limiting site conditions.

7) The proposed flexibility reserved for deciding the final form of waste classification limits is objectionable if it is to be based on the kinds of arguments advanced in the case of Cs-137 (Main Report. 7.2.5). The quality of the source characterization data used in this EIS provides no real assurance that an additional dilution factor of 10 to 20 is warranted, particularly in light of the anticipated everaging and scaling practices to be used by waste generators.

8) A more complete discussion of waste processing than is presented in Appendix G - section 5 is needed. Emissions as a function of processing temperature, off-gas treatment systems, type of process equipment used, and operating conditions should be discussed.

9) A proposed regulation in 10CFR61 requires separation of units so that there is "no interaction between them." The draft EIS should discuss whether this is possible in terms of hydrocarbon (such as methane) migration, ponding of water from subsidence and subsequent movement into adjacent regions, etc.

10) The draft fails to address all types of wastes which may need disposal before 2000. It would appear likely for example that at least some pilot plant reprocessing wasto might need disposal. Since UFs facility westes presently contain Ra and Th and since these will be present as daughters of uranium as the uranium ages these radionuclides should be considered in the regulations as possible. In addition wastes from the burger might help the present regulations as possible. In addition wastes from the present gas

#### Docketed Comment Number: 25

#### Commenter: Los Alamos National Laboratory

Response(s): Item 1 - Prior to responding to this comment, it is useful to briefly review both the 10 CFR Part 61 rulemaking action and the DOE greater confinement disposal (GCD) study program. The Part 61 rulemaking is intended to cover land disposal of radioactive wastes (generally referred to as lowlevel radioactive waste) which are not covered by other regulations. That is, the scope of the Part 61 regulation excludes disposal of uranium mill tailings. disposal of high-level and transuranic waste in geologic repositories, and disposal by the many other possible methods defined in 10 CFR Part 20 (e.g., disposal by transfer to another person, disposal by release to air or water, disposal of H-3 and C-14 by less restrictive means). The current Part 61 regulations provide overall requirements for land disposal as well as a number of specific requirements for disposal of waste reasonably near the earth's surface. Concentration limits for near-surface disposal for a number of radionuclides were set forth in the proposed 10 CFR 61. Space is left at appropriate points in Part 61 to provide for additional specific requirements which may be developed in the future for disposal by other methods than nearsurface disposal.

NRC expects that only relatively small quantities of wastes currently being sent to operating near-surface disposal facilities will be generally unacceptable for near-surface disposal under the Part 61 concentration limits. However there may be larger quantitites of such wastes generated in the future from such activities as decommissioning nuclear power plants or plutonium recycle. NRC intends to examine in the immediate future the impacts of disposal of such waste streams by disposal methods which may offer greater confinement capability than near-surface disposal. These methods may include, for example, deeper disposal, use of engineered structures, or mined cavity disposal. NRC expects that this analysis would be performed in a similar manner as the current analysis.

6174

DOE defines greater confinement disposal as "the disposal of LLW in such a manner as to provide greater confinement of radiation, reduced potential for migration/dispersion of radionuclides, and greater protection from inadvertent human and biological intrusions in order to protect the health and safety of the public." (Reference 2) Greater confinement disposal (GCD) is intended for a few higher activity waste streams which are being generated or may be generated in the future from commercial or defense operations. DOE has defined this disposal method very broadly to include disposal at greater depths, use of engineered barriers, waste containment, and waste solidification. In Reference 2, DOE also stated their intention to demonstrate the concept by constructing and operating demonstration GCD facilities--one in an arid western environment and another in a humid eastern environment. Preliminary designs and concepts for these demonstration facilities would indicate that while the design facilities would involve only minor modifications to existing practices or experience, facilities separate from existing DOE nearsurface disposal facilities are envisioned. -. . シー・ション などの しょうせいしん · · · ·

NRC staff found a lack of clarity in the commentor's statements that the rule is flawed because the waste classification system is linked to specific disposal requirements. NRC's regulations are based upon the principle that progressively restrictive disposal requirements should be imposed on progressively more hazardous waste. Thus, in the draft Part 61 regulations, Class B waste is required to be stabilized either as part of disposal facility design or through a stable waste form or package. Class C waste must meet the stability requirement as well as a requirement for an intruder barrier (lavering). The stability requirement accomplishes a number of safety objectives, including protection of groundwater, enhancement of overall site stability, reduction of long-term care costs, improved operational safety, and reduction of potential inadvertent intruder impacts. The intruder requirement improves operational safety as well as reduces potential inadvertent intruder impacts. The above criteria also reduce impacts due to potential intrusion by burrowing animals and deep rooted plants as well as reduce potential impacts (already negligible) from release of tritiated methane. The commenter states that there are classes of waste (possibly Class B and Class C Waste) which should be disposed

 المراقع المحمدين معادلين المراقع المراقع المحمدين معادلين المراقع المراقع المحمد المراقع المحمد المحمد المراقع المحمدين محمد المحمد المحمد المحمد المراقع المحمدين محمد المحمد المحمد المحمد

by "some form of greater confinement disposal system." The commentor further questions "technical fixes" such as waste form and layering. However, NRC notes the "technical fixes" that the commenter questions accomplish similar objectives as those set out by DOE for "greater confinement disposal." For example, waste containment and solidification are defined by DOE as one method of achieving greater confinement. Deeper burial (e.g., layering) is another method suggested by DOE of achieving greater confinement. In addition, the commentor has provided no rationale for his supposition that a "greater confinement disposal system" will be any better tested or understood than the "technical fixes" in Part 61. Thus, the concept of "greater confinement" of some wastes, as suggested by the commentor, is already an intrinsic part of the Part 61 regulation. •...

Item 2 - Neither NRC nor Part 61 encourage dilution. In addition, NRC staff does not believe that requiring disposal of waste in some non-specific type of "greater confinement disposal facility" presents a reasonable alternative to the potential for dilution of waste to meet a particular waste class. That is, one is not an alternative to the other. It is even possible that a "greater confinement disposal facility" would increase the potential for waste dilution. Assuming that NRC required large quantities of waste to be disposed into such a facility, then waste dilution to avoid potentially more expensive requirements could be used.

With respect to the second part of the comment, in establishing generic requirements for low-level waste disposal, NRC recognizes that there is a wide variation in low-level waste characteristics, including waste form, waste volumes, radionuclide quantity or concentration, and chemical content. Some requirements must be established on the basis of a radionuclide concentration. For example, concentration limits can be used to establish de minimis requirements for certain waste streams, or to establish different requirements for wastes suitable for near-surface disposal facilities. Assuming that disposal by some non-specific "greater confinement disposal facility" was required for some or all (except de minimis) waste, then concentration limits would ultimately be about the only practical means to identify such waste.

Item 3 - NRC staff recognizes the difficulty of making detailed measurements of radionuclide concentrations in waste. To assist waste generators in complying with the waste classification system, NRC has prepared a technical position which outlines acceptable methods by which a licensee may demonstrate compliance with the classification system including the use of scaling factors. NRC staff continue to believe that a compliance program may be implemented by licensees in a reasonable manner. The use of scaling factors is discussed further in the response to Item 9 below.

Item 4 - As the commenter states, there are uncertainties in the radioactive waste data base. This data base was used to assess various alternative performance objectives and technical criteria for the draft Part 61 regulations. Despite these uncertainties, however, NRC staff believe that the data base is the most complete data base yet devised for low-level waste. Regulation of radioactive waste disposal is an ongoing process; it did not start with nor will it end with the present Part 61 effort. Additional information, as it becomes available, will be incorporated into NRC's program. For example, updated information on distributions of activity in LWR process wastes has been included in the calculations for the final EIS.

NRC staff also believe that the uncertainties do not preclude making an intelligent decision on Part 61 requirements. All decisions are made under the restrictions of available information. NRC staff believe that the data base and assumptions are conservative, although an effort has been generally made to avoid over conservatism. NRC staff does not believe that the uncertainties warrant adoption of much more restrictive requirements as are apparently advocated by the commentor.

In response to one of the specific parts of this comment, in the impacts analyses methodology, NRC made certain assumptions regarding the relative ability of certain waste forms to resist airborne dispersion by an inadvertent intruder and uptake by plant roots. This was done to explore the relationship of waste form with potential inadvertent intruder impacts. (Previous work had suggested that a poor waste form would tend to reduce intruder exposures. Through their analysis, however, NRC staff determined that this was exactly the opposite.) In most of the calculations in the EIS and in establishing

.

الا المداد

the waste classification limits, however, no credit was taken for improved waste forms to resist airborne dispersion or plant uptake. This was a conservatism that was included in consideration of uncertainties of the long-term performance of waste forms.

Regarding the final waste classification limits for Cs-137, it must be remembered that the limits in Table 1 were formulated based upon a number of considerations. One of the considerations was protection of the potential inadvertent intruder. This consideration was the main consideration, for example, in determining limiting allowable concentrations for near-surface disposal. However, other equally valid considerations are costs to licensees (especially small entities), groundwater migration, site stability, long-term social commitment, and operational safety. The need to reduce the potential for groundwater migration, reduce long-term social commitment, and improve site stability were especially important to developing limits for Class A waste disposal. (These points were perhaps not set out as clearly in the draft EIS as would be desired. An effort to clarify NRC's intentions has been made in the final EIS.)

The Cs-137 limits as calculated from intrusion considerations were judged in the draft EIS to involve a large cost impact to licensees. Cs-137 limits used in the final rule are similar to existing license conditions at operating disposal facilities. Cs-137 is only moderatively long-lived ( $\stackrel{\sim}{\sim}$  30 years half-life) and the increase made essentially no impact on long-term intruder impacts.

Item 5 - The rule has been amended to include specific reference  $\cdot$  to ALARA in the performance objectives for protection of populations (§61.41) and safety during operations (§61.43).

Item 6 - NRC agrees with the need to address the potential non-radiological as well as radiological hazard of LLW. For most LLW (probably greater than 90%) the predominant or controlling hazard potential is radiological. For those presenting an associated non-radiological hazard, NRC believes the requirements in Part 61 provide a high degree of safety in disposal. In addition, existing conditions in effect at the three operating sites prohibit

•

the disposal of waste having a higher non-radiological than radiological hazard. NRC plans to address this issue further in coordination with EPA. In the interim, the existing state license conditions should help minimize the potential for disposal of chemically hazardous wastes at the LLW sites and the Part 61 requirements coupled with coordination with EPA should assure an adequate level of safety in the disposal of radioactive wastes also containing associated chemically toxic material.

Item\_7 - The data bases for waste stream characterization and documentation of the data and methodology for pathway analyses were presented in NUREG/CR-1759 published in November 1981. (Reference 3.) Los Alamos National Laboratory was included in the general mailing of this multi-volume report. الجواري الجاري والمنافع المتحمد الجار

and the second second second Item 8 - Part 61 requirements are not intended to specifically apply to existing closed sites. The performance objectives regarding longterm stability and releases to the environment as well as technical requirements on conditions for closure, post-closure care and institutional control can, however, serve as guidance in the final closure and post-closure care for such sites. The implementation of the requirements at any specific site would need to be considered on a case-by-case basis considering site-specific conditions.

The application of Part 61 requirements to existing operating sites must also be handled generally on a case-by-case basis. NRC believes uniformity must be achieved in the application of the waste form and classification; design and operation; and manifest reporting and recording keeping requirements in the future operation of the existing sites. NRC plans to work closely with the Agreement State regulatory authorities to achieve uniformity in application of these and other requirements. N 4 1 .

Item 9 - The use of scaling factors to estimate concentrations of trace radionuclides in waste streams is believed to be a reasonable approach based on existing information. This approach is also believed to be conservative. In work in many previous documents, often only the principal radionuclides in low-level wastes were characterized. By including trace radionuclides, a more conservative estimate of impacts has been obtained. NRC also

. •

believes that the use of inferential measurement techniques, including the use of scaling factors to estimate concentrations of hard to measure isotopes, can be a practical and reasonably implementable approach to showing compliance with the waste classification requirements. A particular set of scaling factors, however, would be best developed for a specific facility's conditions. Such inferential measurement techniques have in fact been developed at Los Alamos National Laboratory and are in use for measurement of transuranic levels in waste. . . . 

154

-

- 1

24

The discussion in Chapter 7 of Volume 2 of the DEIS regarding the use of scaling factors was included to communicate NRC's recognition of the need to implement a workable approach to compliance with the waste classification requirement. Another intent was to provide a stepping off point for public comments on the subject. It is possible that NRC was not sufficiently clear on this matter. In any case, based upon input on the draft rule, DEIS, and from other sources, NRC staff believe that a compliance program may in fact be implemented by licensees in a reasonable manner.

### 

Finally, NRC staff does not believe that the commentor's "greater confinement disposal" system necessarily holds any inherent advantages in terms of easing the burden to licensees or enhancing the prospect for guality assurance and enforcement. NRC staff also found some lack of consistency with statements relating to the constructive role of "greater confinement disposal" alongside of near-surface disposal, or the many advantages that greater confinement disposal has regarding contaminant migration and human reuse of a site. In the first place, regardless of what advantages an undefined "greater confinement disposal" system offers, it does not follow that the burden of "measurement and waste form modification" would be eased for waste generators. There would still have to be an upper and lower limit for "greater confinement disposal" to be complied with, and waste form would in any case be of concern from operational safety considerations. Neither has it been shown that quality assurance or enforcement would be enhanced.

a that we want as the second second In any case, the concepts of greater confinement disposal which are included in DOE's own broad definition of the term are already incorporated into the

and the second and the second second second and the second second

Part 61 rule. The rule incorporates improved stability of some wastes, deeper disposal of some wastes, and sets a concentration limit for near-surface disposal.

<u>Item 10</u> - Enforcement of the requirements proposed in 10 CFR 61 will be carried out in a similar manner as applied to all NRC-licensed activities--through the NRC inspection and enforcement program. The rule proposes no new requirements in this regard.

Quality assurance of waste classification is, as the DEIS carefully points out, a subject which can result in a number of operational difficulties. The rule requires the waste generator to implement a QA program to assure compliance with §§61.55 and 61.56 and to include in this program provisions for management audits. The adequacy of each proposed licensee's QA program will be determined on a case-by-case basis. NRC staff does not understand the concluding part of this comment about decoupling waste classification and disposal requirements to an extent that permits greater confinement disposal technology to play a constructive role.

Item 11 - The very approach suggested in the last sentence of this comment is the approach NRC has followed in developing regulations for near-surface disposal of LLW. NRC intends that wastes determined to be unacceptable for near-surface disposal should be disposed of with greater controls such that the performance objectives would be realized.

A basic dilemma faced in determining the concentrations of waste acceptable and not acceptable for near-surface disposal are the pathways which should be assumed for analysis. The intrusion pathways which could be considered range from very trivial events (e.g., walking across the site) to events which could cause relatively significant exposures (e.g., an archaeologist working in the waste for extended periods of time reclaiming artifacts). Each pathway may also have a different probability of occurrence. Rather than assess probabilities of occurrence, NRC conservatively assumed that a limited number of intrusion scenarios would occur based upon considerations of typical human activities. However, given the fact that such intrusion may never occur, NRC assumed that reasonably conservative actions on the part of the intruders occur. Thus, NRC did not assume the worst, most conservative case and most of the actions discussed in Items 1-6 were not considered. NRC did consider the use of water by an intruder drawn from an onsite well. (See for example, Volume II, Section 5.2, page 5-4; and Section 5.2.2, page 5-13 of the DEIS).

Item 12 - Actually, there are a number of factors which go into the pathway dose conversion factors for calculating dose equivalents from ingestion of radionuclides. Starting with a given concentration of a radionuclide in a biota access alocation, there are a number of transfer factors which relate the transfer of the radionuclide through intermediary stages to man (for example, transfer of radionuclides from soil to plants to cattle to man), where a dose may be calculated. In the analysis, the dose conversion factors were taken from Regulatory Guide 1.109 and NUREG-0172. (References 4 and 5.) This is consistent with NRC's policy in licensing of uranium mills and other fuel cycle facilities. However, the transfer factors for the various radionuclides were determined based upon review of a number of other sources. (References 6-11.) This is discussed in Volume 3 of Reference 3. Thus, the pathway dose conversion factors for ingestion are considerably updated from Regulatory Guide 1.109.

Item 13 - On the contrary, it is the practice of assuming that radioactive waste exists in a form immediately available for plant uptake, as assumed in other documents, that does not have a clear basis in the literature. All of the analyses that have been performed to date by various parties have used transfer data from references such as Regulatory Guide 1.109, which was originally written to calculate impacts to the public from releases of trace quantities of radionuclides from nuclear power plants. The radionuclides would be so released either as fine particulates (submicron range) into air or as dissolved in water. The radionuclides can arrive at a point where they may be ultimately ingested by humans through settling out of the air, for example, or water may be used to irrigate a garden. Radionuclides may be then transferred into plant roots and this process may be quantified through use of numerical transfer factors. Values for the transfer factors are generally determined through experiments in which radioactive tracers are added to soil in a form immediately and readily available to plant roots. On the other hand, at a former low-waste disposal site, radionuclides within disposed waste are contacting or contained within a number of different forms such as spent resins, trash, activated metals, etc. When emplaced within a disposal trench, radionuclides are not in a form which can be immediately taken up by plant roots. That is, there must be a transfer factor which relates the presence of radionuclides within waste to their presence in a form readily accessible to plant roots. Certainly radionuclides bound up within a block of cement or contained within activated metal cannot be immediately accessed by plant roots. NRC staff believes that to not consider the existence of this transfer factor is to overestimate plant uptake.

This is consistent with NRC's overall approach regarding consideration of impacts to a potential inadvertent intruder. The potential for inadvertent intrusion should be considered, but the potential for severe economic impacts should also be considered. NRC staff believe that inadvertent intrusion to the extent considered in the EIS is very unlikely. Merely to assume that it occurs is conservative. Therefore, NRC staff believe that a reasonable approach is required for setting forth typical scenarios for intruder exposure. It accomplishes no good to multiply one conservatism after another merely for the sake of being conservative.

Formulation of numerical values for a transfer factor between waste and a form available for plant uptake is somewhat uncertain (as is the entire intruder scenario). As a first approximation of this transfer factor, NRC staff have used leaching data from disposal sites at Maxey Flats, Kentucky and West Valley, New York. The physical conditions under which the data were collected involved a condition in which waste was continuously saturated for a number of years.

Item 14 - In the draft EIS, different potential disposal site environments were considered. It was observed that the environmental conditions at a particular site affected the potential for dispersion of radionuclides as dust. This was accounted for numerically by development of a transfer factor,  $T_{so}$ , which relates the concentration of a radionuclide in soil to the concentration of that radionuclide in air inhaled by an intruder. Of the four hypothetical sites considered in the draft EIS, three were located

. .

1

in humid environments while one site, the one with the highest value of  $\rm T_{so},$  was located in an arid environment.

In response, NRC disagrees with the commentor's suggestion that generic concentration limits derived from consideration of potential inadvertent intrusion should be based upon the most conservative site conditions. As stated in the response to Item 13 above, NRC staff believe that a reasonable estimate of intruder exposures is preferable to the most conservative. NRC expects that over three-quarters of the waste generated in the country will be generated in humid environments. Assuming that regional disposal of radioactive waste is implemented, this means that over three-quarters of waste will also be disposed in humid environments.

In addition, NRC staff believe that there is no technical basis for <u>not</u> considering differences in waste form and intruder behavior at arid sites. In formulating a reasonable set of intruder scenarios, NRC staff believe that likely human activities must be considered, which is different from considering the most conservative possible set of human activities. Merely multiplying conservatisms by conservatisms leads to unrealistic results which may lead to severe economic impacts. Regarding arid sites, one would expect that degradation of waste into a readily dispersible form would proceed at a much lower rate at an arid site than at a humid site. This would tend to reduce potential inadvertent intruder impacts in two ways: First, the waste would be in a less dispersible form, and second, the fact that more of the waste is in a form recognizable as something other than soil would tend to reduce the likelihood that an intruder would spend significant time in contact with the waste.

In any case, there are operational techniques which may be used to further reduce the potential for intruder exposures. In PRC's analysis, an intruder barrier equivalent to 5 meters of soil or low activity waste was assumed to be effective for only 500 years. After 500 years, some credit was taken for the potential for dilution of waste disposed below 5 meters. This is a conservatism for Class C waste. This is even more conservative for disposal sites located in an arid environment, since the depth of disposal is generally much greater than for a humid site. Thus, waste containing higher concentrations of transuranic radionuclides may be placed at greater depths, further reducing potential intruder exposures due to transuranics. This would be relatively easy to accomplish at a western site because of the greater disposal depths possible.

Item 15 - NRC staff disagrees. The analysis shows that the dilution factor is warranted. In any case, Cs-137 is not a particularly long-lived radionuclide, and the additional dilution factor makes no difference in long-term impacts.

<u>Item 16</u> - Section 4 of Appendix D to the DEIS provides a more complete discussion of processing impacts considered in the DEIS and the data presented in Section 5 of Appendix G. In addition, Volume 2 of the <u>Data Base for Radioactive Waste Management</u> (Reference 3) contains additional information.

For purposes of analysis, NRC assumed that only incineration resulted in additional potential population exposure as a result of processing. Other processes, such as evaporation, compaction, solidification and packaging were assumed to result in no potentially significant additional population exposures to those already considered and analyzed as a part of each facility's license. Since there was no potentially significant incremental change from existing practice, NRC did not specifically analyze it.

Item 17 - The staff considers the separation to be physically possible, and the staff anticipates that in some cases Class A wastes may even be disposed of at sites separate from those for Class B and C wastes.

Item 18 - NRC recognizes that several waste streams and specific radionuclides, not addressed as a part of this first effort, need to be addressed to the extent possible. NRC has attempted in this FEIS to address radionuclides and daughter radionuclides present or expected to be present in waste streams to be generated over the next 20 years. In addition NRC plans to further analyze specific waste streams (e.g., certain reprocessing and decommissioning streams) as a part of subsequent work to that already performed. Such work may result in amendments to Part 61 for disposal of such wastes. . • •



.........

State of California

January 12, 1982

EDMUND G. BROWN JR.

R. Dale Smith, Chief

and Safequards

Low-Level Waste Licensing Branch

U.S. Nuclear Regulatory Commission Washington D.C. 20555

Division of Waste Management Office of Nuclear Material Safety

AVCATHENLO FERIA

Dear Mr. Smith:

We are pleased to transmit the comments of the State of California on the Nuclear Regulatory Commission's proposed rulemaking on land disposal of low-level radioactive waste [10 CFR Part 61] and the related draft environmental impact statement (EIS). Because these two documents are closely related, this letter transmits comments on both. However, to facilitate differentiation of the issues, each will be discussed separately.

#### PROPOSED RULEMAKING ON LAND DISPOSAL OF LOW-LEVEL RADIOACTIVE WASTE (10 CFR Part 61)

The comments of various State agencies on the proposed rulemaking are included below. I would like to raise the following additional points.

1) Part 61.52(a)(6) states that the "waste must be placed and covered in a manner that limits the gamma radiation at the surface of the cover to levels that are within a few percent above the natural background levels of the site" (emphasis added). This terminology is unnecessarily vague. We recommend that the term "a few percent" be replaced with a specific number.

2) Part 61.55. With regard to waste classification, we share the view that segregating waste into different classes can be beneficial from the standpoints of protecting public health and maximizing disposal economies. However, the language of the rulemaking does not appear to definitively rule out the possibility of a "low-level" disposal site for transuranic-contaminated waste. Part 61.55(d) states that waste with a concentration exceeding the values shown in the accompanying table is not acceptable for near-surface disposal

• ,

**.** .

• -

R. Dale Smith

. .

"without specific Commission approval pursuant to Part 61.58 of this Part". Part 61.58 is one very brief paragraph which appears to permit the Commission to authorize "other provisions for the classification and characteristics of waste on a specific basis" measured against criteria and an evaluation process which are not specified in any way. It is therefore unclear to us under what conditions the Commission would exercise its authority under "Part 61.58 and whether or not waivers could be granted for waste that exceeds the concentrations in Column 3 of Table 1. Our view is that transuranic-contaminated waste should under no circumstances be considered low-level waste and should not be included in low-level waste disposal sites. Rather, this material should be disposed of at the specificallydesignated sites operated by the federal government to receive and dispose of transuranic-contaminated waste.

-2-

3) Part 61.54, similarly, in one brief paragraph appears to permit the Commission to authorize provisions other than those set forth in detail in Parts 61.51 through 61.53 of the proposed rulemaking without any discussion of the considerations that would go into such a decision. One can only wonder why the specific elements of this rulemaking are included if the Commission is empowered at the same time to unilaterally change the requisite requirements for <u>segregation and disposal</u> of waste on the basis of what appears to be an arbitrary finding. This part should be clarified or deleted.

4) Part 61.62 -- Funding for disposal site closure and stabilization. The financial arrangements, while on the right track, do not appear to us to be sufficiently comprehensive. It is unclear whether the annual review by the Commission of the financial arrangements would include the requirement that the size of the post-closure funding be increased on an annual basis to account for inflation and unforeseen problems and costs. The financial surety arrangements mentioned in subparagraph (g) (surety bonds, cash deposits, certificate of deposits, etc.) are not instruments which increase in value over time to compensate for the effects of inflation. We suspect that a device such as a sinking fund would be a preferable vehicle, but this receives no mention. Additionally; there is no mention whatsoever of the funds that would be required or the source of such funds if problems should occur at the site which would require considerably greater post-closure expense than that budgeted on an assumption of normal operation. Surely the experience of the State of Kentucky with Maxey Flats testifies to the importance of making contingency funds available in the event that serious problems occur. This issue should be addressed and the rule-making changed accordingly.

5) Subpart F -- Participation by state governments and Indian tribes. We are disturbed by the tenor of Subpart F. As drafted, it appears to set up an adversary relationship

. . .

R. Dale Smith

January 12, 1982

between the states and the federal government. Rather there should be full cooperation between state government and federal agencies in all phases of low-level waste management. This rulemaking as drafted does not lay the groundwork for that cooperation. For example, the state proposal for participation required under Part 61.72(b) and (c) calls for a submission by the state of various specific items of information at a time so early in the process that all the state's concerns may not yet be apparent because of lack of information. While we understand the Commission's desire to avoid an unnecessarily protracted participation by a hostile state, nonetheless the legitimate interests of state governments should be accommodated in a more thorough and flexible manner. The regulations as drafted do not accomplish this. Note also in Part 61.71 the statement that "upon request of a state or federal government body, the director may make available Commission staff to discuss with representatives of the state ... " (emphasis added). As a first step in the right direction, surely the word "may" should be replaced with "shall." This section should be com-pletely revised to facilitate collegiality between the federal government and the states.

-3-

6) Part 61.82 -- Commission inspections of land disposal facilities. We thoroughly endorse the notion that the Commission should be afforded an opportunity at all reasonable times to inspect radioactive wastes and the premises, equipment, etc. An explicit provision should be added that host states enjoy a similar right.

THE 7) Part 2, Subpart 2.754 (a) (b) (c). The intent and consequences of these parts are unclear. They appear to authorize an initial decision by the Commission that would preclude effective appeal by either a concerned party or state. Immediate affectiveness, as it has been implemented by the Commission in reactor. licensing, has had the effect of denying states effective participation, discouraging cooperative efforts between "state and federal governments, and rendering state's appeals "ineffectual, since a facility would commence operation before appeals had run their course. The implications of these sections, should be clarified.

... Below you will find additional detailed comments of various State agencies on the proposed rulemaking. With regard to site suitability described in Section 61.50(a)(5), the criteria should be changed to require a lower risk of flooding. Currently, the section would allow waste disposal in a floodplain that is likely to be flooded less than once every 100 years. The current ratings of flood risks are crude at best. For example, in California we have had floods rated as a 100-year flood and as a 300-year flood, both within the last 25 years. This experience has led many people to suggest that our estimates of flooding potential are much too low.

-4-

Based on the California experience, we would suggest that the 100-year floodplain discussed in the regulation should be increased to at least a 300-year floodplain and, preferably, to a 500-year floodplain. Where the purpose of the disposal site is to keep the wastes isolated for a period of at least 500 years, surface flooding of that site should be avoided within our best estimates of what would be likely to happen within that 500-year period. The experience at Maxey Flats, Kentucky, should convince people that flooding of the disposal site should be avoided.

Third, the performance objective in Section 61.2 concerning protection of individuals from inadvertent intrusion should be strengthened. Some kind of permanent sign or warning device should be in place at the perimeter of the site. The warning sign or symbol should be designed to last 500 years and to remain effective as a communicator, even if the language spoken in the area changes within that period. An example could be a combination of the skull and crossbones and the symbol for nuclear radiation.

Fourth, although the regulations describe minimum requirements for waste characteristics to be accepted at a disposal site, the regulations do not appear to require some kind of checking of the condition of the materials at the site. A site could experience the problems found in the past when sealed steel drums were delivered for disposal and no one knew what the drums contained. If there is no program for checking the contents of the drum, either at the site of origin or at the disposal site, the requirements for waste characteristics may well be ignored by many of the waste generators.

#### OFFICE OF EMERGENCY SERVICES

Following closure, the draft assumes the State becomes the site owner (pp. 3-36 of NUREG-0782, V-2.). However, paragraph 61.59 of Part 61 states either the state or the federal government shall become the site owner. Governmental ownership is certainly desirable; however, the apparent conflict should be clarified.

-5-

Paragraphs 61.1 and 61.3 indicate that licenses will be issued by the NRC. In paragraph 61.70 through 61.73, provision is made for a state or tribal government to participate in the licensing process, yet it is quite clear the NRC retains sole authority to issue the license. This suggests that a local jurisdiction has neither a voice in determining whether or not a site is established in their locale nor the conditions under which it is established and operated. The NRC should take steps to facilitate participation by affected local governments, including consideration.

If the State government has little or no real control during the functional life of the site, there is some question whether it would wish to assume responsibility for the site when it was closed. This would be especially questionable if the new site owner (i.e., the State) was expected to fund the cost for maintenance and monitoring.

Although several methods are mentioned for providing funds to the institutional authority, the rule makes no provision for it. In fact, the Commission admits it has no authority to "...require land disposal facility licensees to provide financial responsibility for activities occurring after the original licensee's responsibilities have ceased and the license has been transferred to another party." We would suggest the Commission ask Congress for authority to require financial assurances for licensees for the active institutional control period.

For additional comments please see Part 7.2 below of the comments on the environmental impact statement.

0

-6-

2 1

-1

.73

.

- 1

4

-

1

#### DEPARTMENT OF CONSERVATION

The California Department of Conservation (CDC) has reviewed the subject document for its geotechnical and procedural aspects. We. . .feel Section 61.72 is very important, providing for State participation in the review of any license application that affects the State. These procedures are very important to assure a real opportunity for the states, and thereby any affected local government, to have an effective input in the lowlevel waste (LLW) disposal process and specific site decisions which inevitably will impact all "host" states.

However, we believe that there is a significant defect in Subpart D. Subsection 61.50, <u>Disposal site suitability</u> requirements for land disposal. These requirements will not provide adequate protection to usable groundwater or to the environment from radionuclides that could be transported from the site by groundwater.

None of the stipulations in the disposal site criteria refer specifically to preventing migration of radionuclides into usable groundwater. Item (7) in Subsection 61.50 states. The disposal site must provide sufficient depth to the water table that groundwater intrusion, perennial or otherwise, into the waste, will not occur. The Commission will consider exceptions to this requirement if it can be conclusively shown that disposal site characteristics will result in diffusion being the predominant means of radionuclide movement and the rate of movement will result in the performance objectives of Subpart C of this part being met."

Our concern is that the above-quoted stipulation is concerned only with groundwater intrusion into the facility and, furthermore, would allow diffusion of radionuclides in groundwater as an acceptable concept in the disposal of waste.

What is lacking in these criteria is the fail-safe approach to planning and design. The uncertainties inherent in geologic, design, and operational factors for any LLW site cast serious doubt on the assumption that the wastes can be guaranteed to be isolated for the prescribed time. If radionuclides should prematurely escape from their confinement at the site, it would be difficult and expensive, if not impossible, to prevent their contaminating the groundwater. Therefore, CDC recommends that Item (7) in Subpart D, Section 61.50 be rewritten as follows:

....

#### Department of Conservation (continued)

1 1

The disposal site must not be located 1/ within basins containing usable groundwater or their recharge areas, or 2/ within geologic formations which will permit the diffusion of radionuclides to the environment, or their transport by groundwater to a degree exceeding the performance objectives of Subpart C.

-7-

We recognize that the adoption of this recommendation will have the effect of decreasing the number and size of the search areas which would be eligible for consideration as potential LLW sites. Nevertheless, we believe that the seriousness of the risk of any radioactive contamination of groundwater warrants this degree of effort to assure that even if radionuclides were to escape, they could not contaminate any usable aquifer.

The regulations also fail to specify in Subpart G, Subsection 61.81 the nature and extent of Records, Reports, Tests and Inspections which will be required to ensure compliance with Subpart D - Technical Requirements for Land Disposal Facilities. Greater specificity is necessary regarding geologic, hydrologic, and other types of surveys and/or research to determine that potential sites comply fully with the regulations.

#### DRAFT ENVIRONMENTAL IMPACT STATEMENT ON 10 CFR PART 61 "LICENSING REQUIREMENTS FOR LAND DISPOSAL OF RADIOACTIVE WASTE"

The Draft Environmental Impact Statement Is an important accompanying document, without which the proposal rulemaking would be difficult to assess. Comments from several state agencies are included below. First, however, I would like to make a few additional points.

-8-

1) Part 2.3 -- Alternative Disposal Methods. The EIS discusses briefly ocean disposal of low-level wastes. Although this disposal alternative is not addressed in detail within the EIS, we want to express our opposition to the use of the oceans for disposal of low-level wastes.

2) Part 4.6.1 -- Institutional Control Requirements. We support the concept of permitting disposal of low-level wastes only on land owned by the federal government or by the states, since the need for control of near-surface disposal facilities will last, in some cases, for several hundred years.

3) Part 5.5.1.2(2) -- Site Characteristics. With regard to the location of future sites; we believe the criteria should be changed to require a lower risk of flooding. We reiterate our comment (see comments above from The Resources Agency) that the 100-year floodplain may not be conservative enough. We suggest that a 300-year floodplain or, preferably, a 500-year floodplain be required to avoid surface flooding of a site.

X)

(૬)

4) Part 5.5.1.3(2) -- Design and Operations. We share the view that prior to any license application, the applicant shall gather information concerning "the ecology, meteorology, climate, hydrology, geology, and seismicity of the site." However, we disagree with the requirement that "for those characteristics that are subject to seasonal variation, data shall cover at least one full year." We believe this should be strengthened. Any locale's susceptibility to changing environmental factors requires that an attempt be made to gather historical data so as to try to accurately reflect how a proposed site has changed over time. We suggest that this section be amended to require collection of historical data going back a reasonable period of time, to the degree such collection is feasible.

5) Part 7.2 -- Waste Classification Based Upon Consideration of a Potential Inadvertent Intruder. The discussion of financial requirements during the operation of the

• . . .

postclosure period touches on most of the relevant issues. However, there is a lack of depth to the analysis, and adequate solutions are not suggested for problems that have been identified. For example, per our comments above, it is clear that a sinking fund or some similar financial assurance mechanism would be the most preferable alternative for ensuring that necessary funds will be available for the lifetime of the site (i.e., including postclosure lifetime). Steps should be taken by the Commission to seek the authority to explicitly require that a sinking fund be established. Instead, the document endorses less satisfactory alternatives while at the same time the Commission recognizes the shortcomings of this approach. Additionally, the EIS, like the draft rulemaking, fails to account for the possibility of serious problems occurring at the site. It does not make contingencies for ... such problems or for the costs which a state would no doubt incur if such problems occurred. This is a major failing of the document and should be rectified. Costs and cost estimates should reflect the possibility of a serious failure of the site -- a failure of greater consequence than those that have already occurred at existing sites.

6) Part 7.2.6 -- Transuranic Isotopes. We support the retention of the 10 nanocurie per gram limit for surface disposal of low-level waste. We believe that wastes that exceed this limit should not be considered low-level waste and should not be buried at commercial low-level waste disposal sites.

7) Part 8.4 -- State, Tribal, and Public Participation. We would like to reiterate our point made earlier in Part 61.71 of the proposed rulemaking that there should be full cooperation between the state and federal governments in all phases of low-level waste management. This cooperation will strengthen the working relationship between the states and the federal government and thereby facilitate the safe establishment of necessary new disposal sites.

Below you will find additional detailed comments of several state agencies on the draft Environmental Impact Statement.

#### STATE WATER RESOURCES CONTROL BOARD

#### General.Comments:

 In California, disposal to land of all but very low level radioactive wastes is prohibited by state law (California Administrative Code, Title 17, Section 30288, attached).

-11-

The entire document fails to emphasize the need to prevent significant movement of pollutants from the disposal site to underlying ground water. The place- 2 ment of an impervious cap over the waste will not preclude gravity drainage of liquid pollutants through a pervious trench bottom. Further, if the trench walls contain pervious beds (even lenses or "stringers"), water from precipitation or other nearby sources can move laterally into the trench, leach out pollutants, and then percolate vertically to underlying ground water. These ground water pollution threats can be by essentially precluded by requiring disposal trenches to have impervious bottoms and sides. An engineered impervious barrier such as a clay liner could be reguired\_for\_each disposal trench. Better yet, the trench site should be in an area having a substantial thickness of clay. (See Class I Disposal Site Criteria, California Administrative Code, Title 23, Section 2510.)

::::-

((D)

#### Specific Comments:

- 1 <u>Summary, rage:11</u>: The abbreviations, "PWR" and "BWR" should be interpreted (re Report Page 3-10, bottom).
- Report, Pages 10-6 and 13 et seg. The southwest "hypothetical regional site" is described as serving the western half of the country. The "High Plains" location, however, is far from the significant concentration of nuclear generating facilities on the west coast. It would be more appropriate for the western hypothetical site to be located near the west coast facilities.

#### 3. Proposed Rule 10 CFR Part 61:

- A. Section 61:40 sets standards to avoid excessive exposure to humans. Excessive exposure to animal life should be avoided also.
- B. Section 61:50 should include criteria requiring impervious material (natural or "engineered") beneath and along the sides of all disposal trenches.
- Attachment

(5)

m.

6

 $(\mathbf{7})$ 

#### SUBCHAPTER 15. WASTE DISPOSAL TO LAND

#### Article L. General Provisions

Arrice 1. Central Providing 2500, Definition of Terms. (a) "Disposal site" means any place next for the dispusal of solid or liquid vastes. It does not include any part of a sewage treatment plant or point of discharge of sewage efflu-ent or fand drainage from pipes or ditcless into waters of the state, (b) "Disposal acca" is that portion of the site which has received or is reservicing water. (c) "Lawdrate" iterating from the waste or fluid resulting from the periodation of liquid through a waste substance, (d) "Usable" ground or sufface water includes potentially usable water.

### Article 2. Classification of Waste Disposal Sites

Article 2. Classification of Wasto Disposal Sites 2510. Class I Disposal Sites. Class I disposal sites are howe at which complete protection is provided for all time for the quality of ground and surface waters from all waters deposited therein and against hazard to public health and wildlife resources. The following criteria must be met to qualify a site as Class I disposal and gases channeling a site as Class I (a) Geological conditions are naturally expalse of preventing verti-cal devolution of the set and usable surface or groundwaters. (b) Geological conditions are naturally expalse of preventing lateral hydraulic continuity between liquids and gases channeling from the waste in the site and usable surface or groundwaters. (b) Geological conditions are usturally capable of preventing lateral hydraulic continuity between liquids and gases channeling from wates in the site and usable surface or groundwaters, or the disposal area has been modified to achieve such capability. (c) Underlying geological formations which contain rock fractures or fissures of questionable permeability must be permanently scaled to provide a competent barrier to the provement of liquids or gases from the disposal wite to usable waters. (d) hundration of disposal areas shall not occur until the site is closed in accordance with requirements of the regional board.

¢٨

# (172.12.37 phaglaton 74, Ho, 17-32670)

2261

(b) Excreta from hulo iduals under going medical diagnosis or the a-py with radioactive material shall be except from any limitations of the tancel in this section, provided that the user provides for appropriate radiological monitoring whenever any waste fine in the user's instan-tion which may carry such exercts is opened

BADSATION.

tion which may carry such exercts is opened 20285. Disposal by lurial in Soil. (a) No user shall dispose or the disactive material by bural in soil unless: (1) the total quantity of radioactive material buried at any tree location and time does not exceed, at the time of barial, 1,000 unlest the amounts specified in Section 20286, Appendix II; and (2) burial is at a minimum depth of four feet; and (3) specesive burials are separated by disfances of at least six feet and not more than 12 burials are made in any part.

th) The department will not approve any application for heense : (b) The department will not approve any application for heense : tercuve radioactive material from other persons for dispession lang owned by the Federal or State Covernment.

30259. Treatment or Disposal by Incineration. No user shall treat or dispose of radioactive material by incineration except as specifics, approved by the department pursuant to Section 3033. Norr: Authority side, Section 102, 394, 2541, Health and Solety Under History: 1. New section filed 7 22-31, effective themeth day thereafter effective No 30).

Article 6. Becords, Reports and Notifications

Article 9. necority, imports and Nonitrations 10237. Records: (a) Each user shall maintain accurate and com-plete written records, as follows: 11) The results of each required calibration, survey and test, (2) Each receipts, transfer, and disposal of a source of radiation. (3) Radiation caposures of all individuals for whom personner monitoring is required under Section 20216. Exposure records the lockept on department Form IIII-2365 or in a manner which incluors all the applicable information required on said form. Each entry shill be for a period of time not exceeding one calcular quarter. (4) Results of medical examinations and bio-assoy's pursuant 4; Section 30277.

Section 30277.

(b) Each required record of dose received by individuals and re incidical examination and bioassay shall be preserved indefinitely re until the Department authorizes their disposal. Each other requirez record shall be preserved for a period of three years following the date of the occurrence that it the subject of such record.

Sent by Bill Friedman, Dat Health 5-0331 3-2757

(c) Disposal areas shall not be subject to washant.

Leachate and subsurface flow into the disposal area shall be contained within the site unless other disposition is made in accordance with requirements of the regional board.
Sites shall not be located over zones of active faulting or where all focures forms of geological change would impair the competence of natural features or artificial barriers which prevent continuity with washe waters.

ral features or artificial barriers which prevent continuity with usable valers. (1) Siles inade suitable for use by man-made physical barriers shall ind be located where improper operation or maintenance of such struc-tures could permit the waste, leachate, or gases to contact usable fround or surface water. (1) Siles which comply with n, b, c, c, f, g, and h but would be subject to innondation by a file or a flood of greater than 100 year frequency may be considered by the regional board as a limited Class I deputa sile.

15. History: I. Amendment of inducetions (a), (b) and (c) filed 12 29-72; effective thirti-eth day thereafter (Register, 72, No. 53).

51

531

# 1TLE 17 Register 78, No. 13-8-28-78)

30350. Appendix II.

APPENDIX B						
Rulimertile	Minet.	Radusarius	7.4	Rollarschie	31.~~ ~~~	
A musicion 111		frie 131	19	54+++ \$#3	••••	
Fallener 131	100	Journal 117		Shot-SR.M.,	1	
Antiment 178	l ii l	Beatres (6 Laures	104	feine II		
Are se 13	100	long-33	140	tuine 21	H	
Armen 18	. 19	1-m-10		Steventierts 65	17	
Armain 18		Atyl ban 10		Pleasant Pl		
Aur -111	Î Ĥ	Inchange III	14	Fiberdam, W		
Barnard-133.manuaras	ы	8-ed.719	01	biraniare. ()	11	
Durum 140	. 20	adrings.119	190	Bulgher-37 anterestates bes	16.	
Fary Baland, same and	194	Plant serm 53		Tealahra 137		
long il	16	Manager	12	Jacks, man alm		
Coluins 100	Î	Meseres) (P).	100	Indermat-11	1.11	
Calenter-Hilm.	- 10	Mostury Minerica	196	Terlarburn ten	8.00	
[	140	Marine 121	. 16	Jerlashane 70	- He	
And the second second		A	144	Telluroute 1754s		
failes II		Sandy purson 114		tobellette forde anne i anne.		
Come Hil	100	Se 14 12		1.4.4.10.17.00		
Comme 113	110	Notel 45	10	1. Rodenter 1	lens.	
( 111		Sala Al	114	Johnson Illes	14	
Comm 131	1,000	Parenter \$1.00.000		Tilletrest. 132	14	
Country 111		Automotive bill	1 3	The Name 200		
Consta 173		Lintums 115	1 11 1	The line tel		
(mm-14		Ihreine 171 a	100	The Page 747	1.44	
Conser 1 17	30	1	104	Thallong 701	11	
Chierme 30	1 10	Chanople 103	100	Thereas tretwalt		
Charley 19		Follower WJ	114	Telester 199.	1 12	
Columb 1900-21, an anna		1 hours and 37		Tan 123		
Erdak St.	ើ	Distances (NI	104	T = 175	1.5	
C-444.00		Things. Itim	106	Tungeren INT.	11	
Cherges 64. despesses.	150	Petian Its	140	Jusesten 143	14	
Di aperante 163		Julman, N/m	108	Tabadra lat.		
17) 97 million 152		1 1610600 107		finbert forteral ar dejiendi,	۲×	
John Hite	1 1 1 1	Patroname 710		Premet. 732 - I cannon 713		
Tatopinen-1829.26	140	Fateren St	19	Torade res 19	1)	
3 mapres 152 13 27		[ ] hasan (presidents   ] .	114	Arma Lita.e.ceeeeeeeeeeee	104	
C WYPPIDE 13 Langester		119000-101900-010 .	[ 19]	Committee and a second	1 23	
Therefore it and a second	100	I turbe there \$ 15.	1 13	Similar 15		
Indulation Illegence		1 ation \$70		110	10	
Hadabarra 168	1 <u>110</u>	Reven 19-148	1 114	Dinter bl		
Conge It. conserver	1			Things W		
The second second		Rinduster Htt	(	2 3 Mellin 23, and a same as a same same .		
1244 199	110	Ret alien 18.	10.	Time Ites	1.5	
Hickory 181	£ 14	Fubreiren 17	10	1	1144	
Il	1 12	Ball rouge 17	1 174	Incorden 13	1.0	
Histogra L	11.11	Kull.reses 283	1 11	2 Termien 75	{ <u>!!</u>	
Indone 1140	1 31	Bathening 104	1 7	I are shown by	}	
Jackson 111-1	1 124	Comerine 111	1 16	and lated shoes at automat of	}	
Inlian 115	1 10	Tomariam 112	100	alpha se ittere af und sura a sette.	1	
fotos 175	1 1-	Frendrum (S	E .10	for a second second second second	M. J	
100mt 179	1 1.	To be made 18	1 10	And the star and the series by a first		
Same 11	ι Τ'	Patrone 14	1	emeridad in factory bars, and bat.	1	
B-Lat 112	1 1	Adapted 11	1.00	the of a base of the state	6	
Judn e-133	1 1	}	1	1	1	
Will I'r fwr ymgan a d fintiwn 2017, 2017, 2019 and 2019 aller Here a mediad a simdiaut in af rafna						

RADIATION

which is receive another to be at a train of the state of the Markov set of the state of the st

#### -15-Department of Conservation (continued)

along with uncertainty in the capability for adequate enforcement of the regulations relative to proper packaging and disposal, we recommend that item 7 in Subpart D, Section 61.50 of [the proposed rulemaking for] 10 CFR Part 61 be rewritten as recommended above.

-14-

#### RESOURCES AGENCY, DEPARTMENT OF CONSERVATION

The California Department of Conservation, based on review by the Division of Mines and Geology, has considered the Draft Environmental Impact Statement with respect to geotechnical aspects and procedural requirements.

In the DEIS, NRC discusses the use of high-integrity disposal package containers with extended containment life (approximately 300 years) for use in the disposal of highconcentrations wastes, as a waste processing option (DEIS, Ch. 5.2.4.8, App. D.4.3). This section also discusses potential use of similar containers for lower concentration wastes, but usage of this type of containerized disposal is not required by the proposed regulations. Also, for less concentrated wastes, the proposed regulations appear to require that the disposal package containers maintain their integrity only during the operational phase of the disposal site trenches (DEIS, App. D.4.3). However, we feel that because the less concentrated wastes could still release radionuclides similar to, or even the same as, those contained in the waste packages for high-concentration wastes, container integrity is essential to preventing the release of radionuclides into groundwater (prior to adequate confined decay time) to insure that the resultant activity level is low enough to not pose a danger to public health and safety.

As discussed in the DEIS, the proposed regulations in 10 CFR Part 61 assume that in the event of early release of radionuclides from disposal containers, or from decontainerized disposal, the site design, including the geologic setting, should be capable of preventing radionuclide migration out of the disposal trenches and into the surrounding groundwater and environment. However, the proposed regulations provide no fail-safe assurance that this will be the case.

Even if the wastes were to be segrecated according to the active life of the different radionuclides and disposed of in containers which could maintain their integrity for the necessary containment time of each of the different classes of radionuclides, there does not appear to be adequate provisions in the proposed regulations for enforcement of this degree of detailed inspection during waste processing. We feel that the potential for migration of radionuclides from the disposal site and subsequent contamination of groundwater in the vicinity of the disposal sites could, coupled with adequate site planning and design, be minimized by containerized disposal of wastes in containers capable of maintaining their integrity for a minimum confinement period of 100 years [10 CFR Part 61, Subpart A, 61.7 (4)]. However, due to the lack of provisions in the regulations to require containerized disposal of all wastes.

0

(H)

(13)

and a second second

#### DEPARTMENT OF HEALTH SERVICES

We continue to be troubled by the cost issues and their presentation in this EIS draft.

First, we are discouraged to find NRC using their own regions for the waste data bases. The states have been working for more than a year now with regions and waste volume projections based on U.S. Department of Energy (USDOE) studies. Comparison, then, with the USDOE data becomes difficult or impossible. However, because we know something of USDOE's efforts, their strengths and weaknesses, there is a need for careful comparison of data and conclusions on such an important matter as this. The final EIS should facilitate those comparisons.

Secondly, costs are based on the 20-year period from 1980 to 2000. We think it important that costs be shown by year from 1986 (when exclusionary authority may be conferred by Congress) through 2000. For some regions (as defined by current state actions, or the USDOE), initial costs may verge on prohibitive. A review of USDOE data indicates that by 1986 only three regions would generate the volume of waste on which the EIS was based. One, Region 5 (USDOE), would not have the waste volume by the year 2000 (see Table I).

Given the history of some existing disposal sites, one key concern should be the assurance of adequate financial resources on the part of the applicant to construct and operate a disposal facility and to provide adequate financial provisions for site closure and long-term care.

The EIS, although it cites no specific cost figures, appears to underestimate the short- and long-term costs of operating and maintaining a low-level waste disposal site, and fails to recognize the problems small companies (as identified in the EIS) have in meeting financial requirements in operating a waste disposal site. It seems likely that few small companies can raise the necessary capital for plant development, set aside trust funds, cash deposits, purchase surety bonds against short-term financial needs and further set aside additional money for 100-year care costs within the life span of the disposal site. The most careful attention should therefore be paid to the financial resources of any applicant who seeks to develop and/or operate a new site.

The "unanticipated contingencies" not addressed by the EIS (i.e., problems occurring at a site) should, we believe, be explicitly addressed either by the NRC or the Congress. To the extent that all national sites meet or exceed a

**W** 

65)

(16

17

8

#### -17~ Department of Health Services (continued)

common design and performance standard, the Congress might accept such a responsibility. That uniformity might, however, require some special handling as was done for uranium mill tailings.

To summarize, the EIS should contain a section specifically developed for informing the Congress on the impact of its impending action as authorized in PL 96-537. That section would chart waste disposal needs and costs by regions as they actually exist or are planned by the states. The conclusion of such a piece might well be that the implementation date of 1986 is too early in terms of waste volume, and unaffordable. Additionally, given the amount of time necessary to bring new sites into operation (4-7 years), the 1986 date in PL 96-537 may be premature, if safe management and disposal of these materials is to be assured.

# 

...

#### -18-

#### R. Dale Smith

÷.

····

.

Thank you for affording us the opportunity to comment of these documents. This is a most timely issue, and one which we are sure with benefit from the careful attention and input provided by all interested parties.

-19-

Sincerely, Whigh the heren

Phillip'A. Greenberg Assistant to the Governor for Energy and Environment

An and a set of the set of the

.

1

Fraction of Representative Site Waste\_Volume by Region\*

TABLE I

	1986	2000
Region 1	1201	230
Region 2	461	971
Region 3	140	2901
Region 4	871	1801
and the second		. `
Region 5	198	40%
en en production de la construction		· .

 Region 6
 52%
 84%

 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 1
 <t

.

· .

\*Reference: Low-Level Radioactive Waste Management Report, prepared by the

U. S. Department of Energy, March 13, 1981.

.

A-63

#### Docketed Comment Number: 26

Commenter: State of California, Governor's Office

<u>Response(s)</u>: Item 1 - There is no conflict between §61.59 of the rule which states that either the state or the federal government shall become the site owner and the assumption on page 3-36, Volume 2 of the draft EIS that the state would be the site owner. The rule identifies the two parties who may become the site owner and the draft EIS merely assumes for the purposes of analysis one of those two parties.

Item 2 - Ocean disposal of low-level waste was discontinued in 1970. Responsibility for licensing this disposal method, should it be resumed, rests with the U.S. Environmental Protection Agency as a result of the Marine Protection Research, and Sanctuaries Act of 1972. The state's opposition to this disposal method is noted.

Item 3-- The staff considers the 300- or 500-year floodplain to be too restrictive since the major impact of flooding is inundation of disposed units which have not been covered or stabilized. Part 61 requires that each disposal unit be closed and stabilized as it is filled and covered. Thus, each disposal unit will be open a comparatively short time. In addition, Part 61 requires that disposal unit covers direct surface water and infiltrating water away from the waste and that the site be designed to eliminate the contact of percolating or standing water with the wastes after disposal. Therefore, given the short time frame that disposal units will be open, the staff considers the 100-year floodplain (Executive Order 11988, Floodplain Management Guidelines) to be adequate protection against inundation. Other site flooding will be handled by the site drainage system.

Item 4 - The staff considered this requirement and determined that the one year period is adequate for data collection. However, in branch technical positions and regulatory guides under development, the staff will advise applicants to collect existing information from government agencies and other resource centers to allow a determination to be made as to the representativeness of the year's field data.

<u>Item 5</u> - The staff is aware that explicitly requiring a licensee to establish a sinking fund for long-term care would be the most effective method of ensuring that sufficient funds be available for long-term care. However, at this time, NRC lacks the statutory authority to require that a long-term care fund be established. The Commission has presented testimony requesting this authority. Until such authority is granted, the Commission cannot require licensees to establish a long-term care fund. ...

The amount of financial assurances that a licensee is required to provide for closure and post-closure care will be determined by the Commission during the licensing period. The Commission will ensure that such funds are adequate and will consider the position of the state and any other parties involved as to the amount and type of the licensee's financial responsibility for closure and post-closure care.

With respect to financial responsibility for long-term care, the amount and type of coverage would be established between the licensee and the site owner in a lease or other type of binding arrangement. The Commission would review this document at the time of licensing and periodically thereafter to ensure adequate financial responsibility for costs during the long-term care period.

Item 6 - A considerable number of comments were received on the limit on near-surface disposal proposed in the draft rule for transuranic radionuclides. These comments were received from persons addressing the draft Part 51 rule as well as those addressing the rationale set out in the draft EIS. Due to this interest, NRC has reevaluated this limit and has determined that the 10 nCi/gm limit may safely be raised to 100 nCi/gm for wastes in which transuranic nuclides may exist in only trace amounts. For example, measurement of transuranic content in wastes from nuclear power plants have indicated that the TRU content is typically well below 10 nCi/gm and only occasionally in the 10-100 nCi/gm range. These latter occurrences have furthermore been associated with past incidents of failed fuel. Fuel performance has since generally improved. For wastes in which the transuranics may be the principal radionuclide within the waste (e.g., from decommissioning former MOX fabrication facilities), additional criteria would be imposed. The disposal limit would be placed at 100 nCi/gm, but such waste would be required
.

Item 7 - NRC staff agrees the states will have a major role in the development of new sites. Under the Low-Level Radioactive Waste Policy Act of 1980 (P.L. 96-573), the states are responsible for providing for the availability of adequate disposal capacity. The Act provides for the establishment of regional compacts to meet this responsibility. Proposals for new sites should come from Policy Act activities and the siting arena which the Act establishes. Thus the state will be involved from the beginning and NRC plans to work closely with the states in the licensing process.

The state will also likely be the landowner and provide institutional control after site closure. The lease and other arrangements made to fulfill the institutional requirements of proposed Part 61 will afford another means of early state involvement. These arrangements also afford a means of continuing state involvement during operation and closure of sites. <u>Item 8</u> - The use of impervious clay liners will trap any infiltrating water in disposal trenches leading to soaking of wastes in standing water and the need to periodically pump and treat the trench waters thus created. NRC does not believe that leachate collection systems and water

treatment are acceptable means of disposal and lead to the need for long-term active maintenance water treatment are acceptable means of disposal and lead to the need for long-term active maintenance. The need for such maintenance is contrary to the performance objectives of Part 61.

 $\underline{Item \ 9} - "PWR" \ is \ an \ abbreviation \ for \ pressurized \ water \ reactors \ and "BWR" \ is \ an \ abbreviation \ for \ boiling \ water \ reactors.$ 

. -

 $\underline{\text{Item 10}}$  - The location of the hypothetical southwest regional site was chosen to be roughly central relative to existing and future nuclear generating capacity in NRC Regions IV and V.

Item 11 - A preferred alternative identified in the draft EIS was that all waste should be placed into a stable form or container to eliminate the need (and cost) for long term active maintenance. It was not selected, however, because stability was not warranted for all wastes on the basis of public health and safety protection and because of the high cost to small waste generators of having to place low activity wastes into a stable form. NRC selected the alternative of only requiring higher activity wastes to be placed into a stable form. The concentration limits calculated for the low activity wastes were determined on the basis of their disposal in an unstable form and their not resulting in the Part 61 performace objectives being exceeded. .

<u>Item 12</u> - The state is correct in its assumption that the proposed regulations do not provide fail-safe assurances. The rule's basis is that waste form, site characteristics, site design and site operation and closure constitute an interactive system which will provide a reasonable assurance that the performance objectives of Subpart C will be realized.

Items 13 - See response to Item 11 above.

Item 14 - The work done in projecting waste data as a basis for the rule was begun well in advance of passage of the Low-Level Radioactive Waste Policy Act of 1980 and subsequent waste volume projections by DOE. Moreover, NRC's regional waste projections were made in support of the development of a regulatory program for land disposal of radioactive waste and for this reason the staff considers the use of these projections appropriate.

Item 15 - The staff believes that a year-by-year breakdown of costs will add little to the overall evaluation of benefits and costs associated with the proposed 10 CFR Part 61. The purpose of the rule is to assure long-term public health and safety and environmental protection, and conclusions about the efficacy of the rule made on the basis of costs incurred in a single calendar year are not considered by the staff to be appropriate.

.

•

With respect to the commentor's observations on waste volume projections, the regional breakdown may or may not give rise to questions about the need for additional disposal capacity. However, the staff feels that waste volume will continue to increase nationally, and the proposed rule is considered necessary to deal with this increase in a manner consistent with NRC's statutory responsibilities.

Item 16 - The staff has reviewed the financial history of the existing disposal facilities, and shares these concerns for the need to require licensees to provide adequate financial resources for closure, postclosure, and long-term care. The proposed rules require licensees to provide financial assurances for closure, post-closure, and long-term care of the low-level waste disposal facility.

<u>Item 17</u> - The Commission staff agrees with the importance of having a licensee possess sufficient financial resources, and they will therefore examine the financial resources of the license applicant at the time of the license application review.

Item 18 - The proposed rule provides that responsibility for the costs of closure and post-closure care at a low-level waste disposal site will be determined during the license application review. The NRC will ensure that such funds are adequate and will consider the position of the state and any other parties involved as to the type and amount of financial responsibility the licensee should provide for during closure and post-closure care. Financial responsibility for all events during the long-term care period will be covered in the lease or other type of binding agreement between the licensee and the site owner.

The staff also thinks that Congress, in passing the Comprehensive Environmental Response, Compensation and Liability Act (CERCLA) has already enacted a program to provide for financial responsibility for the cleanup of unanticipated contingencies at a low-level waste disposal site, such as a release of radioactive materials.

. .

Item 19 - As stated in the introduction, the scope and purpose of this EIS was to examine and provide a decision basis for the requirements in Part 61 to ensure safe disposal of LLW. It is not a generic EIS and it is not intended to provide a planning basis for the identification and selection of new sites. Such work is being performed by DDE, and NRC has not and does not plan to prepare its own independent assessment. NRC and DDE have cooperated closely in this area and plan to continue to do so in the future.

.

. .

DURE POWER COMPANY : Power Building 422 South Crutech Street, Charlotte, N. C. 20242

WILLIAM & PARSER, JR. Vice President Stear Production January 13, 1982





· · · ·

Secretary of the Commission U.S. Nuclear Regulatory Commission Washington, D.C. 20555

Dear Mr. Secretary:

Subject: Licensing Requirements for Land Disposal of Radioactive Wastes; Proposed 10CFR61 Supplemental Duke Power Company Comments File: GS-811.02

Duke Power Co. is pleased to supplement its previous comments, dated October 23, 1981, on this subject and on the Draft Environmental Impact Statement NUREG-0782, in accordance with the additional opportunity presented to the public to do so.

First, we want to inform you that we generally subscribe to the comments submitted to you on this subject from the AIF and the EEI/UNWMG.

Our most important comment on the DEIS is that in no way neither technically, nor quantitatively, does it support the values listed in Table 1 of the proposed regulation (i.e., the values in the Table are not derived in the EIS). This, of course, calls the conservatism of the Table into serious question as well as some of the conclusions reached which are reflected in the proposed regulation.

Although originally written for somewhat different purposes, we also want to include as our comments the information presented on the two enclosures to this letter. One item in the enclosures describes a mechanism by which the NRC can establish a generic "de minimus" category for low level contaminated trash.

Thank you for the opportunity to submit additional comments. We expect that they will receive full and careful consideration by the Commission.



## Docketed Comment Number: 27

Commenter: Duke Power Company

<u>Response(s)</u>: <u>Item 1</u> - Contrary to the assertion of the commentor, the values listed in Table 1 of the proposed Part 61 rule are derived in the DEIS-specifically in Chapter 7 of the Main Report (Vol. 2). Further examination of the values in Table 1 has been performed in the Final EIS.

. . .

.

## Docketed Comment Number: 28

. .

Commenter: Arkansas Power and Light

<u>Response(s)</u>: This comment was inadvertently docketed for both the rule and the EIS, although the comment addressed only the rule. The commenter's concerns were reviewed and responded to in the development of the final rule.

.

## Docketed Comment Number: 29

Commenter: Stone and Webster

.

<u>Response(s)</u>: This comment was inadvertently docketed for both the rule and the EIS, although the comment addressed only the rule. The commenter's concerns were reviewed and responded to in the development of the final rule. •

Docketed Comment Number: 30 Commenter: Health Physics Society

.

<u>Response(s)</u>: This comment was inadvertently docketed for both the rule and the EIS, although the comment addressed only the rule. The commenter's concerns were reviewed and responded to in the development of the final rule.

 A second sec second sec

and the second second

A-69

•

and the second second second second second

.



Betty Johnson 1907 Stratford Lane Rockford, IL 61107

COMMENTS ON DRAFT ENVIRORMENT& IMPACE STATEMENT ON CFR PART 61 "Licensing Requirements for Land Disposal of Radioactive Waste" NUREG -0782, vol. 1-4 DISCRET

Because it may help prevent repeating serious mistakes made in the past at low level waste land disposal sites, it is highly desirable to have specific requirements for land disposal of radioactive wastes-specifically for LLW in shallow burial sitesas proposed in 10 CFR Part 61. Mistakes made at Maxey Flats, Ky., Sheffield, II., West Valley, N.Y., Hanford, Wa., and other burial sites have seriously affected the health and safety of the public and it is important that they not be repeated.

Therefore I recommend that the most careful and strict requirements for compaction, stabilization, reduction of waste volume, and segregation of waste be used as outlined in these proposed regulations. In spite of these strict regulations it seems likely that groundwater, air, soil and human and animal life will be contaminated over the long term. This is likely because institutional arrangements cannot be made to last for the hundreds and thousands of years that longlived, toxic radioactive wastes must be isolated. These proposed regulations set up cost/benefit analysis and exceptions to regulations and minim amounts of transuranics and other long lived radionuclides that may be included in the waste stream<sub>g</sub>all of these may lead to pollution from these wastes. In this draft EIS groundwater contamination is considered to be the most dangerous. It is likely that this will occur because of the fact stated in this report that three fourths of the waste will be buried in a humid environment. Also many of the proposed solutions have not been and cannot be thoroughly tested for the length of time some elements in this radioactive waste will remain toxic.

term strategy that involves phasing out land disposal."

It is imperative that we find a place to put radioactive waste. Therefore, I would like to suggest use of the alternative of above ground «tructural containment as suggested in NREG/CR-0308 UC 209-02, "Screening of Alternative Methods for the Disposal of Low Level Radioactive Wastes". If possible sites used for above ground structural containment should be those that are already radioactively contaminated and also contaminated land areas that are no longer being used and could contain waste, such as civilian reactors, Manhatton Project and AEC facilities, surplus or obsolete DOE nuclear sites, and sput down nuclear reactors of utilities, such as Dresden I. This would offer the following advantages:

 Savings in cost of decontaminating and transporting wastes from these sites.

2) Use of existing setups for security and monitoring.

3) Mastes put in storage could be made retrievable; burial mistakes are more difficult to remedy.

4) These above ground structures could help prevent accidental intrusion because they could be easily identified, labeled, etc.

5) Past acceptance of the sites for radioactive type activities by local populations could minimize opposition and make site selection less difficult. 6) Valuable natural resources would be saved from radioactive contamination. Agricultural land . water, air and mineral resources, such as sait, that will be needed as world population grows would be protected from radiation pollution .

There are several hundred sites of this type which might be suitable, unless there was poor siting of these facilities from a geological (i.e. groundwater and earthquake) or social (i.e. high population areas like Zion, IL. or Indian Point, N.Y.) standpoint. One or more of these sites could be used for small experimental land burial sites to try out proposals contained in CFR Part 61. Large sites of even operating commercial reactors or other facilities could also be used for limited land buriel of LLW to minimize transportation and new site lacation. Then the

A-70

p. 2

p.4

p. 3

operating facilities on these sites are closed [i.e, reactors after 20 to 40 years of operation[they could be used as waste burial sites on either land or in structural containment.

## Additional Comments \*

- 3-30 Is a picnic ground in the planning map for this waste disposal site appropriate? This use could make intrusion more likely and the plannings could cause proprems.
- 4-6 Even unlikely "events" must be guarded against. Looking at 200 years of U.S. history and the record of nuclear facilities in a time span of less than 50 years shows loss of records, and many accidents and mistakes from unpredicted events.
- 4-12. How can it be assumed that these radiation impacts will be very small? Radiation is cumulative and concentrated as it moves up the food cabin, and new discoveries are still being made about its long-term effects.
- 4-34-37. Waste disposal sites should never be put on land that is needed for food production now or in the future. As world population grows every available bit of agricultural land will be needed.
- 5-16.1 agree that it is very important to prevent closing of landfill sites prematurely without sufficient funds to protect the site. Should insurance be mendatory?
- 5-117. Urea formaldehyde is toxic and should not be used for stabilization. Why add to the potential for pollution with another toxic material? Federal Resgister Proposed Rules:
- p. 38083--Says the inadvertant intruder can't be reasonably protected against and that it may be one or a few individuals. There could be many individuals, so this assumption is faulty. 55 mrem/yr maximum individual exposure limit is not acceptable for children who might be the intruders.
- p. 38085---Smoke detectors contin transuranic Am-241 in concentrations greater than 10 nCI/g and they are being generated in significant volumes. they should be required to be disposed of in low level waste repositories.

 Notations of sctions and/or pages placed by these comments are not cross-indexed and should be applied to other references to these same problems in the draft EIS. Institutional Control--There should be some sort of automatic signal for leakage required. As past experience has shown, operators of the waste disposal can't be depended on to report accurately and monitor carefully. Financial assurances--See comment on 5-16.

- p. 38085. Shipping regulations proposed are good, but may not preclude accidents. Preoperational Phase of site selection--There is no adequate provision for meaningful public participation. There should be a representative citizen's committee and mandatory public hearings at every step of the site selection and licensing. Site selection should be governed by Federal and state rules for hazardous waste disposal. The state and local governments must concur with the disposal option.
- p. 38088 How to minimize duplication with superfund requirements? Incorporate these requirements where possible and add those needed for long lived hazard of radioactivity.
- p. 38091. I question whether or not institutional control can be relied on for periods up to 100 years, and diposing of A and B waste without protection; past experience with sites less thum 100 years old find this did not work. Also 500 years calulation is not long enough for the most long lived waste.
- p. 38092. 61.53--Environmental monitoring. There is a need to also monitor the health of the human population in the area. The monitoring system must have an automatic alarm for early warning monitoring. Could the wate be color dyed or given an odor?

I participated in the 1980 workshop held by the Midwestern Regional Office of the Council of State Governments for the central region and Midwest., and I am pleased by the careful efforts that have been made to get public input on proposed CFR 61 regulations.

13 Bet

المان المحمد المحمد

Ø

Ю

Ø

Docketed Comment Number: 31

## Commenter: Betty Johnson

·- ·

<u>Response(s)</u>: <u>Item 1</u> - Based upon past experiences (both good and bad) and analyses of LLW disposal, NRC selected requirements leading to long-term site stability. These types of requirements have been found to work well at existing sites, (e.g., the Barnwell, South Carolina site). NRC recognizes, ho ever, the difficulties in demonstrating and assuring that the requirements will continue to work well. With this in mind, NRC plans to assure that Part 61 requirements are properly implemented through a structured licensing program including periodic updates and inspection of operations. This will enable the operation. Also, Part 61 provides for a period of observation and maintenance after active operations cease to ensure the site has reached e stable condition before releasing the site operator from his responsibilities.

Item 2 - The EIS and requirements developed for Part 61 do not deal with identification of specific near-surface disposal alternatives for actual disposal of LLW. Rather, they establish the safety limits and technical requirements which should be applied in disposal to ensure public health and safety over the long term. Each of the options mentioned in the comment can be evaluated against the requirements in Part 61 as well as against the types of social and political factors raised. Part 61 does not preclude their use but establishes the requirements on how they should be sited, designed, operated, closed and cared for over the long term.

Item 3 - The picnic area shown in Figure 3.3 of the DEIS, Conceptual Layout of Support Facilities, is not intended to be a public facility and therefore, would have no effect on the probability of disposal site intrusion. The picnic area is intended for the use of site workers during operations and no waste would be disposed of in the area indicated.

<u>Item 4</u> - A basic problem in determining the requirements for disposal of waste was the choice of pathways and events which should be considered in the analyses. The commenter suggests no specific "unlikely event"

. . . .

٠.

,

which should be guarded against. As a part of the analyses, NRC considered both credible and potentially unlikely events (e.g., intrusion into a closed disposal facility). In addition, NRC analyzed potential accidents which might occur at a site during operations such as dropping a container of waste releasing a portion of its contents and a fire in a disposal trench. In the final EIS, NRC also analyzed one additional pathway, trench overflow and surface water transport. NRC staff believes the pathway and events considered in the analyses in the FEIS are adequate to assess the impacts of disposal of LLW.

• . .

Item 5 - The staff believes that the comment refers to a statement in the first complete paragraph on page 4-12, Volume 2 of the DEIS. This statement notes that for the purposes of the EIS, operational releases from other waste processing operations (e.g., compaction and solidification) would be very small and as such were not included in the analysis. Only potential additional impacts from waste incineration were included in the analysis.

Perhaps the statement in the DEIS was not sufficiently clear as to NRC's intent. For purposes of the Part 61 rule, NRC wished to analyze the costs and impacts of different waste form and packaging options. For example, incineration of combustible waste material combined with solidification of the resultant ashes has the advantage of placing the waste into a stable form, helping to promote disposal site stability and reduce potential groundwater impacts. On the other hand, waste incineration is not currently practiced by most waste generators, and involves an additional expense as well as an airborne release pathway.

For the EIS, it was recognized that waste processing activities such as waste compaction or solidification are currently being carried out by a number of licensees such as nuclear power facilities. As such, these activities are already being included as part of assessing the potential impacts of operation of a particular facility. In such assessments, the effects of radionuclide being incorporated into food chains are considered. Such assessments have indicated that the potential impacts of such "normal" waste processing operations are indeed small. Waste incineration, however, has not been extensively used by waste generators, and so estimates of potential airborne impacts were included in the draft EIS.

· .

٠.

In any case, the purpose of the EIS was not to perform a detailed environment survey of all waste management operations, but to address costs and impacts only to the extent that they relate to development of the Part 61 rule on waste disposal. In this case, it was determined that waste incineration was a useful means of helping to ensure disposal site stability.

Item 6 - The siting requirements developed in Part 61 require consideration of projected population growth and future development to determine their effect on the ability of the disposal site to meet the performance objectives. Almost any type of land with proper treatment and sufficient water could potentially be used for agriculture. As such, given the commentor's assumption, all potential sites could be eliminated from consideration. Since the number of sites required for LLW is small and the hazard duration for the waste is short, they should not have an impact on the overall U.S. or world agricultural production. In addition, Part 61 would allow productive uses of the site provided such uses would not affect site integrity or lead to disturbance of the disposed waste. If at some future time it is decided to use the site for productive uses such as farming, the potential impacts of doing so would have to be weighed and balanced against the benefits.

1000 1000 000 00

Item  $7^{2}$ - The staff agrees with the importance of requiring the licensee to provide financial responsibility for closure, post-closure, and long-term care of the disposal site, and the proposed rules require that the licensees obtain these types of financial assurances. However, at this time, the Commission does not require LLW disposal site operators to obtain third party liability insurance, although the operators of all the operating LLW disposal sites currently carry this type of coverage from the nuclear insurance pools as a normal business practice. The Commission staff is currently conducting a generic review of mandatory insurance for a variety of types of material licensees, and depending on the results, the Commission may require LLW operators to obtain this type of coverage.

<u>Item 8</u> - Urea-formaldehyde (UF) has been used as a solidification agent for liquid radioactive wastes. In 1979 the State of Nevada prohibited the acceptance of urea-formaldehyde at the Beatty facility. In 1981 both the

States of South Carolina and Washington also prohibited its acceptance. Urea-formaldehyde was prohibited at the three commercial disposal sites due to its inability to consistently produce a solidified product which would meet the disposal site free-standing liquid requirements.

. .

يتعاد

. . . .

• •



.....

- 1 - L

Atomic Industrial Forum, Inc. 7101 Wisconsin Avenue Washington, D.C. 20014 Telephone: (301) 654-9260 TWX 7108249602 ATOMIC FOR DC

January 18, 1982

071115752

•••

32 J. 10 Mags (46 FR 38081

U.S. Nuclear Regulatory Commission Washington, D. C. 20555

Secretary of the Commission

Attention: Docketing and Service Branch

Re: Notice on Proposed Rulemaking "Licensing Requirements for Land Disposal of Radioactive Waste" 10 CFR Part 61 (46 Federal Register 38081) and "Draft Environ-mental Impact Statement," NUREG-0782

Dear Sir:

The enclosed comments on the proposed rule and the draft EIS were prepared by two AIF subcommittees on low level radioactive waste.

The consensus of the subcommittees is that the proposed rule appears reasonable; however, some suggestions and some exceptions to the proposed rule and to the draft EIS were made and are reflected in the enclosures.

Sincerely,

John H. MacMillan Chairman, AIF Committee on the Nuclear Fuel Cycle

JHM: gpg Enclosures

it shall whether by card.

Comments on the Proposed NRC Rulemaking: "Licensing Requirements for Land Disposal of Radioactive Waste", 10 CFR Part 61, and on the "Draft Environmental Impact Statement", NUREG 0782 The AIF Subcommittee on Low-Level Radioactive Waste and the AIF Subcommittee on Solidification of Low Level Reactor Radwaste

## Comments on "Supplementary Information":

- p. 38084 The first paragraph in this section under "Disposal
- Site Design, Land Disposal Facility Operation and Disposal al Site Closure Requirements" calls for operations and design which will result in the elimination of ongoing active maintenance after closure, requiring only minor custodial care. "Active maintenance" is not clearly de-fined here or even in the definition in Para. 61.2 "Defi-nitions". In the definition active maintenance is defined in Para V.C ritions". In the definition, active maintenance is des-cribed as a "significant remedial" action. It is suggested that the postclosure maintenance requirement should be site specific and incorporated in the license conditions.
- p. 38084 The first paragraph under "Waste Characteristics and Classification" points out the desirability of the physi-cal integrity of the waste and the site lasting until Para V.C radioisotopes decay "to levels where they are no longer of concern from the migration standpoint". What stan-dards should be applied to satisfy this requirement?
- p. 38085 In the section "Waste Characteristics and Classifica-Para V.C tion", there is a discussion of a "de minimus" classification for wastes which would be exempt from 10 CFR Part 61. It is noted that the NRC in the next two years will work to define these wastes and "to provide for additional waste exceptions as appropriate." We support the need for a "de minimus" concept and encourage the expeditious establishment of suitable criteria for this concept. A "de minimus" classification would result in the conservation of valuable disposal site burial space while at the same time protecting the health and safety of the public. In this connection, the Commission is urged to permit case by case reviews of requests for specific applications of the "de mimimus" concept during the period criteria are being developed. We will be pleased to be of assistance in the establishment of "de minimus" levels and would like to call attention to the minimus" levels and would like to call attention to the

. . . . . . .

×.

report by the Forum's National Environmental Studies Project entitled "De Minimus Concentration of Radionuclides LENE DISTRICTENT Maste Management Group sponsored a study, "Sug-gested Concentration Limits for Shallow Land Burial of Radionuclides", which should be of value.

In reference to the "Manifest Tracking System" we urge p. 38086 the Commission to ensure that any changes in manifest Para F tracking systems are compatible with existing systems in order to minimize or eliminate possible duplication.

The five phases of the life-cycle for a typical land p. 38086 Para G disposal facility discussed in the proposed rule are: preoperational, operational, closure, post-closure obser-vation, and institutional control. Some discussion is needed to indicate that several of these phases may proceed simultaneously at the same site as part of the normal disposal site operation.

· · .

21.22 ÷...,

There should be a sixth phase in the life cycle for a Incre should be a sixth phase in the file cycle for a land disposal facility identified in the regulation; namely - release for uncontrolled use. This phase, which occurs after the radioactive contents of the landfill have decayed, should be stressed. It is further impor-tant to stress the need to keep toxic or mutagenic chemical forms out of landfills intended for radioactive waste. Chemical wastes are present forever and the land used must be restricted forever. Radioactivity decays away in time and land use can be recovered.

The proposed rule requires a license renewal every five p. 38087 Para G years. For a facility of this type, a five-year license renewal policy appears unreasonably short. Disposal sites should be provided with a full term license, subject to appropriate review. The fiscal basis for site operation and monitoring assumes a reasonable operating life. Licensing similar to that under 10 CFR 50 would appear to be more appropriate.

The proposed rule states that the Institutional Control Board has a responsibility to "keep people off the p. 38087 Para G site". This approach may be unnecessarily restrictive. Limited use of the land may be desirable in the future. A more appropriate action might be to control access to 1. 1º\* the site. The control board should have the flexibility to institute suitable control options depending on the particular condition existing at the site.

## Comments on Part 61:

• •

p. 38089 See our comments on Para. V.C p. 38084 on "active main-Para 61.2 tenance".

-3-

. . .

+2 ----

122

يرس

1.1

p. 38090 In the definition of "Disposal" isolation of radioactive Para 61.2 wastes from the biosphere should be clarified. A better definition would be: "Confinement of the wastes with no provision made for subsequent retrieval."

p. 38091 The primary safety objectives for a near-surface disposal

- Para 61.7 facility should be redefined in a manner that will (a)
- **(b)(1)** keep the site personnel dose as low as reasonably achievable and (b) keep the environmental impact and population dose below specified limits. In the existing statement of objectives the word "prevention" should be replaced with "minimize". To minimize migration of radionucludes is to provide a means, towards achieving the primary safety objectives. "Prevention of exposure to "<u>inadver</u>-tent intruders" is a special case of (b) above, and should be regarded as a secondary objective.

In the first sentence the word "eliminated" should be p. 38091 In the first sentence the word "eliminated" should be Para 61.7 modified or omitted. The requirement may not be possible. (b)(2) · · · · ·

p. 38092 It is not apparent what is required for "demonstration" Para 61.13 or how analysis will be accomplished. This section

- should be clarified. This comment applies to Paras. 61.23(f),(i) and (j). **(b)**
- p. 38094 The section on "changes" is too restrictive. The Para 61.25 licensee should be allowed to make changes when deemed necessary, providing they do not decrease the level of protection to the public and provided they are brought to the attention of the Commission in a timely manner (i.e., Subpara.(d) provisions for changes similar to those in 10 CFR 70.32 (d)).

p. 38095 The closure period should be included in the closure -Para 61.29 plan rather than the regulation. The required period of observation should be a function of specific site charac-

teristics. 

-4-	-	4	•	
-----	---	---	---	--

o. 38095 2ara 61.30 5)	The requirement for transfer of the license to Federal or State custodial care should be part of the closure agreement and not the regulation. Transfer should be contingent upon licensee satisfaction of required conditions. Transfer "when the agency is prepared" leaves an open-ended commitment by the licensee which is not warranted.	p. 3809 Para 61
0. 38096 Para 61.50 (a)(3)	Projections of population growth should be limited to useful demographics. Projection to 100 years or even 50 could be a useless exercise or worse, could rule out an acceptable site.	
0. 38096 Para 61.51 (a)(4)	Replace "prevent" with "minimize".	
2. 38096 212 61.52 (2)(1)	Replace "no interaction" with "no significant inter- action".	
0. 38096 212 61.51 2)(6)	Replace "eliminate" with "minimize".	p. 3809 Para 61
2. 38096 2ra 61.52 (a)(4)	The word "orderly" should be deleted or sxplained.	<b>(a)</b> (7)
). 38096 Para (a) (7)	"Accurately located" depends on the state of the art. The word "accurately" should be defined or not in- cluded.	p. 3809 Para 63 (b)(1)
5. 38097 Para 61.55 Pa)	Table 1 has a footnote eliminating wastes containing chelating agents in concentrations greater than 0.1% except as specifically approved by the Commission. This requirement eliminates most routine decontami- nation techniques to reduce occupational exposures and thus adversely affects ALARA programs. It is not clear why this restriction is being imposed. Guidance on acceptable packaging and disposal techniques for these agents is needed.	

.

-5-

:. :

19097 Table 1 is a specific list of radioisotopes with . 55 their respective concentration limits for three classifications of wastes, Class A segregated waste, Class B stable waste, and Class C intruder waste. While some of the concentration limits shown are reasonable, demonstrating compliance for others would be difficult (for example Ni-59 and Ni-63) because of problems in sampling and long delay periods for off-site transport which consequently result in additional increases in personnel radiation exposures. Additionally, the measurement of TRU in the 10 nanccurie/gram range in the presence of other interfering radionuclides is very difficult. Detection of 10 nanocuries/gram can be accomplished readily if TRU isotopes are the only ones present. Further, the present policy of volume reduction increases the concentration of radionuclides in the waste and in some cases may cause the waste to exceed the Table 1 concentration limits. For these reasons, the concentration limits in Table 1 should be reevaluated to determine their ability to be achieved in a realistic situation and in a cost-effective manner. . . 

. 38097 ara 61.56 a)(7)	Change to read "that does not <u>significantly</u> exc atmosphere at 20 degrees C". If this requirement not modified, waste generators may be required to package gases under reduced pressure or elevated temperatures. The basis for the 100 curies should provided.	eed is be
	provided.	

38098 By specifying a 5% limitation on the physical waste a 61.56 form, the proposed rule may well be unintentionally mandating a high integrity container for Class B wastes. Experience has shown that drums and liners can normally be filled to about 80% of their volume. Demonstrating compliance with a 95% criteria could be difficult. Also, the 50 psi compressive load criteria may eliminate bitumen media as a waste stabilization process. The compressive load criteria may also be more appropriately related to individual disposal site overburden characteristics rather than specified as a generic criteria.

· · ·

## p. 38100 Eliminate radioactive waste already disposed of and Para 61.82 covered from NRC inspection requirement. p. 38100 It is not understood how this paragraph applies to Para 2.103 10 CFR 61. p. 38102 Regulatory requirements suggest rewording section to Para 20.311 "conduct operations in a manner which assures com-(d)(3)pliance with Paras. 61.55 and 61.56 of this chapter." Existing wording implies that a separate quality assurance program will be used. (f)(5)Same comments as suggested for 20.311(d)(3). Same comments as suggested for 20.311(d)(3).

-6-

(a)

and

1.1

•

p. 2-2 and 2-3

. . ••

· • •

should be specified in the regulations. Para 2.2 Prescriptive requirements that limit site inventory or quantities shipped to protect against excessive migration of radionuclides in ground water are site specific and should be incorporated in site licenses. p. 2-8 As noted, de minimus levels may be readily applied Para 2.4 to other waste streams. - We urge the NRC to develop these levels as well as levels for other radionuclides. p. 4-49 Consensus of opinion indicates that the institutional Para 4.3.6 control period may reasonably range from 100 to 300 years. Since this parameter is somewhat arbitrary it should be the last parameter selected in the equation for determining prescriptive requirements.... p. 4-65 The 500 rem/yr limit to an inadvertent intruder is Para 4.5.4 too conservative when considering the low probability of acquiring such a dose commitment. p. 7-3 A "generic on-site-specific waste classification Para 7.1.1 system" will be too conservative. Waste generators shall normally only need to consider one site to dispose waste. They shall normally only need to be conversant with the classification system specific to that site. Publishing generic classification systems shall not enable waste generators to ignore site specific limitations since these can be expected to be complex and many. A site specific waste classification shall allow optimum site utilization. p. 7-13 On this page and on several other occasions the ALARA p. 7-13 On this page and on several other objective re-Para 7.2. concept is incorrectly used to justify excessive re-strictions. Levels should be defined as ALARA when the costs to reduce radiological parameters to these levels are justified by the benefits accrued and where costs to reduce radiological parameters below these levels are not justified. It is not necessarily ALARA merely because it is lower. It is not necessarily

ALARA when industry has learned to comply to an excessive restriction. It is not reasonable and therefore not ALARA to require industry to reduce environmental impacts 2-3 orders of magnitude below impacts from other conventional industries.

-7-Comments on Draft Environmental Impact Statement on 10 CFR Part 61

"Licensing Requirement for Land Disposal of Radioactive Waste"

NUREG 0782

Generic performance objectives should be limited to

occupational and environmental impact limitations and

(1)

**(**2)

3

A

 $\bigcirc$ 

6

++

j.

. . . .

-8-

n. 7-16

p. 7-16 Para 7.2	The 10 nCi/gm limit for disposing of TRU waste is not ALARA even though industry has complied with this reg- ulation. The only alternative requiring indefinite storage of TRU waste exceeding 10 nCi/gm is not rea-	
p. 7-23	Industry will have direction a	j

- ill have difficulty in economically arranging waste to ensure that it complies to the conditions Para 7.5 of a particular category:
  - This may lead to generators assigning conserva-0 tive estimates to waste concentrations and consequential under utilization of a waste site. (8)
  - Inability of regulatory authorities to assay con-٥ tainers of waste renders control by assay unenforceable.
  - The scaling factors recommended to simplify waste ٥ analysis are not applicable to industries generating a wide range of wastes.

# General Comments on the Above:

- The product of a large number of conservative estimates ٥ will be unreasonably conservative even if individual estimates are only marginally conservative. A better method for combining parameters is to use best estimates of each parameter and to propagate uncertainty errors to generate upper and lower confidence boundaries. A simplified ver-sion of this approach using a range of values for each parameter (e.g. as used in the BEIR III report), is pref-(9) erable to compounding conservative estimates.
- Since the inadvertent intruder is identified as the criti-٥ cally exposed group for most radionuclides more effort should be directed into determining the probability of intruder scenarios occurring. Waste concentration limits should then be relaxed if these interaction probabilities are factored into the impact calculations.
- Credit should be given for improving critical waste forms to reduce plant uptake in the intruder-agriculture scenariq.

-9-

- There is no discussion of the potential impact of techno-logically enhanced natural radiation (TENR) due to cali-0 bration operations or changes in pH. Radon and potassium-40 levels in groundwater may be increased due to site operations. The EIS should show that such an impact would not be likely to violate EPA drinking water quality regulations and then TENR can be explicitly excluded from. 10 CFR 61.
- Although there is provision for exceptions to the proposed ٥ limits on a case by case basis there should be some calculations to show the impact expected from radioactive gas. Also the wide range in toxicity of labeled compounds should also be addressed.

(B)

Ð

(16)

6

 $(\overline{x})$ 

- 0 Provision should be made to allow waste generators to categorize waste by an inventory process. The quantities of waste generated in a year or present in an individual shipment could be determined with greater accuracy than by making separate determinations for individual containers. This comment is particularly relevant to very low level waste and to radionuclides that are restricted by the ground water migration scenario.
- Scenarios assume that all waste is emplaced just before site closure. However, in practice waste will accumulate over a 20 to 60 year period and a considerable fraction shall have decayed before site closure. Hence a relaxation in concentration limits should be considered when applied to the short half-life radionuclides received during the initial period of burial activities.
- Studies are quoted in the EIS which indicate that the av-۵ erage radioactivity concentrations in-waste can be expected to be from 1 to. 10% of the maximum concentration. Hence concentration limits should be relaxed by at least an order of magnitude and still provide adequate intruder protection. · . . . . . . . . . . 1
- We have been unable to find any justifications for the ٥ quantity limits for containers. Individual container limits appear excessively conservative.
- For waste buried as received the DOT limit should apply ٥ since airborne release and nonoccupational exposure is the controlling factor.
- For waste produced on-site the limit for individual con-۵  $\textcircled{\baselineta}$ tainers should be 10 times the DOT limit since it is occupational exposure which provides the limiting scenario.

(10)

•

•

## Docketed Comment Number: 32

.

## Commenter: Atomic Industrial Forum

<u>Response(s)</u>: Item 1 - NRC staff agrees. This is the very approach adopted and used by NRC. Part 61 establishes overall performance objectives which define levels of protection for individuals and populations which should not be exceeded in the disposal of LLW. Part 61 also states that the inventory of specific nuclides disposed of at a particular site will be controlled on a site-specific basis.

 $\underline{\text{Item 2}} \sim \text{NRC} \text{ agrees that development of de minimis levels for radioactive waste is of importance and has programs underway in this area. (Also see the response on this issue under Item 4, Comment 11.)$ 

Item 3 - NRC analyzed a range of institutional control periods from 50 to 300 years. Based on the analysis NRC found no compelling reason to select one particular institutional control period over another. Use of a longer institutional control period would theoretically allow higher concentrations of some radionuclides to be disposed of as Class A waste. The limit of 100 years proposed in Part 61 was selected because:

- It agreed well with previous estimates on the effective length of institutional controls made by EPA;
- (2) It was consistent with the consensus arrived at from the regional workshops on Part 61; and
- (3) Public comments on the preliminary draft of Part 61 was that 100 years was about the right time period.

Based on public comments received on the proposed Part 61 rule, NRC has not changed the 100 year institutional control period. NRC does not believe raising the institutional control period to 300 years is acceptable since it raises long-term care costs, it may place an undue burden on future generations, and it creates greater uncertainty in analyzing LLW disposal. (It is more

difficult to postulate what may happen over such a long time period.) In addition, several commentors stated they believed 100 years was too long. Raising the institutional control period to 150-200 years would have the effect of raising the radionuclide concentrations for Class A wastes based on intruder protection considerations. Since the stability requirement for Class B waste is of more importance for other reasons besides intruder protection (i.e., migration, operational safety and long-term stability), NRC does not believe changing the 100 year time frame for active institutional control is warranted.

Item 4 - NRC has not adopted a higher dose limit for protection of an inadvertent intruder. Rather, given the hypothetical nature and low probability for such an event occurring (e.g., passive institutional controls should prevent it from occurring), NRC has performed a more realistic analysis in determining concentrations of specific radionuclides acceptable for nearsurface disposal. Based on the analysis, concentration limits for certain radionuclides were raised. The impacts to an inadvertent intruder over the long term, however, were not found to change. They are still only a few millirem at 500 years.

Item 5 - NRC staff disagrees with the comment. Although waste generators normally send waste to one site, there have been numerous instances in the past in which waste shipments originally headed for one site have actually been disposed in another. Waste generators have also made use of 2 or more different waste disposal facilities within a year. This has occurred, for example, as a result of temporary closure of disposal facilities. A site-specific classification system would increase the potential for some wastes being acceptable at one site and not in another. Such a system would also be difficult to enforce and would tend to increase the number of shipment violations (e.g., wastes delivered to a site in forms or having radioactivity concentrations unacceptable at that site). This situation would be worsened if the number of disposal sites grows as a result of development of regional disposal compacts.

÷

NRC staff believes that case-by-case consideration of exemptions to Table 1 should be allowed for specific waste forms and specific disposal site conditions. Provisions have been included in the Part 61 rule to allow this. However, NRC staff still believes that the most workable overall approach is a generic waste classification system rather than a site-specific one.

Item 6 - NRC staff does not believe it has necessarily incorrectly used ALARA. In response to the comment however, NRC has deleted reference to ALARA in the two instances specifically noted in the comment (Paragraph 7.2.6 of Chapter 7 and Paragraph 6.3 of the summary). This change has been incorporated into Appendix E, "Errata."

 $\underline{\text{Item 7}} - \text{The reader is referred to the response to Item 6.}$  Comment 26.

 $\underline{\text{Item 8}} = \text{Compliance with waste classification is an important}$  issue and NRC has taken a number of steps to clarify the matter. These include:

- reducing the radionuclides listed in Table 1 to those which are really needed;
- specifically allowing indirect measurement techniques;
- allowing averaging of C-14, Tc-99, I-129 and other radionuclides over the volume of the waste;
- In addition, NRC staff is developing a technical position on waste classification to set out further guidance on compliance with the waste classification system.

<u>Item 9</u> - The staff believes the current approach is the most reasonable considering the level of available information and the generic nature of the analysis.

Item 10 - NRC has not performed a probabilistic analysis of inadvertent intrusion. Rather, NRC has conservatively assumed a few individuals will be exposed through inadvertent intrusion involving reasonable productive uses to which the land could be put in the future. As discussed in Item 4 above, NRC has performed more realistic analyses involving inadvertent intrusion which has increased concentrations for certain radionuclides.

(The reader is also referred to Item 3, Comment 24.)

Item 11 - The staff included provisions in proposed Part 61 for improved waste form to be considered on a case-by-case basis. At the present time there is insufficient data for most waste streams on the relationship of improved waste form to reduced plant uptake to allow a credit to be factored into the intruder-agriculture scenario for such improvements.

Item 12 - Technologically enhanced natural radiation (TENR) has been defined as those natural sources of radiation that would not normally occur without the presence of some technological activity not expressly designed to produce radiation. Examples would include the release of concentrations of natural radioactivity to the environment from the combustion of coal and natural gas to produce electricity and from the mining industry. NRC staff is unsure of the appropriateness of application of such a concept to the construction activities which might take place at a LLW near-surface disposal facility. The staff believes that impacts from TENR at a near-surface disposal facility would be similar to those resulting from any large construction project. In addition, they would be very site-specific with respect to the concentration of natural radioactivity that would be present in the specific site soils and ground water. As such, NRC has not addressed them in this FEIS.

Item 13 - NRC considered the full range of potential pathways of release of radioactivity to the environment. The pathways considered included groundwater migration, plant and animal intrusion, wind and surface water transport and gaseous releases. (See Appendix M of the DEIS for details.) Based on consideration of existing data and measurements, NRC concluded that gaseous

## 

releases were not a significant pathway for release and thus no detailed analyses were performed in the EIS. NRC did not specifically analyze the range in chemical toxicity of labeled compounds due to a lack of information about the specific compunds used and more importantly, due to the lack of an accepted methodology of assessing biological effects of exposure to such compounds.

<u>Item 14</u> - This comment relates to the basic issue of compliance with the waste classification requirements. NRC recognizes the difficulty in directly assaying for some radionuclides on a routine basis and will allow indirect measurement techniques. NRC has prepared a branch technical position which outlines methods and procedures acceptable to NRC staff for complying with the waste classification requirements.

<u>Item 15</u> - The staff does not agree with the commenter on this point. Rationale for the staff's position is contained in the response to Item 10, Comment 38.

<u>Item 16</u> - Several commenters have remarked about the concentration limits in Table 1. As noted earlier, the staff has conducted more pragmatic analyses and as a result the concentration limits in Table 1 have been revised.

<u>Items 17-19</u> - The 100 Ci limit is based on the license conditions for the disposal of gaseous wastes now in effect at the Hanford and Barnwell disposal sites. These limits have not resulted in unsafe environmental conditions at the disposal sites nor have they resulted in overly restrictive situations for waste generators.

The 100 Ci limit is consistent with an accident evaluation assuming a dropped package producing occupational exposures to site workers. The DOT limits, however, are established based on accident doses to the public. For tritium and Kr-85 in uncompressed gaseous forms, the DOT limits are 1,000 Ci in a Type A package and 50,000 Ci in a Type B package (49 CFR 173.389(1) and 173.390(a)). For gaseous waste forms the occupational exposure case is the limiting condition. . . . .

· .

 US Ecclogy, Inc. 9200 Sheibyville Road, Suite S28 P O. Box 7246 Courtrille, Kentucky 40207 502 426-7160



· . ·

Washington, D.C. 20555 Dear Mr. Smith:

US Ecology has reviewed the proposed NRC rules for the land disposal of radioactive wastes (10 CFR 61) published in the Federal Register on July 24, 1981, and comment as follows:

 <u>Disposal Site Design, Land Disposal Facility Operation, and</u> <u>Disposal Site Closure Requirements</u>. Page 38084, Col. 1.

This section as well as others in the proposed regulation refer to "eliminating the need for active maintenance after site closure." It is impossible to ensure that the need for no active maintenance can be achieved, so we would suggest that the word "minimize" be used in place of "eliminate."

- 2. Waste Characteristics and Classification.
  - a. Page 38084, Col. 2.

This section states that the stability of the waste and the disposal site should last long enough for the radioisottopes to decay to levels where they are no longer of concern from a migration standpoint. Without any reference to appropriate standards it is the regulatory agency's interpretation which must be used as a standard. References to applicable standards should be given.

b. Page 38084, Col. 3. The limit of 10 nanocurie per gram for TRU waste is excessively conservative and is counter productive. Limits of this nature may dissuade waste generators from practicing volume reduction.

- 2 -

c. Page 38085, Col. 2. Decreased burial efficiencies will be experienced in hurial facilities if the Commission fails in the establishment of a lower limit Class A segregated waste concentration. It is suggested that de minimus quantities be established to properly categorize those wastes that are suitable for alternate disposal methods. (e.g. sanitary landfills).

. . . .

-J. Financial Assurances. Page 38085, Col. 3.-

The Commission indicates that certain types of surety mechanisms are acceptable. It is recommended that representatives of the commercial insurance industry be given definitive criteria against which to base any bonds in order to ascertain whether or not underwriters will accept the risk and default provisions set forth in these regulations.

4. Manifest System. Page 38086, Col. 2 and Pages 38102, 38103, Section 20.311.

The manifest tracking system provides for three (3) copies of the manifest. It has been the experience of this company that a greater number of copies is needed. A total of 5 identical manifests would provide a copy for the generator, carrier, an advanced copy to the intended recipient, one copy with the shipment and one copy as a receipt for the waste.

5. Regulatory Flexibility Act. Page 38088, Col. 1.

The lack of a de minimus quantity as well as any lower limits on the Class A segregated waste concentrations conflicts with the Department of Transportation Regulations governing the transport of radioactive material. 49CFR section 173.389 defines radioactive material as "any material or combination of materials which spontaneously emit ionizing radiation. Material in which the estimated specific activity is not greater than 0.002 microcurries per gram of material, and in which the radioactivity is essentially uniformly distributed is not classed as Radioactive material."

6. 61.2 Definitions

Page 38090, Col. 1. . "Disposal" is defined as isolation of radioactive wastes from the biosphere. This is not possible as the "biosphere" can be interpreted as meaning part of the world in which life can exist. Many microorganisms will exist in and around all the waste. It is suggested that disposal be defined as isolation from the biosphere's inhabited by man or his food chain.

1/22/82 smp

#### 7. 61.5 Interpretations

#### Page 38090, Col. 3:

By placing the responsibility for all regulatory interpretations upon the General Counsel and preventing other Commission personnel from making interpretacions, the Commission could significantly disrupt operations by slowing the regulatory process. To avoid this, the regulations should establish a period, for example 10 days, within which the General Counsel must respond to requests for interpretation.

- 1 -

1 1

8. 61.7 (c) (2). Page 38091, Co. . . . . .

Since closure plans are to be submitted with the initial license application and periodically updated until final closure occurs, the licensee should not have to apply for a license amendment to close the site. The closure plan will have been periodically reviewed by the licensing agency throughout the disposal site operational phase, therefore, closure of the disposal site in accordance with the final updated and approved closure plan should be sufficient.

. . . 9. 61.13. Page 38092 الم المعالية المعام المع المعام . التعليم المالية المعام المعام المعام . 

The definition of what is required for "demonstration" or how analysis ... will be accomplished should be provided. The comments apply to-61.2 (1) (1) (j).
10. <u>61.24 Conditions of Licenses</u>.
61.24 (f). Page 38094, Col. 2.

This section refers to pussession of radioactive materials and should be clarified to indicate that buried waste is no longer in the possession of the licenses. The second second

аналанан талан Вен каландар 11. 61.25 Changes.

61.25 (a). Page 38094, Col. 3. This section on changes is too restrictive. The licensee should be allowed to make changes when deemed necessary providing they do not decrease the level of protection to the public and provided they are brought to the attention of the Commission prior to implementation.

12. 61.29 Post-Closure Observation and Maintenance: Page 38095, Col. 1.

. . . . . . . . . . . .

•.

If the Commission or licensing agency closely monitors the closure per 61.28 and is familiar with the site, the minimum period of five years for post-closure observation and maintenance could be shortened commensurate with the condition of the disposal facility as closed. Provisions should be made for a period of less than five years if conditions warrant upon request of the licensee.

### 13. 61.30 Transfer of License.

•

> 61.30 (a) (5), Page 38095, Col. 2. This section will allow the long-term custodial care agency even though prior commitments have been made to delay in accepting its responsibility. The use of the phrase "when the agency is prepared" in effect negates all prior commitments of long-term custodial care agency.

- 2 -

. . .

#### 14. 61.50 Disposal Site Suitability Requirements.

61.50 (a) (2). Page 38096, Col. 1. Modeling of any site can only be approximate. Some statement as to what criteria should be used to define "capable of being modeled" is required.

61.50 (a) (3). Page 38096, Col. 1. A time frame for projecting population growth and future development is needed. Is this time frame the 100 year custodial care period or the 500 year engineered barrier life span?

61.50 (a) (5). Page 38096, Col. 1. Clarification is warranted as to the requirement for well-drained and free of areas of flooding and frequent ponding. Is this a selection criteria based on "as is" conditions or "as can be engineered."

61.50 (a) (11). Page 38096, Col. 1. · · · If the environmental monitoring program is masked then it is not going to be a proper indication of the stability of the site, therefore, the use of the phrase "significantly masked" is not appropriate.

## 15. 61.51 Disposal Site Design for Land Disposal.

61.51 (a) (4). Page 38096, Col. 2. No cover will totally prevent all water infiltration. This section should require that covers be designed to minimize water infiltration.

\_\_\_\_\_ 61.51 (a) (5). Page 38096, Col. 2. Th a humid area some minor erosion will occur. This is one of the criteria which must be taken into account when selecting and closing a disposal site. It is not possible in a humid area to ensure that erosion will not result or that active maintenance will not be necessary in the future.

· · · · s - 2 - 1 · • ·

!

## 16. 61.52 Land Disposal Facility Operation and Disposal Site Closure

- - -

61.52 (a) (4) Page 38096, Col. 2. The word "orderly" should be deleted or defined. Crderly placement is often inconsistent with the principles of ALARA.

61.52 (a) (7) Page 38096, Col. 3. The requirement to accurately locate each disposal unit is reasonable if an allowed tolerance is indicated.

#### 17. Table I Page 38097.

Having no lower limit activity concentrations for Class A segregated waste conflicts with the definition of radioactive material used in 49CFR.

#### 18. 61.55 Waste Characteristics

61.56 (a) (7) Page 38097, Col. 1. 49CFR allows limited quantities of radioactive gases to be packaged in greater quantities than 100 curies. This is inconsistent with the requirements of these proposed regulations.

#### 61.56 (b) Page 39098, Col. 1.

By specifying a 5% limitation on the physical waste form, the proposed rule may well be unintentionally mandating a high integrity container for Class B wastes. Experience has shown that drums and liners can normally be filled to about 80% of their volume. Demonstrating compliance with a 95% criteria is unrealistic as a practicle matter. Also, the 50 psi compressive load criteria may eliminate bitume media as a waste stabilization process. The compressive load criteria would be more appropriately related to individual disposal site overburden characteristics rather than specified as a generic criteria.

#### Draft EIS Comments

In general, the Environmental Impact Statement (EIS) states current practices and also cites alternate methods. It is unclear whather the Commission accepts either case or prefers the alternates. It is important for the Commission not to specify a method as any proposed site for disposal will be unique.

1. Page 5-106

Vibratory compactors are recommended by the EIS. Such compactors are good for granular soils but are less effective for cohesive soils. The Commission should specify a degree of compaction or required permeability rather than a method of compaction.

· ·· .

#### 2. Page 3-35, Section 3.6.3.6 Security

Full time security personnel have not been used at the present commercial burial facilities. The full time security provisions such as fences, signs and periodic law enforcement visits have provided effective security and have not resulted in any security violations that would warrant the employment of full-time security personnel. The use of radio communication to contact emergency and law enforcement agencies is not warranted as the fastest and casiest method is the telephone. Many years of experience has shown that the use of telephone commucation in emergency situations has proved reliable.

- 6 -

3. Page E-13, Section 2.1.

It appears that the Commission is basing site selection on the ability of the locations to fit their computer model. Due to projected site complexity, it is not realistic to require a location to fit present computer models.

Ð

6

B

4. Page Q-16 and 17, Section 4.1.1, Table Q-7.

The cost of building demolition (\$300,000) seems high in view of the fact that the three buildings to be demolished cost only \$423,850 to build (See Q-3) and about half of that would have been materials.

5. Q-16, Section 4.1.2.

Vibratory compaction equipment will be effective only in granular fill.

6. Page K-45, Section 5.1.

The last paragraph of this section factually states that this fund is inadequate to pay for long-term care of the site but does not properly state that monies were placed in the State's general'fund and not earmarked for the use intended.

7. Page K-50, Section 5.5.

The statement that the fund is inadequate is not based on current facts. The surcharge has increased by 10% in 1981 and the proper investment of the funds will take care of the effects of inflation. Further increases can and are being implemented which will provide substantial funding to the State.

J. (J. Acoville President

JJS:db

C

## Docketed Comment Number: 33

#### Commenter: U.S. Ecology

<u>Response(s)</u>: <u>Item 1</u> - The purpose of the DEIS was to examine current and some past practices and alternatives which could be applied in the near-surface disposal of waste. Based on the examination preferred alternatives were selected which are reflected in Part 61 as performance objectives and technical and other requirements. The minimum technical requirements must be met in the siting, design, operation and closure of a near-surface disposal facility and allow flexibility in how they might be met at any particular site. The requirements generally reflect current practices with some additional improvements (e.g., requiring deeper burial of certain wastes exceeding specified concentrations).

Item 2 - Page 5-106 of the DEIS discusses "using commercially available compacting equipment such as vibratory compactors." This is only an example, which indeed, may be most applicable to granular soil. With respect to a degree of compaction or required permeability, these will be determined on a case-by-case basis due to differences in designs and site soils. Additional guidance will be provided in a technical position paper on site design and operations.

Item 3 - Full-time security personnel are in fact used at one of the present operating commercial disposal facilities. Use of full-time security personnel at future disposal facilities will be at the discretion of the site operator and the licensing authority. Similarly, radio communication capability with emergency and/or law enforcement agencies is present at one of the current operating commercial disposal facilities and will likewise be a discretionary feature at future sites.

Item 4 - The NRC is not basing site selection on the ability of a site to fit any computer model. However, modeling of a site is one basic tool in site evaluation. As indicated in a December, 1980 interagency workshop on modeling and low-level waste management, NRC believes that adequate modeling capability already exists for non-complex sites. (Reference 12.) For the complex sites, either existing or new models must be available to adequately represent the site conditions and potential migration from the site. In the case of new models, the applicant should expect that NRC will review the model as well as its use to predict site performance.

· · ·

Item 5 - The consideration of cost of building demolition was estimated for the draft EIS based upon the cost for building demolition estimated by Chem-Nuclear Systems, Inc. as part of their preliminary site closure and stabilization plan for the Barnwell disposal site. (Reference 13.) In this closure and stabilization plan, an estimate of \$525,000 was made for decontamination and demolition of most of the onsite buildings (some would be left standing for use by the site owner following license termination), conservatively assuming that a private contractor was hired to perform these services. Recognizing that CNSI carries out a number of additional activities at the Barnwell facility besides waste disposal (e.g., truck, transporter and cask refurbishment, mobile solidification units, development of solidification agents), a building demolition cost of \$300,000 was estimated for the reference disposal facility.

It should be recognized that actual closure costs such as building decontamination and demolition costs would be specific for a particular site and should be evaluated specifically for that site. Whether or not the assumed decontamination and demolition costs for the reference disposal facility are overconservative does not change the essential conclusions reached in the draft EIS regarding the need to consider and plan for facility closure prior to site licensing and the need to assure that funds will be available to carry out closure activities. In the interest of completeness, however, these costs have been reduced to \$200,000 for the final EIS. This change does not alter the conclusions reached in the EIS but is reflected in the cost/benefit analyses and estimates of unmitigated impacts.

Item 6 - See staff response to Item 2 above.

<u>Item 7</u> - The staff agrees with this comment and has included it in the Appendix E, Errata in this volume.

• •

.

•••

· · · · · ·

·

<u>Item 8</u> - The staff has revised its statement for the FEIS and included it in the Errata Appendix of this volume.



STATE OF WASHINGTON

DEPARTMENT OF ECOLOGY

Mail Stop PV-11 . Olympia, Washington 98504 . (206) 459-6000

January 13, 1982

Thank you for the opportunity to comment on the draft environmental

impact statement on 10 CFR Part 61 "Licensing Requirements for Land

Disposal of Radioactive Waste". As the state's coordinator for



DONALD W. MOOS Director

IOHN MILLMAN Covernor

.



ALANJ CIBBS Secretary

(1)

(2)

STATE OF WASHINGTON DEPARTMENT OF SOCIAL AND HEALTH SERVICES Olympia, Washington 98504

January 13, 1982

Barbara Ritchie TO: Department of Ecology PV-11

- Wancy P. Kirner, Supervisor Radioactive Materials Unit Radiation Control Section
- COMMENTS ON PROPOSED 10 CFR 61 ENVIRONMENTAL SUBJECT:

The environmental impact statement for 10 CFR 61, "Licensing Requirements for Land Disposal of Radioactive Waste," appears to be well written and it adequately supports appropriate regulation of a radioactive waste disposal site having more than 10 inches of rain each year. The environmental impact statement and its proposed regulation, however, fail to accurately address realistic concerns and place realistic conditions on the operation of a radioactive waste disposal site at an arid location ... Among the major problems of an arid site are slumping and wind erosion. More emphasis should be placed on mitigating these two impacts, absent infiltration of ground and surface water.

While it is conceded that the stability of the waste form enhances safety, the strong reliance placed upon scenarios involving groundwater infiltration appears inappropriate for an arid site such as the low-level waste disposal site located near Richland, Washington. Without the liklihood of groundwater or surface water transport of radionuclides, segregation of class A and class B wastes seems to be unnecessary when weighed against the burden of operating separate disposal units. For the arid site, a case can even be made to allow co-mingling of waste classes in an attempt to lower the average concentration of the most hazardous wastes. The same logic can likewise be used for class C wastes at an arid site, provided class C wastes are placed at deeper depths and solidified in a relatively leachfree matrix or otherwise segregated for their hazardous lifetime. The proposed regulation and its supporting environmental impact statement do not appear, however, to make a convincing enough argument for the establishment of a third disposal unit to handle only class C

FROM:

- IMPACT STATEMENT

NEPA documents, we notified all state agencies of the availability of the EIS. The Department of Social and Health Services was the only agency to respond. A copy of their comments is attached. If you have any questions, please call Ms. Nancy Kirner, Department . \* of Social and Health Services, at (206) 753-3459 or Mr. Greg Sorlie,

Sincerely,

Dennis Lundblad, Supervisor **Comprehensive Management Division** 

Attachment

DL:1c

IOHN SPELLMAN

Governor

cc: Ms. Nancy Kirner Mr. Greg Sorlie

Mr. R. Dale Smith. Chief

Washington, D.C. 20555

Safeguards

· Dear Mr. Smith:

. .

Low-Level Waste Licensing Branch

Office of Nuclear Material Safety &

U.S. Nuclear Regulatory Commission

Department of Ecology, at (206) 459-6016.

· · · · · · · ·

Division of Waste Management

.

and the second 

1. 1. <del>1</del>. 1. 1. 1. 1. 1.

Barbara Ritchie Page 2 January 13, 1982

wastes at an arid site. Washington does not believe that the segregation of waste by classification should become a matter of compatibility.

Since the draft environmental impact statement on 10 CFR 61 was written, the state of Washington has increased its perpetual care and maintenance funding. Descriptions of Washington's funding mechanisms should be revised on pages 9-6 and K-50 to reflect these recent changes as follows:

- a. The Perpetual Care and Maintenance (PC&M) Fund contribution by U S Ecology, currently at \$0.275 per cubic foot, will increase to \$1.75 per cubic foot. This renegotiated contribution will provide a total fund of approximately \$6 million by July 1, 1985. The total accumulation will be determined by the actual volume of waste disposed.
- b. In addition, US Ecology will contribute \$0.25 per cubic foot of waste disposed to a newly established Contingency Closure Fund. A contribution at this rate will yield a fund of \$800,000 by July 1985, assuring that the state of Washington will be able to adequately close the radioactive waste disposal operations conducted at the site if and when needed.
- c. On Jaunary 15, 1982, the company will post a surety bond in the amount of \$500,000 for a period of one year to protect the state if the company should leave the site without meeting closure conditions as stated in the license.

Thank you for the opportunity to incorporate our concerns.

NPK/db

#### Docketed Comment Number: 34

Commenter: State of Washington, Department of Social and Health Services

<u>Response(s)</u>: <u>Item 1</u> - NRC believes the requirements established in Part 61 for achieving long-term stability will be effective at both humid and arid sites and the requirement for segregation of compressible from stable wastes should reduce slumping of trench covers for trenches containing stable wastes. The control of surface erosion at both a humid and arid site will need to be considered on a site-specific basis through establishment of a proper surface water drainage system at a humid site and establishment of a cover that is not easily eroded by surface winds at an arid site.

Item 2 - As discussed above, NRC believes the segregation of compressible and stable wastes will help reduce trench cover slumping and associated increased potential for wind erosion of the cover due to cover instability and slumping. As such, NRC does not believe the rule should be changed. NRC believes, however, that site-specific considerations can and should be addressed through the licensing process.

Item 3 - The staff agrees with the points raised in this comment and has included them in the Appendix E, Errata in this volume.

 $(\mathbf{r})$ 

cont.

(3)

AICHE

# NUCLEAR ENGINEERING DIVISION AMERICAN INSTITUTE OF CHEMICAL ENGINEERS

07:15

าวม

(46 FR 3808

(46 FR 51776

.....

January 11, 1982

Secretary of the Commission U.S. Nuclear Regulatory Commission Washington, D.C. 20555

Attention: Docketing and Service Branch

Dear Sir:

The Nuclear Engineering Division (NED) of the American Institute of Chemical Engineers appreciates this opportunity to comment on the proposed regulation related to land disposal of radioactive waste (10 CFR Part 61) published at 46 FR 38081 on July 24, 1981.

The American Institute of Chemical Engineers is a professional organization representing over 50,000 chemical engineers, many of whom are members of the Nuclear Engineering Division. A number of these persons have had decades of experience in most (if not all) of the technologies related to land disposal of radioactive waste.

NED heartily concurs with the urgent need for regulations codifying requirements for land disposal of radicactive waste. However, we believe that Subpart D of the proposed regulations is so basically flawed that it should be deleted and the proposed regulation rewritten and republished for comment.

Generally, the regulation should present not "technical requirements" but "performance objectives." If technical requirements persist in the regulation:

- Overconservatism now embodied in the proposed technical
- requirements should be considerably reduced.
- Cost/benefit considerations should be based on the entire
- fuel cycle as now defined by the Administration and recognized
- by the Commissioners of the NRC.
  - The definition of transuranic waste should be made realistic
  - recognizing the entire fuel cycle with a rational application of the ALARA principle.
  - Full recognition of the beneficial consequences of "layered"

Secretary of the Commission January 11, 1982 Page Two

disposal should be reflected.

## PERFORMANCE OBJECTIVES

The primary flaw in the proposed regulation consists of the abortive attempt by the NRC to formulate generic limits for disposal. This will deprive a potential disposal facility operator of the opportunity to either:

- Take advantage of natural or engineered features of his operation to permit him to accept materials of above-normal radionuclide content and dispose of them with quite adequate protection of the public in both the short and long term, or
- Limit his receipts to less contaminated material to permit him to avoid some of the more costly proposed NRC requirements, still with adequate safety, to be able to offer a service less costly to his customers (and, therefore, the public).

 $(\mathbf{r})$ 

2

An unacceptable reason given by the NRC not to adopt a performance objective appears in Section 2.2.1 of NUREG-0782 (hereinafter, the NUREG), where it is stated, "In addition, it may not be totally clear to an applicant or interested person how to design and operate a disposal facility to meet the  $\neg_{a}$ general objectives." We submit that anyone so naive or lacking in thorough understanding of all technical aspects of waste disposal should not be involved in a responsible technical position related to a land disposal operation. The role of the regulating agency should be that of regulation, not the establishment of design bases.

We have felt repeatedly in past years that NRC regulations resembled more and more a "cook book" approach. As with other nuclear operations, waste disposal cannot be left to unqualified persons to be carried out simply by rote. It must be done under the supervision of highly qualified management.

Accordingly, there should be no reason why the NRC should not adopt only performance objectives for waste disposal rather than the limits and criteria proposed. To the contrary, it appears that the former would offer worthwhile benefits. Granted, Subpart D does allow departure from its requirements if it can be shown that there will be compliance with Subpart C; however, the very existence of the requirements of Subpart D will inevitably present a major obstacle to an applicant endeavoring to benefit from an especially good site or operational provisions.

## UNDUE CONSERVATISM

## We feel that there is gross overconservatism in the NUREG. The proposed

Secretary of the Commission January 11, 1982 Page Three

requirements are, to a great extent, not cost-effective and extend far beyond ALARA requirements. There are many examples of this, including:

- Both NUREG-0456 and NUREG/CR-1005 are frequently referenced in the NUREG. Indeed the conclusions of all three documents bear many similarities. However, in the two referenced documents no credit was taken for the waste package nor for waste stabilization or solidification. The proposed regulation carries with it requirements for all of these. That all documents are similar at the bottom line but so very different in basic premises shows the extreme conservatism in the NUREG.
- It is well known that the mixing of <sup>129</sup>I with stable (<sup>127</sup>I) iodine results in directly proportionate reduction in iodine related dose to the thyroid and other organs. This is recognized in the NUREG (for example, on pages 4-16 and 5-73) but then ignored in consideration of <sup>129</sup>I limits.
- As addressed below, the only scenarios for transport of radionuclides from a disposal trench when the waste is emplaced at moderate depths (say 10 meters) below grade are the water pathways. Sections 61.51(a)(4) and (6) require that the disposal site design be such as to prevent water infiltration and to eliminate the contact of water with waste. If the proposed regulation is adopted, it seems fair to presume:that a prospective disposal site operator would be required by the NRC to give reasonable proof that it is probable that these requirements can be and will be met. If so, then there exists no credible water pathway, making limits proposed quite irrational.

#### SOURCES OF WASTE

The NUREG pointedly ignores significant sources of waste. It references what is now known to be an interim federal policy against recovery of valuable. irreplaceable energy resources by reprocessing spent fuel. This is no longer the policy, as was recognized by the Commissioners in their Second Prehearing Memorandum and Order dated November 6, 1981 in the waste disposal confidence rulemaking (PR-50, 51). In that document, the Commissioners stated, "On October 8, 1981, the President issued a statement outlining a policy favoring commercial reprocessing." It is well known that the operation of a reprocessing plant generates sizeable quantities of "low level" waste. One must include in such quantities the waste also discharged from the operation of the plutonium and "high level" waste solidification facilities.

Further, it is the policy of the current Administration that the nation should

Secretary of the Commission January 11, 1982 Page Four

proceed with a breeder program. This will require the processing of plutonium for juel, generating more waste not addressed in the NUREG. . .

(4`

Indeed, the NRC in its proposed 10CFR Part 60 recognizes the possibility of reprocessing in the definition of "high level radioactive waste." (60.2)

The nature and quantity of reprocessing, high level waste solidification, and plutonium processing wastes mandates that they be included in consideration of land disposal regulation.

## TRANSURANIC (TRU) WASTES

The definition of the waste form called "transuranic" must be redone for at least several reasons:

- The quantity of wastes from those portions of the entire fuel cycle ingored in the NUREG, much of which would likely be in the 10 to 100 nCi/cm<sup>3</sup> range, must be taken into account in assessing the cost-benefit balance on the ALARA principle. One of the more significant components of these wastes is, for instance, leached fuel cladding hulls. It is likely that the transuranium content of hulls will be below 100 nCi/cm<sup>3</sup>. To unnecessarily identify them as being excluded from land burial could result in large and wasteful expenditures of money. Other components of these wastes likely will fall in the same range.
- It appears that, in tying the 10 nGi limit to lung dose, the inhalation pathway has been used. With layered disposal (see below) this pathway for transuranics should not be applicable. The water pathway would lead to, first, only an ingestion pathway, and second, only to the requirement for an inventory limit rather than a concentration limit.
- The "fact" that the 10 nCi limit is readily complied with now loses its meaning when all real (even if not now generated) wastes are taken into account and the principle of ALARA is sincerely applied.

#### VALIDITY OF PROPOSED LIMITS

As noted above, the NUREG frequently references NUREG-0456 and NUREG-CR/1005. Both of these documents (with deep involvement of

З,

(2)

and

Secretary of the Commission Janaury 11, 1982 Page Five

#### NRC personnel) concluded:

- Intruder-construction and intruder-agriculture scenarios would require concentration limits to be imposed on waste to be disposed of by land burial,
- The potential for other intruders and the potential (likely or not) man-rem consequences to a few individuals lead to the conclusion that such a scenario is unimportant in the future.
- Water migration scenarios would require not concentration limits but total site inventory limits to be imposed.
- None of these scenarios is expected to take place while there is institutional control (taken in the NUREG to be 100 years).
- The "layered" mode of disposal, as described in the NUREC, obviates the need to address any intruder scenario for the higher activity wastes.
- There is no need for special packaging or stabilization of wastes to make the above conclusions valid (the referenced NUREGs assumed no such requirement).

Therefore, the limits proposed in the draft 10CFR 61.55 must be recomputed. There should be a clear recognition that waste buried at the bottom (say 10 meters deep) of an operation is only subject to sitespecific inventory limits (a performance objective). Further, credit should be given for compliance with the requirements of 61.51 "disposal site design for land disposal."

11

Also, the consideration of subsidence should be redone We can see no valid reason why, during the period of institutional control with monitoring by the NRC, any subsidence could not and would not be promptly corrected. In line with this, we believe that even without the proposed requirements to prevent or minimize subsidence, all significant subsidence could be expected to take place well before the expiration of the 100 year period (after site closure) of institutional control. Accordingly, the costly requirements proposed in Part 61 are unnecessary both during the priod of institutional control and after.

· · · · · · ·

Finally, througout the NUREG, numbers on concentrations, doses, etc., are expressed to as many as three or four significant figures. Such a presentation is very misleading in that it implies that our knowledge is that precise (which it is not) or that such accuracy is needed (which it is not).

We do note with approval that the discussion on 46 FR 38085 indicates that

Secretary of the Commission January 11, 1982 Page Six

de minimis waste classifications will be established. Hence, potentially lower cost disposal options may develop for materials at lower concentration levels. We encourage establishment of such de minimis limits outside the scope of Part 61. This approach should allow consideration of both the form and type of waste.

In conclusion, 10CFR Part 61 should be rewritten based on simple performance objectives. Any competent applicant has available all the necessary tools (hydrological, geological, climatological, etc., data, codes, etc.) to support appropriate site-specific limits for his proposed operation. We urge that he be given the latitude to take advantage of site-specific benefits or to search for an improved site which would give him an opportunity to offer a better service at a reduced cost.

172

129

122

We request the opportunity to answer any negative response by the NRC to any aspect of this discussion.

Sincerely,

R. J. neuma

R.I. Newman Chairman, Nuclear Waste Task Force Past Chairman, Nuclear Engineering Division

(6)

Car

6

#### Docketed Comment Number: 35

<u>Commenter</u>: American Institute of Chemical Engineers, Nuclear Engineering Division

<u>Response(s)</u>: <u>Item 1</u> - The reason cited by the commenter for NRC not adopting purely performance objective requirements is only one of several. Others included the additional time required to develop only performance objective requirements, the additional time that would be required in licensing specific facilities due to the large number of factors that would need to be considered to determine compliance, and the fact that, while workable, it would not allow for establishment of more detailed prescriptive requirements in those areas where specific guidance is known to be needed. Based on the comments received on the rule the majority of commenters supported the combined approach of setting overall objectives to define an acceptable level of performance and minimum technical requirements on the siting, design, operation and closure of a LLW disposal facility.

<u>Item 2</u> - NRC staff believes that the commentor has not provided an adequate basis for his assertions regarding either the level of conservatism in the draft EIS or the cost-effectiveness of the requirements in the draft Part 61 rule. The commenter has given three examples to back up his assertions. NRC's responses to each of these examples are as follows:

 Besides the draft EIS, the commenter references two other NRC publications, NUREG-0456 and NUREG/CR-1005. (References 14 and 15.)

While there are some similarities between the earlier work and the draft EIS, the draft EIS applies more recent dose assessment methodology and an improved handling of groundwater impacts. A number of important conclusions are also quite different-for example, the importance of waste form in limiting the consequences of potential contact of waste by an inadvertent intruder is not considered in the earlier work. This results in some large differences in the limiting concentrations calculated for near-surface disposal. In this regard, NRC staff believes that by not considering waste form, it is the earlier work which leads to overconservative conclusions rather than the draft EIS. NRC staff believes that adoption of the limits proposed in NUREG/CR-1005 would have had a greater impact on most waste generators.

• • •

(2) The potential for reduction of thyroid dose due to dilution of radioactive iodine (<sup>129</sup>I) with stable iodine (<sup>127</sup>I) was briefly considered in the draft EIS. Work performed by Leddicotte, <u>et. al.</u>, was cited, for example. (Reference 16.) At the time, NRC staff believed that while information gaps and time limitations precluded taking credit for dilution by stable iodine in the draft EIS, it was a matter that obviously needed to be further examined. This potential for dilution by stable iodine has in fact been examined in more detail in the final EIS.

As it turns out, whether or not dilution by stable iodine is considered makes essentially no difference in determining whether a waste generator meets waste classification limits obtained from consideration of exposures to an intruder. Iodine-129 typically exists in only trace quantities in low-level waste well below (typically one or more orders of magnitude) the maximum concentration limits. However, dilution with stable iodine may make a large difference in long-term groundwater impacts and consequently any site-specific inventory limits.

(3) In the draft EIS, the effectiveness of layering waste in order to reduce potential intruder exposures was considered insofar as this technique could be implemented in all regions of the country. It is not clear what type of waste the commentor is referring to in his statements regarding disposal at a depth of 10 meters. Since this would be unreasonable for all wastes, NRC staff believes that the commentor is referring to Class C waste. NRC has not required in the draft rule that Class C waste be placed a minimum of 10 meters below grade as such a requirement is not necessary and might also preclude disposal of Class C waste in many humid disposal sites. Rather, NRC has required 5 meters of depth which provides an adequate level of protection against intrusion for the limited period of time required. It is also not clear that the only credible pathways (as the commentor asserts) for waste disposal at a depth greater than 10 meters are water pathways. It is true that

pathways such as erosion or human intrusion from housing construction are probably extremely unlikely. However, this does not preclude other activities such as a well being drilled onsite which passes through waste, bringing contaminated material to the surface. Disposal of waste at greater depths will be considered by NRC in subsequent work.

Finally, it appears that NRC staff was not sufficiently clear regarding their intent with paragraphs 61.51(a)(4) and (6) in the draft Part 61 rule. The intent was that requirements in Section 61.51 such as preventing water infiltration or eliminating contact of water with waste be considered as <u>objectives</u> to be strived for rather than absolute criteria. This intent is being clarified in the final Part 61 rule.

Item 3 - The requirements and classification system developed for Part 61 can be applied to any waste whether currently generated or to be generated in the future. The requirements define safe disposal and establish minimum controls which should be applied to ensure safe disposal of waste regardless of type or point of generation. Some wastes are not considered generally acceptable for near-surface disposal and will need to be analyzed further. A preliminary analysis indicated that certain reprocessing wastes may fall into this category. (See response to Item 4 below.) NRC plans to address disposal of such wastes through subsequent work which may result in amendments to Part 61 setting out requirements for the disposal of such wastes.

Item 4 - Due to the volume of comments received regarding TRU waste disposal, the limits for transuranic waste have been reexamined in the final EIS. However, some of the commentor's rationale for his assertions may be briefly examined. First, NRC staff continue to believe that the option of reprocessing of spent reactor fuel and recycle of the recovered plutonium is not likely to be a significant source of waste for at least several years. Secondly, NRC staff compared the limits in 10 CFR 61 with some estimated concentrations in a number of waste streams which could be projected to result from plutonium recycle activities. These estimated concentrations were obtained from work on the subject performed by DOE. (Reference 18.) This analysis projects that cladding hulls, for example, would contain transuranics at levels greatly exceeding 100 nCi/gm--e.g., nearly 700 nCi/gm. Thirdly, the commenter incorrectly assumes that all waste possibly containing TRU isotopes would be layered. It is possible that much of this assumption comes from a lack of clarity in Table 1 of Section 61.55. In any case, NRC staff believe that the concept of layered disposal as defined in the Part 61 rule does not automatically exclude potential inhalation exposures.

. ..

 $\underline{\text{Item 5}} - \text{As discussed in response to other comments by this commenter, NRC staff does not believe that the commenter's conclusions fall from the premises stated.}$ 

Item 6 - NRC does not believe the consideration of subsidence should be redone. The requirements for reducing subsidence are intended to reduce the need for active and costly maintenance during the active institutional control period. By placing greater emphasis on stability as a part of operations, long-term maintenance costs are reduced. In addition, if consideration of subsidence was neglected, the very same requirements for long-term stability would be needed to help ensure safety during operations, reduce potential for migration and reduce potential exposures to an inadvertent intruder.

· 1 Item 7 - NRC staff apologize for any difficulties regarding the presentation of the results in the draft EIS. Most of the calculations were performed with the assistance of a digital computer. Since the computer was inherently able to handle a large number of significant figures, it was convenient to retain several significant figures throughout intermediate calculations and in final printouts of results. (This was believed to be more useful than the practice of rounding at intermediate steps which could have been the case if hand calculations were involved.) Retaining three or four significant figures in the computer output facilitated debugging the computer programs and checking the reasonableness of results. In the draft EIS, tabulated results were considered "intermediate results" and were generally reproduced as printed by the computer. These tabulated results were used to provide a backdrop for discussions and help reach conclusions. Rounding up to a more reasonable number of significant figures was accomplished as part of the discussions and conclusions reached and as part of setting forth particular numerical requirements (e.g., Table 1) in the draft Part 61 rule.

A-93

·. .

· . .

· . .



CONFERENCE OF RADIATION CONTROL PROGRAM EIRECTORS, INC. P. O. Box 148 Concord, New Hampshire: 03301 Jul 20 P1:12 January 15, 1982 Mr. R. Dale Smith, Chief Low-Level Waste Licensing Branch

Low-Level Waste Licensing Branch United States Nuclear Regulatory Commission Washington, D.C. 20555

#### Dear Dale:

The following comments relate to the Proposed Rule for a new Part 61, and other related amendments, as published in Volume 45, No. 142 of the Federal Register, dated July 24, 1981.

The Nuclear Regulatory Commission is to be commended on this proposed regulation. This new proposed rule provides a vast improvement in guidance and requirements for the control of land disposal of radioactive waste. The existing regulation on the disposal of radioactive waste contained in Part 20 provides only a limited guidance that has resulted in problems at both NRC and state licensed burial sites.

The Conference has taken the position since the late sixties that more federal guidance and criteria is needed in the area of shallow land disposal of radioactive waste. The Conference has parsed various resolutions addressing this concern. Such guidance is imperative for uniform management of the country's radioactive waste.

Many of the improvements in the proposed Part 61 are responsive to some of the recommendations made by the Conference. Proposed Part 61 also establishes, in the NRC regulatory system, many "state of the art" improvements that have been developed by the states in the operation and regulation of low-level radioactive and hazardous waste burial sites.

The Conference concurs and supports the following proposals contained in the new Proposed Part 61:

- An improved waste classification system that divides the present all-inclusive "low-level" waste into several categories based on hazard evaluation.
- Confirmation of the definition of transuranic wastes as recommended by the Conference several years ago.

Mr. R. Dale Smith January 15, 1982 Page 2

- 3. Technical requirements for burial that become more stringent based on the increasing hazard of the radionuclide concentration in the waste.
- 4. Technical requirements on stability of waste packaging.
- 5. Technical requirements on burial site operations that minimize voids in trenches with emplaced waste.
- 6. Technical and financial requirements associated with the site closure phase and the post-closure observation and maintenance phase.

•.

- Defining a period of time for institutional control, and relating the classification and disposal of waste to this time frame.
- 8. For design purposes of new proposed sites, and until specific recommendations are forthcoming from the Environmental Protection Agency, the Conference supports the objective in Proposed Part 61 that any movement of radioactivity should not result in calculated doses in excess of 25 mrem/year to an individual at the site boundary, and support the application of the EPA drinking water standard to the nearest public drinking water supply. We also recommend the application of these dose limits as guidelines for existing sites. Of course, the application of ALARA should be applied to a near surface burial site, as with other licensees.

We strongly support the proposed amended requirements to Part 20 for the certification and use of shipping manifests to track waste shipments.

We offer the following specific comments on the proposals.

- (1) 61.2 Definitions:
  - a. Should include a definition for "minor custodial care."
  - b. "Disposal" As stated, the temporary storage of waste could meet the definition. Disposal into land generally connotes long term or permanent removal of the waste from the biosphere. Possibly the words "long term" should be added before the word "isolation."
  - c. "Waste" The definition should include levels of concentrations of radioactive materials specific to a particular waste stream below which regulatory control is no longer regulred. This position was previously

1

1

Mr. R. Dale Smith January 15, 1982 Page 3

transmitted to the Commission in the form of Resolution III which was adopted at the thirteenth annual Conference meeting.

Although the definition for waste may not be the area in which the following concerns should be addressed, we strongly urge the Commission to consider the total hazard, both radiological and chemical, in the classification of acceptable waste for a radioactive burial site. Although this concern is somewhat addressed under 61.56, criteria or guidelines are needed which specifically consider the appropriateness and/or procedures of combining waste which is highly toxic, chemically, but low in radioactivity, with chemically nontoxic radioactive material.

Consideration should be given to a definition of "toxic chemical/radioactive waste" which may require different handling and burial requirements. This concern was expressed to the Commission in the form of Resolution II adopted by the Conference.

#### (2) <u>61.7 Concepts:</u>

. . . .

Under section (c) (4), "The Licensing Process," the concept of license transfer to a state or federal agency after finding of satisfactory disposal site closure is discussed. Under this concept, if transferred to a state, the NRC would be licensing a state government until institutional care is not required. This concept for a low-level waste site may need further discussion and refinement before implementation. For example, what criteria will be used to judge the adequacy of the state government licensee? If "state government" violates the conditions of the license, what enforcement actions would be taken? Additionally, why would the license be terminated if transferred to the Department of Energy, but not terminated if transferred to a state? Experience and history has shown that states have been as effective as the federal government in assuming responsibility for long-term care of existing sites. Therefore, consideration should be given to license termination after transfer to a state government.

(3) 61.12 Specific Technical Information:

Section (d) would require a description of the design basis natural events or phenomena. Requirements should be

Mr. R. Dale Smith January 15, 1982 Page 4

> placed on the applicant to consider the maximum creditable accident anticipated, and a description of actions that would be taken should such an event occur.

## (4) 61.50 Site Characteristics:

Consideration should be given to a nonsuitability requirement for burial into areas high in natural radioactivity.

#### (5) 61.80 Records:

Consideration should be given to the requirement for the maintenance of a duplicate set of vital records at an alternate location in case of destruction by fire or other loss of primary records.

We would also like to comment on the Summary Draft E.I.S., NUREG-0782, Volume 1. We believe the Draft E.I.S. adequately supports the need for the Proposed Part 61, and identifies impacts. Our specific comments are as follows:

#### 1. Page 15, Impact Measures:

Another pathway which should be considered is trench overflow and/or pumping of water from trenches.

2. Page 16, Table S.4:

Impact measures should include trench overflow or pumping of trenches, and the release of tritiated methane.

3. Page 30, 5.1.2., 4th paragraph:

The "bathtub" problem not only "leads to costly long-term trench pumping," but may also release radioactivity in the process.

#### 4. Page 55, Impacts on the Public:

An additional beneficial impact with the implementation of the requirements of the Proposed Part 61 is the reduction of potentially large, long-term financial cost for taxpayers in states in which sites are located.

#### - GENERAL COMMENTS -

There is an urgent need for written criteria and/or guidelines as to what constitutes acceptability in meeting the performance

· ·

Mr. R. Dale Smith January 15, 1982 Page 5

standards of the new Proposal. We recommend immediate consideration be given to the following:

- 1. Standards for acceptable solidifying agents for liquid low-level waste.
- 2. Acceptable testing procedures to determine if solidified liquids meet the above standard.
- 3. Comprehensive licensing guidelines for waste reduction methods such as incineration and compacting units.
- Guidelines for types of acceptable or optimum geological formations for the land disposal of low-level radioactive waste.
- Standards or criteria for the optimum design of a site, considering dry vs. humid climates.
- 6. Guidelines as to acceptable or optimum burial practices.
- 7. Guidelines relating to the minimum acceptable health physics program for a near surface burial facility.
- Guidelines relating to the minimum acceptable <u>nonradiological</u> occupational protection program for a near surface burial facility.
- 9. Guidelines relating to the specific areas that must be considered for emergency planning for a near surface burial facility.
- 10. Guidelines on environmental contamination trend analysis, and recommended protective actions based on potential increasing environmental levels of specific radionuclides.
- 11. Standards or criteria which specifies what constitutes stabilization and a decommissioned site.
- 12. Guidelines which identify the minimum acceptable activities to be performed by a government agency after closure.
- Guidelines on methods of determining financial needs for long-term care.
- 14. Guidelines as to the minimum acceptable environmental monitoring programs for a near surface disposal facility.
- 15. Guidelines on the application of ALARA for near surface facilities.
- 16. Guidelines on ground water modeling for near surface facilities.

Mr. R. Dale Smith January 15, 1982 Page 6

Again, I would like to commend the Commission for the excellent work contained in the new Proposed Part 61. Such regulations have been needed for many years. We appreciate the opportunity to comment on this very vital need to our country.

Yours very truly,

John R. Stanton, Chairman Conference of Radiation Control Program Directors, Inc.

JRS/CMH/pch

cc: Board Federal Liaisons Executive Secretary G. Wayne Kerr

#### Docketed Comment\_Number: 36

Commenter: Conference of Radiation Control Program Directors, Inc.

<u>Response(s):</u> Item 1 - In the draft EIS, a scenario was considered in which a disposal facility experiences a severe water accumulation problem in disposal trenches. Potential costs of managing this accumulation problem were calculated, which included pumping leachate from trenches, processing the leachate through an evaporator, and solidifying and disposing of the concentrated evaporator bottoms. In the draft EIS, however, potential radiological impacts from overflow of leachate from disposal trenches or from operation of the evaporator were not estimated. However, the potential occurrence of such scenarios was part of the basis for NRC's position in the draft Part 61 rule regarding the need for disposal site stability.

It should be recognized that actual impacts due to trench overflow would not be expected to occur to any great extent. Rather, the cognizant officials in charge of radiological health (either on a State or Federal level) would take steps (e.g., leachate pumping and treatment, recapping disposal trenches) to minimize or eliminate such impacts. However, the calculated impacts represent levels of potential human exposures which could occur if such steps to eliminate water accumulation were not taken. Experience has shown that these steps can be very expensive. Of course, operation of an evaporator to reduce the volume of the accumulated leachate would indeed involve release of radioactive material to the environment.

An estimate of potential radiological impacts from trench overflow and evaporation has been performed in the Final EIS. These calculated impacts add support to NRC's position regarding the need for disposal site stability.

Item 2 - As stated in Item 1 above, an estimate of potential impacts from trench overflow and operation of a leachate evaporator have been included in the final EIS. These potential impacts provide further support to NRC's position regarding the need for disposal site stability. From data available to NRC at the time the draft EIS was written, however, it appeared that potential impacts from tritiated methane released through disposal trench

covers are at negligible levels. NRC staff is not aware of any new data which would contradict this belief.

Item 3 - As stated in Item 1 above, an estimate of potential impacts from trench overflow and operation of a leachate evaporator have been included in the final EIS.

Item 4 - The staff concurs with the commenter's opinion and will include this suggestion in the final EIS.



Docketed Comment Number: 37 (Not assigned)

.

Commenter: N.A.

Response(s): N.A.

RELRAD

New England Nuclear'

182 J." 27 92:56

·62 .... 27 P2:54

•

January 21, 1982

R. Dale Smith, Chief Low-Level Waste Licensing Branch Division of Waste Management Nuclear Regulatory Commission Washington, D. C. 20555

Dear Mr. Smith:

I enclose this letter to alert you to the fact that New England Nuclear chaired the NELRAD 10 CFR 61 Subcommittee and was actively involved in the development of the detailed comments. As such New England Nuclear endorses the report and recommends due consideration be given to it.

Sincerely,

NEW ENGLAND NUCLEAR lum Brantley

VUP. Administration, NENC

JCB/da

· · ·

549 Albany Street, Boston, Massachusetts 02118 Telephone 617-482-9595 Telex 94-0996

. ..

JUCRET RULIDER PR-2, et al PROPOSED RULE PR-2, et al (46 FR 5000) BROFOSED BULE PR-61 38 46 FR 51776 January 21, 1982

2. Dale Smith, Chief Low-Level Maste Licensing Branch Division of Waste Management Nuclear Regulatory Commission Washington, D. C. 20555

lear Mr. Smith:

I enclose comments on 10 CFR Part 61 and the accompanying Draft Environcental Impact Statement. These comments are written from the generator's respective, compiled from safety officer input from several waste generiting firms in New England.

These collective comments are a product of NELRAD activity. NELRAD is a consortium of New England firms and institutions who use radioactive zzterials and have a common need for a nearby low-level radioactive waste disposal facility. Our group was formally organized in 1981 to support the efforts of the six New England states in complying with the Low-Level Eacloactive Waste Policy Act.

We are pleased to be a part of the process that allows review of proposed regulations. In general, we approve the intent of 10 CFR Part 61 and encourage expeditious progress.

Sincerely, A

Siznis D. Stelluto Executive Director, SELRAD

.05/da

Enc.

P. O. BOX 1267 CONCORD, MA 01742-1267 Tel. (617) 371-0358

то:	NELRAD Committee
FROM:	NELRAD 10 CFR 61 Subcommittee
SUBJECT:	Comments on Proposed Rulemaking on Land Disposal of
	Low-Level Radioactive Waste.
DATE:	1/20/82

The enclosed report presents detailed comments on 10 CFR 61, the Draft Environmental Impact Statement (NUREG-0782) and the accompanying summary of the proposed rule. The detailed comments are preceded by a summary of the main points.

Members of the subcommittee which prepared this report are:

F.N. Brenneman (to Dec. 1981)	Northeast Utilities P.O. Box 270 Hartford, CT 05101
M. Galanik	77 Main St. Room 208, 238, MIT Cambridge, MA 02139
D. Gomer	Nuclear Metals, Inc. 2229 Main St. Concord, MA 01742
L.R. Smith (Chairperson)	New England Nuclear Corp, DuPont 549 Albany St. Boston, MA 02118
J. Stelluto	NELRAD 549 Albany St. Boston, MA 02118
E. Tarnuzzer	Yankee Atomic Electric Company 1671 Worcester Rd. Framingham, MA 01702

Thanks are due to K. Bennert, J.D. Bernardy, J.C. Brantley and C.B. Killian for their helpful comments and to K. Thomas for processing this report.

1 R. Emith.

L.R. Smith

#### SUMMARY OF COMMENTS ON THE PROPOSED RULEMAKING ON LAND DISPOSAL OF LOW-LEVEL RADIOACTIVE WASTE: PREPARED BY THE NELRAD 10 CFR 61 SUBCOMMITTEE.

#### 10 CFR 16 Scope.

We believe that the development of performance standards in 10 CFR 61 is the best approach to establishing licensing requirements for land-disposal of low-level radioactive waste. We agree that only essential generic prescriptive requirements should be included in the regulations and all site specific requirements should be incorporated in individual facility licenses.

#### Site Licenses and Inspection.

The scope and sequence of activities in establishing safe operations and ensuring proper closure of a facility appear realistic. We recommend that safeguards be strengthened by:

- a. granting disposal facilities a full term license with appropriate review instead of subjecting a license to the public hearing process every five years.
- assigning a full-time NRC inspector to each LLW site during the operational phase.
- c. encouraging active monitoring and review of site records by state authorities.

#### Probability of Inadvertent Intrusion.

The method used to establish generic prescriptive requirements to protect inadvertent intruders is appropriate for estimating the lower boundaries of concentration limits. However, these limits are unnecessarily conservative because the probability of intruders encountering radioactivity has not been factored into the calculations.

We believe that better estimates of maximum permissable concentrations can be made if the following considerations are included in the calculations:

a. The probability of inadvertent intruders encountering critical waste forms.

(1)

b. The concentration of radioactivity in waste sent to a site exhibits a log-normal distribution with an average concentration at least an order of magnitude lower than the maximum permissable concentration.

If these probability factors are included in the calculations, maximum permissable radioactivity concentrations are expected to be at least an order of magnitude higher than those presented in 10 CFR 61 and will still provide sufficient protection to the inadvertent intruder.
### Site Selection and Utilization.

The site could be better utilized if credit was given for the decay of short lived radionuclides during the operational period.

It is clear that local resources and demographic developments are important considerations when selecting a suitable site. However, demographic predictions can be unreliable therefore we recommend that the NRC consider zoning requirements to restrict activities that ray adversely affect the site hydrology and environment.

The potential impact of changes in natural radioactivity in ground water due to site excavations was not conspicuously covered in the DEIS. We recommend that these effects be considered.

Clarification of the Intent of 10 CFR 61.

We have indicated several instances where the intent of the regulations is not clear. The following improvements are recommended:

- replace absolute statements by achieveable practical ones. a.
- use units and terminology recommenced by scientific standard ь. setting organizations (eg. ICRU and ICRP).
- ć. clarify performance objectives by specifying internal and external dose equivalent limits to individual organs as suggested by the Waste Concentration.

Waste generators are concerned that the difficulty in accurately assaying radioactivity in individual containers will cause overly conservative values to be assigned to shipments resulting in poor utilization of the site. We recommend that the NRC consider relaxing concentration limits on individual containers and accept inventory methods designating average concentrations in waste shipments. Using inventory averaging methods would also enhance the generator's ability to determine if waste concentrations were below "de minimis" levels. The establishment of "de minimis" levels for radionuclides and waste forms should be encouraged for better site utilization. . . .

The 100 Ci per container limit appears excessively conservative. We recommend that DOT limits be adopted since the most restrictive potential impact scenario appears to be individual exposure from accidential breach of containment during transportation to the site.

### Use of ALARA Concept.

in several instances the ALARA concept is improperly used to justify excessive restriction. We recommend that optimum levels be defined at which an operation could be described to be ALARA. Imposing further restrictions yielding small benefits at great cost is not ALARA. For example, the proposed 10 nCi/g limit for TRU contaminated waste is not ALARA

. ..

- . simply because industry has complied with this regulation. The EPA has suggested that 100 nCi/g is an appropriately conservative limit. We recommend that the EPA's suggestion be adopted unless a better limit is derived.

3

.: 1

# Manifest Tracking.

We recommend that a reasonable procedure be developed to integrate enforcement agencies into the control or supervision of the manifest tracking systems.

### Conclusion.

• .• .•

• • • • •

(2)

In general, we recommend that 10 CFR 61 should not duplicate existing regulations but should reference them and be compatible with them.

Our final conclusions is that 10 CFR 61 will provide a reasonable and necessary regulatory frame-work for low-level radioactive waste disposal. We submit these comments in the hope that they shall improve both safety and cost effectiveness.

•

 $V_{1} \geq V_{1}$ 

### SPECIFIC COMMENTS ON THE PROPOSED RULEMAKING ON LAND DISPOSAL OF LOW-LEVEL RADIOACTIVE WASTE: PREPARED BY THE NELRAD 10 CFR 61 SUBCOMMITTEE

1. Page 38084, Col. 2.

This section specifies that waste stability should be sufficient to ensure that the residual radioactivity is "no longer of concern from the migration standpoint" We recommend that the quoted phrase be replaced by appropriate dose limits.

- 2. Page 38084, Col. 3.
  - a. The 10 nCi/g limit for transuranic waste is too conservative. The EPA suggests that 100 nCi/g is a conservative limit. Overly conservative limits may dissuade waste generators from practicing volume reduction of waste potentially contaminated with TRU.
  - b. Although industry has been able to comply with the 10 nCi/g limit for TRU waste we do not agree that this limit is ALARA. On page 7-13 of the DEIS and in several other instances the ALARA concept is misapplied to justify excessive restrictions. Operations should only be said to be ALARA when the cost to reduce impacts from these operations is justified by the benefits accrued and when further costs to reduce impacts are not justified. Compliance with an excessive restriction or achieving a lower level of impact are not necessarily ALARA. Reduction of environmental impacts 2-3 orders of magnitude below comparable impacts from other conventional industries is not reasonable and therefore not ALARA.
- 3. Page 38085, Col. 2.

The establishment of de minimis levels for other waste streams and radionuclides should be encouraged since this should lead to improved utilization of disposal sites.

4. Page 38087, Col. 2.

Disposal sites should be provided full term license with appropriate review. The financial planning necessary for long term site monitoring assumes a reasonable operating life. Hence the license should not be subject to the public hearing process every five years with the possibility that renewal may not occur.

5. Page 38087, Col. 2.

"Keep people off the site" should be replaced by "control access to the site" to allow maintenance, surveillance and other appropriate activities.

6. In the following sections the words "assurance" and "assure" should be replaced by "ensurance" and "ensure" respectively. This is to indicate that positive action should be taken to achieve an objective rather than merely persuading that an objective can be achieved.

Page	38089,	Col.	3,	line	5,	61.2.			
Page	38090,	Col.	1,	line	51,	61.2.			
Page	38091,	Col.	1,	line	14,	61.7	(Ь),	(3).	
Page	38093,	Col.	з,			61.23	(b),	(c),	(d).
Page	38094,	Col.	1,			61.23	(e),	(g).	
Page	38095,	Col.	1,			61.28	(b).	-	
Page	38095,	Col.	2,			61.30	(a),	(2),	(5).
Page	38095,	Col.	3,			61.50	(a),	(1).	
Page	38096,	Col.	2,			61.51	(a),	(2).	
Page	38097,	Col.	2,			61.55	(b),	(c),	(1).
Page	38097,	Col.	з,			61.56	(b).		

Page 38090, Col. 2, line 16.
 Page 38091, Col. 1, line 3, 61.7 (a), (1)

Instead of "15-20" meters a single value should be used. "15-20" may be confused as meaning burial below 15 meters and above 20 meters from the ground surface.

- 8. Page 38090, 61.7 (a), (1).
  - a. We agree that both performance objectives and prescriptive requirements are necessary. Performance objectives should be limited to occupational and environmental impact concentrations and should be specified in the regulations.
  - b. Generic prescriptive requirements are appropriate to limit LLW concentrations and to protect inadvertent intruders. These and prescriptive requirements which provide financial surety should also be incorporated in the regulations. In the cases where prescriptive requirements are adopted from other existing or proposed regulations these should be referenced or incorporated in 10 CFR 61.
  - c. Other prescriptive requirements which limit site inventory or which protect against excessive migration of radionuclides are site specific and should be incorporated in site licenses. 10 CFR 61 should specify that site licenses will incorporate these site specific prescriptive requirements.
- 9. Page 38091, Col. 1, 61.7 (b), (1).
  - a. We recommend that the primary objectives for disposal of LLW are: To isolate LLW from the biosphere in a manner that maintains:
    - i. personnel dose equivalent commitments ALARA;
    - environmental impact and personnel dose equivalent commitments below specified limits.

• 5

- b. "Prevention of migration of radionucides" is an important strategy for achieving those primary safety objectives. "Prevention of exposure to inadvertent intruders" is a special case of (a) (ii) and should be called a secondary objective.
- 10. Page 38091, Col. 1, 61.7 (b), (2).
  - Omit "eliminated or" since it is not possible to reduce water access to zero.
  - b. This paragraph should be written more clearly. In particular it should be clear when "stability" refers to trench structure or the waste itself.
- 11. The generic term "radionuclide" should replace "isotope" and "radioistope" in the following sections:

Page 38091, Col. 2, line 1, 61.7 (b), (2). Page 38097, Table 1 Page 38097, Col. 2, 61.55 (a), (1).

12. Page 38091, Col. 2, 61.7 (b), (3).

Replace "would" by "could" since intruder risk has a statistical basis.

- 13. Page 38091, Col. 2, 61.7 (b), (4).
  - The first sentence is ambiguous. It may be rewritten thus: "Institutional control of access to the site is required for at least
    100 years after closure of the burial site;
  - b. Page 4-49 of the DEIS indicates that consensus of opinion expects that the institutional control period may reasonably range from 100 to 300 years. Since this parameter is somewhat arbitrary it should be the last parameter selected in the equation for determining prescriptive requirements.
- 14. Page 38091, Col. 3, 61.7 (c), (1).

"Established administrative procedures" should be cross referenced to enable recognition of specific procedures.

15. Page 38093 Col. 1, 61.13 (b).

"Demonstration" should be replaced by "reasonable indication" since it is not possible to demonstrate the achievement of performance objectives until long after a site has been closed.

16. Page 38094, Col. 1, 61.23 (e).

Replace "should" by "to".

17. Page 38095, Col. 2, 61.40.

Eliminate "reasonable assurance exists that".

- 18. Page 38095, 61.41 and 61.42.
  - Specify whether "annual" and "year" refer to a calender or a sliding year.
  - b. i. "Dose" should be defined to mean "dose equivalent",
    - It is not stated whether "dose" refers to internal, external or a summation of these commitments.
    - We advise that the ratio of dose equivalent limits to various organs should follow ICRP recommendations.
- 19. Page 38095, 61.42.
  - a. Since the inadvertent intruder is identified as the critically exposed individual for most radionuclides, more effort should be directed into determining the probability of intruder scenarios occuring. Waste concentration limits could then be relaxed if these interaction probabilities are factored into the impact calculations.
  - b. We agree with the proposed dose limit provided that waste concentration limits are calculated to ensure, with reasonable probability, that the inadvertent intruder does not receive more than 500 mrem/yr.

2

2.3

1.3

38

- c. Dose equivalent limits following ICRP guidelines should also be specified for the inadvertent intruders' thyroid, skin, bong and other organs.
- 20. Page 38095, 61.50 (a), (1).

The second sentence should be clarified. The "long-term performance objectives of Subpart C" should be specified or cross referenced individually.

21. Page 38096, 61.50 (a), (3), (4), (11).

If industrial or other activities which may adversely disturb the ground water should not be located near the site, consider the need to establish zoning restrictions to exclude these activities.

22. Page 38096, 61.50 (a), (5).

"Coastal high-hazard area or wet land" should be defined or a definition else-where in the regulations referenced.

23. Page 38096, 61.50 (a), (6).

Clarify whether "upstream drainage area" refers to onsite or other locations.

24. Page 38096, 61.50 (a), (7).

It is not appropriate to specify that ground water intrusion cannot contact waste. This section should be rewritten to specify the maximum permissable probability for ground water intrusion as is similarly accomplished when defining 100 year flood plains etc.

8

25. Page 38096, 61.51 (a), (4).

Replace "prevent" by "minimize".

26. Page 38095, 61.51 (a), (6).

Replace "eliminate" by "minimize".

27. Page 38096, 61.52 (a), (1).

This sentence requires clarification. "no interaction" should be defined since migration of released radionuclides could be considered a form of interaction.

28. Page 38096, 61.52 (a), (3).

It is not clear whether 5 meters refers to the distance from the top or bottom surface of the cover. Does cover include an impervious cap?

29. Page 38096, 61.52 (a), (4).

Remove "orderly". If it is intended that the waste should be emplaced in a specified manner the intent should be described more explicitly.

- 30. Page 38096, 61.52 (a), (6).
  - Replace "radiation... levels" by "exposure rates" or "dose rates". Include exposure to X-rays and bremstrahlung. Consider contributions from neutrons.
  - b. "A few percent above... background" is too vague. Specify a limit and require adherance to the ALARA principle.
- 31. Page 38096, 61.52 (a), (8).

If concentration limits are relaxed, more emphasis should be placed on isolating critical radionuclides from ground water and preventing access to potentially contaminated ground water. The latter might be accomplished by extending the buffer zone in the direction of ground water migration. 32. Page 38096, 61.53 (a).

We recommend that the potential impact of technologically enhanced natural radiation due to excavation operations or changes in pH should be considered in the DEIS. Radon and  ${}^{40}$ K levels in ground water may be increased due to site operations. The DEIS should show that such an impact would not be likely to violate EPA drinking water quality regulations. Then T.E.N.R. can be explicitly excluded from 10 CFR 61.

(6)

(6)

э

33. Page 38097, 61.53 (d).

Place this section before (b) to indicate that it refers to all phases under the licensee's control.

- 34. Page 38097, Table 1.
  - a. Use scientifically accepted notation or provide a definition of the convention employed (eg. tritium is properly indicated by the symbol  $^{3}$ H).
  - b. Table 1 is unclear and could be improved by:
    - i. lining up decimal points in a column.
    - ii. using larger type
    - iii. Indicating units by subheading instead of by note.
  - c. Table 1 should reference a list of de minimis levels for particular waste streams and other disposal methods for waste exceeding table 1 categories.
  - d. The limit on concentration for diluting agents should be specified as 0.1% of the container volume.
  - e. For a 55 gallon drum the concentration limits specified in  $\mu$ Ci/cm<sup>3</sup> should be multiplied by 200,000 ml to determine the maximum permissable total activity expressed in  $\mu$ Cl.
  - f. The use of scaling prefixes for units should be minimized. In particular multiple prefixes should be eliminated and in fractions a single prefix should be placed in the numerator. (eg. instead of  $\mu$ Ci/ml write Ci/m<sup>3</sup>: instead of  $\rho$  Ci/ml write mCi/m<sup>3</sup>
  - g. As suggested on page 5-76 of the DEIS, site utilization maybe improved if site licenses specified the option to dilute <sup>129</sup> by inoculating potentially contaminated waste with <sup>127</sup> sufficient to reduce potential thyroid uptake and exposure by 2 to 3 orders of magnitude.

### 35. Page 38097, 61.55

a. In the DEIS the product of a large number of conservative estimates will be unreasonably conservative even if individual estimates are only mildly conservative. A better method for combining parameters is to use the best estimates of each parameter and propagate uncertainty errors to generate upper and lower confidence boundaries. A simplified version of this approach using a range of values for each parameter (eg. as used in the BEIR III report) is preferable to compounding conservative estimates. Credit should be given for improving critical waste forms to reduce plant uptake in the intruder-agriculture scenario. Studies are quoted in the DEIS which indicate that the average radioactivity concentration in waste can be expected to be from 1% to 10% of the maximum concentration. Hence concentration limits should be relaxed by at least one order of magnitude and will still provide adequate intruder protection.

IV

- b. If concentration limits are to be included in the regulations we encourage the development of concentration limits for other radionuclides and compounds. However a "generic nonsite-specific waste classification system" will be too conservative. Waste generators should normally only need to consider one site to dispose waste. They should normally only need to be conversant with the classification system specific to that site. A site specific waste classification system should allow optimum site utilization.
- c. Industry will have difficulty in economically assaying waste to ensure that it complies to the conditions of a particular category.
  - i. This may lead to generators assigning conservative estimates to waste concentrations and consequential under utilization of a waste site.
  - ii. The inability of regulatory authorities to assay containers of waste renders control by assay unenforcable.
  - ili. The scaling factors recommended to simplify waste analysis are not applicable to industries making a wide range of custom products.
- 36. Page 38097, 61.56, (a).

Clarify whose health and safety is being referred to.

37. Page 38097, 61.56 (a), (1), (7).

Individual container limits appear excessively conservative and should be justified in the DEIS.

- For waste disposed as received the DOT limits should apply since airborne release and non-occupational exposure is the controlling factor.
- For waste processed on site the limits for individual containers should be 10 times the DOT limit since it is occupational exposure which provides the limiting scenario.

38. Page 38097, 61.56 (a), (5).

Add "normally" to read "waste must not contain or be normally capable of generating..." This is to ensure that plastics with high ignition points but which are capable of emiting toxic burn products are not unnecessively excluded from the waste.

11

- 39. Page 38097, 61.56 (a), (7).
  - a. Add "significantly" to read: "... at a pressure that does not significantly exceed one atmosphere at 20°C." If this sentence is not modified, waste generators may be constrained to packaging gases under reduced pressure or elevated temperatures.
  - b. It is not clear whether the 100 Ci limit applies only to gases. This should be clarified. 100 Ci limit per package is too restrictive for certain radioactive gases. eg.  $14CO_2$  and 3H.
  - Although there is provision for exceptions to the proposed limits on a case by case basis, calculations should be included to show the impact expected from radioactive gas. Also the wide range in toxicity of labeled compounds should be addressed in the DELS.
- 40. Page 38098, 61.56 (b), (1).
  - The intent of "within 5%" should be clarified.

1

41. Page 38098, 61.56 (b), (2).

"Non corrosive liquid" should be defined as it is in the DEIS, ie., "pH between 4 and 10 and incapable of significant galvanize and chemical reaction".

- 42. Page 38098, 61.58
  - a. Provision should be made to allow waste generators to categorize waste by an inventory process. The quantities of waste generated in a year or present in an individual shipment could be determined with greater accuracy than by making separate determinations for individual containers. This comment is particularly relevant to very low contamination levels and radionuclides which are restricted by the ground water migration scenario.
  - b. Scenarios assume that all waste is placed just before the site was closed. However in practice waste will accululate over a 20 to 60 year period and a considerable fraction shall have decayed before site closure. Hence a relaxation in concentration limits can be applied to short lived radionuclides received during the initial period of burial activities.

(8)

- 43. Page 38100, 61.82.
  - a. We recommend that a full-time NRC inspector or agreement state agency inspector be assigned to a LLW site during the operational phase.
  - b. The State authorities should be encouraged to munitor the disposal site and review site records.
- 44. Page 38102, 20.311

From the language in the proposed rule, it is not clear how enforcement agencies would be involved. One possible procedure would require the site operator to return a receipted copy of the manifest system to the generator vice merely notifying him of receipt. If the generator were then required to maintain a file of all shipment manifests and backrouted receipts, the enforcement agency could check for compliance at each generator's place of business. Such a system closes the loop on the process and allows one to readily check for compliance during a regular facility inspection. It has the further advantage of not burdening the enforcement agency with volumes of manifests. Additionally, generators could reserve making payment to shippers until the backroute is received. This would provide a strong economic incentive to comply with the manifest tracking system.

In any event, some reasonable procedure should be developed to integrate enforcement agencies into the control or supervision of the manifest tracking system.

### Docketed Comment Number: 38

Commenter: New England Nuclear

<u>Response(s)</u>: Item 1 - The staff has reviewed the basis for the concentration limits and has modified certain aspects of the analysis. These changes are reflected in Chapter 5.0 of the FEIS.

 $\frac{1 \text{ tem } 2}{1 \text{ tem } 2}$  - This comment is essentially the same as Item 5 of this letter. The staff response is contained under Item 5 below.

Item 3 - The staff does not believe it has necessarily used ALARA incorrectly. In response to the comment, however, NRC has deleted references to ALARA in the instances cited. This change has been incorporated into Appendix E, Errata.

<u>Item 4</u> - NRC analyzed a range of institutional control periods from 50 to 300 years. Based on the analysis, NRC found no compelling reason to select one particular institutional control period over another. Use of a longer institutional control period would theoretically allow higher concentrations of some radionuclides to be disposed of as Class A waste. The limit of 100 years proposed in Part 61 was selected because:

- It agreed well with previous estimates on the the effective length of institutional controls made by EPA;
- (2) It was consistent with the consensus arrived at from the regional workshops on Part 61;
- (3) Public comments on the preliminary draft of Part 61 were that 100 years was about the right time period.

Based on public comments received on the proposed Part 61 rule, NRC has not changed the 100 year institutional control period. NRC does not believe raising the institutional control period to 300 years is acceptable since it raises long-term care costs, it may place an undue burden on future generations, 🔀 🖉 a state of the state of t

and it creates greater uncertainty in analyzing LLW disposal. (It is more difficult to postulate what may happen over such a long time period.) In addition, several commenters stated they believed 100 years was too long. Raising the institutional control period to 150-200 years would have the effect of raising the radionuclide concentrations for Class A wastes based on intruder protection considerations. Since the stability requirement for Class B waste is of more importance for other reasons besides intruder protection (i.e., migration, operational safety and long-term stability), NRC does not believe changing the 100 year time frame for active institutional control is warranted.

Item 5 - Technologically enhanced natural radiation (TENR) has been defined as those natural sources of radiation that would not normally occur without the presence of some technological activity not expressly designed to produce radiation. Examples might include the release of concentrations of natural radioactivity to the environment from the combustion of coal and natural gas in the generation of electric power or from the mining industry. The staff is not sure of the appropriateness of application of such a concept to the construction activities which might take place at a nearsurface LLW disposal facility. The staff believes that impacts from TENR at a near-surface disposal facility would be similar to those resulting from any large construction project. In addition, such impacts would be very sitespecific with respect to the concentration of natural radioactivity that would be present in the specific site soils and ground water. As such, NRC has not addressed them in this FEIS.

Item 6 - The potential for reduction of thyroid dose due to dilution of radioactive iodine( $12^{9}I$ ) with stable iodine ( $12^{7}I$ ) was briefly considered in the draft EIS. Work performed by Leddicotte, <u>et. al.</u>, was cited, for example. (Reference 16.) At the time, NRC staff believed that while information gaps and time limitations precluded taking credit for dilution by stable iodine in the draft EIS, it was a matter that obviously needed to be further examined. This potential for dilution by stable iodine has in fact been examined in more detail in the final EIS.

As it turns out, whether or not dilution by stable iodine is considered makes essentially no difference in determining whether a waste generator meets waste classification limits obtained from consideration of exposures to an intruder. Iodine-129 typically exists in only trace quantities in low-level waste well below (typically one or more orders of magnitude) the maximum concentration limits. However, dilution with stable iodine may make a large difference in long-term groundwater impacts and consequently any site-specific inventory lists.

Item 7 - The staff believes the current approach is the most reasonable considering the level of information and the generic nature of the analysis. The staff included provisions in proposed Part 61 for improved waste form to be considered on a case-by-case basis. At the present time there is insufficient data (for most waste streams) on the relationship of improved waste forms to reduced plant uptake to allow a credit to be factored into the intruder-agriculture scenario for such improvements.

Several commenters have remarked about the concentration limits in Table 1. As noted earlier, the staff has conducted more realistic analyses and as a result, the concentration limits in Table 1 have been revised. +3

- 2

1.75

38

274

-

÷.

Item 8 - The 100 Ci limit is based on the license conditions for the disposal of gaseous wastes now in effect at the Hanford and Barnwell disposal sites. These limits have not resulted in unsafe environmental conditions at the disposal sites nor have they resulted in overly restrictive situations for waste generators. The 100 Ci limit is consistent with an accident evaluation assuming a dropped package producing occupational exposures to site workers. The DOT limits, however, are established based on accident doses to the public. For tritium and Kr-85 in uncompressed gaseous forms, the DOT limits are 1,000 Ci in a Type A package and 50,000 Ci in a Type B package (173.389(1) and 173.390(a)). For gaseous waste forms the occupational exposure case is the limiting condition.

<u>Item 9</u> - NRC considered the full range of potential pathways of release of radioactivity to the environment. The pathways considered included

groundwater migration, plant and animal intrusion, wind and surface water transport and gaseous releases. (The reader is referred to Appendix H of the DEIS for details.) Based on a consideration of existing data and measurements, NRC concluded that gaseous releases were not a significant release pathway and thus no detailed analyses were performed in the DEIS. NRC did not specifically analyze the range in chemical toxicity of labeled compounds due to a lack of information about the specific compounds used and more importantly, due to the lack of an accepted methodology of assessing biological effects of exposure to such compounds.

Item 10 - The assumption that the waste is placed just before the site is closed is a conservative assumption for purposes of setting waste classification limits. NRC staff believes, however, that the practical effect of relaxing concentration limits for short-lived isotopes (based upon considering accumulation over a 20-40 year period) would be minimal. Such a consideration would have a practical effect only upon the limits for Class A waste containing the isotopes Co-60 and tritium. However, there are other concerns such as site stability, long-term environmental releases, and exposure during handling which argue against raising the Class A limits for these and other isotopes.

.

. · ·

.



Mr. R. Dale Smith Chief, Low Level Waste Licensing Branch Division Waste Management U.S. Nuclear Regulatory Commission Washington, D.C. 20555

· • •

•

PROPOSED RULE PR -61

Dear Mr. Smith:

Attached are comments of the State of New York on U.S. Nuclear Regulatory Commission's September 1981 Draft Environmental Impact Statement on 10 CFR Part 61 "Licensing Requirements for Land Disposal of Radioactive Waste"

In general, we find that this draft EIS was well prepared and required the expenditure of significant effort. The forthcoming Final Environmental Impact Statement should serve as a useful reference document on land disposal of low level radioactive waste if the New York State comments are adequately considered.

Thank you for providing New York State the opportunity to comment on your document.



PDR PR 61 46FR51776 PDR

Acknowladsod by care. 1/21/7 2LN

Comments of the State of New York on the U.S. NRC Sept., 1981 Draft Environmental Impact Statement on 10 CFR Part 61 "Licensing Requirements for Land Disposal of Radioactive Waste (NUREG 0782) ....

**(1)** 

6

(3)

G

### 1. General Comment

The DEIS adequately supports the need for 10 CFR 61 and identifies impacts. However, there is a tendency in the DEIS to compare proposed requirements with practices in use in the late sixties or early seventies. Many of the requirements in Part 61 have been implemented by individual states with regulatory control of low level radioactive waste burial sites. There is some scattered acknowledgement of these actions in the text (see page 35) but the broad recognition of this fact should be emphasized in the DEIS.

The DEIS should also acknowledge that several states, such as New York, have had continuing development of requirements for hazardous waste disposal sites. Many of the requirements in Part 61 are required in State regulation of hazardous waste. For example, the manifest system to be established by Part 20 is also required for hazardous wastes by both EPA and New York State. Vol. 1, p. 53 only refers to EPA's requirements.

### 2. General Comment

The advantages of below grade stability for polyethylene drums should be weighed against the hazards encountered in transit and storage.

### 3. General Comment

Important radionuclides-radium-226, thorium and other naturally occurring radioactive elements are not listed. Their disposal should be covered.

### 4. General Comment

It could be argued that the disposal of large quantities (tons) of source material is not in our best national interest and such material should be stored for later retrieval. The DEIS should address this point.

·: .

### 5. General Comment

Both the proposed rule and the draft EIS should adequately address the possibility of food chain contamination via wildlife.

### 6. General Comment

The EIS should describe the potential health and environmental effects of the various rem doses. It is otherwise impossible to compare advantages or disadvantages of the alternatives.

# 7. General Comment

One important parameter, which should be covered is the heat build up in Class C intruder waste.

-2-

۰.

8. Vol. 1, Page 15 Impact Measures

The last paragraph refers to a long term radiological exposure through potential leaching and transport of the wastes through the groundwater. Another pathway that should be noted is trench overflow and/or pumping of water from the trenches. The pumping of water is referred to on page 30.

9. Vol. 1, Page 16, Table S-4

This table should include the overflow or pumping to surface waters and release of tritiated methane.

10. Vol. 1, page 20, (second and last paragraphs)

The phrase (State or Federal Government) should be added in parenthesis after the phrase "site owner."

11. Vol. 1, Page 23, Controlling the Disposal of Specific Waste Streams

The first paragraph notes most of the longer-term hazard is caused by transuranics isotopes and then states "If these waste streams are eliminated... long term impacts...are only a few mrem/yr. (3 to 5) after 500 years."

The DEIS should acknowledge that the transuranics are banned at this time by action of the states.

However, due to its industrial importance, options for disposal of transuranic nuclides above 10 nCi/g should be addressed.

12. Vol. 1, Page 30, Section 5.12

The fourth paragraph notes that the "bathtub" problem can lead to costly long term pumping. It should also note that this pumping leads to releases to the environment.

13. Yol. 1, Page 53, Section 7.3

Reference is made to the EPA manifest tracking system for hazardous waste. The DEIS should consider the impact and benefits of a common manifest system for radioactive wastes and hazardous wastes.

14. Vol. 1, Page 55, Impacts on the States

Some of the identified actions have been accomplished by New York to some degree by DEC's Part 380 modification that incorporates the requirements of Part 360 for hazardous wastes into Part 380 by reference.

# 15. Vol. 1, Page 55, Impacts on the Public

An additional beneficial impact is the reduction of potential financial risks for taxpayers in states that own burial sites.

-3-

### 16. Vol. 1 Page 56, Impacts on the Public

The adverse impact of allocation of federal and state resources should be offset by establishing fees to cover these costs.

### 17. Vol. 2, Section 103.1 Hypothetical Regional Sites

It would be valuable to include, under Geology, the seismic characteristics of the four regions discussed. This is particularly important when considering the location of the disposal sites and the engineering of the sites.

 $\overline{\partial}$ 

ଚ

(9)

(10

 $\mathbb{P}$ 

Docketed Comment Number: 39

Commenter: New York State Department of Environmental Conservation

<u>Response(s)</u>: <u>Item 1</u> - The staff recognizes that significant improvements in regulatory requirements have been made by the states since the first commercial radioactive waste disposal site was licensed in 1962. Moreover, the staff feels that these improvements were sufficiently noticed in the DEIS. The staff closely interacted with the states to utilize as fully as possible their operating experience and regulatory insights.

With respect to the commentor's second point regarding the role of states in toxic or hazardous waste management, the staff is aware that several states have indeed played prominent roles in this area. This experience may or may not, however, be applicable to the management of radioactive wastes.

Item 2 - High density, high molecular weight polyethylene (HDPE) containers are commonly used in the chemical industry for the shipment and storage of hazardous or corrosive materials. Because of the chemical resistance of HDPE, this material would also provide protection from degradation in a burial environment which is not an aggressive corrosion environment for HDPE. HDPE has been shown to also resist microbial attack. Radiation testing has shown that the mechanical properties of polyethylene do not exhibit significant changes up to radiation exposures of 10<sup>8</sup> rads. The lifetime accumulated dose for all but a few high activity wastes is below 10<sup>8</sup> rads. These properties of HDPE could increase delay time and trench stability to allow for additional radionuclide decay.

One manufacturer of HDPE containers has qualified a 55-gallon size HDPE drumto meet the DOT Specification 7A requirements. The DOT Specification 7A requirements qualifies a container as a Type A package for the shipment of radioactive materials. The standard Specification 17C and 17H 55-gallon carbon steel drums are also capable of meeting Specification 7A. Therefore, for transportation purposes the HDPE containers can be considered to be equivalent to standard carbon steel drums.

For waste storage HDPE drums provide good corrosion protection which could minimize leakage and waste spills. However, because of the viscoelastic properties of polymers, adequate creep strength must be designed into the container to prevent material deformation and possible rupture under the proposed loads during the storage period. Chemical industry storage experience indicates that properly designed containers can be provided at competitive costs for storing materials for extended periods. Storage of radioactive materials in HDPE containers would require specific consideration of the proposed storage period, stacking arrangements, container weight, and the specific container design to minimize any adverse effects from material creep.

Item 3 - In the initial effort to develop the draft Part 51 rule and draft EIS, NRC staff concentrated on the principal moderate and long-lived radionuclides and principal types and forms of radioactive waste. Guidance for disposal of other radionuclides and waste forms such as radium-226 and other naturally occurring elements will be addressed subsequently.

٠-,

-\*

Item 4 - NRC agrees that for the reason stated and also to reduce the volume of waste required for disposal, consideration should be given to storage of large quantities of source material for possible later use. However, the purpose and scope of the EIS for Part 61 is that it serve as a decision document for the performance objectives and technical and other requirements of Part 61. It is not a generic EIS and is not intended to judge the utility or political considerations involved in deciding whether a particular material should or should not be considered a waste. As such, Part 61 establishes the requirements to ensure safety in disposal of material that could be considered waste. In addition, based on a review of the various waste streams in Appendix D of the DEIS containing source material, those indicating large quantities of source material contain depleted uranium not natural uranium.

<u>Item 5</u> - Wildlife could conceivably be contaminated by radioactive materials at a low-level waste disposal facility in several ways: ingestion of contaminated water, consumption of contaminated plants, burrowing into buried waste and/or direct exposure to exposed waste. The performance

objectives and technical requirements of proposed 10 CFR Part 51 would minimize the probability as well as the severity of such an occurrence. Assuming the objectives and requirements of the rule are met, the staff believes that wildlife contamination on a significant scale is unlikely to occur and that this event does not pose a threat to public health and safety.

Item 6 - In the DEIS, NRC did not convert or express exposures in terms of risks because of the difficulty of accurately assessing risks to future populations from exposures incurred at future times and the small number of individuals involved who could receive a potential exposure. In the DEIS, NRC compared calculated doses on a common basis to exising standards which are expressed in terms of dose equivalent. NRC has, however, attempted to express the overall impacts of Part 61 in the EIS in a clearer manner such that comparison of alternatives and unmitigated impacts are easier to discern and understand. In addition, in response to this comment and to place in perspective the potential risk associated with doses calculated in this FEIS, NRC has included a section in the summary which provides dose response relationships as set forth in International Commission on Radiation Protection Publication 26. The reader can use these relationships to estimate the level of risk associated with doses calculated for various alternatives.

Item 7 - As part of the work performed for the final Part 61 rule, an analysis was performed on the practicality of some of the limits proposed in Table 1. This analysis included both surface radiation levels and potential heat buildup.

Item 8 - NRC staff does not believe that trench overflow treatment and release of pumped leachate represent a long-term radiological exposure hazard. Trench overflow is a short-term potential hazard which would be avoided by leachate pumping and treatment. The leachate pumping and treatment operations, however, would involve short-term releases to the environment. Pumping and treatment operations would also be quite expensive. This does not mean, however, that NRC believes that such potential short-term releases are not important. The impacts from trench overflow and leachate treatment have been calculated for the final EIS and support NRC's position regarding the need for disposal site stability.

.

Item 9 - Potential impacts from trench overflow or releases due to leachate pumping and treatment have been included in the calculations for the final EIS. From data available to NRC at the time the draft EIS was written, however, it appeared that potential impacts from tritiated methane released through disposal trench covers are at negligible levels. NRC staff is not aware of any new data which would contradict this belief. Thus, potential impacts from tritiated methane were not calculated for the final EIS.

. . . .

<u>Item 10</u> - This change has been made, and the reader is referred to Appendix E, Errata in this volume.

Item 11 ~ The staff acknowledges that burial of transuranics above 10 nCi/gm is, in fact, banned at this time. In preparing limits for disposal for the final 10 CFR Part 61, the routine disposal of transuranics between 10 and 100 nCi/gm has been allowed as Class C waste.

<u>Item 12</u> - Potential releases to the environment from leachate pumping and treatment have been considered in the final EIS.

Item 13 - The proposed new §20.311 of 10 CFR Part 20 would establish a manifest tracking system for radioactive waste shipments. The provisions of §20.311 are general in that required information is listed and general requirements to forward and use manifests are included. The requirements were developed to provide information needed for disposal and to be compatible with existing DOT requirements for radioactive shipments.

The staff reviewed the U.S. Environmental Protection Agency (EPA) regulations for hazardous waste manifests. The same general approach for tracking was used. The staff also looked into the Uniform Hazardous Waste Manifest proposed by a joint EPA and DOT rule change published March 4, 1982 (47 FR 9336). The proposed rulemaking would require the use of a specific manifest form for all hazardous waste shipments. The form was proposed to provide relief to shippers from the individual state requirements on manifest contents. Individual states developed their own form which meet Federal requirements but also required additional state information. Thus shippers crossing states were faced with potentially differing requirements on forms in each State. The proposed form is a one page document carefully tailored to provide required and needed information on hazardous wastes. The form may be used as an NRC manifest or DOT shipping Daper for radioactive wastes by using additional lines to provide information on radioactive materials. A few minor procedural and terminology changes were made to the final Part 61 rule to conform to the proposed EPA/DOT manifest.

The manifest requirements in §20.311 of the NRC rule change are compatible with DOT supplementary requirements for radioactive material. Neither DOT or NRC prescribe specific forms for information on radioactive waste shipments. . . . . . .

Item 14 - The staff recognizes these actions. However, modification of existing Agreement State programs to assure compatibility with 10 CFR Part 61 is an action separate from that mentioned by the commenter.

Item 15 - The staff concurs wih the commenter's opinion and will include this beneficial impact in the final EIS.

.. •

Item 16 - Allocation of federal resources will be offset by licensing and inspection fees as set forth in 10 CFR 170. With respect to the allocation of state resources, each Agreement State that would be the location of a new facility will be able to establish its own system of fees or other compensation on a case-by-case basis to allow regulatory costs to be compensated. These arrangements are within the state's discretion and are not addressed by 10 CFR 61 or the EIS.

Item 17 - The seismic characteristics for each of the four regions is included in the detailed descriptions of the sites contained in Appendix J of the DEIS. (See, for example, the last paragraph of Section 1.4.1 of Appendix J, "Geology" for the Southwestern Site.) It was not included in the summary descriptions contained in Chapter 10 of the DEIS.

. ·



900 YE72

'82 FEB -4 P1:28

COMMONWEALTH of VIRGINIA

J & JACKSON JR ADMINISTRATCH

Council on the Environment

903 NINTH STREET OFFICE BUILDING A-CHMONO 23219 804-785-4500

WARDER NORDER

U. USED RUL

January 12, 1982

Mr. R. Dale Smith Chief, Low-Level Waste Licensing Branch Office of Nuclear Material Safety and Safeguards U.S. Nuclear Regulatory Commission Washington, D.C. 20555

Dear Mr. Smith:

The Commonwealth of Virginia is in the process of reviewing the Draft Environmental Impact Statement on New Licensing Requirements for Land Disposal of Radioactive Waste (10 CFR Part 61). The Council on the Environment is responsible for coordinating the State's review of federal environmental impact statements and responding to appropriate federal officials on behalf of the Commonwealth. The following agencies have responded to our review request at this time:

Department of Agriculture and Consumer Services Department of Conservation and Economic Development Department of Health

Office of Emergency and Energy Services Virginia Research Center for Archaeology.

In addition, we expect comments from at least two other state agencies in a few days.

The Draft EIS gives rise to only limited concerns on the part of the responding agencies above. There is presently no site for disposal of low-level radioactive waste in Virginia. If any sites are planned for use, the Virginia Research Center for Archaeology should be contacted before site disturbances begin.

It is possible that farm operations might one day intrude into a radioactive waste site; effective site controls could foreclose this possibility.

The document's discussion of geologic requirements for waste disposal is adequate.

Mr. R. Dale Smith January 12, 1982 Page 2

۰.

We expect to furnish additional comments within two weeks. Thank you for the opportunity to review this document.

Sincerely,

.

Enclosures

CC: The Honorable Maurice B. Rowe, Secretary of Commerce and Resources Mr. Earl A. Finch, Department of Agriculture and Consumer Services Mr. Randolph Turner, Virginia Research Center for Archaeology

Mr. Bruce B. Meador, Department of Conservation and Economic Development

JBJ/CHE/all

### REVIEW INSTRUCTIONS:

- A) Please review the document carefully. If the proposal has been reviewed earlier (e.g., if the current document is a FINAL EIS), please consider previous comments.
- B) Prepare your agency's comments in a form which would be acceptable for responding directly to a project sponsoring agency.
- C) Use the space below for your comments. If additional space is needed, please attach extra sheets.

Return your comments to:

Charles H. Ellis III Environmental Impact Statement Coordinator Council on the Environment 903 Ninth Street Office Building Richmond, Virginia 23219

CHARLES ENVIRONMENTAL IMPACT STATEMENT COORDINATOR COMMENTS NRC

ADEQUATELY

8 (DATE) (SIGNED) 21 . • (TITLE) (AGENCY)

CC GEVE Ridge, MR

- 8. REVIEW INSTRUCTIONS:
  - A) Please review the document carefully. If the proposal has been reviewed earlier (e.g., if the current document is a FEMAL EIS), please consider previous comments.
  - D) Propage your agency's comments in a form which would be acceptable for responding directly to a project sponsoring agency.
  - C) Use the space below for your comments. If additional space is needed, please attach extra sheets.

Return your comments to:

Charles H. Ellis III Environmental Impact Statement Coordinator Council on the Environment 903 Ninth Street Office Building Richmond, Virginie 23219

ENVIRONMENTAL IMPACT STATEMENT COORDINATOR

### CONMENTS

We have made a review of the Draft Environmental Impact Statement on 10 CFR Part 61 "Licensing Requirements for Land Disposal of Radioactive Waste."

Currently no site for this Class of waste exist in Virginia.

The Draft EIS for licensed sites in the U.S. for the disposal of radioactive low level waste (LLW) does not present any problems for agriculture as far as we can determine. The exception to this would be a Very low probability of inadvertent intrusion of agricultural operations, including farm livestock into a closed (terminated) disposal facility through the first 100 years and into a 500 year span. However, since these sites are relatively small, <u>dedicated</u> sites; it would seem that the period of institutional control would be "open ended" to extend throughout a "safe" time frame that would be determined by site monitoring.

(SIGNED) (DATE) January 6, 1982 Earl A. Finch (TITLE) Resource/Environmental Economist (AGENCY) Dept. of Agriculture and Consumer Services Cç ...

 $\infty N C$ CN TH-ENNRONMEN Commonwedth of Viano TO: File FROM: C. Ellis SUBJECT: NRC Draft EIS on Licensing Requirements for Land Disposal of Radioactive Waste Virginia Research Center for Archaeology comments, per Randy Turner on January 5, 1992, are as follows: "If any land disposal sites are planned for Virginia, the Virginia Research Center for Archaeology, as the archaeological agency 2 representing the Virginia State Historic Preservation Officer, should be contacted pursuant to federal legislation on archaeological resources." CHE/all



### Docketed Comment Number: 40

Commenter: Commonwealth of Virginia, Council on the Environment

<u>Response(s)</u>: <u>Item 1</u> - The commenter is correct that disposal of LLW under Part 61 should not present any problems for agriculture. The commenter is also correct that there is a low probability of inadvertent intrusion into the site for agricultural purposes. NRC staff believe that active institutional control of the site following site closure should preclude such intrusion from occurring during the 100-year institutional control period. After that time passive institutional controls such as continued government land ownership and land records should reduce the potential for such inadvertent intrusions.

Item 2 - In the NEPA-mandated environmental review and licensing process of a proposed near-surface disposal facility, NRC will operate under the requirements of Federal laws and regulations for the protection of cultural resources. Among other things, these requirements include coordination with the State Historic Preservation Office, conduct of a pre-construction cultural resources survey and the identification of mitigating measures to protect any known or encountered resources.



United States Department of the Interior OFFICE OF THE SECRETARY WASHINGTON, D.C. 20240-82 FE3 -2 P3:18 IAN 28 1982 ene ER 81/2250 Secretary of the Commission DOCKET NUMBER **U.S. Nuclear Regulatory Commission** PROPOSED RULE DOCKET NUMBER Washington, D.C. 20555 PROPOSED RULE Dear Sir: We have reviewed the proposed licensing requirements for Land Disposal of Radioactive -Waste (10 CFR 61).

### **General Comments**

751O

5 11

102;

an Deser

In general, we are in agreement with the overall technical philosophy and strategy employed in the rule. It appears to be a major step forward in improving low-level radioactive waste management. We concur that many acceptable sites should be possible In most areas throughout the country and we agree, in general, with the flexibility and conservatism of the combined prescriptive and performance objective approach. Finally, we agree that waste classification is a cornerstone for a good waste management rule and that the classification scheme proposed is sensible and practical in terms of radionuclide content.

Our principal concern with the waste classification system is that it fails to address nonradioactive toxicity of the waste. We believe that any waste classification scheme should be based on total hazard. It would seem inappropriate for a particular waste to be declared as Class A radiologically when it might contain toxic metals or organic compounds with potential harmful effects several orders of magnitude greater than those of the radionuclides. Perhaps the rule should either prohibit components with greater potential toxicity than the radionuclides or provide for additional classification options based on other-than-radiological toxicity. This position would be consistent with recommendations of the Conservation Foundation Dialogue Group on Low-Level Radioactive Waste and the Department of Energy's Task Force on Radioactive Waste Management.

We believe that waste volume reduction is an important element in reducing overall magnitude and complexity of the problem. We therefore suggest giving it greater emphasis and perhaps offering additional incentives. We also believe that Federal/Defense generated low-level radioactive waste should come under the same earth-science guidelines and criteria.

### Participation of Bureau of Indian Affairs (BIA) and the Indian Tribes

The Secretary's trust responsibilities apply to waste disposal sites that are to be located on Indian reservations. The Bureau of Indian Affairs (BIA) provides the support services necessary to carry out the Secretary's trust responsibilities. Therefore, BIA area directors or their authorized representatives should be invited to participate in the review of applications for the location of low-level radioactive waste disposal sites on Indian reservations. In addition,

8202090148 820128 PDR PR 2 46FR38081 PDR

### Secretary of the Commission

regulatory policy should also be coordinated with BIA's Office of Trust Responsibilities with respect to locating, licensing, operating and maintaining commercially operated disposal sites on Indian lands. Because of variations and changes in Indian land ownership, BIA's involvement will be essential.

Specific comments on the proposed rule and the environmental impact statement are attached separately.

We hope these comments will help you in the preparation of a final statement.

Sincerely,

Bruce Blanchard, Director **Environmental Project Review** 

2

Enclosures

# PROPOSED RULE

The following more specific comments are addressed to the rule itself but would also apply to corresponding sections of the Summary and the environmental statement.

### Section 61.2, Definitions

<u>Buffer Zone</u>. The buffer zone definition should include depth as well as lateral boundaries, and should be described as a three-dimensional zone. The performance standards might then apply to releases beyond the boundary of the buffer zone. Unrestricted use of land and resources beyond the buffer zone (laterally or at depth) would then be allowable during and after site operations.

### 61.50 Disposal Site Suitability Requirements for Land Disposal

(aX2) We suggest being more specific in the "modeling" requirements. Do you mean physical scale model? numerical ground water/solute transport model? conceptual model? (There are many kinds of possible models.)

(aX?) We endorse this option but suggest using "molecular diffusion" in place of "diffusing" and/or defining maximum hydraulic conductivity allowable such as  $10^{-6}$  cm/sec.

### 61.51 Disposal Site Design for Land Disposal

(aX4) It is impossible to totally "prevent" infiltration; suggest using "minimize" in place of "prevent."

(aX6) This requirement appears inconsistent with 61-50(aX7). The option of disposing in the saturated zone should be mentioned again.

### 61.52 Land Disposal Facility Operation and Disposal Site Closure

(aX8) We believe the location of the buffer zone should be determined on the basis of site performance. The zone ideally would be enclosed within a three-dimensional surface surrounding and underlying the burial site. Our concept of the buffer zone is a zone that provides a controlled/restricted-access volume of earth material around and under the site, beyond which unrestricted use of land and resources, surface or subsurface, could be allowed during and after site operation. The 100 foot lateral extent listed in the rule appears somewhat arbitrary.

### 61.53 Environmental Monitoring

(a) We believe that "geochemistry" should be listed with the other subjects (ecology, meteorology, climate, hydrology, etc.). Although geochemistry is often an implied aspect of hydrology, we believe it deserves specific mention because it plays such an important part in radionuclide mobility in ground water.

### 6155 Waste Classification

General: We believe waste should be classified according to total toxicity as described above under "General Comments."

2

(d) It is not clear what the disposition will be of wastes which exceed Class C concentrations. What type of disposal is envisioned by the Commission for those wastes?

### 6156 Waste Characteristics

General: The non-radiological toxicity of the waste needs to be considered here, we believe. It is apparently ignored.

DRAFT ENVIRONMENTAL IMPACT STATEMENT		
General Comments		Page 7-8, paragraph 4, line 5 - "C-137" should read "Cs-137."
Evidence of the life expectancy and other advantages of polyethylene drums over steel or others should be presented. Will they simply delay the compaction/degradation problem? The same question also applies to wooden boxes—are they significantly better than cardboard?		Page 7-22, Mixtures of Radioisotopes, next-to-last line – Insert closing parenthesis.
During extended crý periods, desiccation cracks tend to form in trench caps due to shrinkage of fine grained minerals. These cracks can extend several feet vertically and can provide avenues for rapid water infiltration. How can this problem be avoided?	0	
We question the long-term viability of using plastic sheeting as an infiltration barrier in the trench cap. How can its integrity be assured? What is the evidence that this technique works consistently?	3	
Also, the extent of hazard presented by low level radioactive wastes is not clearly indicated i the DEIS. While low level radioactive wastes are indicated for some of the more dangerous isotopes, a general definition of hazard or risk should be presented in the introduction to understand impact analyses.	" (4)	· · ·
Cultural Resources Protection	1.0	· · ·
Section 2.3 of Appendix E of the DEIS discussing a reference disposal facility makes only pressing reference to historic areas. Section 3 should make specific reference to historic and archeological resources.	6	
Minor Comments		
<u>Volume 1</u>		
Page 3, 1.5 Scoping for the EIS, line 2 - "501.7" shoud be "1501.7."	6	
Page 55, The role of the U.S. Geological Survey should be defined.	( <del>)</del>	
<u>Volume 1</u>		
Page xi, top line - "Chapter 10" should read "Chapter 9."	16	
kage 3-23, 3.5.2, line 16 - Identify the levels of "little high energy gamma emitting radionuclides."	$  \Phi  $	΄.
Line 19 - Identify the level of "large quantities of high energy gamma emitting radionuclides."	' @	· · · ·
Page 3-31, paragraph 2 - Would trench liners be required?	10	
Page 4-8, 4.3.2, line 11 - "of" should read "at."	ID	
Page 4-69, Requirement, 2 - Delete "economically."	i 🚯	
Page 5-69, table - Units should be presented.	113	· · · · · · · · ·
Page 6-11, last paragraph, line 5 - Change ".033" to ".33."	16	

(B) (T)

•

### Docketed Comment Number: 41

Commenter: U.S. Department of the Interior, Office of the Secretary

<u>Response(s)</u>: Item 1 - High density, high molecular weight polyethylene (HDPE) containers are commonly used in the chemical industry for the shipment and storage of hazardous or corrosive materials. Because of the chemical resistance of HDPE, this material would also provide protection from degradation in a burial environment which is not an aggressive corrosion environment for HDPE. HDPE has been shown to also resist microbial attack. Radiation testing has shown that the mechanical properties of polyethylene do not exhibit significant changes up to radiation exposures of 10<sup>8</sup> rads. The lifetime accumulated dose for all but a few high activity wastes is below 10<sup>8</sup> rads. These properties of HDPE could increase delay time and trench stability to allow for additional radionuclide decay.

2

One manufacturer of HDPE containers has qualified a 55-gallon size HDPE drum to meet the DOT Specification 7A requirements. The DOT Specification 7A requirements qualifies a container as a Type A package for the shipment of radioactive materials. The standard Specification 17C and 17H 55-gallon carbon steel drums are also capable of meeting Specification 7A. Therefore, for transportation purposes the HDPE containers can be considered to be equivalent to standard carbon steel drums.

For waste storage, HDPE drums provide good corrosion protection which could minimize leakage and waste spills. However, because of the viscoelastic properties of polymers, adequate creep strength must be designed into the container to prevent material deformation and possible rupture under the proposed loads during the storage period. Chemical industry storage experience indicates that properly designed containers can be provided at competitive costs for storing materials for extended periods. Storage of radioactive materials in HPDE containers would require specific consideration of the proposed storage period, stacking arrangements, container weight, and the specific container design to minimize any adverse effects from material creep.

Wooden boxes provide much more protection to workers during the waste handling operation at the disposal site than do cardboard boxes. In some cases, contaminated syringes have penetrated cardboard boxes during transit causing injury to workers. While wood boxes are not expected to provide substantially greater protection against subsidence than do cardboard boxes, they do provide greater safety during handling. •

. •

Item 2 - There are techniques available, particularly layered systems which will prevent or significantly restrict development of dessication cracks in trench covers. These are currently being studied under an NRC contract with the the Illinois State Geological Survey.

Item 3 - The proposed Part 61 does not require the use of moisture or infiltration barriers. Such barriers were considered as an alternative but were rejected because of the staff's concern for their long-term viability. They may be approved, however, for use on a case-by-case basis. If they were to be used for Class A wastes with relatively short radiological hazard, plastic membranes may have sufficient proven design life. Other methods such as filter fabric may be viable for longer periods in a layered cover due to the advantages of such a cover, i.e., nearly constant temperature, moisture content, soil water chemistry, protection against burrowing animals or root penetration and protection against radiation from above or below.

Item 4 - See staff response to Item 5, Comment 15.

Item 5 - The staff recognizes that construction and operation of land disposal sites for radioactive waste disposal may result in impacts to cultural resources. In preparing the draft EIS, the staff felt that these impacts were site-specific in nature and could not be adequately assessed in the absence of a specific site proposal. In the NEPA-mandated environmental review and licensing process of a proposed disposal facility, NRC will operate under the requirements of Federal laws and regulations for the protection of cultural resources. Among other things, these requirements include coordination with the State Historic Preservation Office, conduct of a pre-construction cultural resources survey and the identification of mitigating measures to protect any known or encountered resources. <u>Item 6</u> - This change has been made. See Appendix E, Errata in

·. •

this volume.

Item 7 - The role of the U.S. Geological Survey has been included in the FEIS.

<u>Item 8</u> - This change has been made. See Appendix E, Errata in this volume.

Item 9 - Identification of these levels is provided in Chapter 4.0, Transportation Impacts, Volume 3 of NUREG/CR-1759, <u>Data Base for Radioactive</u> <u>Waste Management</u>. (Reference 3.)

Item 10 - Identification of these levels is provided in Chapter 4.0, Transportation Impacts, Volume 3 of NUREG/CR-1759, <u>Data Base for Radioactive</u> <u>Waste Management</u>. (Reference 3.)

<u>Item 11</u> - The commenter's reference is to a description of a reference disposal facility. Trench liners would not be required and are not a requirement of Part 61.

.

 $\underline{\text{Item 12}}$  - The reader is referred to Appendix E, Errata in this volume.

<u>Item 13</u> - The staff has considered this comment and determined that it would be appropriate to delete the word "economically" both here and in the proposed 10 CFR Part 61.

<u>Items 14-17</u> - The reader is referred to Appendix E, Errata in this volume.

A-121

.

Utility Nuclear Waste Management Group

1111 19th Street, N.W. . Washington, D.C. 20036 + (202) 828-7669

PR 61 (46 FR 5776)

0000000000

U. S. Nuclear Regulatory Commission Page 2 February 4, 1982

Another matter of almost equal importance is discussed in the enclosed comments under the heading of "Educational and Public Information Aspects." One of the main functions of an environmental impact statement is to provide the public with an informed insight into the nature, scope and magnitude of relevant issues. The DEIS, however, falls so far short of this goal as to render it inadequate as an environmental full-disclosure statement.

Such a deficiency is not merely cosmetic, nor is it of only legal and academic importance. It can have real, undesirable effects. By failing to present the matter of low-level waste disposal in full and appropriate context, the DEIS is counterproductive to the NRC supported goal of opening new regional burial sites. If knowledgeable, widespread community acceptance of burial sites is to be achieved, the NRC must make the effort necessary to properly inform the public of how reasonable an undertaking the establishment of such a site really is. The environmental impact statement on Part 61 provides an excellent opportunity to do so and should be fully utilized to help dispel basic misapprehensions concerning the perceived dangers of low-level waste disposal.

In concluding, the UNWMG acknowledges that the enclosed comments tend to be rather critical. In addition, we recognize that they are somewhat belated. It is hoped, however, that these comments will be received and utilized by the NRC in the constructive spirit in which they are offered.

Sincerely L. Stanford Ε. Program Manager

(1)

RELS:1k Enclosure

MEMBERS -labama Power Company Arizona Public Service Company Balumore Gas & Electric Co. DIA Erston Laison Company Carolina Power & Licht Company The Cloveland Electric Ruminating Company Company Consolidated Láson Co. of story fork, Inc. The Detroit Educe Co.Juany Duke Fourt Company Dudweine Light Company Fichda Forer Carbonation Florida Pa .er & Light Company Georgia Power Company Guil States Utilities CO.AD-MY Nousica Lighting & Power Company illinois Power Company Iowa Lierthe Light & Power Cumpany Lone Island Lighting Company for A. Jeles, Department of Rater & Fower Middle South Services. Rebraska Public Pome District Nistara Hobawk Power Corporation forment militi a Northern State, Power Company Pennsylvania Power & Light Company Philadelphia Electric Company Portiand General Liectric Company Power Authority of the Slace of New York Public Corvice Electric & Gas Come my Sacra nento Municipal Utility District SHUTTS Southern Critil, min Edison Company Texas Litr'iti.s Company Tole to Edison Company Virginia Liectric & Powe Company Wisconsin Public Service Corporation Wisconsis Elector Tone Company tanker Atomic P 8202110309 820204 Company 61 46FR51776

CZ FH -0 29:44 and February 4, 1982 --GUSTET PHONOD U. S. Nuclear Regulatory Commission Washington, D.C. 20555 PHOPDSED RULE Attn: Chief, Low-Level Waste Licensing Branch (46 FR S1776 Division of Waste Management Office of Nuclear Material Safety and Safequards Re : Draft Environmental Impact Statement on 10 CFR Part 61 "Licensing Requirements for Land Disposal of Radioactive Waste, \*\* NUREG-0782 (46 Fed. Reg. 51,776) Dear Sir: The enclosed comments are submitted on behalf of the Utility Nuclear Waste Management Group (UNWMG) is connection

On January 14, 1982, the UNWMG submitted its comments on the NRC's proposed licensing requirements for land disposal of radioactive waste, 10 C.F.R. Part 51 (46 Fed. Reg. 38,081). In preparing those comments, the proposed rule was used as a point of departure. That is, the regulations in the proposed Part 61 were basically taken as a given. From that perspective, the comments focused on offering suggestions aimed at making the rule, as proposed, more workable.

Review of the Draft Environmental Impact Statement (DEIS) on Part 61, however, has raised some fundamental questions concerning the basic desirability of certain aspects of the proposed rule. This is addressed in the enclosed comments under the heading: "Cost/Benefit Analysis."

In essence, the UNNMG is concerned that a failure to analyze and compare the costs and benefits of various alternatives in a systematic and rigorous fashion has resulted in a proposed rule which, if adopted, would greatly increase the cost of disposal with few -- if any -- benefits. Such an outcome would be wasteful and, thus, inconsistent with one of the basic concerns underlying NEPA; i.e., the efficient management and utilization of available resources. DS10 ADDI

A Program Administered by Edison Electric Institute

PDR

with the above-referenced matter.

A-122

2.0.5-1

# Enclosure

.

Comments of the UNWMG on che "Draft Environmental Impact Statement on 10 CFR Part 61 'Licensing Requirements for Land Disposal of Radioactive Waste'" (NUREG-0782)

The UNWMG has reviewed the Draft Environmental Impact Statement (DEIS) on Part 61 and, as a result, developed comments in the following areas:

o Basic Presentation

o Educational and Public Information Aspects

o Cost/Benefit Analysis

Each of these matters is discussed below in ascending order of importance.

Basic Presentation

As organized and written, the DEIS is almost incomprehensible. The multiplicity of cases, alternatives, and activity spectra, which are neither carefully defined nor used consistently, has led to a very disjointed treatment.

We would recommend that a much smaller number of cases -spanning the expected range of activity levels -- be selected for presentation. The presentation should then be developed to lead to summary tables patterned after those suggested in Attachment A. The UNWMG further suggests that the more simplified discussion of a smaller number of cases and spectra

include data on specific isotopic contributions to doses, since actions appropriate to control one isotope may not be at all appropriate for another.

### Educational and Public Information Aspects

One of the more important functions of an environmental impact statement is to provide the public with a reasoned insight into the nature, scope and magnitude of relevant issues. The DEIS, however, falls so far short of this goal as to render it inadequate as an environmental full-disclosure statement.

First, the entire document reads as though a serious public health and radiological protection problem were being addressed whereas, in truth, the shallow land burial of lowlevel nuclear waste is essentially a non-problem in these<sup>1</sup> respects. Using the population dose estimates presented in the DEIS, we have calculated the annually expected <u>statistical</u> deaths from current shallow land burial practices and, in Attachment B, compared this risk to other causes of death. As can be noted, the risk from shallow land burial is four orders of magnitude less than that associated with activities as prosaic as the use of skate boards.

Of course, we fully recognize the institutional issues that, unfortunately, surround and confuse the matter of lowlevel waste disposal. However, an EIS should, at least, · · · ·

(3)

. .

6)

- 3 ~

3 - C - C - C

strive to put the area under consideration into a proper and understandable scientific and technical perspective. Unfortunately, the existing document not only fails to do this but, as indicated earlier, exacerbates existing misunderstandings and misconceptions regarding shallow land disposal.

It is important to recognize that this deficiency is of more than just legal and academic importance. It can have real and undesirable effects. By failing to present the matter of low-level waste disposal in its full and appropriate context, the DEIS is counterproductive to the NRC supported goal of opening new regional burial sites. If knowledgeable, widespread community acceptance of burial sites is to be achieved, the NRC must properly inform the public of how reasonable an undertaking the development of such a site really is; not just with numbers, but on a conceptual, comparative basis as well. The EIS on Part 61 presents an excellent opportunity to do so, and should be fully utilized to help dispel basic misconceptions concerning the dangers of low-level waste disposal.

Another misconception, enhanced by and threaded throughout the DEIS, concerns the oft assumed dominance of the water pathway. (See, <u>e.g.</u>, Vol. I, p. 17; Vol. II, p. 5-1.) Only a very careful reading reveals the fact that, with the exception of four isotopes, all isotopes are limited by intrusional pathways and the possible doses are restricted to a very few individuals. This distinction is very important to public understanding, for the perception of widespread water contamination is quite a different matter than that of a few individual intruders being exposed. Once again,

- 4 -

a disservice is done to the public by the portrayal of a less than accurate -- or, at least, easily misunderstood -- picture.

The basic lack of perspective embodied in the DEIS is, perhaps, best illustrated by the following sentence taken from page one of the Summary volume:

### Space disposal, <u>although feasible</u>, is not developed to the point of routine technical and economic application.

(Emphasis added.) Such a statement adds materially to public misconceptions. Even the suggestion that space is considered a possible alternative conveys the impression that we are dealing with a problem of truly cosmic proportions. Space disposal of low-level waste will never be "feasible," within the dictionary definition of that word. Space disposal, if mentioned at all, should be dismissed with a single sentence acknowledging its inapplicability to the subject at hand.

### Cost/Benefit Analysis

An EIS should specify, in a clear, concise and meaningful way, the costs and benefits associated with the various alternative actions considered. In this respect, too, the

(4)

3

(and)

DEIS is plainly deficient.

We have -- as best we could given the difficulties arising from the presentational shortcomings of the DEIS, mentioned earlier -- performed a cost/benefit analysis of 14 of the alternatives to present practice (the "no-action case"). The results of this analysis are summarized in <u>Attachment C</u>, Tables 1216/1, 1216/2 and 1216/3.

- 5 -

As the tables show, not a single alternative considered in the DEIS meets the Appendix I cost/benefit test criteria (\$1000/man-rem, escalated to about \$1700/man-rem to account for inflation since 1975), and many of them are highly nonbeneficial. This valuable and important information is nowhere presented in the DEIS, and can be extracted only with considerable effort.

In addition, it should be noted that the analytical results presented in the tables do not take into account the effect on occupational exposure of the various alternatives. Total occupational exposure, over the operational lifetime of the burial site analyzed in the DEIS, routinely runs ten times the non-occupational exposure calculated over 2000 years. Since most of the alternatives involve increased waste handling, with concomitant increases in occupational exposure, it is likely that, when this factor is taken into account, many of the alternatives will increase total exposure rather than decrease it. To assure that these impacts are all clearly identified and evaluated, the UNWMG believes that the final EIS must contain a clear, concise and meaningful cost/benefit analysis, plus a detailed evaluation of alternatives based on the results of that analysis.

- 6 -

. . .

(cait

18

۰.

73

2

In concluding; we wish to emphasize that -- while the foregoing comments are wholly critical -- they are offered in a genuine effort to be constructive. Providing for the safe, cost-effective disposal of low-level waste in an environmentally sound manner is a matter of national importance. As such, it is a matter concerning which the public deserves to be fully informed, in as clear and complete a manner as possible. While we are hopeful that the final EIS will adequately address the relevant issues, the current draft, as discussed above, can only be viewed as being far from adequate.

The UNWMG, of course, remains willing to assist in the constructive revision of the DEIS and would be pleased to provide any further assistance which the Commission deems desirable and appropriate.

Ð

Cort



### Attachment A

SUMMARY TABLE

Max	cimum Inc	lividual	Dose	<u> </u>	Con	t Beneficia	ality
Alter-	mrem/	Path-	Dominant	∆ Dose*	Δ Cost*	Referred	Cost Benefit Ratio
native	year	way	Isotopes	man-rem	\$	to Case	\$/man-rem

# SUMMARY TABLE

	Required	•	Part 61
Alternative	By		Reference

.

.•

. . .

.

•

• .

$ \begin{array}{c} Cost Beneficiality of Alternatives Ordered on Dase cost S104 Costs Costs Cost S104 C$				Table 1216/1					
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		Cost Be	reliciality	of Alternati	ves Ordered	on Dose			
marres         from Ref fr New Back           cratics         2.508-01         1.208-03	Description	2000 year Pop Dose		Dosea fr Nov Bee	Cost \$10 <sup>8</sup>	8 7	ista fr New Ba	Cost/Bene 46 £/mo	fit Patlo A-rem
cattle         2.500.01         -         -         4.28         1.000.06         -         2.500.01         -           state         Transition         1.1000.01         1.2000.01         1.2000.01         1.2000.01         2.500.01         2.500.01         2.500.01         2.500.01         2.500.01         2.500.01         2.500.01         2.500.01         2.200.01         2.700.01         2.600.01         2.200.01         2.700.01         2.600.01         2.200.01         2.700.01         2.600.01         2.700.01         2.700.01         2.600.01         2.700.0	4	Nun-r CA						from Ref	Lr New Base
Baste Fut         1.14E-00         1.12E-00         1.2E-00         1.2E-00         1.2E-00         1.2E-00         1.2E-00 <th< td=""><td>sent Practice</td><td>Z.50E+03</td><td>1</td><td>1</td><td>4.28</td><td>907300 C</td><td>1</td><td>10735 6</td><td>1</td></th<>	sent Practice	Z.50E+03	1	1	4.28	907300 C	1	10735 6	1
isponal linked latent 4.05 2.702-07 1.920-04 1.5203 Pacility linked 1.42201 1.002-01 4.53 2.702-07 1.920-04 1.5203 Covera 9.346-01 1.42201 1.402-01 4.53 1.5026-07 1.2026-07 1.202-03 1.5203 Covera 9.346-01 1.422-01 1.402-01 4.31 6.002-06 9.002-06 1.8203 Covera 9.306-02 1.522-01 1.402-01 4.31 6.002-06 9.002-06 1.8203 a 9.106-02 1.522-01 1.402-01 4.33 5.102-07 4.526-07 1.520-03 a 9.106-02 1.562-01 1.002-01 4.39 5.102-07 4.526-07 1.520-03 a 6.622-03 1.662-01 1.002-01 4.39 5.102-07 4.526-07 1.520-03 a 6.622-01 1.662-01 1.002 00 7.31 9.5102-07 4.526-07 1.520-03 7.726-03 1.662-01 1.002 00 7.31 9.5102-07 4.520-07 1.520-03 Covera 8.622-02 1.662-01 1.002 00 7.31 9.5102-00 4.520-03 2.520-03 Covera 8.622-02 1.662-01 1.002 00 7.31 9.5102-00 4.500-03 2.520-03 Covera 8.622-02 1.962-00 1.002 00 7.31 9.540-00 4.120-06 7.120-03 4.020-04 Covera 8.622-02 1.962-00 1.002 00 7.31 9.540-00 4.120-06 7.120-03 4.020-04 Covera 8.622-02 1.962-00 1.002 00 7.31 9.540-00 4.120-06 7.120-03 4.020-04 Covera 8.622-02 1.962-03 1.402-03 6.53 4.210-03 4.000-03 7.120-03 4.020-04 Covera 8.622-03 1.962-03 1.402-03 6.53 4.210-03 1.120-03 7.2000 7.2000 7.2000 7.2000 7.2000 7.2000 7.2000 7.2000 7.2000 7.2000 7.2000 7.2000 7.2000 7.2000 7.2000 7.2000 7.2000 7.2000 7.2000	a Segregation to Bottom . wound Waste Form	1.146+03	1.205+03	1.60E+02	16.7	3.632+08	3.60540	8 2.7E+05	2.36+06
Accility         1.422-01         1.422-01         1.422-01         1.422-01         1.422-01         1.422-01         1.422-01         1.422-01         1.422-01         1.422-01         1.522-01	cted Disposal	1.102+03	1.406+03	4.005+01	5:1	2.706407	2.40240	74 1.9E+04	1.25405
mm         9.200.00         1.580.00         1.602.00         1.602.00         1.502.00         1.502.00         1.502.00         1.502.00         1.502.00         1.502.00         1.502.00         1.502.00         1.502.00         1.502.00         1.502.00         1.502.00         1.502.00         1.502.00         1.502.00         1.726.00         1	waste Facility	1.08EH03	1.422+03	Z.00E+01	2	1.50E407	1.20640	24 9.6EHO3	3.36+04
Circle         1.00E/01         1.00E/01         4.34         5.00E/06         3.00E/03         3.12E/01         4.22E/03         1.22E/01         4.22E/03         1.22E/01         4.22E/03         1.22E/03         1	regation	9.20E+02	1.582+03	1.405+01	10.1	6.002+06	3.002+0	6 3.8E+03	1.95403
Cited Disposal         Station         Lowerd         Lowerd <thlowerd< th=""> <th< td=""><td>paction</td><td>9.106402</td><td>1.596403</td><td>1.005+01</td><td>71</td><td>6.00E406</td><td>3.00540</td><td>6 3.8E+03</td><td>7.75406</td></th<></thlowerd<>	paction	9.106402	1.596403	1.005+01	71	6.00E406	3.00540	6 3.8E+03	7.75406
Concerts         E.G.E.V.1         T.G.E.V.2         T.G.E.V.2         T.G.E.V.2         T.S.E.V.3         T.S.E.V.3 <th< td=""><td>ontaincrized Disposal</td><td>9.036402</td><td>1.64E403</td><td>1.805+01</td><td>7.80</td><td>3.52E+08</td><td>3.015+01</td><td>8 2.2E+05</td><td>7.96+06</td></th<>	ontaincrized Disposal	9.036402	1.64E403	1.805+01	7.80	3.52E+08	3.015+01	8 2.2E+05	7.96+06
Convers         0.665.402         1.648-601         1.0028         00         7.71         3.648-60         4.008-60         7.184-65         4.187-	reșetan sucțion	8.61E+02	1.64E+03	1.006 00	7.70	3.42E+08	2.91E+0	8C 2.1E+05	7.SE+06
posel 0d 2.508-01 5.208-02 8.591 4.318-08 2.718-06 1.78-05 4.38-05 7.26-05 0.38-05 1.78-05 3.208-02 1500 15008-04 15008-04 7.68-07 3.75-05	roved Covers ent-walled Trench	8.60E+02 5.20C+02	1.64E+03	1.008 00	6.38	3.64E+08 2.10E+08	1.596+0	6 2.1E+05 8 <sup>c</sup> 1.1E+05	4.26+06
04 2.50E401 5.20E4024 1900 19005404 19004404 19005407 3.1544	an Disposal	8	2.506+03	5.206+02	8.591	4.31E+08	2.71E+0	B 1.7E+05	4.36405
	an Disposal Joe	83	2.506+03	5.20E4024	19061	19006108	190061	1.66407	
	eferred to Case 4B.								
ted to Case 4B.	Referred to Case 6.								
ed to Case 43. ed to Case 6.	Given credit for zero-not	strictly true.							
ed to Case 4B. ed to Case 6. credit for reco-mot atrictly true.									

.

Attachment C

•

•

•

. . .

# Tables 1216/1, 1216/2, 1216/3

· · · ·

A-127

•

				<b>4</b> 9 08		Alter	
	T Cost Benefic Ordered on Cost	able 1216/2 iality of Altern Ecnefit to Refer	atives ence Case	Ocean Disposel* Space* • Ocean and space given o	Present Fractice Same Segregation to Bottor Compaction Improved Covers Stacked Disposal Stacked Disposal Nat Naste Facility Doconclainerired Disposal Cement-walled Trench Improved Waste Form	r, bescription	Differential C
Alter Ref 1 4A	Present Practice Some Segregation to Bottom Segregation	Cost Benefit Base, S/man-r	to SA	nueqit tut zeto qo	a 2.502+03 9.102+03 9.342+02 9.342+02 9.342+02 9.002+02 9.002+02 9.622+02 8.622+02 1.142+03 1.142+03	Pop Dote Tan-Ten	bst Beneficiality
4B 4C 4D 4E 6 5	Compaction Improved Covers Stacked Disposal Hot Waste Facility Decontainerized Disposal Comentarial Lod Magnetic	3.82+03 3.82+03 9.62+03 1.92+04 2.12+04 3.22+04	4B preferredslightly lower dose at same cost	5,20E+02 5.20E+02	I.COE403 I.COE403 Increase Increase Increase Increase Increase Increase Increase	A Pop Dose from Prev Case	of Alternates Orde
7C 7D 7B	Compaction Improved Covers Segregation	1.1E+05 2.1E+05 2.1E+05	7D preferredslightly lower dose at same cost	8.59 1900 rue. Both	7776444444 77847555334126 91079555334126	Cost \$10 <sup>8</sup>	red on Coa
7 <b>A</b>	Improved Waste Form	2.2E+05 2.7E+05		2.212+88 19002+08	3.002.46 3.002.46 9.002.46 2.102.46 2.102.46 2.102.40 2.102.40 1.527.60 1.527.60 1.527.60	A Costs from Prev Ca	t Beneficiality

.

٠

Table 1216/3 at Deneficiality to Reference

.

٠

	4.32405 3.72408	2.212+98 19002+08	8.59 1900	5,20E+02 5,20E+02	00	
Referred Referred Referred	8 8 8 9 8 9 8 9 8 9 8 9 8 9 8 9 8 9 8 9	3.002.46 5.0	4.24 4.31 4.34 4.58 5.39 7.80 7.80 7.80	1. 202403 1. 902403 Increase Increase Increase 1. 002401 J. 002401 J. 002401 J. 002402 Increase Increase	2.502+03 9.102+03 9.102+02 9.142+02 1.002+03 1.002+03 9.002+02 8.602:02 8.602:02 8.602:02	a]

Referred to 5 Referred to 5

.

Referred to 48 Referred to 48

A Costs from Prev Case

Cost/Benefit Ratio

· ·

A-128

### Docketed Comment Number: 42

Commenter: Utility Nuclear Waste Management Group

<u>Response(s)</u> <u>Item 1</u> - Contrary to the commenter's allegations, the staff believes that low-level radioactive waste, if not managed and disposed of properly, may indeed jeopardize public health and safety and the environment in addition to posing long-term economic burdens. In developing the proposed rule and the DEIS, the staff explicitly sought to explore a broad range of alternatives to assure that both the Commission's mandate under the Atomic Energy Act and the requirements of the National Environmental Policy Act were met. The staff feels that it achieved these goals in both cases and notes that this commenter was alone among nearly 50 others in suggesting that the DEIS was inadequate as an environmental full-disclosure statement. The staff would like to add that the comments of several state and federal agencies with extensive experience in the preparation of environmental impact statements did not indicate that the DEIS was inadequate in strategy or content.

Item 2 - NRC staff does not believe the DEIS, as written, is "almost incomprehensible." The number of alternatives considered and analyzed in the DEIS is large. NRC staff purposely selected a broad range of alternatives for analysis to ensure completeness in consideration of possible approaches and variations in disposal technologies which could be applied. NRC staff agrees, however, that the number of alternatives can be reduced in number and has presented fewer alternatives in the FEIS. NRC staff also agrees that the use of summary tables will help in their clear presentation and has included additional summary tables in the FEIS.

· · · · · · ·

With respect to including data on specific isotopic contributions to dose, NRC recognizes that actions taken to control one isotope may not be appropriate for another. The majority of wastes, however, contain a mixture of isotopes and not just a single isotope.

<u>Item 3</u> - The EIS for Part 61 is not a generic EIS for LLW disposal. Rather it is a decision document for the technical requirements that should be applied in the disposal of LLW. As such, many issues generally

 related to LLW disposal, such as promotion of new disposal sites, may not have been addressed at all or may not have been addressed in detail.

NRC staff agrees the EIS should provide the public with a reasoned insight into the nature, scope and magnitude of relevant issues. In this regard, the staff does not believe the EIS falls so far short of this goal as to render it inadequate as an environmental full-disclosure statement. Further, NRC staff does not believe that LLW disposal can be dismissed as a "non-problem." The commenter's attempts to consider it as such and to relegate LLW disposal to a place among the most innocuous of society's activities are inappropriate.

The EIS should present information and analyses about LLW disposal in an understandable scientific and technical perspective. NRC has tried to improve the public information aspects and method of presentation used in the EIS. The commenter claims that a "misconception" reflected in the EIS--"the oftassumed dominance of the water pathway"--presents a disservice to the public by the "portrayal of a less than accurate--or, at least, easily misunderstood-picture." NRC staff does not believe it has portrayed a less than accurate or easily misunderstood picture and the commenter, by implying that only a few individual intruders might be exposed is not accurately portraying the full range of exposure pathways accurately. As stated in Section 3.8.2 of the DEIS, the various ways that a person can be exposed to radioactive waste may be divided into three principal categories:

- (1) Activities involving the processing and handling of the waste prior to disposal. This would include activities involved in the handling, processing, and packaging of the waste at its point of generation; transport of the waste from the point of generation to disposal; and activities at the disposal facility involving emplacement of the waste at the disposal facility (processing of waste at facilities other than the generating licensee's facility would also be included).
- (2) Man contacting the waste after disposal (i.e., intrusion into the disposal facility leading to exposure to disposed waste). This would include activities of man that would lead to his intruding into the disposal facility either purposefully (such as an archeologist in the future

intentionally digging into the sites attempting to reclaim artifacts from the disposed waste) or inadvertently (such as an unknowing individual who might attempt to use the land for reasonable productive purposes in the future--e.g., farming or housing).

(3) The waste entering one of several natural environmental pathways back to man. This would include the potential leaching and transport of the waste through the ground water; intrusion and dispersion by plants and animals; long-term erosion of the site with eventual uncovering of the waste and surface water and air transport; and release of gaseous decomposition products from the waste containing radioactive species (e.g., tritiated methane gas).

The first mode involves primarily short-term considerations and the second and third, long-term considerations.

Later in Chapter 5 of the DEIS, NRC again reviews the various pathways of release to the environment and notes that of these, the most significant pathway is ground-water migration. Gaseous releases do not have a large impact and can be reduced by assuring stable site conditions. Impacts from plant and animal intrusion are site-specific and can be reduced through engineering designs applied to reduce ground-water migration and potential intruder exposures. Erosion is a slow, long-term process which can be controlled through proper siting and good operational techniques.

NRC staff continues to believe that the ground-water migration pathway is probably of most significance in relation to other environmental release pathways and has the potential of exposing the largest number of individuals. In addition, contamination of groundwater has been documented in more than one instance from the disposal of hazardous and solid wastes. As such, it is a real potential pathway of exposure which should be considered, not a hypothetical pathway which might occur in the future, such as inadvertent intrusion.

Item 4 - Summary tables illustrating the results of the numerical "cost-benefit" analysis have been included in the final EIS.



• • •

Dear Sir:

(46 FR The Tennessee Valley Authority (TVA) is pleased to provide comments on the proposed amendments to 10 CFR Parts 2, 19, 20, 21, 30, 40, 50, 51, 61, 70, 73, and 170 concerning licensing requirements for land disposal of radioactive waste as noticed in the July 24, 1981 Federal Register notice (46 FR 38081-38105) and comments on NURSG-0782.

We appreciate the opportunity to comment and our specific comments are enclosed.



### ENCLOSURE

·:- .

- 1. Page 38086, last paragraph in column 1-This paragraph would require a licensee (discosal facility operator) to provide financial responsibility for a disposal site even after it has been turned over to a Government agency for long-term institutional care and monitoring. Since the institutional care period could last as long as 100 years, a licensee's money could be tied up for 100 years. We believe it would be preferable to have the licensee turn over the site and any required money at the time that the license is transferred to the Government.
- 2. 10 CFR 61.41, first sentence-We believe the term "general environment" should be defined and suggest the definition should be similar to the same term in 40 CFP. Part 190.
- 10 CFR 61.51(a) (4) -- We suggest changing "prevent" to "minimize" and, in 10 CFR 61.51(a) (6), changing "eliminate" to "minimize." 3.
- 10 CFR 61.51(a) (7)—We believe this item should be deleted since many activities usually take place at a disposal facility other than disposal of radioactive waste. Some facilities are used for transportation equipment storage, empty container storage, and a center for radwaste services to nuclear plants. This regulation could have a significant impact on present disposal sites and conflicts with 10 CFR 61.11(c) (4).
- <u>10 CFR 61.52(a) (4)-(10)</u>—It appears that these regulations apply to Class A, B, and C waste. However, 10 CFR 61.52(a) (2) says that these regulations apply to Class B waste. We believe this should be clarified.
- 6. 10 CFR 61.52(a) (8) We believe the designation of a minimum distance of 100 feet for the buffer zone does not appear to be necessary. We support. the need for a buffer zone but the distance criteria should be determined by site specific and other factors of the disposal facility to meet the performance objectives of Subpart C of this part.
- 7. 10 CFR 61.55, Table 1-The limits for some of the isotopes in this table are the same for all three classes of waste. We succest the regulations should contain statements that any waste containing these isotopes in concentrations exceeding these limits cannot be disposed of at a near-surface disposal facility.
- 8. 10 CFR 61.55(b)(2) and 61.56(b) The stability requirement of at least 150 years for Class B stable waste appears inconsistent with the institutional control period of up to 100 years (see paragraph 61.59(b), page 39097). As discussed in paragraph 61.7(b)(4), page 38091, Class 3 stable waste contains the ". . .types and quantities of radioisotopes that will decay during the 100-year period to levels that do not pose a danger to public health and safety." Therefore, for waste in this classificacation, there does not seem to be a need for stability requirements beyond 100 years.

A-131

6

Also, it is not obvicus why the stability requirement for Class C intruder waste is given as at least 150 years. This waste classification specifies a maximum concentration of radionuclides so that at the end of 500 years, the remaining radioactivity is at a level that does not pose a danger to public health and safety (see paragraph 61.7(b) (5), page 38091). Therefore, for this waste in this classification, the stability requirements should be for 500 years.

.

9. <u>10 CFR 20.311(d) (5)</u>—It appears this regulation requires that a copy of the shipping manifest be sent to the intended recipient (possibly by mail) at the time of shipment. If shipments are sent by truck it is very likely that the truck will get to the disposal facility before the advance copy of the manifest. Therefore, this method would not provide a way of providing prior notification of the intended shipment.

We believe the substance of the information required by this regulation is already required by the Department of Transportation (DOT). We suggest the procedure be revised to require only the submittal of copies of the material sent to DOT. This requirement should be placed in 10 Part CFR 61 or 10 CFR Part 71 and not 10 CFR Part 23.

10. NUREG-0782, Volume 3, Table E.2, Page E-21—We believe the concentration should be in µg/m<sup>2</sup> instead of mg/m<sup>2</sup>. Also the annual average secondary standard of 60 µg/m<sup>2</sup> for sulfur dioxide was revoked in 1973. The current annual average primary sulfur dioxide standard is 80 µg/m<sup>3</sup>. Also, this table includes only four of the seven pollutants for which standards have been developed. We suggest inclusion of the remaining three—carbon monoxide, ozone, and lead.

### Docketed Comment Number: 43

Commenter: Tennessee Valley Authority

<u>Response(s)</u>: <u>Item 1</u> - The staff noted this correction. The reader is referred to Appendix E, Errata in this volume.

1



A-133

Mr. R. Dale Smith February 9, 1982 Pace 3

- In Section 5.2.1 the implication is made that the <u>cotential</u> for airborne releases will be reduced if the number of perimeter air samples is increased. It will in fact only increase the probability of detecting such releases.
- In Section 7.4, Table 7.3 specifies "a stable waste form will maintain its physical dimensions within 5% and its form, under . . . . compressive load of 50 psi . . . .changes." We feel that 50 psi may be overly restrictive in some cases and that the actual pressure should be selected on the basis of the geology and operating parameters for a specific site.

The statement is also made that "waste which exceeds this concentration (Table 7.2) . . . is not generally acceptable. . . .this part." The term "benerally acceptable" is not clear. Perhaps some words could be added to reflect the possibility that a particularly suitable proposed site may be able to accept those wastes, even though most facilities probably would not.

- Even though Table E.4 in Appendix E reflects 1980 annual salaries, for certain of the staff members they seem to be low by about 20% even without considering inflation.
- In Section 5.2.3 of Appendix E no credit is given for a combination of random and selective placement of wastes in the trench at the reference site. This combination is used at existing facilities resulting in a trench volume use efficiency of from 35 to 50%.

Fertaining to the same section, vehicle survey results before and after decontamination are documented, although not usually on the RSR.

- In Section 5.2.5.1 of Appendix E the subject of contamination in wounds is discussed. Standard health physics practice requires that potentially contaminated wounds be surveyed for beta and camma contamination and then receive a separate alpha survey if alpha emitting isotopes are suspected.
- Section 6.3.1 of Appendix E discusses the decontamination and dismantling of site buildings. If a site is operated in compliance with Parts 20 and 61 of ICCFR and the facility license conditions, and if it utilizes accepted health physics practices, the efforts and materials required for decontamination should be minimal. In addition, site buildings could be used for custodial purposes or leased for business purposes. These uses would reduce the projected costs.
- In Section 2.3.3.3 of Appendix F decontainerized disposal is discussed as an option. Such disposal at existing facilities has been limited to bulk wastes with very low activity. The disposal of large volumes of such wastes may be more appropriately conducted at a facility other than shallow land burial if their activity levels are low enough.

Mr. R. Dale Smith February 9, 1982 Page 4

The NRC estimates 1-2 years for site screening and 1-2 years for their review of the license application package. It is critical to the availability of low-level waste disposal capacity in this country that the review-period be expedited as much as possible. Considering the 1985 date for excluding out-of-region waste from regional facilities (Low-Level Waste Policy Act, 1980) the 4-5 year site development process faces extreme scheduling problems. We unge the Commission to consider ways in which the review process could be expedited and licenses issued as quickly as possible. As an example, the proposed procedure for tendering a notice of intent to apply for a license should specify a commitment to respond (e.g., 60 days after submittal) to avoid unnecessary delays.

23

Q

ኤ

- Although the discussion of funding for institutional control (Appendix Q) does present a low cost scenario, it is still probably considerably higher than would be the case if improved operating and closure procedures were employed during the life of the site. The discussion does mention that the costs for the low cost scenario may be conservatively high. We would recommend increased emphasis on this qualifying statement.
- Staff operating costs in Appendix Q are based on 70 employees. We understand that the reference facility represents a site that can meet minimum regulatory requirements, but feel that an estimate of 100 people would more accurately represent operating needs for a site receiving 1.7 million cubic feet of waste per year. Environmental monitoring costs also appear to be low. Site monitoring, the offsite environmental program, and personnel monitoring would total about \$300,000 per year if sent out for third party contractual analysis.
- Appendix Q readily admits than an investment in a disposal site is a
  risky proposition for an investor. In order to compensate the investor
  for this high risk, a high return on the financial investment is required. This return should not be based on a "cost plus" philosophy
  as in the NRC study. Instead, it should be based on the time value of
  money (TVH). The TVM concept implies that a dollar invested in the first
  year of a 20-year period is not equal in value to a dollar or profit
  returned in year 20. The investor must demand substantially more than
  a dollar in return because of the economic, business, and financial risks
  that are incurred each year.

The disposal price for receiving wastes at a new facility must be set at a rate which compensates the investor for the risks involved. This price should be such that the net present value of the cash flow equals zero at the risk-adjusted rate of return.

 Section 3.4.3.2 discusses the projected growth in non-fuel cycle waste based on existing data. The increasing costs of disposal and transportation to distant sites have provided the incentives to find ways to generate less waste, to reduce the volumes that are generated, and to make maximum use of NRC's new rule on scintiliation fluids and animal carcasses. It is not clear whether this trend has been factored into the waste projections.

(15)

ſΖι

(2z

Mr. R. Dale Smith February 9, 1982 Page 5

- In Section 3.6.5 the reference facility costs include 200 acres of land which is calculated to be adequate for the site lifetime. This assumes that essentially all of the site can be used for disposal, a situation which we expect is rare in much of the country. Chem-Nuclear would suggest the purchase of 400-500 acres of land to provide for unusable areas (e.g., due to topography) to allow wider buffer zones, and to give flexibility in the event the waste projections are underestimated. This section also assumes \$1200/acre, a price which is probably low for many sections of the country.
- Section 4.3.5.2 examines the alternative of thicker disposal cell covers by adding 3 meters of over burden. Consideration should be given to the effect of this on the water table. Experience at Oak Ridge National water table of the sector provides a participation results the water
- Laboratory found that raising the surface elevation caused the water table to also rise to follow the topography. This practice could cause the groundwater pathway to become a greater concern.
- In Section 5.2.4.8 what is the difference between the assumptions for Case 10A and Case 1087 "They appear to have the same waste spectra and designs.

We believe that these comments will assist you in the preparation of the final EIS. Please contact me if you have questions, or would like additional clarification.

CHEM-NUCLEAR SYSTEMS, INC. CHEM-NUCLEAR SYSTEMS, INC. Jolu A. Toype Lesile K. Poppe Corporate Health Physicist

LKP:slj

### Docketed Comment Number: 44

Commenter: Chem-Nuclear Systems, Inc.

<u>Response(s)</u>: <u>Item 1</u> - The staff believes that the emphasis placed on past problems at closed waste disposal sites in the DEIS is both appropriate and necessary to establish a historical framework of the need for the proposed action and to identify the types of problems which have occurred and could occur again without proper safeguards. The staff does not believe the emphasis to be undue, nor does the staff believe it has resulted in overly conservative requirements. The performance objectives and technical requirements were developed through independent analyses which considered both benefits and costs in arriving at a reasonable regulation.

Item 2 - The staff has assumed a basic level of knowledge with respect to the structure of matter, radioactive decay and other related subjects on the reader's behalf in preparing this document. In Chapter 2.0 of the FEIS the lay reader is referred to appropriate works which provide a discussion of these subjects adequate for the purposes of understanding the FEIS.

14

. . .

11

23

<u>Item 3</u> - Section 1.2.2 discusses ocean disposal in its historical context, but does not evaluate its acceptability as an alternative disposal method. In Section 2.3, ocean disposal is briefly discussed as an alternative to land disposal and it is noted that jurisdiction over ocean disposal rests not with NRC, but rather with the U.S. Environmental Protection Ágency. The benefits and liabilities of this disposal method are, therefore, not appropriate subjects for further discussion in this FEIS.

<u>Item 4</u> - The staff agrees with the commenter's observation and has incorporated this suggestion in the FEIS.

<u>Item 5</u> - These aspects were generally addressed in the DEIS regarding handling and worker safety. NRC staff believes that wastes containing such materials should be excluded from disposal facilities due to concern for general handling safety.

Æ

ମ୍ଦ

60

<u>Item 6</u> - Presently, DOE disposal facilities for the disposal of LLW are exempt from NRC and Agreement State regulatory jurisdiction. NRC recognizes the comment and notes that legislation would be required.

Item 7 - The staff recognizes the difficulty which the commenter is alluding to. This comment, however, does not affect the analysis of the EIS.

Item 8 - As the commenter states, a linear relationship has been assumed between the electrical generation capacity of nuclear power plants and the volume of waste generated by the plants. However, NRC staff does not believe that this assumption leads to nonconservative results. For one thing, data which was principally used to estimate volumes of waste generated by nuclear power facilities was based upon a survey of a number of different power plants of varying ages (Reference 18). Thus, the effects of the tendency for older plants to generate larger volumes of waste was already considered to a certain extent.

More importantly, perhaps, the electrical generating capacity projected in the draft EIS to the year 2000 was estimated in a very conservative manner. In NRC's estimates of the projected capacity by the year 2000, two scenarios were considered: a "low" scenario and a "high" scenario (Reference 3). The "low" scenario assures that construction continues on power reactors which are already under construction but that any additional construction of power reactors essentially ceases until at least the late 1980's. The "high" scenario assumes that all facilities either planned or deferred indefinitely as of December 31, 1979 are constructed. These two scenarios resulted in a projected capacity by the year 2000 of 146,000 Mw(e) and 169,000 Mw(e), respectively. The high scenario was assumed for the draft EIS, although it was recognized at the time that it was very likely an overestimate of actual capacity by the year 2000.

This viewpoint has been confirmed by a number of more recent events. Since the development of the high scenario, a number of nuclear power plants which were either planned or under construction have been cancelled, deferred, or delayed. A partial list includes, for example:

- o North Anna-4 (901 Hw(e) PWR): cancelled
- o Allens Creek (1150 Mw(e) BWR): delayed
- o WNP-4 (1218 Hw(e) PWR): cancelled
- o WNP-5 (1242 Mw(e) PWR): cancelled
- o Hartsville-B1 (1233 Mw(e) BWR): construction deferred
- o Hartsville-B2 (1233 Mw(e) BWR): construction deferred
- o Yellow Creek-2 (1285 Mw(e) PWR): construction deferred
- o Bailley (644 Mw(e) BWR): cancelled
- o Callaway-2 (1120 Mw(e) PWR): cancelled
- o Sharon Harris-3 (900 Mw(e) PWR): cancelled
- o Sharon Harris-4 (900 Mw(e) PWR): cancelled
- o Hope Creek-2 (1067 Mw(e) BWR): cancelled
- o Pilgrim-2 (1150 Mw(e) PWR); originally listed as "deferred," now cancelled.

.

In addition, the schedules for completing many of the other power plants under construction has slipped.

Thus, NRC expects that the use of the high scenario will give a very conservative estimate of the electrical generating capacity, and thus give a very conservative estimate of the volume of waste likely to be generated by nuclear power stations.

Item 9 - Urea-formaldehyde (UF) has been used as a solidification agent for liquid radioactive wastes. In 1979 the State of Nevada prohibited the acceptance of urea-formaldehyde at the Beatty facility. In 1981 both the States of South Carolina and Washington also prohibited its acceptance. Urea-formaldehyde was prohibited at the three commercial disposals due to its inability to consistently produce a solidified product which would meet the disposal site free-standing liquid requirements.

A statement regarding the prohibition of UF has been added to the FEIS to address this item.

<u>Item 10</u> - The transfer of title is a legal issue that is independent of responsibilities assigned in the proposed rule. The staff was not trying to set legal precedent. Other forums exist for addressing title
such as contracts or leases between disposal facility operators and the landowner.

Item 11 - As stated by the commenter, exposures to transportation workers are typically a small fraction of those received by workers involved in disposal operations. One of the main reasons for this is that transportation workers generally do not perform unloading and disposal activities at disposal sites. However, the transportation impacts calculated for the draft EIS include the following components: loading operations at the waste generator's facility, transport to the disposal site, and time spent at the disposal site during unloading. Of these three, loading operations at the waste generator's facility contribute the largest exposures.

. . . .

Item 12 - NRC did not intend to imply that records would only last for 100 years: NRC has assumed that active institutional controls can only be relied on for 100 years. They may last much longer. NRC has also assumed that other passive institutional controls such as government land ownership; records and deed restrictions would last for a much longer period of time than 100 years. NRC believes this is adequately addressed in Section 4.3.6 of the DEIS:

The first the second state of the second state

Item 13 - The question raised with respect to movement away from the mounded crest is dependent largely on the degree of slope and soil erosion. The observation is valid but would have to be considered on a site-specific basis. Part 61 does not require the use of moisture or infiltration barriers. Such barriers were considered as an alternative but were rejected because of concern for their long-term viability.

<u>Item 14</u> - The commenter's observation is accurate and has been reflected in Appendix E, Errata, of this volume.

Item 15 - The requirement that a stable waste form maintain its physical dimensions within five percent under a 50 psi compressive load was intended to ensure that degradation of waste containers and waste forms would not lead to severe trench subsidence. The five percent value was selected as a conservative deformation level which would be expected to produce only minor

subsidence effects. These minor subsidence effects could be easily resolved by an inexpensive post-closure maintenance program at the burial site.

The NRC staff believes that trench subsidence needs to be carefully controlled in order to minimize water infiltration and major remedial care operations. In order to achieve this objective in a pragmatic way, the NRC staff has chosen to delete from the rule the prescriptive requirement for a five percent dimensional limit. The requirement that wastes remain structurally stable, however, remains in place. In order to provide guidance to waste generators on acceptable waste forms the NRC staff is preparing a Branch Technical Position (BTP) which defines acceptable test methods and criteria oriented to ensuring structural stability. The 50 psi limit for solidified waste forms will remain in the BTP as a value that would assure compliance but not a requirement.

In addition to the BTP, 10 CFR 61 Section 61.52(a)(4) and 61.52(a)(5) requires that wastes be emplaced to maintain package integrity and that void spaces be filled to reduce subsidence. The filling of void spaces will minimize the creep effects in asphalt or polymeric solidified products since lateral stability will be provided by the fill material. If there are no voids, there will be no space for viscoeleastic wastes to flow into.

Ľ,

.

<u>Item 16</u> - In the final EIS, NRC has emphasized the concept set out in 61.58 that wastes exceeding the Class C limits may be acceptable given special treatment or disposal at greater depths or with special facility designs.

<u>Item 17</u> - Based upon input from the commenter, the staff members for which unusually low salaries were originally estimated were identified. Appropriately higher salaries were assigned to these staff members as listed below:

· · · ·

Occupation	Salary originally assumed	Modified salary
QA technician	\$14,000	\$25,000
Radiation safety technician	\$15,000	\$25,000
Security personnel	\$12,000	\$18,000
Waste shipment scheduler	\$16,000	\$20,000
QA and safety supervisor	\$26,000	\$30,000
Site manager	\$40,000	\$45,000

These revised salaries were used in the calculations performed for the final EIS. Use of the modified salaries has not altered the overall conclusions reached in the draft EIS but have been included in the interest of completeness and accuracy.

Item 18 - As the commenter states, at existing disposal facilities a combination of stacked and random disposal is generally used. Typically, waste packages having rectangular dimensions are stacked in place (e.g., wooden boxes, steel bins) to conserve trench space, while low activity drummed waste is generally disposed in a random manner to reduce radiation exposures to site personnel. Higher activity waste in drums and large cylindrical waste containers such as resin liners are generally stacked or at least emplaced carefully to reduce the potential hazard to site personnel. This results in a trench use efficiency which varies depending upon the particular mix of waste received during the filling of a particular trench. A good rule of thumb with this waste emplacement technique is a volume use efficiency of about 50%.

NRC did not consider relatively small variations in trench use efficiency in the draft EIS. This was believed to be in keeping with the scope of the EIS which is to help determine overall requirements for waste disposal in the Part 61 regulation. For the purposes of the EIS, then, two general alternatives for waste emplacement were considered. One, termed "random" disposal, assumed the existing practice by which some waste is stacked and some is disposed randomly. A volume use efficiency of 50% was assumed for this alternative, consistent with current experience. The second alternative, termed "stacked" disposal, assumed neat stacking of all waste containers. A volume use efficiency of 75% was assumed for this alternative, coupled with a higher level of personnel exposures at the site.

The commenter's point regarding a factual description of waste disposal operations at a typical facility may be addressed in Appendix E by deleting the first two sentences in the first paragraph of Section 5.2.3 (p. E-47) and replacing these two sentences with a new paragraph consisting of the following:

Waste is emplaced in the trench, and backfilled with dirt removed during trench excavation. Typically, waste packages are emplaced with the aid of construction equipment such as cranes and forklifts, using a combination of stacked and random disposal. Waste packages such as wooden boxes or steel bins having rectangular dimensions are generally stacked in place while low activity drummed waste is generally emplaced in a more random manner. Special care is taken during emplacement of higher activity waste such as high activity ion exchange liners to ensure operational safety. This combination of random and stacked disposal is termed "random disposal" in this EIS (to distinguish it from a placement alternative of fully stacked disposal: see Appendix F) and results in a trench volume efficiency of about 50%.

<u>Item 19</u> - Disposal sites may have different recording practices. Some record the vehicle survey results on the shipment manifest (e.g., the radioactive shipment record, or RSR) and some record the survey results on a separate form. This point may be generalized in Appendix E of the DEIS by modifying the last sentence in Section 5.2.3 (p. E-98) so that it reads as follows: The results of the survey are recorded.

Item 20 - This point may be clarified in Appendix E of the DEIS by modifying the next to last sentence in the next to last paragraph (p. E-50) of Section 5.2.5.1. as follows: "If through a site accident, a worker may receive an open wound and the wound is suspected of having become contaminated, a radiation survey is also performed. The survey is performed for beta and gamma contamination, and also for alpha contamination if alpha emitting isotopes are suspected." Item 21 - The consideration of cost of building demolition was estimated for the draft EIS based upon the cost for building demolition estimated by Chem-Nuclear Systems, Inc. as part of their preliminary site closure and stabilization plan for the Barnwell disposal site (Reference 13). In this closure and stabilization plan, an estimate of \$525,000 was made for decontamination and demolition of most of the onsite buildings (some would be left standing for use by the site owner following license termination), conservatively assuming that a private contractor was hired to perform these services. Recognizing that CNSI carries out a number of additional activities at the Barnwell facility besides waste disposal (e.g., truck, transporter and cask refurbishment, mobile solidification units, development of solidification agents), a building demolition cost of \$300,000 was estimated for the reference disposal facility.

· .

It should be recognized that actual closure costs such as building decontamination and demolition costs would be specific for a particular site and should be evaluated specifically for that site. Whether or not the assumed decontamination and demolition costs for the reference disposal facility are overconservative does not change the essential conclusions reached in the draft EIS regarding the need to consider and plan for facility closure prior to site licensing and the need to assure that funds will be available to carry out closure activities. In the interest of completeness, however, these costs have been reduced to \$200,000 for the final EIS. This change does not alter the conclusions reached in the EIS but is reflected in the cost/benefit analyses and estimates of unmitigated impacts.

<u>Item 22</u> - NRC staff agrees with this comment and believes the disposal of such wastes can be individually evaluated for disposal on site at the point of generation or at a central facility such as a sanitary landfill.

,

Item 23 - The staff is in fact considering methods of expediting the review process and certain changes to the licensing procedures of the rule have been made to achieve this end. While no new LLW disposal facility has been recently licensed, NRC staff believe that if an applicant takes action in the pre-application phase to involve the local community, the public and the state (and/or compact), prepares a detailed and complete application, and

 promptly responds to additional information needs, it may be possible to license such a facility in a time frame of about 15 months.

Item 24 - In Appendix Q and for purposes of the EIS analysis, NRC staff assumed three cost scenarios corresponding to three levels of long-term care: a low cost level, a moderate cost level, and a high cost level. These levels were used to help assess the potential long term costs of various waste management alternatives. As the commentor states, NRC staff believes that the estimated costs corresponding to the low level of long-term care may be conservatively high. Perhaps one of the reasons for this is that most of the experience with waste disposal has been with high levels of longterm costs resulting from site stability and subsidence problems. Less experience has been gained with potential long-term costs associated with stable site conditions.

Item 25 - The reference site staffing (70 employees) was not intended in all cases to represent a disposal facility which can meet minimum regulatory requirements. One of the purposes of the draft EIS was to examine the long-term consequences of unstable waste and disposal unit conditions as well as analyze the effects of many of the improvements that have been implemented at disposal sites over the past several years. As the commenter has stated earlier in this comments, the analysis in Chapter 5 of the DEIS examines some past practices and the reference case in Chapter 5 corresponds more to operating practices of several years ago then to currently operating practices. As discussed above, this was deliberate.

. . .

12

::4

28

~1

In any case, the assumed personnel level of 70 employees at the reference disposal facility is meant to represent the minimum staff needed to comply with the existing safety requirements of 10 CFR Part 20. The minimum personnel level projected to meet the disposal facility stability and long-term environmental impact requirements proposed in 10 CFR Part 61, however, would be larger. NRC's analysis considers the additional personnel requirements associated with typical site operations which would be required to improve site stability and reduce groundwater releases. Many of these typical site operations are being carried out today.

÷ +

.

Another point is that the commenter's site operations include several activities which are not directly related to waste disposal. They include, for example, delivery vehicle refurbishment and routing, mobile solidification operations, and development of new solidification agents. NRC staff is not sure if this has been fully recognized by the commenter.

In any case, the basis for the comment was requested by NRC staff and was provided by the commentor. Based upon discussions with the commenter, the base number of site employees has been raised to 80. The job functions, salaries, and impact on total operating costs are addressed in Appendix D of this volume.

Environmental monitoring costs have been reevaluated and a revised level included in the analysis.

<u>Item 26</u> - The cost analysis has been modified to consider the time value of money concept. This has been done for the sake of completeness and accuracy. The overall conclusions reached for the Part 61 rule, however, have not been altered.

Item 27 - In the analysis, NRC first considered a "base case" projection of wastes generated from both fuel cycle and non-fuel cycle sources. In this base case, termed "waste spectrum 1," waste volumes generated by fuel cycle sources were linked to electrical generating capacity, while waste volumes generated by non-fuel cycle sources were generally based upon a linear projection from a number of existing data sources. Subsequent waste spectra considered the effects on waste volumes, radionuclide concentrations, waste stability, etc. from various waste processing alternatives. These alternatives included use of different waste solidification media, use of improved liquid waste evaporation techniques, compaction of compressible waste, incineration of compressible waste and ion exchange media, and use of high integrity containers. Thus, the effects of volume reduction were considered in the analysis.

NRC also briefly examined the possible effects of the recent modifications to 10 CFR Part 20 to allow disposal by less restrictive means of less than .05 µCi/gm of tritium and carbon-14 contained in liquid scintillation fluids and animal carcasses. The modification was determined to not alter NRC's conclusions regarding the draft Part 61 rule.

It is true that increased disposal costs have encouraged many waste generators to reduce waste volumes through waste minimization techniques as well as volume reduction techniques such as compaction. One way to account for this would be to assume a curved rather than a linear waste volume generation rate. That is, as time passes, the rate of volume increase would lessen. NRC did not take this approach, however, as the increased complexity did not seem to be warranted by any potential improvements in the analytical results. Unlike fuel cycle wastes, non-fuel cycle wastes are not linked to electrical generating capacity. Compared to fuel cycle sources, non-fuel cycle sources are much more numerous and the levels of use of volume minimization and reduction techniques much more difficult to assess. In addition, much of the waste is generated by small entities such as hospitals, universities, research facilities, and radioisotope manufacturers. Uses of radioisotopes may change, new manufacturing facilities may be built, etc.

For purposes of determining requirements for waste disposal in the Part 61 rule, then, it was believed to be sufficient to assume a linear growth relationship for fuel cycle waste. This approach is believed to be reasonable considering the level of data and at the same time conservative with respect to impacts from waste disposal.

Item 28 - Although the reference facility costs did include purchase of 200 acres of land, only about 140 acres was actually assumed to be used for the disposal. Thus, some consideration was given in the draft EIS to the possibility of unstable areas and to flexibility. Also, the precise amount of land purchased and the price paid for the acreage are not significant as far as the final conclusions reached.

For completeness, however, the assumed acreage purchased for the reference facility has been raised to 400 acres and at an assumed price of \$1400/acre. As the commenter states, the larger acreage provides for wider buffer zones than assumed for the reference facility and gives some additional flexibility in operations.

• • •

Item 29 - The comment is well-taken. Areas where significant thicknesses of fill will be placed will be evaluated on a site-specific basis for water table rises. Native cohesionless soils will not typically experience the response observed at Oak Ridge. In addition, a capillary barrier can be installed under the thick fill to prevent a rise in the water table.

Item 30 - The commenter's observation is accurate. The text in Section 5.2.4.8 contained a typographical error which indicated that Waste Spectrum 2 should be assumed for Case 10A: The text should have stated that Case 10A assumed Waste Spectrum 1. This has been noted in Appendix E, Errata, in this volume.

۰.

۰.

A-141



Docketed Comment Number: 45

<u>Commenter</u>: Atomic Energy of Canada, Limited.

<u>Response(s)</u>: This comment was inadvertently docketed for both the rule and the EIS, although the comment addressed only the rule. The commenter's concerns were reviewed and responded to in the development of the final rule.

.



A-143

			ANNING DIVISION
		(STA	TE CLEARINGHOUSE)
	·		MIS-4 Review and Comment
• .		DATE: 11-23-81	
<b>TO:</b>	Martin Castillo		
	Environmental Improvem	ant Division	
FROM:	Betsy Reed, Planning B	Tread	
RE:	<u>№ 82 11 17-015</u> Dra SAI NUMBER PRO	ft Environmental Impact Statement - I UECTTITLE for Land Disposal of Ra	icensing Requirements dioactive Waste
	Planning Division	· · · · · · · · · · · · · · · · · · ·	
	LEAD AGENCY		
Please revi	iew and comment on the above a	pplication and return to the sender by12-	-31-81
1. Does th 1	us plan duplicate any programs Yes (If yes, please identify these No	w <u>nich nava simuar goals and objectives to the pr</u> programs.)	oporeu airpin-i dont
2. Does th	he proposed application conform	with a comprehensive plan developed for the are	a in which it is lorated?
	Not applicable	· · · ·	
}	Yes No life o plana avalais is what	way it is not compatible )	
••	vo (11 no, piezze explain in white	way to is not compactively	
	, Part . S.		
3. Does th	be proposed application conflict Yes (If yes, please cite the confli No	with any applicable statute, order, rule, or regula ting statute, order, rule or regulation.)	tioh (federal, state or local)?
			tion
4. Descri	pe any suggestions or means of i	mbioning of strengtoening the biologed abbuca	
	•		
			•
×	lemant on this		
No	property of a supported.	project	
Pro	posal is supported with recomm	endations.	
Pro	posal is not supported.		
FW	rther information needed, review	anabended and abbucant pormer of redream	
	ale of many available the set for the set	d my personal and/or personmendations shows	
	Sis of my review. Loave indicate	a my response and/or recommendations above.	
	2. t. I Lasts	ENVIRONMENTAL REVIEW COO	RDINATOR
Signature	of Reviewer	Title	
DECEMB	ER 4. 1981	ENVIRONMENTAL IMPROVEMENT	T_DIVISION
Date		Agency	1 . mbies . es san lies-e
Approved	i July, 1979 - DFA	· · · · ·	1 - yellow - SPD com
UNLIEVELY			2 + pink 1 + lead agency

									DMB Approval I	Ne. 25-R0218	
T	FEDI	ERAL	ASSISTANC	٤	2. Applicant's	s. Numbe	1. State opplication identifier	e Num	w M 82 11 1	7-015	
Ì	1. Type Of Action	Press	plication			b. Date 19 Year Month Day		b. Date Annie	Yee med 19 8	r menth day 1 11 17	
1	(Mark appropriate box)	ert Disprintentitien Of Intent (Opt.)		Leave Blank		•	<b>4</b>	·- ·			
Ì	4. Logal Ap	plicent/l	locipient				S. Federal I	Employee	Identification N	• ]	
	a. Applican	t Name	US Nuc	lear Re	gulatory Commission		6. Program				
	E. Street/P.	,0, <b>8</b> 05	• • •				(From a. Num		<u>••• 17 7 •]dd</u> o		
	d. City f. State	City 1 Washington			<ol> <li>Coun</li> <li>Zip C</li> </ol>	ty 1 Jode 1	Catalog Nuc		lear Regulatory	ALOTY	
;	h. Contact	Contact Person t D. C. (Name & Iglephone no.)				20555		Count	ission.		
Ŧ	7. Title and	l dosoripa	ion of applicant's	project			8. Type of	appliant/	resignant		
/ Recto	DELS of R	Lice	nsing Requ ctive Vast	irement e	a for La	nd Disposal	La franzen Durana in furder familierten Agener Galeren Durana in furder familierten Galeren durana in finder			•	
Cont.	Acti	on be	ing consid	ered in	this ZI	S is the issura	tssursuspecty: Federal				
Ĩ	OT A	10	regulation	, part	61, to t	he Us-NRC rules	S. Type of	-	LAIG COTOL	rate atter 🗠	
1 mail +	ing	PIOCE	dures, per	formance Waste	a object	ives of "low-	A-Basie Gri B-Suppier C-Look	A-Basis Grant D-Insurance B-Supplemented Grant E-Other C-Losh Enter appropriate letter(d)			
ă	10. Area of	project i	mpest (Nemes of	CUINES, COUR	int, said, etc.	11. Estimated number of persons	12. Type a	f applicati C Revi	pliation Revision E-Australiation		
	Stat	evide				beneficing	8-Renewal D-Co		Enter appropriate letter		
	13, Propose	d Fundin	1	14. Congr	minoral Districts Of: 15. Type of shange		For 12c or 12e F=Other Specify:				
	b. Automat	•	.00. 00.	01 -	- 02	01 - 02	S-Decrease Dollars C-Ingrease Duration				
	c. State .00 15. Project d. Local .00 to to		Surt 2 OI <sup>ath</sup> ÓI	17. Project Duration	E-Concellation Ever appro-			AU			
	e. Other		<u>00</u>	18. Estores Lo be s	ted date ubmitted	Year month date	19. Emistin	g ladaral i	dentification m		
	20. Federal	a agency 1	Josepon evideon ca	(Name, ci	ty, state, sip o	ode)	1		21. Annertis a	4444	
-	Nucl	ear R	egulatory	Counis:	sion						
atter	The	belief, c	lats in this pres	oplication/	pursuent all respo	to instructions therein, to ness are attached:		clearingh	pure and reg	ionae stieched	
Cartifi	Cartifies That	by the p cant and with the	poverning body o the applicant w attached essure	f the apple- nil comply mass if the	(1) (2)	;					
÷	22.	a. Type	d name and title			b. Signature			c. Dete segned		
actie	Cartifying represen-			•				Year month day			
-	24. Agency	200.000	· · · ·			t			2 <u>6.</u> Y	ur month der	
	28. Organis	Internet L	int -		•	27. Administrative effi			received 19 28. Federal se		
c Hon	28. A44rm	-				I			30. Federal grant		
~						1	- Yana i		identifugt	ien .	
A a	31. Action taken		a. Fedéral	\$	.00	31. Action data 1	inn month day		Starting 19		
- Total	Do. Rejet	ctind	b. Applicant		.00	35. Contact for addition	nel information e memberi		34. Y Ending	ar math day	
1	C. Returned for <u>c. State</u> amendment     d. Local     Deterred     S. Other			<u>00.</u> 00.				arto 19 37. Romarka (	ubind		
1			.00	1							
Construction of the second section, as the second section, as the second section, as the second section, as the second section of the secti			.00 ny comments r agency respo 25, it has been	00 1 manus received from clearing- v response is due under provisions (Name and teld k bann or is being might,			-96 Otficial ione numbers				
-	474.101		<u> </u>		•		<u> </u>		readend Ray - 4	74 bar 1 (14 m	
	-24-101		ALSO PLEAS	E COUPI	LETE REVE	RSE SIDE)	Inscribed b	y GSA, F	unders Form 4 Tederal Manager	24 rogs 1 (10-71) unt Circular 74-7	
		•	•		,						

·...

.

•

Docketed Comment Number: 46

4 +

•

.

.

<u>Commenter</u>: State of New Mexico, Department of Finance and Administration, State Planning Division

.

.

<u>Response(s)</u>: This comment was docketed and reviewed by the staff. No items were found in the comment which require a response.

. . . . .

. .

.

. .

A-145

κ.

..

.



.

· • • • •

# 46 FR 51776



TEXAS ENERGY AND NATURAL RESOURCES ADVISORY COUNCIL

February 12, 1982

Mr. Dale Smith Division of Waste Management U.S. Nuclear Regulatory Commission Washington, D. C. 20555



Dear Mr. Smith:

The Texas Energy and Natural Resources staff has reviewed the Draft Environmental Impact Statement for the "Licensing Requirements for Land Disposal of Radioactive Waste" and has no comment to make. This agency, however, appreciates the opportunity to review and comment on this report. It is obvious that a great deal of time and effort were necessary to produce it.

If I can assist you in any other way, please feel free to call.

Sincerely, Executive Assistant

Enclosure

ES:iml

Docketed Comment Number: 47

: •

Commenter: Texas Energy and Natural Resources Advisory Council

<u>Response(s)</u>: This comment was docketed and reviewed by the staff. No items were found in the comment which require a response.

. .

Co-Cheirmen: Stilliam P. Clamenca, Jr., William P. Hobby Governor Lieutenans Governor Vice-Chavman: Bill Clayton Specker of the House

Executive Director: Milton L. Nolloway

A-146

PR-61 46 FR 51776

TEXAS WATER COMMISSION

Dorsey B. Hardeman

Lee B. M. Bi

DOCIDET WELEER PR

Fair McDonald, Charman

TENAS DEPARTMENT OF WATER RESOURCES 1700 N. Congress Avenue Austin, Texas

TEXAS WATER DEVELOPMENT BOARD Louis A. Beecherl, Ir., Chairman John H. Garrett, Vice Chairman George W. McCleskey Glen E. Roney W. O. Bankston Lonnie A. "Bo" Pilgrim

Harvey Davis Executive Director

February 23, 1982

Mr. R. Dale Smith. Chief Low-Level Waste Licensing Branch Division of Waste Management Office of Nuclear Material Safety and Safeguards U.S. Nuclear Regulatory Commission Washington, D.C. 20555

Dear Mr. Smith:

Re: U.S. Nuclear Regulatory Commission (NRC) ... Draft Environmental Impact Statement (DEIS) on Proposed 10 CFR Part 61, Relative to Federal Licensing Requirements for the Land Disposal by Shallow Land Burial and Alternative Disposal Methods of Low-Level Radioactive Waste. (NUREG-0782, Volumes 1-4), September 1981. and the second ه در در د

Fursuant to instructions received by this Department on January 18, 1982, from the Governor's Budget and Planning Office, we have reviewed the referenced document from the standpoint of our statutory statewide responsibilities relative to water resources planning, development, conservation, and management, under the Texas Water Code and the Texas Administrative Code. We hereby submit directly our review comments, as requested. Therefore, we have directed our staff review to the following portions of the proposed regulations, 10 CFR Part 61 (46 FR July 24, 1981, pp. 38081-38105), insofar as they relate to water resource matters:

Supplementary Information Section V.C. (Summary of Rule--Minimum Technical Requirements): 1 .....

Disposal Site Suitability Requirements. Disposal Site Design. Land Disposal Facility Operation. Disposal Site Closure Requirements.

Mr. R. Dale Smith. Chief Page 2 February 23, 1982

#### Subpart D (Technical Requirements for Disposal Facilities):

Section 61.13 (Technical Analyses).

Section 61.50 (Disposal Site Suitability Requirements for Land Disposal).

Section 61.51 (Disposal Site Design for Land Disposal).

Section 61.52 (Land Disposal Facility Operation and Disposal Site Closure).

We offer the following staff review comments:

- 1. The above-listed portions of the proposed regulations provide adequate criteria, guidelines, and policies for achieving a comprehensive evaluation of the potential impacts of proposed low-level radioactive waste disposal facilities on local water sources, to ensure the selection of sites that will afford the effective exclusion, isolation, and containment of the waste from aqueous pathways to the biosphere.
- 2. We feel that insofar as water quality protection considerations are concerned, the proposed regulations will, together with present and anticipated new implementing State regulations to be promulgated by the Texas Department of Health, and/or by the Texas Low-Level Radiation Waste Disposal Authority (creation authorized in S.B. 1177 of the 67th Legislature of the State of Texas, effective June 1, 1981), will provide the required consistent Federal and State criteria, guidance, and standards for the timely, comprehensive consideration of State water resource protection. We understand that the above-mentioned two State agencies have the primary responsibility for implementing the proposed 10 CFR Part 61, for the State of Texas, as an "Agreement State" under the provisions of 10 CFR Part 150. Also, we understand that this Department will be required to participate in the governmental interagency review of disposal facility site selection determinations, and facility project plans and specifications, to the extent of certifying the potential water quality impacts thereof. We offer the following suggested additions to the technical water-related guidelines in the regulations:
  - a. Guidance on the interim or temporary emergency storage of the wastes, as distinct from the permanent disposal of waste, in the event of a temporary, unavoidable denial of the permanent disposal facility due to equipment malfunctions or other unforeseen events.

. .

1946 (2016) 13 - M. C. M. B. C. M. ALAMINET, LA 1771 (1917) 14 (2016) 212 212 212 212

•. . .. .

Mr. R. Dale Smith, Chief Page 3 February 23, 1982

- b. Policy guidance on determinations which must be made regarding the best allocation of activities which can be sanctioned in the vicinity of any selected temporary storage or final waste disposal facility site, from the standpoint of public health, safety, and welfare.
- c. Guidance regarding the identification and plotted map locations of springs, sinkholes, swamps, mines, pits, quarries, deep waste-injection wells, water wells, test borings, excavations, monitoring wells, pipelines, and other underground facilities.
- 3. We feel that the referenced DEIS adequately fulfills the administrative, analytical, and coordinative requirements of Section 102(2)(C) of the National Environmental Policy Act of 1969, and the implementing regulations, 40 CFR Farts 1500-1508.

We appreciated the opportunity to review and comment on the referenced document and the related proposed 10 CFR Part 61. Please advise if we can be of further assistance.

Sincerely yours,

Harvey Davis Executive Director

### Docketed Comment Number: 48

Commenter: Texas Department of Water Resources

<u>Response(s)</u>: This comment was docketed and reviewed by the staff. No items were found in the comment which require a response.

2.00

#### VIRGINIA ELECTRIC AND POWER COMPANY RICHMOND, VIEGINIA 23261



Very tryly/yours, • · · · · · · · · · · · . **.** · and the state of the ·. : R. H. Leasburg : 1 ۰. 100 ۰. Sec. Sec. 

. . .

• . . •

Actional Actional by cond.

#### Docketed Comment Number: 49

Commenter: Virginia Electric and Power, Company .. ,

• ·

Response(s): This comment was docketed and reviewed by the staff. The submittals which the commenter references--those of the Atomic Industrial Forum and the Utility Nuclear Waste Management Group--have been responded to by the staff. No additional items were found in this comment which require a response.

.

• . •

1.12. · · ·

1.5

~ 1 M

۰. .

Hephone 312/972-6677

(um-4)

41 PR 38081

and

46 FR 5177

DOCKET

PROFORED RULE

DOCKET HURLEY

PEDEDSED RIH



March 4, 1982

Mr. G. W. Roles U. S. Nuclear Regulatory Commission Mail Stop SS-197 Washington, D. C. 20555

Dear Mr. Roles:

Subject: Comments by Argonne National Laboratory on Environmental Monitoring Costs in Draft EIS on 10 CFR Part 61

Reference: Ltr. J. H. Kittel to R. Dale Smith, "Comments by Argonne National Laboratory on NRC Proposed Licensing Requirements for Land Disposal of Radioactive Waste (10 CFR 61), and supporting Environmental Impact Statement (NUREG 0782)" December 14, 1981.

In response to your recent telephone inquiry, attached are the environmental monitoring cost estimates developed by Argonne National Laboratory for a reference LLW disposal site. These estimates formed the basis for our comments in the reference letter that the direct operation cost for environmental monitoring shown in Table 3.6 of NUREG 0782 is believed to be inadequate.

Please let me know if further information is needed.

Yery truly yours,

1271

J. Howard Kittel, Manager Office of Waste Management Programs

JHK:pf

cc: S. A. Mann, DOE-CH

#### DRAFT EIS ON 10 CFR PART 61

#### COMMENTS ON ENVIRONMENTAL MONITORING COSTS

The environmental monitoring costs given for the Reference Site in Table 3.6 (page 3-39) are \$543,000 over 20 years, or \$26,700 per year. The environmental monitoring program for the Reference Facility is discussed in Appendix E, Section 5.2.6 (pages E-55 and E-58). A preliminary estimate of the costs can be obtained from the sampling and analysis schedule in Table E-10 (page E-57) and is given in the following table. For this cost estimate, we have assumed that 1) the particulate air sample to be analyzed daily for gross betagarma activity is counted on-site as part of routine operations, 2) the other samples are sent to an outside commercial firm for analysis, and 3) the cost per analysis are average charges by firms performing this work, expressed in 1981 dollars.

#### Annual Environmental Monitoring Costs

Sample	No. of Locations	Frequency of Analysis	No. of Analysis	Cost per Analysis	Total Cost
External Gamma-TLD	50	Quarterly*	200*	\$25	\$ 5,000
Atmosphere - Particulate	1	Weekly	52	80 (7-spec)	4,160
Atmosphere - Charcoal	1	Weekly	52	20 ( <sup>131</sup> I)	1,040
Soil and Vegetation	10	Quarterly	40	20 (β-γ) 20 (α) 40 (tritium)	800 800 1,600
Off-Site Wells	5	Semi-annually	10	80 (Y-spec) 20 (a) 40 (tritium)	800 200 400
Site Boundary Wells	10	Semi-annually	20	80 (Y-spec) 20 (a) 40 (tritium)	1,600 400 800
Disposal Area Wells	15	Quarterly	60	80 (γ-spec) 20 (α) 40 (tritium)	4,800 1,200 2,400
Trench Sumps	58	Monthly	70**	80 (Y-spec) 20 (a) 40 (tritium)	5,600 1,400 2,800
					\$35,800

<sup>\*</sup>Paragraph 5.2.6.4 (page E-58) states 12 locations are to be analyzed monthly, for a total of 296 analyses. The number in this table was taken from Table E-10

\*\* Assuming water was present 10% of the time.

The University of Chicago

ARCONNE UNIVERSITIES ASSOCIATION

In addition to the purely analytical costs, expenses of sample collection, sample preparation, quality control, maintenance of sampling equipment, record keeping, supplies, and other expenses may increase the analytical costs by a factor of up to two. If the analyses were performed in-house by personnel who also perform other work, such as the sample collection and preparation, some cost saving would result. We believe a more realistic estimate is \$60,000 per year for the total program.

2

.

۰.

7.

• • • • •

· · · · ·

#### Docketed Comment Number: 50

### Commenter: Argonne National Laboratory

<u>Response(s)</u>: The environmental monitoring costs submitted by ANL were requested by NRC to establish a more accurate basis for cost projections in the FEIS. The ANL estimates were considered along with other estimates in developing the revised monitoring costs contained in the FEIS.

.

~

UNITED STATES ENVIRONMENTAL PROTECTION AGENCY WASHINGTON, D.C. 20460 122 MAR 18 1982 '82 111 19 F12:16 En ETE PR-2etal

Mr. Samuel Chilk Secretary of the Commission Nuclear Regulatory Commission Washington, D.C. 20555



Dear Mr. Chilk:

In accordance with Section 309 of the Clean Air Act, as amended, the U.S. Environmental Protection Agency (EPA) has reviewed the Nuclear Regulatory Commission's (NRC) proposed Licensing Requirements for Land Disposal of Radioactive Waste (46 FR 38081) and the accompanying Environmental Impact Statement (NUREG-0782).

EPA believes it is appropriate for NRC to use both performance objectives and prescriptive requirements in its proposed regulations. These proposed regulations and the supporting analyses are an important step in solving the nation's lowlevel radioactive waste (LLW) problems. The information presented by NRC will be of considerable assistance to EPA in its low-level radioactive waste standards program.

EPA is concerned that the handling of low-level waste licensing in Agreement States may become inconsistent, so EPA recommends that NRC help the Agreement States adopt consistent state regulations and procedures.

NRC solicited comments on possible duplicative requirements for effluent releases and broker activities under the Comprehensive Environmental Response, Compensation and Liability Act of 1980 (CERCLA). This "Superfund" law exempts from notification "any release of source, special nuclear, or byproduct material ... in compliance with a legally enforceable license, permit, regulation, or order issued pursuant to the Atomic Energy Act of 1954" (CERCLA Section 101(10)(K)). Radioactive releases from nuclear waste disposal facilities which are not in compliance with an NRC license, permit, regulation, or order fall within the reporting requirements of CERCLA. Furthermore, as part of the notification regulations under CERCLA, EPA is planning to develop a notification scheme for releases of radioactive materials not licensed under the Atomic Energy Act of 1954 or the Uranium Mill Tailings

Acimowhered by come: 13-12-582

Radiation Control Act of 1978. EPA wishes to minimize duplicative reporting requirements for releases reported to other agencies. EPA intends to work with NRC to minimize duplicative reporting requirements to the extent possible.

EPA has regulatory responsibility for the disposal of hazardous wastes under the Resource Conservation and Recovery Act, as amended (RCRA). RCRA, Section 1004(27), specifically exempts "source, special nuclear, or byproduct material as defined in the Atomic Energy Act of 1954, as amended." Additionally, Section 1006(a) of RCRA states that "Nothing in this Act shall be construed to apply to ... any activity or substance which is subject to the ... Atomic Energy Act of 1954 (42 U.S.C. 2011 and following) except to the extent that such application (or regulation) is not inconsistent with the requirements of such Acts." RCRA does not address the issue of hazardous chemicals mixed with radioactive materials. We believe the most positive way for NRC and EPA to establish jurisdication over these wastes is in a memorandum of understanding. Such a memorandum would enable both agencies to avoid excessive costs and duplicative licensing of wastes. Furthermore, close coordination of EPA's RCRA and NRC's nuclear waste requirements is necessary in areas such as manifest tracking systems, groundwater protection, technical requirements, and financial assurances, since some NRC licensed wastes may be disposed of in EPA permitted facilities. A memorandum of understanding could serve as a vehicle for accomplishing this.

Hazardous and toxic chemicals are frequently present in these nuclear wastes. EPA is particularly concerned that these hazardous and toxic non-radioactive chemicals and their health impact are not considered in this proposed rule and EIS. We consider the rule and EIS deficient in this regard. Accordingly, EPA has rated this draft EIS ER-2 (environmental reservations and additional information requested).

EPA has divided its enclosed comments into major and minor comments on both the proposed regulations (10 CFR 61) and the supporting draft Environmental Impact Statement (EIS). Should you have any questions on our comments, please call Dr. W. Alexander Williams (755-0790) of my staff.

Sinserely yours,

Paul C. Cahill Director Office of Federal Activities

#### Major Comments on 10 CFR 61

1. The 500 millirem per year performance objective for an inadvertant intruder limit is not appropriate as a "regulatory limit." It is not a regulatory limit which will be monitored against for compliance. Nor is it a triggering level for an action such as an accident-related Protective Action Guide. The 500 millirem per year as applied in Part 61, is the design basis for the waste classification system. We believe that if "As Low As Reasonably Achievable" (ALRA) principles were applied, the exposure to the inadvertant intruder would be lower than 500 millirem per year. This is evident from XRC's own analyses in the regional case studies, which use realistic assumptions and the proposed prescriptive requirements. • •

(Ź)

2. Setting an individual exposure limit at the site boundary is appropriate. The 25 millirem per year is in the correct range of values if, as the NRC indicates, that range includes 1 to 25 millirem per year. We are assured by the NRC analyses and their own statements in the DEIS that they should be able to establish a regulatory limit compatible with any future EPA standard established using ALARA principles.

3. It is not appropriate to adopt the contaminant level of 4 milliren per year from the National Interim Primary Drinking Water Regulations as a performance objective for contamination of public drinking water supplies from LLW disposal. The National Interim Primary Drinking Water Regulations were established in the context of levels of contamination, which if exceeded, would require mitigating action by the water supply authority. It is also not appropriate to allow one group of radioactive materials users to contaminate a water supply to a limit which would preclude other releases from nuclear power plants. hospitals, and other users. It does not appear from the NRC analysis that the LLW disposal requirements are so sensitive to this limit that a lower value would be difficult to meet. It should also be noted that the 10 picocurie per liter value for uranium and therium is not part of the National Interim Primary Drinking Water Regulations. EPA would not object to the use of the National Interim Primary Drinking Water Regulations for protection of groundwater. In forthcoming RCRA land disposal regulations we expect to use the drinking water regulations as one aspect of groundwater protection. . . .

4. The NRC's intentions in applying 10 CFR 20 to the operational phase of LLW disposal is unclear. This arises out of the lack of specificity in Section 61.43 in view of the requirement of Section 61.41. This ambiguity is enhanced in the DEIS: see Vol. 1, Section 5.1.3 (page 35); Section 5.2 (bottom of page 38); Vol. 2, Section 6.3 and Section 6.4 (page 6-13); and Section 6.6 (page 6-18).

الم المراجع الم المراجع المراجع

A-153

\*82 WR 19 Fi2:16

DOLISIE

Detailed Comments

of the U.S. Environmental Protection Agency (EPA) on the U.S. Nuclear Regulatory Commission's proposed licensing

Requirements for Land Disposal of Radioactive Waste (10 CFR 61) and draft

Environmental Impact Statement

(NUREG-0782)

It is our understanding, after talking to NRC staff, that the ottsite exposures during operation of the disposal site (approximately 20 years) should be held to the same limits as the long-term offsite exposures. To make this clear, it should be stated that Section 61.43 (use of 10 CFR 20 limits) applies to occupational exposures only. Section 61.41 should be broadened to include direct radiation exposure.

2

5. Section 61.59.b makes it clear that active institutional controls may not be relied on for more than 100 years following transfer of control of the disposal site to the owner. This is appropriate for risk assessment and as a basis for design criteria. However, the regulations should contain a positive requirement that active institutional controls should be established for this period, since this is the design basis of the facility. It is also clear from the DEIS (Vol. II, page 4-69) that institutional maintenance of records of the nature of the hazard is desirable over a longer period of time. This should also be made a positive requirement of the license transfer to the site owner, although it is realized that it cannot be depended upon.

6. Class A segregated wastes will be put in separate trenches from the stable Class B wastes and will have potentially permeable trench covers. The active maintenance of such trenches can be expected to be extensive for many years. NRC should indicate how they plan to develop long-term stablility of the Class A trenches. Although Class A wastes may not present a serious radiological public health hazard, they may, in the eastern United States, under certain hydrogeological conditions, cause site instability problems, pose a "public nuisance", and, more important, the non-radioactive chemical portion of Class A wastes may cause significant ground and surface water pollution just as sanitary landfills may.

7. We urge the NRC to give a high priority to fulfilling its pledge to conduct performance, safety, and cost/benefit analyses for other readily available disposal methods such as hydrofracture, deep well injection and disposal in a mined cavity. These alternatives could show lower intruder impacts. Promulgation of 10 CFR 61 should not be delayed to permit consideration of these alternatives, however.

8. Because it is based solely on the intruder scenario, Table 1 presents some practical incongruities. For example, tritium and some of the short halflife isotopes would have heat rates that clearly would be impractical ( i.e., self boiling of tritiated water and, decomposition of the solidifying medium). Also, it does not appear that some of these high specific activities exist in any actual waste stream. 9. Part 61 shows no consideration of hazards from other chemicals and toxic substances that may be associated with the waste. This is a particularly significant omission because some of these materials may nave essentially infinite lives compared to many of the radioactive constituents. As a minimum, Part 61 regulations should indicate that these materials must be handled in a manner compatible with RCRA regularements.

3

10. The NRC proposes to develop a manifest tracking system which is. "somewhat similar" to the EPA's hazardous waste manifest system. NRC and EPA should seriously consider coordination and possible integration of the two systems. At the least, the two systems should be compatible because of the possible future need to transfer wastes from NRC regulatory authority to EPA authority, or vice versa (i.e., slightly ranoactive LLW whose primary hazard is from non-radioactive substances) and for future interaction between NRC and EPA on the ocean oisposal of LLW.

#### Minor Comments on 10 CFR 61

1. (Sect. 61.52.a;6): It states that, "Waste must be placed and covered in a manner that limits the gamma radiation at the surface of the cover to levels that are a few percent above the background levels of the site." This would be difficult to enforce in its present form due to the ambiguity of "a few percent" and the variability of background levels.

2. (Sect. 61.55, Table 1): The table should have a title and appropriate labels for the columns.

3. Methane, carbon dioxide, and other waste decomposition gases generated within a shallow disposal trench can build up sufficient pressures to directly affect the stability of any engineered trench covers, particularly if the trench covers are impervious and not properly vented. Gases can also be generated from the decomposition of wastes in arid disposal sites, even in the absence of significant precipitation.

Gases from sanitary landfills have travelled underground for hundreds of feet. Gases from Class A trenches, therefore, have the possibility of affecting the stability of the Class B trenches if proper precautions are not taken. Gas generation is a long continuing process which commonly extends for 50 years or more, requiring active repair and maintenance work on the trenches. Part 61 does not indicate any consideration of this phenomena.

4. We recommend that the following additional or similar requirement be added to Subpart D, "a disposal medium with a permeablility sufficiently low to cause the accumulation of water in the trench should not be used, especially for Class B wastes, unless compensating measures are taken to prevent or reduce the leaching of radioactive materials from the waste such as (1) installing an impermeable trench cover to keep water out of the trench or (2) preparing wastes in a low-leachable form." 1.1.1.1.1

4

5. Part 61 or its explanatory preamble should contain a clear statement that, "The intent of site selection is to insure that off-site migration or releases of radioactivity from waste disposed in a facility shall not exceed specified health and safety limits. Although it is reasonable to expect success of a properly sited facility in providing the necessary confinement, this cannot absolutely be guaranteed in advance of actual use." 

#### Major Comments on DE15 for 10 CFR 61

, in the second second

1. The need for cooperation between EPA and NRC in the development of a final 10 CFR 61-is indicated by the intention of NRC to include "Specific concentration limits for the disposal of important naturally occuring and accelerator produced nuclides" in a planned regulatory guide on the classification of waste. (See page 42 of DEIS Vol. 1 and page 7-22 of Vol. 2). Natural radionuclides, particularly radium, are expected to be included in EPA regulations under the Resource Conservation and Recovery Act. Positive steps should be taken to ensure compatibility under these two jurisdictions . . . . .

, • , • . . . بسأحيها والمراجع ومردان 2. In the preamble of the regulation (p. 38091, column 2) and in the DEIS, it is indicated that engineered barriers such as concrete covers are assumed to have an effective life of 500 years. Outside of comments received at public hearings, there is no indication of any basis for such an assumed lifetime. NRC should discuss data which confirm the lifetime of engineered barriers and criteria for approving barrier designs.

5.1 . · · 3. Honitoring at LLW disposal facilities will be an important activity and will require a regulatory guide from NRC. This should be added to NRC's commitment to prepare regulatory guides in the future. (Vol. 2, Section 2.2 p. 2-4). This guide should establish "action levels", for elevated levels of radioactive materials in the environment to indicate when increased monitoring and corrective actions should begin.

4. The NRC calls for Class B wastes to be stable for 150 years. We believe that criteria should be given that will reasonably assure a 150 year waste stability.

tability.

5

5. Actual experience at existing sites has shown that under certain hydrogeological conditions, such as outlined in Base Case 3, the direct overflow of contaminated water to land surface has been a very important, if not the dominant, pathway. The "direct to land surface overflow" pathway also has a very short travel path and does not give the benefits of delay for decay afforded by the longer travel paths and slower travel times of the ground-water pathway. The short- and long-term impacts of the "overflow to land surface" pathway should be. evaluated and compared with the ground-water pathway for Base Case 3 and for all similar test cases.

6. (Vol. 4; Append. M, Sect. 3): The DEIS should make estimates of maximum individual and population health risks, projected over time, for each of the pathways evaluated. As a part of this evaluation, the integrated activity moving through each pathway and its resultant population dose should be presented.

7. (Vol. 2, Sect. 3.8, p. 3-48, and Append. G, p. G-98) ): The doses for transportation are based on a 1972 USAEC report on the transportation of LLW from nuclear power plants. We believe that the quantities of wastes and level of radioactivity in them are considerably higher in actual experience per shipment than was predicted in the 1972 estimates. Therefore, estimated dose levels should be adjusted accordingly. Also, Table 5.5 and the discussion on p. 5-15 should be adjusted accordingly. .

8. The DEIS considers the time value of money in the estimation of the postoperational (closure and institutional control) costs but does not consider the time value for the design and operational costs. The operational costs which occur in the years 1 through 20 should also be discounted. Otherwise, combining the operational and postoperational costs to represent the total disposal costs for each alternative may result in an incorrect cost ranking of alternatives.

<u>.....</u>

. '

9. The cost estimates for institutional control need to be reexamined. After the several manipulations involving constant dollars, inflation and a nominal interest rate are made. It is unclear on what basis the final institutional control costs are stated.

10. The DEIS is deficient throughout in its failure to account for and assess the potential environmental impact and health risk from the non-radioactive chemical, hazardous, and toxic materials in the LLW. The contamination of ground and surface waters and risk to inadvertant intruders could be significant from non-radioactive materials. Souten's set 1

#### . . . . . . Minor Comments on DEIS for 10 CFR 61

1. The short-term impacts of LLW disposal have been underestimated.

In the Summary, in Chapters 3 and 4"of Volume 2, and in Appendices D and G, the radionuclides considered are only those with long half-lives 

3

Ð

(5)

(b)

or those occurring in "significant" quantities in LLW. For short-term impacts, such as radiation doses delivered during the operational phase of a waste disposal facility (including occupational, population and accident doses), the source term should include the presence of shorter-lived, gamma-emitting radionuclides:

6

2. The ground-water pathway is not significant for many arid zone sites. In arid regions, attention should be given to the upward translocation of radionuclides by plants and animals and by the upward "wicking" effects of the strong evaporation potential on capillary water in the soil. Transport, and possibly erosion, by wind should also be considered.

3. The EIS does not give any specific distances for separating the Class A and Class B trenches. This distance could be significant in assuring overall site stability.

. . . . . .

•

4. It is difficult to follow the projections of waste volumes as given in Tables D.25-D.26, which are based on Tables D.9 and D.11-14. For example D.9 gives untreated waste volumes by "region" while Tables D.25-26 list wastes by "spectrum".

5. Carbon-14 from Light Water Reactors (LWR) would more appropriately be scaled to Co-60. In the LWR, C-14 is produced both in the fuel and coolant. Except in cases of gross fuel failure, C-14 in the fuel stays there. Consequently, C-14 that ends up in LLW originates from activation in the reactor coolant. Therefore, it would seem more appropriate to scale C-14 to an activation product such as Co-60.

6. (Vol. 2, p. 6-7, last sentence); An explanation or reference should be given for the assumed release fraction 0.1 of the radioactivity within the waste packages involved in a fire.

7. (Vol. 2, Chap. 6, Sect. 6.2): Accident probability numbers would be helpful in placing the accidents and potential consequences in perspective. Data is available to at least make reasonable probability estimates.

9. (Append. E, Sect. 2.2): Consideration No. 1 should be expanded to include "and other discontinuities in the geologic media which increase the perseability significantly." For example, sand lenses or layers and desiccation cracks, as well as other unexpected features may be encountered, in addition to those listed.

9. (Append. E., Sect. 3.2.2): Texture and mineral composition are critical parameters for radionuclide retention by a geologic medium. It would be helpful to plot the composition of the Schwinn Formation on a United States Department of Agriculture (USDA) Textural Classification Chart in addition to the descriptive terminology of sandy loam and loamy sand. The advantage of using the USDA chart is that it shows the percent clay, silt and sand fractions of the medium which are textural parameters correlative with permeability and the sorption potential of a disposal medium.

7

(21)

65

66)

68

10. (Appendix J): Insufficient data are presented to make reasonable predictions about the sorption capabilities of the geologic media. For example, only the percentage of the silt-size fraction was given when the clay-size fraction of the medium is more important in assessing its capabilities to retain radionuclides. The Eh and pH are not presented either.

11. (Appendix'J, Sect. 1.1.1): In the first sentence, change "underlain" to "overlain."

12. <u>(Append. J. Sect. 1.1.3)</u>: The factors that contribute to increased permeability should be briefly described in this section. For example, in glacial tills, this includes sand lenses and desiccation cracks.

13. (Append. J., Sect. 1.1.9, 1.2.9 and 1.3.9): In a detailed site evaluation, the following additional information is needed: texture of the medium described in percentage of sand, silt, and clay-size fractions; mineral composition and organics described quantatively for each size fraction and a weighted average for total sample; cation exchange capacity correlated to clay mineral and organics content; Eh and pH of medium; chemistry of the groundwater; and naturally occurring radionuclides in the medium.

14. Executive Order 12291 requires government agencies to use a 10 percent real discount rate in developing their regulatory impact analyses. The DEIS uses an "implied" real discount rate of approximately 1 percent in the cost analyses. We believe the use of 1 percent in the DEIS should be reconciled with the requirements of the executive order. In doing so, the DEIS should present a sensitivity analysis of alternative discount rates, possibly using 1, 5 and 10 percent.

15. (Appendix Q): The methodology for calculation of capital, operational, closure and institutional costs appears reasonable. What appears to be lacking is a consistent treatment of the date attached to each cost. For example, capital costs are presented in 1980 dollars, operational costs are provided as sums of money appropriate to the 20 year operating life of the site but no particular date is attached to this amount, and closure costs are presented as 1980 costs inflated to the end of site closure. In order to evaluate disposal costs appropriately, a consistent time treatment of money is needed.

16. The last term in the long equation for postoperational costs on p. Q-44 should be corrected.

(4)

*1*6)

(17)

(zı

17. In the discussion of financial arrangements, the license applicant is to prepare an estimate of the amount of money required for closure and long-term care. We urge the use of a very careful review process on these cost estimates because past experience has shown that the licensee, with one exception where the regulator took an active part, has traditionally underestimated closure and long-term costs.

18. (Appendix G, pp. G-12/13): The dilution factor Q was taken to be equal to its pumping rate in the intruder well pathway. This dilution factor should be taken as the total groundwater flow within the plume of contamination rather than the well pumping rate.

. .

19. The methodology used to develop trench infiltration should be presented somewhere in the text or appendices. The reference cited in Appendix G on p. 68, paragraph 2, is insufficient. There is no indication that the "externally produced" value is ever allowed to vary within the analysis.

20. A clearer explanation of the basis for the indices used in calculating the interaction factors in Appendix G is needed.

21. (Append. G. p. G-68): The rate of infiltration was calculated by multiplying by the number of days for which the precipitation exceeds 0.01 in/day. This method is very simple but, in our opinion, does not meet the state of the art for estimating infiltration rates. Our technical staff is available to discuss other possible methods for calculating infiltration.

22. (Vol. 2, p. 5-22): The dose rates through the population well and surface water pathways for Case 1A (Table 5-6) is calculated to be approximately 10 times lower than for Case 1 (Table 5-3) simply due to replacing the backfill material on moderately permeable soil used in Case 1 with more permeable soil. This fact does not agree with current knowledge of the leaching process. This phenomena was explained by the waste-water contact time being longer in Case 1 than in Case 1A and a resultant higher rate of leaching. In reality, the pellicular water in the waste, which contains dissolved radionuclides, will be maintained in between the interstices of the waste all of the time. Therefore, any added percolation of gravity water will be mixed with the pellicular water during the course of percolation and will be independent of the true velocity of the water movement. Therefore, the rate of radionculide release should change very little because of changes in the permeability of the backfill material.

23. (Vol. 1, p. 19, Table S.5): The values listed under Body & Bone need to be explained in a footnote as "the number of years at which this impact level exists."

1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 -

. .:

3

(P)

3

3

(33)

39)

24. At several points in the DEIS (e.g. Vol. 1, p. 23), the comment is made that "the potential hazard quickly drops to about 1000 millirem per year to bone at about 500 years following facility closure." The term "quickly" does not seem compatible with a 500 year time period.

9

•••

(36)

(37)

(38)

25. (Vol. 1, p. 20): The statement, "Maximum annual thyroid doses are in the range of 850 mrem at the intruder and population wells, 270 mrem at the population well, and 12 mrem at the surface water body" is ambiguous. This leaves in doubt what the dose is at the population well.

26. (Vol. 1, p. 26, Table S.7 5 Vol. 2, p. 4-41, Table 4.19): Several of these costs (those with double asterisk) are not comparable with the others because they are based only on the 10% volume of waste employing that technique. To make the costs comparable, costs for the other 90% of the wastes should be figured and a weighted average presented.

27. (Appendix N): EPA's authority under the Marine Protection, Research, and Sanctuaries Act of 1972 for ocean disposal of radioactive waste should be listed.

28. (Appendix N): EPA's proposed guidance for Occupational Exposures should be discussed. (Federal Register, Vol. 46, No. 15, Friday, January 23, 1981, page 7836).

29. (Appendix N, p. N-5): EPA's schedule for its Low-Level Waste Standard is currently under review and may be revised. We will notify NRC of any such revision.

30. (Appendix N, p. N-19): National Primary Drinking Water Regulations is incorrect, it should be <u>Interim</u> National Primary Drinking Water Regulations.

31. (Appendix G): Tables G-3 through G-10 are poorly titled and measurement units are not well identified.

#### Docketed Comment Number: 51

#### Commenter: U.S. Environmental Protection Agency

<u>Response(s)</u>: <u>Item 1</u> - As noted in the response to Item 3, Comment 11, the staff considered toxic and hazardous chemicals to the extent practicable in developing the requirements of 10 CFR 61. However, the primary focus of the rule is the safe land disposal of radioactive wastes.

The staff recognizes that toxic and hazardous chemicals may be present in radioactive wastes and has incorporated provisions in the rule which reflect this fact. (These provisions are cited in Item 3, Comment 11 and are not repeated here.) We believe that the rule in its current form provides an adequate level of protection with respect to those substances which may be associated with LLW and should not cause EPA to have any further reservations about the rule or its supportive EIS in this regard. The Commission intends to continue working closely with EPA on this matter.

<u>Item 2</u> - The staff has modified the language of §61.41 and 61.43 to clarify the purpose of applying 10 CFR 20 to the operational phase of LLW disposal and to remove any ambiguity between these two sections of the rule. It is the Commission's intent that the provisions of Part 20 will apply to all aspects of radiation protection during operation except for releases of radioactivity from the site which will be governed by the more stringent requirements of § 61.41. The specific sections mentioned in EPA's comments have been changed to reflect this.

<u>Item 3</u> - NRC does not intend to specify concentration limits for naturally occurring and accelerator-produced nuclides in Part 61. As EPA notes, NRC does not have statutory authority over such materials. In response to comments, however, and to provide for safe disposal of certain long-lived naturally occurring and accelerator-produced materials by the Agreement State (which do regulate such materials), NRC plans, as a minimum, to develop guidelines for disposal of Ra-226. This information will be made available to EPA to help ensure compatibility.

,

. .

Item 4 - Based on the analyses carried out in the DEIS, exposures to a potential inadvertent intruder from almost all waste streams are in the range of a few millirem after a few hundred years. NRC thus selected 500 years as a general upper bound upon which reliance on any particular intruder barrier would be allowed. As such, 500 years serves as a design objective. NRC staff has no specific engineering data on the expected maximum lifetimes of various intruder barriers which might be used but believes that depth of cover and concrete engineered structures should be capable of providing protection for that length of time. Specific guidance which should be considered by an applicant relating to intruder barriers will be set out in a technical position paper prepared by the staff on facility design and operation.

<u>Item 5</u> - NRC's future regulatory guides will include a guide on environmental monitoring of LLW disposal facilities. With respect to the establishment of "action levels" for elevated concentrations of radioactive material in the environment, the staff feels that this is a site-specific subject and will require applicants to specify such levels as a part of the license application. These levels will be reviewed and acted upon by the staff for each proposed LLW disposal facility.

<u>Item 6</u> - NRC agrees and recognizes the need for additional guidance regarding waste form stability. In this respect, NRC is preparing a technical position on waste form which will provide guidance, criteria and acceptable tests to demonstrate waste stability.

<u>Item 7</u> - An estimate of the potential impacts from leachate overflow to the land surface from disposal cells has been included in the FEIS.

<u>Item 8</u> - In the draft EIS, NRC expressed radiological impacts associated with operation of a near-surface disposal facility in terms of responses to individuals and populations. NRC did not covert or express these exposures in terism of risk because of the difficulty of accurately assessing risks of exposures to future populations and the small numbers of individuals who could receive a potential exposure. The staff reconsidered its decision on this issue, but has not changed its position. Expressing exposures in terms of risk would ••••

involve new work and time which is not warranted given the urgent need for Part 61 and the limited additional information which would be provided. In the EIS, NRC compared calculated doses on a common basis to existing standards which are expressed in terms of dose equivalent. NRC has, however, attempted to express the overall impacts of Part 61 in the FEIS in a clearer manner such that comparison of alternatives and unmitigated impacts are easier to discern and understand. ÷.,

In addition, in response to this comment and to place in perspective the potential risk associated with the doses calculated in this FEIS. NRC has included a section in the summary which provides dose response relationships as set forth in International Commission on Radiation Protection Publication 26. The reader can use these to estimate the level of risk associated with doses calculated . . for the various alternatives. , · · · · 

· 1.

Item 9 - NRC staff believes that the assumptions used in approximating transportation exposures are adequate for the purposes of the EIS. The EIS is not a generic environmental impact assessment of low-level waste transportation and disposal. Rather, it is a decision document with which to consider alternatives for establishment of performance objectives and technical criteria in the Part 61 rule. As part of considering alternatives, a number of costs, radiological impacts, and other impact measures associated with various disposal technology cases were calculated. Transportation impacts were not the principal impact measures used to arrive at decisions but were included for completeness. Issues related to transportation safety are beyond the scope of the EIS.

. •

The document referenced by the commenter (Reference 19) was used to estimate an average exposure per shipment mile. This was multiplied by the number of shipments (which is a function of waste volume and activity) and by the average distance to the reference disposal facility. This was used to obtain exposure estimates which could be compared from one disposal technology alternative to the next. Thus, the absolute values of the transportation exposures calculated are not as important as the comparison between alternatives. This is believed to be in keeping with the scope of the Part 61 rule and EIS. NRC staff members are aware of more sophisticated calculational methodologies to calculate expoand the second second

 sures due to waste transportation (References 20 and 21), but believe that such methodologies are more appropriate for rulemaking actions on transportation safety than for rulemaking actions on low level waste disposal. In any case, no matter what the levels of radioactivity in different forms of waste, the established DOT radiation limits at vehicle surfaces would still apply. Wastes having higher surface radiation levels require more shielding.

.

In addition, a more recent environmental statement (Reference 20) was examined. This statement, which calculated transportation impacts in a much more sophisticated manner than Reference 19, estimated a total dose (except doses from stopovers, which as in the Part 61 EIS are considered separately) to the public of 520 person-rems from shipment of radioactive material in 1975. Out of a total of 2.19 million packages shipped, only about 7% contained radioactive waste. Assuming 40 waste packages per shipment and an average shipment distance of 400 miles, one can estimate an average exposure of .0008 person-millirem/shipment mile, which is a factor of about 10 less than that assumed for the DEIS. Thus, NRC staff believe that the calculated transportation exposures have been handled in an acceptable manner. • 1. . - , ·

Item 10 - For the FEIS, disposal costs have been recalculated using a time value of money approach on capital and operational costs.

1

and the second Item 11 - In the DEIS, three levels of site maintenance during the institutional control period (and corresponding costs) were examined: high, medium and low. The institutional control costs were presented as the amount of money that would have to be collected as a surcharge on received waste over the operating life of the site to pay for a given level of maintenance. Money collected from the surcharge is assumed to be placed into an interest-bearing sinking fund. The average interest rate and inflation rate assumed over the operating life of the site and institutional control period were 10% and 9%, respectively. and the second second

Based upon revised estimates of monitoring, personnel and other costs as discussed in Appendix C of this volume, revised institutional control costs have been calculated for the FEIS.

. .

1.11 Sec. And a sec. and the second . :

<u>Item 13</u> - NRC staff recognizes that the short-term impacts may have been underestimated since only a few short half-lived nuclides were considered. NRC staff does not believe this is significant, however, since short term actions taken to protect against exposure to the longer half-lived nuclides considered in the analyses should also be adequate to protect against exposure to the shorter half-lived nuclides that would also be present in the waste.

Item 14 ~ As stated in the Branch Technical Position on Site Suitability, Selection and Characterization (Reference 22), the staff considers the contact of soil moisture with the waste containers and waste to be the major mechanism for potential release of radionuclides. With respect to arid sites, the staff considers the movement of soil moisture and vapor upward by evaporation potential or by plant uptake to be the most significant pathway for migration of radionuclides into the general environment.

Item 15 - The staff agrees that the distance between Class A and Class B trenches could be significant in assuring overall site stability. However, the staff feels that site-specific characteristics must govern the determination of an appropriate distance between these types of waste trenches. For this reason, the EIS does not contain a specific figure for separation.

<u>Item 16</u> - The projections of waste volumes are not based on Tables D.11-14. It is possible, however, that NRC was insufficiently clear in Appendix D of the DEIS regarding waste volume projections.

Table D-9 presents volumes projected to the year 2000 of 36 waste streams in each of the 4 regions considered in the EIS. The volumes are shown "untreated," which means as-generated volumes prior to further treatment or packaging for shipment. For example, volumes for PWR and BWR concentrated liquids are given as-generated prior to solidification. The volumes which will be shipped, however, are a function of the waste spectrum considered. A "waste spectrum" is a term used to indicate the total volume and properties of the 36 waste streams after they have been processed by a set of selected waste treatment options. Each spectrum corresponds to a general level of waste performance in terms of waste stability, resistance to wind mobilization, resistance to leaching, and physical, chemical, and radiological properties. The spectra differ significantly in waste volumes, radioactive concentrations, and performance.

To obtain the volumes shipped in a given region, one would multiply the volume of each waste stream by factors which correspond to any changes in volume of the stream associated with processing according to a given spectra. These factors, termed volume increase factors (VIF) and volume reduction factors (VIR), are given for each of the four spectra in Table D.21. For example, the volume of PWR concentrated liquids projected to be generated in region 1 according to the level of processing associated with waste spectrum 2 is as follows:

Volume = 4.87 E+4 \* VIF/VIR

= 4.87 E+4 \* 1.82/6.0 = 1.48 E+4m<sup>3</sup>

Table D.25 presents the volumes by waste spectra (that is, as multiplied and divided by the appropriate VIF and VIR, respectively) summed over all four regions. That is, this table presents the projected total to the year 2000 of the country's waste as a function of waste spectra.

Table D.26 then presents the above volumes normalized to one million  $m^3$  of waste for waste spectrum 1. The volumes in Table D.26 represent a "reference" waste distribution and were used for the generic calculations performed in Chapters 4-6 of Volume 2 of the DEIS. The regional calculations performed in Chapter 10 of Volume 2 of the DEIS were performed using the volumes presented in Table D.9 as modified by the appropriate VIF and VIR presented in Table D.21.

<u>Item 17</u> - NRC staff believe that C-14 can be scaled to either an activation product or to a fission product is somewhat arbitrary. Like any other fission product, C-14 can escape fuel rods and enter the reactor coolant through imperfections in the fuel rods. Such imperfections can range from tiny pinholes to gross fuel failure. Thus, NRC staff believe that the current scaling is sufficient for the FEIS. Ongoing studies by NRC in which samples of reactor process wastes are being analyzed for a number of trace radionuclides

. . .

.

(including C-14, I-129, and Tc-99) should help provide a more accurate estimate of C-14 levels. Additional consideration will be given at that time to the use of scaling factors and to selection of appropriate scaling isotopes.

Item 18 - The basis for the assumed release fractions is presented in Appendix G of the DEIS and Reference 23.

Item 19 - EPA staff were contacted regarding the intent of this comment. particularly as regards their comment on the availability of data. Their intent was that NRC staff should review several years of accident reports at nuclear power stations and other licensees and, based upon any available information on accidents involving waste management activities, to arrive at accident probabilities which could be applied to low-level waste disposal facilities.

12.5

, ·· It is possible that an assessment of accident probabilities, assuming that such information is available, could have some use as supplemental information. However, NRC staff question whether such data can be applied in a straightforward manner to low-level waste disposal facilities. NRC staff does not believe, however, that it would be appropriate or desirable to delay preparation of the final EIS and rule until such data could be assembled and manipulated. NRC staff believe that even assuming that accident probabilities could be acquired, the overall conclusions of Chapter 6 of the DEIS are not likely to change. These conclusions were basically that actions taken to reduce potential inadvertent intruder exposures, reduce long-term environmental releases, and improve long-term site stability and reduce institutional control costs would also tend to improve safety during site operations.

Item 20 - The consideration referred to by the commenter is one of general applicability, but only one of many which the staff has recommended for siting of new disposal facilities. The technical requirements for site suitability in Part 61 have been elaborated upon in a Branch Technical Position on Site Suitability, Selection and Characterization (Reference 22). The staff's quidance within this paper includes the following:

.

and the second "... the hydrogeologic unit used for disposal should not have continuous permeable or impermeable anomalies such as faults or fracture zones. sand lenses, weathered horizons, or Karstic features that provide preferential pathways for or barriers to ground-water flow,"

14 J 1 . . - - -. . . . 1.1.1.1 · · · · .

Item 21 - The descriptive information provided in Appendix E of the DEIS is for a reference near-surface disposal facility located on a hypothetical site. This information is intended to provide a general picture of various site environmental conditions in sufficient detail to approximate base case costs and impacts at the reference facility. The description of site soils from the Schwinn formation is considered to be sufficiently detailed for a lay person to comprehend site conditions, and the staff feels that addition of the USDA chart would add little in the way of new information to Appendix E of the DEIS.

Item 22 - For each of the hypothetical sites described in Appendix J of the DEIS, average cation exchange capacity at the subsurface media have been provided as one estimate of the sorption capabilities of the geologic medium. Assuming that these were measured volumes, the cation exchange capability values are more useful than percentage of silt or clay in estimating sorption capabilities. Further, the reported values for silt content represent the entire fraction passing the 200 sieve (i.e., silt and clay sizes).

Item 23 - This change has been made and is included in Appendix E, Errata, in this volume.

· · · ·

.

Item 24 - As noted earlier in a comment on Appendix E of the DEIS, the descriptions in Appendix J of the DEIS are not intended to be all-inclusive. Consequently, although the commenter has identified some information that is not included in Section 1.1.3 of Appendix J. the staff does not feel that this level of detail is necessary or appropriate to the appendix material. 

12

Item 25 - As noted above, the descriptions in Appendix J of the DEIS are not intended to be all-inclusive. The Branch Technical Position on Site Suitability, Selection and Characterization (Reference 22) discusses the informational needs projected by the staff for preparing a detailed site evaluation. These needs may be more sharply focused when a specific site proposal is submitted. However, no EIS text changes are considered by the staff to be necessary with respect to this comment.

. . . . . . . . . .

الأبور المردية الأخرار وخرجا المحاج المراجع والمرجع والمحاج والمراجع and the second **`**. Item 26 - The commenter is not quite correct in his comment. In order to assess the long term costs for institutional control, NRC assumed that funds would be collected as a surcharge on waste received at a disposal facility and placed into an interest-bearing sinking fund. In the calculations, a 10% interest rate was assumed. (States are often restricted by law regarding the types of securities which may be invested in; securities offering very high interest rates may not be allowed.) Using this interest rate, and an assumed average inflation rate of 9%, a real interest rate of about 1% is obtained. This real interest rate is not the same thing as the discount rate.

Regarding Executive Order No. 12291 (Reference 24), it is not clear that the proposed Part 61 constitutes a major rule as defined in the order and, therefore, it is not clear that Part 61 falls under the scope of the order. In addition, the Commission is currently considering steps to be followed to generally implement the order. Until a decision is reached, the Commission has adopted an interim procedure of following criteria developed from Executive Order No. 12044 (Reference 25). As part of the analysis conducted for the FEIS, however, the staff has reviewed Executive Order No. 12291, as well as the draft Office of Management and Budget (OMB) Interim Regulatory Impact Analysis Guidance on implementing Executive Order No. 12291 (Reference 26).

The OMB guidance states that an annual discount rate of 10% should be used. The guidance also states that where it appears desirable, other discount rates also may be used to test the sensitivity of the results.

The NRC staff believe that a 10 percent discount rate is unrealistically low when applied to siting and operating a low-level waste disposal facility. Support for this belief is provided by comments on the draft EIS from the operator of an existing disposal facility. (See commenter 44.) This commenter's position is that operation of a disposal facility is a high risk venture and that the discount rate should reflect this risk. Subsequent input from this commenter indicates their position that a 10% discount rate is unrealistically low. A considerably higher rate would be used by the commenter for siting and operating an actual low-level waste disposal facility (Reference 27). A comparison between disposal costs associated with a number of different discount rates is, however, included in Appendix C to the final EIS.

Item 27 - See the response to item 10 above.

<u>Item 28</u> - As the commenter has noted, the equation on page Q-44, Appendix Q of the DEIS contains two typographical errors. The last term in the equation should be the same as the equation for LTC on p. Q-42. That is:

 $\frac{iTT_{0}PV_{80}(1+j)^{m}}{V_{w}[(1+i)^{1T}o-1](1+i)^{1T}c}$ 

<u>Item 29</u> - The staff agrees. The Commission will review carefully all aspects of a license application for waste disposal under 10 CFR 61, including the estimates of closure and long-term care funds.

Item 30 - There is no discussion of a dilution factor or a pumping rate on the indicated pages. For the intruder well, the dilution volume is a function of site specific conditions and is given as the total volume of water percolating through the disposal units under natural conditions. The well pumping rate (7700  $m^3/yr$ ) was taken as a lower bound as it is about the minimum required for a well to be considered useful. This approach gives reasonable yet conservative results.

Item 31 - The discussion on p. 68 of Appendix G of the DEIS and in the reference cited is believed to be sufficient. The variation of the percolation rate in the analyses is summarized on the p. 69 of the appendix.

Item 32 - Additional information is presented in Reference 23.

<u>Item 33</u> - NRC staff believe that more sophisticated methods of calculating infiltration rates are more appropriate for an analysis of a specific site where actual site data may be acquired and used. Given the generic, non site-specific nature of the EIS analysis, the approach taken by NRC is believed to be sufficient for the purpose of rulemaking. A complex model cannot compensate for a lack of site specific data and is furthermore of less practicality when considering a large number of alternatives.

Item 34 - The Case 1A reduction in concentration of approximately 10 times that calculated for Case 1 results from the use of a clean, permeable sand backfill. This sand backfill reduces concentrations by three mechanisms First, the trench cap is more stable due to less consolidation and settlement in the filled trenches. Second, the greater porosity of the sand backfill as compared with the surrounding site soil will result in some reduction of water entering the backfill due to the wick effect. And third, the greater posority will result in less pellicular water (i.e., less film in contact with and leaching the wastes) than a finer textured backfill as used in Case 1. This then leads to less pellicular water being flushed off by infiltrating water.

 $\frac{1 \text{ tem } 35}{1 \text{ tem } 35} - \text{ The requested footnote is already contained in Table 5.5,}$  Appendix S of the DEIS.

<u>Item 36</u> - The commenter's point is well taken and it is noted in Appendix E, Errata, in this volume.

Item 37 - The sentence contains a typographical error. It should read, "Maximum thyroid doses are in the range of 850 mrem at the intruder and boundary wells..." This change has been made in Appendix E, Errata, in this volume.

Item 38 - This comment has been noted by the staff.

Item 39 - The U.S. Environmental Protection Agency (EPA) has authority under the Marine Protection Research and Sanctuaries Act of 1972 (P.L. 92-532) for regulation and permitting of ocean disposal activities. EPA is developing a regulatory program and permitting system for resumption of ocean disposal of radioactive wastes. NRC did not include a specific reference to this authority in Appendix N of the DEIS since the scope of the EIS was limited to land disposal. The staff has made no change to the FEIS as a result of this comment. <u>Item 40</u> - Appendix N of the DEIS reviews a number of proposed and promulgated regulations and guidance applicable to LLW disposal. Since EPA's Proposed Guidance for Occupational Exposures was of more general applicability. NRC staff did not specifically indicate it in Appendix N.

Item 41 - NRC will maintain close coordination with EPA on the development of these standards.

 $\underline{\text{Item 42}}$  - This comment has been noted and included in Appendix E, Errata, in this volume.

.....

::

רי: דיאר

.....

Item 43 - This comment has been noted by the staff.

ł

# **References:**

- General Research Corporation, "Study of Chemical Toxicity of Low-Level Wastes," NUREG/CR-1793, November 1980.
- Oak Ridge National Laboratory, "Proceedings of the Third Annual Information Meeting, DOE Low-Level Waste Management Program, November 4-6, 1981, New Orleans, LA," ORNL/NFW-81/34, December 1981.
- 3. Wild, R., "Data Base for Radioactive Waste Management, 3 volumes, NUREG/CR-1759, Dames and Moore, Inc. for U.S. NRC, November 1981.
- 4. U.S. Nuclear Regulatory Commission, Regulatory Guide 1.109, "Calculation of Annual Doses to Man From Routine Releases of Reactor Effluents for the Purpose of Evaluating Compliance with 10 CFR Part 50, Appendix I," Revision 1, October 1977.
- Hoenes, G. R., J. K. Soldat, "Age-Specific Radiation Dose Commitment Factors for a One-Year Chronic Intake," NUREG-0172, Battelle Pacific Northwest Labratories for U.S. NRC, November 1977.
- Healy, J. W., and J. C. Rodgers, "Limits for the Department of Energy Transuranic Wastes," Los Alamos Scientific Laboratory, LA-UR-79-100, January 1979.
- 7. U.S. Nuclear Regulatory Commission, Office of Standards Development. Draft Regulatory Guide 1.109, "Calculation of Annual Doses to Man from Routine Releases of Reactor Effluents for the Purpose of Evaluating Compliance with 10 CFR Part 50, Appendix I." March 1976.
- U.S. Nuclear Regulatory Commission, Office of Standards Development, "Calculational Models for Estimating Radiation Doses to Man from Airborne Radioactive Materials Resulting from Uranium Milling Operations," Task RH 802-4, For Comment, May 1979.

- 9. Moore, R. E., et. al. "AIRDOS-EPA: A Computerized Methodology for Estimating Environmental Concentrations and Dose to Man from Airborne Releases of Radionuclides," EPA 520/1-79-009, December 1979.
- 10. Greenhouse, N. A., and E. T. Lessard, "A Dosimetric Analysis of Pathways to Man From the Environmental Release of Low Level Radioactive Wastes," Prepared for Dames and Moore by Occupational and Environmental Health Analysts, Inc., 1980.

. . . .

- in the second 11. Ng, Y. C., et. al., "Transfer Coefficients for the Prediction of the Dose to Man via the Forage-Cow-Milk Pathway from Radionuclides Released to the Biosphere." University of California, Lawrence Livermore Laboratory Setting to Report No. UCRL-51939.
  - 12. Oak Ridge National Laboratory, "Modeling and Low-Level Waste Management: An Interagency Workshop," Oak Ridge National Laboratory, June 1981.

ter setter i setter i

and the second second

1997 - 1967 - 1967 - 1967 - 1967 - 1967 - 1967 - 1967 - 1967 - 1967 - 1967 - 1967 - 1967 - 1967 - 1967 - 1967 -

. .

- 13. Chem-Nuclear Systems, Inc., "Preliminary Site Stabilization and Closure Plan for the Barnwell Low-Level Radioactive Waste Disposal Facility," Chem-Nuclear Systems, Inc., May 1980.
- 14. Adam, J., V. Rogers, "A Classification System for Radioactive Waste • • • Disposal--What Waste Goes Where?," NUREG-0456, Ford, Bacon and Davis, Utah for USNRC, June 1978.
- 15. Rogers, V., "A Radioactive Waste Disposal Classification System." NUREG/CR-1005, Ford, Bacon and Davis, Utah for USNRC, September 1979.
  - 16. Leddicotte, G. W., et. al., "Suggested Quantity and Concentration Limits to be Applied to Key Isotopes in Shallow Land Burial," Management of Low-Level Radioactive Waste, Volume 2, Pergamon Press, New York, 1979.

· · · · ·

17. U.S. Department of Energy, "Technology for Commercial Radioactive Waste Management," DOE/ET-D028, May 1979.

. .

 Phillips, J., <u>et. al.</u>, "A Waste Inventory Report for Reactor and Fuel-Fabrication Facility Wastes," ONWI-20, NUS corporation for US DOE, March 1979.

· • ;;

••••

- U.S. Atomic Energy Commission, <u>Environmental Survey of Transportation of</u> <u>Radioactive Materials To and From Nuclear Power Plants</u>, WASH-1238, December 1972.
- 20. U.S. Nuclear Regulatory Commission, <u>Final Environmental Statement on the</u> <u>Transportation of Radioactive Material by Air and Other Modes</u>, NUREG-0170, December 1977.
- 21. Finely, N. C., <u>et al.</u>, <u>Transportation of Radionuclides in Urban Environs:</u> <u>Draft Environmental Assessment</u>, NUREG/CR-0743, July 1980.
- 22. Siefken, D., <u>et al.</u>, <u>Site Suitability, Selection, & Characterization</u>, <u>Branch Technical Position--Low-Level Waste Licensing Branch</u>, NUREG-0902, April 1982.
- Oztunali, O. I., <u>et al.</u>, <u>Data Base for Radioactive Waste Management</u>: Vol. 3, Impact Analyses Methodology Report, NUREG/CR-1759, November 1981.
- 24. Executive Order 12291, <u>Federal Register</u>, Vol. 46, No. 33, February 19, 1981.
- 25. Executive Order 12044, Federal Register, Vol. 43, No. 53, March 24, 1978.
- 26. Office of Management and Budget, Draft Interim Regulatory Impact Analysis Guidance for Implementing Executive Order 12291, June 5, 1981.
- 27. May 7, 1982 telephone conversation between Leslie Poppe, Corporate Health Physicist, Chem-Nuclear Systems, Inc., and G. W. Roles, U.S. NRC.

# APPENDIX B

.

.

· · ·

# STAFF ANALYSIS OF PUBLIC COMMENTS ON PROPOSED 10 CFR PART 61 RULEMAKING

# PUBLIC COMMENTS ON PROPOSED 10 CFR PART 61

On July 24, 1981 proposed 10 CFR Part 61 was published in the <u>Federal Register</u> (46 FR 38081). Proposed Part 61, "Licensing Requirements for Land Disposal of Radioactive Waste," contains requirements for disposal of low-level wastes. The original comment period expired October 22, 1981 but was extended to January 14, 1982 (46 FR 51776) to coincide with the comment period for the supporting draft environmental impact statement (NUREG-0782). The following commenters responded and copies of the original comment letters are included in this appendix following the staff's analysis of the issues raised therein:

#### Commenter Docket No. Page Homestake Mining Company ..... 1. B-185 2. Robert Reynek (?)..... B-185 Marvin Lewis..... <sup>•</sup> 3. B-186 New Mexico-Secretary for Health & Environment..... · 4. B-186 <sup>°</sup>5. Advance Medical & Research Center, Inc. ..... B-187 The Procter & Gamble Company..... 6. B-188 ć**7.** Bethlehem Steel Corporation..... University of California, LA..... Environmental Law Project..... B-189 B-190 8. 9. B-191 Advisory Committee on Reactor Safeguards..... B-193 10. Environmental Protection Agency..... , B-195 11. Department of Commerce, NOAA, Boulder..... 12. B-195 13. 14. Exxon Nuclear Co. Inc. 15. B-201 PA, Department of Environmental Resources..... B-202 16. Dow Chemical..... 17. B-204 Isham, Lincoln & Beale for Commonwealth Edison..... 18. B-210 Department of the Environment, London..... B-211 19. The Surety Association of America..... 20. B-220 Joseph H. White III..... 21. B-221 Kerr-McGee Corp. ..... B-222 22. D. M. Mathews ..... 23. B-222 Sargent & Lundy ..... 24. B-223 United Technologies/Packard..... 25. B-224 NASA, JFK Space Center ..... B-226 26. Northern Illinois University ..... 27. B-226 Department of Energy ..... 28. B-227 Nuclear Diagnostic Laboratories..... 29. B-228 University of NC ..... B-229 30. Florida Power & Light Company..... B-229 31. Wisconsin Electric Power Company..... B-231 32. 33. Alabama Power..... B-232 Law Engineering Testing Company..... 34. B-233 Commonwealth Edison.... B-234 35. Union of Concerned Scientists.... B-236 36. Sierra Club, Radioactive Waste Campaign..... B-238 37.

38.	Ohio Environmental Protection Agency	B-242
39.	Union Carbide Corp/Medical Products Division	B-243
40	Pathfinder Mines Corp.	B-245
17	Chom-Nucloan Systems Inc	B-246
41.	Them nuclear systems, inc	D-240
42.		B-247
43.	Los Alamos National Laboratory/Safeguards Systems Group	B-249
44.	Bechtel National, Inc	B-250
45.	New Mexico-Secretary for Health & Environment (See 4 also)	B-253
46.	Joel Jaffer.	B-254
47	Arizona State (learinghouse (multiple state agencies)	B-254
10	Ruke Deven Company	D 254 D-257
40.	Date Fower Company	D 050
49.	Howard University	_ B=259
50.	Altonso Scarpa	. B−259
51.	Ontario Hydro	B-260
52.	Arkansas Power & Light	B-261
53.	American College of Nuclear Physicians	B-262
54	The National Association of Insurance Brokers Inc	B-263
55	Kantucky Special Advisory Committee on Nuclean Iscues	B-264
55.	Render Deseter Composition	D 204
50.		D-200
57.	R/A Services, Inc.	8-265
58.	Mr. & Mrs. D. W. Willoughby (misdocketed and cancelled)	B-266
59.	Mr. & Mrs. M. E. Vega (misdocketed and cancelled)	B-266
<b>60.</b> <sup>•</sup>	Mr. & Mrs. J. Johnson (misdocketed and cancelled)	B-266
61	Joette Lorion (misdocketed and cancelled)	B-266
62	Harry Lawrocki	B-266
62	Doportmont of the Anny	· D-200
03. ČA	Zala M lange	D-207
64.	Zerra M. Jensen	B-20/
65.	Department of Planning and Economic Development, Hawaii	B-268
66.	Union Oil Company of California	B-268
67.	Stock Equipment Company	B-272
68.	Argonne National Laboratory	B−274
69.	SD State Planning Bureau.	B-278
70	Georgia Institute of Technology	B-279
70.	Louise Competie	B-201
71.	Louise dorening.	0 201
12.	Mr. & Mrs. Michael A. Norsworthy (misdocketed and	
	cancelled)	B-283
73.	Oswald U. Anders	B-283
74.	Yakima Indian Nation	B-285
75.	University of Texas Medical Branch	B-286
76.	Paul F. Hadala & Don C. Banks	B-287
77	Georgia Yuan	B-288
70	University of Amizona	D 200
70.	Constant file and file and file and file and the second second	0_200
79.	St Dept of Health and Environmental Control	B-298
80.	Birmingham Audubon Society	B-300
81.	Utility Nuclear Waste Management Group	B-302
82.	Werner & Pfleiderer Corp	B-311
83.	Dow Chemical (see 17 also)	B-313
84	Middle South Services. Inc.	B-317
25	Northeast Iltilities	R-310
00.	Nuclean Menitoning Suctome & Management Componation	D-333
00.	Analisan Muslean Casista	0-323
87.	American Nuclear Society	8-323
88.	Valore, et al. for lownship of Lower Alloways Creek	B-330
89.	General Electric	B-333
90.	Amy S. Hubbard	B-337
91.	Florida Power Corp	B-338
	· · · · · · · · · · · · · · · · · · ·	

•

. . .

. .

.

•

.

.

B-ii

92.	Power Authority of the State of New York	B-339
93.	State of California	B-340
94.	Arkansas Power and Light Company (see 52 also)	B-349
95.	Stone & Webster Engineering Corp	B-350
95.	Health Physics Society	B-351
97.	Don't Waste Wasnington Legal Detense Foundation	B-354
98.	WISCONSIN Electric Power Lompany (see 32 also)	8-355
99. 700	Atomic Inductrial Forum The	B-35/ D-264
100.	ALOMIC INdustrial Forum, Inc.	D-304 R-269
101.	Nuclear Engineering Division/American Institute of Chemical	D-200
102.	Engineers	B-371
103.	Conference of Radiation Control Program Directors	B-374
104.	P. Ziemer for EG&G Idaho, Inc. Program Review Committee	B-377
105.	U. of Texas System Cancer Center	B-378
106.	Carolina Power & Light Company	B-380
107.	The American Society of Mechanical Engineers	B-382
108.	Nuclear Diagnostic Laboratories, Inc. (see 29 also)	B-388
109.	North Carolina Radiation Protection Commission	B-390
110.	New England Nuclear	B-392
111.	Colorado Department of Health	B-400
112.	State of Washington	B-400
113.	The American Society of Mechanical Engineers (see 107 also).	B-402
114.	U.S. Department of the Interior	B−404
115.	Terrescond Vollar Authority	B-406
117	Tennessee valley Authority	B-421 D-422
119	Atomic Enorgy of Canada Limitod	D-423 R-121
110.	11 S Department of Friendy (see 28 a)so)	D-424 R-425
120	Argonne National Laboratory (see 68 also)	B-435
121	Argonne National Laboratory (see 68 and 120 also)	B-436
122	U.S. Environmental Protection Agency (see 11 also)	B-438
123.	General Research Corporation	B-444
124.	Werner and Pfleiderer Corporation (see 82 also)	B-446
125.	Valore, et al. for Township of Lower Alloways Creek	
	(see 88 also)	B-447
	• •	

· .

•

. . .

•

• • •

..**.** 

· · · · i ••• 1. C. 1. 1. 1. 1. 1. ile son Besser Ale ; . ••• . . . ; ٠, . •

# Statistical Analysis of Commenters on Part 61

Type of Commenter/Commenters/Docket Number

- 1. States New Mexico (4) & (45) Nevada (14) PA (16) Ohio EPA (38) Arizona (47)
  - KY Legislative Research Comm (55) Hawaii (65) SD (69)
- 2. Utilities Florida Power & Light (31) Wisc Electric (32) & (98) Alabama Power (33) Commonwealth Edison (35) Duke Power (48) Ontario Hydro (51)
- 3. Industry Homestake Mining (1) Proctor & Gamble (6) Bethlehem Steel (7) Exxon (15) Dow (17) & (83) Kerr-McGee (22) & (115) Packard (25) Union Carbide (39) Pathfinder Mines (40) Breeder Reactor Corp. (56) R/A Services Inc. (57) Union Oil of California (66) Stock Equipment (67) Werner & Pfleiderer (82), (124) Nuclear Monitoring System (86) General Electric (89) Atomic Industrial Forum (100) NEN (110) General Research Corporation (123)

SC (79) California (93) Power Authority of NY (92) Attorney General of NY (99) Conf of Radiation Control Program Directors (103) NC (109)

Colorado (111) Washington (112) Texas (117)

Arkansas Power & Light (52) & (94) Utility NWMG (81) Middle South Services (84) NE Utilities (85) Florida Power (91) Carolina Power & Light (106) Tennesse Valley Authority (116) 4. Individuals Lowroski (62) Reynek (2) Lewis (3) Jensen (64) Gorenflo (71) Quigg (13) White (21) Anders (73) Mathews (23) Hadala & Banks (76) Klotz (42) Yuan (77) Jaffer (46) Hubbard (90) Scarpa (50)  $(x,y) \in \mathbb{R}^{n}$ 5. Federal Agencies/Labs EG & G Committee (104) EPA (11), (122) U.S. Department of the Interior (114) NOAA Boulder (12) NASA (26) DOE (28), (119) Los Alamos (43) Dept. of Army (63) Argonne (68), (120), (121) 6. Medical Advanced Medical & Research Center (5) American College of Nuclear Physicians (53) U of Texas Medical Branch (75) U of Texas Cancer Center (105) 7. University U of California (8) GA Tech (70) Northern Ill (27) U of Ariz (78) U of NC (30) Howard (49) 8. Brokers/Disposal firms Nuclear Dignostic Labs (29) & (108) Chem-Nuclear (41) US Ecology (101) 9. Legal groups Environmental Law Project (9) Commonwealth Edison legal group (18) 10. Surety Surety Assn of America (20) Nat'l Assn of Insurance Brokers, Inc. (54) 11. Engineering firms Sargent & Lundy (24) Law Engineering (34) Bechtel (44) Stone & Webster (95) Public interest groups 12. Concerned Scientists (36) Sierra Club (37) Birmingham Audubon (80) Don't Waste Washington (97)

B-v
13. Professional organizations American Nuclear Society (87) Health Physics Society (96) American Institute of Chemical Engineers (102) American Society of Mechanical Engineers (107) & (113)

# 14. Others

.

ACRS (10) Dept of the Environment, London (19) Yakima Indians (74) Township of Lower Alloways Creek (88), (125) Atomic Energy of Canada Limited (118)

# INDEX OF ISSUES

The analysis of the individual issues raised by the commenters is structured, to the extent practical, according to the part of the rule being addressed by the commenter. The issue numbers are assigned based on the part of the rule being addressed, e.g., issues addressing Subpart A, General Provisions, are numbered A-1, A-2, etc. The issue numbers are indicated on the annotated set of comments following the discussion of issues. 

 $10^{-1}$ 

<u>Issue Number</u>	<u>Issue</u>	<u></u> :	 ··· Page

Subpart A

-5 . .

A-1	Scope and applicability of the rule	B-1
A-2	Agreement State compatibility	B-5
A-3	Commencement of construction	B-6
A-4	Commission authority to regulate	°
A-5	Pre-emption of State laws	B-8
A-6	Official interpretations of Part 61	B-8
A-7	Exemptions to the rule	. B−9
A-8	Other than near-surface disposal methods	B-9

· ;

Subpart B

B-1	Contents of application; meaning of "demonstrate"	
	compliance	B-12
B-2	Groundwater as major pathway	B-15
B-3	Time limits on licensing process	B-16
B-4	License conditions and safety-related changes	B-17
B-5	License renewals	B-19
B-6	Closure procedure or burden	B-21
B-7	Length of post closure observation and maintenanc	e
	period	B-22
B-8	License transfer	B-23
B-9	Hearings	B-26
- B-10	Post-operational monitoring	B-27
B-11	License each disposal unit	B-28
B-12	Signing application under oath	B-29
B-13	Number of copies of application and EIS	B-30
Subpart C		

C-1	Systems analysis/performance objectives versus	
• •,	prescriptive requirements 'B-	31
C-2	Need for EPA standard B-	33
C-3	Performance objective for environmental	
	protection B-	34
C-4	Performance objective for intruder protection B-	38

;

C-5		Protection of individuals during operations;	0 40
C-6 C-7		ALARA for the performance objectives Stability of the disposal site after closure Need for additional performance objectives	B-43 B-44 B-46
<u>Subpart D</u> :	§61.50	Disposal Site Suitability for Near-Surface Disposal	B-47
D-50-1 D-50-2 D-50-3 D-50-4		Complexity siting requirement Ground water requirement Surface water drainage and flooding Colocation with hazardous waste disposal facilities	B-47 B-47 B-51
D-50-5 D-50-6 D-50-7 D-50-8 D-50-9		or other nuclear facilities Tectonics Demographic requirements Transportation siting requirement Properties of site soils Time spans for siting and design requirements	B-53 B-54 B-55 B-56 B-56 B-57
<u>Subpart D</u> :	§61.51	Disposal site design for land disposal	B~59
D-51-1 D-51-2 D-51-3 D-51-4		Site design Design life of markers and monument Alternatives for Class A or Classes A and B Intruder barrier engineering	B-59 B-60 B-60 B-61
Subpart D:	§61.52 site clo	Land disposal facility operation and disposal osure	B-62
D-52-1 D-52-2 D-52-3 D-52-4 D-52-5	·	Segregation of wastes Factors in stability of the site Basis for 100 foot buffer zone Ancillary activities at the site Operational requirements on direct gamma levels and miscellaneous comments on operation and	B-62 B-64 B-67 B-68
D-52-6		Closure plan	B-89 B-71
<u>Subpart D</u> :	§61.53	Environmental monitoring	B-73
D-53-1		Environmental monitoring	B-73
<u>Subpart D</u> :	§61.55	Waste classification	B-77
D-55-1 D-55-2 D-55-3 D-55-4 D-55-5 D-55-5 D-55-6 D-55-7 D-55-7		Waste classification-basis for numbers De minimis levels for waste TRU classification (10 nCi/gm limit) Waste classification - Ra-226 Case-by-case approval of disposal of waste in greater than Class C concentrations Compliance with waste classification requirements Averaging concentrations over packages	B-77 B-79 B-82 B-89 B-90 B-93 B-95
D-55-8		requirements Waste classification by total hazard	B-96 B-99

.

•

•

.

.

•

# B-viii

	D-55-10 D-55-11		кadionuclide inventory vs. concentration limits Cost and regulatory burden of classification	B-101
	D-55-12		requirements Waste classification - definitions, clarifications	B-102
	D-55-13		and miscellaneous other comments Stability and disposal requirements for Class A	B-105
	D-55-14	;	waste vs. Class B and C wastes Hold wastes for decay	B-107 B-109
	<u>Subpart D</u> :	§61.56	Waste characteristics	B-110
-	D-56-1 D-56-2 D-56-3		Use of absorbents Chelating agents Definitions of pyrophoric, hazardous, and	,B-110 B-110
	D EC A	. ^	explosive	B-113
	D-56-4		Lardboard containers	-B-114
	D-50-5		Wastes in a gaseous form Deckeding standards	B-112
	D-56-7		Hazardous biological pathogenic and infectious	, D-111
:	0.007		waste treatment	R-118
	D-56-8		Waste stability requirements	B-119
	D-56-9		Five percent deformation limit	B-122
	D-56-10		Void spaces in waste containers	B-123
:	D-56-11		Compressive load requirement	B-124
	D-56-12		High integrity container (HIC)	B-126
••••	D-56-13		Free standing liquids	B-127
•	D-56-14		Consistency with other NRC regulations and	
• _			guides	B-128
1) z	D-20-12		Radiolytic decomposition of waste form	B-129
2	D-56-16 D-56-17		Establishment of technical criteria and standards Enforceability of and compliance with	. <b>B-13</b> 0
•			restrictions on waste form	B-131
	D-56-18	. '	Proposed container design	B-131
•	D-56-19		Polymetric solidification agents	B-132
•	D-56-20		Solidification of ion-exchange media and wet	
14			solids	B-132
	D-56-21		Characteristics of volume reduced wastes	B-133
	<u>Subpart D</u> :	§61.57	Labeling	B-135
	D-57-1		Labeling	B-135
	<u>Subpart D</u> :	§61.59	Institutional requirements	B-137
	D-59-1 D-59-2		State responsibility after license transfer 100 year limit on the institutional control period; permitted activities during the	B-137
	D-59-3		control period Landownership - miscellaneous	B-138 B-141
	<u>Subpart E</u>			B-143
	F-1		Financial assurances	8-143
	Ĕ-2		Optional financial report	B-150

•

•

•

. .

....

.

Subpart F		B-151
F-1 F-2	State and tribal participation Local government role and rights	B-151 B-156
Subpart G	•	B-157
G-1	Records, reports, tests, and inspections	B-157
Part 2	· · · ·	B-160
Part 2-1 Part 2-2	Part 2 - general Issuing licenses	B-160 B-161
<u>10 CFR Part 20</u> : §2	0.311 Transfer for disposal and manifests	B-163
M-1 M-2 M-3 M-4	General comments on §20.311 The quality assurance requirements in §20.311 Burden on small entities Manifest system a burden	B-163 B-168 B-169 B-170
General Comments	· · · · ·	B-173
GEN-1 GEN-2 GEN-3 GEN-4 GEN-5 GEN-6 GEN-7 GEN-8 GEN-9	Absoluteness of criteria Additional regulatory guidance needed Exempt waste in storage Development of new sites <u>De facto</u> disposal sites Need for GEIS Extend the comment period Data base on wastes Hearing transcripts to incorporate for the record	B-173 B-176 B-178 B-179 B-180 B-180 B-181 B-181 B-181 B-182
Editorial comments		B-184
ED-1	Editorial comments	B-184

.

. .

.

- :

B−x<sup>.</sup>

•

# ISSUE A-1

Issue:	Scope and applicability of rule
Commenters:	Homestake Mining Company (1)
	Marvin I. Lewis (3)
	New Mexico Secretary for Health and Environment (4)
	Advisory Committee on Reactor Saféguards (10)
	Nevada Department of Human Resources (14)
	Dow Chemical (17)
	Union of Concerned Scientists (36)
	Pathfinder Mines Corporation (40)
	Bechtel National, Inc. (44)
	Alfonso Scarpa (50)
	Union Oil (66)
ť	South Carolina Department of Health and Environmental
· ·	Control (79)
· · · · ·	Birmingham Audubon Society (80)
	American Nuclear Society (87)
	Larolina Power & Light Company (105)
2. T. 1. 11.	American Society of Mechanical Engineers (107), (113)
	Department of Energy (119)

<u>Rule Citation</u>: §61.1 Purpose and scope §61.2 Definitions

11

<u>Summary of comments</u>: Commenters suggested changes to the scope of the rule to both include and exclude topics and materials. Suggestions for additions to the scope included exempted source material and criminal penalties of up to 25 years for violations. Suggestions for deletion from the scope included mill tailings, fuel cycle wastes, and fissile materials. In addition, one commenter expressed the view that the rule should apply only to near-surface disposal until specific requirements for other methods are formulated. One suggested that above ground engineered disposal be included in the scope.

Several commenters addressed the applicability of the rule to existing sites. One commenter was concerned that the abandoned sites being cleaned up by the Department of Energy (DOE) and existing commercial low-level waste sites were not included. One emphasized the difficulty in developing case-by-case guidance for decommissioning the commercial sites where operations have been terminated. One commenter cited the financial burden on operators of existing sites if the five year postclosure observation and maintenance requirement is imposed. One commenter agreed that existing sites should have little or no difficulty in complying with operational and waste characteristics requirements. One commenter was concerned about migration from the West Valley site. One commenter expressed the view that existing sites should be exempted or granted an interim status in the rule to avoid instant noncompliance. South Carolina noted the need to apply Part 61 requirements on a case specific basis at the Barnwell site.

~ ~ ~

. . .

Commenters also addressed the applicability of the rule to Agreement States and individual licensees. Three commenters recommended that the regulation recognize and clarify that Part 61 does not apply to persons licensed by Agreement States. Three requested changes to §61.1(a) to allow individual licensees to operate a burial site under a Part 61 license and one felt that such individual burials should be governed by requirements as stringent.

Analysis of Comments: The two suggested additions to the scope of the rule were not needed to address the commenters' concerns. The Union Oil Company concern about source material was directed toward § 61.10 language and the need for a de minimis provision. The Company objected to the use of "containing" because the contained activity could be so very small in view of the concentration limits which define source material in 10 CFR Part 40. In Part 40 source material generally means containing more than .05 weight percent of uranium or thorium or a combination of the two. The Company noted that wastes containing traces of material made up of just over .05 percent uranium or thorium would be far less active than most natural ores. Part 61 is intended to permit the disposal of all source material, including low concentrations or exempted amounts. The staff agrees that this is an option needed by licensees. The proposed language in Part 61 would not negate any exemption or relief provided generating licensees in other parts of the Commission's rules or in a license. Part 61 applies only to wastes transferred to a land disposal facility. Thus no change was adopted for exempted source material.

The suggestion to include criminal penalties is addressed by the authority citations following the table of contents. The Commission imposes civil penalties but refers cases involving potential criminal acts to the Department of Justice for action. The authority citation provides the legal basis for Justice cases, and each of the sections listed in the second paragraph may involve criminal penalties if violated. Specifying specific criminal penalties is beyond Commission authority.

Only clarifying changes were needed to address most of the concerns of the commenters suggesting deletions from the scope of the rule. The disposal of uranium mill tailings and related source material by a licensee operating under a Part 40 license is currently regulated under the terms of 10 CFR Parts 20 and 40. No change in this situation was intended. Language to emphasize the exemption of all but small quantities of mill tailings was added to the final rule. Waste disposal by individual licensees is currently licensed under the terms of Part 20 and Parts 30, 40, and/or 70 and no change was intended. Fuel cycle waste transferred for disposal at a land disposal facility would have to meet the requirements in Part 61 as required by the new §20.311 of 10 CFR Part 20. Fuel cycle waste disposed of by other methods would not be affected and no change to the rule is needed.

The Birmingham Audubon's concerns over criticality from fissile materials are understandable but from a practical point of view, criticality limitations for special nuclear materials have never been a serious problem at the sites. The criticality potential is evaluated but has little, if any, impact on waste limits or emplacement. The materials in question are primarily enriched uranium and plutonium. The rule establishes limits in the classification scheme on transuranics, hence plutonium, that preclude any criticality concerns. Enriched uranium is not a factor in the classification scheme in § 61.55 because of its relatively low radiological hazard. Thus the staff sees no reason to further restrict special nuclear materials.

••••

The commenter who expressed the view that the rule should apply only to nearsurface disposal is correct that Part 61 applies to other than near-surface disposal. The performance objectives and the procedural (licenses), institutional, financial, state and tribal participation, and records, reports test, and inspections subparts apply equally to all land disposal of low level wastes. The technical requirements are specified only for near-surface disposal, but sections are reserved for other than near-surface methods. Above ground engineered disposal falls in this other category and the staff continues to believe that engineered disposal above ground is sufficiently unique that the specific technical requirement should not be mandatory. The staff considered whether the regulation should cover only near-surface disposal when the rule was proposed. Based on staff judgement and staff experience providing technical assistance to Kansas on a mined cavity application, the staff believed that applications for other disposal methods were credible and that the regulation could be framed to provide the needed flexibility. The comments offered no specific problems with the scope as proposed. Staff still believes the flexibility wise. Alternative disposal methods are further discussed under Issue A-8.

Applicability of the rule to existing sites is a complex issue. The application of the requirements in the rule to existing sites was intended to be a case-bycase determination. The regulation was modified to clarify the applicability to existing sites and address concerns for instant noncompliance.

Lewis expressed concern that the abandoned sites being cleaned up by DOE are not covered by Part 61. Commission authority to regulate Department of Energy waste management activities is limited to the activities covered by 10 CFR Part 60, "Disposal of High-Level Radioactive Wastes in Geologic Repositories". Therefore, Part 61 cannot be modified in scope to cover DOE cleanup activities.

The staff recognizes that problems exist at existing sites. The experiences at these sites provided part of the basis for the proposed rule in a lessons learned sense. The siting, design, operational, financial, and licensing procedures are intended to collectively prevent past problems at future sites.

Part of the confusion concerning the applicability of the rule to Agreement States resulted from language in the Supplementary Information portion of the Notice of Proposed Rulemaking for Part 61. Section V.H. of the Supplementary Information states: "Although nearly all disposal at existing facilities is carried out under State licenses, it would be the Commission's intent that in the future all disposal would be expected to comply with the provisions of Part 61." Section 61.1(b) as proposed referred to Part 150 in outlining exceptions. The intent was not to imply that the rule itself applied to persons licensed by Agreement States. The commenters are correct in noting that Agreement State licensees. Agreement States must have programs compatible with the Commission, and the assumption concerning future disposal was based on the expectation that the manifest and disposal requirements would be implemented by the States. Section 61.1(b) was modified to clarify the applicability in Agreement States. The requests to extend the scope of Part 61 to include disposal by individual licensees was apparently based on the belief that severe limits on burial quantities are imposed on individual licensees by 10 CFR Part 20. The deleted §20.304 did limit the quantities that could be buried without prior Commission approval. Disposal of large quantities was authorized pursuant §20.302. All individual licensee burials are currently licensed pursuant to §20.302. Section 20.302 does not impose specific limits on the types or quantities of waste which may be authorized for disposal. Limits are established on a case-by-case basis. The Part 61 and accompanying rule changes leaves the quantity flexibility in §20.302. Thus the basis for the request appears to have been a misunderstanding.

Licensee burials of their own wastes are approved on a case-by-case base as noted in the preceeding paragraph. The performance objectives and other requirements in Part 61 will certainly be considered in the evaluation of such proposals. However due to the wide variety of possible proposals, mandatory compliance was not proposed. In general, the hazard and quantities of wastes will be significantly less so that siting and design features needed for multiple user sites may not be required. The flexibility to make these judgements without granting exemptions to a regulation was retained.

### Rule Changes:

1. Amend 61.1(b) to read:

(b) Except as provided in Part 150 of this chapter, which deals with an assumption of certain regulatory authority by Agreement States, and  $\S61.6$  "Exemptions," the regulations in this part apply to all persons in the United States. The regulations in this part do not apply to (1) disposal of high-level waste as provided for in Part 60 of this chapter; (2) disposal of uranium or thorium tailings or wastes (byproduct material as defined in § 40.4(a-1)) as provided for in Part 40 of this chapter; or (3) disposal of licensed material as provided for in Part 20.

2. Amend 61.2:

"Waste", for purposes of this part, means those low-level radioactive wastes containing source, special nuclear, or byproduct material that are acceptable for disposal in a land disposal facility. For the purposes of this definition, low-level waste has the same meaning as in the Low-Level Waste Policy Act, that is radioactive waste not classified as high-level radioactive waste, transuranic waste, spent nuclear fuel, or byproduct material as defined in section 11 e.(2) of the Atomic Energy Act (uranium or thorium tailings and waste).

3. Add to §61.1(a):

Applicability of the requirements of this regulation to Commission licenses in effect on the effective date of this part will be determined on a case-by-case basis and implemented through the terms and conditions of the license or orders issued by the Commission.

### ISSUE A-2

Issue: Agreement State compatibility

<u>Commenter</u>: Pennsylvania Department of Environmental Resources (16) State of North Carolina (109) U.S. Environmental Protection Agency (122)

Rule Citation: General

64.

<u>Summary of Comments</u>: The Pennsylvania Department of Environmental Resources (Department) expressed the view that the NRC formally specify which portions of Part 61 must be adopted by Agreement States for compability. The State of North Carolina urged maximum flexibility for Agreement States in adopting Part 61 requirements. EPA urged consistency.

<u>Analysis of Comments</u>: The Department expressed concern that states contemplating changing Agreement State status be able to assess the impact of adopting all or part of the rule and that the rule must be compatible with state and compact activities under the LLW Policy Act. The Department was silent on how the NRC should articulate its compability policy.

The State of North Carolina agreed that "most, if not all, of the technical requirements...should be imposed by such [Agreement] states." However the State felt that the administrative and procedural aspects would pose an unnecessary burden on the states. The EPA urged that Agreement States adopt consistent state regulations and procedures to assure consistent handling of low-level wastes.

On January 23, 1981 the NRC published modified criteria concerning Agreement States to provide for States to enter into agreements for low level waste only (46 FR 7540). These criteria are silent on the issue of compatibility with Part 61. (Part 61 was not published as a proposed rule until July 24, 1981.) Draft guidance has been provided to all states on these limited agreements. The Commission lacks the authority to impose Part 61 as minimum national standards and to require identical regulations.

The NRC staff would expect that the technical requirements of Part 61 would form the basis for state regulations. Procedural flexibility is a recognized need for States. However, the conerns reflected in the procedures established in Part 61 should be addressed in the state regulatory program.

The Commission has held meetings with Agreement State authorities to plan for the orderly implementation of the new requirements in Agreement States. Since the provisions in § 20.311 of 10 CFR Part 20 could affect all waste generators, both in Agreement States and states regulated by the Commission, the effective date of this section should be delayed to permit the necessary changes in Agreement State regulations and license conditions of the Agreement Stateregulated disposal facilities. Waste form, waste classification, and manifest requirements would be affected.

Since all other provisions of the proposed rules would pertain only to applicants or licensees for new Commission-licensed disposal facilities, there are no reasons to delay the effective date of these requirements. The Commission is working with the Agreement States to develop model regulations to be adopted by the Agreement States in accordance with their agreements to maintain compatible State regulations.

Proposed Rule Change: None.

# ISSUE A-3

Issue: Commencement of Construction

<u>Commenters</u>: Harry Lawroski (62) Union Oil Company (66) Birmingham Audubon Society (80) Utility Nuclear Waste Management Group (81) Middle South Services (84)

Rule Citation: § 61.3(b), § 70.23(a)(7)

<u>Summary of Comments</u>: Most of the commenters recommended providing more flexibility to commence construction of the disposal site at the applicant's risk. One commenter strongly supported delaying construction.

<u>Analysis of Comments</u>: The language quoted by Lawroski on commencement of construction was taken from 70.23(a)(7) of Part 70. (A proposed change to delete "commercial waste disposal of land burial "from this paragraph was included in the notice as an editorial change since licenses for disposal of wastes from others will be issued pursuant to Part 61 when Part 61 becomes effective not Parts 30, 40, and 70.) The language states that early commencement "shall be grounds for denial." The applicable language in proposed Part 61 is in § 61.3, License required, which states:

(b) Each person shall file an application with the Commission and obtain a license as provided in this part before commencing construction of a land disposal facility. Failure to comply with this requirement may be grounds for denial of a license.

The Part 61 language "may be grounds" does not make denial mandatory. Further, the Commission may grant exemptions to any requirement in Part 61 under § 61.6, Exemptions. Thus flexibility exists without modifying the proposed lauguage.

Union Oil argued that if an applicant has all other required permits except an NRC license, he should be able to construct the facility at his own risk. This flexibility would allow development of the site at the same time associated facilities not requiring approval are developed.

Birmingham Audubon Society strongly supported the requirement for a license before commencing construction.

The Utility Nuclear Waste Management Group expressed concern over the severity of denial and recommended deleting the second sentence in § 61.3(b). As noted above denial is not mandatory.

Middle South Services also recommended more flexibility to construct and deletion of the second sentence of 61.3(b) on denial.

The Commission is concerned that the actual spending of funds for construction or apparent spending or other financial committments not influence the licensing decision and NEPA evaluations. Also, construction of disposal units or modification of the intended buffer zone prior to final review and licensing decision is of particular concern. Further, neither existing site operators nor persons claiming to be potential applicants commented on the provision. Thus the suggested change to rule was not adopted.

Rule Change: None.

# ISSUE A-4

<u>Issue:</u>

. . .

Commission Authority

<u>Commenters</u>: Pennsylvania Department of Environmental Resources (16) Joel Jaffer (46)

U.S. Environmental Protection Agency (122)

Rule Citation: General and 61.7(c)(4)

<u>Summary of Comments</u>: One commenter suggested that NRC should request authority from Congress to regulate commercial low level waste sites if transferred to DOE. One stated "I don't think the Commission has any authority to license the disposal of radioactive wastes which remain toxic for lengths of time, and for proposed purposes and scope, beyond the NRC's authority." The EPA suggested close coordination between agencies for mixtures of wastes containing both hazardous and radioactive materials.

<u>Analysis of Comments</u>: The Pennsylvania Department of Environmental Resources correctly observes that NRC does not have the authority to regulate disposal sites if the Department of Energy (DOE) is to provide institutional control and that such lack accounts for the license termination after closure where DOE is custodial agency as discussed in § 61.7(c)(4). The commenter suggests that authority to license DOE during institutional control be requested from Congress in order to ensure adequate protection of public health and safety.

While the suggestion has merit, as a practical matter the Commission is not aware of plans for DOE custodial care of existing sites or new sites that might be licensed pursuant to Part 61. The question is an academic one now and can be addressed at some future time should such transfer become more likely.

Mr. Jaffer challenged the Commission's authority to license long-lived materials. The Commission clearly does have the authority to license use and disposal of byproduct, source, and special nuclear materials. These materials include long lived radionuclides. The NRC and its predecessor AEC have always licensed disposal as a necessary adjunct to use of materials. The Commission has and does license commercial or other disposal sites involving disposal of wastes from other persons. Proposed Part 61 only codifies and elaborates on how the Commission will continue this activity.

The U.S Environmental Protection Agency noted that the EPA has regulatory responsibility for the disposal of hazardous wastes under the Resource Conservation and Recovery Act (RCRA), that RCRA exempts materials regulated by NRC, and that RCRA does not address hazardous chemicals mixed with radioactive materials. EPA suggested a memorandum of understanding as a vehicle for coordinating the two agencies' programs. NRC agrees that close coordination is needed and intends to pursue the matter with EPA.

# ISSUE A-5

Issue: Pre-emption of State laws

Commenter: Environmental Law Project (9)

Rule Citation: none

<u>Summary of Comment</u>: The commenter recommended that authority for some form of final approval of a license by a local or preferably state government be included in the regulation.

<u>Analysis of Comment</u>: The commenter is concerned that Commission authority is pre-emptive over State laws and that the Commission might issue a license in spite of state wishes and in spite of state plans to meet its responsibilities under the Low-Level Radioactive Waste Policy Act.

Under the Policy Act, it is quite clear that the Commission is to license sites in non-Agreement States. Under the Commission's licensing authority, no final decision is vested in local governments and there is no provision for State veto. Issuing the license is a decision the Commission must make. This responsibility does not mean that State concerns will not be considered. States have several avenues for expressing concerns and can exercise a <u>de facto</u> veto by not accepting the role of landlord and long term custodian.

Rule Change: none

### ISSUE A-6

Issue: Official interpretations

<u>Commenters</u>: Union Oil Company of California (66) U.S. Ecology (101)

Rule Citation: § 61.5

<u>Summary of Comments</u>: The commenters were concerned about the restrictions on interpreting the regulation articulated in § 61.5.

<u>Analysis of Comments</u>: U.S. Ecology and Union Oil expressed concern that the requirement in § 61.5 that only written interpretations by the General Counsel are binding could unnecessarily delay the regulatory process. U.S. Ecology suggested a 10 day time limit for obtaining General Counsel responses.

The language in § 61.5 is a statement of legal fact whether restated in Part 61 or not. The need for official interpretation is rare and staff routinely explains or clarifies the intent of rules through regulatory guides, technical positions, correspondence etc. Section 61.5 does not restrict these activities.

Because of the rarety of such interpretations, it is impossible to predict the nature or subject of them or the time that may be required to develop an official written response. Thus, the time limit suggested was not adopted.

Rule Change: none

### ISSUE A-7

Issue:

Exemptions to the rule

**Commenters:** 

# Birmingham Audubon Society (80) State of California (93) U.S. Department of Energy (119)

Rule Citation: §§ 61.6, 61.7(b)(5), 61.54, 61.55

Summary of Comments: The Birmingham Audubon Society expressed opposition to case-by-case exceptions on waste concentrations that exceed Table 1 values and felt that general exemptions under § 61.6 should involve extensive noticing and hearings. The State of California was opposed to the flexibility to approve alternative requirements on design and operations in § 61.54 and DOE offered explicit support.

Analysis of Comments: The Audubon Society was concerned that wastes remaining dangerous longer than 500 years should not be allowed in near-surface facilities. The staff shares the Society's concerns but believes that special, and usually more stringent, requirements on waste form and disposal methods can mitigate the potential danger. This issue is discussed more fully in the waste classification issues. The intent for more restrictive requirements versus generally acceptable was clarified in the final rule but the flexibility to consider wastes on a case-by-case basis was retained.

The Society recommended that the Governor and State legislative be notified if any exemptions are granted under § 61.6 and that public hearings be held in the nearest large city. Staff agrees that exemptions should be carefully considered and will do so but does not believe that procedural changes are needed to assure that officials and the public are informed. If the exemption is requested in the initial application, the required distribution assures notification. If a major health and safety issue is involved after the license is issued, § 61.25 assures notification. 

The State of California expressed the view that approval of alternative design and operations pursuant to § 61.54 was arbitrary. Since the section includes a requirement that the alternate proposals meet the performance objectives, the same degree of safety is assured and staff does not agree that flexibility to provide unique or innovative solutions is arbitrary. Rule Changes: none

ISSUE A-8

Issue:

Other than near-surface disposal methods 网络经济性

A :

· · · ·

<u>Commenters</u>: Advisory Committee on Reactor Safeguards (ACRS) (10) Department of the Environment, London (19) Northern Illinois University (27) Zelia M. Jensen (64) Nuclear Monitoring Systems & Management Corporation (86) U.S. Department of the Interior (114) Kerr McGee (115) U.S. Department of Energy (119) U.S. Environmental Protection Agency (122)

## Rule Citation: General

<u>Summary of Comments</u>: The ACRS, Department of the Environment, Kerr-McGee, DOE, and EPA supported the development of criteria and flexibility for disposal methods other than near-surface for more hazardous wastes. Northern Illinois University and Zelia Jensen were opposed to near-surface disposl methods and favored alternate methods for all wastes. Nuclear Monitoring suggested use of the Corporation's specific retrievable storage system for more hazardous waste. The U.S. DOI questioned what would be done with wastes exceeding Class C limits.

<u>Analysis of Comments</u>: The ACRS offered general support for the "establishment of criteria for deeper land burial and disposal in mined cavities." The staff agrees and notes that while the performance objectives, institutional, financial, and procedural requirements apply to any form of land disposal, the specific technical requirements developed so far cover only near-suface disposal and that staff plans that future additions to Part 61 will specify technical criteria for other types of land disposal, such as the use of deep mined cavities, if necessary.

The Department of the Environment supported the flexibility of the systems approach to allow the combination of factors in disposal to determine the disposal methods based on the nature of the wastes. The Department also supported the concept of a range of disposal methods including existing cavities and intermediate depth disposal. Kerr-McGee also supported the development of requirements for other the near-surface disposal, particularly for the disposal of transuranic wastes from decommissioning facilities. Such disposal would be more economic than in a Part 60 high level waste geologic repository.

The DOE supported the concept of alternatives for more hazardous wastes but expressed concern that separate facilities may not be necessary. The DOE noted that other factors in the method of disposal, such as waste form, may provide the greater confinement needed. The staff agrees and did not intend to limit additional assurances to depth of burial only. Such requirements are similar to the considerations for protecting near-surface Class C wastes from intruders. The proposed rule provided depth or other means such as engineered barriers for Class C wastes. The case-by-case considerations provided for when concentrations exceed § 61.55 limits for Class C wastes would also involve a range of factors in providing additional protection, not just depth. Separate facilities were certainly not intended but no change to the rule is needed to allow other than near surface disposal at a near-surface facility.

The EPA urged analysis for other disposal methods such as hydrofracture, deep well injection, and mined cavities but stated that Part 61 should not be delayed for these analyses. Staff agrees with this EPA view. The Northern Illinois University and Jensen were opposed to near-surface disposal for radioactive wastes. The University referenced experience at existing sites and uncertainties in understanding the mechanisms for migration as the basis for dismissing near-surface disposal in favor of disposal in "deep mined cavities in areas of low precipitation." Jensen felt geologic repository disposal of all wastes would solve the problem of waste disposal and not leave it to "future generations to resolve." The staff disagrees with these two commenters and believes that wastes can be safely disposed of in near-surface facilities with reasonable assurance of protection of the public health and safety and with minimum burden on future generations.

The Nuclear Monitoring Systems and Managemet Corporation has developed a specific technology "which is capable of Storing, Monitoring, and Recovering [SMR] both high-level and low level radioactive wastes." A copy of a document prepared for the State of Texas was submitted for the Commission's information. The commenter supports a systems solution for near-surface land burial and use of their specific SMR technology for wastes not suitable for near-surface burial. Staff did not evaluate the specific SMR technology and has no comment on the merits of the proposal.

The Department of Interior questioned what alternatives exist for waste exceeding Class C limits. Such wastes can (1) be considered on a case-by-case basis for disposal at the near-surface facility with additional measures for protection, (2) stored by the licensee until alternate disposal methods are developed, or (3) stored by another licensee or DOE until alternate methods are available.

### Rule Changes: none

• •

17

# ISSUE B-1

Issue:

. .

Contents of applications; meaning of "demonstrate" compliance

**Commenters:** 

ters: Union Oil Company of California (66)
Birmingham Audubon Society (80)
Utility Nuclear Waste Management Group (81)
American Nuclear Society (87)
Health Physics Society (96)
State of New York, Department of Law (99)
Atomic Industrial Forum (100)
U.S. Ecology (101)
Conference of Radiation Control Program Directors (103)
The American Society of Mechanical Engineers (107) & (113)
Department of Energy (119)

Rule Citations: §§ 61.11, 61.12, 61.13, 61.23

<u>Summary of Comments</u>: Three commenters suggested adding more prescriptive requirements. Two commenters suggested clarification or definition of terms. Five commenters addressed the intent and meaning of the requirement to "demonstrate" compliance with objectives and requirements. One suggested addressing accidents in design plans and one questioned whether the experience requirement would lead to monopolies by present companies.

Analysis of Comments: The Union Oil Company of California raised three points about § 61.12. One was the need to define "vicinity" in paragraph (a) to limit the bounds of surveys for data since adjacent properties may not always be accessible. The need for data on adjacent properties will be very site specific and depend on features such as the complexity of the subsurface and ground water flow paths. The characterization of sites will be addressed in regulatory auides and other guidance. The general reference is needed to make it clear that data needs may not be limited just to the disposal areas. A second point questioned the need for information on auxilliary buildings not an integral part of the actual disposal process as proposed paragraph (f) would require by a literal interpretation of "land disposal facility." The intent was not to require detailed architectural drawings of support facilities or auxilliary Staff agrees that the emphasis and key is the relavance to disposal buildinas. operations. General descriptions and layouts for auxilliary features would be Clarifying the reference to methods of construction to focus on sufficient. disposal units should help (See rule change 1). The third point suggested limiting the requirements in (h) to "known" natural resources to avoid the implication that extensive exploration is required. Staff agrees and the suggestion was adopted.

The Birmingham Audubon Society suggested that additional specificity be provided for 61.11(b)(2) (specific personnel qualifications), 61.12(b) (minimum standards for design features), 61.12(1) (minimum corrective measures for migration), and 61.13(b) (minimum standards for adequate intruder barriers). The commenter's suggestion would make the rule very prescriptive in nature.

Technical requirements for design features for near surface disposal are addressed in Subpart D, § 61.51 so further modification of 61.12(b) is not needed to address the Society's comment. Personnel qualifications needed will vary depending on the types and quantities of wastes and on the design and operation of the facility. Expertise is provided by employees and consultants. Staff prefers to address this question in regulatory guides where options and flexibility prevail. Minimum corrective measures warrented will not only be site specific, they may be different for each incident. A general requirement in § 61.53(d) on environmental monitoring keys actions to assuring that performance objectives will be met. Case-by-case flexibility must be preserved. Applicants can propose a variety of intruder barriers. Generic standards for reducing intruder risks or for discouraging intruders involve factors that are too subjective and staff prefers to provide general guidance in supporting guides.

The Utility Waste Management Group comments focused on the use of the terms "demonstrate" and "demonstration." The commenter refers to use of the terms in § 61.13(b) and 61.23(f), (i) and (j). The commenter suggested clarifying what is meant by demonstration by specifying that analysis resulting in reasonable assurance is intended. Adoption of language similiar to proposed <!: ` § 60.101(a)(2) on findings was suggested. Section 60.101(a)(2) states:

(2) While these performance objectives and criteria are generally stated in unqualified terms, it is not expected that complete assurance that they will be met can be presented. A reasonable assurance, on the basis of the record before the Commission, that the objectives and criteria will be met is the general standard that is required. For § 60.111, and other portions of this subpart that impose objectives and criteria for repository performance over long times into the future, there will inevitably be greater uncertainties. Proof of the future performance of engineered systems and geologic media over time periods of a thousand or many thousands of years is not to be had in the ordinary sense of the word. For such long-term objectives and criteria, what is required is reasonable assurance, making allowance for the time period and hazards involved, that the outcome will be in conformance with those objectives and criteria. • •• •

. 4. <sup>1</sup>. .

11 11 1.

The commenter's point is somewhat related to "absoluteness" (see Issue GEN-1) and to what "demonstrate" means. Staff agrees that absolute guarentees are not possible and were not intended by the use of demonstrate.

In § 61.23 "Standards for issuance of a license," the "reasonable assurance" criterion is included in findings for the four performance objectives ((b)-(e)). The commenter correctly notes that the same language should be reflected for all findings. The commenter's suggestion to add the "reasonable assurance" standard to § 61.13 was adopted. The Atomic Industrial Forum and the American Society of Mechanical Engineers expressed similiar views.

The Health Physics Society strongly supported the requirements for specifying a radiation protection program and environmental program. The Society also stressed that the environmental program and occupational exposures should be consistent with ALARA. (See changes to Subpart C made because of other comments on ALARA.)

The Conference of Radiation Control Program Directors suggested revision of § 61.12(d) concerning design basis natural events or phenomena to add consideration of the maximum credible accident anticipated and planned response to such accidents. The response from a radiation safety point of view is covered by § 61.12(k). Section 61.12(d) requires identification of the natural phenomena or events used as a design basis, which includes the maximum credible event concept.

The American Society of Mechanical Engineers also addressed two additional points. One is the need to clarify 61.11(c)(4) to indicate the time periods to address when describing plans for use of the facility for purposes other than disposal of radioactive wastes. The intent was to focus primarily during operations to identify activities that might impact disposal. However, any plans that might impact disposal, monitoring, etc. should be addressed. Plans prior to operations would be addressed in the environmental report and considered as part of the overall impacts of the proposed action. Plans following operations would be addressed in the closure plan and in the application for license transfer to the custodial agency. No change to the rule was made since potentially all time periods are relevant and none could be excluded. The second point suggested adding prescriptive requirements for the quality assurance program requirement in 61.12(j). As discussed earlier and in other issues, the rule focuses on the performance objectives and general requirements and includes only limited prescriptive details. The flexibility thus provided has received broad support. Guidance will be provided in regulatory guides and no rule change was adopted.

The Department of Energy expressed concern that the requirement for being qualified by training and experience could lead to monopolies by present companies and suggested replacing "and" with "or." Such was not the intent. For example, experience handling comparable quantities of radioactive materials, experience constructing similiar facilities, experience disposing of hazardous wastes could all be used to demonstrate that the applicant is qualified. The change was not adopted. The State of New York felt that past experience was not emphasized enough.

## Rule Changes:

•••

- 1. Change 61.12(f) to add after include: ", but not be limited to," and after construction: "of disposal units"
- 2. Amend § 61.23(f)-(j) to read as follows:

(f) The applicant's demonstration provides reasonable assurance that the applicable technical requirements of Subpart D of this part will be met.

- 11 - 2

(g) The applicant's proposal for institutional control provides reasonable assurance that such care will be provided for the length of time found necessary to ensure the findings in paragraphs (b)-(e) of this section and that the institutional control meets the requirements of §61.59.

(h) The information on financial assurances meets the requirements of Subpart E of this part.

(i) The applicant's physical security information provides reasonable assurance that the requirements of Part 73 of this chapter will be met, insofar as they are applicable to special nuclear material to be possessed before disposal under the license.

(j) The applicant's criticality accident information provides reasonable assurance that the requirements of §70.24 of Part 70 of this chapter will be met, insofar as they are applicable under the license.

3. Change 61.13(b) to add a reference to "reasonable assurance."

4. Amend 61.12(h) to add "known."

# ISSUE B-2

 $(i,j) \in \mathbb{N}^{n-1}$ 

· · · ·

<u>Issue</u>: Groundwater as Major Pathway

<u>Commenters</u>: Department of the Environment, London (19) Utility Nuclear Waste Management Group (81)

Category:

Rule Citiation: Secton 61.13 Technical Analysis, paragraph (a)

<u>Summary of Comments</u>: The Department of the Environment commented that paragraph 61.13(a) states that ground water is probably the most significant pathway for radioactive transport back to man and requires particular study. They state that although transport is very significant, recent sensitivity analyses suggest that other pathways such as human intrusion, erosion and vegetation uptake may be more significant. The UNWMG recommended deleting the reference to the groundwater pathway as unnecessary.

<u>Analysis of Comments</u>: NRC recognizes that other pathways of exposure, particularly intrusion events by man into the disposal facility, will result in significantly higher exposures to the individuals involved than the ground water pathway. Such events may also bring disposed wastes to the surface where they may be transported off-site by surface water or incorporated into crops grown at the site. Intrusion into a disposal facility is relatively non site specific and NRC has incorporated specific requirements in Part 61 which, if met by a licensee, should assure that an inadvertent intruder would not receive an unreasonable exposure (i.e., greater than 500 mrem to the whole body) assuming reasonable uses of the land. The requirements principally involve the waste classification requirements which include requirements to bury waste presenting high potential hazard to an inadvertent intruder at deeper depths or with additional intruder barriers. The type of pathways analyzed in developing the requirements included construction of house at the site, living in the house constructed and consuming food grown at the site.

The release of radioactivity to the environment through air, soil, ground water, surface water, plant uptake and burrowing animal pathways is relatively site specific and depends primarily on the natural characteristics of the site and its environs. Of these, NRC considers ground water transport to be of greatest potential significance in comparison to the potential releases to the environment which may occur through the other pathways. In addition, requirements in Part 61 and controls instituted at specific sites directed at controlling migration will also reduce the potential for releases through the other pathways (e.g., thicker, denser trench caps). As noted above, intrusion into the site including the cultivation and consumption of food crops at the site will generally yield higher exposures and has also been addressed in Part 61.

The Utility Nuclear Waste Management Group noted that the reference to the ground water pathway was gratutious and recommended deletion. The staff agrees and has removed the reference.

<u>Rule Change</u>: Delete from 61.13(a): "For near-surface disposal, the ground water pathway will generally be the most significant in terms of releases of radioactivity."

### ISSUE B-3

Issue: Time limits on licensing process

. . .

<u>Commenters</u>: Ohio EPA (38) Union Carbide (39) University of Texas Medical Branch (75) Health Physics Society (96) Atomic Industrial Forum (100)

Rule Citation: Section G of Supplementary Information and Subpart B

<u>Summary of Comments</u>: Most of the commenters recommended specifying periods of time in the rule for parts of the licensing process. One commenter suggested a specific licensing step for unrestricted use following institutional control.

Analysis of Comments: The Ohio EPA noted that the only time period listed for Commission action is a reference in the Supplementary Information to the statement that completeness review of tendered applications generally will be made within 30 days. No other time periods are listed in the notice and none in the rule. Union Carbide expressed concern about meeting the January 1986 deadline in the LLW Policy Act and the need for specifying definite times to assure that compacts can meet the deadline. (The January 1986 date is the time compacts can begin excluding non-compact wastes.) The University Texas Medical Branch recommended a time limit to approve or deny the application based on financial considerations and noted that "Applicants who have options on land or own land that could be used for other purposes may incur substantial financial losses due to unreasonable delays." The Medical Branch also suggested that applicants be compensated for delays. The Health Physics Society recommended limits on hearings.

The Commission is sympathetic to the problem of uncertain time frames in the licensing process, the potential for delays, and the need for expedicious action. However, the review and licensing process involves a number of uncertain or potentially uncertain time periods such as the quality of the application, extent of supplemental tests or data that may be required of applicants, State or tribal participation schedules, Commission use of contractors to aid a specific timeframe that might curtail other rights does not seem justified. The Commission is building in-house expertise and computer capabilities to minimize the use of contractors to minimize delays and uncertainties in this area and will expedite its review to the extent possible. Schedules will be developed for all parties on a case-by-case basis for each application. Issues related to compensation must be addressed through-the courts or other means and are inappropriate for a rule such as Part 61

\* \* \* \* \* \* \* \* \* \* \* \* \* \* \* \* \* \*

The Atomic Industrial Forum suggested a sixth phase for the life cycle of a disposal facility - release for unrestricted or uncontrolled use after institutional control. The commenter noted that such release would result from radioactive decay to safe levels if sufficient controls on the non-radiological properties are imposed. The rule as proposed does provide for a discrete period of institutional control and license termination at the end of the control period. The proposed rule is silent on what the landowner may do with the land at this point. Reliance on passive controls such as land records and intruder barriers continues after the end of the active institutional control period and, for purposes of analysis, are assumed to cease functioning after 500 years for nearsurface disposal. The landowner could allow reuse of the surface after license termination and even limited use of the surface during institutional control. The commenter suggested that some of the phases - preoperational, operational, closure, post closure observation, and institutional control - may take place simultaneously. Closure activities will occur during operations and no other overlaps should occur so no clarification seem warrented. . .

Rule Change: None.

: •

# ISSUE B-4

<u>Issue</u> :	License conditions and safety related changes
<u>Commenters</u> :	Bechtel (44) Duke Power (48) Utility Nuclear Waste Management Group (81) Middle South Services (84) American Nuclear Society (87) General Electric (89) Stone and Webster Engineering Corporation (95) Atomic Industrial Forum (100) U.S. Ecology (101)
	The American Society of Mechanical Engineers (107)

Rule Citations: §§ 61.24(g) and (h) and 61.25(a)

<u>Summary of Comments</u>: Three commenters suggested deleting § 61.24(h) which allows the Commission to add additional requirements into license conditions. Nine commenters (all but Stone and Webster) suggested modification of 61.25(a) to allow nonsafety changes. Stone and Webster suggested alternate language for § 61.24(g) to provide for license certifications instead of Commission inspection prior to beginning disposal.

<u>Analysis of Comments</u>: The commenters (Bechtel, American Nuclear Society, and American Society of Mechanical Engineers) who recommended deletion of 61.24(h) were concerned that the applicant's, or licensee's rights under § 2.105 to petition for a hearing on additional requirements or conditions was being bypassed. No such bypass was intended and the provision in 61.24(h) in no way restricts the rights under § 2.105. No change is necessary.

Stone and Webster Engineering Corporation expressed the view that the Commission should prepare a value/impact analysis in support of any changes to the license made pursuant to 61.24(h). The staff agrees that the impacts of its actions must be evaluated but does not believe that a prescriptive requirement to specify how that evaluation will be documented is appropriate. The nature of the change could range from correcting drafting or typographical errors to major changes requiring an EIS or supplement to the EIS.

The nine commenters who expressed the view that § 61.25 was too restrictive were concerned that minor changes that would not affect the public health and safety are necessary during routine operation of the disposal facility. They were also concerned about changes in administrative or support facilities being restricted. Several recommended the following addition to § 61.25(a) that would allow changes without prior notification:

# (4) changes that do not impact public health and safety can be made immediately with subsequent notification of the Commission in a timely manner.

U.S. Ecology, a current site operator, indicated that notification of the Commission prior to implementation was acceptable.

The commenters' concerns can be addressed through the wording in the specific license conditions in the license issued for each land disposal facility. The issue really boils down to whether the Commission should be aware of all minor changes not requiring Commission approval 60 days before implementing (as proposed), at the time of implementation, or after implementing, or whether a further hierarchy of conditions or other clarification is warrented. The commenters are correct that the proposed language requires <u>all</u> changes to the facility and procedures to be classified as one of the three categories. The three categories were established based on the relative importance to public health and safety. Many day to day changes may involve matters of no concern from a health and safety prespective as the commenters noted. Descriptive information on the facility or procedures may have been submitted in the application for illustrative purposes. An editorial change to require the important features to be addressed by conditions failing in the categories but not all will accomplish the same objective as adding a fourth category as suggested by the commenters.

Stone and Webster suggested an elaborate process of certification, notification, and notices on an alternative to § 61.24(g) which states

"No radioactive waste may be disposed of until the Commission has inspected the land disposal facility and has found it to be in conformance with the description, design, and construction described in the application for a license.

The wording of the alternative suggests concern about Commission delays. If budget constraints or other unforeseen problems arise to prevent timely Commission inspection, other arrangements such as State reports or an exemption to

the provision can be made. The staff views this final inspection to be very important and the provisions in § 61.24(g) are intended to be administratively simple: i.e., no notices or other procedural steps are required. A letter or inspection report will meet the condition. Thus, the suggestion was not adopted.

# Rule Changes:

.

1 2 1 4

Amend second sentence of 61.25(a) to read:

"The license will include conditions restricting certain subsequent changes to the facility and the procedures authorized which are important to public health and safety."

. . . . .

# ISSUE B-5 . .

Issue:

.

Commenters:

### License renewals

Pennsylvania Department of Environmental Resources (16) Union Carbide Corporation (39) Duke Power Company (48) Oswald U Anders (73) Utility Nuclear Waste Management Group (81) Middle South Services Inc. (84) Northeast Utilities (85) General Electric (89) General Electric (89) Stone and Webster Engineering Corporation (95) Don't Waste Washington Legal Defense Foundation (97) Atomic Industrial Forum (100) Carolina, Power and Light Company (106)

The American Society of Mechanical Engineers (107) North Carolina Radiation Protection Commission (109) New England Nuclear (110)

Department of Energy (119)

Rule Citations: Summary Section V.G. § 61.27 Application for renewal or closure; 61.25(b); 61.7(c)(2)

Sumary of Comments: All of the commenters except Stone and Webster, Don't Waste Washington, and DOE were concerned about the burden, particularly if hearings are involved, of the implied 5 year renewal requirement and recommended life of the facility licenses. Stone and Webster addressed renewal intervals linked to actual receipt of waste and DOE suggested specifying the renewal interval. Don't Waste Washington supported the clear statement of responsibility for buried wastes in § 61.27.

Analysis of Comments: Under section V.G. "Life Cycle of a Typical Land Disposal Facility" of the Supplementary Information portion of the regulation, the operational phase is discussed. The discussion indicated that at intervals specified in the licensee, the license would be renewed following § 61.27. A parenthetical reference to 5 years as the normal term for materials licensees was made and a practice of offering the opportunity for public hearings was mentioned. The offering of hearings would be required for renewals under the provisions of proposed 61.25(b). The renewal interval is not specified in the regulation

but is implied by the parenthetical reference. Section 61.27 outlines procedures for submitting and acting on applications for renewal but does not require renewals or specify intervals. Section 61.7 "Concepts" under (c)(2) also discusses periodic renewals.

The commenters believed that the regulation provides adequate evaluation and control of the licensee and any potential changes without the burden of frequent hearings which might be involved if a 5 year interval is adopted in the license.

The Commission included the renewal provision in the rule to provide an opportunity to review the operating history and determine whether wastes should continue to be received and disposed of at the site. The renewal provides incentive to update the license to reflect developing technology and to fully factor operating experience and new site information and site performance into periodic reassessments. The renewal provides a greater degree of assurance that the licensee and the Commission will perform the reassessment.

The commenters' belief that adequate controls exist independent of the renewal process was based in part on the tight control of changes and opportunities for notices and of opportunities for requesting hearings in § 61.25(a). (Note that 9 commenters believed that proposed § 61.25(a) does not provide reasonable flexibility. See issue B-4 for a discussion of changes made to provide additional flexibility.) Further monitoring of site activities is provided through inspections and reports filed by licensees. Comprehensive annual reports are proposed in § 61.80(h). Annual financial reports are to be submitted under § 61.80(g). An annual financial re-evaluation of closure funding by the Commission is required by § 61.62(c).

A relevant legal point is that licensees or interested parties can request hearings on any Commission licensing action under existing rules. In proposed requirements such as those in § 61.25 the Commission is advertising the opportunity to request hearings and providing opportunity to request hearings before taking action. Any licensing action taken as a result of Commission review of reports or inspections would be subject to requests for hearings.

The staff reconsidered the issue of how to assure periodic reassessments and reduce the potential hearing burden on licensees. Three alternatives to the proposed renewal provisions were considered. One was to add a section to § 61.24 "Conditions of licenses" to require submittal of periodic reassessments or to require a condition in the specific license issued to the applicant for submittal of such periodic reassessments. All references to renewal would be deleted and reliance placed on the new requirement in § 61.24. The second was to expand the scope of reporting requirements in § 61.80 to cover additional areas of concern. All references to renewal would also be deleted. A third was to delete the requirement to automatically notice renewals in § 61.25(b).

The second altenative to rely on reports provides the greatest assurance of reduced opportunity for hearings. No opportunity for a hearing exists unless the Commission issues orders or otherwise amends the license based on evaluation of the reported information. Updates or reassessments required by conditions of licenses could appear to be equivalent to submittal of an application. In such case, the opportunity for hearings could exist. Thus relying on conditions of licenses reduces the assurance of reassessment while not completely assuring no hearings. The third alternative would keep renewals to assure reassessment but publicize the opportunity for hearings only when significant changes to license conditions were made, e.g., when renewal involved changes in any conditions covered by 61.25(a)(1). The need for periodic reassessment has been dramatically proved by experience with existing sites and other materials licensees and the Commission believes it must be assured by renewal.

アドレビスの開始に対

Stone and Webster recommended that the renewal interval begin when actual receipt of waste begins rather than when the license is issued since the applicant does not begin construction until the license is issued. This change was not adopted since the rule does not specify the renewal period (5 years was given as an example only) and such factors can be factored into the expiration date establishing the renewal period. Additionally, construction should only involve a few months.

The Department of Energy suggested that § 61.27(d) or some prior section indicate the amount of time for which a license is issued. Section 61.27 refers to "any" expiration date on the license. As proposed, Part 61 would not have mandated an expiration date but 5 year renewals were referenced as typical in V.G. of the Supplementary Information, as noted earlier. A new paragraph was added to § 61.24 "Conditions of licenses" to establish the requirement for an expiration date in the license. Flexibility to set the date on a case-by-case basis was retained to allow periods longer or shorter than 5 years.

Don't Waste Washington endorsed the language in § 61.27 stating the licensee's continuing responsibility for site closure, postclosure observation, and transfer to the site owner whether the authority to continue to receive waste is renewed or not. Staff agrees and no changes were made to change the language other than grammatical.

### Rule Changes:

1. Add to § 61.24 a new (j):

The authority to continue disposal of wastes shall expire at the date stated in the license except as provided in § 61.27(a) of this Part."

2. Delete "license renewal" in § 61.25(b).

### ISSUE B-6

Issue:	Closure	procedure	a	burd	e'n
		•	`	: •	

Commenters:	Ontario Hydró (51)
	Oswald U. Anders (73)
	U.S. Ecology (101)

Rule Citation: §§ 61.25 and 61.27 and 61.7(c)(2)

<u>Summary of Comments</u>: Two commenters suggested not subjecting the licensee to opportunity for hearings at closure and one suggested deleting the requirement for a license amendment to allow closure.

Analysis of Comments: Ontario Hydro expressed the view that hearings before issuing the license should be sufficient and that a negative result from hearings on closure would impose an unfair financial burden on the licensee. Anders expressed a general veiw that the procedures proposed in Part 61 are over-regulation and will lead to perpetual litigation. Closure was singled out as the final blow - the lack of the right to "walk away in frustration." U.S. Ecology noted that closure plans will be reviewed in the initial application and periodically updated and expressed the view that no license amendment should be required.

In response to these comments and the comments concerning the regulatory burden of the renewal provision (Issue B-5), changes to license conditions (Issue B-4), the burden of multiple hearings (Issue B-9), and concerns of States and local citizens and governments (e.g., Issue F-2), the staff re-examined the procedures proposed to assure up-to-date information and periodic reassessment.

For most licensees, closure will be a continuing activity all during operations and final plans will not usually involve major activities. Broad input and a last chance for the most affected persons to agree on the plans for long term care seems prudent and in the best interest for providing assurances to the custodian so that the custodian is willing to make the required upfront committment. The requirement for a closure amendment and noticing closure was retained.

Rule Changes: None.

## ISSUE B-7

Issue:

Length of Post Closure Observation and Maintenance Period

Chem-Nuclear Systems, Inc. (41) Commenters: Bechtel National Inc. (44) American Nuclear Society (87) Atomic Industrial Forum (100) U.S. Ecology (101) The American Society of Mechanical Engineers (107) U.S. Department of Energy (119)

<u>Rule Citation</u>: § 61.7(c)(3) "Concepts" and 61.29 "Post Closure Observation and Maintenance"

Summary of Comments: None of the commenters took exception with the need for a period of post closure observation and maintenance by the licensee. The Atomic Industrial Forum and the American Society of Mechanical Engineers suggested that since the period may need to be extended or possibly shortened based on site specific conditions, it should be included in the site closure plan rather than the rule. U.S. Ecology commented similarly about the need to consider site specific conditions at closure which could warrant a shorter period. They suggested that provision be made for a period of less than 5 years if conditions warrant upon request of the licensee. The Department of Energy questioned the basis for applying the 5 year period in all cases since a shorter or longer time period might be required depending on site specific conditions.

The American Nuclear Society, The American Society of Mechanical Engineers and Bechtel national, Inc. commented that the requirement, as written in requiring a minimum of 5 years was open ended and did not provide sufficient guidance to establish adequate funding for the time required. They suggested that a specific time frame should be set such that the licensee could plan for the time required. Chem-Nuclear Systems commented similarly noting that a clear position on the time frame is necessary for an operator to set aside funds to cover those activities.

Analysis of Comments: The comments do not take exception with the need to observe and maintain the site after site closure to help ensure it is in a stable condition prior to transfer of control to the site owner. They do take exception, however, with the fact that specifying a minimum of 5 years results in an open ended requirement which can affect a licensees planning activities and also the fact that site conditions may warrant a shorter or longer period of observation and maintenance. NRC did not intent to establish an open ended requirement and believed that by specifying a minimum of 5 years, a specific limited time frame would be established which could be extended, if needed, based on site specific conditions. NRC staff agrees that the length of time required will be a function of site specific conditions at closure and the post operating history of the site.

Rule Change: Based on the staff's analysis of comments, § 61.29 has been modified to read as follows:

§ 61.29 Post-closure observation and maintenance.

Following completion of closure authorized in § 61.28, the licensee shall observe, monitor, and carry out necessary maintenance and repairs at the disposal site until the site closure is complete and the license is transferred by the Commission in accordance with § 61.30. Responsibility for the disposal site must be maintained by the licensee for 5 years. A shorter or longer time period for post closure observation and maintenance may be established and approved by the Commisson as a part of the site closure plan based on site specific conditions. 5.4.4 1.5.200 - 5.4.5.4

### ISSUE B-8

# Issue: License transfer

<u>Commenters</u>:

::

· · ·

. .

Environmental Law Project (9) PA Department of Environmental Resources (16) Chem-Nuclear Systems, Inc. (41) Arizona State Clearinghouse (47) State of California (93) Atomic Industrial Forum (100) U.S. Ecology (101) U.S. Ecology (101) Conference of Radiation Control Program Directors (103) U.S. Department of Energy (119) General Research (123)

· · · · ·

· · · ·

Rule Citations: § 61.23 (g), 61.30, and related citations 61.24(a), 61.26.

<u>Summary of Comments</u>: One commenter was concerned that adequate safeguards are specified in the event of transfer of the license from one operator to another. Two commenters were concerned about requirements for specific plans and assurances related to transfer of the license to the custodial agency and three were concerned about delays in accepting transfer by the custodial agency. The Conference raised several questions concerning the concept of license transfer to a state or federal agency after closure. They questioned what criteria would be used to judge the agency, what enforcement actions might be taken, and why a state might be licensed for custodial care but not DOE. The Conference suggested considering termination after closure. Pennsylvania questioned whether the rule would allow the operator to remain the licensee for long term care if so desired by the site owner. 'California expressed concern about a state assuming responsibility without adequate funds. (See Issue E-1). The DOE suggested that custodial license conditions be added to the rule in § 61.31.

<u>Analysis of Comments</u>: The commenter concerned about license transfer to another operator was primarily concerned about transferees being subject to all requirements that the initial licensee was. The commenter wanted more explicit assurance. The proposed rule requires a license amendment for transfer of the license in § 61.24(a), "Conditions of licenses." A general requirements for issuing amendments is in § 61.26(b) which says that the Commission will apply the criteria set forth in § 61.23. Section 61.23 specifies the standards for issuing the initial license. Thus the proposed rule does require equivalent assurances.

Chem-Nuclear Systems, Inc. who questioned whether a specific plan for institutional care is required and how an applicant can assure compliance with such a plan raised good points. Section 61.59(a) outlines certain minimum specific activities which the custodian must perform. The amount of detail required concerning activities during institutional control will be dictated by two factors. One is the degree of reliance on the activities to assure protection of the public health and safety in the technical analyses. The second is financial planning. Section 61.63 requires Commission approval of the arrangements and adequacy of financial assurances for institutional control. Costs must be estimated for institutional care to determine adequacy. Thus some preliminary plans would be required. Finalization of such plans would be part of the application for amendment to transfer to the site owner pursuant to § 61.30. The staff does not agree with DOE that license conditions for the custodial agency should be added. The conditions will be site specific, and flexibility is needed. Detailed conditions were not added.

The applicant cannot assure the actions of the site owner. The applicant should demonstrate that close coordination has occurred with the Federal or State government who owns or will assume ownership of the disposal site. The certification required in § 61.14 is intended to be "informed" consent and agreement. The applicant can only be expected to provide the basis for informed certification and financial planning and evidence of the arrangement. Just as the Commission determines that the applicant's proposals provide reasonable assurances, the applicant is demonstrating the custodial committment provides reasonable assurance that institutional care will occur.

The commenters were concerned about delays in transfer under § 61.30 because the custodial agency might impose more stringent requirements or use "when the agency is prepared" to effect delays. Such delays could deplete financial reserves set aside. The commenters (U.S. Ecology and Chem-Nuclear Systems, Inc. the two companies currently operating sites and the Atomic Industrial Forum) correctly identify a weakness in the proposed licensing procedures. The dilemma the Commission faces is how to assure that the custodial agency has the commitment, staff, funds, authority, etc. to carryout the institutional care before releasing the site operator without imposing an unfair burden on the site operator as the commenters suggest. In the draft EIS, alternative licensing options were discussed in Chapter 8, including a co-licensee arrangement up front. The co-licensee experience with the West Valley, New York reprocessing facility did not solve this dilemma. None of the commenters offered a solution other than making transfer a "part of the closure agreement and not the regulation." Contingency planning for delays in transfer will have to be a risk associated with doing business unless case-by-case solutions are found. The proposed rule provided maximum flexibility to address the issue on a case-bycase basis and no changes were made.

17.57

نغاد مربعه ور بار تغاد

Two factors important to the transfer to the landowner are the limits on NRC authority and the importance of early and continuing involvement by the landowner. The turnover to the landowner is a contractural arrangement between the operator and owner that NRC cannot mandate. The NRC can object to transfer only. As pointed out in the analysis of comments on § 61.59, since the landowner (usually the State) ultimately is responsible, it behooves the landowner to be involved during the lifetime of site operations. Continuing awareness and involvement should minimixe any last minute problems in transfer.

The Arizona commenter recommended that license transfer occur after operations to receive wastes cease. The commenter is concerned that the lack of economic incentive to carryout closure and postclosure activities could prompt the operational licensee to use legal means to escape responsibility. The commenter suggested that provisions for the landowner to assume responsibility and marked funds and then carryout closure, post closure observation and institutional care should be added as a requirement in the rule to eliminate reliance on the operator.

The rule as proposed has sufficient flexibility to allow a number of options including licensing options. The license can be transferred any time not precluded in the rule. The landowner and custodian can be joint licensees and share or shift responsibilities at any point in the process. The State may be the operator. The Arizona or Pennsylvania scheme--operator as long term care licensee--may happen under the proposed rule but is not mandated. The Pennsylvania proposal has some drawbacks, but if the custodial period is short because of waste restrictions or method of disposal (e.g., sealed mine), it might be preferable. The Commission prefers to keep the flexibility for operator/ custodian roles in view of the LLW Policy Act activities and negotiations.

With respect to the questions raised by the Conference, criteria to be used to judge the State readiness for license transfer will be focused on having staff, authority, and funding in place. Any state can be a licensee - the key is readiness to implement the custodial program. Possible enforcement actions include any the Commission is authorized to take against licensees. The state would qualify as a "person" under Commission rule and would be a licensee in the same status as other licensees. The only reason for licensing a state and not DOE is lack of authority (see Issue A-4). The Commission lacks the authority to license DOE to provide institutional control.

••••

. •

••••

The General Research Corporation expressed concern over the lack of more prescriptive procedures for closure and postclosure activities. The commenter was concerned that end points of these phases are not well defined and unanticipated problems may arise to interfere with orderly transfer for institutional control. The commenter raises some interesting points to factor in specific site planning but staff believes the level of detail and attempts to postulate all possible problem scenarios and include coverage to be too prescriptive and speculative for a rule.

Rule Changes: none

#### ISSUE B-9

<u>Issue</u>: Hearings

<u>Commenters</u>: Environmental Law Project (9) Oswald U. Anders Ph.D. (73) Northeast Utilities (85) Health Physics Society (96)

Rule Citation: § 2.105, V.G. of the Supplementary Information, § 61.25

<u>Summary of Comments</u>: One commenter believed the rule should require that public informational hearings be held near the proposed site. Two commenters believed that the multiple opportunities for hearings are a burden and will virtually eliminate the private sector. One expressed a general concern about the burden of hearings and questioned how the scope might be limited.

<u>Analysis of Comments</u>: The Environmental Law Project is concerned that public hearings be mandatory. The suggested hearings would be in addition to the adversary hearings which can be requested under existing regulations. The Project also believed that residents of the area should be given two months notice for the hearings and a local public document room should be established.

The Commission considered the alternative of mandatory hearings of the "adversary type" as prescribed under existing regulations rather than such hearings being optional. Establishing a new type of hearing and prescribing the conduct of a new type of informational hearing was not considered. Nothing in the proposed rulemaking would prohibit holding such hearings. The applicant, the State, the local government, or the Commission could hold informational hearings.

In view of State activities under the Low-Level Radioactive Waste Policy Act, a Commission requirement for local hearings could be duplicative and cause additional delays. Developing the rules for such hearings would be difficult and would require supplementary proposed regulations. The proposed regulations require notification of local officials. Local officials and citizens can comment on and provide input for the environmental impact statements for the site. Thus the suggestion was not adopted.

The Commission also considered mandatory local public document rooms. Local document rooms have been established for existing Commission disposal site licensees so a requirement would codify current practice. In the DEIS, case-by-case flexibility was indicated as preferable in case the state has made other arrangements or lack of interst or willingness for a local group to accept

responsibility for maintaining the docket files. The Project offered no specific arguments that would negate these views expressed in the DEIS. No change was proposed.

The Northeast Utilities expressed concern that the multitude of public hearings and the associated uncertainties and burden would result in no private entity being willing to develop a site. Northeast Utilities reviewed the five-phases that make up the life cycle of a site and counted seven opportunities for hearings assuming five year renewals and 25 years of operation. (The seven identified would be a minimum number, depending on how many changes require notice and opportunity under § 61.25.) The commenter's concerns are addressed under issues B-5 (renewals) and B-6 (closure) for these parts of the cycle. No issue was taken with opportunity for initial hearings. The remaining opportunity discussed by the commenter is at transfer to the custodian.

Oswald U. Anders expressed similiar concerns about the private sector viability under the proposed regulatory scheme, including multiple opportunities for hearings.

As discussed for renewals (Issue B-5) and closure (Issue B-6), the licensee or their interested party can request hearings on any Commission action. The Commission has a choice in whether to advertise the opportunity and delay action for a specified period to allow requests as in 61.25(a)(1). In response to the concerns raised by commenters, the required number of notices in § 61.25(b) has been reduced by deleting renewals.

Limiting the timing and scope of hearings would involve changing the existing process defined by 10 CFR Part 2. No changes to Part 2 are appropriate for one category of licensee.

Rule Changes: See Issues B-5 and B-6.

### ISSUE B-10

<u>Issues</u>: Post-Operational Monitoring <u>Commenters</u>: Environmental Law Project (9) University of North Carolina (30)

> Commonwealth of Pennsylvania, Department of Environmental Resources (16) Duke Power Company (48)

Rule Citation: § 61.29 and 61.59

<u>Summary of Comments</u>: Five comments were received on this particular issue by the commenters:

- (a) "...the five-year active monitoring phase required by the proposed rule § 61.29 is far too short..." (9)
- (b) "At least ten years of some form of active monitoring seems to be required, and perhaps more..." (9)

- (c) "Some specific provision for additional monitoring if determined necessary by geologic or other conditions would be helpful." (9)
- (d) "It should also be made clear that active controls, especially environmental surveillance, could continue beyond the 100 year time period if the custodial agency so desires." (16)
- (e) The environmental monitoring program should only continue during the five-year post closure period if environmental monitoring is intended to be a period of sampling of wells and other environmental samples for radioactivity. If environmental monitoring is deemed to be required for the period of 100 years, it should be severely limited to occasional samples of the ground water pathway." (48)

<u>Analysis of Comments</u>: Environmental monitoring at proposed near surface disposal facilities under 10 CFR Part 61 is intended to be comprehensive and extend beyond the post-closure observation and maintenance period. § 61.53(a) requires a preoperational monitoring program and § 61.53(b) extends this program throughout the facility construction and operational period. § 61.29 requires that the site licensee maintain complete responsibility for the site for a post-closure observation and maintenance period of 5 years and, as part of this responsibility, the licensee must monitor the site. NRC has some discretion over the length of this post-closure period and may extend it as conditions warrant. With respect to Comments (a), (b) and (c), NRC is not bound to a 5-year monitoring program, but depending upon site characteristics, facility inventory or other factors, will determine the length of required post-operational monitoring on a facility-specific basis. (See change made, Issue B-7.)

The staff disagrees with comment (e). The nature of the monitoring program and extent of sampling will also be depend on site characteristics and will be determined on a site specific basis.

With respect to Comment (d), the staff feels that the language of § 61.59(b) is clear: The Commission will determine the length of the period of active controls. As mentioned above, the length of this period will be determined based on facility-specific characteristics. Although the custodial agency may continue active controls beyond the period required by the Commission, the Commission has in establishing the performance objectives and technical requirements of 10 CFR Part 61, operated on the basis of not assuming reliance on active controls for more than 100 years following transfer of control of the disposal site.

Rule Change: See Issue B-7.

• • •

### ISSUE B-11

<u>Issue:</u> License each disposal unit

Commenter: Arizona State Clearinghouse (47)

Rule Citation: Subpart B

Summary of Comment: Arizona suggested that a licensing approach oriented to individual disposal units be considered.

Analysis of Comment: Under Arizona's proposed scheme, only limited numbers of disposal units would be authorized for use. Use of additional units would not be authorized until filled units were adequately closed and stabilized. Such an approach would guarentee that closure was completed during the operating life of the site. However, it would involve additional burdens for the licensee and the Commission since specific license amendments would be required. These amendments would be an administrative burden and would subject the licensee to additional requests for hearings since hearings can be requested concerning any Commission licensing action. The proposed rule already provides for tight control of licensee activities and periodic reassessments. The licensee car be required through the license to complete closure as units are filled.

From a technical point of view, the site closure activities will be very site specific. Some features of closure can be accomplished as you go, others will involve larger areas of the site or the entire site or longer timeframes. The applicant will have to identify closure activities and schedules in order to adequately develop funding. These plans will be part of the activities covered by the license and license conditions.

The Commission agrees with the thrust of the commenter's point that closure should be completed to the extent practical during operations to minimize the work that must be done after operations cease and economic incentives are gone. The Commission prefers to keep site specific flexibility and minimize administrative burdens in implementing this approach. Thus the suggestion was not adopted.

Rule Change: None.

### ISSUE B-12

Issue:

Signing application under oath

Ontario Hydro (51) Commenters: General Electric (89) The American Society of Mechanical Engineers (107)

Rule Citations: § 61.20(a), 61.24(b)

Summary of Comments: The commenters questioned the need to sign the license application and statements under oath.

Analysis of Comments: Knowingly submitting false information to a federal agency is a crime. Signing under oath focuses attention on the responsibility for the submitted information and provides added weight if the information misrepresents facts or circumstances should the Commission be forced to take action. Considering the long term importance of information associated with waste dispoal, such focus and assurance seem reasonable.

Rule Change: None.

# ISSUE B-13

Issue: Number of copies of application and EIS

Commenter: Union Oil Company of California (66)

Rule Citation: § 61.20

<u>Comment</u>: "The applicant should be required to provide only enough copies of the application and EIS to satisfy the distribution requirement."

<u>Analysis of Comment</u>: Section 61.20 requires the applicant to file 3 copies of the application and environmental report and retain 85 copies for distribution as directed. The distribution and basis for directing the applicant to distribute copies is in proposed § 2.101(b). The numbers of officials and interested parties will be different for each applicant. The 85 copies were judged to the adequate to cover expected distribution needs. The number was specified up front so that the applicant would know that multiple copies will be required and can seek bulk rates for printing or copying. The current provisions of 10 CFR 51.40(c) require the applicant for disposal facilities to retain 85 copies of the environmental report for distribution. This number was retained. Submission of 3 copies instead of 15 as required in § 51.40(c) was adopted because of Paperwork Reduction Act limits. The Commission has no better number and the 85 copies requirement was kept.

Rule Change: None.

# ISSUE C-1

Systems Analysis/Objectives vs. Prescriptive Requirements Issue: Department of the Environment, London (19) Commenters: Commonwealth Edison (35) Los Alamos National Laboratory (43) 11 Ontario Hydro (51) Georgia Yuan (77) General Electric Company (89) Don't Waste Washington Legal Defense Foundation (97) American Institute of Chemical Engineers (102) Conference of Radiation Control Program Directors (103) Carolina Power and Light Company (106) New England Nuclear Corporation (110) United States Department of the Interior (114) U.S. Department of Energy (119) U.S. Environmental Protection Agency (122)

# Rule Citation: Subpart C

<u>Summary of Comments</u>: Almost all commenters supported the approach of addressing disposal from an overall systems standpoint; establishing overall performance objectives (radiological protection standards) and minimum technical requirements; and leaving considerable flexibility in how an applicant or licensee would design and operate a site. For example, the U.S. Department of the Interior stated they agreed, in general, with the flexibility and conservatism of the combined prescriptive and performance objective approach.

The New England Nuclear Corporation, The General Electric Company and Carolina Power and Light all offered a similar comment that the development of performance standards in 10 CFR 61 is the best approach to establishing licensing requirements for land-disposal of low-level radioactive waste. They agreed that only essential generic prescriptive requirements should be included in the regulations and all site specific requirements should be incorporated in individual facility licenses. New England Nuclear further commented they agreed that both performance objectives and prescriptive requirements are necessary. Performance objectives should be limited to occupational and environmental impact concentrations and should be specified in the regulations. They also stated that generic prescriptive requirements are appropriate to limit LLW concentrations and to protect inadvertent intruders. These and prescriptive requirements which provide financial surety should also be incorporated in the regulations. In the cases where prescriptive requirements are adopted from other existing or proposed regulations these should be referenced or incorporated in 10 CFR 61. Other prescriptive requirements which limit site inventory or which protect against excessive migration of radionuclides are site specific prescriptive requirements. Detailed prescriptive requirements should not be in the regulation.
The U.S. Department of Energy commented, however, that the proposed regulation is restrictive by setting both overall performance objectives and technical requirements, which could significantly increase the costs and occupational health impacts without any significant benefits in increased safety. DOE recommended that primary emphasis should be on the overall performance objectives, and the Commission should provide applicants with flexibility to propose specific subsystem performance criteria based on a systems approach which considers site characteristics, design, and operating practices.

The Don't Waste Washington Legal Defense Foundation questioned the justification for relying primarily on performance objectives in a area where concern for public health and safety is so great. They further questioned why additional technical requirements could not be used in the rule. The American Institute of Chemical Engineers commented that the rule should present only performance objectives and no technical requirements. They argued that the establishment of technical requirements would constrain a licensee, result in overconservatism, and prevent an operator from taking advantage of natural or engineered features of a site which could allow disposal of higher concentrations without affecting public health and safety protection.

Yuan commented that while the goal of flexibility has merit, the proposed rule does not provide enough specific guidance. She particularly felt that the site suitability requirements would be difficult to apply because they are too general and depend on meeting the performance objectives.

Analysis of Comments: The approach the NRC has followed is to set overall performance objectives to define an acceptable level of safety leaving an applicant or licensee flexibility in choosing design features and operating practices to achieve the objectives. The NRC also established some minimum prescriptive requirements that were judged necessary in all cases in light of past operating experience with waste disposal and based on specific controls needed for disposal of waste according to the classification system established by Part 61. Based on the comments, this approach appears acceptable and NRC plans to continue its use. The U.S. DOE provided no specific details with respect to their claim that Part 61 could significantly increase costs and occupational health impacts without any significant benefits in increased safety. NRC has examined the Part 61 requirements as modified based on comments filed on the draft rule and further staff analysis, and believes that they do provide a significant benefit in increased safety over the long-term and a reduction in long-term care costs for a small increase in costs to some waste generators to improve waste forms and a small increase in disposal costs. With respect to the comments from the American Institute of Chemical Engineers that the technical requirements will constrain a licensee and not allow full consideration of site characteristics and design features, the NRC believes that the flexibility provided in the technical requirements will not unduly constrain a licensee. They will, however, help ensure a uniform level in safety in disposal and that problems which have occurred in the past will not occur to the same extent in the future.

Rule Change: None

# ISSUE C-2

Issue: Need for EPA Standards

Commenters:

1

Catherine Quigg (13) Conference of Radiation Control Program Directors (103) North Carolina Radiation Protection Commission (109) Kerr-McGee (115)

U.S. Environmental Protection Agency (122)

1 1 5 5

## Rule Citation: Subpart C

<u>Summary of Comments</u>: Quigg questioned whether NRC is preempting EPA standards setting authority. North Carolina urged a high priority for EPA efforts. The Conference supported NRC objectives in the absence of EPA standards. Kerr-McGee expressed the view that the exposure and emission standards proposed for Part 61 are premature and beyond the agency's authority to the extent that they are not already embodied in 10 CFR Part 20. The EPA supported the performance objective and prescriptive requirements approach used and noted that the information presented will assist EPA in its standards program.

<u>Analysis of Comments</u>: Quigg's comment is addressed in the following analysis of Kerr-McGee's comment.

The Kerr-McGee comment is based upon the transfer to EPA, by Reorganization Plan No. 3 of 1970, of Atomic Energy Act authority to establish generally applicable environmental standards for the protection of the general environment from radioactive material. Such EPA standards can take the form of "limits on radiation exposures or levels, or concentrations or quantities of radioactive material, in the general environment outside the boundaries of locations under the control of persons possessing or using radioactive materials" (the quoted language is from Reorganization Plan No. 3). The commenter argues that this transfer of function leaves NRC no authority to issue exposure and emission standards. Until EPA acts NRC is alleged to be frozen into the standards of 10 CFR Part 20 promulgated prior to 1970.

The comment sweeps too broadly, overlooking the legislative history of Reorganization Plan No. 3, which clearly establishes that the role to be played by EPA consists of establishing base levels of radiation exposure or concentrations of materials regulated under the Atomic Energy Act when the materials have been dissipated to the general environment and are no longer under control at a specific site. (See Hearings, "Reorganization Plan Nos. 3 and 4 of 1970" Subcommittee on Executive Reorganization and Government Research, Committee on Government Operations, U.S. Senate, 1970, at pp. 136-142.)

A subsequent memorandum dated December 7, 1973, from Ray L. Ash, Director of the Office of Management and Budget further clarified the relationship between AEC (now NRC) and EPA as to standard setting authority. EPA was ordered, on behalf of the President, to discontinue preparation of any standards for facilities, and to restrict its role under Reorganization Plan No. 3 to setting standards for the total amount of radiation in the general environment from all facilities combined in the uranium fuel cycle. When viewed against this more closely defined division of labor it is clear that the standards in proposed 10 CFR 61.41 fall within the ambit of NRC authority. The standards govern release of radioactive materials from within the boundaries of locations under the control of licensees. The Supplementary Information to the proposed rule also notes that the standards are anticipatory of EPA ambient standards. If and when EPA issues ambient standards the release rates in Part 61 will be adjusted, if necessary, to be consistent and in harmony with the EPA standards.

The Commission also rejects the notion that the standards of 10 CFR Part 20 are frozen in place until EPA takes final action on related ambient standards. There is no law of which the Commission is aware that stipulates that it must await the action of another agency before it can discharge its statutory duties to protect the health and safety of the public from the hazards of source, byproduct, and special nuclear material disposed of in commercial near-surface burial grounds.

Rule Changes: None

ISSUE C-3

Issue:

Performance Objective for Environmental Protection

Marvin Lewis (3)

Commenters:

New Mexico Department for Health and Environment (4) Catherine Quigg (13) PA, Dept. of Environmental Resources (16) Department of the Environment, London (19) Joseph H White III (21) Union of Concerned Scientists (36) American College of Nuclear Physicians (53) Union Oil Company of California (66) Argonne National Laboratory (68) Georgia Institute of Technology (70) South Carolina Department of Health and Environmental Control (79) Birmingham Audubon Society (80) State of New York Department of Law (99) Conference of Radiation Control Program Directors (103) New England Nuclear (110) Colorado Department of Health (111) Kerr-McGee (115) U.S. Department of Energy (119) U.S. Environmental Protection Agency (122)

<u>Rule Citation</u>: § 61.41, Protection of the general population from releases of radioactivity.

<u>Summary of Comments</u>: There was no clear pattern or recurring issue in the comments. Marvin Lewis commented that there is no absolute amount of leakage which would be considered unsafe and the entire contents of a site could leak out slowly into the environment without violating the proposed rule. Joseph White, III asked what amount of off-site migration would be acceptable to NRC.

The State of New York advocated a "zero discharge" criterion for ground and surface waters or at least an ALARA criterion.

The EPA commented that the establishment of an individual exposure limit at the site boundary for enviornmental releases as proposed in § 61.41 was appropriate. They stated that the 25 mrem/yr limit is in the correct range of values (1-25 mrem/yr was analyzed by NRC) that should encompass any future EPA standard for LLW disposal facilities. Based on NRC's analysis, NRC does not anticipate any need to change the technical requirements of Part 61 to meet such an EPA standard. The Department of the Environment, London commented that the standards specified in the rule appeared to be reasonable. The approach of setting an individual dose limit for inadvertent intrusion and separate limits for ground water releases is consistent with the proposed U. K. approach to radiological protection standards for disposal options and with ICRP principles. The Conference of Radiation Control Program Directors supported the objective as proposed.

12

As discussed under Issue C-2, Kerr-McGee Corporation commented that the standards established in § 61.41 were premature and beyond the NRC authority to the extent that they are not already embodied in 10 CFR Part 20. Catherine Quigg asked why NRC didn't wait for the EPA to set its limits and, in effect, preempted the EPA's authority by "anticipating" the EPA standards. The Colorado Department of Health also expressed the view that this performance objective was not sufficiently justified and noted the legal challenge to the 25 mrem criterion.

Kerr-McGee also commented that the proposed limits in § 61.41 are too stringent and unsupported. They specifically noted that the apparent use of the EPA 40 CFR 190 standard for fuel cycle facilities to derive a standard for LLW disposal facilities is not correct since:

- EPA specifically excluded operations at waste disposal facilities from Part 190. NRC may not promulgate its own standard in the absence of action by EPA;
- (2) The limits proposed are too low since they are a small fraction of the existing limits in Part 20 which have not resulted in any untoward results. They are also a small fraction of natural background exposure which leading authorities believe is not hazardous or harmful; and
- (3) The limits deviate from that specified in EPA 40 CFR Part 190 since they do not exclude radon and its daughters excluded by 40 CFR Part 190. Without this exclusion they note that the rule may be unduly stringent when applied to the disposal of either uranium or thorium ore residue wastes at a LLW disposal facility.

The Pennsylvania Department of Environmental Resources pointed out that the same ALARA principles applied in the development of the 40 CFR Part 190 standards may also be applicable in the case of LLW disposal facilities since engineered barriers and other site design features could further reduce potential exposures in a cost effective manner.

The Department of the Environment, London commented that the dose limits for ground water releases are consistent with current U.S. practice and, when

combined with the requirement that EPA Drinking Water Standards should not be exceeded, should ensure the doses are as low as reasonably achievable. The New Mexico Department for Health and Environment recommended that the EPA drinking water standards should be applied to both existing and future potential public and private drinking water supplies. The Union of Concerned Scientists similarily commented that the EPA drinking water standards should be extended to all actual or potential water supplies outside the site boundary. The American College of Nuclear Physicians recommended that the EPA drinking water standards should be applied at the site boundary. They pointed out that the "nearest public drinking water supply" criterion might change after establishment of the site causing potential danger of retroactive design limitations. South Carolina noted that the rule should clarify whether the EPA drinking water limit or 25 mrem/yr apply at the site boundary. EPA commented that it was inappropriate to apply the EPA drinking water standard in § 61.41 as proposed by NRC and stated it should be deleted from § 61.41. Kerr-McGee recommended that the 10 pCi/l limit for uranium and thorium in drinking water should be deleted from § 61.41. Argonne suggested including standard deviation for the limit for uranium and thorium.

New England Nuclear Suggested clarifying changes concerning the use of "annual" and "dose." The DOE commented that the basis for the drinking water limits should be provided.

<u>Analysis of Comments</u>: With respect to the comments of Marvin Lewis and Joseph White III, the performance objective in § 61.41 defines an acceptable level of safety regarding releases to the environment from all environmental pathways of release from the site. It thus defines a safe level for releases from the site. Since migration is the principal environmental release pathway; the performance objective also defines an "acceptable" amount of migration that would be allowed at the site.

Kerr-McGee's comments that the performance objectives were premature and beyond NRC's authority were discussed and addressed under Issue C-2. The EPA, under its generally applicable standards setting authority, has responsibility to prepare a standard that will set limits on radioactivity in the general environment from disposal facilities. Presently, there exists no such EPA standard. In the absence of a standard, NRC examined a range of limits within that expected for the EPA standard and selected a proposed performance objective that establishes a release limit for the site boundary. The performance objective thus takes the place of EPA standards and will be replaced by the EPA standard when it is developed. Under its regulations development authority, NRC may establish such limits on releases as it deems necessary to ensure protection of the public health and safety. As such, NRC developed the performance objective under its general authority to establish such limits for radiation protection purposes. In a rule making action, NRC is not solely limited to existing standards in Part 20 and NRC does not intend to withdraw any portions of the proposed rule that may be related to the performance objective.

NRC did not adopt the 40 CFR Part 190 standard for application to LLW disposal facilities and as such is not subject to any limitations that are expressly implied by EPA in the application of that standard. Rather, NRC used the 40 CFR 190 standard to help establish a range of dose guidelines that should be analyzed in selecting a performance objective for Part 61. Based on the analyses, NRC selected 25 mrem (whole body and other organs except thyroid)

and 75 mrem (thyroid) as the preferred performance objective for environmental releases. The performance objective represents what is achieveable and ALARA at a LLW disposal facility and as such is a small fraction of other dose limits set out in 10 CFR Part 20. The actual performance at individual disposal facilities may be less than 25 mrem/yr based on site specific conditions. NRC does not believe it is too stringent or low and based on EPA's comments, believes it is close to any standard EPA may develop in the future. NRC, thus, has made no change to this part of the performance objective on environmental releases, but has, however, added the ALARA concept for emphasis

As suggested by Kerr-McGee, Part 61 does not apply to the bulk disposal of uranium and thorium mill tailings or wastes (byproduct material as defined in § 40.4(a-1)) which are covered by Part 40. Disposal of other uranium and thorium wastes and small amounts of tailings is permitted by Part 61. Also, NRC has not addressed as a part of this rulemaking the subject of the linear non-threshold model, radiation hormesis, or other detailed aspects dealing with radiation dose response relationships. They will be addressed in other forums.

With respect to comments on the application of EPA drinking water standards to the nearest public drinking water supply, NRC heavily weighted the EPA comment that its use in § 61.41 was not appropriate and believes that it should be deleted from the performance objective. NRC intends, as a part of the review of an application for a site, to consider and evaluate water usage near the site including application of appropriate standards.

With respect to comments on "annual" and "dose," NRC did not express the limit in terms of effective dose since NRC is presently evaluating, as a part of development of proposed amendments to Part 20, whether and how NRC will implement this approach. Until this work is completed NRC does not plan to use this approach in individual rulemaking actions and no change will be made to § 61.41 in this regard. Based on the final decisions made in amendments to Part 20, however, compatible changes may need to be made to the performance objectives in Part 61. NRC considers the performance objective dose limits expressed in units of rem, to mean dose equivalent. The term annual or year refers to any period of 12 consecutive months.

<u>Rule Change</u>: Based on the staff's analysis of comments, the performance objective for protection of the general population from releases of radioactivity has been revised to delete reference to the EPA drinking water standard and include ALARA. The revised performance objective reads as follows:

§ 61.41 Protection of the general population from releases of radioactivity

Concentrations of radioactive material which may be released to the general environment in ground water, surface water, air, soil, plants, or animals 'must not result in an annual dose exceeding an equivalent of 25 millirems to the whole body, 75 millirems to the thyroid, and 25 millirems to any other organ of any member of the public. Reasonable effort should be made to maintain releases of radioactivity in effluents to the general environment to as low as is reasonably achievable. ISSUE C-4

Performance Objective for Intruder Protection Issue: Commenters: Marvin Lewis (3) Catherine Quigg (13) Department of the Environment, London (19) Commonwealth Edison (35) Union of Concerned Scientists (36) Union Carbide Corporation (39) Bechtel National Inc. (44) Ontario Hydro (51) Argonne National Laboratory (68) Oswald U. Anders (73) Birmingham Audubon Society (80) American Nuclear Society (87) Amy S. Hubbard (90) New York State Department of Law (99) Paul Ziemer for EG&G Idaho, Inc. (104) **Program Review Committee** Carolina Power and Light (106) New England Nuclear (110) The American Society of Mechanical Engineers (113) Kerr McGee Corporation (115) U.S. Department of Energy (119) U.S. Environmental Protection Agency (122) Rule Citation: § 61.42 - Protection of individuals from inadvertent intrusion Summary of Comments: There was no clear pattern or recurring issue in the comments. The Union of Concerned Scientists stated that the nature of the intrusion events should be specified in the amendments. The American Society of Mechanical Engineers inquired about the time frame that should be applied to the inadvertent intrusion to evaluate the 500 mrem/yr. Marvin Lewis

commented that 500 mrem seemed like a large exposure for making a small mistake and questioned how a 500 mrem exposure could be ensured without on-site security. The Birmingham Audubon Society commented that instead of using 500 millirem per year, it should be no more than 10 percent of background per Catherine Quigg inquired how the NRC arrived at the 500 mrem limit. year. She asked if NRC took into consideration the greater health risk to children, pregnant women or the fetus, or ill, elderly people from such a dose. She stated there is no basis for NRC projecting that only one, or at most, a few persons would be exposed and asked how NRC could predict how many people might be farming or digging in a certain plot of land over the next 200-300 years. Amy Hubbard commented similarly noting that present trends in population growth, soil erosion, and water resource needs will result not in inadvertent intrusion but actual permitted uses of the land and water at a site. She stated the analyses of intruder exposures were deficient since they did not consider a more plausible family intruder where some numbers of the family would spend more time at the site and some family members (children and pregnant women) would be more sensitive to radiation. She concluded that the analysis was inadequate for present day society and its applicability to the future is even more uncertain.

The Department of the Environment, London commented that the approach of setting an individual dose limit for inadvertent intrusion is consistent with the U.K. approach and with ICRP principles. They questioned, however, whether the limit was consistent with ICRP dose limits because the rule did not state whether the limit is in terms of effective dose. They stated this point should be clarified. The New England Nuclear Corporation also recommended that NRC clarify the performance objectives by specifying internal and external dose equivalent limits to individual organs as suggested by the ICRP and that dose be defined to mean "dose equivalent". They inquired as to whether "annual" and "year" refer to a calendar or a sliding year.; Onario Hydro supported the 500 millirem dose limit as only a few people could conceivably receive that dose. The New England Nuclear Corporation commented they agreed with the proposed dose limit provided that waste concentration limits are calculated to ensure, with reasonable probability, that the inadvertent intruder does not receive more than 500 mrem/yr. They further noted that since the inadvertent intruder is identified as the critically exposed individual for most radionuclides, more effort should be directed into determining the probability of intruder scenarios occuring. Waste concentration limits could then be relaxed if these interaction probabilities are factored into the impact calculations. The New York State Department of Law stated they were impressed with the innovative approach to disposal regulation through protection of various populations at risk; general populations, intruders and employees. The American Nuclear Society and Bechtel National Inc. commented on the absoluteness of the performance objective and recommended using the word "should" in place of the word "must".

6

The Union Carbide Corporation commented that with continuing passive institutional controls such as deed restrictions, the dose limit of 500 mrem was too low. The Union Carbide Corporation recommended it be raised by a factor of 10 to the occupational exposure limit of 5000 mrem. Carolina Power and Light commented similarly noting that the 5000 mrem/year limit currently applied to radiation workers represents an acceptably low risk and should be the standard applied, that there is a low probability of intrusion and a limited number of individuals would be involved. The Department of Energy, the LLW Management Program Review Committee, Argonne National Laboratory, and Oswald U. Anders all commented that the inadvertent intruder scenario is given too much weight. Reasons stated included the fact that it leads to unreasonably low concentrations for radionuclides in Table 1, exercises inordinate control over potential disposal site options, and safety regulations for the general population should not be dictated by the hypothetical actions of a very small number of individuals. The LLW Management Program Review Committee noted that the intruder scenario is more analogous to an accidental exposure pathway than to a chronic exposure pathway. They sugggested that more emphasis should be placed on requirements to reduce the likelihood of intrusion such as use of durable monuments large enough to warn potential intruders. Oswald Anders also noted the inadvertent intruder could be sufficiently warned by a big enough monument with an inscription in deep relief on it about the disposal site. For the same reason, the U.S. Department of Energy recommended that the NCRP 25 rem accidental exposure limit be used as the intruder dose limit rather than 500 mrem. They also noted that the 500 mrem limit leads to unreasonably low concentrations and increases the costs of disposal. Oswald Anders commented that the inadvertent intruder's needs are readily taken care of by placing the required cap onto the burial trench. He stated no truly inadvertent intruder would burrow to more than 7 feet under the ground without considerable deliberation and knowing what he is doing. The Kerr-McGee Corporation also objected to

restrictions embodied in the intruder performance objective. They argued that the inadvertent intruder hypothesis is predicated on the assumption that the government will fail which is inconsistent and antithetical to our constitution. They also argue that this is doubly the case since the proposed regulations require licensees to provide funding for institutional controls of indefinite duration.

The EPA commented that the 500 mrem dose limit was not appropriate as a regulatory limit. A licensee would not be able to monitor or demonstrate compliance with a specific dose limit imposed in the objective today that applies to an event that might occur several hundred years from now. They recognized use of 500 mrem as a basis for determining the concentration limits in Table 1 of Part 61, noting that, given ALARA actual exposures to an inadvertent intruder would be lower then 500 mrem per year.

Commonwealth Edison commented that this performance objective lacked adequate provision for protection of the general public from potential releases to the environment. They noted that although terrorists are not "inadvertent" intruders, they believe there should be a plan to safeguard a site from terrorists in that disposal sites may be more susceptable to covert acts than nuclear power sites. Amy Hubbard also commented that acts of terrorism and sabotage were neglected.

<u>Analysis of Comments</u>: With respect to comments that NRC express the dose limit in terms of effective dose, NRC did not express the limit in terms of effective dose since NRC is presently evaluating, as a part of development of proposed amendments to Part 20, whether and how NRC will implement this approach. Until this work is completed NRC does not plan to use this approach in individual rulemaking actions and no change will be made to § 61.42 in this regard. Based on the final decisions made in amendments to Part 20, however, compatible changes may need to be made to the performance objectives in Part 61. NRC considers the performance objective dose limits expressed in units of rem, to mean dose equivalent. The term annual or year refers to any period of 12 consecutive months.

NRC staff agree with EPA's position that it is not appropriate to set out a specific limit in the performance objective although a limit should continue to be used as a basis for the waste classification concentration limits in § 61.55. A dose limit in the performance objective will be essentially impossible to monitor against and ensure compliance with for several reasons. Inadvertent intrusion into a LLW disposal facility is a hypothetical event which may never occur. NRC's objective is to ensure that if it should occur, the exposure to the individuals involved would not be unacceptable high. Since it is not possible to control today what may happen several hundred years in the future, NRC is controlling the concentration of waste disposed of today such that if someone should contact it through reasonable uses to which the site could be put in the future, the individuals involved would not receive a high exposure. NRC used 500 mrem as a basis for determining the concentration limits in waste which would result in actual potential exposures to an inadvertent intruder of a few 100 mrem at 100 years and a few mrem at 500 years. Thus the performance objective is to keep potential exposures to an inadvertent intruder to a dose limit that is not unreasonably high given the accidental and hypothetical nature of the event. Using 500 mrem as a limit to calculate concentration limits yields potential exposures of a few mrem after 500 years.

In addition, there is no practical way in which a licensee could monitor and assure compliance with an actual exposure limit and there is no way to absolutely predict the type of intrusion event which might occur, when it might occur, and how many people might be involved.

. . . . . .

The majority of the commenters, including Argonne National Laboratory, the Union Carbide Corporation and the U. S. Department of Energy, Oswald Anders, the LLW Management Program Review Committee, and the Carolina Power and Light Company, in their comments about considering probability of occurrence, expressed concern about weighting too heavily inadvertent intrusion in determining disposal requirements for waste. Several expressed concern about the basis for the 500 mrem limit and some recommended that a higher dose limit (e.g., 5000 mrem or 25 rem) should be applied.

NRC's selection of the 500 mrem limit was based on (1) public opinion gained through the 4 regional workshops held on the preliminary draft of Part 61; (2) its acceptance by national and international standards organizations (e.g., ICRP) as an acceptable exposure limit for members of the public; and (3) the results of analyses presented in Chapter 4 of the draft EIS. In considering EPA's comment above that the intruder dose limit is not appropriate as a regulatory limit and considering ALARA (given the mix of waste disposed of at a site) actual potential exposures at 500 years would only be a few mrem; comments on the nature of intrusion events and ability to predict what with actually happen; and comments that NRC has weighted intruder considerations too heavily, NRC has changed its approach to addressing inadvertent intrusion. No change has been made in the intruder scenarios considered since they reflect, to the extent that we can predict today, reasonable , probable and productive uses of the disposal facility site. As such, probability of occurrence of specific events will not be quantitatively assessed. The NRC believes that the primary concern of those who feel that the intruder protection objective is too restrictive is the effect that this has on the concentrations of certain nuclides that are acceptable for disposal in a near surface facility and the need to meet waste form requirements such as stability for some wastes. With this in mind and in response to other comments, the NRC has reevaluated the calculations that establish the waste classification concentration limits to eliminate unnecessarily conservative assumptions with the result that the analysis is more realistic and the limits for several important nuclides have been raised. With this action, the NRC believes that most of the concerns of those who encouraged higher exposure limits or less emphasis on protection of intruders will have been met. St. Company prover

As previously discussed, the actual exposures to an intruder if such events should occur at 500 years would be a few millirem. NRC did not specially . consider the greater health risk to children, pregnant women or the fetus, and others except to the extent that they may already be considered as a part of the 500 mrem exposure limit recommended by various national and international authorities such as the ICRP.

2.5 - 222 Several commented about the nature of the intrusion events, (Union of Concerned Scientists), how many people would be involved, how NRC predicted this (Catherine Quigg) and questioned the adequacy of site security to preclude inadvertent intrusion. (Marvin Lewis). Kerr-McGee objected to the restriction in the performance objective since it assumed failure of the government which is inconsistent with our constitution. For purposes of

1.11

analysis, NRC considered 3 intrusion events. These were selected based on evaluation of the broad range of events possible, those considered by other investigators, and the likelihood of occurrence. The 3 events can be characterized as intruder construction (exposure to workers constructing a house at the site), intruder agriculture (exposure to individuals living in the house constructed and consuming food grown onsite), and intruder discovery (exposure to an individual who digs into the waste, realizes something is wrong and ceases his excavation activities). NRC assumed that only a few individuals would be exposed though such activities based on the number of people normally required to construct and live in a house. With respect to site security and institutional control, during the operational and post closure observation and maintenance periods, the licensee would be responsible for maintaining site security and control of the site. This would include fencing, posting, security guards, and controlled entrance and exit from the site. During the following 100 year institutional control period the government land owner would be responsible for site security which would include maintaining the fence and warning signs, physical surveillance and control over access to the site. NRC does not assume that the government fails at the end of the 100-year institutional control period, but rather that the government ceases active control over access to the site. Thus, Kerr-McGee's assumption has no foundation in the rule. The rule does not presuppose collapse or failure of government, but rather places a restriction on the character of radioactive material disposable by near surface disposal that serves to relieve government of the burden of actively excluding persons from the site in perpetuity. The lifting of institutional control under the rule results from the fact that an inadvertent intruder will not be subjected to an unacceptable high dose of radiation (defined to be 500 mrem for purposes of analysis). "Based upon the kinds of material expected to be buried and public opinion, the NRC staff determined that the maximum period of institutional control will be 100 years after closure. Kerr-McGee also states that funding is required for institutional controls of an indefinite duration. This is not so. Active institutional controls are not indefinite (100 years as noted above) if the concentration limits of § 61.55 are adhered to. Other passive controls such as government land ownership, records, deed restrictions and covenants which will have little to no cost would continue after active controls cease and would serve as an additional means for preserving and transmitting information about the site.

Finally, as noted above, NRC also did not directly consider the probability of various intrusion events occurring except to the extent of considering reasonable, probable productive uses to which the land could be put. Unusual activities such as an archaeologist reclaiming artifiacts at the site were not considered. NRC agrees with the comments on use of "permanent" markers at a site to inform of the material disposal of at the facility.

<u>Rule Changes</u>: Based on the staff's analysis of comments, the performance objective for protection of the inadvertent intruder has been revised to delete specific reference to 500 mrem. It reads as follows:

Design, operation and closure of the land disposal facility must ensure protection of any individual inadvertently intruding into the disposal site and occupying the site or contacting the waste at any time after active institutional controls over the disposal site are removed. In addition, words have been added to Part 61 to indicate that monuments are required to warn against intrusion.

# **ISSUE C-5**

Issue:

Protection of Individuals During Operations: ALARA for the Performance Objectives

Commenters:

Dow Chemical Company (17), (83) Department of the Environment, London (19) New York State Department of Law (99) Conference of Radiation Control Program Directors (103) New England Nuclear (110) U.S. Environmental Protection Agency (122)

Rule Citation: §61.43 Protection of Individuals During Operations; Subpart C

Summary of Comments: EPA commented that this section was not clear with respect to whether releases to the environmental during operations would be covered by Part 20, as implied here, or by the performance objective in §61.41 on protection of the general population from releases of radioactivity. EPA further commented that it was their view the performance objective on protection of the population should apply both during and after operations. The DOW Chemical Company commented that in an effort to reduce the uncertainties associated with pathway analysis, the NRC should make it very clear that the spirit of the ALARA concept applies to all standards and technical criteria. The Department of the Environment expressed the view that the approach used to develop Part 61 reflected ALARA principles throughout. New York and the Conference commented that ALARA should guide all site activities and all of Subpart C. New England Nuclear cautioned against using ALARA to impose excessive restrictions.

Analysis of Comments: NRC agrees with EPA's view on application of §§61.41 and 61.43. NRC intends that §61.43 would require that everything in 10 CFR Part 20 with respect to occupation safety, control of radiation in restricted areas, control of radiation in unrestricted areas, and maintaining radiation exposures and releases of radioactive materials to unrestricted areas as low as reasonably achievable would apply to the disposal facility. Releases of radioactivity to the environment during operations and over the long term after operations cease would be governed by §61.43. Although §61.43 conflicts with limits in Part 20, NRC believes the lower limits in §61.43, which are achievable, should apply in the spirit of ALARA.

Rule Changes:
1. Add to §61.41:

... Every reasonable effort should be made to maintain releases of radioactivity in effluents to the general environment as low as is reasonably achievable.

Add to §61.43: 2.

Except for releases of radioactivity in effluents from the land disposal facility, which shall be governed by §61.41 of this part every reasonable effort shall be made to maintain radiation exposures as low as is reasonably achievable.

# ISSUE C-6

Issue: Stability of the Disposal Site After Closure

<u>Commenters</u>: Dow Chemical Company (17) Joseph H. White, III (21) Sierra Club (37) Duke Power Company (48) Northeast Utilities (85) New York State Department of Law (99)

## Rule Citation: §61.44

Summary of Comments: The New York State Department of Law commented that they particularly support the notion of site stability as a criterion for successful disposal. If the site is stable over long periods of time, the likelihood of excessive long-term maintenance costs is significantly lower. They also point out that, as history has shown, site owners (States in this case) have been and probably will be saddled with unrecoverable multi-million dollar bills each decade to maintain facility integrity. Joseph H. White, III inquired what would be done if stability is not met. Duke Power Company commented that stability, as defined, applied to the waste and disposal site. They recommended clarification to show that the more important aspect is really stability of the trench, not the waste. Dow Chemical Company commented that placing reliance on stability of the disposed waste will be possible only if stability is specified and enforced by the standards and technical criteria during the disposal operation. Thev pointed out that techniques which provide stable waste forms are available today but due to lack of regulatory standards and technical criteria they are not in general use. They noted that these techniques result in monoliths which place the radionuclides into forms which are less likely to be dispersed and less likely to be released to and transported by ground water. These monoliths are more likely to be recognized and therefore avoided or properly investigated and handled by intentional or inadvertent intruders. They also commented that these factors are within the concept of ALARA and reduce the need to rely on the many unknowns and uncertainties associated with pathway analysis.

The Sierra Club commented that the goal of long-term stability and avoidance of continued, active maintenance at the site is central to Part 61 and noted the goal was enthusiastically shared by environmentalists, industry and government. They expressed concern, however, that there is no basis in experience with burial sites located in regions of moderate to high rainfall that indicates this goal can be achieved by the means outlined in Part 61. They further commented that unless the recommended measures have been <u>demonstrated to work</u> at an already existing facility, then the measures are examples of wishful thinking and not concrete tactics whose implementation will result in the desire goal. To establish regulation on the basis of a hoped-for future dream and not on the basis of actual experience is a dangerous process that may lull the unwary into a false sense of security.

They commented that the goal of stability supposedly will be achieved through specific site design features. They cite for example, in § 61.51(a)(4) that "Covers must be designed to prevent water infiltration" but note that at none of the existing sites and after twenty years of experimentation has water

infiltration through covers been prevented. They note an apparent misunderstanding in Part 61 regarding the inevitable slumpage of trench contents. Without this understanding--that slumpage leads to cover indentations, cracks and collapse--then it is possible to make the mistake that cover "design" can solve the problem. Likewise, they note there is a fundamental misunderstanding that slumpage can be avoided by techniques of placing materials into the trench or by packaging of materials. They finally comment that unless the relationships are clearly seen between gradual deterioration of packaging, slumpage, trench cover collapse and water infiltration, no meaningful design criteria can be established.

Northeast Utilities suggested that Class A waste disposal areas be exempted from the stability objective because the wastes are not required to be stable.

.

Analysis of Comments: NRC has stated the stability performance objective as a goal. The site characteristics, facility design and operations, and waste characteristics should work together to assure protection of the public health and safety without the need for continued active care and maintenance. As discussed under Issue Number GEN-1, the absoluteness of the language needs modification. However, only minor custodial care (and minor maintenance) should be required over the long term. As Mr. White inquired, there may be cases where this goal will not be met in an absolute sense which may require some additional maintenance to achieve a stable site condition.

. .

With respect to the comments by Duke Power, Dow and the Sierra Club, each seemed to reflect one part of the combination of site characteristics, design and operations (e.g., trench and covers) and waste form characteristics which would work collectively to ensure long-term site stability. That is, NRC does not expect reduced infiltration through trench covers that are constructed on waste forms and containers which are not structurally stable and which have not been placed into a trench and backfilled in a manner to help achieve a stable disposal environment. It is with this view in mind that NRC in Part 61 separated out the higher activity wastes requiring that they be placed into a structurally stable form or container. Waste must also be disposed such that disposal unit design, waste emplacement, backfilling and other operational techniques will lead to a stable site condition. Lower activity wastes that do not need to meet the structural stability waste form requirement are required to be disposed of in separate disposal units such that they do not contribute to the slumping and collapse of trench covers. (This is one of the primary reasons for trench cover collapse problems that have occurred at several existing sites.) In this way a stable foundation is created for a better designed trench cap which can retard infiltration and direct water away from disposal units. Disposal units containing lower activity compressible wastes may require some increased maintenance to achieve a stable site condition but the hazard presented by such wastes is low. Also for such cases, greater emphasis will be placed on appropriate design and operations to achieve a stable site condition given the compressible and degradable nature of the waste involved.

Rule Change: Based on the staff's analysis of comments, NRC changed the stability performance objective (§61.44) by deleting the words "disposed waste and the" and adding the words, "to the extent practicable" after eliminate. " In this way stability of the disposal site itself and the absoluteness of the requirements is addressed.

## ISSUE C-7

<u>Issue</u>: Need for Additional Performance Objectives <u>Commenters</u>: Los Alamos National Laboratory (43) State of California (93) Health Physics Society (96)

# Rule Citation: Subpart C

<u>Summary of Comments</u>: The Health Physics Society recognized the need to assure radiation protection for employees, possible intruders, and the general public during land disposal facility operations and beyond. They recommend that each of these groups be properly protected using limits in 10 CFR 20, as currently written or as proposed, rather than proposing a system of new and unique limits related to waste disposal. They noted the establishment of a unique system of dose limits for a separate area of the nuclear fuel cycle seemed unnecessary and may imply a special need for protection in the minds of the public; thereby exacerbating an already confused public perception of radioactive waste disposal issues. The Los Alamos National Laboratory commented that it would be desirable to add a performance objective to minimize intrusion by plants and animals which have the potential for transporting radionuclides to the food chain. The State of California noted that §61.40 sets standards to avoid excessive exposure to humans and that excessive exposure to animal life should be avoided also.

Analysis of Comments: The NRC staff agrees with the comments of the Health Physics Society that new radiation protection limits should not be developed if existing limits can be applied. This is the approach that NRC has followed. With respect to occupational safety, NRC has required that disposal facilities comply with the same limits in Part 20 which other licensees must comply with. With respect to releases to the general environment from disposal facilities, however, NRC does not have any existing standards in Part 20 or EPA standard to apply. Existing standards in Part 20 apply to effluents where a licensee can exercise direct control over the releases. They don't necessarily apply to the migration and release of material from a disposal facility from less readily controllable means. At the same time, protection of the inadvertent intruder is a new consideration applicable to disposal facilities, but not necessarily applicable to other types of facilities NRC licenses. Thus, NRC was faced with examination and determination of radiological limits involving somewhat different considerations than those already covered in Part 20. As discussed in the EIS, however, NRC did consider the range of existing standards in establishing performance objective for environmental and intruder protection. NRC recognizes the need to minimize intrusion by plants and animals and considered this as part of the draft EIS analysis. NRC established no specific requirements in this area because of the site specific nature of the type of animals and vegetation which should be considered. NRC plans to review, on a site specific basis, to ensure that burrowing animals and deep routed plants which could serve to transport radionuclides offsite are not a problem.

#### Rule Changes: None

# **ISSUE D-50-1**

1

Issue:

....

Complexity siting requirement

Commenters:

Ontario Hydro (51) Union Oil Company of California (66) Georgia Institute of Technology (70) Georgia Yuan (77) South Carolina Department of Health and Environmental Control (79) U.S. Ecology (101) U.S. Department of Interior (114) U.S. Department of Energy (119)

<u>Rule Citation</u>: 61.50(a)(2)

<u>Summary of Comments</u>: Six of the eight commenters questioned the vagueness of the requirement: i.e., what does it mean to be capable of being characterized, modeled, analyzed, and monitored? Two of the commenters (South Carolina Department of Health and Environmental Control and U.S. Department of Energy) suggested examples or rewordings to answer the preceding question. The U.S. Department of the Interior queried what types of models were included, i.e., physical scale, numerical, or conceptual; and, Union Oil Company of California asked questions such as: what role will modeling play, what site characteristics and events will be modeled, and are there NRC-approved models?

. . .

<u>Analysis of Comments</u>: The staff recognizes the vagueness of Section 61.50(a)(2) and has attempted to provide additional explanation in the draft technical position paper on site suitability and site characterization. The staff is revising that technical position paper and the examples and suggested rewordings offered by South Carolina Department of Health and Environmental Control and the U.S. Department of Energy will be incorporated into that revision. (See page 5 of NUREG-0902, "Site Suitability, Selection, and Characterization.")

The staff does not believe that a concise statement can be made in the regulation to remove the vagueness noted by the commenters. Rather, technical position papers on these subjects will provide direction. In addition the staff is developing an in-house modeling capability and will share that capability through pre-qualification of prospective computer codes.

<u>Rule Change</u>: No changes are recommended for Section 61.50(a)(2).

# ISSUE D-50-2

• •	
Issue:	Ground water requirements
· · ·	
<u>Commenters</u> :	Department of the Environment, London (19) Sierra Club, Radioactive Waste Campaign (37) Ontario Hydro (51)
• •	Union Oil Company (66) Argonne National Laboratory (68) Georgia Institute of Technology (70)

Paul F. Hadala and Don C. Banks (76) South Carolina Department of Health and Enviornmental Control (79) State of California (93) Peter Skinner for Attorney General of New York (99) New England Nuclear (110) U.S. Department of Interior (114) U.S. Department of Energy (119)

## <u>Rule Citation</u>: 61.50(a)(4), 61.50(a)(7), 61.50(a)(8)

<u>Summary of Comments</u>: Four commenters (Ontario Hydro, Argonne National Laboratory, Department of Interior, and Department of Energy) concurred with the exception to Section 61.50(a)(7) to permit disposal below the water table where diffusion dominated the ground-water flow system. The Department of the Interior recommended using the term "molecular diffusion" and both they and Ontario Hydro specified a soil hydraulic conductivity of less than  $10^{-6}$  cm/sec as appropriate. The State of California, on the other hand, took considerable exception to disposal below the water table and spoke in favor of total containment. They suggested a revision of Section 61.50(a)(7) as follows:

The disposal site must not be located 1/within basins containing usable groundwater or their recharge area, or 2/within geologic formations which will permit the diffusion of radionuclides to the environment, or their transport by groundwater to a degree exceeding the performance objectives of Subpart C.

The Department of the Environment, London, commented that no requirement on ground water other than meeting Section 61.41 was needed. One commenter recommended a siting requirement on soils which will preclude or reduce leachate migration by attenuation. Two commenters (Georgia Institute of Technology and Peter Skinner for the Attorney General of New York) recommended separate mention of requirements for impervious and porous soils. Both cited the ability of porous soils to drain readily and thereby avoid prolonged contact of infiltrating water with the waste. The former commenter suggested the impervious soils could be treated to keep water out of the disposal units; the latter commenter suggested a leachate collection and treatment system for the impervious soils.

Three commenters requested that Section 61.50(a)(7) provide a minimum depth to the water table, and, the Department of Energy stressed the importance of avoiding the transection zone of the water table.

The South Carolina Department of Health and Environmental Control recommended that the natural resources considered under Section 61.50(a)(4) specifically include ground water and aquifers underlying the sites and suggested examples of ground water characterization data needs. Two concerned citizens (Commenters 76) suggested revision of Section 61.50(a)(8) to conform to the wording used in the staff's draft technical position on site suitability and site characterization. Union Oil Company suggested that §61.50(4) address areas of "known" natural resources.

New England Nuclear suggested addressing changes in natural background radioactivity due to construction and assigning a probability limit on ground water intrusion.

Analysis of Comments: The staff appreciates the approval for the exception to permit disposal below the water table where molecular diffusion dominates the ground-water system. The staff disagrees with the comment received from the State of California and would point out that a clay encapsulated disposal unit proposed by the State is somewhat comparable. The basic difference is that the type of site envisioned by the staff as satisfying the exception is an inactive flow system so the water which would contact the solid/solidified wastes would move on the order of less than one foot per year. Given the low hydraulic conductivity and affective porosity of the soils, very little water would actually contact the waste or flow from the disposal units. The travel time will result in sufficient reduction of concentration of the small amounts released; and fine-grained soils will typically provide significant attenuation for most radionuclides. The staff interprets the State of California comment to be largely based on experiences with disposal of liquid wastes in fine-grained soils above the water table. The staff considers that experience is not applicable to the disposal required under 10 CFR Part 61.

S Same Ch

. . .

. . .

The comment from the Department of the Environment, London raises the very interesting question of why, if the performance objectives can be shown to be met, do there have to be any requirements related to water access or diffusiondominated transport. The commenter states that it is only necessary to ensure the releases via ground water will not result in failure to meet the performance objectives. ... Given the latter statement, all that is really needed in the regulation are the performance objectives. However, the staff has identified in 61.50 and 61.51 certain characteristics which will if present (or in some cases if absent) enhance the ability to meet the performance objectives and, consistent with the ALARA concept, result in potential releases which are both within acceptable limits and as low as reasonably achievable through prudent site selection and site design. Since the staff considers the water (soil moisture) to be the prime catalyst for degradation of waste containers, leaching of wastes, subsidence, and transport, particular attention has been given to limiting the access of water into the disposal unit. st toute and st

The staff does not consider a specific siting requirement on such characteristics which promote attenuation of radionuclides to be appropriate. Whereas the staff does agree that attenuation is advantageous for some radionuclides, others such as H-3, C-14, Tc-99 and I-129, are not significantly attenuated. Rather, the rule uses siting requirements which will keep water away from the wastes, result in low volumes released, and provide long travel time for decay.

Proposed sites with impervious or porous soils will be evaluated on a case basis. The staff agrees that site design features can effectively limit access of water to wastes; however, the staff takes exception to any design which relies upon a leachate collection and treatment system as violating Section 61.44. Sump systems may well be installed in the disposal units to collect precipitation entering while the trench is open. The sump systems can provide monitoring locations and, if necessary, leachate collection systems. However, the staff does not want the disposal facility designed so as to rely on the sump systems to make the facility meet the performance objectives and/or to prevent bathtubbing. Rather, the disposal facility should fully meet the performance objective on isolation (Section 61.41) without the sump systems.

The staff agrees with the Department of Energy comment on avoiding alternative wetting and drying within the disposal unit due to fluctuations of the water

table. However, the staff does not feel that a numerical value can be assigned to a minimum depth to the water table. That particular depth will vary according to soil type due to the height of the capillary fringe above the water table. The staff will review the minimum depth on a case basis, and, if the soil type changes across the site as at the Sheffield site, on a trench basis based on soil moisture and tensiometer measurements.

As indicated in the draft technical position paper on site suitability and site selection, the term natural resources does include ground water and aquifers underlying the site. The staff agrees that the wording in Section 61.50(a)(8) can be improved as suggested. The intent of the section is to require, through site requirements, a travel time for potential releases at least approximately equal to the time required for groundwater to travel to the site boundary. During the travel distance, decay, sorption, precipitation, and other processes could act to reduce concentrations to within acceptable limits. In addition, the travel distance to the site boundary provides space in the buffer zone for remedial actions, if needed, prior to releases to the surface environment.

Staff agrees with the ground water data needs identified by South Carolina and they are reflected in the BTP (NUREG-0902). The data needs are too prescriptive to include in the rule, however.

Staff agrees with Union Oil that the applicant should not have to perform extensive exploratory evaluations looking for natural resources and that "known" resources would be addressed.

New England Nuclear's suggestions regarding addressing changes in natural background due to construction and addressing ground water intrusion on a probabilistic basis as was done for flooding are interesting points. Changes in ground water background will be a very site specific issue and is judged to be of most importance in the environmental monitoring program. (See Issue D-53-1). Staff did not adopt this suggestion. (See responses to DEIS comments: Commenter 32, Item 12 and Commenter 38, Item 5.) Staff also did not adopt the probabilistic approach for intrusion in favor of retaining the flexibility of an objectives and reasonable assurance approach.

#### Rule Changes:

- Section 61.50(a)(4) should be changed to delete "economically significant," and insert "known". Conforming changes should also be made to §61.12(h) (see Issue B-1).
- 2. Section 61.50(a)(7) should be changed as follows:

The disposal site must provide sufficient depth to the water table that ground water intrusion, perennial or otherwise, into the waste will not occur. The Commission will consider an exception to this requirement to allow disposal below the water table if it can be conclusively shown that disposal site characteristics will result in molecular diffusion being the predominant means of radionuclide movement and the rate of movement will result in the performance objectives of Subpart C of this part being met. In no case will waste disposal be permitted in the zone of fluctuation of the water table.

- 3. Section 61.50(a)(8) should be changed to read as follows:
  - The hydrogeologic unit used for disposal shall not discharge ground water to the surface within the disposal site.

## GISSUE D-50-3

Surface water drainage and flooding

Commenter:

Issue:

• •

۰.

. . . . .

New Mexico Secretary for Health and Environment (4), (45) Sierra Club, Radioactive Waste Campaign (37) Duke Power Company (48) Union Oil Company (66) Georgia Yuan (77) South Carolina Department of Health and Environmental Control (79) Utility Nuclear Waste Management Group (81) Northeast Utilities (85) State of California (93)

U.S. Ecology (101) New England Nuclear (110)

U.S. Department of Energy (119)

Rule Citation: Supplementary Information, V.C.(3) Surface Water 61.50(a)(5), 61.50(a)(6) 

- Summary of Comments: The commenters raised four questions on the siting requirements related to surface water drainage. These comments can be summarized as:
  - (1) definition of terms, specifically upstream drainage areas, costal high-hazard area, and wetland,
  - the state (2) the adequacy of the exclusion of waste disposal based on the 100-year floodplain rather than the 300 or 500-year floodplains, -1 g
  - 1945 Anto (3) whether engineering drainage modifications can be made in order to meet the requirements, and see a set of the set o
  - (4) the vagueness of the terms, such as generally free, minimized, and decrease.

<u>Analysis of Comments</u>: The second sentence in Section 61.50(a)(5) implements Executive Order 11988, Floodplain Management Guidelines. The terms "coastal high-hazard area" and "wetland" are defined in the Executive Order. The term "upstream drainage area" can be defined in conventional hydrologic terms as all the land surface which drains, either by channel flow or sheetwash, across the near-surface disposal facility. Since these definitions either exist already in federal regulations or are standard definitions, the staff does not see the ineed to re-define or reproduce the definitions in 10 CFR Part 61.

The comments on the adequacy of exclusion of waste disposal from the 100-year floodplain fell in two groups, namely those, such as the State of California, who felt the probability of occurrence was too high given the length of radiological hazard and others who misinterpreted the meaning of the 100-year floodplain and assumed it meant one flood of that magnitude was certain within a 100-year period. With respect to the probability of occurrence, the 100-year floodplain is that land which would be inundated by a flood having a 1 in 100 chance of occurring in any particular year. The staff feels the major hazard due to flooding is associated with the period of site operations when disposal units are open. Given that Section 61.51(a)(6) required that the contact of water with wastes must be minimized during and after disposal and Section 61.52(a)(9) requires that disposal units be closed and stabilized as each unit is filled and covered, the disposal units will be open a comparatively short time. Once closed, the covers and site drainage system will provide protection against flooding. The staff considers the 300 or 500-year floodplains to be unnessarily restrictive; and, the staff questions whether an adequate data base or standard methods of determining the 300 or 500 year floodplains exist.

The question on engineering modifications raised by U.S. Ecology and Union Oil will be addressed more fully in technical position papers related to site suitability, selection and characterization and to site design and operations. The requirements on well-drained, free of areas of standing water, and minimal upstream drainage areas relate primarily to the site after construction. However, natural areas of poor drainage or frequent ponding can be indicative of seasonally high ground-water levels. In addition, any engineering drainage modifications must have a lifetime in excess of 150 years and must be consistent with Section 61.44.

With respect to the vagueness or non-prescriptive nature of the requirements, as questioned by Georgia Yuan, the staff recognizes that potential disposal facilities may represent a wide variety of site characteristics. Whereas the staff prefers, for example, sites which are well-drained and not subject to frequent ponding, the staff views the Barnwell site with its shallow bays perched on surficial clay deposits as a suitable site for low-level waste disposal. Basically then the staff anticipated that "non-prescriptive" siting requirements will be site screening tools which will be met in most cases and which, if not met fully, will require a site-specific evaluation. The staff finds this preferrable to treating the "prescriptive" siting requirements as exclusionary.

The State of New Mexico supported the wording of §61.50(a)(6) and suggested editorial changes to the Supplementary Information to be more consistent with §61.50(a)(6) to insert "The potential for flooding should be low". Staff believes this issue should be addressed by clarifying the real concerns raised by potential flooding as indicated in recommended rule change 1 below.

Rule Changes:

1. Wording in Section V.C.(3) Surface Water of the Supplementary Information should be revised as follows if it is used in the future:

Areas which are poorly drained, subject to flooding, or downstream of significant upstream drainage areas should be avoided to reduce the potential for inundation or exhumation by erosion of the disposal units.

- 2. Insert a reference to Executive Order 11988 in §61.50(a)(5).
- 3. No changes are recommended for §61.50(a)(6).

## ISSUE D-50-4

Issue:

.....

Colocation with hazardous waste disposal facilities or other nuclear facilities

Commenters:

Ohio Environmental Protection Agency (38) Bechtel National, Inc. (44) South Carolina Department of Health and Environmental Control (79) Utility Nuclear Waste Management Group (81) American Nuclear Society (87) U.S. Ecology (101) American Society of Mechanical Engineers (107) Colorado Department of Health (111)

<u>Rule Citations</u>: 61.51(a)(7) 61.50(a)(11)

<u>Summary of Comments</u>: Comments on this issue covered two separate rule citations,  $\overline{61.50(a)(11)}$  dealing with masking of the environmental monitoring program and 61.51(a)(7) dealing with using the disposal site for radioactive wastes only.

The South Carolina Department of Health and Environmental Control commented that the requirement on masking of the environmental monitoring program should not be interpreted so strictly as to prevent location adjacent to existing federal facilities. U.S. Ecology suggested an editorial revision.

Most of the other commenters agreed that the requirement on disposal of radioactive wastes only (Section 61.51(a)(7)) was too restrictive and that colocation, without co-mingling, could be advantageous.

The Utility Nuclear Waste Management Group noted that proposed Part 61 would not preclude location of a disposal facility at a reactor site and the ANS noted that location adjacent to another nuclear facility is not prohibited.

Analysis of Comments: The staff agrees with the comments from South Carolina, ANS, and the UNWMG; however, the staff cautions that the environmental monitoring program for a near-surface disposal facility must be able to identify and quantify releases. Should adjacent activities also result in releases, the environmental monitoring program must be able to differentiate the source and quantity of releases. U.S. Ecology objected to the use of "significantly masked" and expressed the view that the monitoring program is either masked or not masked. Only not masked should be acceptable. Staff chose to leave the flexibility for locations near other nuclear facilities but would repeat the caution on identifying and quantifying releases and interpreting results.

With respect to colocation, there appears to be a misinterpretation of the requirement. The requirement indicates that the disposal site, which is subject to NRC licensing, shall be used only for disposal of radioactive wastes. It does not preclude colocating facilities, such as at Sheffield. However, the hazardous waste site must be separate from the licensed radwaste site and potential interactions between the sites must be fully evaluated in terms of the site performance objectives. This issue is also discussed under issue D-52-4.

<u>Rule Change</u>: No wording changes are recommended for Sections 61.50(a)(11) or 61.51(a)(7). Note that 61.51(a)(7) should be relocated in the rule to §61.52 on operations.

#### ISSUE D-50-5

Issue: Tectonics

<u>Commenter</u>: University of California, Los Angeles (8) Georgia Institute of Technology (70) Georgia Yuan (77) U.S. Department of Energy (119)

Rule Citation: 61.50(a)(9)

<u>Summary of Comments</u>: Three of the commenters raised the question of the relevance of seismic or volcanic hazards to low-level waste disposal given the orders of magnitude difference between time frames between those geologic phenomena and the hazard of the low-level wastes. The UCLA comment indicated that Section 61.50(a)(9) may completely eliminate all potential sites in California; yet, it is highly unlikely ground faulting would result in releases given the other siting and waste packaging requirements.

The Department of Energy and Georgia Institute of Technology suggested rewording the requirements as follows: "Active seismic faults or volcanic sites are unsuitable."

<u>Analysis of Comments</u>: The staff deliberated to great lengths the question of whether, given all the minimum technical requirements on site selection and waste packaging, surface faulting would result in significant releases of radionuclides. The staff has also considered indirect effects of faulting, such as creation of barriers or highly transmissive drains for ground water flow.

The staff also gave serious consideration to the UCLA comment that the proposed requirement could possbly eliminate all potential sites in California. That comment ignores the key words in the requirements which are "with such frequency and extent to significantly effect the ability of the disposal site to meet the performance objectives...preclude defensible modeling and prediction of long-term impacts."

Rather than adopt the wording suggested by the Department of Energy and Georgia Institute of Technology, the staff has provided a mechanism for site-specific evaluation of such factors as recurrence intervals, probabilities, liquefaction potential and ground accelerations to compare against a 500-year radiological hazard, dimensionally stable waste container requirements, solidified waste forms, and disposal above the water tables. The minimum technical requirement would not arbitrarily eliminate potential sites so much as (1) provide a site screening test which will be met in most cases and (2) mandate a thorough evaluation of site performance in areas of known tectonic hazards.

Rule Change: No changes are recommended for Section 61.50(a)(9).

#### ISSUE D-50-6

Issue: Demographic requirement

<u>Commenters</u>: Birmingham Audubon Society (80) Atomic Industrial Forum (100) U.S. Ecology (101) New England Nuclear (110)

# <u>Rule Citation</u>: 61.50(a)(3), 61.50(a)(11)

. . .

. • . \*•

<u>Summary of Comments</u>: The Department of Energy and New England Nuclear have expressed reservations about the reliability of projections of population growth for 100 years or more. U.S. Ecology recommended specifying a timeframe for population projections to clarify whether 100 or 500 years is intended. In addition, New England Nuclear has indicated that zoning requirements should be mandated in order to restrict off-site activities which may affect the performance of the disposal facility.

The Birmingham Audubon Society indicates strong approval for the requirement as written.

<u>Analysis of Comments</u>: The comments on the reliability of long-term projections are very interesting, since a significant portion of the staff findings to support approval of a license application will be based in large part upon longterm projections equivalent to the duration of the radiological hazard. The staff recognizes each of these projections, whether demographic, hydrogeologic or other, has a degree of uncertainty. Part of the staff review of any projections focuses on this uncertainty and how it has been handled by the applicant.

In addition, the staff considers the previous experience with commercial lowlevel disposal sites to illustrate that suitable sites can be reasonably found in areas of low population density and minimal population growth potential.

With respect to requiring zoning restrictions on land adjacent to a nearsurface disposal facility, the staff considers this to be unnecessary since site selection and an appropriately-sized buffer zone around the disposal site can provide sufficient separation from near-by activities. This should be especially true when one considers the type of site which meets the entire set of site suitability requirements. The site should have very limited water resources, either surface or subsurface, insignificant mineral resources, a low population density to draw a work force from, and then there should be no existing nearby facilities which could adversily impact the site or the environmental monitoring program.

With respect to specifying a timeframe for projections, the staff considers such a provision to be overly prescriptive for the rule. The projections should address the combination of factors discussed in the preceding paragraphs. Both 100- and 500-year intervals are a part of the disposal scheme but should be addressed in different levels of detail and approach in view of the uncertainties and changing institutional control and intruder protection measures related to each period. <u>Rule Change</u>: No changes are recommended for Sections 61.50(a)(3) or 61.50(a)(11).

# ISSUE D-50-7

11100

Issue: Transportation siting requirement

<u>Commenters</u>: Pennsylvania Department of Environmental Resources (16) Ohio Environmental Protection Agency (38)

Rule Citation: Proposed new requirement for 61.50(a)

<u>Summary of Comments</u>: The Pennsylvania Department of Environmental Resources and the Ohio Environmental Protection Agency have recommended an additional siting requirement on accessibility of near-surface disposal facilities to major transportation routes. The Pennslyvania DER expressed concerned that the largest contributor to population exposure is transportation from the waste generator to the disposal facility. In addition, the Pennsylvania DER recommended a separate section for minimum institutional requirements, such as transportation routes, public water supplies, and population density.

<u>Analysis of Comments</u>: Transportation requirements are addressed in D. O. T. regulations, and given that they are met, the issue of accessibility of major transportation routes becomes primarily an economic consideration to be considered in site selection and the evaluation of alternatives required under NEPA. The other two:"institutional" requirements recommended by Pennsylvania are already present in Sections 61.50(a)(3) and 61.50(a)(4).

Rule Change: No changes are recommended.

## ISSUE D-50-8

Issue: Properties of site soils

<u>Commenters</u>: South Carolina Department of Health and Environmental Control (79) Conference of Radiation Control Program Directors (103) U.S. Environmental Protection Agency (122)

Rule Citation: Proposed new requirement for 61.50(a)

<u>Summary of Comments</u>: Each commenter has proposed additional siting requirements related to characteristics of the soils. South Carolina Department of Health and Environmental Control has proposed requirements that the mechanical and physical properties of the soils be compatible with certain uses required primarily in Section 61.51(a), e.g., suitable for compaction. The Conference of Radiation Control Programs Directors have recommended exclusion in areas of high natural radioactivity. The Environmental Protection Agency recommended a requirement that the site soils be permeable enough that water infiltrating into a trench can drain through the trench bottom rather than accumulate.

<u>Analysis of Comments</u>: The staff feels each of these suggested requirements is unnecessary. The requirements in Section 61.51(a) which South Carolina addresses will be met, but not necessarily with on-site soils. Off-site soils may be trucked in, as at Sheffield, or engineered features may be used. Requirements such as being capable of supporting the construction equipment or being amenable to the surface water drainage are implicit if stabilization is to be accomplished.

ាក់ខេត្ត សម្ត្រ។

With respect to areas of high natural radioactivity, these areas would be excluded if they could be shown to violate Section 61.50(a)(11). Otherwise, the staff sees no valid reason for excluding these areas.

The staff agrees with the concept of the EPA comment and has included a recommendation in the technical position paper on site suitability, selection, and characterization (NUREG-0902, p. 8) that the bottom of the disposal unit should drain at least as readily as water can infiltrate into the disposal unit. However, the staff feels that this can be accomplished by site characteristics, design features, or most frequently a combination of both. Therefore, the staff has given the applicant flexibility while requiring in 61.51(a)(6) that the contact of percolating or standing water with wastes after disposal must be minimized to the extent practicable.

Rule Change: No changes are recommended.

## ISSUE D-50-9

Issue:

Time spans for siting and design requirements

<u>Commenters</u>: Advisory Committee on Reactor Safeguards (10) Bechtel National, Inc. (44) South Dakota State Planning Bureau (69) Birmingham Audubon Society (80) American Nuclear Society (87)

<u>Rule Citations</u>: Supplementary Information, V., C., (5) Stability; 61.12(d); 61.50; and 61.51

<u>Summary of Comments</u>: The Advisory Committee on Reactor Safeguards has raised the general question of how long do the various siting or design requirements have to be satisfied. This comment was repeated in various terms by Bechtel National, the American Nuclear Society, and the Atomic Industrial Forum. Bechtel National and the American Nuclear Society requested that the design basis natural events or phenomena be identified and that the length of hazard associated with these be stated.

The South Dakota State Planning Bureau offered the suggestion that the stability requirement be specified as 100 years; whereas the Birmingham Audubon Society commented that waste stability should be required for the 500-year duration of the radiological hazard.

<u>Analysis of Comments</u>: As indicated in Column 1 on page 38084 of the <u>Federal</u> <u>Register</u> notice, the siting, design, and waste package requirements relate to both stability of the disposal site and control of releases within acceptable limits. Over the time frame of the radiological hazard, reliance must be placed primarily on the site since the waste package and design features will decrease in effectiveness. Therefore, each of the siting requirements in Section 61.50(a) should be considered applicable over the indefinite future and should be evaluated for at least a 500-year time frame. The time requirements on design and waste packages, however, should be applied to a shorter period. Given any necessary passive maintenance to maintain the design features in an effective manner, the staff would anticipate a gradual decrease in effectiveness of the design features and waste package such that they will continue to provide a significant though decreasing contribution to isolation for approximately 300 years. This should be particularly true because the design features will have been observed, repaired and modified, if necessary, to assure a stable disposal site prior to the end of the institutional control period.

Bechtel National and the American Nuclear Society request that the time frame for which design bases natural events or phenomena must be considered be stated. This comment raises much the same question as raised by the ACRS on time spans for siting requirements. The staff believes that the time frame for considering the design bases natural events or phenomena is equivalent to the minimum time of concern for siting requirements (500 years). The design life of the particular design feature would thus be evaluated over the period (300 years) when the feature is significantly contributing to isolation and a follow on period of contribution of less significance. Additional guidance will be provided in a BTP on site design.

Rule Changes: No changes are recommended.

()

••

# ISSUE D-51-1

Site design Issue: . Chem-Nuclear Systems, Inc. (41) Commenters: Bechtel National, Inc. (44) Arizona State Clearinghouse (47) Duke Power Co. (48) American Nuclear Society (87) Peter Skinner for Attorney General, State of New York (99) Atomic Industrial Forum (100) U.S. Ecology (101) American Society of Mechanical Engineers (107) Texas Department of Health (117) e 1. ( U.S. Department of Energy (119)

# Rule Citation: 61.51(a)(3), 61.51(a)(4), 61.51(a)(6), 61.62(a)

<u>Summary of Comments</u>: Five of the commenters objected to the absoluteness of the requirements in 61.51(a)(4) and 61.51(a)(6) and suggested replacing the words "prevent" and "eliminate" with "minimize." Several indicated the words "and improve" should be removed from 61.51(a)(3) or similar concerns about mandatory improvements. Several commenters questioned the use of "eliminate" when referring to erosion in §61.51(a)(5) and active maintenance in §61.62(a).

The Texas Department of Health requested preferential consideration be given to progressive slope design for burial; and Peter Skinner indicated concern that the rule does not provide specific guidance for engineered features which would deal with leachate generation and that site areas used for disposal of Class A waste will require more maintenance.

<u>Analysis of Comments</u>: Sections 61.51(a)(3), 61.51(a)(4), and 61.51(a)(6) are design objectives. Given these are design objectives, the actual achievement will be to minimize, rather than absolutely prevent or eliminate. The staff anticipated that at most sites, design features can be used to improve natural site characteristics, particularly with respect to surface water drainage and limiting infiltration into the disposal unit. Additional guidance will be provided in the technical position paper on site design and operations. Staff agrees that improvements may not be mandatory and inserted "where appropriate" in §61.51(a)(3).

The commenters on §§61.51(a)(5) and 61.62(a) are referred to the definition of "Active maintenance" in Section 61.2. It is the staff's intent that the need for active maintenance be eliminated and staff believes that this goal as reflected in the definition can be achieved.

With respect to progressive slope design for burial, the regulation does not specify the type of disposal unit. Alternative types of disposal units have been evaluated in NUREG's CR-0308 and CR-0680. Additional guidance will be provided in the technical position paper on site design and in an NRC-sponsored symposium in September, 1982 on the same subjects. The staff will provide more specific guidance on avoiding leachate generation and addressing the inherent instability of Class A waste in the technical position paper on site design and operations. (See the discussion of stability under Issue D-52-2 also.) The site designer should give particular attention to the design of that portion of the site used for disposal of Class A wastes and use innovative designs to provide long-term stability.

#### Rule Changes:

- 1. In Sections 61.51(a)(4) and 61.51(a)(s6), change "prevent" and "eliminate"
   to "minimize." (See Issue GEN-1 also.)
- 2. In Section 61.51(a)(3), add "where appropriate" after "and improve."

# ISSUE D-51-2

Issue: Design life of markers and monuments

<u>Commenters</u> :	Union of Concerned Scientists (36)
	State of New Mexico (45)
	State of California (93)
	U.S. Department of Energy (119)

# <u>Rule Citation</u>: 61.52(a)(7), (9)

<u>Summary of Comments</u>: The Union of Concerned Scientists recommends that "warning signs" with a 500-year design life be employed as a deterrent to inadvertent intrusion. The U.S. Department of Energy also recommends requiring permanent monuments for the site.

The States of New Mexico and California request that there be a provision for a "permanent" identification monument with a design life of 500 years. They recommend this be high enough to be visible above the contour of the disposal site.

<u>Analysis of Comments</u>: There are few "signs" in the traditional sense that have design lives of anything approaching 500 years. The staff would consider granite monuments, near the survey marker control points required in 61.52(a)(7), which have the radioactivity symbol and descriptive data engraved on them as permanent markers. Staff also notes that the later the monuments are installed, the longer they should last during the passive control period. Thus installation after the licensed period of active institutional control may be appropriate and this flexibility was provided by the suggested change.

<u>Rule Change</u>: Addition to subsection 61.31(c)(2): "permanent monuments or markers warning against intrusion have been installed."

## ISSUE D-51-3

<u>Issue</u> :	Alternatives for Class A or Classes A & B Waste
Commenter:	Ohio Environmental Protection Agency (38)
Rule Citation:	§§61.50, 61.51, and 61.52

Summary of Comment: The commenter suggests that the NRC consider less stringent siting, design, and operational requirements for a facility designed to contain Class A or Classes A and B wastes than one containing Class C wastes. The commenter indicated that the present rules are designed to ensure containment of Class C wastes.

Analysis of Comment: The proposed rule assigned lesser operational requirements to the disposal of Class A wastes. However, the stability and design requirements for Class B and C wastes have the same basis, with additional intruder protection added for Class C wastes. The staff does not agree that the Class B requirements can be lowered to those for Class A wastes.

The commenter is correct that the siting requirements apply equally to all classes of waste and staff believes that this position continues to be reasonable in view of the minimum nature of the requirements. Section 61.54 provides flexibility to approve alternative design and operational requirements and would be the vehicle for considering proposals such as less restrictive measures for disposal of Class A waste only.

Rule Change: No changes are recommended. 

## ISSUE D-51-4

Intruder barrier engineering Issue: PA Department of Environmental Resources (16) Joseph H. White III (21) Ohio EPA (38) Commenter: Chem-Nuclear Systems, Inc. (41) University of Arizona (78)

U.S. Department of Energy (119)

# Rule Citation: Section 61.7(b)(5)

Summary of Comments: The University of Arizona and Department of Energy point out the usages of the term "engineered barrier" in 10 CFR Parts 60 and 61 are inconsistent. Ohio and Pennsylvania noted that "engineered barriers" may also be used to describe features that limit or control water movement. Chem-Nuclear Systems Inc. indicated their support for the use of "engineered barriers" as an alternative to deeper disposal for Class C waste. White questioned what types of barriers may be used. . . : <sup>1</sup> 1.11

12530

. .

Analysis of Comments: The usage in 10 CFR Part 61 for the term "engineered barrier" is inconsistent with 10 CFR Part 60. The usage in 10 CFR Part 61 should be changed to "intruder barrier." 113 0 1

Options other than depth involve use of natural or man-made barriers having an estimated protection lifetime of at least 500 years. Options might include multiple layers of soil, clay, gravel, and boulders or caissons capped with concrete. Additional guidance on such options will be addressed in the Branch Technical Position on design and operation being prepared.

Rule Change: Change "engineered barrier" to "intruder barrier" in 61.7(b)(5), in 61.52(a)(3). · . . . 

## ISSUE D-52-1

Issue:

#### Segregation of wastes

Commenters:

Joseph H. White III (21) Alabama Power (33) Union Carbide Corporation (39) South Carolina Department of Health and Environmental Control (79) State of New York (99) New England Nuclear (110) Utility Nuclear Waste Management Group (81) State of Washington (112) U.S. Department of Energy (119)

## Rule Citation: § 61.52(a)(1).

<u>Summary of Comments</u>: Alabama Power is concerned that the waste classification requirements should not prohibit mixed classes of waste being shipped on one transport vehicle. For example, if a container of Class B or Class C waste needs to be transported by a shipping cask the commenter felt that, any remaining space in the cask should be allowed to be filled by available waste containers, including Class A waste containers. Otherwise, transport costs would be needlessly raised. The commenter believes that waste containers should be segregated at the disposal site rather than segregated by the generator into different transport vehicles.

Union Carbide stated that Class A material would form a good shielding buffer from the more radioactive Class B material. By segregating the waste classes, burial trenches for Class B may be filled with dirt simply to provide shielding which could have been provided by the Class A material, resulting in better land use.

The State of New York, New England Nuclear, the Utility Nuclear Waste Management Group, and DOE questioned the meaning and intent of the term "interaction" in the requirement in § 61.52(a)(1). The Utility Nuclear Waste Management Group suggested defining sufficient separation as no interactions that can result in premature failure of the disposal facility. The State of Washington questioned the need to segregate by waste class at arid sites, indicating that comingling should be allowed. White questioned how wastes would be segregated.

The South Carolina Department of Health and Environmental Control recognized waste segregation as a viable option to further assure meeting the performance objectives over the long-term; however, waste segregation has several short-term drawbacks. These include increased operational exposures due to the absence 'of shielding provided by the Class A wastes, a need for at least two trenches open simultaneously and a resulting need for additional handling equipment, and higher radiation levels at the trench boundaries with a resultant significant increase in occupational exposure.

<u>Analysis of Comments</u>: The intent of the rule is not to prohibit waste from more than one class from being shipped on the same transport vehicle.

Consistant with transportation requirements, the staff has no objection to comingling different classes of waste in transport.

In response to Union Carbide's comment, the staff notes that the Class A wastes are segregated to provide better long-term stability of the disposal site because Class A waste, such as ordinary trash-type wastes, is less stable. If mixed with higher activity Class B wastes, the deterioration of Class A wastes could lead to failure of the cover system and permit water to penetrate the disposal unit, resulting in potential migration of the higher activity wastes. Therefore, the long-term potential for waste migration is considered a more important factor than the optimization of short-term operational parameters. This same reasoning is applicable to the comments raised by the South Carolina Department of Health and Environmental Control. The operational exposures can be directly monitored and controlled such that applicable standards are not exceeded. In addition, recent advances have been made in the technology to remotely place wastes with a variety of lifting devices, thereby providing some mitigation of potential occupational exposures.

In identifying the need to clarify the term "interaction," the commenters noted that it was vague and unenforceable. The intent of the provision was to protect Class B and C wastes from the less stable Class A wastes. Class A waste as proposed may contain absorbed liquids, dewatered resins, and biological materials, for example, that may release solvents, water, known chelates, or other mobile components. These mobile components may interact with other wastes in the disposal unit. If the other wastes are Class A and of lesser hazard, the consequences of the interaction are minimized. In addition, Class A wastes are less stable and more prone to degradation, will consolidate and provide a less stable support for disposal unit covers, leading to increased infiltration of precipitation and increased potential for surface water intrusion through voids in the covers. When the additional moisture from any of these sources leaches the waste, some of the solvents and organics are mobilized along with soluble radionuclides. The amounts and range of these mobilized materials will be very site specific and depend on factors such as soil types, hydraulic gradients, precipitation, and cover designs. The language in the rule was modified to reflect the desired protection of Class B and C wastes and to reference meeting the performance objectives instead of the absolute "no interaction." In addition, the rule was modified to permit disposal of stable Class A waste with Class B and C wastes.

The State of Washington regulates the arid disposal site located near Richland, Washington. The State noted that ground water or surface water are not significant factors at arid sites, and, segregation of Class A wastes seems to be unnecessary when weighed against the burden of operating separate disposal units. The State noted that co-mingling of Class A and B wastes would dilute the Class B wastes and have potential benefit. The State's observations may have merit for arid sites, but they are difficult to adopt in a rule that must address sites located in all parts of the country. However, the Commission anticipated the need to consider alternative disposal requirements and included § 61.54, "Alternative requirements for design and operations" to provide for consideration of such alternatives, without granting exceptions to the rule, provided the performance objectives of Subpart C are met. State requirements could certainly contain similar provisions for alternatives and remain compatible with the rule.

## Rule Changes:

 The concern raised by Alabama Power has been addressed by changing the wording of § 61.52(a)(1) to more clearly indicate the intent as follows:

"...placing in disposal units which are sufficiently separated from disposal units for the other waste classes..."

2. The concern about the meaning of the term "interaction" has been addressed by relating the requirement to the performance objectives as follows:

> "...so that any interaction between Class A wastes and other wastes will not result in the failure to meet the performance objectives in Subpart C of this Part."

3. The concern about comingling waste classes has been partially addresses by adding the following statement to § 61.52(a)(1):

"This segregation is not necessary for Class A wastes if they meet the stability requirements in § 61.56(b) of this Part."

## ISSUE D-52-2

Issue: Factors in Stability of the Site

Commenters:

Advisory Committee on Reactor Safeguards (10) Nevada, Department of Human Resources (14) University of North Carolina (30) Sierra Club (37) Duke Power Company (48) South Carolina Department of Health and Environmental Control (79) Utility Nuclear Waste Management Group (81) State of New York (99) Atomic Industrial Forum (100) U. S. Ecology (101) . . . . Conference of Radiation Control Program Directors (103) American Society of Mechanical Engineers (107) State of North Carolina (109) New England Nuclear (110) State of Washington (112) Department of Energy (119) Environmental Protection Agency (122)

<u>Rule Citation:</u> § 61.52(a)(4) and (5)

<u>Summary of Comments</u>: The ACRS requested that Section 61.52 be reevaluated since subsidence is the result of many factors, including primarily the manner of placement of waste packages in the trenches. DOE questioned why have these requirements, which increase costs, if the overall performance objectives are met; and, the Utility Nuclear Waste Management Group felt the specific stability static requirements were unnecessarily prescriptive as to the specific method to minimize subsidence. Numerous commentors (Sierra Club, Duke Power Company, Utility Nuclear Waste Management Group, Atomic Industrial Forum, U.S. Ecology, American Society of Mechanical Engineers, and New England Nuclear) objected to placing wastes in an "orderly" manner. Several indicated their comment was based on increased costs and/or occupational exposures due to handling the waste packages to meet the requirement. Several asked for an explanation of a specific method; and, several indicated stacking may not benefit long-term package or trench integrity due to corrosion and/or decay.

The Conference of Radiation Control Program Directors indicated support for § 61.52(a)(5). Duke Power Company requested an editorial change; and, the University of North Carolina indicated filling void spaces with earth material will not be successful unless there is mechanical compaction of the backfill. The South Carolina Department of Health and Environmental Control requested that the staff specify what material, other than earth materials, are suitable for use as backfill. The State of Nevada Department of Human Resources indicated that subsidence may be delayed 50-100 years at arid sites; and, the State of Washington recommended that more emphasis should be placed on avoiding slumping and wind erosion at arid sites. EPA indicated that gases can build up sufficient pressures over a period greater than 50 years to affect the stability of trench covers, thus requiring active maintenance and repair of breaches.

1.1.1.1.1.1

The State of New York and EPA questioned the long-term stability of Class A disposal units and indicated that, even if the overall performance objective on releases was met, the Class A disposal units will pose a significant nuisance condition. The State of New York indicated that the appearance of Class A disposal units will lead to public and local government steps to close the sites and will undo the favorable treatment of Class B and C wastes. Further, the EPA raised the question of releases of both radiological and longer-term non-radiological contaminants, while the State of New York indicated that decay during the period of institutional control cannot be relied upon to reduce the hazard of Class A wastes if there is migration during that period. The State of New York and the Sierra Club recommend the same requirements for Class A wastes as for Class B and C wastes.

4 64 Analysis of Comments: The ACRS correctly points out that subsidence is the result of many factors, including primarily the manner of placment of the waste packages in the trenches. The staff addressed these lessons learned by proposing requirements on many of these factors, including placement of the waste package so as to maintain package integrity during placement, to minimize void spaces between packages, and to permit filling of the void spaces (§ 61.52(a)(5)). Although the Department of Energy suggested that meeting the overall performance objective of stability is sufficient, the staff has taken the approach throughout the regulation that the best way to assure the entire system (site characteristics, design, operations, waste classification, waste form, etc.) will meet the overall performance objectives is to place requirements on each of the components of the system. The staff has attempted to provide flexibility on how the specific requirements will be met; and, this is reflected in the rewording of § 61.52(a)(4). 一, "我们不是?" . •

In addidition, the staff will provide guidance on preferred methods of meeting the specific requirements in a branch technical position paper on design and operations. The BTP will stress that, while no one method can prevent subsidence, the combined effect of the total system should be a dramatic reduction in the potential for catastrophic or significant failure of trench caps as observed at several closed low-level waste disposal sites. Also, Section 61.12(b) requires, in the application, specific information which relates to subsidence. Section 61.13(d) requires specific technical information related to long-term trench stability.

. . . .

1

The staff agrees that there are alternatives, such as backfilling individual layers of waste packages after placement or grouting between the wastes, to orderly placement of the waste packages. Likewise, there are alternatives, such as the use of granular backfill between individual layers of waste packages, to mechanical compaction of the backfill. The staff will address means to avoid the longer term slumping or subsidence at arid sites in the BTP on design and operations. In addition, technical analyses pursuant to Section 61.12(b) and 61.13(d) should address Nevada's and Washington's concerns in the application phase. Experience has shown that short-term (5-10 year) subsidence in humid sites is more dramatic and requires a greater amounts of maintenance than subsidence at arid sites. Long-term subsidence at either type of site will likely be gradual and can be dealt with through minor maintenance activities by the custodian.

With respect to the generation of gases raised by EPA, several specific requirements such as on trench covers and waste characteristics should reduce the rate of gas generation such that it may diffuse through the soil or trench cap without pressure buildup. In addition, sumps constructed for drainage while trenches are open and used for monitoring and/or remedial pumping after trench filling may provide sufficient venting to the atmosphere, especially if granular backfill is used. The effects of trench gas, like the effects of gradual deterioration of waste packages, are addressed by the requirements for continued maintenance during the post-closure observation and institution control periods.

The concerns expressed by the State of New York and EPA about the long-term stability and appearance of Class A disposed units is well-taken. The staff would point out that the siting and design requirements (§ 61.50 and 61.51) are applicable to all waste classes. In addition, the rule has been revised so that all waste classes must be disposed of in accordance with § 61.52(a)(4) through (11). Further guidance on Class A disposal units will be provided in the BTP on design and operations and in an NRC-funded symposium on design and operations in September 1982. Innovative thinking and techniques are needed for disposal of segregated Class A wastes; and, the BTP and symposium should provide an appropriate forum.

## Rule Change:

1. Revise § 61.52(a)(3) as follows:

"All wastes shall be disposed of..."

2. Revise § 61.52(a)(4) to eliminate "orderly" placement and to stress the objectives rather than prescribe methods. The revised section should read:

"Wastes must be emplaced in a manner that maintains the package integrity during placement, minimizes the void spaces between packages, and permits the void spaces to be filled.

## ISSUE D-52-3

÷.

Issue:

2

Basis for 100 foot buffer zone

**Commenters:** Joseph H. White III (21) Law Engineering Testing Co. (34) Paul F. Hadala and Don C. Banks (76) Birmingham Audubon Society (80) New England Nuclear (110) U.S. Department of the Interior (114) Tennessee Valley Authority (116)

## Rule Citation: § 61.52(a)(8), 61.2

Summary of Comments: All commenters generally supported the concept and purposes of a buffer zone; however, there was disagreement on whether a specific distance should be required. White asserted that 100 feet is too small. Law Engineering and TVA questioned the basis for the 100 foot buffer zone and stated that the buffer zone should be based on site performance objectives. Hadala and Banks and the Audubon Society suggested that the minimum buffer zone size be increased to at least 300 feet. The Department of the Interior suggested a threedimensional zone based on site performance; and, New England Nuclear suggested that the buffer zone extend farther in the direction of ground-water migration. , f.

Analysis of Comments: The proposed prescriptive requirement of a minimum buffer zone of 100 feet in § 61.52(a)(8) was arbitrarily selected. The intent was to provide adequate space for monitoring or remedial action and adequate physical separation from off-site activities. The intent was to evaluate the needed size on a site-specific basis, emphasizing that 100 feet was an absolute minimum. Certainly the distance would vary both from site to site and directionally at a given site. Greater distances were anticipated in the direction of groundis water flow where contingency actions might be required or in directions needed for surface water management or erosion control measures. [In addition,] discussions with the Corps of Engineers indicated that 100 feet may not be sufficient for purposes of remedial action. Therefore, the prescriptive 100 feet was dropped and the purpose of the buffer zone was expressed. The buffer zone must be adequate to meet the performance objectives. In addition, the Department of Interior comment that the buffer zone include depth as well; as lateral boundaries was adopted in the definition. Unrestricted use of land and resources beyond the three-dimensional buffer zone is possible during and after site operation, thereby, reducing the impacts of the disposal site. . . .

White also questioned what mitigative measures may take place in the buffer zone as discussed in § 61.7(a)(2). The possible measures are site/situation specific and the staff felt that speculation in § 61.7(a)(2) concerning what specific measures may be employed was inappropriate. ..... ...

. .

. 2111

. . . · ·

## Rule Changes:

Amend § 61.52(a)(8) to read: "A buffer zone of land must be maintained 1. between any buried waste and the disposal site boundary and beneath the disposed waste. The buffer zone shall be of adequate dimension to carry out environmental monitoring activities specified in § 61.53(d) of this Part and take mitigative measures if needed."
- 2. Add to § 61.12(b) "and adequacy of the size of the buffer zone for monitoring and potential mitigative measures."
- 3. Revised definition of buffer zone in § 61.2 as follows:

"...licensee and that lies under the disposal units and between the disposal units and the boundary of the site."

ISSUE D-52-4

Issue: Ancillary activities at the site

<u>Commenters</u>: Duke Power Company (48) South Carolina Department of Health and Environmental Control (79) Birmingham Audubon Society (80) Utility Nuclear Waste Management Group (81) Middle South Services (84) The American Society of Mechanical Engineers (113) Tennessee Valley Authority (116)

Rule Citations: § 61.12, 61.13, 61.51(a)(7); 61.2

<u>Summary of comments</u>: South Carolina and ASME addressed the need for on site surge storage provisions and consideration of the radiological impacts of satellite activities, such as trucking terminals. The Audubon Society offered strong support of § 61.51(a)(7), and the UNWMG, Middle South Services, TVA, and others suggested clarification of this provision on exclusive use for disposal of radioactive wastes in order to clearly state that the provision addresses waste types and not ancillary activities.

<u>Analysis of Comments</u>: South Carolina suggested adding a provision to § 61.12(f) to require contingency plans for surge storage of wastes because natural phenomena and unplanned events might interrupt operations and delay disposal. The ASME made a similar comment for the definition of disposal facility in § 61.2. In such cases, storage would be required so that trucks could be off loaded and released. Existing site operators generally have provisions in their licenses for storage of up to 6 months, although prompt disposal is encouraged and is the usual practice. Paragraph 61.12(f) requires a description of "methods and area of waste storage" but makes no distinction between routine and contingency storage plans.

South Carolina also suggested adding evaluation of the radiological impacts of ancillary or satellite activities such as trucking terminals to the required analyses in § 61.13. The State noted that such ancillary activities could have significant radiological impacts. Ancillary activities, such as incinerators, trucking terminals, or supply services, which might be located at the disposal facility or in close proximity to the disposal site, would be licensed under 10 CFR Parts 30, 40, 70, et al., as appropriate. The performance objectives in Part 61 would not necessarily apply since Part 61 deals with disposal of wastes. The impacts of the ancillary activities, both radiological and nonradiological, would be addressed in the site specific EIS to the extent they are known or anticipated. Paragraph 61.51(a)(7) states "The disposal site shall be used exclusively for the disposal of radioactive wastes." The wording resulted in confusion as to whether other waste types or other activities were prohibited. The confusion centers on the term "disposal site," which is defined in § 61.2 to mean "that portion of a land disposal facility which is used for disposal of waste." The disposal site consists of disposal units and a buffer zone; therefore, the requirements of § 61.51(a)(7) apply only to the disposal units and the buffer zone.

The intent was to prevent co-mingling of radioactive wastes with other types of wastes such as chemical or hazardous wastes. There was no intent to prohibit ancillary activities, such as incineration or other waste treatment, at the disposal facility provided they are not located within the perimeter of the buffer zone. In addition, there was no intent to prevent co-location of other facilities outside the buffer zone. Thus, a hazardous waste disposal site could be developed at the same location and use common administrative facilities as is currently done at the Beatty, Nevada site. The disposal sites for the two types of wastes must be separate and the wastes must be properly segregated. Concerns such as masking the environmental monitoring data so that site performance cannot be monitored must be addressed in such plans in accordance with § 61.50(a)(11). The provision to describe such ancillary activities in § 61.11(c)(4) was intended to flag these activities so their impacts could be assessed.

<u>Rule Changes</u>: Section 61.51(a)(7) was revised and moved to § 61.52(a)(11). Section 61.52(a)(11) restricts disposal to only radioactive wastes.

# ISSUE D-52-5

•

•.

<u>Issue</u> :	Operational requirements on direct gamma levels and miscellaneous comments on operation and closure
<u>Commenters</u> :	Joseph White (21) Union of Concerned Scientists (36) Union Carbide Corp (39) State of New Mexico (45)
·	S.C. Department of Health and Environmental Control (79) Birmingham Audubon Society (80) Northeast Utilities (85)
· · · · · · · · · · · · · · · · · · ·	State of California (93) Atomic Industrial Forum (100) U.S. Fcology (101)
•	The American Society of Mechanical Engineers (107) New England Nuclear (110) Department of Energy (119)

U.S. Environmental Protection Agency (122)

<u>Rule Citations</u>: § 61.52(a)(2), (a)(3), (a)(6), (a)(7), (a)(9)

Summary of Comments:

. . .

.

.....

Direct gamma levels: Proposed § 61.52(a)(6) required that disposal take place in a manner that limited gamma radiation at the surface to levels that are

at the

.

. •

within a few percent above background. White questions the meaning of a few percent above background Union of Concerned Scientists; States of New Mexico, South Carolina and California; New England Nuclear and DOE suggest gamma radiation levels should be some specific level above background. The Birmingham Audubon Society suggested 1%. The DOE suggested 10% or 1 mrem/hour. Union Carbide suggests limiting gamma radiation to limits established in 10 CFR Part 20. New England Nuclear suggested use of "exposure" or "dose rates" to include other types of radiation. EPA indicated a few percent was vague and didn't taken into account the variability of background radiation levels at any particular site.

<u>Miscellaneous Comments</u>: South Carolina suggested adding a requirement in § 61.52(a)(9) that erosion control measures be carried out. The American Society of Mechanical Engineers (ASME) recommended defining "adequate" in this paragraph.

Northeast Utilities and New England Nuclear (NEN) questioned whether the disposal unit cover referenced in § 61.52(a)(3) includes an impervious cap. NEN also questioned whether the 5 meters refers to the distance from the top or bottom of the cover.

U.S. Ecology stated that the requirement in § 61.52(a)(7) to accurately locate each disposal unit is reasonable if an allowed tolerance is indicated. The ASME and AIF recommended clarifying "accurately located" in this paragraph.

<u>Analysis of direct gamma level comments</u>: The staff agrees that the requirement should be reexamined. The staff considers the permissible levels of radiation in unrestricted areas, as specified in § 20.105 of 10 CFR Part 20, to be appropriate for application at the time of transfer of the disposal facility license to the site owner for the period of institutional control. Although access to the site is restricted during this period of institutional control, the persons working at the site should not be considered radiation workers; therefore, the higher occupational exposures for radiation workers are not appropriate. When ALARA considerations are applied to § 20.105, the levels will probably not be significantly above a few percent of background. The applicant or licensee is allowed greater flexibility and more specific guidance on an acceptable upper limit by the reference to §20.105.

<u>Analysis of miscellaneous comments</u>: Paragraph 61.52(a)(9) requires closure and stabilization measures to be carried out as disposal units are filled and covered. South Carolina suggested specific reference to erosion control measures which are an important part of closure and stabilization. The staff does not feel specific reference to erosion is needed; however, erosion control would be included in the site closure plan which must be submitted for approval as part of the application for approval and will be periodically reassessed during operations of the disposal facility. "Adequate" was replaced with a reference to the approved closure plan to clarify the requirement.

Paragraph 61.52(a)(2) addresses the intruder barrier and waste emplacement for Class C wastes. The Northeast Utilities recommended clarifying the term "cover" in § 61.52(a)(2) to indicate whether the cover includes an impervious cap. New England Nuclear raised the same point. The requirement in § 61.52(a)(3) was addressing the intruder barrier depth only, not materials or design. Design requirements for covers are provided in § 61.51(a)(4). New England Nuclear also recommended clarifying the 5-meter distance.' The 5-meter distance is intended to be from the accessible surface to the top of the wastes, i.e., the thickness of material between a potential intruder and the wastes. The rule was clarified to so indicate.

The suggestion to add a tolerance limit to paragraph 61.52(a)(7) was not adopted. All measures, surveys, etc. by nature have some range of error. In keeping with an objective approach, a more prescriptive requirement was not adopted. Guidance on surveying accuracy, including both elevation and location, will be provided in the technical position paper on design and operations.

### Rule Changes:

1 12

1. § 61.52(a)(6) replace "gamma radiation" by "radiation dose rates" and replace the reference to a few percent of background with a requirement to comply with § 20.105.

· . . .

2. Amend § 61.52(a)(9) to read: "Closure and stabilization measures as set forth in the approved closure plan must be carried out as each disposal unit (e.g., each trench) is filled and covered."

3. § 61.52(a)(2): insert "top" before "surface of the cover."

## ISSUE D-52-6

Issue: Closure plan

Commenters:

Environmental Law Project (9) Birmingham Audubon Society (80) The American Society of Mechanical Engineers (113) General Research Corporation (123)

<u>Rule Citations</u>: Paragraph 61.12(g) requires a conceptual closure plan with the license application and § 61.28 requires a final revision of the closure plan. Related citation: § 61.52(a)(9).

<u>Summary of Comments</u>: The Environmental Law Project seems to be requesting a detailed closure plan with the license application which can be periodically amended and updated. The Audubon Society suggested a requirement for alteration of the site or disposed waste before closure if necessary to protect the public health and safety. The ASME suggested making the rule sufficiently prescriptive to eliminate the need for a closure plan and adding procedures to assure that the closure is addressed and settled up front when the license is issued. General Research Corporation expressed concern that Part 61 may not establish adequate requirements for the specific content of the closure plan and for revision during operations.

<u>Analysis</u>: An applicant is required to submit a closure plan with the license application. Parts of the plan will be conceptual rather than detailed. However, staff agrees that it would be appropriate to periodically "flesh out" the conceptual portions of the plan as more information about the site becomes available during site operations and to reflect changing technology. The annual review of the adequacy of the funding for closure required by § 61.62(c) will be keyed to any changes and updates in the activities to be funded after operations cease and the status of measures completed as units are filled. An important concept and requirement related to concern about updating or revising the plan is the requirement to carry out closure and stabilization measures as each disposal unit is filled in paragraph (a)(9) of § 61.52. Closure is actually a two phased activity. Much of the work will be done as the units are filled and closed. This work can be defined and described in detail in the application. The activities which must be delayed or are best delayed until after operations cease and all units filled and covered will be more conceptual in nature and will need to be updated. However, the delayed activities must be sufficiently defined and described to estimate costs in order to meet the requirements of Subpart E. As a minimum, the plan and its basis will be updated at license renewal and as part of approving closure pursuant to § 61.28.

The provisions of § 61.28 are adequate, without any changes, to give the NRC authority to require any changes in the site or wastes at closure that are needed to protect the public health and safety. Thus, no change was adopted in response to the Audubon comment. The General Research Corporation raised several topics that might be addressed in the rule such as how can activities during long-term care be factored in and what happens if the closure plan is not approved. While the points are valid questions, they are potentially very site and circumstance specific. Additional guidance on the closure plan will be considered as guidance for the format and content of the application is developed. The Corporation notes that the rule does not state when the closure plan must be revised, although several opportunities are provided when the plan may be reviewed. The staff considers this flexibility appropriate.

Change in Rule: None

## ISSUE D-53-1

Environmental monitoring Issue: 1 1. · · · Commenters: Commonwealth of Pensylvania, Department of Environmental Resources (16) Joseph H. White III (21) Union of Concerned Scientists (36) Bechtel National, Inc. (44) Union Oil Company of California (66) Argonne National Laboratory (68), (121) American Nuclear Society (87) Attorney General State of New York (99) The American Society of Mechanical Engineers (107) North Carolina Radiation Protection Commission (109) New England Nuclear (110) . . . . U.S. Department of the Interior (114)

Rule Citation: § 61.13 and 61.53

.

<u>Summary of Comments</u>: One commenter noted with respect to Subsection 61.13 that analyses of release pathways should be conducted so that they may be validated by data acquired from subsequent monitoring. Monitoring should be conducted with this end in view and validation should be required at set periods. (36)

5 M - 1 - 2

The other comments received all relate to § 61.53. One commenter noted that the requirement in § 61.53(a) for monitoring data covering a simple twelvemonth period for seasonally variable characteristics is insufficient. Seasonal variations having an impact on future environmental monitoring would better be related to established normals, maximums and minimums as provided by appropriate agencies (16). The State of North Carolina (109) expressed the view that the 12 months begin when the application is filed and continue during the review period.

Another commenter noted that § 61.53(a) should include among preoperational monitoring programs the areas of land use, local population density and predicted future movements of population. (21)

· · · / . · /

With respect to § 61.53(c) two commenters suggested that the last sentence of that section be rephrased to read: "The monitoring system must be capable of providing early warning of migration of radionuclides <u>from the disposal units</u> <u>before they egress the site boundary</u> (emphasis added)." (44) (87) Similar concerns were expressed by another commenter (107).

One commenter suggested that § 61.53(d) should be amended to include language specifying when performance specifications should be readied and to whom they should be submitted (16) and one (68) suggested an explicit reporting requirement in § 61.53.

۰ ..

B-73

The State of New York suggested that § 61.53 be rewritten to provide detailed prescriptive requirements, for groundwater monitoring, in particular (99). Two other commenters also addressed providing more details. Union Oil (66) objected to the term "information" in § 61.53(a) as unduly broad and recommended specifying the type of monitoring in § 61.53(b). Argonne (68) suggested clarifying that baseline data must cover both radiological and nonradiological characteristics. Argonne also provided cost data on environmental monitoring for the FEIS (121). Union Oil also questioned the need for monitoring during construction.

New England Nuclear (110) recommended that technological enhanced natural radiation be excluded from § 61.53(a).

The U.S. Department of the Interior (114) recommended adding "geochemistry" to  $\S$  61.53(a).

•• •

<u>Analysis of Comments</u>: In general, the comments received on this aspect of the rule have not required substantive change in the applicable sections of the rule. Analysis of individual comments follows in the order that the comments were listed above.

With respect to the comment on § 61.13, it is the staff's intent that analyses of release pathways will be conducted by the potential applicant with respect to specific verifiable locations: i.e., environmental monitoring stations. Results of monitoring will be made available to the Commission on an annual basis as per §61.80(h)(2)(ii), and verification of site performance will be made by NRC based upon the applicant's technical analyses, the results of the applicant's environmental monitoring and NRC release limits.

In establishing the period for pre-operational monitoring in § 61.53(a), NRC intentionally set a minimum time period of 12 months for seasonally variable data. If site conditions warrant additional monitoring to establish an accurate baseline of environmental data, NRC may require such monitoring on a site-specific basis. It is also NRC's intent that site monitoring results be related by the applicant to data acquired over a longer term by appropriate agencies to determine the representativeness of the site data. This intent will be elaborated upon in regulatory guides to be issued subsequent to this rulemaking.

 $\sim$ 

The State of North Carolina (109) expressed concern about the additional time the 12 month preoperational monitoring program would require in view of the January 1986 exclusionary date in the Low-Level Waste Policy Act. The State suggested changing the rule to require that the applicant "shall have implemented such preoperational monitoring by the time a license application is submitted." Since NRC review would probably take about 15-18 months and construction cannot begin until the license is issued, the State believed adequate data could be developed while review takes place. Section 61.53(a) requires 12 month data only for those characteristics subject to seasonal variation. The NRC can grant exemptions to the rule that are consistent with health and safety under § 61.6. For example, if regional data on seasonal characteristics was exceptionally good and applicability to the site clearly established, the NRC would consider a request for an exemption to the 12 months data requirement. An applicaton with no preoperational data to confirm site characteristics would involve demonstrations that are very speculative in nature. Further, the site characterization and preparation of an application will take at least a year. These activities and the monitoring take place in parallel. The data submitted would not have to be based on 12 months data from every monitoring point developed. The rule is sufficiently flexible to permit phased development of data based on interim findings. Thus from a practical view, the requirement should not be a burden or pacing item and the suggestion was not adopted.

The site suitability requirements of § 61.50(a)(3) necessitate gathering of population and land use data by the applicant. This data is generally available from existing sources (such as federal, state or local agencies; aerial photographs or other remote sensing; or topographic maps) and therefore, monitoring of these subject areas under § 61.53(a) is not necessary.

The rule in § 61.53(c) states that the monitoring system must be capable of providing early warning of migration of radionuclides <u>from the disposal site</u>. The disposal site as defined in § 61.2 contains both the disposal units and the buffer zone. The rule's intent is that monitoring should take place within the buffer zone so as to detect any migration before radionuclides would reach the site boundary. Hence, the commentor's point of clarification on this item is accurate and the language of the rule should be revised accordingly.

With respect to the comment on § 61.53(d) relative to when performance specifications should be readied and to whom they should be submitted, the rule sets forth in Subpart C the performance <u>objectives</u> to be met under the rule. The applicant (or licensee) must submit plans as part of the license application for taking corrective measures if migration of radionuclides indicate that the performance objectives of Subpart C would not be met. Action levels, evaluation of results, and special reports in addition to the annual report required by § 61.80(h) will be addressed on a site specific basis and specific license conditions. Thus no new reporting requirements were added to the rule.

The State of New York (99) suggested that § 61.53 be totally rewritten to provide specific prescriptive requirements for monitoring programs. The commenter suggested adding specific requirements for groundwater programs such as number of wells, depths, zones, and gradient orientation. A monitoring layer beneath the disposal units was also suggested. The commenter correctly notes that the NRC has experience with monitoring systems at existing sites. This experience suggests the highly site specific nature of monitoring programs. Similar considerations apply to Union Oil's (66) comments on more specificity. Argonne's (68) point about radiological/nonradiological is addressed to some extent by the list of topics or subjects in 61.53(a). The addition of geochemistry in response to commenter 114 further emphasizes nonradiological characteristics. One additional factor to consider is that environmental impacts and monitoring are a key issue under NEPA and will be addressed under the requirements of 10 CFR Part 51. The focus of Part 61 is the health and safety performance objectives based on radiological considerations. With respect to Union Oil's comment on the need for monitoring during construction, staff believes such monitoring is needed. As the list of subjects in § 61.53(a) indicates, the monitoring intended is more comprehensive than just monitoring pathways <u>per</u> se. Information on site characteristics important to understanding the site and predicting long term performance and events is intended. For example, excavation of a trench can yield important data on soil and ground water properties. Additional data on seasonal variations can be obtained.

· . · . · . The NRC does plan to address monitoring programs in regulatory guides and believes that this is the better forum for addressing the details.

New England Nuclear (110) recommended that technologically enhanced natural radiation due to excavation operations or change in pH be explicitly excluded from Part 61 in § 61.53(a). Specifically, Radon and K-40 levels in ground water may be increased due to site operations. As proposed, § 61.53 does not address this phenomonon. The commenter expects that the enhancement should not violate EPA drinking water quality regulations and may therefore be excluded. Such a specific prescriptive requirement is not in keeping with the approach of § 61.53. Details such as altered radon and K-40 levels prior to receipt of waste would be documented and evaluated as part of the site specific data. Any guidance on this issue would be included in a regulatory guide. No change to the rule was adopted based on the comment.

The U.S. Department of the Interior (114) recommended that the technical area "geochemistry" be added to the list of subjects listed in § 61.53(a). Staff agrees and the addition was adopted.

### Rule Changes:

1. The last sentence in § 61.53(c) should be changed from:

> "The monitoring system must be capable of providing early warning of migration of radionuclides from the disposal site.

to:

"The monitoring system must be capable of providing early warning of releases of radionuclides from the disposal site before they leave the site boundary."

2. Add to 61.53(a) after "geology": "geochemistry"

## ISSUE D-55-1

Commenters:

Issue

Waste classification - basis for numbers D. M. Mathews, Ph.D (23) Sargent and Lundy (24) Wisconsin Electric (32) Sierra Club (37) Ohio EPA (38) Bechtel (44) Duke Power (48) Argonne National Laboratory (68) Georgia Institute of Technology (70) The University of Texas Medical Branch (75) American Nuclear society (87) General Electric (89) Health Physics Society (96) Atomic Industrial Forum (100) American Institute of Chemical Engineers (102) Carolina Power & Light (106) The American Society of Mechanical Engineers (107) New England Nuclear (110) Kerr-McGee (115) Atomic Energy of Canada (118) U.S. Department of Energy (119)

Rule Citation: Table 1 of Section 61.55.

Summary of Comments: This group of commenters in one way or another objected to the numbers set out in Table 1. Commenter 24 stated that the waste concentration scheme and associated concentration limits would have a substantial impact on the nuclear power industry's waste disposal costs, but gave no basis for the statement. Commenter 24 suggested replacing the values in Table 1 with those from FBDU's study NUREG/CR-1005, "rather than the current "arbitrary" values. Commenter 44 thought that the logic for the concentrations was not apparent, and cited as an example the proposed limit  $(0.8 \text{ Ci/m}^3)$  for C-14. As stated by the commenter, "it would appear, for example, that carbon-14 which contains less than 0.8 microcuries per cc may be disposed of as segregated waste but that any concentration greater than 0.8, even if it is only a tiny increase, immediately requires that the disposer seek special permission from the government for disposal. The abrupt demarcation needs explaining so that the logic · · · · of it can be understood." · • • • Lag and the top top the

11 1

Commenter 38 thought that depleted uranium should be classed as a heavy metal poison, not as a radiation hazard, and questioned why it should be controlled at all. Commenters 68 and 107 questioned the status of low-activity bulk solid wastes containing uranium and/or thorium and referenced the draft BTP published October 23, 1981 (46 FR 52061) on uranium and thorium wastes. Atomic Energy of Canada questioned the entries for uranium and the potential for excluding natural uranium oxides because of the different limits for uranium as a metal. The DOE raised similiar concerns about uranium fluoride compounds. Similarly, commenter 38 thought that "iodine-129, as found (which means diluted with reactor-produced stable iodine) is of such low specific activity it is less of a radiation hazard than natural potassium or rubidium."

Commenter 23 did not see the line of reasoning associated with the numbers in Table 1, and made a comparison with NCRP total body burdens for I-129 and Sr-90. He requested a statement regarding how the numbers were chosen. Commenter 23 was also distressed that there was no consideration in the table regarding the chemical composition of the isotope under consideration. "It is my contention for example that iodine-129 in the form of lead iodide which has a very low solubility would be of less hazard than iodine in the form of potassium iodide which has a rather high solubility." Commenter 70 suggested that physical form be considered for H-3 and C-14 limits.

Commenters 75 and 96 and others felt that the numbers for the beta emitters not specifically listed were unduly restrictive. Commenter 37 felt that the half-lives of the listed nuclides had not been adequately considered. Commenter 106 and others felt that decay during operations had not been adequately considered. Commenter 89 felt that the system and supporting rationale is workable.

<u>Analysis of Comments</u>: The basic problem is that the commenters could not see the basis for the numbers in the table as published in the <u>Federal Register</u> Notice. However, the notice clearly states that the basis for the numbers is contained in NUREG-0782, the draft environmental statement for the rule. Many of the comments are general and did not identify specific values in Table 1 that were of concern.

The FBDU values in some cases are even more conservative than those in NUREG-0782. Other problems have been identified with the FBDU work. The FBDU numbers do not form a workable classification system in that they do not consider stability or waste form. The FBDU work contains some errors (for example, their treatment of direct gamma radiation is in error by at least a factor of two and possibly higher, and the lung model FBDU uses is in error for transuranics by about a factor of 10) that have been corrected in the NUREG-0782 work.

The comment on the demarcation between acceptable and nonacceptable quantities of C-14 reflects the very nature of <u>all</u> numerical limits in that on one side of the limit you are ok but on the other side you are not. It's possible that the commenter is confused by Table 1, with the same number for C-14 and other isotopes in all three columns. It would probably be better to set out two tables for Table 1.

The basis for limits for specific isotopes and which isotopes to include in the table(s) was re-examined in the FEIS and the commenters are referred to the EIS and the discussion of comments on the DEIS on this issue. Staff generally agrees that natural uranium should be deleted from the table. The proposed final version of the rule allows disposal of low specific activity residues (except for large quantities of mill tailings wastes) as Class A waste. Under DEIS calculations, iodine-129 is important since it can give high intruder exposures (and population impacts based upon intrusion) and dominates the ground water pathway: The possibility that the calculated impacts from iodine-129 are exaggerated by not considering dilution with stable iodine was included in preparing the FEIS.

As far as chemical forms of isotopes effecting the results is concerned, the commenter is probably correct. However, staff does not have the data necessary (and neither, undoubtedly, do most waste generators) to draw these distinctions in the classification scheme. Similarly for the physical form of H-3 and C-14.

## **Rule Changes:**

(a) Modify Table 1 to make two tables.

(b) Change limits and/or delete some nuclides (e.g., possibly uranium, C-14).

## ISSUE D-55-2

Issue: De minimis levels for waste

Commenters: PA, Department of Environmental Resources (16) Dow Chemical (17) United Technologies - Packard (25) Dow Chemical (17) University of North Carolina (30) Alabama Power (33) Sierra Club (37) Ohio EPA (38) Union Carbide (39) Bechtel (44) Duke Power Co. (48) Ontario Hydro (51) American College of Nuclear Physicians (53) Kentucky Special Advisory Committee on Nuclear Issues (55) Union Oil Company of California (66) Stock Equipment Company (67) Stock Equipment Company (67) Argonne National Laboratory (68) Georgia Institute of Technology (70) Utility Nuclear Waste Management Group (81) Middle South Services, Inc. (84) Northeast Utilities (85) American Nuclear Society (87) General Electric, Wilmington (89) Arkansas Power and Light (94) Health Physics Society (96) Don't Waste Washington Legal Defense Foundation (97) Wisconsin Electric Power Company (98) Atomic Industrial Forum, Inc. (100) U.S. Ecology (101) American Institute of Chemical Engineers (102) Conference of Radiation Control Program Directors (103) Carolina Power, and Light (106) American Society of Mechanical Engineers (107) North Carolina Radiation Protection Commission (109) New England Nuclear (110) U.S. Department of Energy (119) New England Nuclear (110) Rule Citation: 10 CFR Parts 20 and 61.

<u>Summary of Comments</u>: Of the commenters who commented on the concept of setting levels for wastes below which there is no regulatory concern, i.e., de minimis wastes, all but apparently one (Commenter 97) supported the concept. In their response, commenter 97 stated that "we agree with the proposed regulation's statement that there should be no generic 'de minimus' category for waste." While the great majority of the remaining commenters supported the de minimis concept firmly, some commenters' support was more of an implied nature. For example, commenter 37 suggested that at places such as medical hospitals and research institutions, wastes having half lives under 190 days could be segregated, stored for decay, and sent to a "regular municipal landfill if there is certainty that full decay has occurred. This would significantly reduce low level dump site volume requirements (currently 25% of total low level waste volume is provided by medical wastes) and provide a more rational segregation." Holding wastes for decay is discussed further under Issue D-55-14.

Some of the commenters supporting the de minimis concept made direct reference to NRC's position that exempting particular waste streams from compliance with the Part 61 regulations was preferable to setting generic de minimis levels for all isotopes. Commenter 16, for example, thought that setting "exemptions on the basis of certain waste streams is of dubious value and is considered to be a poor excuse for failure to deal with the controversial topic." As another example, commenter 51 thought it unfortunate that the rules did not establish a generic de minimus category, as it would (in the commenter's opinion) have a greater economic advantage than case-by-case decisions. This commenter then remarked, however, that as there is not yet a concensus on a generic de minimus level, any level chosen would be premature. A number of other commenters suggested that a de minimis classification be added to the Part 61 regulation, perhaps as an additional column to Table 1.

Considering all the comments, the fundamental concern appeared to be not whether a generic or a case-by-case approach should be taken, but that action to develop de minimis standards should be taken as soon as possible. As stated by commenter 81, "the establishment of a 'de minimus' category of low-level waste (LLW) whether upon a generic or case-by-case basis, would be extremely useful and would result in considerable savings of time, money and valuable burial space at disposal sites without any corresponding increase in risk to the public health and safety." Several commenters (e.g., commenter 100) suggested that NRC "permit case-by-case reviews of requests for specific applications of the 'de minimus' concept during the period criteria are being developed." Some suggested specific values for specific waste streams or radioisotopes. For example, commenters 68 and 107 suggested that low specific activity wastes containing uranium or thorium should not have to meet the proposed waste form requirements for Class A, B, and C wastes.

A potential pitfall in development of de minimis levels was observed by commenter 25 and others, who noted a problem with the de minimis standard for H-3 and C-14 in relation to DOT transportation standards. We permit disposal of waste containing H-3 and C-14 in liquid scintillation fluids in concentrations less than .05 uCi/gm without reguard to its radioactive content. However, DOT defines radioactive material as material with a specific activity greater than 0.002 uCi/gm (49 CFR 173.389(e)). Thus the H-3 and C-14 shipments must be identified as radioactive and hazardous disposal sites and landfills are reluctant to accept radioactive shipments. This situation led the commenter to note that "since the DOT regulations require a radioactive hazard label, the relaxed requirements of paragraph 20.306 are null and void for any wastes which must be transported from the generator's site."

i ·

Analysis of Comments: NRC agrees with the importance of setting timely standards for disposal of certain wastes by less restrictive means. NRC agrees with the commenters that establishment of such "de minimis" levels would reduce costs of disposal for many licensees and would also conserve space in disposal facilities which are otherwise designed for wastes having much higher activities, and also believes that such levels can be set and enforced so that the public health and safety is ensured. NRC staff also believes that establishment of de minimis levels is important in enhancing overall stability of a disposal facility, and therefore in reducing potential long-term site maintenance (and corresponding costs). As stated elsewhere, disposal facility stability would be best facilitated by requiring that all wastes be placed in a stable form. NRC staff, however, believe that this would be costly to many small entities and is difficult to require of waste streams which may contain only small amounts of radioactivity or are only suspected of being so. NRC has therefore compromised by requiring that high activity wastes (i.e., Class B and C wastes) be stabilized in a segregated manner from Class A wastes, for which no waste stability requirements are imposed. The result is improved overall site safety and stability relative to disposal of waste without consideration of classifica-...tion. Some maintenance would still be required for Class A disposal units, but such disposal units would only contain low activity waste.

NRC plans to further examine cost-effective methods to improve the overall stability of Class A disposal units. Some methods might include improved packaging. It is apparent, however, that eliminating waste streams that do not have to be disposed by rigorous methods will have a significant effect in improving overall disposal site stability. Restricting disposal to wastes which truly need to be disposed under the Part 61 requirements would also improve the cost-effectiveness of future potential improvements in Class A waste stability.

Regarding the issue of setting de minimis levels on a generic or on a case-bycase basis, NRC staff still believe that the current policy of examining waste streams on a case-by-case basis will result in the quickest and best results. It is recognized that setting generic limits may be a desirable goal, and NRC plans to work toward this goal over the next few years. However, the question of setting generic limits involves resolving a number of issues and questions such as limits for public exposures, pathways by which the public may be exposed, and ensuring compliance with the de minimis requirements. "NRC staff believe that the process of resolving such issues and questions on a generic basis would be expidited by first examining such issues with respect to a few specific waste streams. From these specific cases, generic requirements may be determined. Attempting to set generic limits and then applying these limits to specific waste streams, conversely, is believed to be less likely to produce workable results on a timely basis.

<u>Rule Changes</u>: Part 61 should not be deferred until additional waste streams or generic de minimis requirements are determined.

NRC staff plan, however, to accelerate work on setting standards for disposal of waste by less restrictive means. Such standards are expected to include

consideration of controlled on-site disposal by licensees as well as disposal into sanitary landfills and hazardous waste disposal facilities.

As a part of this, NRC staff intends to examine disposal of some specific waste streams, and will accept petitions for rulemaking or applications from licensees. In making such petitions or applications, licensees or petitioners should provide at least the following information:

- o a description of the process by which the waste is generated;
- a description of the waste generated, including chemical characteristics;
- o the radionuclide content of the waste, including principal <u>as well</u> <u>as trace contaminants;</u>
- o a description of the potential change in the radionuclide content as a function of process variations;
- o a description of the process control and quality control programs by which the licensee would ensure compliance.

Waste streams in which the radionuclide content is well known and relatively nonvarient are generally preferred.

The preceding guidance should be included in the discussion of the final rule.

### ISSUE D-55-3

Issue:

TRU classification (10 nCi/gm limit)

**Commenters:** 

Catherine Quigg (13) Department of the Environment, London (19) Florida Power & Light Company (31) Wisconsin Electric Power Company (32) Alabama Power (33) Ohio EPA (38) Union Carbide Corp./Medical Products Division (39) Duke Power Company (48) Utility Nuclear Waste Management Group (81) Northeast Utilities (85) American Nuclear Society (87) Power Authority of the State of New York (92) State of California (93) Arkansas Power and Light (94) Health Physics Society (96) Wisconsin Electric (98) Atomic Industrial Forum (100) U.S. Ecology (101) American Institute of Chemical Engineers (102) Conference of Radiation Control Program Director (103) Carolina Power and Light (106) American Society of Mechanical Engineers (107)

North Carolina Radiation Protection Commission (109) New England Nuclear (110) Colorado Department of Health (111) Kerr-McGee (115) U.S. Department of Energy (119) Argonne National Laboratory (120)

Rule Citation: Table 1 of Section 61.55. The state of the state

 $\{ t_{i} \}_{i \in \mathbb{N}}$ 

<u>n F</u>actoria d

. ....

۰. .

÷.,

Summary of comments: This subject created a lot of discussion, with a number of related points addressed. Of the commenters, 4 thought that the 10 uCi/gm limit should be either retained or lowered, while most of the remaining suggested that the limit be raised. Strend ter

The commenters who suggested that the 10 nCi/gm limit be raised presented a number of points. Commenter 19 commented that the limit "could be too restrictive for Class C wastes," and also stated that it wasn't clear whether the limit applied to Class A and Class B wastes as well. This commenter (19) was also concerned that the rule didn't allow case-by-case consideration of higher levels for disposal for improved methods (the commenter notes that allowing case-by-case considerations is in the rule but erroneously says that it doesn't apply to TRU waste). That is, there is no allowance for disposal between shallow land disposal and a high level waste repository at intermediate depth. The commenter thought that such a tact would prove to be very expensive in the UK.

Several commenters stated that the technical justification for the 10 nCi/gm limit had not been demonstrated since it came from the old comparison with natural radium (31, 32, 33, 39, 106, 107, 115). Many said that the number was >>> conservative or arbitrary. Other commenters observed that assay procedures for alpha-emitting transuranic isotopes were difficult, and one stated that it had not been determined that the detectable limit for alpha-emitting TRU radionuclides is down to less than 100 nCi/gm, thus making the 10 nCi/gm limit not measurable and therefore not enforceable (33). The difficulties of routinely measuring for TRU radionuclides at such low levels was also noted by other commenters (e.g., 96). Along these lines, several commenters objected to the logic that since 10 nCi/gm is achievable, it should be continued. Commenters (e.g., 110) stated that "operations should only be said to be ALARA when the cost to reduce impacts from these operations is justified by the benefits occured and when further costs to reduce impacts are not justified. Compliance with an excessive restriction or achieving a lower level of impact are not necessarily ALARA." Some of commenters (e.g., 38, 87, 107) stated that the limit led to great quantities of waste that is only suspected of containing TRU isotopes (mainly due to origin) to be needlessly stored. As stated by commenters 87 and 107, "much of the waste presently stored as transuranic waste is segregated from low-level waste on the basis of waste origin since the 10 nCi/gm limit is too low for accurate measurement and certification. However, segregation according to the 100 nCi/gm limit could be achieved, eliminating expensive retrievable storage and deep geologic disposal of 'suspect' transuranic waste."

As implied above, many if not most of the commenters suggested that the limit be safely raised to 100 nCi/gm. One (31) suggested that based on studies the limit could be as much as 50 times higher. These commenters stated that raising the level to 100 nCi/gm was technically defensible and would furthermore provide some advantages. As discussed above, one advantage would be enforcability. That is, with current measurement techniques, it was very dififuclt if not impossible to certify that one was below 10 nCi/gm but much less difficult to certify that one was below 100 nCi/gm. In discussing this, commenter 106 observed that although 10 nCi/gm is achievable during normal power plant operations, there are documented reports (EPRI Project 613, August 1980) of levels associated with unusual fuel performance which were occasionally in the 10-100 nCi/gm range. The commenter felt that the very small amount of waste that falls in this range could be readily disposed of safely. The commenter also believed that "all waste from operating power reactors could safely be assumed to fall below the 100 nCi/gm level, thereby achieving compliance by definition and making it unnecessary to perform direct or indirect measurements which are technically very difficult." 11100

Another advantage stated by the commenters was that a 100 nCi/gm limit would encourage volume reduction through incineration and other means while conversely, a 10 nCi/gm limit would discourage volume reduction. Commenters expressed the view that it may be necessary to dilute wastes in order to meet the concentration limits and that discouraging volume reduction would be contrary to NRC's policy on volume reduction as published in the <u>Federal Register</u> (46 FR 51100) on October 16, 1981.

The commenters cited a number of reports, documents, and ongoing activities as providing justification for their contension that the 10 nCi/gm limit be raised to at least 100 nCi/gm. Reports cited included those by Leddicotte, et. al. ("Suggested Concentration Limits for Shallow Land Burial of Radionuclides"), Adam and Rogers ("A Classification System for Radioactive Waste Disposal--What Waste Goes Where," NUREG-0956), and Rogers ("A Radioactive Waste Disposal Classification System," NUREG/CR-1005). One commenter (81) notes that the latter two documents have been criticized since they used an older ICRP-2 lung model rather than a more realistic ICRP-30 (Task Group) lung model. Commenter 81 then cited a recent publication by Dunning and Killangh ("A Comparison of Effective Dose Equivalents From Three Major Internal Dose Compilations," <u>Radiation Protection Dosimetry</u>, Vol. 1 No. 1, 1981) and contended that this publication demonstrated that "conclusions derived on the basis of the ICRP-2 lung model are valid for the transuranics." Staff does not agree with the commenters contention.

Other documents cited by commenters included:

- o a proposed revision of DOE Manual Chapter 0511 (dated 7/30/81);
- o a recent mark-up of House Bill HR 5016 by the House Science Committee;
- o the proposed EPA regulation for high level waste disposal, 40 CFR 191.

Others thought the NRC analysis of the TRU limit in the draft environmental impact statement was excessively conservative. The most common comment in this regard was that NRC was too conservative in that it did not consider dilution by other (lower activity) wastes, and that if dilution was considered, the allowable concentration could be increased by an order of magnitude or more. 15月19日在短期中市中

One commenter (110) stated that studies are quoted in the DEIS which "indicate that the average radioactivity concentration in waste can be expected to be from 1% to 10% of the maximum concentration." Another commenter (81) referenced a DOE document by Healy and Rodgers ("Limits for the Burial of the Department of Energy Transuranic Wastes," LA-UR-79-100, January, 1979) which indicated that a dilution factor of 20-60 could be expected for DOE trash wastes. 112 1. -

Two commenters believed that there were some errors in NRC's calculations. For example, commenter 102 thought that an inhalation pathway would not be applicable with layered disposal. This commenter also compares conclusions in the DEIS with supposed conclusions in the documents NUREG-0456 and NUREG/ CR-1005. Commenter 115 believed that NRC's method for calculating airborne dispersion by an intruder was in error and suggested another technique based upon resuspension by wind. Similarly to commenter 102, commenter 105 also stated (erroneously) that since all transuranic waste would be Class C waste, such waste would always be disposed a minimum of 5 meters below the earth's surface and could therefore never be contacted by an intruder.

One commenter questioned whether the transuranic concentration limit is a cumulative limit for all transuranic isotopes (except Pu-241) or if it is based upon a concentration per transuranic isotope. Another commenter (31) urged the Commission to perform an in-depth evaluation of TRU isotopes in reactor plant radioactive waste streams prior to adopting the 10 nCi/gm concentration limit, and to evaluate the practicality of techniques for demonstrating compliance. . - -• • • •

Commencers that supported the 10 nCi/gm limit or did not want it raised included commenters 13, 93, 103, and 111. Commenter 93 expressed general support for the ' 10 nCi/gm limit for near surface disposal of low-level waste, and stated that "wastes that exceed this limit should not be considered low-level waste and should not be buried at commercial low-level waste disposal sites." Commenter 103 expressed support for "confirmation of the definition of transuranic wastes as recommended by the conference serveral years ago." Commenter 111 stated that "transuranic wastes should be limited to 10 nCi/gm regardless of its decay mode due to the fact that the daughters may be hazardous." 

Commenter 13 had a number of concerns, and included some enclosures to support her positions and concerns. (Note that commenter 120 clarified some of the points the commenter attributed to him.) The commenter stated that based upon experience at Maxey Flats, INEL, and ORNL, which indicated that plutonium migrated much more quickly than originally anticipated, one would expect to see the concentration of transuranics allowed for shallow land burial regulated downward.

The principal concerns, however, appeared to be that:

A Star Car

- (1) much greater than 10 nCi/gm of TRU would be considered for land burial; and . . . .
  - (2) the limit for Pu-241 is too high since it decays to Am-241, which the commenter considers to be extremely toxic.

The first of the above two concerns is based upon two items. One is section 61.58, which allows "other provisions for the classification and characterization of the waste, disposal site, and method of disposal, if it finds reasonable assurance of compliance with the performance objectives. .." The second is the footnote in Table 1 which states that the concentration may be averaged over a waste package (multiply by 200,000 for a 55 gallon drum). The commenter noted that this allows up to 2 million nanocuries of TRU or 70 million nanocuries of Pu-241 per 55 gallon drum. These issues are addressed as part of Issues D-55-5 (Case-by-case approval of disposal of wastes in greater than Class C concentrations), D-55-7 (Averaging concentrations over packages), and in the FEIS.

Other specific comments/questions on TRU waste from commenter 13 were:

.

- o Why didn't the NRC put "less than 10 nCi/gm in columns 1 and 2 on page 38085. Doesn't it apply?
- o Why didn't the NRC say that TRU waste above 10 nCi/gm should be given retrievable surface storage until its new regulations for intermediate waste are available?

Commenter 85 suggested that 163 day half-life Cm-242 should be exempted from the limit. Commenters 87 and 107 suggested removable surface contamination limits.

The DOE commented that the TRU limit should be based on pathway analyses and that DOE is reassessing its use of the 10 nCi/g limit.

<u>Analysis of comments</u>: The commenters raised a variety of issues. Most of the comments were written without consideration of the draft EIS. In the draft EIS, staff made an effort to come up with numbers for individual transuranics, but this effort suffered from two basic limitations: (1) the analysis did not consider decay chain daughters and (2) the analysis did not go from individual isotopic concentrations and calculate an effective cumulative limit for all alpha-emitting isotopes. These limitations were factored into the final EIS. Staff agress that Cm-242 deserves special consideration and has addressed it in the same manner as Pu-241.

There has been some discussion by commenters and others regarding raising the TRU limit to the range of 100 nCi/gm. The best reason for doing this appears to be the ability to measure transuranics. It's easier and less expensive for them to show that they are under 100 nCi/gm than under 10 nCi/gm.

Work in the draft EIS did not consider dilution due to depth of disposal and by other waste streams; the calculated limit for many if not most alpha-emitting TRU isotopes was about 10 nCi/cm<sup>3</sup> (about 6 nCi/gm). Staff agrees that not considering that Class C waste (which has been disposed at a minimum depth of five meters) would still be difficult to contact even after 500 years was an unnecessarily conservative assumption. Other considerations include dilution by lower activity waste streams, improvements in health physics methodologies, and differing disposal site environmental characteristics. In order to provide more realistic estimates of the consequences that will result from disposal of TRU wastes, the analyses were reevaluated in the final EIS, with the result that the near-surface disposal limits for transuranics and other isotopes were raised by a factor of 10. The following discussion addresses some of the specific questions raised by the commenters who supported a raised limit:

.

.

0

0

· 0

0

1.

.

.з.,

.

. :

bo Commenter 19 questioned whether the limit applied to Class A and B waste as well as Class C waste. The answer is yes. 

Commenter 19 was also concerned that there was no disposal method between shallow land burial and a repository. A partial answer is that developing requirements for disposal by such intermediate methods (and correspondingly higher isotope concentrations) is planned. Licensees will have to continue to store wastes not acceptable when evaluated on a case-by-case basis until a commercial or DOE facility is available. The commenters concerns are valid but not new or a result of Part 61.

- o The justification for the 10 nCi/gm limit (commenters 31, 32, 33, 39) was addressed in the draft EIS.
- The comments (31, 33, 38) that assaying for alpha-emitting TRU isotopes 0 is difficult, impossible to routinely directly measure down to 10 nCi/gm, and leads to a lot of suspect waste are good comments and support a higher limit.
  - Commenter 39 questioned the logic that the 10 nCi/gm limit should be continued because it is achievable since there have been occasions when wastes could not meet this limit and were not acceptable at the commercial disposal sites. Unless there is no limit, there may be some waste streams which are over the limit and this point is not sufficient reason to change the limit. · `-۰. .....
  - As proposed and as finalized, the limit is a cumulative limit for all TRU (except Pu-241 and Cm-242) isotopes rather than an individual isotopic limit. The cumulative limit is more conservative but also contributes to the need to raise the limit.

:

Commenter 31 suggests that we should take a close look at TRU isotopes in reactor waste, as well as the practicality of determining compliance, prior to adopting the 10 nCi/gm limit. Because of the license conditions in effect at the disposal sites, licensees have had to comply with the limit for years. However staff is continuing to look at these issues and follow work being done by EPRI and others in this area. Raising the limit for Class C wastes should alleviate some of the commenter's concerns.

Commenter 13 expressed concerns about plutonium migration, disposal above 10 nCi/gm, the Pu-241 limit, applicability of the 10 nCi/gm limit to columns 1 and 2 in Table 1, and requiring storage of TRU wastes above the limits.

· · . -

In response the comment regarding "migration" of TRU isotopes at Maxey Flats and ORNL (migration has not been reported at INEL), the experience at Maxey Flats and ORNL does not prove that TRU: isotopes migrate faster than expected. Maxey Flats and ORNL are similar situations in that in both cases we have had 2. bath-tubbing as well as disposal in fractured formations. The bathtubbing at ORNL has led to surface seeps. On one hand, we have had extended periods of contact between water and waste (plenty of time to form a semi-organic leachate leading to concern regarding chelating agents) and on the other hand we have

had instances of rapid flow through fractures. We also have had lots of surface contamination at these sites. However, consider the situation at other places such as the HLW tanks at Hanford and the Barnwell and Sheffield sites. Where there are no fractured formations and percolating water has flowed through the waste and soil rather than being allowed to bath-tub, plutonium and other TRU isotopes have shown a definite propensity to migrate little if at all (even with organic chemicals present). 11100

Case-by-case determination of higher limits than 10 nCi/gm is addressed in Issue D-55-5. In approving any exceptions or alternatives to the technical requirements in Part 61, meeting the performance objectives rather than the numerical concentration limits will be the bottom line. Thus the response to the question about what criteria will be used to evaluate proposals to dispose of higher concentrations is, the performance objectives form the "criteria." With respect to the Pu-241/ Am-241 issue, the submitted information included no calculations or pathway analysis. Based on work associated with the EIS, staff continues to believe that it can be demonstrated through analysis that the 350 nCi/gm number for Pu-241 is conservative. The applicability of the 10 nCi/gm in TRU limits for Class A and B wastes needs to be clarified in a revised table. The proposed rule and table were somewhat confusing with respect to columns 1 and 2. There is no need to require that licensed TRU waste be stored. The requirements already exist under existing rules. If the waste cannot be transferred out of the licensee's possession, the licensee must safely store it.

Commenter 13 also questioned the numerical TRU concentration limit for "nearsurface" that would be approved on a case basis. NRC is not in a position at this time to set such a limit, and there is furthermore no compelling reason to set one now. A concentration limit for intermediate depth disposal will be considered at a later time. NRC would want to caveat any future limits to allow for flexibility and future improvements. If NRC gets a license application in the meantime, a site specific limit can be included as part of addressing the license application. Similiar arguments apply to the question on numerical limits on the depth of disposal.

The ANS and ASME suggestion to add a 100 picocurie per square centimeter  $(pCi/cm^2)$  limit for transferable contamination of TRU nuclides was based on proposed revisions to the DOE Manual Chapter 0511. The surface contamination limit could reduce the potential exposure for an archeological or scavenger type intruder. Part 61 did not attempt to protect such intentional intruders who would be looking for identifiable waste such as lathes. Protection of the inadvertent intruder was considered and surface contamination is not important in the scenarios. However, such a limit is not unreasonable for DOE wastes as an ALARA approach in view of the more frequent disposal of contaminated equipment of interest and of the TRU contamination in DOE wastes that is primary and not incidental to other nuclides.

In response to commenters 102 and 115, NRC's intent regarding the 10 nCi/gm limit in the proposed rule was that the limit apply to all classes, not just Class C. Given the uncertainties regarding natural and human actions over long time periods, and the long half lives of many of the transuranic isotopes, NRC believes that it is unreasonable to assume that Class C waste could never be contacted by humans. Although the commenters assertions regarding the draft EIS analyses did not accurately describe what was actually done, the analyses en l'an el ser el m

1.1.1.1.1.1

were reassessed for the final EIS to determine whether there existed excessive conservatism. In regard to the methodology used to calculate airborne dispersion, such dispersion is assumed to result from mechanical disturbance of the soil, not from wind resuspension as mistakenly asserted by commenter 115.

- Rule Changes:
- 1. 1. 1. 1. 1. 1. 1. Raise the limit for TRU includes with half lives greater than five years to 100 nCi/g for Class C wastes.
- 2. Clarify the case-by-case approval provision.
- 3. Clarify the TRU limits for Class A and B.
  - 4. Add a separate limit for Cm-242.

## ISSUE D-55-4

Issue:

11

## Waste classification - Ra-226

Commenter:

Commonwealth Edison (35) Bechtel (44)

New Mexico Secretary for Health and Environment (45) American Nuclear Society (87) American Society of Mechanical Engineers (107) North Carolina Radiation Protection Commission (109)

1 : . . .

11.

## Rule Citation: Table 1 of 61.55.

Summary of Comments: Basically, the commenters want to know what to do with waste contaminated with or containing Ra-226, a radioisotope which is not currently listed in Table 1. Commenter 35 states that they possess several radium-226 sources used at their fossil fuel stations for flow rate determinations. Commenter 45 states that it is not clear whether Ra-226 will be permitted for disposal and in what concentrations. Commenters 44, 87, 107, and 109 all request a value or concentration limit for Ra-226 disposal.

Analysis of Comments: As the commenters have observed, there is no waste disposal concentration limit for Ra-226. It appears that there are two types of radium wastes to be considered: (1) small concentrated sources such as sealed sources or radium dials, and (2) wastes which contain small amounts of radium incidental to other radioisotopes such as mining or manufacturing residues. The former would in general not fall under the auspices of the Atomic Energy Act of 1954, and would also not appear to be generated in significant quantities. The EPA has a program for collection of discrete radium sources. Disposal of the latter type of waste is probably more common and may or may not involve material regulated under the Atomic Energy Act. NRC has not placed limits for such material in § 61.55 because such wastes are believed to generally not occur in sufficient quantities to warrant it. However, the staff sees no reason to exclude small amounts; of uranium or thorium mill tailings wastes that might result from laboratory assay, research activites, environmental sample analyses, etc. Therefore, a provision for disposal of small quantities of tailings waste as Class A waste should be added. For purposes of this provision,

a small quantity could be defined as 10,000 kilograms containing not more than 5 millicuries of radium-226. This radium concentration is typical of uranium mill tailings (0.5 nanocuries per gram). The quantity of radium-226 is that contained in 150 pounds of natural uranium at equilibrium with its daughter products using a specific activity of 6.77 x  $10^{-7}$  curies per gram from Appendix B of 10 CFR Part 20. 10 CFR Part 40, § 40.22 permits persons to possess and use under general license 150 pounds of source material per year. Permitting the disposal of such a quantity in a near-surface disposal facility is judged to be acceptable. For larger amounts, specific approval would be required.

Suggested Rule Change: Amend 61.1(b)(2) to read:

Class C concentrations

(2) Disposal of uranium or thorium tailings or wastes (byproduct material as defined in § 40.4(a-1)) as provided for in Part 40 of this chapter in quantities greater than 10,000 kilograms of uranium tailings or wastes containing less than five (5) millicuries of radium-226.

### ISSUE D-55-5

Case-by-case approval of disposal of waste in greater than

<u>Issue</u>:

1. 1

<u>Commenter</u>: Catherine Quigg (13) Los Alamos National Laboratory (43) Birmingham Audubon Society (80) Northeast Utilities (85) State of California (93) North Carolina Radiation Protection Commission (109)

Rule Citation: Section 61.55, Table 1 of Section 61.55, and Section 61.58.

Summary of Comments: The commenters' concerns were related to a footnote in Table 1 and paragraph 61.55(d) which indicates that greater concentrations than Class C limits may be determined to be acceptable for near surface disposal under certain conditions. The footnote to Table 1, for example, states "Until establishment and adoption of other values or criteria, the values in this table (or greater concentrations as may be approved by the Commission in particular cases) shall be used in categorizing waste for near-surface disposal." Paragraph 61.55(d) states "Waste that has a radioisotope concentration that exceeds the values shown in Column 3, Table 1 of this section, is not generally acceptable for near-surface disposal and shall not be disposed of without specific Commission approval pursuant to § 61.58 of this part." Section 61.58 states that "The Commission may, upon request or on its own initiative, authorize other provisions for the classification and characteristics on a specific basis, if, after evaluation of the specific characteristics of the waste, disposal site, and method of disposal, it finds reasonable assurance of compliance with the performance objectives in Subpart C of this part."

In their responses, the commenters either asked for clarification of the requirements (43, 85, 109) or were opposed to any exceptions in near-surface disposal requirements (13, 80). Commenter 80, for example, stated that "there should be no exemptions in near-surface disposal prohibitions against the higher level wastes." Commenter 13 was concerned that the Part 61 requirements would allow large quantities of transuranic isotopes to be disposed by near-surface

disposal. This concern appeared to be motivated by Section 61.58 as well as by another footnote in Table 1 which states that radionuclide concentrations may be averaged over the volume of a package and for a 55-gallon drum the concentration limits may be multiplied by 200,000 to determine allowable total activity. - Commenter 13 noted that this allows up to 2 million nanocuries of TRU or 70 million nanocuries of Pu-241 per 55 gallon drum. (Note that while the commenter correctly calculates the maximum activities that the concentration limit would allow in a drum, intentional dilution to meet this limit was not intended and concentrated sources are not a common waste form in non DOE wastes. (See issue D-55-7.) Commenter 13 also questioned who in NRC would make a case-by-case decision and what the criteria would be to judge whether a particular site was suitable, and questioned what the maximum limit on transuranic concentration NRC will allow for land disposal. Commenter 43 was concerned that the definition of waste that might be included in land disposal was too open ended, and that according to Section IV under Supplementary Information, "high-specific activity wastes, such as those produced presently during the cleanup operations at TMI-2, will qualify for land disposal as 'Class C Intruder Wastes.'" Commenter 93 throught that TRU-contaminated waste should in no circumstances be considered low level waste and each waste should be disposed of at specifically-designated sites operated by the federal government. Commenter 85 questioned what criteria would be used for approvals under § 61.58 and Commenter 109 was concerned about special treatment of certain licensees that might result from case-by-case approvals.

Analysis of Comments: The concentration limitations and other requirements in Subpart D are intended to help ensure that the performance objectives established in Subpart C are met. That is, the concentration limits and other requirements are not the end in themselves, but are a means of achieving the end. The Class C limits were developed using the performance objectives as criteria to ensure safe disposal of waste considering the degree of isolation provided by "normal" near-surface disposal. Obviously, to ensure that the performance objectives are met, disposal of higher concentration of isotopes than those listed in column 3 of Table 1 would have to be by disposal technologies having higher isolation capacity than "normal" near-surface disposal. Such improved disposal technologies could, depending upon the particular radioisotopes, involve better waste forms or packaging, or disposal by methods having additional barriers against intrusion.

While there are some minor changes which should be made to the rule to clarify NRC's intent, NRC still believes that the best overall approach to the rule is the existing framework in which requirements are established which apply to the majority of the waste, but some flexibility is allowed in meeting the performance objectives. The principal reasons for this position are as follows:

....

- 1. The approach allows for potential improvements in disposal technology, and also allows for consideration of licensees which may produce unique wastes.
- 2. The approach is in keeping with the philosophy of the Regulatory Flexibility Act of 1980 (Public Law 96-345) which emphasizes objectives and flexibility to reduce burdens on the public.
- 3. NRC will be looking next at setting regulatory requirements in the form of amendments to 10 CFR 61 for licensing disposal by methods offering greater isolation than near-surface disposal. These methods could include,

for example, intermediate depth disposal or use of mined cavities. NRC staff expect that the regulatory requirements developed will include setting limiting concentrations for isotopes of significant concern. In the meantime, it is possible that license applications will be received for disposal by such improved methods. NRC staff wish to retain the flexibility to be able to address these license applications in the existing framework of the rule. It is not desirable to arbitrarily prohibit NRC from considering such applications, especially since there is a current shortage in disposal capacity.

For similar reasons and in response to Commenter 13, NRC staff does not plan at this time to establish an absolute concentration limit for land disposal of transuranic or other radionuclides. In the near future, NRC intends to analyze and develop technical criteria for disposal by disposal methods offering greater isolation than near-surface disposal. As part of these efforts, NRC expects to develop concentration limits for disposal by these methods; these concentration limits are of course expected to be higher than limits established for near-surface disposal. In any case, NRC staff expect to incorporate flexibility into future requirements to allow for alternative ways to meet the performance objectives as well as potential improvements in technology.

Commenter 43 stated reservations regarding the definition of wastes acceptable for near-surface disposal. The commenter was particularly concerned that some high-specific activity wastes from the Three Mile Island (TMI) cleanup would quality as Class C wastes. While the commenter did not specifically state which TMI wastes he was concerned about, staff assume that he is referring to the EPICOR-II first stage liners. These wastes contain organic resins which are highly loaded with Cs-134, Cs-137 and Sr-90. The loadings on these resins would qualify these wastes as Class C. The commenter's concern, staff assumes, involves the radiolytic degradation of the organic resins.

The NRC is preparing a Branch Technical Position (BTP) on Waste Form which recommends a restriction against the generation of organic resins which would have total accumulated doses of greater than  $10^8$  Rads. At this dose level organic resins begin to undergo substantial degradation. The BTP guidance includes loading of organic resins in excess of  $10^8$  Rads when it has been demonstrated that the specific resins will not suffer substantial degradation. Staff views this type of detail to be overly precriptive and restrictive for the rule.

The EPICOR-II first stage resins will receive total accumulated doses in excess of 10<sup>8</sup> Rads. Due to their unique nature, the DOE has agreed to accept these wastes for research and development and disposal purposes. See Issue D-56-15 also.

## Suggested Rule Change:

 Change the language in the footnote in Table 1 reading "...or greater concentrations as may be approved by the Commission in particular cases..." to read "...or greater concentrations as may be approved by the Commission in accordance with § 61.58..." ISSUE D-55-6

মাহা সংগ্ৰন্থ হয়

. . ..

Issue:	Compliance with waste classification requirements
Commenter:	United Technologies - Packard (25)
 	Florida Power & Light (31)
	Bechtel (44)
	Duke Power Co. (48)
	Arkansas Power and Light (52)
· · · · ·	Department of the Army (63)
· · · ·	Utility Nuclear Waste Management Group (81)
•	Middle South Services (84)
	American Nuclear Society (87)
•	Florida Power Corporation (91)
,	Power Authority of the State of New York (92)
č . t.	Arkansas Power and Light (94)
	Health Physics Society (96)
	Wisconsin Electric Power Company (98)
· · · · · · · · · · · · · · · · · · ·	Atomic Industrial Forum (100)
• •	Carolina Power and Light (106)
	The American Society of Mechanical Engineers (107), (113)
and a second	North Carolina Radiation Protection Commission (109)
	New England Nuclear (110)
	U.S. Department of Energy (119)

Rule Citation: Section 61.55 and Table 1.

. :

Summary of Comments: These commenters' concern was how one determines compliance with the waste classification requirement. That is, how does one measure and report all the radionuclides contained in all the miscellaneous waste streams, and how accurate must one be? Most commenters were concerned that the regulations would require them to routinely measure for every isotope in Table 1 within each package of waste. As stated by commenter 25 in regard to trash wastes, "this mixture is not amenable to analysis, therefore any statement of activity can only be a rough estimate, and the combining of radios of mixed radionuclides will certainly increase the uncertainty." Commenter 31 was concerned with the practicality of demonstrating compliance for non-gamma emitting isotopes. Commenter 31 felt that any requirement to perform complete assays on each waste shipment would (1) result in significant increases in personnel radiation exposures (from collection of more and larger samples and from increases in sample handling times), (2) result in transportation difficulties (transport casks tied up longer while awaiting completion of assays prior to shipment), and (3) increase need for temporary on-site storage. Similar sentiments were expressed by other commenters, including commenters 84, 91, 94, and 98.

Some commenters mentioned specific isotopes which they felt would be very difficult to routinely measure and demonstrate compliance with (e.g., Commenters 92 and 100). These isotopes included Ni-59, Ni-63, Nb-94, Tc-99, I-129, and Cs-135.

Commenters 44, 87, and 107 and others felt that classification of external dose measurements should be allowed, particularly for wastes such as trash. Commenter 48 suggested that rather than sampling every container to determine compliance, the waste could be generally classified by source. Only under unusual or abnormal conditions would more detailed calculations have to be made. That is, "PWR ion exchange resins might be said to be Class B-Stable waste under normal conditions of reactor operation where there is less than X% failed fuel in the reactor and where the total activity in the waste container is less than Y curies." Commenters 106 and 119 (and others) suggested that the rule specifically state that indirect measurement techniques would be allowed. Commenter 81 strongly supported "the concept of using key isotopes which can be externally measured and without opening the waste package." Some suggested that the concept of error limits be included for measurements.

Commenter 52 felt that it would be impractical to "positively determine a given activity level as being characteristic of a given isotope," and therefore the proposed waste classification methodology is also impractical. Others stated that the limits in the table would tend to discourage volume reduction and were therefore too low. One commenter suggested classification limits be developed for every site (110). This commenter was also concerned that the classification system "may lead to generators assigning conservative estimates to waste concentrations and consequential under utilization of waste site." In addition, the commenter was concerned that the "recommended scaling factors" in the DEIS "are not applicable to industries making a wide range of custom products."

Commenters 44 and 87 (and others) questioned whether the classification scheme could be combined with the transportation system requirements in 10 CFR Part 71.

Commenter 63 indicated no compliance difficulties for the Army.

Analysis of Comments: This issue, along with transuranics and de minimis, is one of the largest issues related to the proposed rule. Staff agrees that additional guidance is needed for licensees. The draft waste classification regulatory guide should be made available for comment as soon as practical and discussed in both the preamble to the final rule and the final EIS. Staff agrees that routine assay of all packages is not necessary or desirable. Alternative methods can be used to show compliance. For example, staff has identified four basic programs which may be used either individually or in combination by licensees. They are: materials accountability; classification by source; gross radioactivity measurements; and direct measurement of individual radionuclides including scaling some radionuclides based upon measurement of others. These methods are discussed in the draft Branch Technical Position on Waste Classification and in the FEIS. Staff also recognizes that the confidence with which licensees assure proper classification will increase as programs are established and verified and more experience is gained with the licensee's specific wastes.

The answer to the question of can the clasification requirements in Part 61 be combined with the transportation requirements in Part 71 is no. The requirements are based on different concerns and different time periods. Part 71 requirements are to get the materials safely to the site. Part 61 classification addresses what happens over the long term in the facility. For example, nuclides that are short-lived may produce high gamma radiation levels which are of concern in shipping and handling but of little concern over the long term and long-lived soft beta emitters such as C-14 that do not pose a handling hazard are of special concern over the long term.

All measurements or calculations of radioactive content involve error limits. Staff judged that this point is sufficiently understood but will consider the issue in the regulatory guide on classification.

Rule Change: Add a provision to the rule that indirect methods may be used to determine compliance with classification. Also the number of isotopes in the rule should be reduced to the minimum needed. For example, Cs-135 could be eliminated. Compliance with classification should be discussed in the preamble to the final rule.

## **ISSUE D-55-7**

the state of the Issue: Averaging concentrations over packages

2

State of the second

Tereta in a sur Commenter: Bethlehem Steel Corporation (7) Catherine Quigg (13)

E. O.

New England Nuclear (110)

U.S. Department of Energy (119)

## Rule Citation: Table 1

Summary of Comments: One commenter requested confirmation of the averaging provision for sealed sources. One commenter was concerned about concentrated TRU waste or "hot spots" allowed by averaging and one noted the need to 🗄 specifically allow averaging TRU over the weight of the package. One suggested averaging concentrations of certain nuclides over shipments or the site rather than packages.

Analysis of Comments: Bethlehem Steel Corporation applyed the averaging provision for sealed Co-60 and Cs-137 sources correctly and averaged the activity over the volume of the package. However, the waste must also meet the physical form and characteristics requirement of the waste class determined by the averaging. 

Quigg, who was concerned about averaging the concentration over the volume of the container, is correct that areas of concentrated activity are permitted. The commenter was specifically concerned about concentrations of transuranic (TRU) nuclides. From a practical point of view, most shipments with trace TRU will generally be fairly homogeneously distributed and incidental to the total activity. Averaging over the packages is physically representative of the waste itself in this case. Only occasional shipments from any generator should involve heterogenous distribution. Possibilities include some decontamination and decommissioning waste from cleanup of mixed oxide fuel research facilities and waste from destructive metallurgical analysis of an occasional fuel rod.

TRU wastes are just not produced in significant amounts where averaging would permit disposal under the proposed or final classification scheme. Further, when the exposure scenarios are examined, the waste must degrade to become accessible in the important scenarios. Mixing with the package contents and

surrounding fill and soil should occur as the intruder disturbs the disposal site. Hot spots are not important in migration pathways such as groundwater transport. Relying on such mixing for the occasional package seems reasonable and relief for the shipper is justified. If frequent shipments with heterogenous distribution were expected, more restrictive averaging requirements might have been proposed. Future changes in waste streams (e.g., if reprocessing is resumed) can be reflected in changes to the rule. Certainly if reprocessing is resumed the Commission will examine the wastes being generated.

Commenter 119 and others noted that the rule should clearly allow averaging TRU over the weight of the package. Staff agrees.

New England Nuclear's suggestion to "accept inventory methods designating average concentrations in waste shipments" was part of the commenter's arguments offered in support of providing relief for assaying individual containers for very low concentrations and hard to measure nuclides which must be restricted because of ground water scenarios (i.e., H-3, C-14, Tc-99, and I-129). Changes to the final rule to delete some nuclides (e.g., Cs-135) and to clarify that indirect methods of determining activities partially address the commenter's concerns. The commenter also emphasized that without some relief, licensees would conservatively over-report the four nuclides H-3, C-14, Tc-99, and I-129 or report that the waste meets the limits so that the values used to keep total site inventories would not represent activity actually present. This conservative over-reporting could exhaust site inventory limits and lead to inefficient use of the site. Staff agrees and plans to address this issue in the BTP on waste classification. Keeping the concept of averaging over the waste volume or mass is important for the rule but details are more readily addressed in guidance documents.

Rule Change: Provide flexibility in the averaging language of the rule.

ISSUE D-55-8

Issue:

Ensuring waste generator compliance with Part 61 requirements and impacts on volume reduction

Commenters:

Nevada Department of Human Resources (14) Dow Chemical (17) Sierra Club (37) Union Carbide (39) Stock Equipment Company (67) South Carolina Department of Health and Environmental Control (79) Power Authority of the State of New York (92) State of California (93) Health Physics Society (96) Wisconsin Electric Power Company (98) State of New York, Department of Law (99) Atomic Industrial Forum (100) Carolina Power and Light (106) North Carolina Radiation Protection Commission (109) The American Society of Mechanical Engineers (113) U.S. Department of the Interior (114)

U.S. Department of Energy (119)

Rule Citation: None specific.

<u>Summary of Comments</u>: Commenters 14, 17, 67, 79, 93, and 96 remarked on the issue of ensuring compliance with the Part 61 rules once promulgated. One

commenter (14), speaking from the viewpoint of a state regulator, noted that the record and history of packaging of radioactive waste by generators has not indicated a durable capability to package waste in conformance with DOT regulations. Given this assertion, and the commenter's opinion that the types and quantities of isotopes in waste packages are at best educated guesses, the commenter doubted that waste generators will be able to properly classify or even properly label the waste according to the proposed regulation. Another commenter (17) spoke from the position of a radwaste solidification media and equipment supplier. The commenter stated that neither the Commission nor the existing disposal facilities have inspection and enforcement programs to assure that the requirements are being met. In the commenter's opinion, standards and technical criteria without inspection and enforcement are useless and unnecessary.

e se la mora para s

Commenter 67 (another radwaste solidification media and equipment supplier) stated that "the key to any program is inspection and enforcement," and suggested that "after the standards are established, enforcement be immediately established to ensure compliance." Like commenter 17, commenter 67 thought that criteria should be based upon ALARA and use the best available technology.

Commenters 79, 93, and 96 made somewhat more specific suggestions. Commenter 93 suggested that the rules require programs for checking the contents of waste drums, either at the site of origin or at the disposal site. Commenter 79 suggested that Section 61.12 of the draft rule ("Specific technical information") include provisions "for the inspection of waste forms to insure that license requirements are being met. These provisions should require periodic inspection of waste packages, administrative procedures, or waste inspection and preparation procedures submitted by the waste generators utilizing the burial facility." Commenter 96 suggested that waste generators warrenty that packages being shipped will meet the long-term 150-year criteria. Disposal facility licensees would also warrenty that "to the best of his knowledge and efforts, the packages as buried will meet the same criteria." Commenter 96 also stated that "the form of the quality assurance programs by both the generator or processor to meet these criteria may need to be spelled out in more detail."

A number of commenters, including 37, 39, 98, 100, 106, 109, 113, and 119 argued that the waste classification scheme tends to discourage volume reduction. The point was made for TRU nuclides (Issue D-55-3) and all nuclides that since volume reduction increases concentrations, treated wastes may fall in a more restrictive and thus more costly class or become generally unacceptable for near-surface disposal. The commenters noted that the Commission encourages volume reduction and referenced the policy statement on this issue published October 16, 1981 (46 FR 51100). Most of these commenters supported volume reduction by implication. The state of New York, DOI, and ASME explicitly supported volume reduction.

<u>Analysis of Comment</u>: Issue D-55-6 is concerned with how waste generators will demonstrate compliance with waste classification. This issue, however, is the other side of this issue--how can regulators, through inspection and enforcement programs, ensure that the requirements in the rule are being met?

The issue of inspection and enforcement to assure compliance with regulations neither started with nor will end with 10 CFR 61. Inspection and enforcement is obviously not something that can be ensured through rulemaking alone. One

of the problems has been that in the past, there was often not much attention given to waste disposal. NRC has recognized the need for more uniformity in compliance with waste transportation regulations by, for example, incoporating DOT transportation regulations into NRC's own regulations. Thus, compliance with transportation regulations may be inspected against by NRC.

Similarly, prior to the development of the Part 61 rule, waste generator compliance with license conditions at disposal facilities has been generally enforced through provisions in 10 CFR Parts 30, 40, and 70 (e.g., § 30.41 Transfer of byproduct material) that provide that no person may transfer licensed material to another person unless that person is properly licensed to receive it. Setting out specific requirements in 10 CFR 61, however, will greatly enhance NRC's ability to inspect against violations of site license conditions. In addition, a quality control program is required of generating licensees in the new § 20.311 of 10 CFR 20 to help ensure compliance by waste generators and other licensees with the Part 61 requirements. This quality control program can also be inspected against. (See Issue M-2.) Adoption of uniform requirement for quality control programs and waste form and classification by Agreement States will greatly bolster the effectiveness of a national system of inspection and enforcement.

In regard to suggestions by commenters 79, 93 and 96, the suggestions would involve adding more prescriptive requirements to the rule on the quality control program the site operator must have. The proposed rule has a general requirement for a quality control program that must include waste receipt, handling, and emplacement. The commenters raise valid issues that should be factored into additional guidance documents being developed. Two important related points are (1) the emphasis the proposed rule places on the generator's responsibility for compliance and (2) the associated inspectable documentation and certifications required. Ensuring compliance will be a collective effort by NRC and Agreement State inspectors, once compatible State regulations are in place. The site operator's program will no longer be the first line of defense. Staff agrees with Commenter 79 that disposal site package inspection, administrative procedures, and generator programs are parts of a compliance program but prefers to leave flexibility in the mix and dependence on these and other parts of the program. ÷

With respect to the impact of complying with Part 61 on volume reduction, staff does not feel that waste classification necessarily discourages volume reduction. While a higher classification of waste might result in more stringent requirements on waste form and disposal methods, there are economic considerations that need to be considered by the waste generator. The cost of processing, shipping, and disposal of a small volume of higher classification waste needs to be compared with the transportation and disposal of a larger volume of a lower classification waste. There is no reason to believe that the balance will always be against volume reduction. For wastes with concentrations that would place them not generally acceptable for near-surface disposal if they were volume reduced, the provisions for specific Commission approval of the disposal of such wastes provides a potential alternative for licensees considering volume reduction.

Rule Change: None.

1 · . .

## ISSUE D-55-9

র জনমা স্থান চা

Issue:Waste classification by total hazardCommenter:American College of Nuclear Physicians (53)<br/>Argonne National Laboratory (68)<br/>General Electric, Wilmington (89)<br/>Conference of Radiation Control Program Directors (103)<br/>U.S. Department of the Interior (114)<br/>U.S. Environmental Protection Agency (122)<br/>Township of Lower Alloways Creek (125)

## Rule Citation: § 61.55

<u>Summary of Comments</u>: These commenters were concerned with materials which may be present in low level radioactive waste which may be chemically toxic or hazardous. Some, but not all commenters, suggested that NRC's waste classification system incorporate a "total hazard "approach which would consider both the radiological and chemical hazard of a substance.

For example, commenter 53 thought that the rule should make "provision for handling of waste" which is both radioactive and requires special handling for other reasons." Commenter 103 also throught that consideration should be given to a definition of "toxic chemical/radioactive waste" which may require different handling and burial requirements," and recommended classification by total hazard.

Commenter 114 throught that any classification scheme should be based on total hazard. As stated by commenter 114, "it would seem inappropriate for a particular waste to be declared as Class A radiologically when it might contain toxic metals or organic compounds with potential harmful effects several orders of magnitude greater than those of the radionuclides. Perhaps the rule should either prohibit compounds with greater potential toxicity than the radionuclides or provide for additional classification options based on other-than-radiological toxicity." Commenter 68 suggested that "a general statement, at least, be included to the effect that releases of chemically-toxic substances shall not exceed any local or Federal standards that exist."

On the other hand, commenter 89 supported the concept of not developing a waste classification system based on total hazard. As stated by this commenter:

While concerns related to both the chemical and radiological components of a waste must be evaluated to determine the proper disposal method, the establishment of a total hazard classification system would be a redundant effort. The U.S. Environmental Protection Agency already addresses chemically-related concerns through the comprehensive regulations associated with the implementation of the Resource Conservation and Recovery Act (RCRA).

The proposed regulations properly address necessary radiation protection considerations and provide adequate guidelines for judging proper treatment or exclusion of nuclear of waste components. This avoids the need to develop a very complex classification system that somehow interrelates radiological and nonradiological concerns on a detailed basis.

### The U.S. EPA stated:

Hazardous and toxic chemicals are frequently present in these nuclear wastes. EPA is particularly concerned that these hazardous and toxic non-radioactive chemicals and their health impact are not considered in this proposed rule and EIS.

and

, i

As a minimum, Part 61 regulations should indicate that these materials must be handled in a manner compatible with RCRA requirements.

Valore, et al. (125) endorced the EPA concerns.

<u>Analysis of Comments</u>: The Commission has stated publically on several occasions that if it were technically feasible to classify waste by total hazard, then it would make eminently good sense to do so. However, unlike the technical methods for radiation, there is no reliable consistent system to relate a "dose" and potential health effects to exposure to chemicals. Therefore, there is no reliable consistent way to relate exposure to chemicals to exposure to radiation. See DEIS response to Argonne National Laboratory, Comment number 11, Item 3 for details. Thus a "total hazard" approach was not adopted.

The nonradiological hazards of the waste are not ignored. The operating disposal sites have general words on site licenses regarding chemical hazards not out weighing radioactive hazards. The prohibition against cardboard and fiberboard boxes was in part due to protection of workers from non-radiological hazards. Paragraph 61.56(a)(8) requires that wastes containing biological, pathogenic, or infectious material must be treated to reduce the potential hazard. The information requirements for shipping manifests in § 20.311(b) include specifying the chemical forms to the extent practicable. This requirement will provide a means of generating a more definitive data base on the chemical form of wastes being shipped for disposal. Finally, we have not observed any situation where migration of chemicals away from a nuclear disposal site has been a problem.

The DOE plans to support research into the development of a classification system for hazardous waste that might be compatible with Part 61. In the meantime, the staff will study the chemical toxicity of low-level waste, with special emphasis on identifying any licensees who generate hazardous wastes subject to requirements of the Environmental Protection Agency and re-examine what could be done, perhaps through processing, to minimize the hazard.

Furthermore, the Commission believes that the technical provisions of Part 61 generally meet or exceed those expected in the Environmental Protection Agency's rules for the disposal of hazardous wastes. Although it is not the Commission's intent to allow disposal of hazardous wastes in a radioactive waste disposal facility, as is noted in the regulation, the Commission recognizes that such wastes may be present in low-level radioactive wastes. It is the Commission's view that disposal of these combined wastes in accordance with the requirements of Part 61 will adequately protect the public health and safety. Such hazardous wastes are expected to be such a small percentage of the total volume that dilution by other wastes would greatly minimize any risks. The Commission intends to work closely with the Environmental Protection Agency to assure continued compatibility. Further, EPA in its response to a resolution of the Conference of Radiation Control Program Directors indicated their willingness to work with other Federal agencies to address this problem.

<u>ISSUE D-55-10</u>

	1330L D 33 10		1997 - 49 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1
Issue:	Radionuclide inventory vs. concent	ration limits	
<u>Commenters</u> :	Union of Concerned Scientists (36) Bechtel (44) Duke Power Co. (48) American Nuclear Society (87) The American Society of Mechanical New England Nuclear (110) U.S. Department of Energy (119)	Engineers (10	7), (113)

Rule Citation: Table 1 of Section 61.55; 61.7(b)(2)

Rule Change: None.

. :

<u>Summary of Comments</u>: The basis for the comments by these commenters are the statements in Table 1 that NRC intends to set maximum site inventories for H-3, C-14, Tc-99, and I-129 and in § 61.7(b)(2) that maximum disposal site inventories will be established. Several of the commenters requested the basis or criteria for setting the inventory limits. Commenters 44 and 87 thought that the criteria should take into account site conditions and locations (and also thought that "the criteria should reflect the fact that if a site is properly selected, a single large site may be more desirable than a series of smaller sites."). Commenter 110 throught that "prescriptive requirements which limit site inventory or which protect against excessive migration of radionuclides are site specific and should be incorporated in site licenses." This commenter also thought that "10 CFR 61 should specify that site licenses will incorporate these site specific prescriptive requirements." Commenter 107 suggested adding criteria and questioned the basis for site inventory limits. Commenter 48 questioned the need for inventory limits, stating that the 500 year intruder barrier should provide the protection required (apparently the commenter did not take into account the ground water aspects). Commenter 36, on the other hand, thought that a maximum site inventory should be determined for every isotope. The reason was again potential intrusion. As proposed by commenter 36, "adoption of a maximum site inventory for all isotopes would provide some protection in the event of inadvertent intrusion of a severity greater than that specified or in the event of other circumstances which might lead to unanticipated releases of radioactivity." DOE (commenter 115) suggested clarifying which nuclides are intended in § 61.7(b)(2).

<u>Analysis of Comments</u>: It appears that at least part of the comments on this issue are caused by confusion over why NRC is differentiating between waste concentrations and inventories. Concentration limits for radionuclides have been established based upon a number of considerations, including protection of a potential inadvertent intruder, operational safety, ground water migration, and long-term site stability. The desire to have the ability to set maximum site inventories for some isotopes is to have additional control over radionuclides that are of concern from a ground water point of view. Iodine-129, Tc-99, and C-14 are both mobile and long-lived. Tritium is of concern due to its extreme mobility and its presence in waste in large quantities. There is no reason to set maximum inventories for all radionuclides because they are of less concern for ground water migration. Controlling the migration of the mobile radionuclides will ensure control of the migration of the less mobile radionuclides. The basis for the establishment of inventory limits for some radionuclides would be to help ensure that the performance objectives for ground water migration are not exceeded. As suggested by commenters 44 and 110, any inventory limits would be site-specific in nature and would be established as part of licensing a particular site.

Conversely, NRC staff believe that there is no need to establish a site inventory limit for every isotope to protect against potential inadvertent intrusion. To begin with, inadvertent intruder exposures are mainly controlled by the concentration of a particular isotope, and to a lessor degree by the site inventory. NRC has also determined that to control such potential exposure to exceptable levels, concentrations limits for every isotope need not be specified. It is sufficient to control the disposal of a few key isotopes. Staff believes that only those isotopes which are of significant long-term toxicity and/or are reasonably expected to be in waste in large quantities should be listed in § 61.55.

## Rule Changes:

- 1. Clarify the footnote to Table 1.
- 2. Clarify § 61.7(b)(2) to reference that the purpose of the limits is to limit potential exposure.

## ISSUE D-55-11

Issue:

Cost and regulatory burden of classification requirements

<u>Commenters</u>: Nevada Department of Human Resources (14) Sargent and Lundy (24) Union Carbide (39) Arkansas Power and Light Company (52) American Nuclear Society (87) Florida Power (91) American Society of Mechanical Engineers (107)

Rule Citation: Technical Requirements and Table 1.

<u>Summary of Comments</u>: The comments pertains to various site operation and waste packaging requirements, in addition to waste classification requirements, which the commenters believe would raise the cost of waste disposal. Some of the scenarios that Commenter 14 discussed which he believes would result from the rule and/or raise costs include:

- o potential active maintenance costs would be assessed for Class A waste;
- o the site operator would consider all other (besides Class A) waste as Class C waste and bury it in a separate trench under 17 feet of earth;

the site operator would have to accept the radioactive waste as classified by the generator because he cannot open the packages for inspection;

0

0

0

o additional bookkeeping would be required, as well as new methods of disposal;

the regulatory agency governing the site would either require waste packages not labeled with a waste classification to be opened by the site operator at his expense, or have the waste returned by the operator at his expense;

the requirement that Class B waste be stabilized will require considerable (expensive) quality control on the part of the generator. For resins solidified in cement, dependence may have to be placed on the container for prevention of deformation, an additional expense.

Commenter 14 also believes that the waste classification and other requirements will increase regulatory burden and expenses. Two reasons are given for the commenter's concern:

- (1) The regulatory agency (in the commenter's opinion) will need to monitor charges collected by the site operator for active maintenance because "if the money was not all used during the five years after site closure, it should be turned over to the institution that will have custody of the site for long-term care and maintenance"; and
- (2) The regulatory agencies governing the generators would have to increase inspections of the actual waste packaging operations to determine that the packaged waste meets the requirements for different classifications. The commenter doubts that regulatory agencies can do more inspection in this area than they are doing now because of budget limitations.

Commenter 24 and others made very general statements that the classification scheme would increase costs of nuclear power and disposal costs in general.

Analysis of Comments: The issue of the cost of implementation of waste classification is strongly related to other issues such as Issue D-55-3 (10 nCi/gm TRU limit) Issue D-55-6 (compliance with waste classification), and Issue D-55-9 (basis for waste classification numbers). Although Commenters 24, 87, and 107 did not provided any clear basis for their claims, from the context of their responses it appears that much of the concern was related to the issue of determining compliance with the waste classification limits. These commenters also suggested that the "arbitrary" numbers in Table 1 be replaced with those from NUREG/CR-1005.

The comment that potential additional active maintenance costs would be assessed for Class A waste is an interesting one. Such an occurrence is possible, and appears to be reasonable from a technical point of view. Class A waste contains waste for which structual stability cannot be assured, and such waste have been demonstrated to principally contribute to subsidence of disposal trenches and subsequent maintenance.
The comment by Commenter 14 that the site operator would treat all Class B and Class C waste as Class C waste is not intended by the rule and NRC staff believe that this situation is unlikely to occur.

Regarding the comment that the site operator would have to accept waste as classified by the waste generator, this is no different than current practice today. Waste generators are already required by site operators to record on Radioactive Shipping Records (RSR's) information such as radionuclide content, presence of special nuclear material or source material, DOT transport class, chemical content, and so forth. Disposal facility operators accept this information as given. Limited opening of packages has occurred particularly at the South Carolina site but such practices are not routine and require special procedures and facilities. The rule does not require opening but does not prohibit it either. Spot checks can be a part of the operator's quality control program. Regarding the comment on regulatory agencies either opening unclassified waste or returning it to the waste generator, again there are already procedures at all waste sites to deal with waste shipments having improperly filled out RSR's. This is no different than the situation today.

Additional bookkeeping will indeed have to be carried out by waste generators and disposal facility operators, as will additional quality control. Increased attention and costs now mean reduced attention and costs later.

The comment that additional costs will result from quality control programs and use of high integrity containers is true. However, such costs are not expected to be large.

In analyzing the comment on state regulatory burden, NRC staff reviewed comments received on the draft rule and EIS from other state agencies (e.g., Commenters 4, 16, 38, 45, 47, 55, 65, 69, 79, 92, 93, 99, 109, 111, 112), particularly those from other states currently having operational disposal sites within their boundaries (79, 112), to determine if others had similar concerns. No such concerns were observed.

It is probably true that the waste classification and other requirements in 10 CFR 61 will increase NRC and State regulatory expense, if for no other reason than the new regulations exist. Any new requirements mean additional criteria to be complied with which means additional personnel hours to assure that the criteria are complied with. The additional costs are justifiable, however. There is a demand for the regulation. To a great extent, the waste form, waste classification, and other requirements in the Part 61 regulation are intended to reduce regulatory and other institutional costs at later dates. The rule trades somewhat higher short-term costs (i.e., costs while the disposal facility is operating) for reducing the uncertain, potentially very high, continuous long-term costs. Consider, for example, how much money and time NRC, Kentucky, and other government agencies have spent with respect to the Maxey Flats site.

NRC would expect that state regulatory agencies would monitor long-term care costs as a normal part of land owner responsibility related to waste classification. The incremental burden is judged small. This is being accomplished today. Costs of a reasonable inspection program should be included in the financial planning by the operator and State.

Inspection of waste generators for compliance with waste classification is more the responsibility of the Commission or the Agreement State regulating the generator. Existing regulatory responsibilities include inspection of the packaging and shipment of radioactive waste. The incremental burden of reviewing a licensee's program for classifying these wastes should be small.

5 T + 1

: .

でアメリアに変更し、

. • . <u>•</u> • •

<u>Rule Change</u>: None

#### ISSUE D-55-12

<u>Issue</u>: Waste classification - definitions clarifications, and miscellaneous other comments

Commenters:

. · ·

. . :

0

0

Nevada Department of Human Resources (14) United Technologies - Packard (25) Ohio EPA (38) UNIO EPA (38) Bechtel (44) State of New Mexico (45) Oswald U. Anders (73) Atomic Industrial Forum (100) U.S. Ecology (101) U.S. Ecology (101) American Society of Mechanical Engineers (107) New England Nuclear (110)

Rule Citation: 61.7(b), 61.55, and Table 1.

Summary of Comment: This group of commenters requested clarifications or definitions on some part of waste classification. Some of the points raised are listed below:

. . ,

Commenter 14. In paragraph 61.7(b), clarify what is meant by Class A waste decaying to acceptable levels during the period when the site is occupied. What are acceptable levels? (There could be no appreciable decay of uranium during the period the site is occupanied.) Does this mean that "receipt of Class A waste at the site would have to be stopped at some point in time before the site is closed, if it is to decay to acceptable levels prior to site closure?"

Commenter 25. Clarify that NRC is not mandating segregation of radioactive waste by radionuclide.

4 50.1

· **O** · Commenter 38. Believes that there is opportunity for confusion since Table 1 on page 38097 classifies waste by isotope whereas the classification on page 38085 is by point of origin.

- Commenters 44, 107. The term "significant gamma radiation" should 0 be defined.
- Commenter 45. 0
  - (a) Footnote 3 to Table 1 doesn't refer to any waste class but only to Class C waste.

- (b) Suggests that 61.55(d) be changed to read "Waste that has a radioisotope concentration that exceeds the numerical values shown in Column 3, Table 1. . ."
- o Commenters 73, 100, 107 and 110. The commenters note that the preamble to the draft rule states that the stability of the disposal site should last long enough for the radioisotopes to decay to levels where they are no longer of concern from a migration standpoint. The commenters believe that standards should be given.
- o Commenter 73. Raised the issue of the number of factors that should be included in the classification scheme and suggested that a much more complex computerized system be used.

#### Analysis of Comments:

- o Commenter 14. The line in question is "Even though the Class A segregated waste is unstable, it decays to acceptable levels during the period when the site is occupied and active maintenance can control water infiltration." This line is conceded to be poorly worded. Class A waste doesn't "decay to acceptable levels during the period when the site is occupied." Class A waste is such that even under the conditions of extreme waste degradation, potential intruder impacts at the end of the institutional control period will meet the intruder performance objective. In addition, the concentrations in the waste are low so that the likely radionuclide inventories in the Class A waste will also be low. Hence the ground water performance objective will be met even given the occurrence of increased water infiltration (relative to Class B and Class C waste) into the waste. It does not mean that "receipt of Class A waste at the site would have to be stopped at some point in time before the site is closed. . ."
- o Commenter 25. NRC is indeed not mandating segregation of radioactive waste by radionuclide.
- o Commenter 38. The classification on page 38085 by point of origin is only for illustration. It's only to give the reader an idea of the types of waste which would be expected in one class or another.
- Commenter 44, 107. The commenters are correct. The term "significant gamma radiation" is imprecise and should be eliminated. As discussed under Issue D-55-10 there is no need to set concentration limits for all nuclides.
- o Commenter 45:

.

(a) The comment is good. Change footnote to read "Maximum concentration for Class C waste, uCi/cm<sup>3</sup>" (or words to that effect).

#### (b) The suggestion is acceptable.

#### Commenters 73, 101, 107:

As noted in response to Commenter 14, the bottom line for disposal and decisions about the site is the performance objectives. Decay and limits on initial concentration must be such that at the end of the institutional control period potential intruder impacts will be acceptable. See Issue C-4 for a discussion of numerical limits for the intruder. For migration, site specific determinations must be made to determine expected compliance with the environmental performance objective. The 25 mr/yr is the standard.

Commenter 73. Commenter 73's suggestion that a complex computerized classification system that takes additional factors such as leachability and biodegradability into account might be feasible for a single waste stream where such a range of factors is known or can be determined. However, the diverse nature of low-level waste and the data available preclude such a complicated approach. Compliance would also be virtually impossible in view of the thousands of waste generators who would have to understand and use such a system. .

<u>Rule Changes</u>: Make changes as discussed above.

#### ISSUE D-55-13

1.17

0

.0

<u>Issue</u>: Stability and disposal requirements for Class A waste vs. Class B and C wastes

:

• • • •

5

<u>Commenters</u>: Pennsylvania Dept. of Environmental Resources (16) Dow Chemical (17) · · · Bechtel (44) Arizona State Clearinghouse (47) American Nuclear Society (87) Washington State Department (87) . .: . . . . . Washington State Department of Ecology (112) Department of Energy (119)

# Rule Citation: 61.52 and 61.55

Summary of Comments: These group of commenters in one way or another had some concerns or suggestions regarding the classification scheme that sets out , additional waste stability requirements for Class B and C waste streams. (The issue of stability and the operational implementation of the classification system is also addressed under Issues D-52-1 and 2.) For example, commenter 43 ... was unsure whether the stability requirements for Class B waste also applied to Class C waste. Commenters 16 and 119 suggested that NRC should clarify whether the requirements in paragraph 4 - 10 of section 61.52(a) applied to all classes of waste or just Class B waste. Commenter 16 also suggested that Class A meet all the requirements of (a) except (4). Commenter 17 thought that the stability requirements should be expanded to consider operational safety. That is, additional requirements on solidification should be imposed (using best available technology) based on assuring transportation and waste handling safety. . ř.

Commenters 44, 47, and 87 were not convinced that Class A and Class B wastes needed to be distinguished. Commenters 44 and 87 both suggested that Class A and B waste be defined solely on the basis of stability rather than stability and concentration so that only Class C waste would have to be stable. There would be no requirements to stabilize Class B waste, but stable waste which just happened to arrive at the disposal facility would be disposed of in a segregated manner from unstable waste.

Commenter 47 stated "if the NRC believes that specific characteristics of each waste necessitates separate burial, this has not been adequately demonstrated, either by their own evidence or by history." Commenter 47 thought that Class A and Class B waste should be mixed together during disposal (possibly to reduce intruder impacts). Commenter 47 spoke from the viewpoint of disposal in an arid climate, and wanted to know if the physical separation between waste classes could be vertical rather than horizontal (Class C on the bottom, followed by Class B and A).

Commenter 112 questioned whether unstable (Class A) and stable waste (Class B and C) need to be segregated at disposal sites such as the Richland, Washington disposal site, which are located in arid environments (the average precipitation at Richland is about 6 inches/yr). The commenter thought that among the most important concerns at an arid site are slumping and wind erosion and that co-mingling of waste classes should be allowed in an attempt to lower the average concentration of the most hazardous waste. The commenter also thought that implementation of the regulations would require a separate disposal unit to handle only Class C waste.

<u>Analysis of Comments</u>: Some of the comments appear to come from a misunderstanding of the rule. The waste characteristics that each class of waste must meet and the emplacement and disposal requirements need to be more clearly stated. The comments regarding vertical segregation and potential relief from segregation requirements at arid sites may have merit, although segregation would tend to reduce the overall level of slumping at an arid site, which is a concern of Commenter 112. Such proposals would be considered under § 61.54 "Alternative requirements for design and operations."

As noted under Issue D-56-7 when addressing the hazardous, biological, etc, properties of wastes, extensive treatment of Class A waste cannot be justified at this time. Stability for all waste is the most desirable option when cost effectiveness is not considered. However, development and implementation of the technology by all licensees will require time and may never be cost effective or even reasonable for small operations. Medical research, university research, and small scale industrial research would be significantly affected. The staff is encouraging the treatment of wastes and also added a provision to the rule that if Class A waste is stable, it does not have to be segregated. Thus a specific alternative would not have to be approved under § 61.54 for stable Class A waste. Staff judges the requirements on Class B and C waste to be justified and no relaxation should be considered.

#### Rule Change:

1

1. Change Section 61.52 so that the applicability of subparagraphs 4 - 10 is clarified.

2. Regarding the stability requirements for Class B and C wastes, the rule need not be changed. Reiterate the points in section 61.55 regarding the minimum and stability requirements in the preamble to the final rule.

S. S. B. Konski and

. • •

# ISSUE D-55-14

<u>Issue</u> :	Hold wastes for decay	
<u>Commenters</u> :	Advance Medical and Research Center, Inc. (5) Northern Illinois University (27) Sierra Club (37)	
•••••••••••••••••••••••••••••••••••••••	R/A Services, Inc. (57) Texas Department of Health (117)	

Rule Citation: §§61.55 and 61.56, Table 1

*.* ••

· • • : •

<u>Summary of Commenter</u>: The commenters were concerned that the waste classification and characteristics requirements in §§61.55 and 61.56 precluded individual licensees from holding short lived isotopes for decay to background levels and subsequent disposal in the ordinary trash. Two commenters noted that hold for decay should be used for most medical/academic wastes and one noted that materials held for decay could be retrieved for reuse.

<u>Analysis of Comments</u>: The comments reflect a need to clarify the purpose, scope, and applicability of the regulation. Part 61 and associated changes would not limit disposal options available to licensees except when transferring wastes containing licensed materials for disposal at a land disposal facility. Options such as on-site burial, hold-for-decay, incineration, or disposal of materials exempted from the provisions of Part 20 remain unchanged.

The Northern Illinois University expressed opposition to near-surface disposal of wastes and noted that most medical/academic wastes can be segregated and held for decay instead of disposal in near-surface facilities. Staff agrees that much of this wastes can be held and has encouraged licensees to adopt this alternative.

Rule changes: None.

> ne setter in an anna 1997. Setter anna 1997 - Setter anna 1997

### ISSUE D-56-1

Issue: Use of absorbents

<u>Commenters</u>: Oswald Anders - 73 State of South Carolina - 79

<u>Rule Citation:</u> 61.56(a)(3)

Section 61.56(a)(3) allows the use of absorbents for Class A wastes provided that the absorbent is capable of absorbing twice the volume of liquid.

The State of South Carolina stated that the requirement should apply only to institutionally generated, aqueous or biological waste forms and not to organic materials.

Mr. Anders stated that absortion was an obsolete technique for immobilizing liquid wastes and was unsatisfactory. He indicated that nuclides could be easily leached from such waste forms.

The NRC staff disagrees that absorbents should only be used for aqueous wastes. There are many organic waste forms which cannot be effectively solidified and absorption is the only effective technique for immobilization. The NRC staff, however, recognizes that certain site specific conditions may restrict the disposal of absorbed organics and agrees with the State of South Carolina that such conditions need to be considered in evaluating the environmental acceptability of these wastes.

Since nuclear power plant waste generators currently solidify aqueous waste streams, Section 61.56(a)(3) is not expected to affect their operations. The NRC staff also believes that nuclear power plant waste generators would generally not use absorption since this technique would result in increased waste volumes to be shipped.

Because the use of absorbents would be only applicable to Class A wastes, pathway evaluations indicate that their use would not result in public health hazards due to leaching of radionuclides.

Because of the low activities of the Class A wastes and the current solidification of liquids at nuclear power plants, the NRC staff believes that it is unnecessary to restrict the use of absorbtion to only institutional waste generators.

Rule Change: None

#### ISSUE D-56-2

Issue: Chelating Agents

<u>Commenters</u>: Advisory Committee on Reactor Safeguards - 10 Wisconsin Electric Power - 32 Alabama Power - 33 Commonwealth Edison - 35 Union of Concerned Scientists - 36 Bechtel National - 44 Duke Power - 48 Utility Nuclear Waste Management Group - 81 American Nuclear Society - 87 Power Authority State of New York - 92 Arkansas Power & Light Co. - 94 Health Physics Society - 96 Atomic Industrial Forum - 100 Carolina Power and Light - 106 The American Society of Mechanical Engineers - 107 State of North Carolina - 109 New England Nuclear - 110 U.S. Department of Energy - 119

Rule Citation: Table 1 of Section 61.55.

<u>Summary of Comments</u>: The utilities, the utility groups, one professional society and DOE stated that the requirement to obtain specific approval to dispose of wastes containing greater than 0.1 percent chelating agents was too restrictive. In addition, they stated that in some cases this could result in utilities deciding against performing decontamination operations which could reduce occupational exposures. Several utilities requested that the basis of the 0.1 percent value be provided. The Union of concerned Scientists stated that disposal of wastes with chelating agents over 0.1 percent should not be permitted under any conditions. One utility stated that generic disposal requirements should be specified rather than case-by-case evaluations. The ACRS stated that the case-by-case evaluation concept was not clearly presented in the proposed rule. Several commenters also questioned packaging requirements and whether the 0.1% criterion was by weight or volume. Several recommended moving the requirements from the footnotes in Table 1.

Analysis of Comments: The proposed regulation 10 CFR 61 stated that disposal of wastes containing greater than 0.1 percent chelating agents would not be permitted without specific approval by the NRC. Since chelating agents have been shown to increase the migration of certain nuclides at certain sites, the NRC staff desired to evaluate the disposal of large quantities of wastes containing high concentrations of chelating agents on a case-by-case basis. This evaluation would consist of a review of the burial site conditions and the form of the wastes. A similar approach was used when the NRC staff reviewed the disposal of wastes which would be generated in the decontamination operations at the Dresden Unit 1 Station. In this case the NRC mandated that, in order to minimize any migration effects which might be exacerbated by chelating agents, the Dresden decontamination wastes should be disposed of at an arid disposal site. At an arid site leaching would be minimized due to the low precipitation rates.

The 0.1 percent chelating agent value is based on the amount of iron in a 55-gal drum which might normally be available to complex with unreacted chelating agents if these chelating agents leached from the waste in an uncomplexed form. If uncomplexed chelating agents complexed with iron from the waste container, it would generally be unavailable to complex with other

radionuclides in the nearby soils. Therefore, radionuclide migration would be minimized.

Because the disposal of wastes containing chelating agents is dependent on the site characteristics of the disposal facility and on the properties of the waste form, the NRC staff has modified the chelating agent disposal requirements to reflect the site specific nature of the disposal of these wastes. The Table 1 footnote that wastes containing greater than 0.1 percent chelating agents would not be permitted unless specifically approved has been replaced by more general requirements. However, in order that disposal of chelating agent wastes are considered in licensing actions, Section 61.12(f) now requires that license applications specifically address methods for handling these wastes. These methods would be developed based on the site conditions and on the allowable waste forms. The manifesting requirements (Section 20.311(b)) now require that waste generators identify wastes which contain greater 0.1 percent by weight chelating agents. This will allow disposal site operators to identify, segregate, and dispose of chelating agent wastes in accordance with individual site license conditions.

The NRC staff believes that the above changes in 10 CFR 61 will not be overly restrictive to utilities but will provide assurance that chelating agent wastes will be properly disposed of. The NRC staff believes that the wastes from decontamination processes which are currently available can be disposed in an acceptable manner. In some cases, though, these wastes may require disposal at an arid site. The NRC staff does not believe that disposal of chelating agent wastes will be or should be the limiting point for utility decisions for performing decontamination operations to reduce occupational exposures.

This modification to 10 CFR 61 would also eliminate the necessity for NRC staff to perform case-by-case evalutions. Instead, specific license conditions at each disposal site would provide the disposal requirements for chelating agent wastes. While this change does not provide detailed generic requirements for waste generators, it does provide flexibility for disposing of these wastes based on the specific disposal site characteristics and the specific waste form properties. The NRC staff believes that the revisions eliminate the confusion regarding the case-by-case evaluations as presented in the ACRS comment.

Several commenters requested specific guidance on packaging chelating agents. Since the disposal of chelating agent wastes are site dependent, there may be many acceptable options for the disposal of these wastes. The NRC staff plans to address acceptable waste form options in a branch technical psoition (BTP). This BTP would provide the basis for disposal site applicants and operators to develop disposal parameters which would provide for the proper disposal of these wastes.

Several commenters requested clarification on whether the 0.1 percent concentration value was to be determined on a percent volume or percent weight basis. One commenter suggested using a percent volume basis. The calculation to determine the 0.1 percent value was performed on a weight basis. This point is clarified in the final rule. Two commenters stated that the chelating agent requirements should be moved from the footnote section of Table 1 and placed on the body of the rule. The revised requirements have been moved into the body of the rule.

511-1613**1**2777

÷

#### Rule <u>Changes</u>:

....

.

- Add to § 61.12: A requirement for addressing wastes containing chelating 1. agents.
- 2. Add to 20.311(b) after "The solidification agent must be specified.": Wastes containing more than 0.1% by weight chelating agents must be identified and the weight of the chelating agents estimated.

#### ISSUE D-56-3

Issue: Definitions of pyrophoric, hazardous and explosive

Commenters: Advisory Committee on Reactor Safeguards - 10 New England Nuclear - 110 ;

Rule Citation: 61.56(a)(4), 61.56(a)(5), 61.56(a)(6)

· · ·

Analysis of Comments: The intent of Sections 61.56(a)(4), 61.56(a)(5) and 61.56(a)(6 is to provide protection to workers during waste handling and disposal operations by prohibiting wastes which are explosive, pyrophoric or generate toxic fumes. · .

One commenter indicated that the restrictions on wastes which are pyrophoric, explosive, or generate toxic gases may be difficult to enforce without clear definitions for these terms. Another commenter suggested new wording (add "normall") for Section 61.56(a)(5) to ensure that plastics are not excluded. Staff agrees. 4 4 4 A A

Definitions of the terms, "hazardous," "pyrophoric" and "explosive" have been added to Section 61.2. The definitions for "pyrophoric" and "explosive" are consistent with those in the transportation regulations 49 CFR 171.8, 49 CFR 173.50, 49 CFR 173.115, and 49 CFR 173.150. The term, "toxic," has been deleted and "hazardous" substituted. A definition for "hazardous" has been included in Section 61.2 to reference the Environmental Protection Agency requirements in 40 CFR 261.

#### Rule Changes:

1

### Proposed Definitions to add to § 61.2: 1.

Pyrophoric: A pyrophoric liquid is any liquid that ignites spontaneously in dry or moist air at or below 130°F (54.5°C). A pyrophoric solid is any solid material, other than one classed as an explosive, which, under conditions normally incident to transportation is liable to cause fires through friction, retained heat from manufacturing, or processing, or which can be ignited readily and when ignited burns so vigorously and persistently as to create a serious transportation, handling, or disposal hazard. Included in this class are spontaneously combustible and waterreactive materials. 

 $\sim 1^{-1}$   $\sim 1^{-1}$ 

Explosive: An explosive material is defined as any chemical compound, mixture, or device, which produces a substantial instantaneous release of gas and heat spontaneously or by contact with sparks or flame.

"Hazardous waste" means those wastes designated as hazardous by Environmental Protection Agency regulations in 40 CFR Part 261.

2. Amend § 61.56(a)(5) to read:

(5) Waste must not contain, or be normally capable of generating, quantities of toxic gases, vapors, or fumes harmful to persons transporting, transporting, handling, or disposing of the waste.

#### ISSUE D-56-4

Issue: Cardboard containers

<u>Commenter</u>: University of California, Los Angeles (8) Sierra Club (37) Ontario Hydro (51) Union Oil Company of California (66) Department of Energy (119)

### <u>Rule Citation:</u> 61.56(a)(2)

<u>Summary of Comments</u>: Ontario hydro (51) questioned why wastes packaged in cardboard or fibreboard boxes are prohibited. If the waste can be segregated then these types of packaging should be acceptable.

University of California, Los Angeles thinks that prohibiting the use of cardboard or fiberboard boxes for low level radioactive waste is unnecessary. The major reason for this conclusion was that cardboard boxes could be introduced directly into an incinerator, while metal drums would have to be unpacked.

Union Oil Company of California suggested Section 61.56(a)(2) should be deleted. For certain low level waste use of cardboard or fiberboard boxes may be suitable for both shipping and disposal.

The Sierra Club supported the prohibition against packaging wastes in cardboard or fiberboard boxes.

The DOE noted that DOT regulations allow cardboard and fiberboard and DOE wastes are safely handled in such containers and suggested deleting paragraph 61.56(a)(2).

<u>Analysis of Comments</u>: The NRC staff after reviewing the comments regarding the prohibition or cardboard and fibreboard containers still believe that such a prohibition is needed. The DOE experience at DOE facilities has shown no difficulty for DOE wastes. However, no extensive handling or shipping are involved and greater administrative controls are possible since DOE wastes are typically buried on site and are generated by a few organizational groups operating under common guidelines. Commercial sector wastes are shipped longer distances and generated by thousands of organizations. Biodegradable packaging has merit in enhancing decomposition of unstable wastes and accelerating the

compacting process and DOE should continue the practice where experience so warrents.

Cardboard and fiberboard containers are found frequently reaching the commercial disposal sites in a ruptured condition. In addition, syringes used in research and in hospitals have sometimes penetrated containers causing injury to disposal site workers. This prohibition is consistent with current license conditions at the Hanford and Barnwell disposal sites.

The requirement to prohibit disposal of wastes in cardboard or fiberboard containers would not affect packaging prior to incineration. In this case such packaging could be desirable. See rule change 6 under Issue M-1 also.

<u>Rule Change</u>: None <u>ISSUE D-56-5</u> Wastos in a gaseous form

<u>Commenters</u>: University of North Carolina - 30 Sierra Club - 37 Los Alamos National Laboratory/Safeguards Systems Group - 43 Bechtel National - 44 Georgia Institute of Technology - 70 American Nuclear Society - 87 Atomic Industrial Forum - 100 U.S. Ecology - 101 The American Society of Mechanical Engineers - 107 New England Nuclear - 110 U.S. Department of Energy - 119

<u>Rule Citation</u>: § 61.56(a)(5) and (a)(7)

2 15 11 Analysis of Comments: The NRC staff has provided requirements for the disposal of wastes in gaseous forms in Sections 61.56(a)(5) and 61.56(a)(7). These requirements are based on license conditions currently in use and are intended to provide protection to disposal site workers and to reduce airborne releases.

Several commenters indicated that there was an inconsistency between Section 61.56(a)(5) and 61.56(a)(7). That is, Section 61.56(a)(5) specifies that wastes shall not be capable of generating hazardous gases, vapors or fumes and Section 61.56(a)(7) authorizes disposal of gases not to exceed 100 Ci per container. Several commenters requested that NRC provide the basis for the 100 Ci limit. Another commenter stated that gases should be processed into liquid or solid forms which would provide better long-term stability since it  $\sim$  would be impossible to assure that containers would last for the 150 years stability period. Other commenters stated that wastes disposed of in a gaseous form should be limited or prohibited and that the general population off-site and downwind should be protected. Staff believes that measures to protect transporters and workers provides ample protection for off-site.

The intent of Section 61.56(a)(5) which specifies that wastes shall not be capable of generating hazardous gases, vapors, or fumes is intended to prohibit the disposal of wastes which are chemically reactive under ambient conditions

and produce hazardous gaseous reaction products. This section was not intended to prohibit the disposal of properly packaged gases such as  $H_3$  or  $Kr^{85}$ , which occasionally require disposal. Section 61.56(a)(5) has been reworded to clarify the NRC staff's intent.

The 100 Ci limit is based on the license conditions for the disposal of gaseous wastes now in effect at the Hanford and Barnwell disposal sites. These limits have not resulted in unsafe environmental conditions at the disposal sites nor have they resulted in overly restrictive situations for waste generators.

The NRC is currently evaluating the significant generators of  $H_3$  gas with the intent to improve packaging designs. The objective of these evaluations are to assure that package designs retain  $H_3$  until it has decayed to insignificant levels. Since  $H_3$  has only a 12 year half-life the package design needs only to retain the  $H_3$  for approximately 120 years to provide a  $10^3$  reduction in radioactivity. The overall structural stability of the package will be oriented to providing trench stability for Class B and C wastes for a 300 year period consistent with the high integrity design concepts.

For  $Kr^{85}$  packages the release by airborne pathways of  $Kr^{85}$  is not as significant as for H<sub>3</sub> since  $Kr^{85}$  is an inert gas which is not concentrated in living tissue. Therefore, complete retention of  $Kr^{85}$  in a waste package over the period required for 10 half-lives of decay is not essential. The NRC staff believes that by limiting the curie content in the waste packages designed for long term stability that public health and safety can be protected.

If, in the future, large quantities of  $Kr^{85}$  are contemplated for disposal, the NRC staff will reevaluate the requirements for the disposal of  $Kr^{85}$  gas in order to assure that disposal is consistent with the disposal site performance objectives.

The American Society of Mechanical Engineers suggested modifying 61.56(a)(7) to make it less absolute. The Society recommended inserting "significantly" before "exceed one atmosphere at 20°C." The requirement of 61.56(a)(7) is consistent with the current license conditions at Barnwell and Hanford. The NRC staff believes that use of the term "significantly" will only increase the difficulties in enforcing this requirement.

The DOE commented that the maximum concentration for Kr-85 in a Class B waste would be 44 Ci/m<sup>3</sup> or 440 Ci/m<sup>3</sup> if contained in a metal. These are the same concentrations as for Cs-137. The DOE commented that this would eliminate disposal in dry wells of Kr-85 immobilized by zeolite encapsulation by deposition on metals. The NRC staff disagrees that 10 CFR 61 elminates this disposal. The regulation 10 CFR 61 applies to near surface burial facilities and requirements for other alternative methods for disposal of low-level wastes would be considered in separate rulemaking actions. At this time the health and safety impacts of the disposal of immobilized Kr-85 in dry wells could be considered.

Rule Change: Change 1 atmosphere to 1.5 atmospheres at 20°C.

### ISSUE D-56-6

is environmente

<u>Issue</u>:

Packaging standards

Commenter:

Department of the Environment, London (19) Union Oil Company of California (66) Stock Equipment Company (67) General Electric (89) Stone and Webster Engineering Corporation (95) The American Society of Mechanical Engineers (107 and 113) New England Nuclear (110) The U.S. Department of Energy (119)

Rule Citation: Section 61.56(a)(1), (a)(7)

<u>Summary of Comments</u>: Paragraph 61.56(a)(1) requires that the waste packages presented for disposal comply with NRC and DOT transportation regulations. This implies that the disposed package could be a Type A, or Type B Package including all related shielding and other transportation-related requirements. Two commenters (95,113) stated it is unlikely that this is NRC's intent but the wording of the paragraph can be interpreted in this manner. They suggest NRC should clarify and reword this requirement.

Two commenters (19 and 119) commented that Section 61.56 is not explicit with respect to disposal of packaging materials. In particular it is not clear whether wastes should be packaged for disposal (thus prohibiting the use of returnable shielded packages) or whether they only need to be packaged for transport and handling prior to disposal.

Stock Equipment Company (67) suggested that all waste containers be constructed of materials that will not support combustion.

Union Oil (66), GE (89), and the ASME (107) suggested that disposal of bulk or unpackaged wastes be permitted.

New England Nuclear (110) suggested that the activity limits on containers implied in § 61.56(a)(1) and specified in (a)(7) are overly restrictive.

<u>Analysis of Comments</u>: The issue of waste packaging requirements as stated in Section 61.56(a)(1) is confusing to the waste generator, carrier, and receivers. NRC staff has, therefore, deleted this section because there is no need to restate in 10 CFR 61 that the waste packages shipped for disposal must comply with NRC and DOT transportation regulations. The Commission did not intend that returnable or reusable shielding must be disposed of.

Stock Equipment Company (67) referenced a recent NRC guidance document, "Radiological Safety/Guidance for On Site Contigency Storage Capacity" which requires that low level dry waste storage be in containers that do not support combustion. The Company suggests that the requirement be applied all wastes that are transported and disposed of. The hazard of fires during handling and transportation is included in the determination of the LSA, Type A, and Type B quantity limits. Since waste packages will meet the NRC and DOT transportation requirements, NRC staff believes it is unnecessary to further restrict packaging materials. Under the rule disposal of bulk or unpackaged waste is allowed provided that the requirements for waste classification and waste characteristics are met. The staff agrees that disposal of bulk wastes may be the most efficient use of space or easiest way to minimize void spaces under certain circumstances and that certain types of wastes may be shipped in bulk under DOT regulations. The requirement for packaged wastes has been deleted.

New England Nuclear expressed the view that the container activity limits in § 61.56(a)(1) and (7) are excessively conservative. Paragraph (a)(1) imposes DOT activity limits on containers by implication. Paragraph 61.56(a)(7) has explicit activity limits on gaseous shipments. NEN raised the point that wastes processed on site after transit and before disposal should be allowed to exceed DOT limits. The staff agrees with this position. The staff intended that Table 1 set limits on allowable disposal concentrations, not quantity restrictions for transportation. The gaseous limits are addressed under issue D-56-5.

Rule Change: Delete § 61.56(a)(1).

#### ISSUE D-56-7

Issue: Hazardous, biological, pathogenic and infectious waste treatment

<u>Commenters</u>: University of Texas Medical Branch (75) University of Texas System Cancer Center (105)

#### Rule Citation: 61.56(a)(8); 61.56(b)

<u>Summary of Comments</u>: The Medical Branch objected to treatment of all biological specimens under 61.56(a)(8) and noted that if liquid scintillation fluids are not allowed, Agreement State licensees in particular, will have a problem. The Cancer Center suggested that liquid scintillation fluids be specifically referenced and clearly exempted from stability requirements in 61.56(b)

<u>Analysis of Comments</u>: The NRC staff agree that treatment (e.g., incineration of liquid scintillation wastes as recommended in NUREG-0656) of wastes is the most desirable option in many instances. However, development and implementation of the technology by all licensees will require time and may never be cost effective or even reasonable for small operations. Medical research, university research, and small scale industrial research would be significantly affected. The staff cannot justify this burden at this time but is encouraging the adoption of alternative volume reduction treatments. Absorbed liquid scintillation wastes are acceptable under § 61.56 and are not routinely subject to the stability requirements since activity levels fall in Class A.

The requirement in § 61.56(a)(8) specifies that biological, pathogenic, or infectious waste material be treated to reduce the non-radiologic hazards of these wastes to the extent practicable. The intent was not to require incineration or similiar treatment of all biological materials. The requirement in (a)(8) is intended as an objective. Specific license conditions or site specific administrative procedures would be expected to address this issue in more detail. For example, the license issued by the State of Washington for the Richland site requires absorbents, lime, and double containment for animal carcasses. Such packaging would be a means of complying with (a)(8). The rule also permits the disposal of absorbed liquid scintillation fluids as Class A wastes. All licensees, however, are encouraged to develop alternative methods of managing

these wastes. The Commission's volume reduction policy statement dated October 16, 1981 also encourages all waste generators to reduce waste volumes to the extent practicable.

. 1

Rule Changes: None.

· 7,

.

۰.

۰.

2

.

.

•••

#### ISSUE D-56-8

· . · · · ·

· · · · ·

·: ·· ·

. **.** . . · · · · ·

.

<u>Issue</u> :	Waste stability requirements	- 
<u>Commenters</u> :	University of California at Los Angeles - 8 Advisory Committee on Reactor Safeguards - 10 DOW Chemical - 17	۰۰، ۲۰۰۰ ۲۰۰۰ ۲۰۰۰
and the second states	Sargent and Lundy - 24	
	University of North Carolina - 30	· • .
; , ,	Alahama Power - 33	
f3+	law Engineering Testing Co 34	
`.	Sierra Club - 37	· · · · · · · · · · · · · · · · · · ·
	Duke University - 48	
	Stock Equipment - 67	
	Argonne National Laboratory - 68	
	Oswald U. Anders - 73	
	Birmingham Audubon Society - 80	
•••	Utility Nuclear Waste Management Group - 81	
	Middle South Services - 84	<u>, 1</u> 4
· • • •	Northeast Utilities - 85	
	American Nuclear Society - 8/	
and the second	Health Physics Society - 96	
1 ( ) · · · · · · · · · · · · · · · · · ·	Atomic Industrial Forum - 100	•
teda en la com	U.S. ECOlogy - 101 American Institute of Chemical Engineers - 102	
	American institute of themital Engineers - 102 Conforance of Padiation Control Program Director	$ac = 102^{\circ}$
S119 1	American Society of Machanical Engineers - 107	5 103
i it is atting a	New England Nuclear - 110	. :
. · · · · · · · · · · · · · · · · · · ·	Tennessee Valley Authority - 116	, <b>*</b> •
	U.S. Department of Energy - 119	j.

<u>Rule Citation</u>: 61.56(b); 61.7(b)(2)

Analysis of Comments: The concept of waste stability is intended to provide long term assurance that the disposal units will maintain their integrity without substantial and expensive remedial maintenance efforts. A stable disposal unit will also minimize the infiltration of water which could enhance migration of radionuclides.

.

Several commenters indicated that the meaning of stability was unclear and requested further specific guidance. One commenter indicated that structural rather than chemical stability was intended.

In order to clarify the term "stability" a definition has been added in Section 61.2. The NRC staff is also preparing a Branch Technical Position (BTP) on waste form which will provide acceptable test methods and criteria to be used to qualify stabile waste forms.

One commenter asked if trash would require stabilization. Trash, which generally has low specific activities, would normally be a Class A waste. However, some trash wastes may have activities which exceed the Class A limits. These wastes would be required to be stabilized or packaged in a container providing stability if the potential for degradation exists.

One commenter stated that if the disposal facility was sited properly waste stability would not be required. The AICE expressed the view that site maintenance for 100 years would take care of subsidence and waste stability was not justified. Anders objected to depending on steel drum waste packages for stability and believed that waste form and efficient backfilling were the key factors.

Siting requirements for a disposal facility are extremely important in assuring that radionuclide migration pathways are minimized. However, even at a well selected site, waste degradation can cause subsidence which could result in expensive remedial action programs. Such expensive maintenance could rapidly deplete perpetual care funds established for remedial care, monitoring and site surveillance. Therefore, trench stability is an important factor to be considered in addition to site suitability.

Two commenters supported the concept of waste stability.

The 150 year stability time period was chosen based on the sum of the site operating period, post closure observation period and the institutional control period. It was intended that wastes remain recognizable and stable over this period.

One commenter suggested that the 150 year stability requirement was overly restrictive since void spaces between waste packages would be filled. This commenter stated that voids were the cause of subsidence and suggested that the elimination of voids would eliminate subsidence. The commenter also stated additional processing would be required to meet this requirement resulting in increased waste volumes. Two other commenters stated that waste dispersion must be prevented and that this was only slightly affected by slumping or deformation.

Two commenters including TVA stated that the stability requirement should only be for 100 years. TVA based the comment on 100 years of institutional control and the decay during that time. One commenters asked what the basis of the 150 year stability requirement since some nuclides would not decay during that period. Another commenter stated that Class A wastes should also have a stability requirement since cesium (Cs-137) and strontium (Sr-90) would also be present in these wastes.

One commenter stated that steel drums can not be expected to last beyond 30 to 60 years. The commenter stated that there was no basis for assuming that steel will not corrode in 150 years. Another commenter stated that high integrity containers have not been tested for 150 years.

The NRC staff has reviewed the 150 year stability requirement with respect to the scenarios used to calculate the waste classification values. The scenario used to calculate the Class A interface assumes that after the 100 year institutional control period (approximately 150 years since the facility began operation), the wastes are unrecognizable. The intruder, therefore, fails to recognize that he is performing construction, agricultural, or residential activities on a waste disposal site.

For Class B wastes, it is assumed that after institutional control is lost, an inadvertent intruder begins construction activities. However, due to the stability requirement, the waste is recognizable. The intruder realizes that he has begun construction in an area where previous disposal activities have taken place and he abandons his effort. The Class B concentrations are based on the intruder receiving 500 mrem in the course of discovering the disposal site.

The Class C intruder scenario assumes that the intruder barrier is effective for 500 years. At 500 years all waste is unrecognizable and the intruder performs the same activities as in the Class A waste scenarios. The difference here, of course, is that the waste has undergone 400 additional years of decay.

Looking at the above scenarios, the stable waste forms should be distinctly recognizable at 150 years for those wastes emplaced when the facility opens. It will not be necessary for the carbon steel drums to remain. For solidified wastes, though, the monolith should still remain intact. At 300 years the wastes should continue to maintain its gross physical properties and also retain a measure of its identity. High integrity containers are designed with a lifetime goal of 300 years. This lifetime also assures that a 10 half-life decay period will be applied for CS-137 and Sr-90 (30 year half-life nuclides) which are removed primarily in ion-exchange resins. Ion-exchange resins are a primary waste stream disposed of in high integrity containers.

Following the 300 year lifetime, waste degradation is expected to gradually occur. Based on the ancient cement structures, it is expected that cement products can exceed this lifetime. High density, high molecular weight polyethylene (used in the current high integrity containers) and polymer solification agents are also expected to exceed this lifetime, based on their chemical resistance, radiation stability and biodegradation properties.

The NRC staff has modified the waste stability period from 150 to 300 years to be consistent with the Class B and C waste classification scenarios.

The presence of Cs-137 and Sr-90 has been considered in the development of the Class A limits for the waste classification system. The fact that Cs-137 and Sr-90 do not decay as fast as the shorter half-live nuclides, such as Co-60, is the reason why the Class A limits for these two nuclides are lower.

The assertion that eliminating voids will provide stability is not complete since chemical, radiologic, and biodegradation will also lead to trench subsidence. It is expected that if currently used solidification agents, cement, vinyl ester styrene, and asphalt, are used with a good process control program that the necessary stability will be achieved. Therefore, volume increases are not expected over current waste generation.

The scenarios used in the waste classification system show that waste stability for Class B and C wastes is the most important factor. A stable waste implies a non-dispersive waste. Therefore, the comments that dispersion is more important than slumping and deformation is only partially correct. Stability will minimize the site maintenance requirements during the institutional control period and the resultant burdens on the perpuetal care funds set up for remedial activities.

Three commenters stated that there was a need to specify tests and criteria to demonstrate the stability requirement.

The NRC staff is preparing a Branch Technical Position on Waste Form which will provide acceptable tests and criteria to demonstrate waste stability.

One commenter asked if wastes would be allowed to degrade if a solidification binder which did not degrade was used.

Wastes within a binder may degrade so long as the gross physical properties of the monolith are not impaired.

#### Rule Changes:

1. Delete 150 years from 61.56(b).

2. Add a 300 year objective for Class B and C wastes to the "Concepts" section.

#### ISSUE D-56-9

Issue:

.

Five percent deformation limit

Alabama Power - 33 Commenters: Union Carbide - 39 Bechtel National - 44 Utility Nuclear Waste Management Group - 81 Werner and Pfleiderer - 82, 124 General Electric - 89 Power Authority of the State of New York - 92 Stone and Webster - 95 Atomic Industrial Forum - 100 U.S. Ecology -101Carolina Power and Light - 106 American Society of Mechanical Engineers - 107, 113 State of North Carolina - 109 New England Nuclear - 110 U.S. Department of Energy (119)

#### Rule Citation: 61.56(b)(1)

Analysis of Comments: The requirement that a stable waste form maintain its physical dimensions within five percent was intended to ensure that degradation of waste containers and waste forms would not lead to severe trench subsidence. The five percent value was selected as a conservative deformation level which would be expected to produce only minor subsidence effects. These minor subsidence effects could be easily resolved by an inexpensive post closure maintenance program at the burial site.

Commenters indicated that the five percent deformation was overly restrictive and impossible to achieve due to the impracticality of filling containers to 95 percent capacity. In order to prevent spills, container filling operations are routinely terminated prior to reaching 95 percent capacity.

Commenters also noted that asphalt and polymetric solidification agents would be incapable of meeting this requirement because of their viscoelastic creep properties. Under static burial site load conditions, asphalt and polymetric material will flow or creep over time periods on the order of years until internal pressures are equalized. Werner and Pfleiderer suggested alternate wording on stability and thermoplastics.

Commenters indicated that the five percent limit would force waste generators into using high integrity containers. Use of high integrity containers would be required because normal containers could not be filled to 95 percent capacity and plastic solidification materials would be unacceptable due to creep.

One commenter indicated that the five percent tolerance limit should be restudied with careful consideration given to the added expenses that would be entailed.

One commenter asked for a clarification on what was actually meant by the term "within 5 percent."

Union Carbide suggested that the 5% apply to volume rather than dimension "since a 5% deflection in the diameter of a drum is inconsequential."

The NRC staff believes that trench subsidence needs to be carefully controlled in order to minimize water infiltration and major remedial care operations. In order to achieve this objective in a pragmatic way, the NRC staff has chosen to delete the prescriptive requirement for a five percent dimensional limit. The requirement that wastes remain structurally stable, however, remains in place. In order to provide guidance to waste generators on acceptable waste forms the NRC staff is preparing a Branch Technical Position (BTP) which defines acceptable test methods and criteria oriented to ensuring structural stability.

In addition to the BTP, 10 CFR 61 Section 61.52(a)(4) and 61.52(a)(5) requires that wastes be emplaced to maintain package integrity and that void spaces be filled to reduce subsidence. The filling of void spaces will minimize the creep effects in asphalt or polymeric solidified products since lateral stability will be provided by the fill material. If there are no voids, there will be no space for viscoelastic wastes to flow into.

The commenters concerns regarding the void spaces inherent in waste containers, are addressed under that specific issue.

Rule Change: Delete 5% limit from § 61.56(b)(1).

ISSUE D-56-10

Issue: Void spaces in waste containers

Commenters:

Sargent and Lundy - 24 Bechtel National - 44 American Nuclear Society - 87 General Electric - 89 Health Physics Society - 96 American Society of Mechanical Engineers - 107

### Rule Citation: 61.56(b)(3)

Analysis of Comments: The proposed regulation specifies that void spaces within containers must be reduced to the extent practicable. The intent of this requirement is to minimize subsidence effects following the degradation of the container or waste product. In some cases waste containers are filled to 50 to 75 percent of the container volume. Our objective is to fill containers to 85 to 95 percent of container volume when it is practicable to do so. Components such as pumps or valves will have large internal void spaces which can not easily be eliminated. These components, however, would not be expected to undergo substantial degradation over short periods of time and will not present severe trench subsidence problems.

Several commenters requested specific criteria on how this requirement would be met and if filler materials were needed. Two commenters suggested deleting the requirement since economics would drive waste generators to package the maximum volume of waste into a container.

Because wastes can be of a very variable nature it is not possible or even desirable to include specific criteria for minimizing voids in a regulation. Where it is possible NRC suggests filling voids with other waste materials. The NRC staff will work with waste generators to suggest acceptable means for eliminating voids. The NRC staff agrees that economics should force waste generators to maximize the amounts of wastes packaged in a container, but also believes that the requirement should remain as an incentive for those waste generators who have yet to establish effective volume reduction programs to minimize volumes of waste shipped for disposal.

Rule Changes: None

#### **ISSUE D-56-11**

Issue:

 Compressive load requirement Commenters: Alabama Power - 33 Union Carbide - 39 Stock Equipment - 67. General Electric - 89 Power Authority of the State of New York - 92 Atomic Industrial Forum - 100 U.S. Ecology - 101 Carolina Power and Light - 106

American Society of Mechanical Engineers - 107 North Carolina Radiation Protection Commission - 109 U.S. Department of Energy - 119

#### Rule Citation: 61.56(b)(1)

Analysis of Comments: The 50 psi compressive load requirement is intended to assure that wastes requiring stability will not be crushed when subjected to the load of waste packages or overburden placed above them. The 50 psi value

is based on conservatively assuming maximum burial depths at the Hanford, WA site (up to 45 feet) and waste or overburden density of 150 lb/ft<sup>3</sup>. Testing performed on acceptable solidified waste specimens indicate that the 50 psi compressive strength should be easily obtained.

ា ស្ដេះស្ដ្រា។

Three commenters indicated that the 50 psi requirement was very rigorous and suggested deleting the requirement. Two of these commenters suggested editorial changes to incorporate their concerns.

Six commenters including Union Carbide and DOE suggested that the 50 psi load requirement be deleted and replaced by a general requirement to reflect actual disposal site conditions and operations.

One commenter suggested restudying this requirement on a cost/benefit basis to determine if it was necessary.

One commenter stated that some waste forms and containers would be unable to meet the 50 psi criterion. Union Carbide stated that few, if any, existing waste containers can meet the stability requirements in the rule.

The NRC staff has deleted the prescriptive requirement in 10 CFR 61 for a minimum compressive strength of 50 psi for Class B and C wastes. This 50 psi specification, however, will remain in the Branch Technical Position on waste form for solidified products as suggested guidance. NRC staff believes that the 50 psi compressive strength should be easily obtained for cement, asphalt, and vinyl ester styrene solidified products.

Section 61.56(b)(1) is now consistent with the recommended revision proposed by the two commenters cited earlier.

Rule Changes: Delete prescriptive 50 psi limit from § 61.56(b)(1). 

# ISSUE D-56-12

Issue:

Commenters:

..

High Integrity Container (HIC) Dow Chemical (17) Arkansas Power and Light Company (52) Stock Equipment Company (67) Argonne National Laboratory (68) Werner and Pfleidonor Company (68) Union Carbide (39) Werner and Pfleiderer Corporation (82) Northeast Utilities (85) State of California (93) State of California (93) Health Physics Society (96)

Rule Citation: § 61.56(b), (1) Disposal container

Summary of Comments: Several commenters asked for more clear design criteria on HIC.

One laboratory (68) commented that NRC's definition in the EIS is only a subjective term. The commenter also asked if NRC will provide a quantitative definition of "HIC" or will this be left to State authorities or the private sector.

One Commenter (82) requested the Commission to reexamine the design criteria for HIC for highly dispersible forms.

The State of California (93) suggested that HIC's should be used for both high and low concentration wastes to prevent the release of radionuclides into ground water.

One comment from a Utility (85) stated the need to clarify whether the HIC alone will meet the stability requirements for Class C wastes (i.e., 500 year stability requirement).

One industrial firm (67) stated that the HIC is subject to differing guidance as to whether it should provide 150 or 300 year service and what the transportation application might require or whether the life requirement can be credibly proven.

Dow Chemical (17) stated that Section 61.56 fails to reflect the concepts of ALARA and best available technology (including whether the HIC has been tested to show that they can be expected to provide stability in the disposal environment for at least 150 years).

The Health Physics Society noted that the implied requirement for this new type of container could impact larger entities such as power plants and radio-pharmaceutical companies.

In Section 61.56(b)(1), it states that "structural stability can be provided by... placing the waste in a disposal container or structure that provides stability after disposal." Arkansas Power and Light Company correctly interpreted this to mean that the use of High Integrity Containers would be an acceptable method of providing structural stability. Union Carbide expressed the view that few if any, existing containers can meet the requirement and the option to rely on long term care should be retained.

<u>Analysis of Comments</u>: NRC staff is preparing a Branch Technical Position (BTP) on waste form. This BTP provides HIC design criteria and is the basis for how the staff would evaluate the acceptability of HIC designs.

The HIC design goal is to provide integrity over the 300 year lifetime. It will withstand various conditions such as corrosion, compaction load, thermal load, radiation, biodegradation, etc. for 300 years. If a proposed HIC container meets all the requirements of HIC design goals, it would provide an acceptable method for demonstrating structural stability. The 300 year design goal is now consistent with 10 CFR 61 scenarios. See Issue D-56-8.

Argonne National Laboratory's concern about the definition of a HIC is clarified by this design criteria in the BTP.

The State of California suggested that HIC should be used for all low waste. Basically, a HIC is designed for Class B and C wastes which require stability. Less hazardous Class A wastes are not required to be disposed of in HIC, but could be to help provide greater overall site stability.

The NRC staff believes that the use of HIC's is consistent with the concept of ALARA and the use of the best available technology. Occupational exposures in

using HIC's are expected to be similar to or less than waste solidification either with mobile or installed systems. For those utilities without the capability to effectively solidify resins, the use of the HIC could be economically advantageous since expensive equipment retrofits would be necessary to provide resin solidification capability.

While there has been no specific testing of HIC's for a 150 year period, there is substantial test data for HIC materials regarding chemical resistance, biodegradability, radiation resistance, and mechanical properties. The NRC staff belives that these data can be conservatively extrapolated to the 300 year design goal to provide assurance that the HIC will properly function over the long-term. Likewise, the NRC believes that test data can also be extrapolated to assure that solidified products can remain structurally stable over the long-term.

The use of HIC's is also consistent with the requirements for transportation provided in 10 CFR 71 and 49 CRF 171-179. HIC's currently approved for use by the State of South Carolina for shipments to the Barnwell site are fabricated from high-density, high-molecular weight polyethylene. Other containers fabricated of high-density, high-molecular weight polyethylene are commonly in use for transporting hazardous chemicals. These polyethylene containers meet the applicable transportation requirements for transporting hazardous chemicals. Likewise, the use of polyethylene containers is not in violation of transportation regulations for radioactive materials.

The NRC staff and the State of South Carolina are evaluating the use of HIC's for highly despersive waste forms such as incinerator ash. This evaluation is addressing the impacts of HIC handling accidents on worker safety. Solidification of highly despersive wastes will be required where occupational exposures from such accidents are unacceptable.

The stability requirements for Class B and C wastes are identical. Therefore, it is unnecessary for HIC's to be designed to remain completely intact for 500 years. The 300 year design goal is discussed in more detail under Issue D-56-8.

#### Rule Change: None

-

#### **ISSUE D-56-13**

<u>Issue</u> :	Free standing liquids
<u>Commenters</u> :	Dow Chemical Company - 17
	Sargent and Lundy - 24
	Stock Equipment - 67
	State of South Carolina - 79
•	American Nuclear Society - 87
	American Society of Mechanical Egineers - 107
	New England Nuclear - 110

**Rule Citation:** 61.56(b)(2)

The proposed rule stated that liquid wastes or wastes containing liquids must be converted into a form which contains as little free standing non-corrosive liquid as is reasonably achievable, but in no case more than 1 percent of the waste volume. Stock equipment commented that all classes of wastes should exclude free liquids. The State of South Carolina recommended implementing the free liquid requirements in their license conditions that is, 0.5 percent non-corrosive liquids in regular waste containers and 1 percent in high integrity containers.

Section 61.56 has been modified to incorporate as a minimum requirement that solidified wastes must have less than 0.5 percent of the waste volume as free liquids. This allowable free liquid volume is to account for potential condensation of water vapor sealed in the containers. The NRC staff believes that this requirement can be easily met using the solidification agents currently in use.

For dewatered products, such as resins, it is very difficult to assure that such products would meet a 0.5 percent free liquid requirement following transport to a burial site. Therefore, for dewatered products 1 percent free liquids will be allowed to account for settling during the transport period.

A requirement to dewater to less than 1 percent free liquids would only increase worker exposures without providing commensurate assurance that the more restrictive free liquid requirement is being met.

Sargent and Lundy, ANS, and ASME suggested deleting the requirement for noncorrosive free liquids or explaining whether the intent was to minimize either non-corrosive or radioactive liquids. New England Nuclear suggested defining non-corrosive liquids using the same terms as used in the DEIS.

The intent of the non-corrosive liquid requirement was to minimize both corrosive and radioactive liquids. Past experience has shown that corrosive liquids can result in container failure. These container failures result in higher worker exposures from repackaging wastes and cleaning spills. The waste form Branch Technical Position defines non-corrosive liquids as liquids having a pH between 4 and 11. Liquids outside of this pH range are corrosive to carbon steel, the most common material used in waste containers.

Sargent and Lundy, ANS, and ASME asked if there was an intent to limit "clean" liquids in the wastes.

The free liquid requirement applies to all liquids, corrosive and non-corrosive, radioactive and non-radioactive. This requirement follows from our intent to minimize the quantity of liquids which are in or percolate through the burial trench.

Rule Changes:

·. •. ..

1. Modify § 61.56(b)(2) to apply the 1% criterion to wastes in a disposal container designed to assure stability and 0.5% for waste processed to a stable form.

#### ISSUE D-56-14

<u>Issue:</u> Consistency with other NRC regulations and guides

<u>commenters</u>: Stock Equipment - 67 American Society of Mechanical Engineers - 113

#### Rule Citation: § 61.56

Both Stock Equipment and ASME commented on the apparent inconsistencies between the proposed 10 CFR 61 and the Effluent Treatment Systems Branch (ETSB) Technical Position 11-3.

The November 1975 version of ETSB 11-3 specified that all wet wastes from Nuclear Power Plants should be solidified. At this time the concept of using a high integrity container for dewatered products had not been developed. ETSB 11-3 was revised in July 1981. This document still requires solidification of evaporator and reverse osmosis (RO) concentrates, but allows the option of either dewatering or solidifying resins and filter sludges. In addition, ETSB 11-3 states that specific burial site requirements may dictate the type of container to be used for these dewatered products.

The revisions to ETSB 11-3 are consistent with the proposed 10 CFR 61 in that: (1) evaporator bottoms and RO concentrates would require solidification in order to meet the free liquid requirements and (2) resins and sludges can be either solidified or dewatered dependent on the type of container used.

Stock Equipment also questioned the consistency with 10 CFR 50, Appendix I and 10 CFR 61. Appendix I to 10 CFR 50 addresses the design objectives and conditions for implementing the ALARA criterion for power plant effluents. Again, the NRC staff believes that Appendix I and 10 CFR 61 are consistent. The waste form requirements of 10 CFR 61 can be met by the use of processing equipment which will maintain effluent releases ALARA. As an example, the high integrity container concept is expected to meet or exceed the Appendix I ALARA objectives at the power plant. In addition, the use of the high integrity container is expected to meet or exceed similar objectives for the disposal site.

Stock also stated that 10 CFR 61 should also be consistent with NUREG-0782 (The Draft Environmental Impact Statement (DEIS) in support of 10 CFR 61), the waste form Branch Technical Position (BTP), 10 CFR 20.311, NRC Guidance for storage facilities, and leach testing standards.

First, the NRC staff beieves that the proposed 10 CFR 61 is consistent with the DEIS. Second, the waste form BTP is being revised to provide acceptable methods and criteria for meeting the waste form requirements in 10 CFR 61. Third, the NRC staff believes that the storage guidance is consistent with 10 CFR 61 but also recognizes that possibly more restrictive packaging measures may be needed to assure the long-term integrity of containers in a storage environment over those measures needed for disposal. Last, leach tests standards and criteria will be incorporated into the waste form BTP.

Rule Change:	None	· , , ,			1 _ ·
	<u>ISSUE D-56-15</u>	ĩ,	-	, <sup>1</sup> , <b>*</b> − −	•
<u>Issue</u> :	Radiolytic decomposition of waste	form		•	· · · ·
<u>Commenter</u> :	Los Alamos National Laboratory/ Safeguard Group (43)		4	•	-, '

#### Rule Citation: 61.2

The commenter indicated that the Class C intruder waste class would include some high specific activity wastes generated in the Three Mile Island Unit 2 (TMI-2) cleanup. The commenter had reservations that such wastes should be allowed for near-surface disposal.

The waste classification system is based on the pathway studies discussed in the DEIS which supports 10 CFR 61 (NUREG-0782). The allowable nuclide concentrations are based on those pathway studies.

Some of the high specific activity wastes generated in the TMI-2 cleanup are unacceptable for near surface disposal because their activities exceed the Class C limits or because of radiolytic decomposition of the waste materials. The Department of Energy has agreed to accept these problem wastes for research and development and disposal purposes.

Based on the NRC staff experience with TMI-2 wastes, wastes which undergo substantial radiolytic decomposition should be prohibited from disposal at near-surface facilities. A Branch Technical position on waste form is being developed which includes provisions to test waste forms to ensure that stability will be maintained in a radiation environment.

Rule Change: None

#### **ISSUE D-56-16**

Issue: Establishment of technical criteria and standards

<u>Commenters</u>: Dow Chemical - 17, 83 Stock Equipment - 67 Health Physics Society - 96 The American Society of Mechanical Engineers - 113

#### Rule Citation: 61.56

<u>Summary of Comments</u>: Two commenters stated the need for technical standards and criteria for waste form and requested that these standards and criteria be prepared immediately. The ASME suggested that more comprehensive standards be added to Part 61 to address factor such as leachability. One of the commenters also requested that enforcement programs must also be established to ensure compliance.

The Health Physics Society suggested that it would be useful to have a separate rulemaking action on the waste stability requirements.

<u>Analysis of Comments</u>: The NRC staff agrees that there is a need for additional technical standards, criteria and guidance on waste form. In order to meet this need, the NRC staff is preparing a Branch Technical Position (BTP) on Waste Form which would provide acceptable test methods and criteria to demonstrate waste stability.

The NRC staff, however, is not contemplating a separate rulemaking action in this area and cannot justify additional standards fo Part 61. Since there are

many different waste forms which require disposal, the NRC staff believes that a BTP is the best approach to providing guidance while still allowing flexibility to deal with unique or differing waste forms.

Rule Change: None

.....

# ISSUE D-56-17

Enforceability of and compliance with restrictions on waste Issue: form

commenters:	ACRS (10)
	State of California (93) Health Physics Society (96)

# Rule Citation: § 61.56

Summary of Comments: The ACRS expressed general concern about the enforceability and implications of all the requirements in § 61.56. The State of California expressed concern about the lack of a requirement to confirm compliance with the waste form provisions. The HP Society comments reflected uncertainty about generator and disposal facility operator roles and responsibilities for waste form and packaging. . 16-5 12 Las

•

Analysis of Comments: The minimum requirements for waste characteristics <sup>3</sup> outlined in 61.56(a) are a mixture of prescriptive requirements and more general objectives. The requirement banning cardboard packaging for example is a straightforward requirement that requires no interpretation and can readily be confirmed. Other requirements such as reducing the hazard of biological material to the extent practicable is more of an objective or statement of policy."

The primary reliance for assuring waste characteristics is the generator's quality control program required by proposed § 20.311. The program is inspectable at the waste generator's facility and includes a certification that wastes meet the requirements for acceptance at the disposal facility. Disposal facility operators will also conduct package inspections to monitor shippers.

Rule Changes: None

 $\mathcal{C}$  . The second se

Issue:

· . . : .

ISSUE D-56-18 Proposed container design

Robert Reynek - 2 Commenter:

Rule Citation: NA

Mr. Reynek proposed a design for a waste container and has suggested testing it. While this comment does not directly apply to the proposed 10 CFR 61 regulation, the proposed design could have merit if further developed. The NRC, however, has no legislative responsibility for development of the nuclear industry. The Atomic Energy Act has given this function to the Department of Energy. ---- .

#### **ISSUE D-56-19**

Issue: Polymetric Solidification Agents

Commenter: Oswald U. Anders - 73

Rule Citation: 61.56

<u>Summary of Comment</u>: Anders recommended requiring solidification of all wastes in plastic polymers and banning urea formaldehyde and absorbents.

<u>Analysis of Comments</u>: Anders points out that virtually all waste can be immobilized in plastic matrices. Aqueous waste can be directly incorporated. Granular solids can also be directly incorporated. He recommends that all wastes be so solidified to reduce leaching and deter intruders.

Adaption of Dr. Anders' recommendation would have the most impact on Class A waste shippers. The stability requirements of 61.56(b) would accomplish his goals for Class B and C wastes. Thousands of small users generate Class A wastes and typically will not have treatment or solidification facilities.

Our pathway evaluations indicate that the use of polymetric binders are unnecessary for Class A wastes. Likewise, absorbents are also adequate for Class A wastes. For Class B and C wastes the NRC staff believes that cement as well as asphalt and other polymetric materials are capable of meeting the stability requirements in 10 CFR 61. Therefore, in order to allow waste generators the flexibility to select the optimum solidification system for their needs, the NRC staff has not specified the use of individual solidification agents.

Urea formaldehyde is currently no longer used as a solidification agent and is no longer acceptable for disposal at the three commercial disposal sites.

Rule Change: None

: •

**ISSUE D-56-20** 

Issue: Solidification of Ion-Exchange Media and Wet Solids

<u>Commenters</u>: Catherine Quigg - 13 Dow Chemical - 17 Stock Equipment - 67 Argonne National Laboratory - 120

### Rule Citation: 61.56(b)(1)

<u>Summary of Comments</u>: Dow Chemical stated that the disposal of ion exchange media by dewatering is not within the concepts of ALARA and the use of the best available technology. This comment would apply to Class A ion exchange media and Class B and C media which are disposed of using high integrity containers. Stock Equipment also supported this concern.

Catherine Quigg suggested that resins should be solidified.

In addition, Mrs. Quigg suggested that ion-exchange resins should not be allowed to be disposed of as low-level wastes since resins can contain high specific activities and contain nuclides like Cs-137 which have half-lives of 30 years. Mrs. Quigg also suggested incinerating or vitrifying ion exchange media. Note that Argonne National Laboratory (120) clarified a quote Mrs. Quigg attributed to the commenter.

11

1

<u>Analysis of Comments</u>: The pathway evaluations include in the source term resins which have been disposed of in a dewatered form. These evalutions consider the dispersive nature and degradation of ion exchange resins. The results indicate that the disposal of ion exchange resins in a solidified or dewatered form is acceptable within the requirements of the waste classification system. Of course, highly loaded organic resins which are subject to substantial radiolytic degradation would be unacceptable for disposal. The resin degradation problem is being addressed in the NRC staff's Branch Technical Position on Waste Form.

The NRC staff believes that for Class B and C wastes, which require stability, that the high integrity container is an acceptable option for providing stability. Since the high integrity container filling operations at the waste generator's site would generally be less complex than solidification, it would be expected that overall occupational exposures (including maintenance of equipment) would be less than or the same as those solidification operations. This statement would generally apply to solidification using either mobile or installed solidification systems. Therefore, NRC staff believes that the ALARA concept is fulfilled by the use of high integrity containers. Additional discussions on high integrity containers are provided in Issue D-56-12.

The waste classification system in 10 CFR 61 determines the allowable concentrations for Cs-137 and for other long-lived nuclides important for disposal. These levels have been determined based on the pathway scenarios discussed in the DEIS which supports 10 CFR 61, NUREG-0782. These pathways include as a source term ion exchange resins.

The incineration of organic resins and the vitrification of inorganic zeolites are options which could be used to process ion exchange media. The incinerations of organic resins is being studied by the DOE and the commercial firms which market incinerators. The DOE is also studying the vitrification of zeolite ion exchange media both as a method for processing DOE generated zeolites and for the zeolite wastes generated in the submerged demineralizer system at Three Mile Island.

Rule Change: None

•

÷

٠ŗ:

#### **ISSUE D-56-21**

Issue: Characteristics of volume reduced wastes

Commenter: Advisory Committee on Reactor Safeguards (10)

Rule Citation: General

<u>Summary of Comment:</u> ACRS pointed out that in the proposed rule, attention should be directed to techniques both for reducing the volumes of wastes

generated and for assuring that the waste that are produced are in, or can be converted to, a form amenable to safe disposal.

<u>Analysis of Comment:</u> NRC is developing a comprehensive data base regarding the acceptability for disposal of waste products from volume reduction techniques. Several research and technical assistance projects are being performed to determine the mechanical properties, leach resistance, radiation stability, and biodegradability of products from current radwaste systems and from advanced volume reduction techniques for low-level waste.

#### ISSUE D-57-1

<u>Commenters</u> : Los Alamos National Laboratory (43) Arizona State Clearinghouse (47) American Nuclear Society (87) Arkansas Power and Light Company (94) Health Physics Society (96) The American Society of Mechanical Engineers Department of Energy (119)	5 (107)

Rule Citation: §61.57

<u>Summary of Comments</u>: The commenters suggested using color coding, different wording to indicate class of wastes, consistency with DOT labeling, minimum standards, and clarification of responsibilities. One commentor questioned whether standard labels will be required. One commenter supported the need and one made a general suggestion that more detail be included.

<u>Analysis of Comments</u>: Los Alamos National Laboratory suggested using color coded labels in § 61.57. The staff considered using color coding for indicating Class A, Class B, or Class C wastes to provide easier identification by operating personnel. The drawbacks are the existing color code labeling requirements under DOT regulations and NRC regulations in 10 CFR Part 20 and the large number of potential generators and service and supply companies potentailly affected. The staff did not want to compromise DOT color codes and the dependence on them in transit. Site operators may develop site specific administrative requirements on package design, weight, labeling, etc. to facilitate handling and emplacement eventhough the rule doesn't have a specific requirement. Thus color codes may be used but are not required. The Los Alamos suggestion was not adopted.

The Arizona comment addressed the need for and potential confusion from the use of "segregated," "stable" and "intruder" in § 61.57 and suggested use of the general term "waste" instead. The suggestion was adopted.

Arkansas Power and Light Company made a general comment that "Differences between DOT and NRC regulations regarding to labeling need to be resolved prior to implementation to avoid confusion." No specific problems were identified by the commenter. The DOT and NRC labeling requirements are related to different activities: handling and disposal. The DOT rules treat all radioactive materials the same and do not distinguish between materials to be used and waste. Differences in labeling for the subset waste minimize the size of the affected population and minimize the number of affected radioactive shipments. The distinctive labeling is needed for ease of proper emplacement at the sites and staff believed it should be retained and apply only to shipments to disposal sites. No change to the requirements in the rule was adopted. The DOE also recommended compatibility with DOT requirements. The Health Physics Society questioned whether NRC plans to require standard labels, warning signs, or other markings to supplement or replace current DOT labels. The commenter agreed that clear classification marking would help site operators. As noted earlier, no standard labels are intended and the labeling is supplementary to DOT. The DOT labels would still be the prime indicator for handling in transit or storage. The classification labeling would not come into play until emplacement. The classification labeling is not necessarily related to occupational hazard. For example large shipments of beta emitting istopes could be Class C but not a significant external exposure hazard.

The American Society of Mechanical Engineers suggested that § 61.57 "labeling requirements should be expanded, clarified, and made more specific." The class labeling is to provide additional assurances. The Class of each package must also be shown in the manifest accompanying the shipment. Flexibility to use printed labels or stencils, choice of colors, etc. to label the package with the words "Class A waste" etc. seems the minimum burden and commensurate with the role played only in emplacement for disposal.

The Department of Energy suggested clarifying responsibility for sorting and labeling wastes and correctly noted that the shipper is responsible. The roles of the generator, collector, processor, and facility operator are complex and are addressed individually in § 20.311 of 10 CFR Part 20 and the discussion of issues related to this section. A cross reference to labeling requirements in § 61.57 was added to § 20.311.

#### Rule Changes:

1. Amend § 61.57 to read:

Each package of waste must be clearly labeled to identify waste class in accordance with § 61.55 and bear the words Class A waste, Class B waste, or Class C waste.

 Amend § 20.311(d)(2), (f)(4) to read: Label each package of waste to identify whether it is Class A, B, or C in accordance with §§ 61.55 and 61.57 of Part 61 of this chapter.

### ISSUE D-59-1

Issue: State responsibility after license transfer

<u>Commenters</u>: Pennsylvania Department of Environmental Resources (16) Ohio EPA (38) Ontario Hydro (51)

<u>Rule Citation</u>: Supplementary Information; also related to § 61.30, 61.31, 61.62 and 61.63.

<u>Summary of Comments</u>: Pennsylvania recommended the option be available to States to turn ownership and responsibility for long-term custody over to the Federal government. The Ohio EPA identified the lack of a clear statement of State responsibility after license transfer. Ontario Hydro questioned state liability for health problems arising from the site after the active institutional control.

<u>Analysis of Comments</u>: The Pennsylvania recommendation that States have the option to turn over responsibility for long-term custody of low-level waste sites to the Federal government if they meet satisfactory criteria reflects the approach in place for mill tailings sites. After stabilization, mill tailings sites are turned from the private sector to the State, if the State wishes, or to the Federal government if the States declines. This process for mill tailings is provided for the Uranium Mill Tailings Radiation Control Act of 1978. No such authority exists for low-level wastes so no change to the rule was made to provide this option.

The Ohio EPA and Ontario Hydro concern is directed at the land owner State responsibility beginning with transfer of the license under § 61.30. Under the licensing scheme in the proposed rule, the State would be responsible for the site and disposed waste. Institutional care would be carried out under license for some prescribed period, up to 100 years. If a commercial or private sector developer and operational licensee was involved, the developer/operator's responsibilities would normally end at transfer. Thus, the state becomes the responsible party upon receipt of the transferred license. Although it is impossible to have absolute guarantees against problems arising after site closure, the requirements of Part 61 are aimed at minimizing the potential for problems to the maximum extent practicable. It behooves the state, as land owner, to maintain awareness of the operations and conditions at the facility, either through some independent oversight as landlord, or by participating with NRC in the review of the initial application as provided in Subpart F of Part 61. If the State was developer, operator, and land owner, the State would be responsible for the site and wastes at all times. The issue of liabilities is further addressed under issue E-1 on financial assurances.

Rule Change:	None				u cela Posti D
	••		- 1944年月1日(1月1日) - 「1951年日日日)		<ul> <li>(1)</li> <li>(2)</li> </ul>
	· ·		and the second second	•	
	· ·	· • •	- Provent Property	· · · ·	

#### ISSUE D-59-2

<u>Issue</u> :	100-year limit on the institutional control period; permitted activities during the control period
<u>Commenters</u> :	Marvin Lewis (3) Environmental Law Project (9) Pennsylvania, Department of Environmental Resources (16) Commonwealth Edison (35) Bechtel National, Inc. (44) Duke Power Company (48) Argonne National Laboratory (68) Oswald U. Anders (73) Birmingham Audubon Society (80) Utility Nuclear Waste Management Group (81) Northeast Utilities (85) American Nuclear Society (87) Amy Hubbard (90) Atomic Industrial Forum (100) Conference of Radiation Control Program Directors (103) The American Society of Mechanical Engineers (107), (113) New England Nuclear Corporation (110) U.S. Department of Energy (119) U.S. Environmental Protection Agency (122)
Rule Citation:	§ 61.7 Concepts and § 61.59 Institutional Requirements

<u>Summary of Comments</u>: The Conference of Radiation Control Program Directors, Inc. concurred with and supported defining a period of time for institutional control and relating the classification and disposal of waste to this time frame. The Environmental Protection Agency supported use of a 100-year time frame for institutional controls. Marvin Lewis inquired as to the minimum time that institutional control could be a surety. The Environmental Law Project philosophically questioned the continued production of wastes that will remain hazardous

longer than the assumed period of government control.

1.1

The Birmingham Audubon Society commented similarly, stating that if the 100-year limit was retained, waste remaining dangerous for longer periods should be disposed of elsewhere, where controls can be maintained for as long as the waste remains dangerous. Amy Hubbard offered a similar comment stating that radioactive wastes remain dangerous for centuries and their safe disposal is dependent upon a stable social and financial structure for at least 500 years which cannot be predicted, much less ensured. Commonwealth Edison expressed the view that 100 years was too long and difficult for a licensee to meet. Bechtel National commented that if a government institution would be available to maintain land ownership and records as discussed in the statement of consideration, then that same institution could maintain a fence for more than 100 years. This would reduce concern about the potential exposure to intruders. The same comment was made by the American Nuclear Society.

The Utility Nuclear Waste Management Group recommended that § 61.59 make it clear that during the period of institutional control the land could be used for other purposes not inconsistent with public health and safety and which would not disturb the integrity of the site. The Atomic Industrial Forum, American Society of Mechanical Engineers and Duke Power commented similarily noting that limited use of the land may be desirable in the future and the government land owner should have flexibility to institute suitable control options and site access depending on the particular conditions existing at the site. Oswald Anders also commented on that issue noting that a proper use of inactive disposal site could be a golf course. The New England Nuclear Corporation also suggested changing the words "keep people off the site" to "control access to the site" to allow maintenance, surveillance and other appropriate activities. Pennsylvania recommended more specificity on allowable site uses.

Northeast Utilities commented that the institutional control period shall be extended for as long as the governing body exists. Doing so would extend the surveillance period and protect against site intrusion until the governing body determines the site could be reopened to the public. The Atomic Industrial Forum commented that consensus of opinion indicate that the institutional control period may reasonably range from 100 to 300 years and 300 years should be selected in the equations NRC used to determine prescriptive requirements. New England Nuclear also offered the same comment. The U.S. Department of Energy stated that the duration of institutional control significantly affects the cost of waste disposal. Raising the period from 100 to 300 years would increase the Class A limit for several radionuclides. They suggested that the NRC reexamine the basis for limiting institutional controls to 100 years.

<u>Analysis of Comments</u>: Four main points are raised in the comments and each of the points is discussed further below:

· . . '

; :

1. Everyone expressed support in one way or another for defining a time frame for institutional control related either to the hazard duration of the waste, cost or assurance of continued government stability.

2. The disposal of waste remaining potentially hazardous after the end of the assumed institutional control period should be disposed of by other methods providing greater controls.

3. Although one commenter strongly supported the 100-year time frame, about thalf suggested raising the limit; most suggested from 100 to 300 years.

4. Part 61 should allow the government landowner flexibility in controlling site access during the institutional control period to ensure all activities are properly carried out and to allow for productive uses of the land which would not affect site integrity.

<u>1 and 2</u>: The approach NRC has followed in defining requirements for safe disposal of LLW is to establish controls for each of the principal components of a "disposal system" - the waste form and package, site characteristics, facility design and engineering, and institutional controls. The comments supported continued use of institutional controls and defining a finite time frame for assumed reliance on such controls. Complete reliance is not placed on any one component part (e.g., waste form) but each acts with the others to collectively ensure safe disposal over the long term. Thus, Part 61 does not assume total and complete reliance on institutional controls to prevent disturbance of the waste. Rather Part 61 assumes reliance on active institutional controls for a limited time frame (100 years) after which the waste form, site characteristics, facility design and operation and "passive" institutional controls, such as records and continued government land ownership collectively continue to provide
the necessary controls. The classification system proposed in Part 61 requires that wastes which will still present a significant potential hazard if disturbed at the end of this 100 year institutional control period must be placed into a stable form or container such that they will be recognizable as waste and less dispersable. Part 61 also requires that wastes which continue to present significant potential hazard at the end of 100 years must be disposed of with further additional controls (e.g., deeper burial) which will further limit the potential of their being disturbed. Thus, Part 61 does provide greater controls for disposal of waste which would continue to present a potential hazard after the end of the assumed 100-year institutional control period.

3: A remaining and basic question is how long reliance should be assumed for active institutional controls. Based on work performed by EPA; public comments on a preliminary draft of Part 61 and an advance notice of proposed rulemaking; and 4 regional workshops, a consensus of public opinion was developed which supported use of a time period of 100 years. Based on the analyses in the EIS, NRC found no overly compelling reason to select one particular institutional control period over another and the limit of 100 years for assumed reliance was based primarily on public opinion. Use of a longer institutional control period would allow higher concentrations of some radionuclides to be disposed of as Class A waste. Using a longer period would, however, increase the costs for long term surveillance and monitoring of the site. In addition, the assumed length of institutional controls and the dose limit used to control exposures to an inadvertent intruder are directly related. Thus, increasing the length of the institutional control period has to be examined in the context of comments that the dose limit established for protection of the inadvertent intruder should also be raised. (See comments on § 61.42.) Based on consideration and balancing of these two aspects NRC has decided not to increase the assumed period of 100 years upon which reliance can be placed on institutional controls. Rather, NRC has reevaluated the calculations that establish the waste classification concentration limits to eliminate unnecessarily conservative assumptions with the result that the analysis is more realistic and the limits for several important nuclides have been raised. With this action, the NRC believes that most of the concerns of those who encouraged higher exposure limits, less emphasis on protection of intruders and use of a longer institutional control period will have been met. (See comments on § 61.42 under Issue C-4 for further details and discussion.)

<u>4</u>: As stated in the EIS, NRC intended and would allow flexibility in control of site access including productive uses of the land provided it did not result in disturbance of the waste or affect long term site performance.

<u>Rule Change</u>: Based on the staff's analysis of comments, the institutional requirements in § 61.59 have not been changed. Section 61.7(b)(4), "Concepts," however, has been modified to allow flexibility in controlling site access including productive uses of the land during the active institutional control period. It reads as follows:

(4) Institutional control of access to the site is required for up to 100 years. This permits the disposal of Class A and Class B waste without special provisions for intrusion protection, since these classes of waste contain types and quantities of radioisotopes that will decay during the 100-year period and will present an acceptable hazard to an intruder. The government landowner administering the active institutional control program has flexibility in controlling site access which may include allowing productive uses of the land provided the integrity and long-term performance of the site are not affected.

-----

1 - 1 **:** - - - -

# ISSUE D-59-3

Issue:

Land ownership-miscellaneous

<u>Commenter</u>: Ontario Hydro (51) Union Oil Company of California (66) Georgia Institute of Technology (70) Louise Gorenflo (71) Birmingham Audubon Society (80) U.S. Department of Energy (119)

Rule <u>Citation</u>: V D of the Supplementary Information, § 61.59

<u>Summary of Comments</u>: Two commenters expressed support of the assurances provided by government ownership and two questioned the need. One questioned NRC authority to regulate federal ownership. One was concerned about the applicant's rights and others questioned whether tribal ownership is permitted.

<u>Analysis of Comments</u>: Ontario Hydro acknowledged that government ownership of land for disposal sites is "certainly an easy way of guaranteeing proper safe use of the land." The commenter also questioned whether NRC could regulate a site if the Federal government owns the land based on the commenter's understanding that one department of the Federal government could not regulate another. The NRC does have the authority to regulate most Federal agencies and routinely issues licenses to VA hospitals, the Departments of the Army, Navy, and Air Force, the EPA, etc. The NRC does not have the authority to regulate Department of Energy (DOE) low-level waste activities. If the DOE will be the custodial agency and provide institutional control, no NRC license would be involved for the institutional control period. Institutional control by other agencies would be licensable. A private sector developer/operator of a site located on land administered by DOE would be licensable. The U.S. Ecology site at Hanford, Washington illustrates. The land is leased by the Federal government to the State and subleased to the operator. The operator is licensed by the State (Washington is an Agreement State) and the NRC.

Gorenflo suggested that quasi-public corporations develop all new sites and assume long term care responsibilities instead of burdening resource limited states. As discussed under Issue E-1, Part 61 provides financial assurances and upfront financial planning to alleviate the resource burden of institutional control. The staff views State or Federal ownership to be a necessary safeguard. The State could fulfill its responsibilities in many ways including a State authority or quasi-public corporation, but the responsibility should be with the more certain established government.

The Birmingham Audubon Society indicated strong support for State or Federal ownership. The staff agrees that this practice should be continued.

The Union Oil Company of California suggested adding provisions to § 61.59 that ownership revert to the applicant if the site is not used for disposal. The Company cited several examples of occurrences that might prompt a change in plans such as adverse public opinion or economics. The Company also felt that the applicant should have the right of first refusal when a site used for disposal is determined safe for other uses. These issues were not addressed in the rule primarily because of the variety of circumstances and roles that may exist in the development of new sites. New sites may be proposed on land already owned by the State or Federal government. These issues may be part of the terms negotiated with the governments. The land value may be one incentive for accepting the institutional committment and responsibility. The rule as proposed does not preclude the return to the applicant. The staff did not adopt the suggestion in order to keep the proposed level of flexibility.

Rule Change: None

## ISSUE E-1

<u>Issue</u> :	Subpart E - Financial Assurances		•	$\sum_{i=1}^{n} (i \in \mathcal{L}_{i})^{n}$
Commenters:	Marvin Lewis (3)			A
<u> </u>	Environmental Law Project (9)	•	· :	
	The Surety Association of America	(20)		
	Joseph H. White (21)			
	Commonwealth Edison (35)			•
n an	Ohio EPA (38)	1		
1	Chem-Nuclear Systems, Inc. (41)	•		
	Ontario Hydro (51)	· .		· · ·
	National Association of Insurance	Brokers,	Inc.	(54)
	Union Oil Company (66)			
	Birmingham Audubon Society (80)	• ,	:	
	Utility Nuclear Waste Management (	Group (81	)	
	Northeast Utilities (85)	• • -	•	
	State of California (93)			· · .
	Don't Waste Washington Legal Defe	nse Found	ation (	(97)
	State of New York (99)	•		· ·
· .	U.S. Ecology (101)		,	
	Conference of Radiation Control P	rogram Di	rectors	s <b>(103)</b>
	Kerr-McGee, Inc. (115)			
	Tennessee Valley Authority (116)			
	U.S. Environmental Protection Age	ncy (122)		
•	General Research Corporation (123)	)		
	•		-	

Rule Citation: Subpart E - Financial Assurances

Summary of Comments: Approximately two dozen commenters responded to the proposed financial assurance requirements for closure and post-closure care. In general, the commenters expressed support for the rule's establishment of financial assurances for closure and for long term care of a LLW disposal site. Commenters mentioned that the existing history of LLW disposal sites revealed a strong need to require licensees to demonstrate evidence of financial responsibility so that the public health and safety were protected and also so that potential liabilities do not rest with state taxpayers. Several commenters felt that the financial requirements should provide more detail.

One of the major points raised by a variety of commenters including the State of New York was that the proposed regulation failed to address financial responsibility for unanticipated contingencies at a LLW disposal site. One group expressed concern that the regulations set the stage for a "tax-payer funded bailout" of poorly-run disposal sites. They felt the industry should bear these costs, and that the regulations should be written to make this explicit. Another commenter noted that the experience of the State of Kentucky with Maxey Flats emphasized the importance of making contingency funds available in the event that serious problems occur. They felt this issue should be addressed in the rulemaking. One State further noted that the rule failed to mention who would be financially responsible if problems occur at the site that cost more than were budgeted on an assumption of normal operation. These

questions covered a variety of different scenarios (i.e., Acts of God, licensee negligance etc.).

With regard to contingencies, one commenter also asked who would assume responsibility for a site and its accompanying waste when it was closed prematurely by NRC, due to rule violation.

Several commenters felt that the rule could resolve the issue of contingencies by requiring insurance coverage, or specific language that licensees would be required to indemnify well users in case of migration.

A variety of comments were received concerning the short term financial assurances required for closure and decommissioning. Several commenters supported the rule's use of a variety of different options for closure, noting that flexibility was crucial if the proposed rule was to function in a reasonable manner.

Other commenters expressed support for the rule's provision requiring that the amount of surety liability change with changes in cost estimates. One commenter also was concerned that the financial surety arrangements increase in value over time to compensate for the effects of inflation.

Commenters expressed support for the variety of alternatives allowed to demonstrate short term financial responsibility. However, several commenters mentioned that no commercial market exists to provide surety bonds of the type required in the rule.

Commenters were also divided about whether the Commission should allow selfinsurance as a financial assurance for closure. Several commenters felt that self-insurance would not satisfy the surety requirements, and they recommended that licensees should be required to place specific funds in escrow to cover costs of decontamination, closure and stabilization. Another commenter suggested that self-insurance be based on an annual submittal of financial reports, i.e., a financial test.

Commenters also expressed support for the need to have a long-term care fund established at the time a license is issued. Some commenters wanted the rule to explicitly require the licensee to set aside funds for long term care. (However, the Commission currently lacks the authority to require a licensee to establish a fund to provide for long term care of the site after the license is terminated.) With regard to this lack of authority, one person suggested that the Commission ask Congress for authority to require financial assurances for licensees for the active institutional control period.

Two commenters addressed the "Superfund" law. White questioned the intent of the statement in the preamble to the rule that some of the requirements in the superfund legislation may be duplicative. EPA addressed this issue, identified releases from Part 61 facilities not in compliance with the license as reportable to EPA and indicated that EPA and NRC should work together to minimize duplicative reporting requirements.

<u>Analysis of Comments</u>: Several commenters expressed support for the financial requirements in the rule. The Don't Waste Washington Legal Defense Foundation stated that such strong assurances are necessary to "discharge our responsibility"

to future generations, and to assure that the public will not bear costs which should be borne by the users of the facility. The State of California also supported the need for financial requirements, noting that "such strong assurances are necessary to discharge our responsibility to future generations, and to assure that the public will not bear costs which should be borne by the users of the facility." en la companya de la companya Antonio de la companya . . . . J. . × .

Chem-Nuclear, Inc. expressed support for the rule's requirement that an applicant assure funding for site closure and post-closure. They felt this should minimize the potential for operator default or abandonment.

10 et 100

1.

÷.,

:.:

Kerr-McGee disputed the Commission's authority to require financial assurances for closure and for long term care. They asserted that the only authority possessed by NRC to require financial assurances stems from the Uranium Mill Tailings Radiation Control Act of 1978. The Commission staff believes the better legal view to be that Section 161 of the Atomic Energy Act, gives the .Commission the authority to promulgate whatever regulations may be necessary and desirable to promote the common defense and security or to protect health or to minimize danger to life or property. If short term financial assurances or other forms of financial sureties are necessary and desirable to achieve the goal of safe closure of a radioactive waste burial site, then such requirements are authorized by the Atomic energy Act of 1954, as amended.

Other commenters also expressed concern that the regulations should be more detailed. The State of California noted that the financial arrangements did not appear to be sufficiently comprehensive, and an individual felt that this section of the rule lacked sufficient detail. U.S. Ecology recommended that definitive criteria be set for financial assurances in order to ascertain whether or not underwriters will accept the risk and default provisions set forth in the regulations. The Commission staff has prepared a draft Technical Position on the Funding Arrangements for Closure and for Long Term Care of a LLW, Disposal Site that provides more definitive criteria for evaluating all financial assurances, including surety bonds. The draft has been circulated for review, and the Surety trade association will be asked to provide comments.

Commonwealth Edison and other commenters also wanted more detail in the rules, noting that it contained no details with respect to the amount of financial assurance that each disposal site licensee is required to obtain., The staff considered this regulatory approach to be ill-advised for two reasons. First, the amount of funds necessary for closure and for long-term care is site specific, and will vary depending on the particular site conditions at the time of closure. Therefore, a specific dollar amount in the rule would not be applicable to all site conditions. Second, a rule with specific dollar amounts would become outdated with changes in inflation. Instead, specific cost estimates for closure will be determined in the licensee's Closure and Stabilization Plan. An experimental state of the st

One of the major points raised by a variety of commenters was that the proposed regulation failed to address financial responsibility for unanticipated contingencies at a LLW disposal site. The Environmental Law Institute expressed concern that the regulations set the stage for a "tax-payer funded bailout" of poorly-run disposal sites. They felt the industry should bear these costs, and that the regulations should be written to make sure this is done. Another commenter noted that the experience of the State of Kentucky with Maxey Flats

emphasized the importance of making contingency funds available in the event that serious problems occur. They felt this issue should be addressed in the rulemaking. The State of California further noted that the rule failed to mention who would be financially responsible if problems occur at the site that cost more than were budgeted on an assumption of normal operation. The State of California was also concerned that State taxpayers rather than waste generators would eventually pay the excessive costs of site maintenance and repair.

These questions cover such a variety of different scenarios (i.e., Acts of God, licensee negligance, etc.) that it is not possible to specifically respond to all of the potential contingencies. However, a general response to the overall issue of responsibility for contingencies at a low-level waste disposal site is possible. These comments cover two different time periods--the post-closure period, when the original license is still responsible at the site, and the institutional control period, when the license has been transferred to the landowner of the site for a period of up to one hundred years. In the case of the post-closure care period, the licensee would be responsible for all activities at the site found necessary by the Commission to protect the public health and safety. Financial responsibility for activities during the institutional control period are a matter to be worked out between the site owner (i.e., the State or Federal Government) and the licensee in their lease or other legally binding arrangement, and it is possible that if the site owner were a state, they would work out an arrangement whereby the site operator would collect a surcharge from waste generators for the institutional control period. rights and responsibilities of the state and the licensee would be determined at such a time.

With regard to contingencies, the Ohio EPA and Ontario Hydro asked who would assume responsibility for a site and its accompanying waste when it was closed prematurely by NRC, due to rule violation, or when the licensee defaults. Responsibility for a site closed prematurelay by the NRC would depend on the situation. Site closure would be a last resort of the Commission, since the agency has other authorities besides closure, such as civil penalties, to require licensee compliance. However, in the event that it would become necessary to close the site for health and safety reasons, the proposed rule provides that the licensee continues to be responsible until the license is terminated. In the event that the licensee's financial condition deteriorated so that he was unable to maintain the site to protect he health and safety, then the Commission would probably require the site owner (either the state or federal government) to assume responsibility at the site. However, regardless of who assumed responsibility of a prematurely closed site, the rules require that a licensee have available at all times during the site life, sufficient financial gurantees to ensure that sufficient funds are available for site closure and decommissioning. These funds would be available to properly maintain the site if the original licensee were unable to do so.

•••

••• `

Several commenters felt that the rule could resolve the issue of contingencies by requiring insurance coverage, or specific language that licensees would be required to indemnify well users in case of migration. The staff agrees that there is a need for licensees to provide financial responsibility for liability coverage for off-site bodily injury and property damage and thinks the public health and safety and the environment would be protected from unanticipated contingencies by such coverage, as well as assisting the State in establishing disposal sites. Four existing LLW disposal facilities currently carry this type of liability coverage, and several other State and Federal agencies, including EPA have imposed similar requirements for hazardous and radioactive waste facilities in order to protect the public health and safety and the environment. However, at the present time, the Commission's only statutory framework for establishing such a requirement is Section 170 of the Atomic Energy Act, also known as the "Price-Anderson" Act. This type of coverage is designed to cover "catastrophic events" primarily for nuclear reactor licensees, and this coverage would be in excess of the risk at a low-level waste facility. Therefore, a third party liability requirement is not established in this regulation. The Commission should strongly encourage licensees to continue to carry third party liability insurance coverage through the conventional insurance market.

The State of New York was also concerned that the post-closure maintenance, leachate collection and treatment, cover repair and other likely costs will be underestimated by applicants and accepted by both site owners and the USNRC due to the pressing need for waste disposal site availability. The staff thinks the license review process will provide a basis for all concerned parties to review the licensee's estimates of costs for closure and postclosure care. All parties would have the opportunity to express their view on the adequacy of the licensee's estimates of costs required for closure and post-closure care.

t

.

÷.

The Conference of Radiation Control Program Directors also recommended that the Commission consider guidelines on methods for determining financial needs for long-term care. General Research Corporation also felt that the contents of the agreements were unclear and that the criteria to be used in evaluating these agreements were not indicated. The Staff has prepared a draft Branch Technical Position on Funding Alternatives for Closure, Postclosure, and Long Term Care that provides guidance on various alternatives States might wish to consider in developing financial arrangements for long term care. Additionally, the draft Environmental Impact Statement (NUREG-0782) accompanying the proposed rulemaking discusses possible financial arrangements for long-term care.

Several commenters expressed opinions on the types of short term financial assurances allowed by the rule. In general, the National Association of Insurance Brokers felt that flexibility was crucial if the proposed rule was to function in a reasonable manner. The proposed rule does allow several types of short term financial assurances; additionally licensees may proposed other financial assurances for closure to the Commission for review.

Other commenters expressed support for the rule's provision that the amount of surety liability should change with changes in cost estimates. The State of California was concerned that the financial surety arrangements increase in value over time to compensate for the effects of inflation. The Don't Waste Washington Legal Defense Foundation noted that Subpart E should be amended to state more clearly that the Commission will review the applicant/licensee's financial assurances periodically, and that the burden of proving adequate financial qualification is on the applicant/licensee. As proposed, the rule would allow the Commission to periodically assess the amount of funds collected for both closure and post-closure care of the site. If necessary, the staff could require the financial assurances to be increased to account for inflation and unforeseen problems and costs. The State of California also was concerned that the short term financial arrangments were not instruments that increase in value over time to compensate for the effects of inflation. As proposed, the rules would allow the Commission to periodically assess the amount of funds collected for both closure and post-closure care of the site. If necessary, the staff could require the financial assurance to be increased to account for inflation and unforeseen problems and costs.

Northeast Utilities suggested that the "pay as you go" funding arrangements for closure should be permitted, instead of surety bonding for an entire site. However, the staff considers that this method of funding does not provide an adequate degree of coverage in the event of premature site closure.

Several commenters mentioned that no commercial market exists to provide surety bonds of the type required in the rule. In developing the rule, the Commission staff is aware that surety bonds of the type proposed in the rule may currently be unavailable. However, the staff included this alternative in the rule in the event that this type of coverage becomes available in the insurance market at a later time.

One commenter noted that insurance could be a viable short term financial assurance against premature site closure. Although not specifically mentioned, the proposed regulations do not exclude the use of insurance for providing financial assurances for closure. If this type of coverage becomes available at a later date, the staff will consider this type of funding assurance if a licensee proposes its use.

Commenters also were divided about whether the Commission should allow selfinsurance as a financial assurance for closure. The State of California and The Don't Waste Washington Legal Foundation felt that self-insurance would not satisfy the surety requirements, and the Birmingham Audubon Society also recommended that licensees should be required to place specific funds in escrow to cover costs of decontamination, closure and stabilization. However, Union Oil felt that self-insurance based on an annual submittal of financial reports, i.e., a financial test should be permitted.

Kerr-McGee also felt that prohibiting self-insurance was arbitrary and unsupported. The staff rejected the use of stand alone "self-insurance" based on the staff's lack of confidence in this method to provide adequate assurances. Further, State officials have informally expresed the need to have tangible funds available from the licensee for site closure, so the State as landowner would not be left financially responsible. While not specifically allowing its use on a generic basis in the rule, the staff will evaluate the use of financial tests proposed by licensees on a case by case basis.

Commenters also expressed support for the need to have a long-term care fund established at the time a license is issued. Commenters suggested that licensees should be required to place in escrow funds necessary to cover costs of institutional safeguards for the duration of those safeguards. One suggested that monies should be collected from a tax or fee structure imposed on the licensee that was similar to the cubic meter surcharge imposed on waste disposed of at the site.

The State of California also suggested that a sinking fund would be a preferable vehicle for funding for the institutional control period. The Tennessee

Valley Authority felt that it would be preferable to have the licensee turn over the site and any required money at the time the license is transferred to the Government. The Commission currently lacks the authority to require a licensee to establish a fund to provide for long term care of the site after the license is terminated. Instead, the Commission can only require a licensee to provide evidence of entering into a lease or other binding arrangement with the siteowner indicating that the two parties have established financial responsibility for long term care between themselves. The proposed regulations do not therefore require a licensee to establish or tie up funds for the long term care period, so the licensee is not required to turn over any required money at the time of the license transfer.

With regard to this lack of authority, the State of California suggested that the Commission ask Congress for authority to require financial assurances for licensees for the active institutional control period. The Commission has raised this issue before Congress several times; for example, see the testimony of Joseph Hendrie before the House Committee on Science and Technology, November 7, 1979; and Statement of John Ahearne, before the House Subcommittee on Energy and Power, July 25, 1980. Additionally, NRC staff provided comments on November 6, 1981 to Congressman Udall's Committee on Interior and Insular Affairs supporting the provision in H.R. 3809 that provided the Commission and the second with this type of enabling authority.

1.1.1.1.1.1

: :

The General Research Corporation outlined several areas where more prescriptive requirements and clarification should be considered. While many of the observations and comments have merit, they represent a level of detail the staff prefers to relegate to Branch Technical Positions and eventually regulatory guides.

. . :

With respect to the superfund issue, the EPA comment is self-explanatory and states:

NRC solicited comments on possible duplicative requirements for effluent releases and broker activities under the Comprehensive Environmental Response, Compensation and Liability Act of 1980 (CERCLA). This "Superfund" law exempts from notification "any release of source, special nuclear, or byproduct material... in compliance with a legally enforceable license, permit, regulation, or order issued pursuant to the Atomic Energy Act of 1954" (CERCLA Section 101(10)(K)). Radioactive releases from nuclear waste disposal facilities which are not in compliance with an NRC license, permit, regulation, or order fall within the reporting requirements of CERLA. Furthermore, as part of the notification regulations under CERCLA, EPA is planning to develop a notification scheme for releases of radioactive 1991 - S. A. materials not licensed under the Atomic Energy Act of 1954 or the Uranium Mill Tailings Radiation Control Act of 1978. EPA wishes to minimize 1 Er duplicative reporting requirements for releases reported to other agencies. EPA intends to work with NRC to minimize duplicative reporting requirements to the extent possible.

NRC staff agree that duplication should be minimized and will work with EPA to eliminate potential overlapping regulatory requirements.

#### Rule Changes:

.

- 1. Amend § 61.62(e) to read:
  - (e) The term of the surety mechanism must be open ended unless it can be demonstrated that another arrangement would provide an equivalent level of assurance. This assurance could be provided with a surety mechanism which is written for a specified period of time (e.g., five years) yet which must be automatically renewed unless the party who issues the surety notifies the Commission, the beneficiary (site owner) and the principal (the licensee) not less than 90 days prior to the renewal date of its intention not to renew. In such a situation the licensee must submit a replacement surety within 30 day after notification of cancellation. If the licensee fails to provide a replacement surety acceptable to the Commission, the Commission will collect on the original surety.

#### ISSUE E-2

Issue: Optional financial report

<u>Commenters</u>: State of New Mexico (4) U.S. Department of Energy (119)

1. . .

Rule Citation: § 51.80(g)

Comment: Delete words "if any".

<u>Analysis of Comment</u>: The commenters suggested that licensees should be required to furnish an annual financial report. Deleting the "if any" flexibility would require the licensee to generate a new report even if such reports were normally generated every two years or were delayed, for example. The optional filing was provided to minimize the burden on the licensee. Section 61.62(c) requires the Commission to review the adequacy of surety mechanism for closure funding annually. Information on financial status will be important to this review and the Commission agrees with the commenters concerns. Section 61.80(g) was modified to require a annual certified financial statement and permit the annual report to meet the requirement. All companies must evaluate finances at least annually for tax purposes so the change should not be a burden.

<u>Rule Change</u>: Amend 61.80(g) to read: Each licensee authorized to dispose of radioactive waste received from other persons shall file a copy of its financial report or a certified financial statement annually with the Commission in order to update the information base for determining financial qualifications.

## ISSUE F-1

## Issue: State and tribal participation

, , · · ·.

Commenters: State of New Mexico (4) & (45) Chem-Nuclear System Inc. (41) American College of Nuclear Physician (53) Department of Planning and Economic Development Hawaii (65) Georgia Institute of Technology (70) Yakima Indian Nation (74) Georgia Yuan (77) Birmingham Audubon Society (80) Utility Nuclean Waste Marte . .**:** · Utility Nuclear Waste Management Group (81) State of California (93) Don't Waste Washington Legal Defense Foundation (97) U.S. Department of the Interior (114) Department of Energy (119)

.

Rule Citations: Subpart F - Participation by State Governments and Indian Tribes (§§ 61.70 - 61.73) . . . f

Summary of Comments:

1 . . .

Five commenters recommended replacing "may" with "shall" in § 61.71 so that the function is an obligation. Two commenters were concerned with relying on Federal Register notices to inform States and tribes. Two commenters requested clarification of Agreement States' role in § 61.70. Three commenters were concerned with limiting participation to those truly involved or affected. One commenter noted that the rule gives ample opportunity for expression of State concerns and three commenters believed additional provisions should be made. One commenter suggested reconsidering the notice of intent. Two felt a better public participation program is needed. One suggested a requirement to comply with state laws. One suggested additional uses of "shall." One suggested Burau of Indian Affairs involvement.

. C. ..

## Analysis of comments:

The commenters are correct that making Commission staff available is an optional function in § 61.71 as proposed. Staff was concerned about possible budget restraints, especially on travel funds, when making the function optional. Changing to "shall" obligates only staff time not travel so the suggested change was adopted. 

The reference to notice in the Federal Register in § 61.72(a) was used to establish a date to begin counting the 120 days. It was not meant to be the requirement for notification. Notification of States and tribes is covered by proposed revised § 2.101(b). Direct notification is required.

Agreement State participation under Subpart F is not specifically addressed. Since Commission licensing would be in non-Agreement States, the most likely affected States are non-Agreement. However, location near a State border or other circumstances might lead to participation by more than one state. Thus, no restrictions based on Agreement status was intended and the language in the rule does not impose any restrictions.

The concern that participation be limited to those truely involved was considered by the staff in developing the decision criteria in §61.73 and true involvement will be considered in the review of proposals.

Georgia Institute of Technology comments focused on the apparent potential for conflict between State, Federal, and tribal laws and questioned which would prevail or is a case-by-case determination required. Overlaping authorities and multiple permitting and licensing are an everyday fact of life. The potential exists for conflict but the constantly changing nature of the requirements and the variety of requirements and authorities preclude any generic statement on this issue. The commenter also questioned the meaning of "relevant tribal law" in § 61.72. Such questions are resolved on a case-by-case basis. No change to the rule was adopted.

Georgia Yuan addressed State and Tribal participation as part of the overall political context and noted that the traditional licensing process that uses formal hearings to resolve issues important to the public and local governments is not a satisfactory solution to gaining public acceptance of sites. She outlined the following weaknesses in the procedures for State and Tribal participation proposed in Subpart F: lack of decision making power, lack of guarenteed influence over decisions, lack of standards for approving participation, and no guarenteed pre-application involvement. She suggested reconsidering the requirement for a notice of intent 3-6 months prior to submitting an application and requiring documentation of early public input and participation.

Yuan addressed, in very general terms, the construction of a successful public participation program. The following goals were suggested:

1. identification of public opposition or support and its causes;

- 2 identification of local or State preferences for locations within the State or region for radioactive waste disposal;
- 3. identification of preferences for State or Federal ownership after disposal operations have ceased; and
- 4. identification of the need to compensate the local population for increased risks resulting from waste disposal.

She encouraged educational opportunities and local citizen involvement not just States and Tribes. She identified NRC use of public opinion and input as a key issue. Her bottom line was "The Commission must seek a more interactive relationship with a broad spectrum of the public before it can begin to define the public interest and rely on it in its licensing decisions."

Yuan's dissatisfaction with the hearing process and public input into regulatory decisions is a generic one - not limited to LLW disposal. Waste disposal is noted as an area of particular concern to the public. While the Commission acknowledges the concerns and problems and that informed public input is particularly important in LLW disposal, it does not believe that the Part 61 rulemaking in the proper forum for resolving this generic problem. The EIS scoping process will be used to address many of Yuan's concerns. State compacting and landlord activities afford additional potential avenues of expression since States will face the same criticisms and must address public input into their decisions. The LLW Policy Act is particularly important in this reguard and will establish the siting arena where early input is so very important.

The State of California expressed general concern about the "tenor" of Subpart F. The State recommended that the Subpart be completely revised to reduce the adversary tenor and facilitate collegially between the Federal government and the States. The State also believed that the 120 day time limit may be too short for States. Staff disagrees based on State and compact responsibilities under the Low Level Radioactive Waste Policy Act of 1980. Replacing "may" with "shall" in § 61.71 was specifically recommended and was adopted as noted earlier. No specific suggestions for addressing the "tenor" were offered. Funding of participation by local governments was suggested as one step in providing a means for local governments to have a voice in whether and how a disposal site is established in their locale. As noted in Issue F-2, the Commission cannot fund intervenors and cannot commit to funding in a rule. Funding must be congressionally approved through the budget process. One State agency stressed the importance of § 61.72 and State participation in the decision process.

The Don't Waste Washington Legal Defense Foundation "believes that the states have and will continue to assert strong leadership roles in management of radioactive waste." The foundation expressed the view that State concurrence should be required for all licensing actions (i.e., issuance, amendment, renewal, termination). Editorial changes to reflect state concurrence and the compliance with state laws that are consistent with the requirements of the Commission were suggested. Changing "may" to "shall" and similarly stronger statements were suggested for §§ 61.7(c)(1), 61.71, and 61.73. The changes to require state concurrence and Commission findings on compliance with state laws were not adopted.

The Commission agrees that state leadership is essential and that the LLW Policy Act is a key factor. The Commission expects to work closely with the States and does not believe that Part 61 should be amended to reflect Policy Act responsibilities or to put the agency in the position of enforcing State laws or determining compliance with State laws. The Policy Act should give the States adequate voice without explicit concurrence provision in the rule. The suggested uses of shall were adopted or clarifying language adopted.

•••

The U.S. Department of the Interior suggested coordination and participation by the Bureau of Indian Affairs (BIA) when sites are to be located on Indian reservations. Area BIA directors and the BIA Office of Trust Responsibilities should be specifically involved. The Department's comments were based on potential sites being located on Indian reservation. Such location is not provided for in the rule. Only State or Federal ownership is allowed. The Federal ownership was not intended to include Indian reservation where the trust responsibilities would come into play. Tribal land ownership and custodial responsibilities was an alternative considered but not adopted. The Department's comment is a good one for assuring involvement of key people to deal with concerns of affected tribes. Disposal sites might be located or lands where residual tribal rights exist or where transportation access involves the reservation. Being host to the site is not a prerequisite for participation under Subpart F. Reference to BIA was added to the list of people to notify when an application is tendered in 2.101(b) to foster early BIA involvement.

#### Rule Changes:

1. §61.71 State and Tribal government consultation.

Upon request of a State or tribal governing body, the Director shall make available Commission staff to discuss with representatives of the State or tribal governing body information submitted by the applicant, applicable Commission regulations, licensing procedures, potential schedules, and the type and scope of State activities in the license review permitted by law.

2. Change last sentence of 61.71 to read: In addition, staff shall be made...

3. Change 61.73(a) to read: "...the Director shall arrange..."

4. Add to 2.101(b)(1)(i): "The Commission will also inform the U.S. Bureau of Indian Affiars when tribes are notified."

#### ISSUE F-2

Issue: Local government government role and rights

<u>Commenters</u>: Township of Lower Alloways Creek, New Jersey (88) State of California (93)

Rule\_Citations: §§ 2.101(b)(1)(i), 61.59, 61.50-73, 61.80

<u>Summary of Comments</u>: The township raised the following issues: advance notice to local governments that an application will be filed, include nearby nuclear activities in the site EIS, the need for additional assurances to local government on landlord performance, receipt of disposal records during operations, involvement in closure amendments, compensation for property devaluation, and annual public seminars. California raised the issue of funding local participation.

<u>Analysis of Comments</u>: Early working drafts of proposed Part 61 included a provision for applicants to file a notice of intent to file an application 3 months before filing. Local officials would have been notified of the applicant's intent. The requirement was not included in proposed Part 61 because it added an administrative burden on the applicant, early state input would probably be involved without it because of LLW Policy Act activities, and a tendering step or other means could accomplish the objective of early notice. Section 2.101(b) requires notification of local officials at the tendered step but the officials will likely be involved much sconer. States and tribes may submit proposals for participation under Subpart F of Part 61. The State and tribal plans for local participation is a required item in proposals. Local governments should be involved in the State efforts at an early stage in the Policy Act compact activities and the applicant should also work closely with local officials to identify key concerns and reflect those concerns in the environmental report and application. The notice of intent concept was not adopted in the final rule since staff believes ample opportunity exists for local officials to be heard and, if not, case-by-case considerations are possible. Local officials can also pass ordinances requiring notification separate from Part 61 if local ordinances and zoning activities do not already provide an opportunity for input.

The township's suggestion that the environmental impact statement (EIS) for a specific site should include consideration of the impacts of other nuclear activities in the area was noted by staff. However, Part 61 does not address the required content of an EIS so no rule change is involved.

The township expressed the view that local governments need additional assurances that the State or federal government will fulfill obligations as landlord and long term care custodian for site security and environmental monitoring The commenter suggested that the State or Federal government enter into a contract with the local community for corrective measures in the event of contamination problems and post bonds. It is beyond the Commission's authority to require such contracts and bonds so the suggestion was not adopted.

The township also suggested approval rights for local governments for long term care funding arrangements. The Commission can certainly consider local views in its review but cannot delegate its decision responsibilities. In some cases, there may be no local community or government and each long term funding arrangement may be unique. For these reasons, no provision for local government approval was added to § 61.63 which addresses financial assurances for institutional or long term care.

On the issue of providing disposal records to local officials during operations, the township expressed scepticism about making arrangements with the State and believes records would reduce public concern. Under proposed 61.80(e), records are to be transferred at license termination to local officials, among others. Thus under the proposed rules, local officials might not receive disposal records until after a hundred years of custodial care. The disposal facility operator is required to keep and report data on disposed wastes all during operation. Annual reports on disposal are required by § 61.80(h). All local officials may not want the burden of receiving, filing, and storing shipping records. Therefore, since the annual reports are on file with the NRC, local officials who wish may request copies of the annual reports.

The township wanted assurances that local officials would be notified of applications for closure and that hearings would be offered. Section 61.25 specifically provides for 30 days notice of hearings for closure amendments. Proposed § 2.104(e) requires notice to local officials for Part 61 licenses but not amendments. A requirement to notice State and local officials was added to § 61.25 in response to the commenter's concerns.

The township's suggestion that compensation for loss of property values or tax bases be provided is beyond the Commission's authority and could not be adopted.

The Township's suggestion that disposal facility operators hold annual educational seminars is a good one for operators to foster good neighbor relationships. Staff has reservations about mandatory seminars since there may be no local community or no interest by the local community and public relations efforts are normally beyond Commission requirements. Chem-Nuclear System, Inc. conducts orientation tours and works closely with local officials in Barnwell, S.C. No State or Federal prodding was necessary to foster this relationship - it is good business practice.

The State of California points out that NRC retains sole authority to issue the license for a Part 61 facility and references state and tribal participation in § 61.70-61.73. The state expressed the view that local jurisdictions should have a voice in the decisions and that the Commission should consider funding such participation. The reference to local jurisdiction is not clear and is probably to State and tribal jurisdictions when considered in context. At any rate, the Commission lacks authority to fund intervenors in cases whether local, State, or Tribal. The participation provisions in Subpart F assisting the Commission in its reviews may or may not involve funding. The Commission cannot commit to funding in a rulemaking.

<u>Rule Change</u>: Add to § 61.25 "The Commission shall provide a copy of the notice of opportunity for hearings in paragraph (a)(1) of this section to State and local officials or tribal governing body specified in § 2.104(e) of Part 2 of this chapter."

### ISSUE G-1

1.1

<u>Issue</u> : S	ubpart G records	, reports, test	s, and insp	Dections	
<u>Commenters</u> : E	xxon Nuclear Com	pany (15)			
	oseph H. White I	II (21)			
A	merican Nuclear	Society (87)			•••••••
S	tate of Californ	ia (93)		.1	
A	tomic Industrial	Forum (100)			
	onference of Rad	iation Control	Program Di	rectors,	[nc: (103)
C	aroline Power an	d Light Company	(106)	-	
and the provide of A	merican Society	of Mechanical E	ngineers (	<b>L07)</b>	
N	ew England Nucle	ar (110)	5	-	1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1
U	.S. DŎE (119)				
U de la composición de U	.S. EPA (122)				
the state of the second se		- ,			

Rule Citations: §§ 61.80, 61.82

at the type to

....

. 1 Summary of Comments: One commenter suggested that any release to unrestricted areas be immediately reported to and investigated by NRC. One suggested duplicate sets of records. One emphasized use of existing forms and documents. One suggested a resident NRC inspector and two, state participation in inspections. One suggested specifying record recipient and one suggested a requirement to maintain records during institutional control and require the operator to transfer records to the landlord at license transfer. Several suggested site that § 61.82(a) be modified to clearly preclude inspection of wastes after disposal. A statement of the constraint of the c

Analysis of Comments: ... Mr. White addressed paragraph (h) of § 61.80 which requires annual reports by each Part 61 licensee. He suggested that any release to unrestricted areas be immediately reported to and investigated by NRC. The listed required contents include in (2)(i), "specification of the quantity of each of the principal radionuclides released to unrestricted areas win liquid and in airborne effluents during the preceding year." The intent "was "to" require licensees to report releases: made pursuant ito § 20.106 of Ler Part 20 and any specific requirements in the license. The report would be of a summary nature and the reporting requirement would not relieve the licensee from reporting incidents as required by 20.403, excessive levels as required by 20.405, or exceeding action levels in operating procedures. 

a fel bena interna interna interna anterna anterna assimilaren el sedaren el astronomia de sedaren el 'As a practical matter, only minimal effluent releases are expected from disposal facility operation. (Effluents are releases from stacks, pipes, filter exhausts etc.) The exhaust from the ventilation system of a storage building is one example. A second is slightly contaminated precipitation which collects in operational trenches and might be pumped out and released. Action levels for these effluents will be established in the licensee's operating procedures. The action levels and releases are subject to ALARA evaluation also. It is unrealistic to assume that these releases will be zero. They may not be at detectable levels but they will not be zero. The commenter's suggestion is not realistic and would place an undue hardship on both the licensee and the NRC and was not adopted. . . . . .. .

The State of California expressed support of NRC opportunity for inspection as provided for in proposed § 61.82. The State also requested an explicit provision "that host states enjoy a similar right." The host State will most likely be landowner and long-term custodian to provide institutional control. Both of these functions provide a means for assuring State inspection rights separate from any explicit provision in the rule. The lease can address the issue. The agreement to assume responsibility for institutional control can also address the issue. An additional mechanism is licensing NARM (naturally occurring and accelerator-produced materials) by the host States. The Commission lacks authority to license these materials and if the State has a licensing program for the materials, it can issue a license for disposal of these materials. Its licensing program should include inspection rights. For the improbable case of Federal landownership and operation, the issue can be addressed on a case-by-case basis through memoranda of understanding between the parties. Since options exist for the States, no change to the rule was adopted.

The Conference of Radiation Control Program Directors suggested a requirement for maintenance of a duplicate set of vital records in § 61.80. The duplicate set should be at an alternate location in case of fire or other loss of the primary records. While the Commission shares the Conference's concern that records are important to both the operator and the custodian, for evaluating monitoring data, closure, remedial actions, etc, it feels that a requirement for duplicating all records and storing them elsewhere is not warrented. No such requirement exists for other records in any of the Commission's regulations at this time.

1.01

Carolina Power and Light Company encouraged the use of existing forms and documents where possible to minimize the administrative burden. Such use is certainly the Commission's intent in § 61.80 and in the proposed new manifest system in § 20.311. The commenter did not suggest any changes to the rule or mention any specific requirements. Staff agrees with the thrust of the comment.

New England Nuclear recommended two additional safeguards concerning inspections. One was to assign a full-time NRC inspector to each site during the operational phase. A second was to encourage active monitoring and review of site records by State authorities. Section 61.82 allows Commission inspection but does not require it or require it at any frequency. The staff agrees that inspection is an important safeguard. The Commission participated in the fulltime inspector program the States of S.C., Washington, and Nevada instituted in 1979 when lack of compliance with DOT shipping requirements was found to be significant. In effect, the inspectors were an independent quality control program on incoming shipments. Such independent quality control on incoming shipments and on all aspects of site operation may not be necessary at all times. Flexibility to adjust priorities should be maintained. The requirement for licensees to bear the expense of inspections under 10 CFR Part 170 is also a factor. The need for thorough inspections and careful quality controls by the licensee and confirmation by NRC are acknowledged but no committment on inspection policy was added to the rule.

The second New England Nuclear recommendation to encourage active monitoring and review of site records by state authorities is a good one. It can be accomplished through memoranda of understanding with States or technical assistance arrangements. If the State is landowner, access to and monitoring records will probably be a part of that responsibility. Such monitoring and review can be one condition of the certification to assume institutional control. No change to the rule is necessary to implement this suggestion. The Departmentof Energy (DOE) suggested specifying to whom the records will be transferred in § 61.80(b). The suggestion was adopted.

The U.S. EPA suggested that in view the importance on the nature of the hazard of disposed waste that a requirement be added to the rule to require transfer of such records at license transfer to the site owner. The rule has two requirements that address this point: (1) § 61.30(a)(3) on license transfer requires transfer of "necessary records for care: and (2) § 61.80 requires maintenance of all records unless disposition is authorized and transfer of records on disposed waste to a variety of officials after institutional control as part of license termination. These provisions collectively provide the "positive" requirement suggested by EPA and no further change is needed.

The staff agrees with the commenters who were concerned that the wording in § 61.82(a) implied that wastes would be inspected after disposal and the rule was modified to clarify that such is not the intent.

#### Rule Changes:

1. Add to 61.80(b) after transferred: "to the officials specified in paragraph (e) of this section."

2. Insert "not yet disposed of" after "inspect radioactive waste" in 61.82(a).

122 5

n ser se an transferante Na Friske en Store

### ISSUE Part 2 - 1

Issue: Part 2 - General

<u>Commenters</u>: Exxon Nuclear Company, Inc. (15) Chem-Nuclear Systems, Inc. (41) Stone and Webster Engineering Corporation (95) Atomic Industrial Forum, Inc. (100)

#### Rule Citations: §§ 2.103, 2.101(b)

<u>Summary of Comments</u>: The three issues raised were the applicability of revised § 2.103 to Part 61 licensees, notification of chief executives for alternative sites, and preparation of value impact statements.

Analysis of Comments: Exxon Nuclear Company and the Atomic Industrial Forum questioned whether revised'§ 2.103(a) applied to disposal facilities licensed pursuant to Part 61 and recommended wording be included in the paragraph similar to that in revised § 2.104(e). Paragraph 2.103(a) deals with two issues. One is the authority to act on applications and issue the license. The first sentence provides this authority and applies to all licenses under any part of 10 CFR including Part 61 when effective. The second issue and sentence deals with who to notify of the action of issuing a license. The current version of 2.103(a) lists facilities, commercial disposal of wastes from other persons, and high-level waste repositories as categories of licensees requiring notification of State, Indian Tribe, and local officials. The proposed revision was to delete the reference to commercial disposal. The commenter is correct that the notification provision does not apply to Part 61 licensees. The requirement to notify officials of license issuance was moved to the proposed new paragraph 2.106(d). Section 2.106 is entitled Notice of issuance so the move was logical editorially. The proposed 2.106(d) also requires notice for major amendments and is therefore more comprehensive than the present requirement in § 2.103. Proposed 2.106(d) also reflects 2.104(e) wording as suggested by the the commenter. No change to the proposed rule is required. A cross reference was added to § 2.103 to clarify the matter.

Chem-Nuclear Systems, Inc. questioned whether the requirement in proposed § 2.101(b) to notify the chief executivies of locations for alternative sites is consistent with the requirements of Part 51 and proposed Part 61. Specifically, the commenter questions whether specific alternatives sites must be identified in the application. If specific sites do not have to be identified, Part 2 should be amended to delete the requirement to notify the officials. The staff's views on alternative sites are articulated in the draft branch technical position in the site selection discussion on pages 9 and 10 of NUREG-0902. The position states that to meet NEPA, there should be comparison between the preferred site and two or three viable alternative sites. It also indicates that the major portion of detailed site characterization efforts are expected to be performed at the preferred site. Thus alternatives will probably be identified and the notification requirement was retained.

Stone and Webster Engineering Corporation referenced proposed § 2.764(e) which would require Commission approval before the Director of NMSS could issue a license or amendments pursuant to Part 61 and indicated that issuance of the license or amendments should be justified by preparation of value/impact statements. The comment is basically a procedural one indicating a specific method of documenting Commission evaluation. Under the existing provisions of 10 CFR Part 51 and under proposed editorial changes to Part 51, preparation of an environmental impact statement is required for the initial license. (See 10 CFR 51.5(a)(6).) A value/impact statement would duplicate this effort already required in the process. Further, major amendments that involve "actions which may significantly affect the health and safety of the public" would involve environmental appraisals (EA) and probably supplements to the initial EIS or a new EIS because of the "significant affect." Amendments are actions where preparation of EISs is optional (See 51.5(b)(4)(iii)). The EIS, EIS supplement, or EA would include economics - evaluation of costs and benefits - which is the point of the value/impact statement - and would be more comprehensive in scope. The Commission can request supplemental evaluation on a case-by-case basis without imposing another administrative burden on all licensing actions. Thus the commenter's suggestion was not reflected in the final rule.

#### Rule Changes:

 Add to § 2.103(a) Note: For notice of issuance requirements for licenses issues pursuant to Part 61 of this chapter, see paragraph 2.106(d) of this part."

#### ISSUE Part 2-2

Issue: Issuing licenses

<u>Commenters</u>: Isham, Lincoln and Beale for Commonwealth Edison (18) State of California (93)

Rule Citation: §§ 2.764, 61.3, and 61.7

<u>Summary of Comments</u>: One commenter suggested the option of immediately issuing licenses when the hearing board directs and one expressed reservations about not waiting for all appeals to be resolved.

<u>Analysis of Comments</u>: Isham, Lincoln, and Beale noted the need for "prompt establishment of a Midwest low level waste facility" and expressed concern that proposed language and discussion did not clearly provide the authority to issue a license while administrative and judicial appeals are resolved. The commenter expressed the view that the rule should allow licenses to be effective upon final review by the Commissioners, at the latest.

The existing provisions of § 2.764 do not apply to materials licensees such as disposal facility licensees, since mandatory hearings, construction permits, and operating licenses are not involved. The proposed change to 2.764 to address Part 61 licensing actions reflects Commissioners' desire to review all significant Part 61 licensing actions prior to issuance. The proposed change to § 2.764 has been moved to form a new § 2.765 to avoid amending provisions not applying to materials licensees.

The Commission agrees with the commenter that the option to issue the license without waiting for resolution of all appeals should exist. No changes to the rule or related amendments are necessary to assure this option, however. The Commission doesn't have to wait for court appeals under existing or amended rules. Who issues the license does not affect this option.

The State of California was concerned that the proposed modifications to § 2.764(a)(b) and (e) would render state's appeals ineffectual and cited the State's experience in Commission reactor licensing cases. The Commission's intent is to weigh the issues and decide whether the license should be immediately effective or not. The option to act before resolution of all appeals does not mean that valid concerns would be ignored or that action will be taken before resolution of appeals.

#### Rule Change:

÷ •

Move the proposed change to § 2.764 to form a new §2.765.

## ISSUE M-1

Issue:

## General comments on § 20.311

Commenters:

. . .

The Procter and Gamble Company (6) University of California, LA (8) · • • • • • . United Technologies/Packard (25) Moward University (49); Union Oil Company of California (66) Georgia Institute of Technology (70) University of Texas Medical Branch (75) General Electric (89) Stone and Webster Engineering Corp. (95) Health Physics Society (96) Atomic Industrial Forum, Inc. (100) U.S. Ecology (101) The American Society of Mechanical Engineers (107) & (113) New England Nuclear (110) Tennessee Valley Authority (116) Texas Department of Health (117) U.S. Environmental Protection Agency (122)

### Rule Citation: §20.311

Summary of Comments: Two commenters addressed chemical form. One addressed exempting Class A wastes. One requested clarification of most of the information requirements in § 20.311(b). One commenter suggested adding dates activities are specified and two suggested adding radiation levels. Two suggested reducing the 60 day disposal facility report period. One requested clarification of Class A waste form requirements. One addressed duplication of manifest transfer requirements when waste collectors are involved. Two were concerned about delays until forwarded copies are received. The Health Physics Society raised several points for clarification. Two encouraged compatibility with existing systems. One commenter addressed the required number of copies and two addressed the inspectability of the system. One suggested relief from classification as A, B or C when transferring to processors. One addressed the logistics of prior notification. Two suggested placing the system in 10 CFR Part 71. 

Analysis of Comments: The Proctor and Gamble Company comments were primarily focused on exempting Class A wastes from most of the manifest requirements as discussed and rejected under Issue 1-4. Specifically, the Company recommended that 20.311(e)(8), (f)(1), (h)(1), and (h)(2) apply only to Class B and C wastes. The Company also recommended inserting "general" before "chemical form" in 20.311(b) which lists required information in manifests to provide additional relief and flexibility. The University of California was also concerned about chemical form and recommended a 500 gram cutoff for specifying chemical form as discussed and rejected under Issue M-4. Some relief was provided by inserting "principal" without quantifying what "principal" means.

United Technologies/Packard suggested that the meaning of the information requirements in § 20.311(b) be clarified (i.e., person generating the waste, type of waste, waste volume and mass, radionuclide identity and concentration, and total activity). The definition from the Environmental Protection Agency's rules in 40 CFR 260.10(a)(26) was suggested for guidance to clarify that the individual who generated the waste was not intended. The use of person was in the same sense as defined in § 61.2 and is the licensee. The purpose of identifying the generator was twofold: 1) to provide a source of information about the waste if questions or problem arise and 2) to enable development of a representative data base showing factors such as actual generators, type of licensee, and state where generated rather than data skewed by large volumes from brokers or waste collectors. Packard suggested "physical description" instead of "type of waste" to clarify intent. This suggestion was adopted. Packard questioned whether drum size (e.g., 55 gallons) was adequate specification of volume. Drum size is acceptable. Disposal charges are usually based on volume and no burden is involved in specifying volume. Packard indicated that the weight of the drum or package might be difficult to determine without large scales. Weight would likely be estimated for handling or freight charges but staff agrees that it might be difficult and that the information is not essential so mass was deleted. The difficulties in specifying radionuclide identity, concentration, and total activity are addressed elsewhere.

Howard University suggested that the specification of activity be as of the generation date. Such specification would be useful for evaluating radioactive decay of short-lived nuclides during storage at the site of generation, and during collection and transport to the disposal site. The suggestion was not adopted. The decay before shipment should be taken into account by the generator preparing the manifest. Howard also recommended reducing the 60 day time period in 20.311(g)(5) for reports by disposal facility generators when shipments do not arrive within 60 days after advance manifests are received. This suggestion was not adopted since the facility operator reports are a backup system for checking to see that shippers, i.e., waste generators or waste collectors have conducted required investigations. Shippers must investigate within 20 days. Georgia Institute of Technology made a similar point.

Union Oil Company suggested that § 20.311(d) be clarified to indicate which sections of § 61.56 are applicable to Class A wastes. Section 61.56 is being restructured to clarify which requirements are applicable to which class of waste alleviating the need to clarify § 20.311(d).

The University of Texas Medical Branch expressed the view that paragraphs 20.311(d)(1) and (2) duplicate the requirements of (e)(3) and (4). Paragraphs (d)(1) and (2) deal with generators preparing and labeling the wastes and (e)(3) and (4) deal with collectors forwording and including manifests. The commenter appears to be objecting to filling out a manifest when the collector is going to fill one out for sending the waste to the disposal site - a duplication of effort. The uncertainty about roles when the collector picks up the shipments at the site of generation also appears to be part of the issue. The commenter is correct that some duplication of effort is involved. However, the duplication serves at least two purposes: (1) emphasizing the waste generator's responsibilities for providing correct information on waste content and (2) documenting the responsibility through the certification required. Thus, the requirement was retained. General Electric and the American Society of Mechanical Engineers expressed similar views on the need to forward a copy of the manifest. The commenters were concerned that shipments must be delayed until the advance copies are received. Such receipt was not intended. The intent was to ship and independently forward a copy at the same time. The wording in 20.311(d)(5) to forward "at the time of shipment" was intended to make this point. When a shipment is picked up f.o.b. the generator's facility, the manifest can be transferred at that time. In addition, the society recommended adding the "radiation level" to the manifest.

1 1

Stone and Webster Engineering Corporation also suggested that radiation levels be included in the required content of manifests in § 20.311(b). The DOT regulations have specific requirements for radiation levels and transport indices in shipping papers and labeling. No exemption from these requirements is given by the rule and in view of the complexity and potential for modification of the DOT regulations, the requirements were not repeated and the suggestion was not adopted.

The Health Physics Society expressed concern about the long term generator liability implied by the certifications concerning waste characteristics in 20.311(c) as required of generators in (d)(1) and waste processors in (f)(3). This issue is discussed under § 61.56 of the rule and issue M-2 on quality assurance. The society also raised several questions about the discussion of the system in the Supplementary Information. The intent and implications of the reference to improving the credibility of decisionmakers was one question. The improved data base will be helpful to licensing agencies, State and compact groups, and disposed facility operators in that actual data on shipments will be available not gestimates or extrapolation from surveys. The staff cannot quantify the incremental cost of manifest system or the value of more informed Incremental costs of the system are small since existing practices decision. cover most of the requirements. The system does require generators to comply with slightly modified DOT requirements as observed by the Society. The rule was changed to emphasize that one set of papers may be used for NRC and DOT requirements in response. The society's question about the meaning of inspectable is addressed in the following discussion of New England Nuclear's comments.

The Atomic Industrial Forum urged that manifest requirements be compatible with existing requirements to minimize duplication. The U.S. EPA noted that the NRC manifest system and EPA hazardous waste manifest system as a minimum must be compatible and encouraged future coordination. Staff concurs.

Minor conforming changes to the information required on the NRC manifest in § 20.311(b) were made to reflect the joint EPA/DOT proposed Uniform Hazardous Waste Manifest (47 FR 9336, March 4, 1982). For example, use of the transporter's EPA hazardous waste identification number was added as an option. Use of a single form and/or use of this joint standard form was highlighted in the Supplementary Information position of the final rule and the rule itself.

U.S. Ecology observed that the requirements of § 20.311 can apparently be met with 3 copies of the manifest but that 5 copes have proven useful in everyday practice. Staff preferred to let the parties involved determine the number of copies needed for other purposes and did not specify numbers of copies.

. .

Section for the sector of the

New England Nuclear recommended that a procedure be developed to integrate enforcement agencies into the control or supervision of the manifest system. A requirement for the site operator to return a receipted copy of the manifest to the generator and to require the generator to maintain files of shipment manifests and backrouted receipts was suggested as a possible procedure. Inspectors could review the files during routine inspections of the generators. The Commission agrees that such a procedure has merit and that the proposed language did not indicate how to generate inspectable records. The intent in the proposed rule was to rely on existing requirements in Parts 30, 40, and 70 (see §§ 30.51(c), 40.61(c)(3) and 70.51(b)(5)) for maintaining records of transfer of materials. These requirements could be met by keeping copies of the manifests. However, retained copies, receipted copies or documentation of receipt by telephone acknowledgement would not be required. The commenter rightly points out that the proposed system does not guarantee inspectable records at the generator's facility and should since the generator has prime responsibility and the rule was changed accordingly. The system does provide an inspectable cross check by having the disposal facility operator report when a shipment has not arrived within 60 days after the advance manifest was received. The Commission can thus compare reported investigation to lists from facility operators and investigate any discrepancies. To guarantee inspectable records at the generator's facility § 20.311(d)(7), (3)(5) and (f)(8) were changed to require the manifest or equivalent documentation such as a computor printout containing the same information and a binding acknowledgement as a specific way to meet existing recordkeeping requirements. Thus the new language would require maintenance of records for the same time periods but would exclude recordkeeping options such as log entries only.

Parts 50, 60 and 72 contain no specific provisions on maintaining records of transfers of materials. Each is noted below:

- Part 50 No specific transfer or records of transfer requirements. All transfers done under Part 30, 40, or 70 possession licenses.
- Part 60 No specific transfer or records of transfer requirements. Only licensee will be DOE should be little or no waste to a Part 61 facility.
- Part 72 No specific transfer or records of transfer requirements except safeguards reports in 72.54 for spent fuel. Will be less than 10 licensees. Should only be a small amount of wastes from water treatment systems that would ever be consigned to a Part 61 facility.

The Texas Department of Health expressed the view that generators shipping to intermediate processors should not have to package and classify wates as Class A, B, or C. The concern was the unnecessary expense of Class B or C packaging. Staff agrees and Class B and C packaging requirements were not intended to apply to transfers to intermediate processors where the waste was to be treated or repackaged. The wording was revised to clarify this point.

The Tennessee Valley Authority noted that copies of the manifest forwarded at the time of shipment to the intended receipient (as required of 20.311(d)(5) for example) by mail would probably not reach the recepient before the waste shipment. It would not be prior notice. The commenter's observation is true, but the purpose of forwarding a copy was to provide a means of cross checking on shipments. The disposal facility operator is required to check the independently forwarded copies against shipments received and report any mismatches. Prior notification was not the prime purpose. Any State or facility operator requirements for prior notification are in addition to the manifest system. However, the manifest can be used for prior notice if desired. The Authority also suggested use of DOT forms and placing the requirements in 10 CFR Parts 61 or 71 not Part 20. The manifest includes additional information not included in DOT papers so this suggestion was not adopted. The manifest should suffice as DOT papers so only one set is required and no duplication should result. The requirement was placed in Part 20 with other waste disposal requirements since it applies to all licensees transferring wastes and Part 61 applies only to disposal facility licensees and not in Part 71 since Part 71 deals with packaging requirements that apply to all radioactive shipments not just waste shipments. The Health Physics Society also suggested considering adding the requirement to Part 71.

.

#### Rule Changes:

1. Change "type of waste" to "physical description of waste" in § 20.311(b), add "principal" before "chemical form", and delete "mass".

- 2. Add to 20.311(b): "The manifest required by this paragraph may be the shipping papers used to meet Department of Transportation regulations, or requirements of the receiver, provided all the required information is included."
- 3. Amend 20.311 (d)(7) to read:

:

- (7) Retain a copy of the manifest and documentation of acknowledgment of receipt as the record of transfer of licensed materials as required by Parts 30, 40, and 70 of this chapter.
- 4. Amend 20.311(e)(5) to read:

(5) Retain a copy of the manifest and documentation of acknowledgment of receipt as the record of transfer of licensed material as required by Parts 30, 40, and 70 of this chapter.

;

- 5. Amend 20.311(f)(8) to read:
  - (5) Retain copies of original and new manifests and documentation of acknowledgment of receipt as the record of transfer of licensed materials as required by Parts 30, 40, and 70 of this chapter.
- 6. Amend § 20.311(d) to read: Any generating licensee who transfers radioactive waste to a land disposal facility or a licensed waste collector shall comply with the requirements of subparagraphs (1) through (8). Any generating licensee who transfers waste to a licensed waste processor who treats or repackages waste shall comply with the requirements of subparagraph (4) through (8).

#### ISSUE M-2

Issue:

The quality assurance requirements in §20.311

Commenters:

The Proctor and Gamble Company (6) Exxon Nuclear Company (15) Dow Chemical (17) United Technologies/Packard (25) NASA, JFK Space Center (26) Los Alamos National Laboratory (43) Georgia Institute of Technology (70) University of Texas Medical Branch (75) Northeast Utilities (85) Health Physics Society (96) Atomic Industrial Forum (100) The University of Texas System Cancer Center (105) The American Society of Mechanical Engineers (113)

### Rule Citation: § 26.311(d)(3), (f)(5)

<u>Summary of Comments</u>: One commenter recommended exemption of certain low concentration wastes from the quality assurance (QA) requirements. One commenter suggested that the requirement be clarified to allow minimal programs for licensees handling minimal amounts of wastes. The resource burden on small operations for Q/A was noted by three commenters. Five commenters also suggested more specificity on requirements. Two commenters suggested the use of "quality control" instead of "quality assurance." One was concerned about management's personal involvement. One was concerned about a separate program being required. One suggested shifting part of the burden on waste form quality control to suppliers.

<u>Analysis of Comments</u>: The proposed requirement for a quality assurance program to assure compliance with §§ 61.55 and 61.56 was intentionally stated in general terms in recognition of the broad spectrum of waste generators who would have to comply. Flexibility to tailor the program to types and amounts of waste was intended. Since it is important that all wastes be properly classified, no exemptions for wastes were provided. The quality assurance program applies only to wastes being transferred for dispoal at a land disposal facility. Wastes being disposed of by other methods or without regard to its radioactive content (e.g., under the provisions of § 20.306) are not covered.

The references to resource burdens were very general and few in number considering the number of licensees potentially affected and to whom copies of the rule were sent. More specific guidance on what should be included in the program is planned for the regulatory guide on waste classification. In view of the diverse waste generator population and range of programs expected, guidance documents would seem more appropriate than prescriptive requirements in the regulation.

The suggested word change from "quality assurance" to "quality control" was based on the accepted use of the terms in Appendix B of 10 CFR Part 50. In the introduction to Appendix B, quality assurance includes the concept of performing satisfactorily in servce while quality control is described as controlling the quality to predetermined requirements. The generator was not

expected to evaluate disposal site performance. The waste characteristics and · 2 classification requirements are the predetermined requirements. Therefore the suggestion was adopted. .:.. 2.4

.

The Proctor and Gamble Company recommended modifying § 20.311(d) to replace "the program must include management audits" with "audit results must be reported to management." The proposed change would change the requirement so that management would not have to personally conduct the audits. The Commis-sion's concern is to make sure the audits are reviewed and corrective measures taken when necessary not just reported and filed. Therefore, wording to address this concern has been added to require management to evaluate audits, but not necessarily conduct them.

Exxon Nuclear Company, Inc. was concerned that the language in § 20.311(d)(3) and (f)(5) implied that a separate quality assurance program he instituted. A separate program was not intended.

The Georgia Institute of Technology suggested that a provision be added for certain blanket authorization based on supplier information on waste form. The Institute suggested that such blanket authorization would help shift part of the burden for compliance with § 61.56 on waste form to suppliers. The staff agrees with the concept that suppliers can develop generic information on products and plans to review topical reports on waste forms and containers. However, how such products and packaging is used influences the validity of product evaluations. The generator will still need to be sure proper procedures are followed and that his waste falls within the range of parameters evaluated for the product. Data supplied by supplier and evaluated by the Commission can be included as a key part of the generator's program but cannot be substituted for it. Thus, the suggestion was not adopted.

#### Rule Changes:

Issue:

. .

Change "quality assurance" to "quality control" in 20.311(D)(3) and (f)(5). 1.

- For 20.311(d)(3) insert evaluation after management in "the program must include management audite " 2. include management audits."
  - \_ <u>ISSUE M-3</u>

Burden on small entities (See Issue D-55-11 on cost of classification, and Issue M-4 on manifest a burden also.) <u>Commenters</u>: Wisconsin Electric (32)

Health Physics Society (96)

\_ <u>Rule Citation</u>: General

Summary of Comment: Wisconsin Electric Power Company expressed the view that the rule would greatly increase disposal costs for small entities without commensurate health and safety benefits. The Health Physics Society noted that the rule will impact small entities but much of the impact will be positive.

S. S. A. A. A. P.

Analysis of Comments: The basis for the Wisconsin Electric comment appears to be the perceived costs of the manifest tracking system (Issue M-4) and increased costs to site operators that will be passed on to the small entity. The commenter also appeared to base his concerns on a misunderstanding about the scope and applicability of the rule based on a reference to "currently permitted to dispose of radioactive wastes into sanitary sewage systems." The proposed rule does not negate such existing options as discussed under Issue A-1.

Wisconsin Electric is not a small entity and the comment was not based on costs to the commenter. No specific data or cost estimates are offered.

No rule changes were proposed based on the comments.

#### ISSUE M-4

Issue:

e: Manifest system a burden

<u>Commenters</u>:

The Procter and Gamble Company (6) University of California, LA (8) D. M. Mathews (23) United Technologies/Packard (25) Nuclear Diagnostic Laboratories Inc. (29) & (108) Wisconsin Electric (32) Union Carbide Corporation (39) American College of Nuclear Physicians (53) Department of the Army (63) Georgia Institute of Technology (70) Health Physics Society (96) Conference of Radiation Control Program Directors (103) University of Texas System Cancer Center (105)

Rule Citation: § 20.311

<u>Summary of Comments</u>: Two commenters were concerned about specifying chemical form. Four commenters objected to shipper responsibility for tracking shipments. Three commenters indicated that the system is a paperwork burden and three a general burden. Three supported the system and one indicated no problems in complying. Two objected to forwarding a copy of the manifest and one was concerned about the implications of generator certifications.

<u>Analysis of Comments</u>: Procter and Gamble expressed the view that the manifest system is an unjustified burden for Class A wastes. Procter and Gamble noted the difficulty in specifying chemical form and the industrial security risk for pharmaceutical firms. The University of California, LA noted the large variety and possible unknown species of chemical forms from research and hospital wastes and recommended a 500 gram lower cutoff. Only chemical forms exceeding the value would have to be specifed. The staff considered exempting Class A waste and the <u>de minimus</u> chemical content concept. The rule as proposed required indication of chemical form " as completely as practicable." It did not have an absolute requirement. Thus, the problems with unknown or hundreds of trace chemical forms would be covered as not practicable. Staff was not able to support an across the board <u>de minimus</u> quantity of 500 grams for specifying chemical form as either adequate or too high. Such a requirement is very prescriptive and would impose a potential or implied compliance burden to quantitatively demonstrate analysis of every component greater than the 500 grams. Class A wastes probably represent more than half the volume of waste currently being shipped. Information on such a large portion of the waste is needed to complete the picture. Class A wastes also probably represent the greatest hazard from non-radiological properties so detailed information is useful for managing accidents or incidents in handling the waste. Thus exemption from chemical form specification was not adopted but some relief was provided as noted in Issue M-1. In addition, a significant percentage of wastes are Class A wastes which the staff believes should not be precluded from tracking and reporting on lost shipments.

Objections to the waste generator being responsible for tracking waste shipments were raised by D. M. Mathews, the American College of Nuclear Physicians, and the University of Texas System Cancer Center. Mathews expressed the view that the waste generator would be unfairly penalized for the truckers' or site operators' mistakes by having to conduct an investigation and file a report on missing shipments. The Nuclear Physicians and Cancer Center objected to not being able to transfer responsibility for wastes to brokers or licensed waste collectors and thus transfer the burden of accounting for shipments. Wisconsin Electric expressed similiar views on carrier reponsibility. The Commission's intent in drafting § 20.311 was to allow the waste collector to acknowledge receipt of the waste to the generator and assume responsibility for tracking the wastes. Clarifying language is added to § 20.311(d)(8) to emphasize investigation only if acknowledgement of recept is not received. The generator is responsible for the information provided about the waste and cannot be relieved of this responsibility.

Nuclear Diagnostic Laboratories (a waste collector), Wisconsin Electric and United Technlogies/Packard were concerned about the paperwork burden. Nuclear Diagnostics' major concern stems from the requirement in § 20.311(e)(2) for making copies of the individual manifests prepared by waste generators part of the new manifest the collector must prepare. Consolidated shipments may include wastes from a hundred or more waste generators so that the papers could be a physical burden. The company indicates that an information retrieval system is in place and requests relief from the requirement of physically incuding generator manifests with the new manifest. The Commission agrees that such relief is warrented and such relief was added to § 20.311(e)(2). Nuclear Diagnostics was also concerned about the apparently needless requirement to forward a copy of the manifest (per § 20.311(d)(5)) to the waste collector who picks up the waste at the generator's facility. Tranfer of a copy with the shipment should be adequate. The Commission agrees and provided an exemption in § 20.311(d)(5) for direct transfers to collectors. However, the generator is still responsible for accounting for all waste transferred and making sure that all waste are acknowledged (e.g., per 20.311(d)(7)(8)). The University of Texas System Cancer Center expressed similiar views.

Wisconsin Electric's concerns about paperwork associated with the manifest system were expressed on behalf of "small entity" licensees. No analyses or data were included in support of the claimed burden. United Technologies/Packard and Georgia Institute of Technology concerns were a general caution.

The Proctor and Gamble Company and University of Texas System Cancer Center expressed the view that the manifest system is a general burden. Proctor and Gamble suggested exempting Class A wastes as discussed above. The Cancer

.

Center's concerns were primarily concerned with the collector or contractor/ generator roles. Proposed changes to 20.311(d)(5) and (8) provide some clarification and relief for collectors of prepackaged wastes. However, the generator is responsible for contracted services and no relief from certifications by the generator was given. The generator can certify based on employee (authorized user) or contractor information but the responsiblity remains with the licensee who packages the waste. If the waste collector repackages or treats the waste, 20.311(f) applies and the waste is the processor's responsibility and the Cancer Center's comments do not apply.

Union Carbide Corporation, the Health Physics Society, and the Conference of Radiation Control Program Directors gave general support to the system. Union Carbide noted that documentation has already become a significant part of the cost of waste disposal. The Health Physics Society noted that the required information and processing is good business as well as good health physics. The Society recommended that the system should be consistent with requirements for shipment of other radioactive shipments and hazardous materials in general. To make the system generic to all radioactive shipments, amendment of Part 71 was recommended. This suggestion to modify Part 71 was not adopted for two reasons. One, the information requested in § 20.311 includes information needed for disposal only. Second, the user of radioactive materials is likely to track shipments he needs without regulatory prodding. Wastes are by definition, materials of no further value to the user and incintive to track shipments was needed. The Commission agrees that consistency with other record systems and requirements is desirable. The Conference stated, "We strongly support the proposed amended requirements to Part 20 for the certification and use of shipping manifests to track waste shipments."

The Department of the Army indicated that current practices cover most of the requirements in the manifest system and that the additional information requirements will be added to documents and forms when required. No problems or burdens were indicated.

In summary, only seven commenters voiced problem or burden or a result of the manifest system and four objected to shippers tracking the shipments. These few comments must be viewed in the context of the 20,000 licensees (both NRC and Agreement State) who were notified of the proposed rule in a mailing that focused their attention on the system and provided a copy of the <u>Federal Register</u> notice with the specific requirements.

#### Rule Changes:

•

- 1. Add to § 20.311(d)(8) a reference to acknowledgement of receipt.
- Add to § 20.311(e)(2): The waste collector may prepare a new manifest without attaching the generator manifests, provided the new manifest contains for each package the information specified in paragraph (b) of this section.
- 3. Add to 20.311(d)(5): or deliver to a collector at the time the waste is collected, obtaining acknowledgement of receipt in the form of a signed copy of the manifest from the collector.

# ISSUE GEN-1

· • •

· . · .

49.77.2**1**0.77

Issue:	Absoluteness of Criteria	· .	
Commenters:	State of New Mexico (4)	,	
er 1.4.	Exxon Nuclear Company, Inc. (15) Department of the Environment, London (19) Bechtel National, Inc. (44) Duke Power Company (48)		• • • • •
	Paul F. Hadala and Don C. Banks (76) Birmingham Audubon Society (80) Utility Nuclear Waste Management Group (81)		
enter Constante Constante enteres	Northeast Utilities (85) American Nuclear Society (87) Atomic Industrial Forum (100)		
	U.S. Ecology (101) The American Society of Mechanical Engineers State of North Carolina (109) New England Nuclear (110)	(107)(1	13)
а <sup>на</sup> с с с с с с с с с с с с с с с с с с с	U.S. Department of the Interior (114) Tennessee Valley Authority (116) U.S. Department of Energy (119)		• • • •

<u>Rule Citations</u>: Various - see summary.

•

<u>Summary of Comments</u>: Most of the commenters were expressing concern with the use of absolute terms in the rule such as "eliminate" and "prevent." One was concerned about lack of absoluteness of "reasonable assurance." One was concerned about subjective interpretations.

۰.

The specific citations and commenters are as follows:

Cit	tations	Part 61 Ter	'n	Suggested Replacements	Commenters
1.	61.51(a)(6)	eliminate	i	minimize	Dept of the Envir. London (19 Bechtel (44) Duke Power (48) Hadala and Banks (76)
			دی (م. ا		UWMG (81) Middle South Services (84) ANS (87) Mechanical Engineers (107) NC (109)
			• •.	* .	NEN (110) Interior (114) Tennessee Valley Auth (116) DOE (119)

Citations		Part 61 Term	Suggested Replacements	Commenters Birmingham Audubon Society (80)	
<pre>2. 61.13(d); 61.23(b),(c),(d),(e); 61.30(a)(2); 61.40; 61.51(a)(2); 61.54</pre>		resonable assurance	conclusive showing		
3.	61.62(a)	assurances	proof	Birmingham Audubon Society (80)	
4.	61.51(a)(4)	prevent	minimize	Bechtel (44) UWMG(81) ANS (87) AIF (100) Mechanical Engineers (107) NC (109) NEN (110) Interior (114) TVA (116)	
5.	61.7(b)(2)	eliminated or	delete	NE Utilities (85) AIF (100) NEN (110)	
6.	61.7(b)(1)	prevention	minimize	Bechtel (44) ANS (87) AIF (100) NEN(110)	
7.	61.52(a)(1)	no interaction	no significant interaction	AIF (100) Exxon (15)	
8. Info pg.	Supplementary ormation 38084 col 1	eliminate	minimize	U.S. Ecology (101)	
9.	61.50(a)(4)	significant	none	NC (109)	
10.	61.52(a)(6)	a few percent	none	NC (109)	
11.	61.52(a)(9)	adequate	approved	NC (109)	
12.	61.40	reasonable assurance exists that	delete	NEN (110) NM (4)	
13.	61.62(a)	eliminate	minimize	DOE (119)	
14.	61.80(h)(2)	report any different	report any significantly different	Exxon (15)	

:

. .

•

.

.

,

Citations	Part 61 Term	Suggested Replacements	Commenters	
15. 61.2 "disposal"	isolation	disposal in an approved facility	Bechtel (44) ANS (87)	

11:

#### Analysis of Comments:

Hadala and Banks most succinctly discussed the implications of paragraph 61.51(a)(6) which states:

(6) The disposal site must be designed to eliminate the contact of water with waste during storage, the contact of standing water with waste during disposal, and the contact of percolating or standing water with wastes after disposal.

The commenters show that a literal interpretation of this requirement would preclude ever issuing a license since infiltration can never be eliminated and all soils have some water content. "Minimize" was suggested as an alternative for eliminate. To emphasize that the goal should be a real target the words "minimize to the extent practicable" were used in the final rule.

The Birmingham Aubudon Society suggested the wording changes without comment. The concern seems to be the need for stronger findings but less than guarentees. "Reasonable assurance" is a licensing standard used throughout the agency and reflects the findings staff expects to make. The changes were not adopted. The Issue B-1 also.

The Utility Waste Management Group expressed concern that literal interpretation of 61.51(a)(4) and (6) leads to requirements that are difficult if not impossible to meet by any existing technology. Similar views were expressed for the citations shown in the chart by New Mexico, Exxon, Bechtel, Duke Power, Middle South Sources, Northeast Utilities, the American Nuclear Society, the Atomic Industrial Forum, U.S. Ecology, the American Society of Mechanical Engineers, North Carolina, New England Nuclear, U.S. Department of the Interior, Tennessee Valley Authority, and the Department of Energy. Staff generally agreed with the comments on the absoluteness of the wording identified and modified the language to address the concerns. See Issue D-51-1 also.

The State of North Carolina recommended that the entire rule be carefully reviewed for the use of the terms such as significant, prevent, eliminate, few, and adequate. The State was concerned about absoluteness, subjective interpretation, and ranges of opinions.

New England Nuclear made specific recommendations noted above and a general comment that absolute statements should be replaced by "achieveable practical ones."
## Rule Changes:

Cita	tion	Adopted change		
1.	61.51(a)(6)	"minimize to the extent practicable"		
2.	61.13(d); 61.23(b).(c),	none		
	(d).(e): 61.30(a)(2);			
	61,40: 61,51(a)(2):			
	61.54			
3	61, 62(a)	none		
ă.	61, 51(a)(4)	"minimize to the extent practicable"		
5	61.7(h)(2)	deleted "eliminated or"		
с.	(1, 7, 0)(2)	nevended so "environtion" net used		
0. 7	$G_1, G_2(d)(1)$	reworded so prevention not used		
7.	61.52(a)(1)	reworded so "no interaction" not used		
8.	Supplementary Information	Section not included in final notice		
9.	61.50(a)(4)	used "known"		
10.	61.52(a)(6)	revised to reflect wording in 10 CFR		
		20.105 so few percent not used		
11.	61.52(a)(9)	"approved" used as suggested		
12.	61.40	no change adopted		
13.	61.62(a)	inserted "to the extent practicable"		
		to modify "eliminate"		
14.	61.80(h)(2)	inserted "significantly" as suggested		
15	61.2	added "inhabited by man and his food		
<b>TA</b> .	V212	chains" to emphasize isolate from		
		man not abcolutoly icolato		
		man not absolutely isolate		

## ISSUE GEN-2

Issue:

Additional regulatory guidance needed

<u>Commenter</u> :	Pennsylvania Department of Environmental Resources (16)
	Dow Chemical (17)
	Ohio EPA (38)
	Chem-Nuclear Systems, Inc. (41)
	Union Oil Company of California (66)
	Stock Equipment Company (67)
	Argonne National Laboratory (68)
	University of Arizona (78)
	South Carolina Department of Health and Environmental
	Control (79)
	Northeast Utilities (85)
	State of California (93)
	State of New York (99)
	U.S. Ecology (101)
	Conference of Radiation Control Program Directors (103)
	American Society of Mechanical Engineers (107), (113)
	U.S. Department of the Interior (114)
	U.S. Department of Energy (119)

Rule Citation: None.

<u>Summary of Comments</u>: These commenters made suggestions on the kinds and urgency of additional regulatory guidance that the commenters felt was needed.

Commenter 67 suggested guidance on onsite contingency storage capacity and leachability test standards and criteria. Commenters 38, 66 and 85 urged that the waste classification regulatory guidance be issued as soon as possible. Commenter 16 suggested the following list:

		0		)	Wast	te	stab	<b>i</b> ]	lity;
•	٠.	• `	•	-		۰.			•••

0

Specific assumptions to use for determining whether the intruder scenario meets the performance objectives;

- o Guidelines and assumptions for setting maximum disposal site inventory limits consistent with the performance objectives;
- Concentration limits for naturally occurring and other isotopes (especially Radium) not specifically addressed in Table 1;
- Specific guidance for the information that is requested in a license application as outlined in Subpart B;
- o Definite standards for the conditions that are required to be met for post-closure license transfer and termination of the license.

Dow Chemical (17) listed properties in addition to free liquids that should be included in the NRC Branch Technical Position on waste form.

Commenter 68 throught the model which NRC would use to evaluate compliance with the groundwater migration performance objective be made available. Commenter 78 suggested interagency agreement on a glossary of terms pertaining to waste management. Commenter 79 supported and recommended development and issuance of regulatory guides which incorporate standards for waste form packaging, high integrity containers, test methods, and approval guidelines and processes. Commenter 79 offered their assistance in developing these needed guidelines. Commenter 93 suggested that guidance on the types of records, reports, tests, and inspections needed to show compliance with Subpart D should be provided. Commenter 99 suggested guidance or requirements regarding soil horizon characteristics as well as guidance for engineering features which effectively respond to leachate management, subsidence control, and aqueous and aeoline erosion.

Commenters 101 and 114 requested a statement (guidance) on what criteria should be used to define the requirement in 60.50(a) that a site should be "capable of being modeled." Commenter 107 requested criteria on developing maximum site inventories, including the isotopes, maximum permissible inventory, and inventory limiting site characteristics. DOE and ASME emphasized the need for future regulatory guides to include quality assurance. Commenter 103 suggested 16 areas where additional guidance should be developed and covered a range of topics from siting to operational health physics.

<u>Analysis of Comments</u>: Agreed, additional regulatory guidance in a number of areas should be made available as soon as practical. Of the above, guidance on waste classification and waste stability is probably the most important. Guidance for license applications and site closure are already being addressed. Guidance on waste classification and waste form including stability have been drafted. Technical positions on site design and operation and financial assurances have also been drafted. A Technical position document has been published for site suitability selection, and characterization (NUREG-0902). Maximum site inventories and disposal of radium need to be considered in guidance being developed. Commenters addressed the issue of radionuclide inventories vs. concentration (Issue D-55-10), and disposal of radium (Issue D-55-4). Formal and separate guides to address these two issues are not needed. Guidance on determining whether the intruder scenario meets the performance objective is not necessary since compliance with the classification scheme provides the mechanism for meeting the objective. The other topics suggested will be considered as additional guidance is developed.

Suggested Rule Change: None.

## **ISSUE GEN-3**

Issue: Exempt waste in storage

Commenter: Union Carbide Corporation (39)

Rule Citation: Packaging and labeling requirements

<u>Summary of Comment</u>: Union Carbide suggested exempting wastes in storage prior to the effective data of the regulation from the packaging and labeling requirements.

<u>Analysis of Comment</u>: Union Carbide described its practice of storing packaged wastes in a shielded facility for up to one year to allow decay of short-lived nuclides and reduce exposures when wastes are shipped. Such reduction of exposures is in the spirit of ALARA. The commenter suggests that if these stored wastes must be repackaged, treated, or relabeled, the exposures involved would not be in the spirit of ALARA and these stored wastes should be exempted from new requirements.

While staff agrees that Union Carbide may have a valid point, the circumstances are very individual. Case-by-case exemptions can be used to provide relief if necessary. A short term detail on implementation is inappropriate for a rule.

Union Carbides point has broader implications, that is should certain parts of the rule be implemented in phases or steps. Application to existing sites was briefly addressed in Issue A-1. For the manifest system and waste Classes to work on a national scale, the 26 Agreement States must adopt conforming rules to make them regulatory requirements. Practical implementation can be achieved for data requirements and waste Class by amendments to the three existing site licenses. The effective dates for the rule will have to be established by working closely with State officials.

Rule Changes: Effective date(s) coordinated with States.

Issue: Development of new sites

<u>Commenters</u>: Joseph H. White III (21) Louise Gorenflo (71) Oswald U. Anders (73)

## Rule Citation: None

<u>Summary of Comments</u>: Joseph H. White III questioned the number and location of new sites expected and the time-line for establishing new sites. Louise Gorenflo was concerned about the lack of requirements for socio-economic impacts of new sites in the rule. Oswald U. Anders questioned whether the requirements in the rule would eliminate private enterprise developement of new sites and Gorenflo suggested that private enterprise not be allowed to develop new sites.

<u>Analysis of Comments</u>: The Commission does not know or control the number or location of new sites that may be proposed. Under the Low-Level Radioactive Waste Policy Act of 1980 (PL-96-573), Congress established a national policy that States are responsible for providing disposal capacity for wastes generated in their State with certain exceptions and that low-level wastes can be most safety and efficiently managed on a regional basis. The Act authorized States to enter into regional compacts to meet this responsibility. Currently, seven compact groups are in various stages of development. The National Governor's Association identified 6 potential regional breakdowns. In the DOE congressionally mandated response to the Policy act (DOE/NE-0015), 5-7 regional disposal sites were estimated to be able to handle wastes through the year 2000. Three sites are currently operating.

Timing of any new sites is also uncertain at this time. The Policy Act provides that compacts can exclude out of compact wastes in January 1986. Staff estimates of the time to license a site is up to two years after submittal of an application. Submittal would be preceeded by 2-3 years of site selection, site evaluation, data collection, and preparation of the application.

Dr. Anders expressed the view that the financial, procedural, and institutional requirements would, as a practical matter, eliminate persons in the private sector from developing new sites. He felt that the government would have to assume responsibility by default because of the regulatory burden. The proposed rule represented an attempt to provide adequate assurances to potential landowners and custodial agencies so that they would be willing to assume the role outlined for them. It attempted to define and clarify existing understandings about roles. The commercial firms presently operating sites did not share Dr. Anders' view in their comments. No comments were received from other potential commercial operators. Eliminating commercial development was certainly not the Commission's intent.

Louise Gorenflo expressed concern over the lack of land use and socio-economic considerations in the rule. The experience of her rural county that was considered for a new LLW facility was relayed. She indicated that certain rural land uses such as tourism and second home development are not compatible with LLW siting. Land adjacent to existing nuclear facilities was suggested. Land use and socio-economic factors are normally considered in the draft and final EISs. Since Part 61 does not address the EIS requirements but defers to 10 CFR Part 51, no changes to the rule were made based on the comment. The proposed procedures require an EIS and her concerns would be addressed as part of that effort.

Rule Changes: None.

## **ISSUE GEN-5**

Issue: De facto disposal sites

Commenter: Alfonso Scarpa (50)

Rule Citation: None.

<u>Summary of Comment:</u> The commenter was concerned that nuclear facilities become de facto disposal sites.

<u>Analysis of Comment</u>: Mr. Scarpa expressed concern that facilities where radioactive materials are used and stored become contaminated and cannot be totally decontaminated. The technology doesn't exit. The residual activity means that these sites are in fact disposal sites. He also made the point that when activity from clean up efforts are consolidated at a disposal site you then have two sites.

Mr. Scarpa offered no solution or suggested rule changes to address his concern. Limits for residual activities in decontamination and decommissioning are the subject of a separate policy development effort and are beyond the scope of proposed Part 61. Residual levels not requiring institutional control or licensing should result from this separate effort. Consolidating wastes that require institutional control to minimize the institutional burden seems to make sense and is the thrust of Part 61.

Rule Changes: None.

.

## **ISSUE GEN-6**

Issue: Need for GEIS

Commenter: Joel Jaffer (46)

Rule Citation: General

<u>Summary of Comment</u>: The commenter believed LLW disposal is hazardous enough to require preparation of a generic environmental impact statement (GEIS).

Analysis of Comment: A comprehensive draft impact statement (DEIS) (NUREG-0782, Volumes 1-4) was prepared in support of the proposed rule. The commenter did not indicate knowledge of or problems with this document. The draft should address the commenter's concerns that extensive analyses were part of the decision process for the rule. The DEIS is not a generic statement on low level wastes, however. It is a decision document for the rule. A final EIS for Part 61 has also been prepared.

## Rule Change: None.

sasti sad a s

## ISSUE GEN-7

<u>Issue</u> :	Extend the comment period	
<u>Commenters</u> :	U.S. Environmental Protection Agen PA Department of Environmental Res	cy (11) ources (16)
9 12 12 12 12 12 9 12 12 12 12 12 7 12 1	Dow Chemical (17) Kerr-McGee Corp (22) U.S. Department of Energy(28) Sierra Club (37)	
Maria Mar	Yakima Indians (74)	ineers (no number)

Rule Citation: Expiration date for comments

Contine of Summary of Comments: Proposed Part 61 was published for comment on July 24, 1981. The comment period was set to expire October 22, 1981. The draft environmental impact statement (NUREG-0782) was not announced until October 22, 1981. Additional time to prepare comments and review the DEIS was requested. One commenter requested consideration of late comments.

Analysis of Comments: The requests for additional time were reasonable and the comment period was extended until January 14, 1982 to coincide with the 90 day comment period on the DEIS (46 FR 51776 October 22, 1981). The Yakima request to submit late comments was acknowledged by letter dated January 25, 1982. Staff indicated that Yakima comments would be considered as fully as possible and if the rule has been finalized when comments are received, they will be considered in future modifications.

# ISSUE GEN-8

.

. . . î

## Issue: Data base on wastes

<u>Commenter</u>: Advisory Committee on Reactor Safeguards (ACRS) (10) Rule Citation: None.

Summary of Comment: The commenter calls for compilation of detailed inventories on the quantities and specific radionuclide concentrations in the low-level radioactive wastes buried in existing disposal sites. The commenter states that although this effort may require the development of instrumentation to identify and assess radionuclide concentrations in waste packages, such data are essential if the NRC staff is to have a clear understanding of current. practices. The commenter further believed that such information is essential if NRC staff are to be able to ascertain the impact of various regulatory actions , particularly the influence of the establishment of "de minimus" concentrations for selected radionuclides in specified types of wastes. Finally, the commenter felt that such information is also essential in order to assess the impact of various restrictions on the types of wastes acceptable for disposal in a given site.

Analysis of Comment: The comment is a straight forward recommendation for the NRC staff to improve their data base on waste characteristics. As noted by the commenter, NRC staff are already working to compile such an improved data For example, NRC staff have completed contracts to acquire propriatory base. disposal facility shipment manifests for the years 1978, 1979, and 1980. Shipment manifests for the years 1981 and 1982 will also be acquired. This analysis has indicated that significant percentages of low level wastes by volume and activity are not identifiable from the records as being generated by a particular waste generator." Such wastes are only identified by the shipment broker. This observation led in part to the requirement in § 20.311 that the original waste generators shipping waste to disposal facilities be specifically identified in the shipment manifests. Other analyses of 1978 and 1979 shipment manifests has led to estimates of the volume-percent distribution of gross activity concentrations in LWR process waste streams. Such estimates have been incorporated into the analysis for the Final EIS.

In addition to the above, NRC staff have ongoing a number of other contractual efforts to improve their data base on waste characteristics. These include the following efforts:

- a contract with Science Applications, Inc. to perform sampling and 0 radiochemical analysis of LWR power plant waste;
- an ongoing contract with Brookhaven National Laboratory to analyze 0 characteristics and potential improved packaging of some specific high activity fuel cycle and nonfuel cycle waste streams, and
- 0 a contract with Pacific Northwest Laboratory under NRC's Office of Research to more completely characterize wastes generated from decommissioning of LWR power plants.

Further information on waste characteristics is expected to be gained as part of further analysis of specific waste streams for possible disposal by less restrictive means (i.e., "de minimis" waste streams").

In summary, although NRC staff believe that the current data base on waste characteristics is sufficient at this time to arrive at regulatory decisions regarding near-surface waste disposal, improvements to this data base are warrented. NRC staff has an ongoing program to acquire such improvements. Some of this additional data has already been used in finalization of the draft Part 61 rule. Additional data acquired will be used to determine potential additional technical requirements, develop additional regulatory guidance, and develop improved procedures to analyze waste disposal and license new and existing disposal facilities.

Suggested Rule Change: None.

## **ISSUE GEN-9**

Issue:

Hearing transcripts to incorporate for the record

Themis Klotz (42) Commenter:

Rule Citation: None. <u>Summary of Comment</u>: The commenter expressed general concern about nuclear matters and requested that the record of two hearings be incorporated into the record.

<u>Analysis of Comment</u>: The hearings referenced by the commenter were 1) Oceanography Subcommittee, Merchant Marine and Fisheries Committee, U.S. House of Representatives, held in Faneuil Hall, Boston, Massachusetts, September 21, 1981 and 2) Falmouth, Massachusetts Board of Health, held at the Town Hall on October 15, 1981.

The Oceanography Subcommittee hearings had not been printed as of February 25, 1982. Subcommittee staff indicated on February 25, that the hearings were at the printers and should be available in about a month. The transcript is available for inspection in House Annex 2, Room 550. The hearings included past ocean disposal of radioactive wastes in Boston Harbor and beyond. Ocean disposal is beyond the scope of proposed Part 61 and permit authority lies with the Environmental Protection Agency. However, the request to place the record of the hearing into the record without comment is noted.

The Falmouth Board of Health hearings referenced did not address radioactive wastes according to Dr. Jones who was chairman of the Board when these hearing were held. The issue was related to hazardous waste and was prompted by an incident involving pesticide spraying of a highway right-of-way. The relevancy of these hearings is unclear but the commenter's request is noted.

.....

## ISSUE ED-1

Issue:

## Editorial Comments

Commenters:

State of Nevada (14) Pennsylvania (16) Dow Chemical (17) Department of the Enviornment, London (19) Joseph H. White III (21) Law Engineering (34) . Union of Concerned Scientist (36) Union Carbide (39) Bechtel National, Inc. (44) Arizona State Clearinghouse (47) Union Oil Company of California (66) Argonne National Laboratory (68) South Dakota (69) Georgia Institute of Technology (70) The University of Texas Medical Branch (75) Utility Nuclear Waste Management Group (81) Northeast Utilities (85) American Nuclear Society (87) General Electric (89) Arkansas Power and Light Company (94) Health Physics Society (96) Atomic Industrial Forum (100) U.S. Ecology (101) Conference of Radiation Control Program Directors (103) Carolina Power and Light Company (106) The American Society of Mechanical Engineers (107) and (113) NC Department of Human Resources (119) New England Nuclear (NELRAD) (110) The American Society of Mechanical Engineers (113) Kerr-McGee (115) Tennessee Valley Authority (116) Texas Department of Health (117) Atomic Energy of Canada Limited (118) Department of Energy (119) U.S. Environmental Protection Agency (122)

<u>Summary of Comments</u>: The commenters made a variety of recommendations for clarifying language, definitions, etc. and identified typographical and numbering errors.

<u>Analysis of Comments</u>: The comments were each considered and adopted where consistent with the intent of the rule.

al 30, 17 H HOMESTAKE MINING GOMPANY 51. 2 450 CALIFORNIA STREET - 314 FLOOR SUCKET BURESEE DD SAN FRANCISCO, CALIFORNIA S4108 -2.19 20. PROPOSED SULE CIT -----In 38081) ൳ 73-170- (46 FR 70. \*ELEPHANE (415) \$81-8180 (T COLER MINIS PROPOSED RULE PR-2, 19, 20, 21 30 July 30, 1981 40,51,61,1 202-0519 USI:20 70, 73,170 a (46 FR 38081) AUG 0 8 1981 Desterne & Ser. Secretary of the Commission U. S. Nuclear Regulatory Commission . . Washington, D.C. 20535 D-5618 Attn: Docketing and Service Branch Gentlemen: The proposed rule published by the Nuclear Regulatory Commission in the Federal Register on July 24, 1981, adding a new Part 61 to 10 CFR, should be revised to more clearly exclude uranium mill tailings and other wasts generated in the nuclear fuel cycle from the definition of "wasts" found at Section 51.2. A – in the Thank you very much. d'un 12 h Since John McMunn Assistant Counsel JSM:hh 110 Conteres 158 AUB 1 2 191 בינה על בכבסשיינסשיים 3118160349 310730 2 46FR39081 PDR PDR 8108140321 810730 PDR PR 2 46FR38081 PDF PDR

. . . .

MARVIN 1 LEWIS 6504 BRADFORD TER PHILA. PA 19149 ECRETARY OF THE COMMISSIO 3 STATE OF NEW MEXICO YRC GOVERNOR'S CABINET SHINGTON, D.C. 20555 SANTA FE PHINULED RULE FR 21, 30 40 DEAR MR SECRETARY; SITUST 51,61, TO, T3, 170 (46R) 10 FLEAS: ACCEPT THE FOLLOWING COMMENTS ON THE PROPOSED DEAR MR SECRETARY: GLORGE S. GOLDSTEIN P.D. SECRETARY NOS VENETING & DAVISONNE RULE 10 CFR GI," LICENSING ROH'TS FOR LAND PISPOSAL 19111 OF RAD WASTE." MY FIRST COMMENT AND CONCERN CENTERS ON AN September 11, 1981 GAMAN OMISSION: THE PROBLEM OF PRESENT, DANGERLUS LUW LEVEL WASTES SITES ARE NOT MENTIONED IN THE PROPOSED RULE. HEST VALLEY L.L.W. SITE IS LEAKING INTO A SAND LEN'S AND EVENTUALLY INTO THE GREAT LAKES Mr. Samuel J. Chilk Secretary of the Commission IWASTE PARER, NEWSLETTER OF THE SIERRA CLUB). U.S. Ruclear Regulatory Commission OCE is ELEANING UP ABOUT 20 ABANDUNED DUMP Aashington, D.C. 20555 SITES. THREE MILE ISLAND IS A DE FACTO' HIGH LEVEL AND PROPOSED RULE FR-2, 19, 20 Attention: Cocketing and Service Branch LOW LEVEL WASTE SITE, 21,30,40 Bear Mr. Chilk: THE PROPOSED RULE ISNORES ALL PREVIOUS AND EXISTING +170 PROBLEMS WITH EXISTING LL WASTE SITES. The following are recommendations and comments from the State of New AT A VERY MINIMUM, THE PROPOSED RULE MUST DISCUSS WHAT HEASINGS Mexico regarding the proposed rulemaking for 10 CFR of for land disposal of radioactive wastes which was published in the Federal Redister on LE PROPOSED TO SOLVE THE PROBLEMS July 24, 1981. 14 LLWASTE SITES. 23ge Colum Line Recommendations and Comments PAGE 10. PROTECTION OF MADVESTENT INTRUDER. 38083 27 1. Suo mrom seems a lot of exposure for 2 Recommend wording read "nearest public or private existing or future potential grinking a small misTARE. water supply." 2. HULL CAN 500 MERM CLAUSLE be unsuged PAGE ON-SITE SECURITY? DETAILS! PROTECTION CE THE ENVIRONMENT 38084 25 Recommend deletion of present text after "Surface Water" and substitution of the D-50-3 following: "The potential for flocating THERE 15 NO ADSOLLTE AMOUNT OR FERCENTATE should be low. The statement at 01.00 LEARAGE BEYOND WITCH. THE SITE WOULD BE UNSAFE. (a)(b) is considered acequate. THERE C-3 LEVERY NUMBER IS AN EXPOSURE OR DILUTION, 38087 3 58 The summary of the procosed rule states FORE, THE ENTIRE CONTENTS COULD LEAK OUT SLOWLY "it would be the Commission's intent that all future disposal would be expected to INTO THE ENVIRONMENT WITHOUT BRENKING THE comply with the provisions of Part 61.4 The Commission should recognize that in CAVEATS OF THIS PROPOSED RULE. A-1 FASE 15 Agreement States reculations must be pro-"ICO YEAR PERIOD OF INSTITUTIONAL CONTROL "ICO YEAR - MAXIMUM" mulgated by local authorities. 19 7-59-2 13023 12 6.1.1(b) states "...the regulations in this HEAT 13 Minimum time that institutiona - CONTROPS part apply to all persons in the United States." 15 A SURETY? An additional clause should be added to this "GE 25 "FINANCIAL SECURITY" - NOT ELOUGH DETHILADD" sentence to read "...States where the Commission vetains authority." E-1 TO BE SURE OF ANYTHING. THIS PROPUSED RULE IS WORTHLESS AND MUST BE REDONE 8108190194 810608 PDR PR 2 46FR38081 PDF Very truly yours, 2109150125 310911 208 29 2 46FR35081 PDF PDR Acknowledged by care. 81:4191. mdi

Advance Medical & Research CENTER, INC. PONTIAC, MICHIGAN 48057 1270 DORIS ROAD 313-373-9120 Mr. Samuel J. Chilk September 11, 1981 Page -2-Line Recommendations and Comments Page Column SITIE! State I A . La tite of the 30.40 38098 33 Recommend the word "eliminated" be replaced by 2 FROMOSED RULE "minimized." Elimination of engoing active GENmaintenance may be the goal but it probably September 15, 1981 will not be met. This fact is recognized in 61.63(a) by the words "any recuired maintenance." Recommend the word "may" be replaced by "shall." 47 Secretary of the Commission. This function should be an obligation of the U.S. Huclear Regulatory Commission <sup>خړي</sup> کې Ofrector. Washington, D.C. 20555 Recommend deletion of words "if any," licensees 38100 15 Attention: Docketing and Service Branch should be required to furnish an annual financial report. To Whom It Hay Concern: Ser the opportunity for comment on the proposed rule. Thank you for providia This letter is in regard to proposed regulation in land disposal of low level radioactive-waste. Sincere' As per Section 61.54, Table I states that with any isotope with a half-life of less than 5 years, the minimum requirements for all classes of waste must be followed as per Section 61.56. Goldstein, Ph.D. George S. Secretary Most medical diagnostic laboratories perform in vitro testing in which the isotope used is of short half-life of about 60 days ( $I^{125}$ ). These isotopes GSG/teb are packaged in kit form of about 10 uCl/100 tube kits. Instead of stating that the instructions in Section 61.56 be followed, can't there be an alternative measure? As an alternate with isotopes of short half-life, make a category whereby they are set aside and allowed to decay to the background level. The waste is then monitored with a Survey Meter and if no counts above background, then dispose of this waste with the usual solid waste. If all solid radioactive waste is to be disposed of as per Section 61.56, there would be tremendous volume of waste that would fill the disposal site rapidly and needlessly. Much of the I125 isotope waste will decay after several half-lives down to acceptable level and this isotope along with FE59, Co57 and I131 and some other isotopes of short half-lives used in a tlinical setting should be given another classification. 1. 2. 2 Sincerely John Pappas Associate Director JP/dg 8109220269 PDR PR 2 46FR38081 810913 PCR . . B-187

## THE PROCTER & GAMBLE COMPANY

WOITDALE TECHNICAL CENTER WOITDALE TECHNICAL CENTER CONTACT INFORMATION OF ASSIGN September 14, 1981 CONTACT INFORMATION OF ASSIGN September 14, 1981 CONTACT INFORMATION OF ASSIGN September 14, 1981 CONTACT INFORMATION OF ASSIGN CONTACT IN

RE: PROPOSED PERH ATIONS: LICENSING REQUIREMENTS FOR LAND DISPOSAL OF RADIOACTIVE WASTE

#### Dear Sir:

M-4

This is to provide the comments of the Procter & Gamble Company upon the proposed regulations covering specific licensing requirements for land disposal of radioactive waste published in the <u>Federal Protster</u> on Friday, July 24, 1981, pages 38081-18104. While Procter & Gamble is not directly involved in such disposal operations, we do utilize small amounts of radioactive materials for research purposes requiring eventual disposal. The subject proposal does contain some provisions that pertain to waste generators, even those involving only small amounts of research materials. Thus, Procter & Gamble is potentially affected.

In general, we agree with the Commission's proposal and waste classification system. Our principal concern with the proposal is the manifest system. The problem is the Commission has failed to carry through its waste classification system into the manifest system. As a result, the manifest system is appropriate and useful for Class C wastes but is too burdensome and of little value for Class A wastes. Certainly the timing and reporting specifies are overkill for the small shipments of Class A wastes that commonly originate from research laboratories and medical facilities. Such overrequirements are in direct conflict with the Government's stated goal of eliminating/avoiding unnecessary regulatory burdens.

We would urge that Glass A wastes be exempted completely from the manifest system of these proposed regulations or at least the overly rigid timing and reporting aspects of this manifest system. In the remainder of these comments, we address specific concerns of the proposal and provide revised wording. THE PROCTER & GAMBLE COMPANY

Nuclear Regulatory Commission Page 2 September 14, 1981

### Specific Comments and Recommendations

#### 20.311(5)

The term "chemical form" is not clear. It would not always be possible or useful to specify an exact chemical species (for example with research wastes from a biological research or medical facility) or to list the many species from multiple experiments. Further, this requirement could represent a serious security problem if the exact identity of research chemicals had to be listed on a public manifest. It should be sufficient for waste disposal purposes to list the <u>general</u> chemical form (i.e., complex organic acid, animal carcass). We recommend the word "general" be inserted before the words "chemical form" at the end of the second sentence of this paragraph.

## 29.311(2)(3)

This wording is not entirely clear. It could be interpreted to require that a personal audit be conducted by canagement as a part of a quality assurance program to assure compliance with \$\$1.55 and \$1.56. We believe it is intended to require (and it is certainly sufficient to require) that audits be done and results be reported to management. Our specific proposed rewording is:

Conduct a quality assurance program to assure compliance with §§61.55 and 61.56 of this chapter; audit results must be reported to management.

#### 20.311(e)(8), 20.311(f)(1), 20.311(h)(1), 20.311(h)(2)

There is a serious overreporting system described by these portions for Clans A wastes. The difficulty is that these are written to cover more serious concerns associated with shipments of Class C wastes and yet are applied with equal rigor to Class A shipments. In particular the one week acknowledgment requirement for receiver of shipments and the requirements for the shippers to investigate and file reports on all shipments missing for 20 days is overreaction with Class A shipments. This overburden will be laid principally on research facilities, hospitals and academia.

We urge that shipments qualifying as Class A wastes be exempted from the acknowledgment, tracing and reporting timing requirements and that it simply be stated that these activities need to be done and documented. Certainly reporting to the Commission Inspection and Enforcement Regional Office should not be required, but the records should simply be maintained by the snipper and be available for the normal NRC inspection.

We recommend that the following wording be adopted:

M-1-

8

D-55-7

1/0 2210

<u>ئ</u>ھ،

B-189

2.

.

THE PEOCTER & GAMBLE COMPANY

Muclear Regulatory Commission Page 3 September 14, 1981

1	20.311(e)(1)	Acknowledge receipt of the waste from the generator; within one week for Class B and C wastes.
M-14	20.311(f)(1)	Acknowledge receipt of the waste from the generator; within one week for Class B and C wastes.
	20.311(h)(1)	Be investigated by the shipper if the shipper has not received notification of receipt; within 20 days after transfer for Class B and C wastes.
	20.311(h)(2)	Be traced and for Class B and C wastes be reported. The investigation shall include tracing the shippent and for Class B and C wastes filing a report with the nearest Commission Inspection and Enforcement Regional Office listed in Appendix D of this part. Each licensee who conducts a trace investigation of a Class B or C waste shall file a written report with the nearest Commission's Regional Office within 2 weeks of the completion of this investigation.

Sumary

We urge the Commission to exempt Class A wastes from the proposed manifest system or at least the timing and reporting aspects of the system so that it is rore appropriate for the small risks and amounts of research wastes involved and avoids unnecessary regulatory paperwork and reporting.

Very truly yours.

Dr. Donald T. Hooker, Associate Director Research & Development Regulatory Services Division

:bal

Bethlehem Steel Corporation

BETHLEHEM, PA 10016

4. E. angelerer, Ja. Se B.		September 14,	1981
Secretary of t	ne Commission Regulatory Commission	Ð	ALLELL
Washington, D	C 20555 PROFOSED RUL	PR-2, 19, 20, 21, 30, 1	to, IT SEP 18 .
Artention:	Docketing and Service Branch	(46 FR 38 D81	
Gentlemen:			

ŝ

. .

73

1

Res Proposed Rulemaking on Land Disposal of Low Level Radioactive Waste, 10 CFR Part 61, published in the Federal Register, Vol. 46, No. 142, Friday, July 24, 1981. pares 38681-38105

Bethlehem Steel Corporation would like clarification on certain points of the proposo rule.

> 1. Please indicate the conditions required to permit sealed sources containing relatively long lived hard gamma emitters such as cobatt 60 or cesium 137, to be classified as Class A - Segregated Waste or Class B - Stable Waste. The language in the "Supplementary information" accompanying the proposed rulemaking seems to imply that sealed sources would have to be classified as Class C - intruder Waste.

The notes below Table 1, on page 38097, appear to indicate that by enclosing a 200 millicurie, cesium 137 sealed source, having a capsule volume of 6 cm<sup>3</sup>, in a 55 gallon drum, it is permissible to average the 200 millicuries over the 55 gallons to obtain a concentration of:

200.000 microcuries = 1.0 uCi/cm<sup>3</sup> 200.000 cm<sup>3</sup>

which means the limits for Class A - Segregated Waste; as opposed to

which exceeds even the limits for Class C - intruder Waste and would be prohibited in a low level waste disposal site. Is our interpretation of this point correct?

· · · · · . . . . .





- 2 -

If our interpretation as given in item (2) is correct and sealed sources can be classified as Class A or Class B, we feel that the proposed rule is reasonable. We would appreciate your comments and clarification in that regard. D-55-7

Sincerely yours,

a.E. moth A. E. Mottift, Jr., Sc.D. Manager of Environmental Health

cc: R.Dale Smith, Chief Low Level Waste Licensing Branch Division of Waste Management U. S. Nuclear Regulatory Commission Washington, DC 20555

UCLA UNIVERSITY OF CALIFORNIA, LOS ANGELES ERCELET + DAVIS + UNINE + LOS ANGELES + ANVERSIDE + SAN DIECO + SAN FRANCISCO SANTA BARBARA + SANTA CRUZ

COMMUNITY SAFETY DEPARTMENT OFFICE OF RESEARCH & OCCUPATIONAL SAFETY LOS ANGELES, CALIFORNIA 8034

DOCAST RUNGER 25070550 RULE 11-2,19,2021 30,40,51, 73,170 6 FR 38081)

8

Secretary of the Commission U.S. Nuclear Regulatory Commission Washington, D.C. 20555 Attention: Docketing and Service Branch

Dear Strs:

Re: Proposed Rulemaking on Land Disposal of Low-Level Radioactive Waste

In general, we think that the proposed amendments to the various parts of Title 10 Code of Feceral Regulations dealing with land discosal of radioactive waste are beneficial and desirable. However, we want to comment specifically on several proposed changes as follows:

Part 20.311 (5): Compliance with the requirement that the shigment manifest contain the cnemical form of the waste is virtually impossible for research and hospital wastes. In some cases the chemical form may literally be unknown and in all cases the variety of cnemical compounds that could be in the waste from one laboratory will be very M-44 large.

> Pernaps this requirement could be rewritten to apply only to pure material present in quantities greater than a specified minimum. That is, containers of known compounds exceeding some specified quantity (say 500 gms.) would be identified, but wasta contaminated with some chemical would not be identified.

(Part 61.50 (a) (9): We appreciate the concern for avoiding areas where tectonic processes (faulting, etc.) could affect the site integrity. However, the present wording would probably make it difficult or impossible to locate a site in California, D-50-5 yet realistically there is no concelling scientific reason not to bury low level. D-50-5 radisactive waste - even where some faulting has or may occur. In view of all of the other restrictions governing site selection and requirements on waste packaging, it seems highly unlikely that even major ground faulting would release radioactive material to the occupied human environment. · · ·

M-1

Secretary of the U.S. Nuclear Regulatory Commission September 17, 1981 Page 2

We hope that this section can be reworded to minimize the restriction on the D-50-5 (existence of faulting and seismic activity in the area of the disposal site.

Part 61.56 (2): <u>We think that prohibiting the use of cardboard or fiberboard boxes</u> for low level radioactive waste is unnecessary and may be short-sighted. It seems to us that one of the most promising methods for disposal of this type of waste is incineration. <u>Cardboard boxes could be introduced directly into an incinerator</u>. D-56-4 metal drums would have to be unpacked.

Perhaps this section could be modified by placing certain limits on the total quantity of activity that can be placed in a cardboard box - rather than totally prohibiting such containers.

Part 61.56 (b) (1): This section would also prohibit cardboard boxes and cossibly some metal drums. Again for low level waste we feel that this structural stability requirement is unnecessary. The size itself, if selected according to the proposed criteria, should provide adequate containment for low level waste. Structural stability of the primary container is relatively insignificant - particularly over a time span of several hundred years.

Incidentally, incineration is again at attractive alternative as it processes the waste into a more stable form (asnes).

' Thank you for your consideration of these comments.

Very Truly Yours,

Dir Res Cer

Walter F. Wegst Director Research and Occupational Safety Certified Health Physicist

#### WFW/ml

D-56-8

## Tel: 919/962-3780, 968-9534

Comments on proposed rules for Land Disposal of Low-Level Waste, 46 P.R. 38081 (1981).

The Environmental Law Project is concerned about the proposed regulations for several reasons. These are:

(1) pre-emption of state laws, particularly with regard to siting:

(2) possibility of transfer of license with inadequate safeguards;

(3) the fact that hearings must be requested, and are otherwise not required; and

(4) the limited time that licensees are required to monitor sites after closure.

. . . .

. .

1. Presention of state law: As we read the Atomic Emergy Act, rarticularly sections 42 U.S.C.A. \$2021(c) and (1). it is our understanding that the Commission still reserves the power to approve and license a disposal facility without and even in direct contradiction of, the benefit of state law. See Pacific Legal Foundation v. State Energy Resources Conservation and Development Commission, 472 F.Supp. 191, 199 (D.C. S.Cal. 1979). which reaffirms the general pre-emptive powers of the Commission. Especially in light of P.L. 96-573, which makes it the responsibility of the states to provide disposal capacity, it is our contention that some form of final approval must be vested with a local, preferably state, government. Specific language to this effect in the regulations would resolve this latent ambiguity in favor of more responsibility for those governments which Congress intended to have primary responsibility for locating disposal sites.

2. <u>Possibility of transfer</u>: The license of an operating disposal site can apparently be transferred under the proposed regulations; although we would presume that the same requirements for

A-5

B-B

#### Page 2

financial security, etc., would be necessarily zet before the Commission gave its consent, it would be much more comforting to have it spelled out in black and white in the final rule. That is, that transferees will be subject to all licensing restrictions that original licensees are subject to, which must be shown prior to granting approval of the transfer. This would also be useful in generating evidence of record in case of future problems with the new operator.

3. Hearings: Hearings may be requested by interested parties subject to the procedures in 10 C.F.R. \$2. 105. Our experience with this procedure, and the nature of the activity contemplated under this rule, make it imperative that public hearings be a routine part of the licensing process. The hearings should be conducted where the site is proposed, not in Washington or anywhere else that might be at the Commission's or applicant's convenience. Therefore, the rule should be amended to require putlic hearings, as well as the normal adversary type hearings, at or suitably near the proposed site. Applicant or the Commission would be required to publicize these hearings at least two months in advance, to give residents an opportunity to make satisfactory investigations of their own. In addition, it might also be helpful to establish public document rocus near waste facilities similar to those now required near nuclear power plants.

R-9

B-10'

4. Time limitations: We think the five-year active monitoring phase required by the proposed rule 361,29 is far too short. Experience with other low-level sites has shown that migration has often gone undetected for long periods of time, and can occur in totally unexpected and unforeseen ways. In addition, past experience with licensee monitoring around power plants leads us to expect that monitoring efforts will be no more than the minimum required by the Commission, and that consequently detection of problems will be unlikely. At least ten years of some form of active monitoring seems to be required, and perhaps more using the experience gained at Maxey Flats. Some

specific provision for additional monitoring if determined necessary by geologic or other conditions would be helpful.

Fage 3

'5. Other: One additional weakness we noted is that licensee does not apparently need to submit a detailed closure plan at the time of application, but that this will only be required as closure approaches. We heartily agree with the chilosophy 7)-52-6 that site operation should be aized at all times toward closure. but would like to see a stricter requirement on applicants at the time of application to have detailed closure plans, subject-to amendment with Commission approval when so required by developments during operation, prepared when applying. Section \$61.53(d) requires the licenses to have plans ready in case of Digration: is this a realistic requirement? Since the probable problem would be groundwater contamination, we think it unlikely that licensees will be able to do anything about it. Specific language that licensees would be required to indemnify well users in case of migration would thus be preferable. Section \$61.31 refers to the duty placed on licensee to E - Iperform "minor custodial activities" after closure. What about major activities? Again, specific language placing financial responsibility for these on the licensee would be useful in removing potential ambiguity. In addition, it appears that by virtue of the omission of major duties (such as cleanup) and the Commission's discretionary power to apply these rules retroactively to existing sites, the stage could possibly be set for a taxpayer-funded bailout of poorly-run existing and inoperative sites, like Maxey Flats and West Valley. We feel that the industry should bear these costs. and the regulations should be written to make sure that this is done. - Finally, an interesting philosophical question is raised

by the statement at 46 F.R. 38085 that one hundred years is the D-59.7 ( longest that "governmental institutions should be relied on to carry out active controls." Given the problems to date of our (inability to solve the high-level waste dilemma, and the fact that even the most optimistic predictions estimate that high-

. . .

Fage 4

level waste (and other low-level waste containing isotopes such as Fu-239 and U-238) can remain toxic for up to 600 years. is this not an admission that we are creating a problem for D-59-2 future generations that we don't think they'll be able to solve, and that we ourselves can barely handle? As a catter of philosophic good sense and good moral conduct, this statement would seem to require the oscisation of any production of long-lived redioactive waste. 

Comments prepared by Daniel F. Read.

÷ . · ...1

× .-

1.191 UNITED STATES NUCLEAR REGULATORY COMMISSION ADVISORY COMMITTEE ON REACTOR SAFEGUARD WASHINGTON, D. C. 20555 September 16, 1981 SHOFTSED RUS Honorable Nunzio J. Palladino Chairman U.S. Nuclear Regulatory Commission Washington, DC 20555 See. 1. 1. 1. 1 SUBJECT: REPORT ON PROPOSED RULE ON "LICENSING REQUIREMENTS FOR LAND DISPOSAL OF RADIOACTIVE WASTE • . Dear Dr. Palladino: During its 257th meeting, September 10-12, 1981, the Advisory Committee on Reactor Safeguards met with the NRC Staff and representatives of the U.S. Department of Energy and the Commonwealth of Kentucky to discuss the proposed rule on "Licensing Requirements for Land Disposal of Radioactive Waste" (10 CFR 61). This was also the subject of a meeting of the ACRS Subcommittee on Waste Management held in Washington, D.C. on September 3, 1981. 1. General Comments On the basis of this review, we offer the following general comments: a. Adequacy of Proposed Rule The proposed rule contains criteria which should assure improvement in the siting, design and operation of near-surface radioactive waste disposal facilities. ' Concurrent with this effort should be a continuation of work to seek better containment, stabilization, and immobilization of low-level wastes as well as completion of the establishment of criteria for deeper land bur-tal and disposal in mined carities. Also to be explored is the A-8 possible disposal of such wastes in the sea. In addition, the processes that result in the production of the wastes need to be addressed. Attention should be directed to techniques both for reducing the volumes of wastes generated and for assuring that The wastes that are produced are in, or can be converted to, a form amenable to safe disposal. D-56-21b. Applicability to Existing Disposal Facilities

.::

đ.

.:1 يتشن

11

7

The rule states that many of the operational characteristics requirements proposed are in effect at currently licensed low-level radioactive waste disposal facilities and that such facilities TOACTIVE WASTE GISposal Factificies and Line Seal Control

and the second second

Honorable Nunzio J. Palladino

September 16, 1981

A-1 should have no difficulty in complying with the suggested requirements. The requirements for sites where operations have been terminated because of disposal problems are not covered by the proposed rule. The NRC Staff has stated that methods for decommissioning the terminated facilities are to be enumerated in regulatory guides issued in support of the proposed rule. It should be recognized, however, that development of satisfactory guidance for such actions may be difficult. Several of these sites contain wastes that include plutonium and other long-lived radionuclides which appear to be in contact with water.

- 2 -

## C. Types and Quantities of Wastes Subject to Disposal

Development of the proposed rule has revealed certain deficiencies in data, particularly with respect to the compilation of detailed inventories on the quantities and specific radionuclide concentrations in the low-level radioactive wastes buried in existing dis-

- GEN-B posal sites. The NRC Staff is currently compiling such data. Although this effort may require the development of instrumentation to identify and assess radionuclide concentrations in waste packages, such data are essential if the NRC Staff is to have a clear understanding of current practices and if they are to be able to ascertain the impact of various regulatory actions, particularly the influence of the establishment of "de minimus" concentrations for selected radionuclides in specified types of wastes. Such information is also essential in order to assess the impact of various restrictions on the types of wastes acceptable for disposal in a given site.
  - 2. Specific Comments

In terms of comments on specific topics within the proposed rule, we offer the following:

#### a. Time Spans for Various Requirements

D-50-9There appears to be a lack of clarity within the proposed rule relative to the time spans over which the various design requirements are to apply. It would be helpful to include a clear statement that restrictions, such as those pertaining to floods, erosion, and water drainage, are to apply through the time of site closure as well as the period of institutional control. In contrast to this, there would appear to be little necessity for specifying a need to observe long-term tectonic changes potentially affecting the site. Honorable Nunzio J. Palladino

September 16, 1981

#### b. Avoidance of Soil Subsidence

D-52-2 The proposed rule implies that the waste form plays a major role in soil subsidence that has frequently been observed in land disposal facilities in the past. Since experience shows that subsidence results from a variety of factors, including primarily the manner in which the waste packages are placed in the ground, this portion

In which the waste packages are placed in the ground, this portion of the proposed rule may need to be reevaluated and revised.

- 3 -

#### Restrictions on Types of Wastes

 $\mathcal{D}$ -56-3 The proposed rule contains a number of limitations on the types of wastes that may be disposed of in a land facility. These include restrictions on pyrophorics, explosives, wastes that generate

D-56-17 toxic gases, etc. There are also requirements on the "stability" of the wastes, without a clear definition as to what this means, and on the minimum compressive strength for the wastes, which could unduly increase the volumes to be buried. We believe that these restrictions and requirements need to be carefully assessed, both from the standpoint of their enforceability and overall implications.

#### d. Disposal of Chelating Agents

The proposed rule would not allow the disposal in surface land facilities of wastes containing greater than 0.1% chelating agents. Since such agents are present in a wide variety of radioactive

D-56-2 wastes (for example, in decontaminating solutions), this provision could exclude many wastes from burial. In our discussions, the NRC Staff indicated that their intent is not to exclude such wastes from burial but to make the disposal of such wastes subject to their approval. The fact that consideration of the disposal of such wastes may have to be handled on a case-by-case basis needs to be emphasized.

A-8 A-8 A-8 A-8 A-8 A-8 A-8 In closing the Committee would like to note that, at the present time, there is a not a solution to the problem to the problem that the same time as a solution of such additional that the same time as pointed out above, it is important that attention be directed to means for reducing the amounts of such additional artention be directed to means for reducing the amounts of such additional for the disposal of the the same time is a pointed out above, it is important that attention be directed to means for reducing the amounts of such wastes are produced, and assuring that those that are produced are in a form compatible with disposal. It is also important to vigorously pursue alternatives to shallow land burial as a means for the disposal of such materials, particularly in view of the problems that have been encountered with land burial sites located in areas of the U.S. with high precipitation.

Sincerely,

J. Carson Mark Chairman

U.S. DEPARTMENT OF COMMERCE UNITED STATES ENVIRONMENTAL PROTECTION AGENCY U.S. DEPARTMENT OF COMMERCE National Oceanic and Atmospheric Administration Environmental Research LABORATCRIES Boulder, Colorade 80303 WASHINGTON, D.C. 20460 September 14, 198 011 CONSTER SEP 1 1 1981 USVED PLOYOSED ROLE I II TO: R. Dale Smith, Chief SFP 23 Low-Level Waste Licensing Branch Division of Waste Management U. S. Nuclear Regulatory Commission Mr. R. Dale Smith Division of Waste Management Washington, D. C. 20555 Office of Nuclear Materials Safety and Safeguards FROM: William Takaraki, Chief Supply Services Division U.S. Nuclear Regulatory Commission . . Washington, D.C. 20555 SUBJECT: Proposed Rulemaking on Land Disposal of Low-Level Radioactive Dear Mr. Smith: Waste . The U.S. Environmental Protection Agency (EPA) is presently reviewing proposed regulations for the Licensing Requirements for Land Disposal of Radioactive Waste. These proposed Reference your letter of 7 August, 1981, subject as above. This Agency uses several extremely low-level radioactive sources that rules (10 CFR Parts 2, 19, 20, 21, 30, 40, 51, 61, 70, 73 are all self-contained sealed units. Disposal or turn-in of these units and 170) were published for public comment on July 24, 1981. are handled through agreement with the University of Colorado located with comments due to you October 22, 1981 (46 FR 38081). adjacent to our facilities. This Agency does not use any liquid radioisotope materials that could apply to subject rulemaking at this time. The Federal Register notice referenced an Environmental Impact Statement (EIS) which still has not been distributed. We have been told that this EIS will be more than 1,000 Based on the above, no comments, suggestions or recommendations are ------GEN-7 pages in length. Accordingly, we would like to request that NRC consider formally changing the due date for comments on . cut submitted relative to the proposed rulemaking. 1000 sh. the rule to coincide with the date that comments will be due on the draft EIS. This would allow both documents to be reviewed simultaneously. should you have any questions concerning our request, please CC: J. A. Kemper contact Dr. W. Alexander Williams of my staff (755-0790). Jack Cooper Thank you for your consideration of our request. 1510 2 Sincerely yours, DSID 1/0 ./o William N. Hedeman. .77 Director Office of Federal Activities 50348 810911 0267 310914 PD9 B-195



. •

B-196

PAGE THREE 100 FR61 CUIGG

COMMENTS ; REGARDING TRANURANIC WASTE RULES IN LOCFR- PART 61.

Why didn't the NRC put "less than 10mCi/g in columns 1 and .... 2 on page 38085. Desn't it apply?

D-55-3 Why didn't the NRE say that TRU waste above 10 nG1/g should be given retrievable surface storage until its new regulations for intermediate waste are available? The NEC's R. Dale Smith says promulgation of these regulations is next on that agency's Lagenda.

3. (IOCFR Part 61 indicates the NRC will decide TRU waste disposal D-55-5' on a case by case basis, giving it shallow land burial if the NRC deems the site suitable. Who in the NRC will make that decision and what are the criteria that will be used to judge whether or not a particular site is suitable;

4. The NEC's Smith told me high level waste will now include TRU waste down to 10 n21/g. Isn't this misleading when a 55 gallon drum of radioactive waste containing 2-million nanocures of TRU or 70-million curies of plutonium-211 will be allowed burial D-55-3 in a low level waste disposal site under the proposed regula tions?

5. In Section 61.55 (d), there is no definition of "near surface" --does this mean that if the dump operator digs deep enough, then land disposal at that site will be approved? 7 . . . . .

6. (what is the maximum limit on transuranic concentration the NAS D-55-5 will allow for land disposal? The rule gives no indication, other than to suggest that above 10 nG1/g, specific Commission approval will be required.

7. Averaging the concentration of radioisotopes over the volume of a waste container may allow heavy actual comentrations or "hot spots" of radioactive waste to be accepted for shallow land burial.

D-55-7 In very small type, the NRC states: "For a 55 gallon drum, multiply the (allowed) concentration by 200,000. Thus, by adding concrete, one should be able to bury just about anything -- as long as you add enough concrete. The multiplication factor of 200,000 will allow 2 million nanocuries per gram of TRU or 70 million nanocuries of plutonium-211 in a 55 gallon drum of Cradioactive waste.

3. / Why 10 nCi/g? We should examine the origin of the 10 nCi/g standard for transuranics. Pro-nuclear, scientists were, quick to tell me at a recent IPA meeting on low level wastes that "longi/g D-55-3 is the same as natural background -- it's like the soil outside

10CPR61

7-55-5

COICO

this building." They are wrong. According to the Atomic Snergy Commission (AEC), Appendix O511, Part 1 on redicactive waste . • \*

"The value of 10 mi/g is derived from the upper range of comentrations of radium-226 in the earth and is subject to modification based on long-term studies of muclide migration in the soil."

Experience at Maxey Flats, Idaho National Energy Lab, and Oak Ridge National Lab indicate that plutonium migrates much more quickly through the soil than originally anticipated. (For details of the Maxey Flats esperiences, see <u>Science magazine</u>, June 26, 1981, "Characterization of Plutonium in Maxey Flats"). One would expect, based on these experiences, to see the commentation of transuranics allowed shallow land burial regulated downward.

·\* • .

5

よいがたい

PAGE FOUR 10CFRPART61 CUIGG

. .

D-567

20

THE STATUS OF SPENT RESINS UNDER LOCFS61

Ion-exchange resins are used to take radioactivity out of primary cooling water; they act like a filter. Spent resins look like little beads of plastic. They are considered low level waste even though they are highly radioactive and contain isotopes with long half-lives like cesium-237.

Under 1007R61, spent reains will continue to be allowed shallow land burial. J. Howard Kittle, an expert in nuclear waste managment at Argonne National Laboratory, told me in a phone conversation (August 4, 1960) that: "It (spent resin) is really in a different category and About be containerized. There is no way to reduce resin volume." Kittle said the NRC requires immobilization of spent resins now or the use of high integrity containers. "But no one knows what a high integrity container is," he said, "and that is why utilities still immobilize their spent resins."

Kittle said: "I strongly favor the consideration of spent resins and . cladding hulls as intermediate wastes. Spent resins require remote handling and clad hulls have to be handled remotely."

Unfortunately, under IOCFR61, spent resins are allowed shallow land burial. This is what industry wants. At a meeting of the American April 25, 1981, John S. Richmond of Wisconsin Public Service Corporatio: said: "Spent resins' should be solidified as polymer or with cement instead of shipped as liquid. It's ok to give them shallow land burial.

But other experts, in addition to Dr. Kittle, disagree. In a private communication with John Feel of the DDB's office of waste isolation in July 1980, he said: "Our focus is on improved shallow land burial. We should quit burying organic chemicals and ion-exchange chemicals. Peel said, "Spent resins should be solidified. You can burn resins in an incinerator or glass smelter at low temperatures. There are two types of resins: you can burn organic resins and melt zeolites."

According b the NRC's R. Dale Smith, (phone conversation Sept. 15, 1981): "what we did in ICCFR-FARTS1 was to establish performance objectives ... allowable exposure. Then we back-calculated to the kinds and amounts of spent resins and other waste that could give that kind of exposure." Smith said, "Under these calculations, most spent resins would fit into the categories of waste that could be given shallow land burial."

Smith went on to say that for spent resins above that limit and inorganic materials, the type of disposal had not been determined -other than the fact that "it should not be shallow land burial." Shith solve wiedged the need for a definition and disposition of intermediate waste, saying "that is next on our egenda."

PAGE FIVE 100FR61 CUIGG

C-4

COMMENTARY ON SPENT RESINS

In my opinion, the NRC has made a serious tactical error in not

D-56-20 level waste rules. If my reading of lOGFR61 is correct, the NRC has already defined TRU waste and spent resins as low level waste. The agency will have a difficult, if not impossible, task of redefining these wastes as "intermediate wastes" in the future.

The allowed exposure to low level radioactive wasta is set by the And at 500 millirem per year per exposed indiridual in 1007Rol. The discussion (page 3800L, Fed. Register, Sept. 2L, 1981) in the proposed rule indicates: "Any waste with concentrations of these isotopes that would cause an exposure greater than 500 millirem must be protected from intrusion by deeper burial or some other barrier."

• • •

Now did the NRG arrive at 500 millirem which is the equivalent of about ten chest x-rays or 20 dental x-rays as a safe dose. Does the NRE take into consideration the greater health risk to children, pregnant women or the fatus, or ill, elderly people from such a dose? There is no basis given for the NRE's projection that only one, or at the most, a few persons would be exposed to the 500 millirem dose. Eow can the NRD predict how many people might be farming or digging in a certain plot of land somewhere in the United States over the next 200 or 300 years?

(And, why doesn't the NRE wait for the Environmental Protection Agency (EPA), which has environmental standards-setting authority, to set its C-2)(IPA), which has environmental standards-setting authority, to set ---limits for radioactive releases to the environment from disposal d facilities? Why is the NRC, in effect, pre-empting the EPA's authority C-3 (by "anticipating" the EPA standards for radicactive releases from muclear dumpsites.?

••

PLUTONIUM IN LOW LEVEL WASTE TO ESCALATE

· . . .

The NAE has separated out from the 10 nGi/g restriction the transuranic element plutonium-241, and allowed it a far greater concentration than was previously permitted in low level waste. On page 38097 of its new regulations, the NAC proposes a concentration of 350 m21/g of plutonium-241 be allowed land burial as Glass G Intruder Waste.

To explain, the NRC states "this concentration of short-lived betaa longer-lived alpha emitter." The NAC goes on to say that "At D-55-3. present, wastes containing transuranic nuclides in concentration greater than 10 nCi/g are not being generated in significant quantities."

> Let us examine these two statements. First of all, plutonium-241 is not short-lived as the NAC states. It has a half-life of 13,2 years, which means it will be toxic in the environment for 132 to 164 years. This is not a short time. As the NRC says, plutonium-241 decays to americium-241, Americium-241 has a half-life of 558, which means = is toxic from 4,580 to 9,460 years. Although plutonium-241 is a

PAGE SIX 1007R61 CUIGG

D-55-31

beta-emitter, its decay products emericium-211 and neptunium-237 (half-life 214 million years) are alpha emitters.

The health risk from alpha radiation comes from its internal penetrating power; it has weak external (beta) penetrating power. If alpha particles are inhaled or ingested, all of their energy strikes a few cells, increasing the likelihood of cell mutation or cell death. Plutonium-2L1 and americium-2L1 emit alpha perticles with strong energies and have high linear energy transfer (LEI); that is they transfer greater energy in a shorter distance. A publication of the Scientists Institute for Public Information, <u>Radiactive Contemination</u> by V. Brodine, 1975, states that "Observations of radiation effects at the cellular level have shown that high LET radiation cannot be repaired."

Americium-21:1 is known to move through the food chain. It should also be noted that any transurances that are given land disposal can, through various natural mechanisms (erosion, intrusion, wind storms), become airborne.

Jeffrey King, MD, researcher in pulmonary physiology at the Harvard School of Public Health, says that americium-241 is in a class with the most radioactive elements and is far more toxic than the most toxic carcinogen. King says, "The special toxicity to the lungs from americium (or plutonium) is a result of the powerful short range, effect of alpha particles on the sensitive bronchial epithelium."

Dr. Karl Z. Morgan, considered the father of health physics in the United States and the former director of health physics at Oak Ridge National Laboratory, says his calculations indicate that when americium and radium-226 may become deposited in the body, "americium-211 is about 16 times more hazardous than radium-226."

After an investigation into the incidence of cancer from plutonium soil contamination downwide from the Rocky Flats plant, Dr. Garl Johnson of the University of Golorado School of Medicine in Denver, concluded: "The consistency of the increase in incidence of all cancer and of certain categories of cancer with increasing concentrations of Pu in soil supports the hypothesis that exposure of the general public to low concentrations of Pu in the environment may have an effect on cancer incidence."

Now comes the NRC with 100FR61 proposing not only that we give concentraions of transuranics over 10 m21/g land burial, on a case by case besis -- but that we also allow 35 times more plutonium-211 than any other transuranic muclide. Gould it be that the NRC and the muclear industry know, as we know, that plutonium -211 is the most common transuranic produced by cornercial nucler power plants? According to a recent study by the Meetric Power Research Institute (EPRI-NP-1052, April 1979), plutonium-221 constitutes 81 percent of the alculated yearly transuranic nuclide production of a 1000 megawatt nuclear power plant.

n a standard de la secondard de

PAGE SEVEN 100 FR61 QUIGG

. . .

The volume and curie amounts of plutonium-211 that will be allowed land burial under ICCPR61 is staggering, when one considers the NRC's allowance that "concentrations may be averaged over the volume of the package. For a 55 gallon drum multiply the concentration units by 200,000 to determine allowable total activity, we are told.

> For plutonium-211, we multiplied 350 nG1/g by 200,000 for a total NRC allowed 70 million nanocuries per 55 gallon drum of plutonium-211 waste.

> It defies reason to understand such regulations by an agency mandated to protect the health and safety of the American public.

## ###

.

Enclosures for the record:

• .

- Letter dated April 30, 1979 from Jeffrey M. King, M.D., to Joseph Strasburg, New York City Council, Legal Division.
- Statement of Karl Z. Morgan, July 28, 1978, regarding americium-241 and radium-225.

ر ر

11

7.

- 3. Report by Carl J. Johnson, M.D., to 5th International Congress of the International Radiation Protection Association, Jerusalem, Israel, March 9-14, 1980.
- L. "Calculated Yearly Nuclide Production From a 1000 Mwe Plant", Table 2.4 from "Activity Levels of Transuranic Nuclides in Low Level Solid Wastes from U.S. Power Reactors," prepared by Science Applications, Inc, MD., for Electric Power Research Institute, Falo Alto, CA., August 1980.





STATE OF NEVAGA DEPARTMENT OF HUMAN RESOURCES DIVISION OF HEALTH BUREAU OF CONSUMES HEALTH PROTECTION SERVICES

303 EAST RIME STREET CARGON CITY, NEVADA - 88716 TELEPMONE: 17021 588-6756

September 24, 1981



R. Dale Smith, Chief Low-Level Waste Licensing Branch U.S. Nuclear Regulatory Commission Wasnington, D.C. 20555

Cear Cale:

In answer to your request, we have enclosed comments on the proposed 10CFR61.

001



Supervisor - Radiological Health Section

JV/jc

Enc.

cc: G. Wayne Kerr (w/enclosures)



STATE OF NEVADA OEPARTMENT OF MUMAN RESOURCES DIVISION OF MEALTH IVISION OF MEALTH



305 East Find Street Called City, NEVABA 40710 Telephone 1702; 843-4750

COMMENTS ON PROPOSED

It appears to be logical to categorize radioactive waste to promote stability of the surface of the site and to protect possible users of the land in the distant future.

To put such a system of classification of radioactive waste into effect, however, takes more than just cromulgating a regulation. It will require the cocceration of the backager, the site operator, the regulatory agency and an institution that will monitor the site for up to 100 years.

The history and record of the backaging of radioactive waste by generators has not indicated a durable capability to package the wasta in conformance with DOT regulations. Further, the types of isotopes and their duantities in the packages of waste are at cest educated guesses. In view of this situation, it does not seem likely that the waste generators will be able to properly classify their waste according to the proposed regulation, or for that matter even label it correctly.

> The site operator will have to accept the radioactive waste as classified by the generator because he cannot open the packages for inspection. The Class A waste would be placed in a designated trench. The site operator would assess potential active maintanance costs for this type of trench and acd a per unit volume charge to the snippers of Class A waste. For similicity, the site coerator would consider all other waste received as intruder waste. This waste would be buried in a different trench under 17 feet of dirt. The additional cost for this type of burial would be passed on again to the generators of the waste. All of the additional bookkeeping required will complicate waste disposal operations and the new methods of cisposal will greatly increase costs of waste disposal. In addition, the regulatory agancy would need to conitor the charges collected by the site operator for active maintenance pecause, if that money was not all used during the five years after site closure, it should be turned over to the institution that will have custody of the site for long term ater and maintenance.

The regulatory agency governing the site would either require waste backages not labeled with a waste classification to be opened by the site operator at his expense, or have the waste returned to the generator at his expense.

The regulatory agencies governing the generators will have to increase their inspections of the actual radioactive waste cackaging operations to determine that the waste packaged does meet the regulaments for the different classifications. It is doubtful that regulatory agencies can do more inspection in this area than they are now doing because of budget limitations.

8-926

9-639

D-55-114

#### Comments on Proposed ICCFR61

 $\mathcal{D} - 55 - 11 \begin{cases} \text{The proposed regulation states that Class B wastes must be stabilized. This will require considerable quality control on the part of the generator in his packaging operations to eliminate voids and to assure solidification. In the case of resins solidified with cement, for example, if this is not done properly, the matrix will eventually crack and the resins could flow under pressure. This means that dependence may have to be placed on the container for prevention of deformation, an additional expense.$ 

Section 61.29 of the proposed regulation states that site operator must actively maintain the site for five years after site closure. The cost of this operation for a new site could be estimated by the site operator and passed on to the waste generators. In the case of existing sites, particularly in arid climates, D-52-2. Sit is not expected that trench caps of trenches containing segregated weste (would crack or collapse for many years, pernaps 50-100.

-2-

A-1 The statement is made in "other considerations" that the proposed regulation would apply to existing sites. In this case, the regulatory agency would require the site operator to actively maintain the site for at least five years after closure at his expense, because there would be no way the site scoreator could recover his costs for this operation. To apply the estimated maintenance costs to current users of the site would make burial cost prohibitive for them at that site. It is not likely that the site operator would agree to this arrangement and probably could not be forcad to do So.

(In 61.7 (b)(2) a statement is made that Class A segregated waste decays to acceptable levels during the period when the site is occupied. What are "acceptable levels"? It is assumed the acceptable levels are below those in D=55-12 (Column 1 of Table 1. In the case of enriched or depleted uranium, nowever,

there would be no appreciable decay during the period the site is occupied. Further, dependent on the definition of "acceptable levels", the receipt of Class A segregated waste at the site would have to be stopped at some coint in time before the site is closed, if it is to decay to acceptable levels prior to site closure.

In summary, a radioactive waste classification system may be effective for new sizes in the future provided an intense education and inspection program can be developed for the generators of the radioactive wasts; and provided that methods can be found for solidifying wastes that need to be stabilized which will not leave residual liquid and not deteriorate in form. Existing sites will require active maintende for a great number of years so the wasta classification system is not meaningful in these cases.

Sother minor comments: Words such as "commentione" adequate complement" and "unreasonable risk" should be eliminated from the regulation. Section 51.23A(g) refers to Section 51.50 which does not exist.

Jann Jacen, Supervisor

ED-

Radiological Health Section

. ... ,

EXON NUCLEAR COMPANY, Inc. "SEARCH AND TECHNOLOGY CENTER Genese Washington Way, Richard, Washington 99332 HONE: (509) 373-7100 October 1, 1981 DOCKET NULLER

> (46FX 3808)) Secretary of the Commission U.S. Nuclear Regulatory Commission Washington, D.C.. 20555 2,19,20,21,30,60,51,77 (1,70,73,17)

Atth: Docketing and Service Stanca

Reference: Froposed Rule, Licensing Requirements for Land Disposal of Radioactive Wasta, Federal Register, Vol. 46, No. 142, pp. 38081-38105.

#### Gentlemen:

We have reviewed the referenced proposed rules and have a number of comments for your consideration prior to preparation of the final rule. In general, the proposed rule appears to be a reasonable bland of performance oriented objectives and technical requirements. The proposed rule should increase and promote public confidence relative to the safe disposal of low level rudioactive wastes.

Specific comments with respect to the proposed rule are given below:

C09C252 511001

7aragraph



A better definition of disposal would be: "Confinement of the wastes with no provisions made for subsequent retrieval."

This subparagraph specifies that there be no interaction between certain disposal units and implies that an absolute condition be accomlished by separation of the disposal units. It would be more precise to state that "Wastes designated as Class  $\lambda$  ... by placing in disposal units which have no <u>significant</u> interaction with disposal units containing other than Class A segregated wastes".

The obligation in the annual report to include information relative to "any instance in which observed site characteristics were different from those described in the application for a license", should be modified to require reporting only of such site characteristics which were "significantly" different...

. . AN APPILIATE OF EXXON CORPORATION

125 1

.

September 24, 1981

Secretary of the Commission

October 1, 1981

-2-

Comment

Paragraph Number 61.32(a) Ambiguities in this paragraph leave open to the Commission the right to inspect waste materials which have been placed and covered within a disposal unit. Since this presumably is not the intent, the paragraph should be modified to specifically limit the inspection of radioactive wastes to materials which have not been placed within a disposal unit and covered. It is not clear how the revised paragraph includes wasta disposal facilities licensed pursuant to the proposed Part 61. Wording similar to that of the revised paragraph 2,104 (e) would appear to be appropriate to more precisely define the affected licensees. (20.311(d)] and To assure compliance with regulatory requirements, while evoiding the implication that a separate (2) (5) GA program exists, the following alternative to paragraph 20.311 (d) (3) is proposed: M-2 "Conduct operations in a manner which assures compliance... chapter; management audits of the operations must be conducted;" Similar words would be appropriate in paragraph 20.311(5)(5). We appreciate having the opportunity to comment on the proposed rules.

Sincerely,

a. Milson, Managar

Corporate Licensing

aNisla



COMMONWEALTH OF PENNSYLVANIA DEPARTMENT OF ENVIRONMENTAL RESOURCES P.C. 2053 Harrisburg, PA 17120



1141

DCC/CTED

6

October 5, 1981

20090525 811 Samuel J. Chilk Secratary of the Commission U.S. Nuclear Regulatory Commission Washington, D.C. 20555

Attn: Docketing and Service Branch

Dear Mr. Chilk:

The Department mas reviewed the proposed regulations covering the licensing requirements for land disposal of radioactive waste (10 CFR 51). Attached are detailed staff comments that contain the specific points the Department is concerned with.

Generally, the use of performance objectives and more general technical criteria instead of the previously suggested specific standards leads to a much more flexible and achievable regulatory system. However, in our coinion, more information is needed in the licensing process.



We understand that the supporting Environmental impact Statzment (EIS) referenced in the proposed regulations has not been issued. We reserve the right to provide the NRC with additional comments on these regulations after we review the EIS.

Thank you for this opportunity to review the proposed regulations.



Sincerel CLIFFORD 1 Secretary of Environmen

3110130163 311005 2 445838081 PDR

#### COMMENTS ON NRC PROPOSED REGULATIONS - 10 CFR 61 by Cepartment of Environmental Resources

The use of general performance objectives and the more general technical criteria instead of the previously suggested specific standards leads to a much more flexible and achievable regulatory system. However, to provide the more definitive information which is necessary for licensees, it is suggested that at least the following topics be addressed in the form of more detailed regulatory guidance:

1. Waste stability;

3.

5.

6.

GE1-2

h.

د

A-2

 Specific assumptions to use for determining whether the intruder scenario meets the performance objectives;

Guidelines and assumptions for setting maximum discosal site inventory limits consistent with the performance objectives; Concentration limits for naturally occurring and other isotores (especially Radium) not specifically addressed in Table 1;

- Socific guidance for the information that is requested in a license application as outlined in Subpart 8. This could be in the form of standard formats and standard review plans such that the licensee would know beforehand the level of detail and, where applicable, minimum acceptable criteria to estated the beforehand the review for the formation of review the set of the
- satisfy the NRC review process; Definitive standards for the conditions that are required to be met for postclosure license transfer and termination of the licensee.

Because of the fact that low level waste (LLM) disposal has become a state responsibility, it would be very helpful to present Agreement States, or States which are contemplating on becoming Agreement or Limited Agreement States for the purpose of regulating a LLW site and for the NRC to formally specify which portions of Part 51 must be accound by the State for compatibility. It is important that this be done as soon as possible so that those states which are contemplating on changing their agreement status and a possible need for specific State Legislation could include in their deliberations the impost that the mandatory adoption of these regulations would have on those decision-making processes. Part H of the Summary seems to imply that all of Part 51 will have to be complied with for compatibility.

It should be assured that these regulations will be compatible with the site selection process that may be either part of a compact arrangement or an individual State process which has been established to meet the provisions of the LLW Policy Act.

Specific Comments on Indicated Sections:

D - 51 + 51.2 Definition of "Engineered Barrier" may be too restrictive in that this term could also be used to describe engineered barriers that limit subsurface migration or surface water intrusion.

A-4 A-4 Solution and safety. 61.7(c)(4) It is recognized that the Energy Reorganization Act precludes NRC licensing of all but those specific DOE activities listed in the Act. Possession of commercial LLW burial sites is not included as one of these specific activities, but NRC should request that Congress consider amending the law to allow licensing of this activity to insure adequate protection of public health and safety.

61.25(b) This provision would allow the opportunity for public hearing every five years during operation whether or not significant changes have occurred. Unless the prerequisites for requiring a public hearing are

6-5 Accurred. Unless the prerequisites for requiring a public nearing are sufficiently stringent, such that this process is not subject to abuse, these hearings could become overly burdensome and potentially lead to unnecessary (cost without a commensurate benefit to health and safety.

B-8  $\begin{cases} 61.30(a) \text{ Does this paragraph preclude the site operator from retaining licensee responsibility during the institutional control period following post$ closure observation and maintenance? If the site owner chooses and the site operator is agreeable, the site operator could remain as the licensee until the requirements are met to terminate the licensee witnout a detrimental effect on public health and safaty.

C-3  $\begin{cases}
61.41 & It should be recognized that the limits of 40 CFR 190 are based on$ a cost/benefit analysis of the potential for reduction of releases from eachstep of the nuclear fuel cycle. These ALARA principles may also be apolicablein the case of LLW disposal since engineered barriers or other disposal sitedesign features could further reduce potential exposures in a cost effectivemanner.

D-50-7 51.50 Consideration should be given to including as a requirement accessibility to major transportation routes. Most assessments have shown that the largest contribution to population exposure from LLW waste management operations is due to transporting the waste from the generator to the disposal site, and therefore this factor should be assessed and minimized if possible. Minimizing the impact of this factor could also positively effect the institutional and political acceptability of a disposal site. The appropriate way to address this consideration may be to include a separate section which lists only institutional requirements such as public water supplies and population density.

D-55-13 wastes need to meet the requirements of paragraph (4) through (10). Certainly Class 3 wastes need to meet at least the same requirements as Class 3, and consideration should be given to requiring Class A to meet most of the acditional requirements. except perhaps paragraph (4).

Section 61.52 (a)(5) the term <u>"reduce</u> future subsidence" should be stated <u>'control</u> future subsidence" "Reduce" is too amoiguous.

ED - 1 Section 61.52 (a)(9) is not a performance specification. It should be rewritten to describe what is to be achieved.

D-53-1-

Section 51.53 (a) Data covering a simple twelve-month period for seaschally variable characteristics is insufficient. Seasonal variations having an impact on future environmental monitoring would better be related to established normals, maximums and minimums as provided by appropriate agencies.

Section 61.53 (d) should include language specifying when performance specifications should be readied and to whom they should be submitted.

D-55-2 D-55-2D-55

51.59(a) It should be noted that nationally recognized groups (State Planning Council and DOE National LLW Strategy Task Force), which have developed policy recommendations for LLW management, nave recommended that the option should be available for States to turn over responsibility for long-tarm custody of LLW disposal sites to the Federal Government if they meet satisfactory decommissioning criteria. NRC should more thoroughly consider this option and make appropriate recommendations for specific Federal legislation if needed.

D-59-2 61.39(b) Consideration should be given to including requirements for, D-59-2 or limits upon, the use of the surface of the disposal site during the institutional control period.

- Jb Surveillance, could continue beyond the 100 year time period if the custodial agency so desires.

-57-2 Section 54.50 deals with performance standards for avoiding environmental harm. Recognized alternative uses should be considered in the assessment of present worth because any size chosen must be restricted for a minimum of 106 years and cossibly over 500 years.

Dow

DOW CHEMICAL U.S.A.

October 12, 1981 DCCTET INTER PROJECT D RECORDED ATER I R-2, 1 Secretary of the Conmission Vashington, D.C. 20555 Attention: Docteting and Secure Based

Attention: Docketing and Service Branch

Subject: DOW COMMENTS, PROPOSED RULEMAKING 10 CFR 61

GEN-7 Dow comments dated September 18, 1981, on the subject rulemaking are enclosed. Commission employees have verbally informed Dow that the comment period will be extended to January 1982. No written confirmation of the extension has been noted in the Federal Register or any other official source. The Dow comments are therefore being submitted at this time to assure that they are within the official comment period. Additional comments will be submitted as necessary and as opportunities arise.

Sincerely,

J. B. Oven Group Leader Nuclear & Solidification Services 517-636-3388

fo

AN OPERATING UNIT OF THE DOW CHEMICAL COMPANY

September 18, 1981

DOW CHEMICAL U.S.A. LARKIN LARCRATORY 1691 N. SWEDE RO. MICHIGAN 48640 September 18, 1981 PROFILED NULE PR-2 et TELLE (46 FR 38081) DODIETED Secretary of the Commission LISNES U. S. Nuclear Regulatory Commission OCT. 19.1981 > Washington, D.C. 20555 Attention: Docketing and Service Branch Subject: PROPOSED RULEMAKING 10 CFR 61

Dow Chemical U.S.A. (Dow) is licensed by the U.S. Nuclear Regulatory Commission (Commission) to possess, use, and transport certain radioactive materials. Dow is actively engaged in developing, testing, and marketing technology for cleaning equipment that is contaminated with radioactive materials. The Commission has reviewed and accepted the Dow topical report, DNS-RSS-001-P-A, which describes Dow technology for solidification of low-level radioactive wastes. Dow recognizes the need for and supports actions which result in proper disposal of radioactive wastes. For these reasons Dow welcomes the opportunity to comment on proposed rulemaking 10 CFR Part 61, Licensing Requirements for Land Disposal of Radioactive Waste, published in the Federal Register on July 24, 1981.

Specific comments are as follows.

Supplementary Information

Section II acknowledges that current regulations do not contain any standards or technical criteria for the disposal of licensed materials and that the need for standards and technical criteria is well documented. Section III acknowledges that the Commission has had a program underway for several years to develop regulations and other guidance for the management and disposal of low-level wastes and references NUREG-0782, a draft environmental impact statement, to provide guidance and support to 10 GFR 61.

support to 10 CFR 61.

AN OPERATING UNIT OF THE DOW CHEMICAL COMPANY

#### Comment

. .

Medical and research facilities, nuclear power plants, and other commercial activities have been generating significant amounts of low-level radioactive wastes each year since the 1950s. The first commercial site for disposal of these wastes was opened in 1962. The lack of regulatory standards and technical criteria resulted in the selection of some disposal sites and use of some disposal techniques without regard for the concept of as low as reasonably achievable (ALARA) and for which there is very little or no technical or economic justification. Some private companies, such as Dow, have developed new or improved disposal techniques which have been tested in private and national laboratories and are being used by some waste generators and service companies that support them. However, due to the lack of enforced regulatory standards and technical criteria most of the waste generators have shown significant reluctance to use the new or improved techniques. There are a few waste generators that have openly stated that they will not change or improve disposal techniques until they are required and Anforced by regulation.

-2-

The reference to NUREG-0782 as guidance and support to 10 CFR 61 is an example of Commission actions which result in waste generators being reluctant to act without specific regulatory direction. Dow attempted to obtain NUREG-0782 and was informed that it is still being drafted and that when the draft is available there will be a 90-day comment period. The comment period for NUREG-0782 will obviously extend beyond October 22, 1981, which is the end of the comment period for 10 CFR 61. To comment on 10 CFR 61 while the guidance and support that is to be provided by NUREG-0782 is still being drafted tends to add to the waste generator's reluctance to change. In the meantime, the waste generators will continue to generate wastes and use disposal techniques which are known to compromise the ALARA concept and the goal of assuring protection of public health and safety during transport and disposal of the wastes.

The Commission must recognize that there is an immediate need for regulatory standards and technical criteria and not allow the development program to continue for several more years. Immediate actions should

GEN-7

September 18, 1981

be taken to establish and enforce standards and technical criteria based on the concepts of ALARA and best available technology for disposal of radioactive waste at the existing disposal sites. The Commission has used the excuse that the existing disposal sites are located in and licensed by Agreement States, therefore, the present regulatory standards and technical criteria are the responsibility of the States. This is not a

~3-

- D-56-11 valid excuse. The States have limited technical and economic resources to evaluate new or improved techniques and establish and enforce standards and technical criteria... Like the waste generators, the States are also reluctant to act without Commission guidance and assistance with these matters. The need for the Commission to provide guidance and assistance to the States has received additional emphasis by passage of the Low-Level Radioactive Waste Policy Act which authorizes the States to enter into compacts in order to establish and operate regional disposal sites. The Commission should take the lead by providing basic standards and technical criteria and strengthen the Agreement States Program with additional technical and economic assistance.
  - 2. (Section IV explains why the term "low-level radioactive wasta" is not used in 10 CFR 61.
  - ED-1- Comment

Dow agrees that wastes should be defined in terms of radioactive and chemical properties and not in traditional terms related to their origin.

3. Section V is a discussion of Commission logic and philosophies related to the use of prescriptive requirements versus performance objectives and the minimum technical requirements related to disposal site selection, operation, and closure including waste characteristics and classification. Land ownership, institutional control, financial assurance, life cycle, and impact of other regulatory requirements and groups are also discussed.

Comment

GEN-2 Dow agrees with most of the logic and philosophy of this section. However, the present needs for regulatory standards and technical criteria can not be overlooked and require the following specific comments.

- 4--

#### September 18, 1981

## Subsection B

D-56-8

Placing reliance on stability of the disposed waste at the time the facility is closed will be possible only if stability is specified and enforced by the standards and technical criteria during the disposal operation. As pointed out in previous comments, techniques which provide stable waste forms are available today but due to the lack of regulatory standards and technical criteria they are not in general use. These techniques result in monoliths which place the radionuclides into forms which are less likely to be dispersed and less likely to be released to and transported by ground water. These monoliths are more likely to be recognized and therefore avoided or properly investigated and handled by intentional or inadvertent intruders. All of these factors are within the concept of ALARA and reduce the need to rely on the many unknowns and uncertainties associated with pathway analysis.

Subsection C

As previously stated, Dow agrees that wastes should be classified by their radioactive or chemical properties. We also agree with the intent and the five disposal site characteristics that are listed. However, without benefit of NUREG-0782 it is not possible at this time to provide specific detailed comments concerning the waste classification scheme.

ED- | It is possible to comment that listing waste streams under Class A, B, or C appears to be an attempt to classify wastes in terms of their origin, and therefore, is in direct conflict with the intent of the classification scheme. It is acknowledged that it is implied that NUREG-0782 contains radioactive and chemical data to support this listing.

D-55-2 The minimum set of the m -5-

September 18, 1981

D-55-2 and questions the need for the Commission to work on this definition over the next two years.

Subsection H

77-55-8

Dow agrees that many of the operational provisions and waste characteristics requirements proposed in this rulemaking are in effect at the existing disposal facilities. The point is that the waste characteristics requirements proposed in this rulemaking do not reflect the concepts of ALARA and best available technology. Just as important, maybe more, is the condition that neither the Commission or the existing disposal facilities have inspection and enforcement programs to assure that the requirements are being met. Standards and technical criteria without inspection and enforcement are useless and unnecessary.

· · ·

.

Part 61 - Licensing Requirements for Land Disposal of

## Radioactive Wastes

To comment on each Subpart and Section would in many instances be to repeat comments which have been made regarding <u>Supplementary Information</u>. In an effort to minimize repetition Dow has limited its comments to the following specific Sections or paragraphs.

1. /Subpart A, Section 61:7 Concepts. (b)

Dow agrees that the primary safety objectives are to prevent migration of radionuclides and exposure to intruders. It is also agreed that stability is a cornerstone of the system to control migration of radionuclides and contributes to intruder protection by providing a recognizable and nondispersible waste form.

Dow has no objections to philosophies of the waste classification scheme. There is some concern that too much emphasis is being placed on pathway analysis and the ALARA concept is being overlooked. However, specific comments can not be made until NUREG-0782 is available.

. . .

Subpart B, Section 61.13 Technical Analyses and Subpart C, All Sections

-6-

Once again, Dow has no objections to the philosophies expressed in these Sections but emphasis on pathway analysis and specific dose values without including the ALARA concept continues to cause concern. Dow has noted that the Commission has mentioned ALARA one time (page 38084) in the Supplementary Information and not at all in Part 61. In an effort to reduce the uncertainties associated with pathway analysis, the Commission should make it very clear that the spirit of the ALARA concept applies to all standards and technical criteria.

3. Subpart D, Section 61.55 Waste Classification

Dow understands that the basis for this Section will be contained in NUREG-0782. Specific comments will therefore be made when NUREG-0782 is available. It is possible to make the following general observations.

The Commission is to be complimented for not including the listing of waste streams under Class  $\lambda$ , B, and Cthat Dow objected to in the Supplementary Information.

Table 1 leaves the impression that the Commission is completely ignoring the needs of today and the objectives of assuring protection of the workers, the general population, and the environment during the operation of the disposal faciltiy. For example: Class A segregated waste may be transported and disposed of with only minimum requirements on waste form and characteristics but may contain up to 140 curies of  $^{60}Co$  in a 55-gallon drum. The minimum requirements on waste form and characteristics are set forth in Section 61.56(a) and will permit packaging wastes which are in a readily dispersible form and which contain significant volumes of liquids that will contribute to radionuclide migration by ground water. Dow understands that 140 curies of 60Co in a 55-gallon drum will decay to about 0.1 millicurie after 100 years of institutional control and is therefore considered to be of little concern to the intruder. However, there is doubt that 100 years of institutional control is possible and 140 curies of <sup>60</sup>Co in a 55-gallon drum is of significant concern during the transport, disposal, and first 50 years of institutional control.

(-5)

D-55-13

September 18, 1981

11

: 77 - 22

-::

#### -7-

September 18, 1981

 $D-55-13 \begin{cases} The point is that there are waste solidification tech$ nologies available and in use today that with verylittle effort and cost can place this 140 curies of<math>50 co into a stable form. Use of these technologies will dramatically reduce the transport and disposal problems of today as well as significantly minimize the potential hazard to the intruder should the institutional control fail in less than 100 years.

> The Commission should not ignore the needs of today and establish standards and technology only for the benefits and protection of intruders after 100 years of institutional control. This Section appears to be in direct conflict with the concepts of ALARA and best wailable technology.

4. /Subpart D, Section 61.56 Waste Characteristics

D-56-13

Dow suggests that minimum requirements for all classes of waste should exclude the presence of liquids. Historically the Commission and the nuclear industry have used the term "no free standing water" to describe this exclusion as a desirable characteristic of the wastes. The Commission's Effluent Treatment Systems Branch (ETSB) defined free water as uncombined water not bound by the solid matrix in Branch Technical Position - ETSB 11-3 dated November 24, 1975. The Commission funded a program at Brookhaven National Laboratory (BNL) to investigate the properties of solidified radioactive wastes and containers. BNL summarized the results of their program for the period from April 1976 through September 1978 in NUREG/CR-0619 SNL-NUREG 50957. The BNL investigation confirmed much of the information that had previously been reported by Dow and others. The Central Electricity Generating Board (C.E.G.B.), in the United Kingdom, has conducted a similar investigation and confirmed much of the BNL information. At the request of the Commission, BNL also provided input relevant to the establishment of free standing water criteria revised October 15, 1979. Dow suggests that the Commission has sufficient justification to limit noncorrosive liquid in the waste, as referenced in paragraphs 61.56, (a), (3) and 61.56, (b), (2), to not exceed one-half percent (0.5%) of the waste volume or one gallon per container, whichever is less.

Dow was informed by a responsible Commission employee that one of the purposes of paragraph 61.56, (b), (2) was to provide for disposal of ion exchange media by merely dewatering to 1% of the volume of the waste. -8-

Dow assumes that the wording of paragraph 61.56, (b), (1) which allows the disposal container to provide stability after disposal is also for this purpose. Dow suggests that the practice of disposing of ion exchange media by merely dewatering, to any level, is not within the concepts of ALARA and best available technology. Dow is not aware that the Commission has any information to indicate that any of the disposal containers now available, including the recently introduced high integrity containers, have been tested to show that they can be expected to provide stability in the dispo-sal environment for at least 150 years. Dow suggests that in preparing Section 61.56 the Commission has -56-124 ignored the information provided by BNL and others. Section 61.56 therefore fails to reflect the concepts of ALARA and best available technology; it is a significant change from Section 61.86 in draft 10 CFR 61 dated November 5, 1979; it is a significant, unnecessary, and unjustified relaxation of criteria now in use at two of the three operating disposal sites; and it will not achieve the objectives of the Commission as stated in the Supplementary Information, Section V Summary of Rule and 10 CFR 61, Subpart A, Section 61.7 Concepts. As written and published, this section will not alleviate the reluctance of the waste generators and the States to use new or improved techniques or provide them with a clear indication of future regulatory standards and technical criteria. This entire section should therefore be rewritten to correct these conditions.

GEN-2 Dow understands that the Commission, with assistance from BNL, is preparing a Sranch Technical Position which is intended to describe a stable waste form. Dow feels the Commission now has the information necessary to accomplish this task in a timely manner. In addition to limiting noncorrosive free standing liquid to less than one-half percent (0.5%) of the waste volume or one gallon per container, whichever is less, the characteristics should define stability in specific terms such as homogeneity, leachability, compressive strength, yield strength, impact resistance, radiation resistance, thermal resistance, biological resistance, and similar properties.

.

B-208

-9-

September 18, 1981

. . .

-10-

September 18, 1981

5. Subpart G - Records, Reports, Tests, and Inspections

. ..

M

As previously stated Dow feels that regulatory standards and technical criteria without inspection and inforcement are useless and unnecessary. It is noted that the Commission intends to amend 10 CFR 20 to include a new Section 20.311, Transfer for disposal and manifests. (Paragraphs 20.311, (d), (3) and 20.311, (f), (5) provide for conducting quality assurance programs to assure compliance with 10 CFR 61 Sections 61.55 and 61.56. The Commission is to be complimented for this intent and is encouraged to follow up to assure that these quality assurance programs are properly performed. Compliance can only be assured by inspection and enforcement at the waste generators site. Dow suggests that this follow up include a full scale preoperational qualification test similar to that which was conducted prior to solidification of decontamination-waste from the Auxiliary Building at Three Mile-Island Unit II during July 1979 or as described in Appendix 2 of ANSI/ANS-55.1-1979 with an appropriate process control program to assure that radioactive waste are solidified in the same manner to produce the same results.t. red to a 

In summary, Dow agrees with many of the goals, objectives, philosophies, and definitions expressed by the Commission in this rulemaking. However, the Commission has failed to ; recognize that there is an immediate need for regulatory standards and technical criteria for the proper disposal of wastes that are being generated and disposed today. Pathway analysis has been emphasized without inclusion of the concepts of ALARA and best available technology. This emphasis has resulted in a waste classification scheme which may limit exposure to the intruder after 100 years of institutional control but fails to consider the need to protect the workers, general population, and environment during transport and disposal operations. The Commission has sufficient technical information from BNL, Dow, and others which when considered with the experience of the nuclear industry justifies immediate action to establish and enforce regulatory standards and technical criteria which will meet the agreed to goals, objectives, philosophies and definitions. Dow representatives are available to provide additional information, assistance, and comments necessary to assist in the timely establishment of appropriate regulatory standards and technical criteria.

Sincerely.

. D. Cenan /J. B. Owen Group Leader

Nuclear & Solidification Services 517-636-3388

fo

· · · · TELLE COCKETED USNRG ISHAM, LINCOLN & BEALE 0CT 191981 > Office of the Secretary ONE FIRST NATIONAL PLAZA FORTY-SECOND FLORE Dortrier & Samo CHEAGO, ILLINGIS 60403 TELET-ONE 313-884-7900 TELEX: 1-5288 ./4 Oct 2 0 1561 102-413-6736 Heb McList Internet 18 Partz-24 Secretary U.S. Nuclear Regulatory Commission Washington, D. C. 2055 חצבונה הבכת PROTOTED RULE PK 46 an The state Attentions: Docketing and Service Branch · · · · · · maser church Commonwealth Edison Company (Commonwealth) submits these comments on the proposed rules for Licensing Requirements for Land Disposal of Radioactive Waste, published in 45 Fed. Reg. 38081. Commonwealth holds operating licenses for saven operating nuclear reactors and construction permits for six more units. Commonwealth and its customers in Northern Illinois have a substantial interest in and need for the prompt establishment of a Midwest low level waste facility. Accordingly, we have the following suggestions. F 4 4 4 4 4 The proposed additions to 10 C.F.R. Part 61 will unnecessarily prolong the time required to obtain final approval of a license to build a disposal facility for low level radio-'active waste, by delaying effectiveness of a license until all administrative and judicial appeals have been exhausted. ye ; Commonwealth urges that proposed 10 C.F.R. 5561.3 and 61.7 be revised to: allow licenses; to become effective immediately. when granted by the Atomic Safety and Licensing Board (ASL2) or, at the latest, upon final review by the NRC. - 1.

Present regulations, 10 C.T.R. §2.764 allow construction authorizations and operating licenses to become effective immediately upon issuance by the ASL3. This "immediate effectiveness" rule fully protects the public health and safety without costly and unnecessary administrative delays. As the rule now stands, construction and operation of the facility is conditioned upon approval by the ASL3, the schministrative body most qualified to judge the compliance of the proposed facility with federal laws; moreover, a procedure is available for delaying construction and operation even after ASL3 approval, if good cause is shown why the approval, should not become immediately effective.

the war and the second second second second second

Part2-

· ·

-2-

• .

. . . .

Proposed 10 C.F.R. \$561.3 and 61.7 will delay effectiveness of a license until review by both the Licensing Appeal Board and the Commission itself, and, as implied in the Supplementary Information preceding the proposed rule, 46 Fed. Reg. 38087, appeals through all levels of the federal judicial system. This could cause a delay of several years between initial approval of the license and the beginning of construction of the facility. The small and speculative benefit to public health and safety which may accrue from this delay is not nearly commensurate with the added cost to the applicant, its customers and the public at large who benefit from 11w generating activities and who urgently need the prompt establishment of new regional llw facilities. At a minimum, the rule should be revised explicitly to exclude the possibility raised by the Supplementary Information, that effectiveness will be delayed until all federal judicial appeals have been exhausted. The judiciary brings no additional technical expertise to review of a nuclear waste disposal license, and judicial review will shed no critical new light on a proposed facility which has been reviewed by, and has the approval of, all reviewing bodies within the NRC. Celay of construction until the end of judicial appeals will add greatly to time and expense, with no concomitant benefit to the public health and safety license.

The second secon

(1) The second secon

الم المركز الم

B-210



I respond to the Nuclear Regulatory Commission's invitation to comment on the above proposals on behalf of the regulatory departments in the United Kingdom with responsibilities for radioactive waste management. They are the Department of the Environment, the Ministry of Agriculture, Fisheries and Food, the Yuclear Installations Inspectorate and the Scottish and Weish Offices. I also annex comments received from other UK organisations.

The proposals made are most interesting, demonstrating the absolute standards approach rather than the case-by-case regulation commonly adopted in the United Kingdom. Document 10 CFR 61 illustrates the setting of overall performance objectives whilst allowing some flaxibility in desiming and operating each individual repository, whereas document 10 CFR 60 appears to set acceptance criteria not always justified by technical evidence. This could, in our view, lead to unduly restrictive or lax standards when research results permit a more rigorous analysis to be undertaken.

The United Kingdom regulates the whole system of disposal of low level radioactive wastes. We deal with disposals on individual sites and on the basis of individual waste types, matching waste to local conditions either in shallow land-burial sites or by disposal to the marine environment. Our practice recognizes that wastes will eventually leach from a disposal site and we aim to ensure that adequate dilution and dispersion takes place to protect critical groups from unacceptable doses.

A - 8

We also recognize the merit of developing a range of repositories offerint levels of isolation appropriate to the nature of the waste utilising, for instance, existing cavities or purpose-built repositories at intermediate depths.

In the case of high-level, heat-generating wastes, the United Kingdom is not committed to the use of the geological route for disnosal. Research is taking place into various options (disposal to various types of geological formation, disposal on or under the ocean bed, very long-term storage at or near the surface). The first step however will be to vitrify such wastes, and it is then likely that the vitified wastes will be stored, using existing technology, for a period of 50 years or even longer, in order to facilitate supervision and secure a substantial reduction in heat output. The best solution can then be chosen on the basis of full and detailed information about the alternatives. We believe that such storage is a realistic and acceptable proposition, and it has gained a considerable measure of public acceptance in the United Kinedon. Because of this, we do not see a need in our own case for the particular concept of 'retrievability' which is incorporated in your proposals: we shall not undertake disposal until we are satisfied that the repository in question will be safe.

In regard to problems of human intrusion into a repository, our view is that institutional controls are required only over that period during which the hazard is primarily due to the fission products. Similarly, our existing control over potential repository sites does not take into consideration present-day population distribution, since we cannot predict population changes in the future.

In taking account of the many technical factors to be considered in a choice between the several options for creating a repository for hest-emitting vastes, we have adopted a systems study approach in order to bring together the many areas of research and operation. As a result we would not wish to set down specific criteria or defined levels for all variables, because these levels will differ from site to site. We think it advisable for any regulatory procedure to make allowance for natural variation.

Comment has been invited on a number of specific issues and those on which we can comment are dealt with in the paragraphs above. Those remaining relate to matters of national concern and do not call for any comment from the UX. There are, however, two matters which should be considered as meeding clarification:

- (a) the ALAPA principle, mentioned as a footnote on p.35289, might nerhans be explained more fully in relation to the anyroach adopted by the ICRP; in particular the need to take social and economic factors into account;
- (b) on the same page, in column 2 <u>A</u> T it is not made clear whether "one part in 100,000" refers to the activity in the year in question or to that which was originally emplaced. This seems to be a critical matter which must be unambiguous.

The detailed comments received from the National Radiological Protection Sourd, the United Kingdom Atomic Energy Authority, and the Natural Environment Research Council's Institute of Geological Sciences are annexed. I must stress that the views expressed are those of the orranisation submitting them.

Yours sincerely F S FEATES

1
## COMENTE ON NEC PROPOSED RALES

.

## by National Radiological Protection Board.

10 CFR Part 60 Discosal of High-Level Radioactive Waste in Geological

## 1. General Comments

- 1.1 The rule has been developed in the absence of radiological protection criteria (environmental standards), for disposal of high-level wastes; the proposed technical criteria are, therefore, arbitrary. This express to setting technical criteria is incorrect in principle. It leads to criteria which are inflaxible because, since they have no clear basis, there can be no basis for changing them. In addition the approach is very likely to lead to criteria which are too restrictive, thus causing nore expenditure on high-level waste disposal than is warranted by radiological protection considerations.
- 1.2 The rule does not define in any detail the means by which compliance with performance objectives is to be demonstrated. As a consequence the proposed performance objectives have little meaning and it is very difficult to decide whether they are appropriate or achievable.
- 1.3 One of the main purposes of the rule is to reduce uncertainties in predicting the performance of geologic disposal systems. While this is a reasonable objective it is over-emphasised. This leads to technical criteria which may be too restrictive, particularly for TRU wastes. It also down-grades the "geologic barrier" to an extent which is incompatible with the reasons for selecting the geologic disposal option.
- 1.4 For the reasons outlined above, the proposed rule is unsatisfactory and should not be adopted in its present form. It would be preferable to leave the rule in "proposed" form until the TA standards have been published and until there is sufficient information available to derive sechnical criteria from these standards. The rule should then be revised.
- 2. Speicific Points
- 2.1 Retrievability The retrievability requirement (p. 35232 and Section 60.111 P. 35289) is porrly defined. It is not clear whether retrieval is regarded as an extreme measure and the intention is only to ensure that wastes can be located and could be retrieved at great cost, or whether a repository struid be designed and operated so that retrieval would be
- 2.2 Human Intrusion It is reasonable to distinguish between deliberate intrusion and indivertent intrusion, and to ignore the former possibility when setting criteria. However, the probability and consequences of indivertent intrusion should be considered when setting technical criteria such as the minimum disposal depth, and is specifying site selection and marking regulaments. The statement (p. 35233, column 2, para. 2) that it "does no use to argue" over measures designed to prevent intrusion or to "speculate" on intrusion scenarios is inconsistent with the discussion which proceeds it. Such arguments and speculations should play a part in deciding whether a respository is acceptable.

-1-

. .

.

- 2.3 Alternative Approaches (to prescribing criteria) The "alternative approaches" (p.35283), column 3, para. 2) considered by NPC should not be considered as alternatives, they represent stages in developing criteria and parformance objectives for geologic disposal. In order to develop a consistent set of criteria it is necessary to set a "single overall performance standard", to derive "minimum performance standards for each of the major elements" from this overall standard via safety assessments and then to derive detailed numerical and engineering criteria from these performance standards for the system elements. In choosing "Alternative 2" NRC are acting prematurely and limiting the designers' flexibility to an unreasonable extent (see also general comments).
- 2.4 Interdependency of Site and Design The NPC view that site and design are so interdependent that it is artificial and misleading to separate siting and design requirements (p. 35234, column 1, para. 3) is very realistic. Separation of the two types of requirement would inevitably lead to inconsistencies and difficulties in meeting criteria.
- 2.5 Siting Requirements (population density) The decision not to include any siting requirements which deal directly with copulation density or proximity to population centres (p. 35284, column 3, para. 4) is both practical and realistic. Given the long time periods of concern in geologic disposal it would be impossible to ensure that any such requirements were met.
- 2.6 Definitions of "Disposal" and "Isolation" Taken together the definitions of disposal and isolation (Section 60.2, P. 35285-3) are consistent with UK concepts and include recognition that isolation implies control of releases to the environment, rather than zero release.
- 2.7 Probabilistic Aspects of Assessments In the description of the contents of assessments (Section 50.21 (ii), p. 35237) there is no mantion of evaluating the probabilities that various events and processes will occur. The implication is that NRC have not reached a decision on the extent to which probabilities should be included in assessments, in spite of the indications that ERA standards will require a probabilistic approach. This section of the rule (and other related sections) require revision to clarify the NRC position on probabilistic assessment methodologies.
- 2.8 "Reasonable Assurance" The requirement for reasonable assurance (Section 60) 101 (a) (2), P. 35286) that performance objectives and criteria will be net does not provide sufficient quidance for applicants. Since the performance objectives and criteria in the proposed rule are arbitrary (see general comments) it is obviously difficult to define clearly how compliance is to be demonstrated. However "reasonable assurance" ortid be statistically defined, taking into account the increase in the uncertainty of predictions with time.
- 2.9 MARA NRC specifically seeks comments on whether an ALRA principle should be applied to the performance objectives for containment and control of releases (Section 60.11, p. 35239). It is difficult to see the logic in applying ALRA to particular parts of the disposal system while criteria for other parts. The ALRA principle should be a major feature of the overall performance objective for the whole disposal system. The principle should then be applied both in decisions on whether the whole system is acceptable and in satting performance objectives for system elements. In the latter case it will be necessary to show that performance objectives are not over-restrictive, in the source that they entail nore financial expanse than is warranted by the reduction in total risk. It seems likely to us that splitation of ALRA to contairmant

-2-

criteria would show than containment for 10<sup>3</sup> years is not a reasonable requirement for either HEN or THU wastes.

- 2.10 Favourable and Potentially Adverse Conditions This approach (Section 60,122 and 60, 123, P. 35290-1) to specifying geologic criteria is a very reasonable and practical one. It allows considerable flexibility in selecting sites, making allowance for the wide variations in geologic environments. The inclusion of a minimum disposal depth of 300m as a favourable condition, rather than a strict criteria, is a particularly good feature of the approach.
- 2.11 Backfill Functions The processed rule requires that the backfill shall serve a number of functions (Section 60.112 (1), (3), (11), P. 35293), without specifying the degree of effectiveness required. It seems unlikely that any backfill material could perform all of these functions effectively. The objective should be to choose a material with the optimum combination of properties.

-3-

. .

10 CTR Part 61, Licensing Requirements for Land Dispesal of Padioective Naste

## General Comments

- 1.1 The rule appears to be satisfactory when viewed in the context of the current US waste management situation. The approach of specifying overall performance objectives (radiological protection standards) and perscriptive requirements is sound. It leads to criteria which are consistent and soundly based, while providing considerable flexibility for selecting the design features and operating procedures leading to an optimum disposal system for particular waste types and sites.
- 1.2 The emphasis on the longer term aspects of near-surface disposal (site 'closure, institutional controls etc.) is entirely appropriate. These aspects have not received sufficient attention in the past.

#### 2. Specific Points

C-

2.1 Performance Objectives - The radiological protection standards specified in the rule (p. 30083 and Sections 61.41, 61.42) appear to be reasonable. The approach of setting an individual dose limit for indivertent intrusion and separate limits for groundwater releases is consistent with the proposed UK approach to radiological protection standards for disposal options and with ICRP principles. However, it is not clear whether the 500 mrm yrl limit in the intrusion case is entirely consistent with ICRP dose limits because the rule only states that the limit is for whole body dose, not whether it is in terms of effective dose. This point should be clarified. The dose limits for groundwater releases are consistent with current US practice and, when embined with the requirement that EPA Drinking Mater Standards should not be exceeded, should ensure the dose are as low as reasonably achievable.

#### 2.2 Minimum Technical Acquirements

D-50-2 2.2.1 Groundwater - The requirements related to access of groundwater to the waste (0.30084 and Sections 61.7, 61.50, 61.51) seem unduly restrictive and somewhat inconsistent with the performance objectives. There should be no requirements to prevent water access, mininise water access or show that diffusion is the predminant releases via groundwater rechanism. It is only recessary to ensure that releases via groundwater will not result in a failure to meet the performance objectives. . Irs.

2.2.2. Waste Classification - The classification of wastes into 3 categories, the basis for this classification and the associated corditioning and disposal requirements may well be appropriate for US wastes. However, the limit of 10 nCl g<sup>-1</sup> (Section 61.55) on alpha-emitting transuranic radionuclides could be too restrictive for Class C wastes. In addition it is not clear whether this limit applies to Class A and Class B wastes as well; if it does the reasons for chorsing the same limit for all wastes should be explained.

D-56-6 2.2.3 Packaging - The requirement that all waste must be packaged to transport standards (Section 61.55 (a) (i), p. 38097) is not emplicit. In particular it is not clear whether wastes should be packaged for disposal (thus prohibiting the use of returnable shielded packages) or whether they only need to be packaged for transport and handling prior to disposal.

-4-

D-55-3

D-56-0 (Section 61.56) does not appear to be consistent with either the maximum institutional control period (100 years) or the requirement that natural or engineered barriers should provide protection against human intrusion for at least 500 years.

.

### Inconsistencies between 10 CTR Part 60 and 10 CTR Fart 61

C-1

There are a number of important inconsistencies between Part 60 and Part 61. The overall effect of these inconsistencies is to support our view that Part 60 has a less satisfactory basis than Part 61.

1. Approach to setting Criteria and Performance Objectives - In Part 61 overall performance objectives (radiation protection standards) and prescriptive requirements are specified. This approach leads to a rule which is generally satisfactory and which has a clear basis in radiological protection as well as practical engineering and administrative procedures. In contrast, Part 60 contains a number of arbitrary requirements, has no clear basis in radiological protection and may therefore be to a large extent uncorable. The major inconsistency between the two rules is likely to create unnecessary difficulties in implementing NUM disposal since it will reinforce the views of those who claim that disposal of this waste cannot be carried out in an acceptable manner.

2. Protection of Individuals and Populations - In Part 61 both individuals and populations are considered when setting radiological protection criteria; limitation of doses to individuals in the event of human intrustion plays a major part in specifying prescriptive requirements. As noted above, radiological protection principles and criteria have played a minor role in developing Part 60. However, this rule does not contain references to the EPA standards for HUM disposal and the indications are that these standards will deal only with health effects in populations. It, therefore, appears that NRC are proposing to use redically different radiological protection criteria for HUM disposal. This is not only inherently inconsistent, it is also inconsistent with accepted radiological protection principles.

3. Human Intrusion - The human intrusion question is discussed in Part 60 but largely dismissed as a factor in disposal decisions. In Part 51 many of the performance objectives and technical requirements are sized at protection of the indivertent intruder. While intrusion is clearly more probable at near-surface disposal facilities and should, therefore, receive more attention in this case, it is inconsistent to dismiss it when specifying requirements for deeper disposal.

4. Institutional Controls - Part 61 does not permit reliance on institutional controls for more than 100 years and aims to minimise the burden of surreillance on future generations. Part 60 requires that a gualquic responsionry should remain open for 110 years. The reasons for these apparently differing views of the extent to which waste management responsibilities should be' passed on to future generations are not clear.

5. Financial Assurances - The financial responsibility for site closure and observation is treated in detail in Part 61. In part 60, where the timescales are longer and the need to establish responsibilities is greater, this question is notaddressed; it is implicitly assumed that finds will be available.

-S-

6. ALARN - The ALARA principle is implicit throughout Part 61 since NTC has analysed the cost and benefits of a range of performance objectives and presecriptive requirements. In Part 61 there is no evidence that this principle has been applied and its application appears only to have been considered in specifying containment and release rate objectives.

7. Intermediate Level Nastes - The net effect of the two rules is to require that wastes with less than 10 nCi g<sup>-1</sup> alpha emitting transuranic radionuclides should be disposed of by near-surface burial and that wastes above this limit (TRU wates) should be treated to produce a highly leach resistant waste form and placed in deep geologic repositories. Although there is a provision in Part 61 for dealing with more active wastes on a casa-by-case basis, it is evident that this provision is not intended to cover

D-55-3 case basis, it is evident that this provision is not intended to cover TRU wastes. Part 60 requires that TRU wastes should be managed and disposed of in essentially the same way as HIW. The two rules, therefore, praclude the option of disposing of TRU wastes at intermediate depth without highly sophisticated and expensive conditioning. This may be unimportant in the current US situation where the rate of generation of TRU wastes is low. It would not seem to be a reasonable requirement for the UK where there are already large volumes of intermediate level ("TRU") wastes in store with more being produced and may lead to problems in the future in the US. The two rules do not provide an approach to the disposal of all the categories of intermediate level wastes which is consistent with the ALWA principle.

-6-

National Radiological Protection Board, Chilton, Didcot, Ducn, Chilton, Chilost, Ducn, Chilton,

30th September, 1981.

B-214

#### COMMENTS ON N.R.C. PROPOSED RULES

BY

#### UNITED KINGDOM ATOMIC ENERGY AUTHORITY

## 10 CFR Part 50, Disposal of High-Level Radioactive Wastes in Geologic Repositories

#### General

We feel that too many firm numbers are being laid down without sufficient experimental and theoretical justification. Particularly if disposel will not take place for many years it is better to set overall dose limits to define the required performance of the multiple barrier. It is then up to designers to optimise the individual elements in the system as models and experimental data are improved over the years. The proposed rule would freeze options too scon. Host of the numbers quoted are probably reasonable but at this stage they should be examples not rules.

We believe that over-emphasis on designing for retrievability could distort the design of a repository. For heat-emitting waste it is better, therefore, to aim for a period of retrievable storage followed by disposal with no intention of retrieval. After this, retrieval is always possible as a last resort but it should never be necessary and should not be allowed to distort the design. This problem might be eased by having separate rules for spent fuel and other high-level waste since wasts should be eventually disposed of but fuel is a resource which may need to be retrieved.

In a number of cases the definitions quoted differ in detail from those published in the IAEA Nuclear Waste Management Blossary (Draft, April 1921). It would be advantageous if the internationally agreed definitions could be used wherever possible.

#### Detailed Peferences to the Text

Page 35181, column 3, para. 2	We are doubtful about specifying 1996 year transit time.	
Page 35381, column 1, pars. 1	10,000 years is a speculative number and mothing is gained by stating it.	
Paçe 35232, bottom of column 2	If waste is to be disposed of on an experimental basis it is reasonable to insist on retrievability in the early stages. We do not, however, believe that 100 year retrievability should be a condition for all future repositories. We favour the alternative of storing the solidified waste for many years and then disposing of it with no intention of retrieval.	Page 15230 Page 15230
Page. 15182, column 3	Ruman intrusion, we believe that a distinction ended to and a network at a	

Page 15293, column 3

Page 35284, column 1

Page 35285, column 3 \*

Page 35296, column 1

Pige 35299

of the 3 "alternatives" at this stage we strongly favour 1, the setting of a single overall performance standard. It should then be possible to nove to 2, setting of minimum performance standards for major elements as more experience and evidence becomes available. ·. . •

-

Overall Description, we have grave doubts about the specification of 300 metres minimum depth which may be totally inappropriate at a specific site.

have value as a resource.

Page 35254, column 3, last para. We agree that there is no point in invoking a population criterion.

Definitions, we would like to see many of these altered to conform to the IAEA glossery.

"HWL" is a printing error.

Performance Objectives. This is the most critical section and we would question the justification for most of the numbers guoted e.g.:

- retrievability 50 years after emplacement is complete.
- 1,000 years containment by the wasts package.
- in 100,000 annual release of any radionuclide. Also this definition is not totally clear, is it in 100,000 of the amount of the radionuclide present after 1000 years or 1 in 100,000 in any year of that which is present at the start of that year.

Favourable Conditions. (f) (4) We are doubtful about specifying this number at this stage since its significance is very dependent on (g) regarding radionuclide transport.

 as above we are concerned that 300 metres may be inappropriate at soma sites.

Potentially Adverse Conditions. (b) the mend to "invotigate" 2 kilometrus provid and SCO metros below the limits 10 CFR Part 61, Licensing Requirements for Land Disposal of Redipartive Waste

.

## General

AFPD Building 10 AERE Harwell

Our main comment is just to note that the definition of Low-Level Waste is not consistent with the IAEA definition and that similarly the categories A/B and C and the limits quoted in Table 1 reflect U.S.A. national practice which differs in detail from that in many other countries including the U.K.

## Detailed Reference to the Text

Detailed Reference to the Te	xc	NUCLEAR REGULATORY COMMISSION
Pages 38089/38090	Definitions, we suggest that wherever possible these should be in line	10 CER PARTS 50 AND 51
	with the IAEA glossary.	COMMENTS
Paga 38097	We doubt the justification for requirin stability for 150 years.	by the
		INSTITUTE OF GEOLOGICAL SCIENCES
		LOKDON

.

institute of Geological Sciences Exhibition Road South Kensington Loncon SW7 205.

0070328 1981

•. .

B-216

Comments on Nuclear Regulatory Commission Document - 10 CFR Part/37 by the Institute of Geological Sciences

#### General Comments

The major features of the proposed rule are considered under six subheads:

- (1) Overall description
- (2) Performance objectives
- (3) Siting requirements
- (4) Design and construction
- (5) Waste package
- (6) Performance confirmation.

There is a geological component embodied in all of these subheads but before considering the detail it is pertinent to discuss two general points. The first relates to Section 60.2 Definitions (pp 35295-6). Definitions are given for "Geologic repository", "Geologic repository operations area", "Geological setting" or "site" and the "Disturbed zone" but these are amplified in later sections of the report e.g. the "disturbed zone" is rather tightly defined on p.35290 (column 3). These subdivisions of the geological environment lead to possible differentiation between the criteria necessary to define the characteristics of the "near" and "far" fields. As they stand the definitions are ambiguous, particularly in relation to the "Geological setting" or "site" which emerges as an amorphous concept of unknown dimensions and characteristics.

The second general point leads directly from the ambiguity in the definition and relates to consequential inconsistencies and variability in the degree of qualitative versus quantitative components in the technical criteria. From the geological viewpoint this can be seen, for example, in pages 35230 and 35251. Sections 50.122 and 60.123 in which the favourable and potentially adverse conditions of the geological setting are considered. Other examples are referred to below in relation to specific points. Some inconsistencies highlight admirably those aspects of the siting of a MLRW repository about which there are insufficient technical data. It can be argued that these sections of the document illustrate the strength of the case for delaying the proposed rules

until more technical data are available. The deficiency of data and the problem of the proof of the adequacy of a repository into the future is recognised throughout the text, for example at p.35288. Section 60.101 paragraph 2. Statements of this nature are entirely valid and seem to negate the technical content of associated criteria.

### Specific Comments

Reference to geological criteria occur throughout the text but they are covered in some detail on p.35289 Section 60.112, p.35290 Section 60.122 and pp.35290-1 Section 60.123 in which the required characteristics of the geological setting are considered and the potentially favourable and adverse conditions are outlined. The generalised characteristics of Section 60.112 (mandatory) are amplified in Section 60.122 which is proceeded by the statement that ..... "a geological setting shall exhibit an appropriate combination of these (favourable) conditions so that, together with the engineered system, the favourable conditions present are sufficient to provide reasonable assurance that such performance objectives will be met". There is no indication of what combination of conditions would be regarded as appropriate and some of the associated statements are so generalised as to cause difficulties in interpretation of their meaning. Most of the criteria are interactive, although this is not emphasised in the text, and many of the specific conditions are the products of a number of interactive factors.

#### Section 60.112

Pars. (a). Some quantitative statement of stability would be desirable if these mandatory characteristics are seen as significant but a stable geological setting has proved difficult to define in a generic sense. It will normally require definition in a regional rather than a global sense.

Paras. (a) and (b). "The start of the Quaternery" was probably chosen as a relatively arbitrary, but convenient time base (dated at about 2 million years ago in the United Kingdom, which was sufficiently far back in the geological record to provide a science time-base. However, there is little scientific basis for choosing that time period, rather than any other specified deriod.

Para. (b). In the land masses of the northern hemisphere there have been a series of glaciations during the Quaternary and it is totally unrealistic to speak of hydrogeologic and geomorphic stability since the start of the Quaternary.

Para. (c). The 1000 year time period appears to be related to assessments of the life-span of the engineered barriers. The pre-waste emplacement rates of groundwater movement, rather than travel times over an arbitrary period of time, are relevant as the base-line from which migration times can be judged. Any final nuclide migration models (of the types outlined in Section 60.21, p.35287 and to be included in the Safety Analysis Report) will attempt to simulate the repository and its geological setting as an entity. Accordingly, it is the groundwater travel times resulting from any perturbations caused by the repository and its waste which should be definitive. There is an inconsistency with respect to groundwater travel times between Section 50.112 para. (c) where the phrase ..... "at least 1000 years" is used, and Section 60.122 para. (f) [4] where the term ..... "substantially exceed 1000 years" appears.

## Section 60.122

Paras. (a) to (c). The phrase "nature and rates of processes" will prove difficult to define and quantify when applied to some of the processes referred to. The distinction made between the tarms "tectonic processes" and "structural processes" in paras. (a) and (b) is fine. In most situations the two terms would be regarded as synonomous and the reason for the distinction is not clear.

Para. (f) (1). The term .... "low groundwater content" is ambiguous and without a rigorous definition as to its precise meaning this condition could preclude the inclusion of many argillaceous formations and crystalline rocks which current research indicates could be satisfactory as hosts.

Para.(j) This "catch-all" condition illustrates the generalised nature of conditions (a) to (i) and emchasises the problems of drafting criteria to cover all of the interactive factors in advance of condition of the necessary research. This point is well recognised in the text (p.35234 3rd column) ..... "The impact of these characteristics on overall performance would be site specific. Thus, the Commission has judged that these should not be made absolute requirements. Presence of all the favourable characteristics icos not lead to the conclusion that the site is suitable to host a repository. Neither is the presumption of unsuitability because of the presence of an unfavourable characteristic incontrovertible". This valid concept should not be negated by earlier statements in the rules which are based on generalised concepts.

Para. (a) (5). The recognition of the timing of movements on fault planes is technically extremely difficult and in many situations it would be impossible to ascertain whether or not a fault had been active since the start of the Quaternary.

Para. (b) (1) and (2). In the cases of argillaceous and crystalline host-rocks the evidence available from opencast and sub-surface mines as well as from prior drilling, may be advantageous in that they increase the amount and reliability of information and may in no way detract from the suitability of the site. This may not be so for evaporites, but the wording should reflect the distinction. In addition the rules may well have implications on the use of existing cavities for the storage of waste.

Para. (b) (5)... Hearly all rocks have, at some time or another, suffered dissolution - it is the degree and the nature of the processes involved which are important. Most rocks are soluble under specified conditions.

Para. (b) (b).. Does this condition rafer to only one fault - it is the nature of the fault which is important, not the mere fact that it exists. In some circumstances the identification of faulting in the disturbed zona may be regarded as advantageous in that the location(s) of the potential for future movement is known.

Para. (b) (3)... Uplift and subsidence following isostatic readjustment after the last ice Age is ubiquitous in the glaciated areas and is common in much of the rest of the world for various reasons. It is the degree, rate and causes of structural deformation which are important.

Para. (5) [10]. No such correlations should be accepted as tefinitize as far as areas remote from plate boundaries are concerned.

Para. (b) (13). It is difficult to anvisage that there are not zones within any potential host-rock in which there are not reducing conditions. Excavation and entry of air into the disturbed zone will change the conditions, at least in the near-field, and any such clause will need to recognise this. Further geological aspects or the technical criteria, especially those concerned with geomechanical, geochemical and the nature and design of backfilling materials are contained in Sections 60.132, 50.133, 60.134 and 60.135 (pp 35293-35295). These are concerned with the design and construction requirements of the repository and include some very definitive statements, such as the requirements of the backfill placed in a facility.

It can be argued that it is neither proven, nor generally accepted, that such requirements are necessary and that, accordingly, these statements are premature.

## Section 60.21

The contents of the Safety Analysis Report which is required to accompany the application for the siting of a repository provide a valuable resume of the types of information necessary to describe and assess the suitability of a site for the burial of HLRU but again this emphasizes the need for completion of the research before comprehensive mandatory rules are issued.

## Conclusion

In its present generalised form the 'Proposed rule' highlights the areas in which more technolal data are required and illustrates the difficulty (and danger) of framing comprehensive technical criteria which define safe limits for complex, interactive parameters. That there will be a need for technical criteria to be defined at sometime in the future cannot be argued, but current research programmes are providing data relevant to the definition of such criteria and the publication of a definitive rule for universal application should await the completion of many, if not all, of those studies. In the meantime individual applications for site licances should be dealt with on a site-specific basis to ensure that performance objectives can be met.

• A second s

Comments on Nuclear Regulatory Commission 10 CFR Part 61 by the Institute of Geological Sciences

The report is very generalised in respect to the geological guidelines for site selection and is not dissimilar to the hydrogeological guidelines issued in the United Kingdom for disposal of controlled wastes. (The licensing of waste disposal sites, Department of the Environment, Waste Hanagement Paper No 4, HMSO, London). The concept of minimising water access into the waste by choosing an appropriate site or by site engineering is stressed, along with the need to assess the potential for flooding, erosion or seismic/ tectonic/volcanic events.

The following paragraphs consider geological/hydrogeological aspects of site selection, and are discussed in turn.

. .

. .

Para. 61.2.	The definition of "engineered barrier" is novel and usage of
	such terms ought to be compatible with document 10 CFR Part 60.
Para. 61.12.	$\sum$ This identifies the broad subject areas which need to be
ED-1	considered during site selection and includes "geologic,
	technical hydrologic, meteorologic, climatologic and biotic
	features of the disposal site and vicinity". The term
	"technical hydrologic" is taken to mean both hydrologic and
	hydrogeologic but a precise definition of the meaning is
	Crequired.
Para, 61.13.	This basically states that groundwater is probably the most
	significant pathway for radioactive transport back to the surface
	and man, and requires particular study. Although groundwater
	/ transport is very significant, recent sensitivity analyses
<b>X</b>	suggest that other pathways such as human intrusion, erosion
B-2	and vegetation uptake may be more significant (Faulth and
	Wilhite, 1980, Prediction of radionuclide migration from
	Savannah River Plant's buried waste. Trans Am. Nucl. Soc.
	Vol. 34, p.119).
	$\sum_{i=1}^{n} \left\{ \left  e_{i} \right  \in \mathbb{N} \right\}$

/ This paragraph outlines criteria for use in site salection. rara. 01.50. covering the practicability of characterisation, modelling and monitoring; mineral reserves; drainage and flooding; erosion/stability; seismicity/tectonics/vulcanism; and hydrogeology. In particular (a) (8), states that "any groundwater discharge to the surface within the disposal site must not originate within the hydrogeologic unit used for disposal". In some circumstances this may be unnecessarily restrictive and may not be required to meet the performance objectives outlined earlier in the document.

Rara. 61.51. Point (a) (6) states that the "site must be designed to eliminate GEN-1 disposal". Again this could prove unnecessarily restrictive in many circumstances and for some wastes.

# The Surety Association of America

100 WOOD AVE. S., ISELIN, NEW JERSEY (\$830 (201) 494-7600

LLOYD-PROVOST General Manager

October 9, 1981

LICENSING REQUIREMENT FOR LAND DISPOSAL OF RADICACTIVE WASTE

Secretary of the Commission U. S. Nuclear Regulatory Commission

Washington, D. C. 20550

responsible firms in the country, if at all.

would appear to be more feasible.

SAULTER REFE

Actualial Departm ROBERT G. HEPBURN, JA. Assistant Secretary GAETON SACCOCCIO anapoan Membership Service ALICE TIERNEY exercised Secretary

Surety Conarter

Secretary Patentity Departs FRANCIS X. LAMUNYCN

QUENTIN W. LERCH

AOBIN V. WELCY

Assistant Secretary-Attorney

Attention: Docketing and Service Branch

Dear Sir:

to contact us.

10210272 3:1009 2334.0

L2:VE

208

This Association is a trade association of more than 500 companies Writing surety bonds throughout the United States.

We have reviewed the Notice of Proposed Rule which appeared in the Federal Register of July 24, 1981 concerning the licensing requirements for land disposal of radioactive waste. The proposed Rule requires the licensee to provide assurances that sufficient funds will be available to carry out disposal site closure and stabilization. One proposed method to provide such assurance would be through the use of corporate surety bonds.

By their very nature these bonds would be of very long term duration

Under these circumstances the alternative methods proposed in the Rule

Somewhat similar bonds have been proposed by the Nuclear Regulatory Commission to provide assurances for the decontamination of uranium mill tailings. To our knowledge no market has developed for those bonds and you might want to check with the people within the Commission who are hanyou might want to check wird the seven to append the experiences with surety bonds. If we can provide any additional indemation, please do not besitate :231-

1. T. S. · • \* \*\*

.

.... 

and would be subject to periodic increases as the cost of future closure rises. In addition we understand the bonds will be required in substantial amounts. All of these factors lead us to success that we do not believe such surety bonds would be available except for the largest, most financially

E-1

B-220

PRIORITY MATERIAL PLEASE ANSWER IN TEN BUSINESS DAYS	August 26, 1981 Joseph H. White III 11 South Merion Ave. EI	(2) (2) 61.7 pg. 38091 (A-2) last sentance first full paragraph: What, specifically, are the mitigative measures referred to?
R.Dale Smith, Chief Low-Level Waste Licensing Branch (46FR 3808 Division of Waste Management Office of Nuclear Material Safety and Safeguard U.S. NRC Washington D.C. 20555 Re: <u>PROPOSED RULEMAILING ON LAND DISPOSAL OF LO</u> COMMENTS AND QUESTIONS, BOTH CENERAL AN	a) Doc D-52- DOC D-52- D-51- D-51- D-51- D-51- D-52- D-51- D-52- D-51- D-52- D-51- D-52- D-51- D-52- D-51- D-52- D-51- D-52- D-52- D-51- D-52- D-52- D-51- D-52- D-5	<ul> <li>3) 61.7 pg. 38091 3-2 last sentance first column; <u>How</u> will the units be seperated?</li> <li>4) 61.7 pg. 38091 B-5 : Besides depth, speciffically what barriers</li> <li>7 will be used to seperate Class C from other wastes?</li> <li>5) 61.25 rg. 38094 : Again, who is responsible for on-site wastes if a license is revoked?</li> <li>5) 61.44 pg. 38095: What is to be done if stability is not met?</li> <li>5(7) 61.52 pg. 38096 (A-6):What percent is a few percent?</li> </ul>
Dear Mr. Smith, <u>GENERAL COMMENTS</u> : (pg. 38087, sectionH.other considerations: "Many provisionsare in effect at existing dispose 1) What are the location of sites within the where I live? Please put an asterisk beside the 2) What state agencies are responsible for 2) What state agency with corresponding re where I live? List agency with corresponding re 3) Which, if any, of the above mentioned sid het are the maxium number of sites proposed: 1) Nationally 2) In the region where I live. Has a time-line for the creation of "new sites NRC yet? If yes, please forward it. On page 38068, a Memorandum of Understanding be mentioned. Please forward a copy. C-3 What amount of "migration off disposal site"	D-52 of the operational D-53 al facilities" Dese in Pennsylvania. The sites in the region the. ED s" been considered by etwoen NRC and DOT is is acceptable by NRC?	<ul> <li>(8) 61.52 pg. 38096 (A-8): 100 Feet is too small a buffer zone.</li> <li>(9) 61.53 pg. 38096 (A): Fre-operational monitoring programs should also include the local population density of the area, and the historic and predicted future movements of the population in the area. Land use of the area, e.g. industrial, farming, residential, must also be addressed.</li> <li>(10) 61.56 pg. 38097, (A-8): What do you mean by "maxium extent practicable"?</li> <li>(11) 61.56 pg. 38097, (A-8): The words, "Potential Hazard" are GAREAGE.</li> <li>Something is either a hazard, or it is not.</li> <li>(12) 61.56 pg. 38098 (3-3): What is ment by practiable? Please send the definition according to NRC jargon.</li> <li>(3) 61.61 pg. 38098 third sentance: please give an example of remonable assurance.</li> <li>(14) 61.80 pg. 38100 (h-2-i0: Any release to <u>unrestricted areas</u>, must be reported immediatly to NRC; likewise NRC must investigate any release to <u>unrestricted areas</u> immediatly.</li> </ul>
E-1 When a site is closed prematurely by NRC, due assumes responsibility for the site and waste a on pg. 38083, you state "superfund registration 1) What are you talking about? Specfic comments on <u>PART 61: Licensing Requires Radioactive Waste</u> ED-1 (1) 61.2 "Active maintenance does not include ing equipmentminor repair of disposal unit of called or refered tops? Passive maintenance?	to rule violation, who already disposed? If may be dup to the time " ments for Land Disposal of replacement of monitor- covers" What are these	This closes my comments and questions at this time. I would again like to stress the necessity of receiving these documents with haste.

• •

ī

2, 19, 20, 21, 30, 20, 51, 61, 10, 73, 1

KERR-MCGEE CORPORATION

NOTET RUNEER P

Divisionality and HEN'LA Immediately Division

August 31, 1981

Mr. R. Dale Smith Chief, Low-Level Waste Licensing Branch Division of Waste Management Office of Nuclear Material Safety and Safeguards U.S. Nuclear Regulatory Commission Washington, D.C. 20535

Dear Mr. Smith:

A check with NRC publications concerning our July request for a copy of Draft Environmental Impact Statement (DEIS) MIREG-0782 has determined that it will not be available for another one or two months which puts receipt possibly into late October.

The potentially late availability of this document makes informed comment on proposed Rule 10 CFR Part 61 concerning shallow land disposal of low-level nuclear waste by the established October 22, 1981 comment date difficult at best. As noted in the Background section of proposed 10 CFR Part 61, NUREG-0782 provides guidance and support for the rule development and is, therefore, an integral part for consideration.

Secause of the importance of the background document to 10 CFR Part 61. Kerr-McGee requests an extension of the proposed rule October 22, 1981 comment date to include an adequata review period of NUREG-0782.

Sincerely.

GEN-7



W.J. Shelly, Vice-President Nuclear Regulation and Control

WUS/enn

box John Stauzer

3110210233 811019 278 24 2 46#R58081 208





. . . .

Secretary of the Comission,

U. S. Nuclear Regulatory Commission Washington, D. C. 20555

Attn: Docketing and Service Branch

Re: Federal Register/Vol. 46, No. 142/Friday, July 24, 1981, Proposed Rules.

Dear Sir:

I have just finished reading the above listed proposed rules and am submitting the following comments with regard to some serious defectences in the manuscript.

I failed to see the logic of 20.311(h). In this section the shipper of a package is penalized because of someone else's mistakes. In the event that a trucker loses the package or in the event that the receiver fails to notify the shipper, then it is (according to paragraph h) the obligation of the shipper to conduct an investigation and file a written report with the commission regional office.

While I agree that shipments of any type must reach their destination I fail to see any logic of picking a whipping boy to penalize for anything that goes wrong. The NRC, the Department of Transportation and any other agency of the government that is involved should do do their job properly and refrain from needlessly penalizing users of radioactive materials. My final comments are in relation to section 61.5 and 61.56. I have been completely unable to establish any line of reasoning for the numbers listed in Table 1 and consequently with the class A, class B, class C, ratings

M-4

D-55-1

Letter to U.S.N.R.C. September 15, 1981 Page 2

D-55-

associated with these numbers. To merely list a sample of the confusion that I encounter, the norp total body burden for iodine 129 is 200 microcuries and the corresponding total body burden for strontium 90 is 20 microcuries. Both these concentrations refer to soluble materials and the values listed in Table 1, column 1 shows an allowable concentration of iodine only 1/50 of the allowable concentration of strontium 90 while the body burden for iodine is 10 times higher than that for strontium 90. In the event that these numbers are somehow related to a classification of hazard or danger, then a statement explaining how they are chosen would be very beneficial.

Moreover, there is no indication of any value being considered for the numbers in Table 1 with regard to the chemical composition of the radio isotope under consideration. It is my contention for example that iodine 129 in the form of lead iodide which has a very low solubility would be of less hazard than iodine in the form of potassium iodide which has a rather high solubility.

Although the number in the Table are definitely not arbitrary, a much better grouping of numbers could be obtained for such an all inclusive use as radioactive waste disposal. It is my hope that these numbers will be reconsidered and these two sections revised before forcing these items on every user-of radioactivity in the United States.

Very truly yours,

Macheil D. M. Mathews, Ph.D.

cc: R. Dale Śmith cc: Senator David Pryor

cc: Representative Ed Sethune

V to an interest SARGENT & LUNDY ENGINEERS SS FAST HONROE STREET DOCYST PURPER 201. 10. 10. 10. CHICAGO, ILLINOIS 60803 · .... Secretary of the Commission -DOCHETED Hen Docketing and Service Section CONFIC . U. S. Nuclear Regulatory Commission 12 Mashington, D. C. 20555 5 199 WOODER IN PR - Mic Nothe PROGOSED BL (Regulation Sunde Dear S Enclosed are our comments 'on; Regulatory Guide 10CFR61 Land Disposal of Radioactive Waste 2004 We appreciate having been given the opportunity to comment. Yours very truly, Second Second الحجم حصيات يسبع والمتحد J. C. Loomis, Head Nuclear Safeguards & Licensing Division JSL:DMS:sp Enclosure Copies R. F. Janecek (1/1) G. P. Wagner (1/1) NSLD File 1B-4(1/1)



#### 10CFR61 - \*LAND DISPOSAL OF RADIOACTIVE WASTE\* COMMENTS ON PROPOSED RULEMAKING

· . .

Section 61.55 - Table 1

D-55-1 D-55-1 The waste classifications scheme presented here with the associated concentration limits presented in Table 1 would have a substantial impact on the nuclear power industry's waste disposal costs and hence, upon the cost of nuclear generated power.

. . .

The concentrations given in this Table are much more limiting than necessary. In the study prepared for the USNRC by Ford, Bacon, & Davis Utah, Inc. "A Radioactive Waste Disposal Classification System." NUREG/CR1005, conservative radioactivity limits for various waste classes were established through detailed hazards analysis. The limits recommended in NUREG/CR1005 should be incorporated into 10CFR61 in place of the arbitrary values in Table 1.

## Section 61.56 Paragraph (b) (1)

D-56-B Their "stability for 150 Years" needs to be modified to indicate what forms of proof are acceptable. Some metal, wooden, and concrete structures can be shown to have maintained their "stability" for 150 years past. Very few of these are applicable to waste packaging. There is no way that deformation alone of the waste form can be a hazard to the public. The key requirement is to keep the waste from being dispersed, which is little affected by "slumping" or a "5t" deformation.

## Section 61.56 Paragraph (b) (2)

D-56-13 The provide the interval of the inter

#### Section 51.56 Paragraph (b) (3)

D-56-10

Does this mean that filler material must be added to packages containing irregularly shaped solid objects? Or is it the intent of this article that all such objects should somehow be chopped, melted, or otherwise compacted? What forms are acceptable, i.e., ash, pellets, compressed trash?

In general, the intent of these two articles is not clear and their applicability to wastes such as dewatered spent resins and incinerator ash or pelletized dryer or incinerator product needs clarification.



#### Gentlemen:

Packard Instrument Company, Inc., is pleased to submit comments on the proposed amendments to Title 10CFR as printed in the Federal Register, Vol. 46, pages 38081-38105. Our comments shall be limited to the proposed sections which affect shippers of radioactive wastes.

Cur general reaction to the proposal is favorable, in that we do not disagree with the Commission's intent. However, we do have some specific observations which we hope will be helpful.

(In the preamble (p. 38085) the Commission refers to the new \$20,306 as an example of the recognition of the need for "de minimis" classification of radioactive wastes. We agree with the need but hope that future efforts at "de minimis" classification will be more successful. In \$20,306 the Commission redefines the term radioactivity to mean (a) in excess of 0.05 microcuries... of hydrogen-3 or carbon-14,

excess of 0.05 microcuries... of hydrogen-3 or carbon-14, per gram of medium, used for liquid scintillation counting or (5) in excess of 0.05 microcuries of hydrogen-3 or carbon-14, per gram of animal tissue averaged over the weight of the entire animal. This appears to be a welcome relaxation of regulations. However, the Department of Transportation defines radioactive material as material with a specific activity greater than 0.002 microcuries per gram of material (4907R \$173.389(e)). Since the DOT regulations require use of a radicactive hazard label, the relaxed requirements of \$20.306 are null and void for any wastes which must be transported from the generator's site. We are concerned that new approaches to "de minimis" classification should avoid ambiguity and should be applicable as intended.

NRC 2 We suggest that \$61.55, Waste classification, needs some clarification. First, is it intended that the section mandate segregation of radioactive waste by radionuclide? D-55-12 This is an inference which some readers have drawn from M-1 Table.1. Segregation by radionuclide is highly impracticable and is impossible in wastes from multiple label experiments. Perhaps an expansion of, and repositioning of the footnote to Table 1 which refers to mixtures could clarify what appears to be an ambiguity. We also have some concerns about the estimation of specific activity. The solid radioactive waste which we generate is D-55-6 or plastic gloves, plastic tubes and capper, glass, rubber mixture is not amenable to analysis, therefore any statement of activity can only be a rough estimate, and the combining of ratios of mixed radionuclides will certainly increase the uncertainty. Is it intended that accurate concentration data is mandated by 161.552 In the case of liquid scintillation wastes, the estimation of M-2.4 D-55-2 Perhaps a "de minimis" classification promulgated by both the Commission and DOT would eliminate the need to consider liquid scintillation wastes as radioactive waste. We are further concerned with 120.311, "Transfer for disposal and manifests." It may be important to establish a manifest M-4 tracking system but one must be careful not to develop a system wherein the paperwork burden exceeds the benefit of (tracking a shipment. the state Several questions arise related to the requirements of \$20.311 (b), such.as, 1. person generating the waste **.** · · · · · 2. type of waste Sincerely 3. waste volume and mass 4. radionuclide identity and concentration 5. total activity M-1 What is meant by "the person generating the waste?" Do you refer to separate individuals? I suggest that an appropriate definition is available from the Environmental Protection Agency in 400FR \$250.10.(a)(25) "'Generator' means any person, by site, whose act or process produces hazardous waste ... " 1ag/ddo The phrase "by site" clearly removes the possibility that individuals would be considered to be generators. . . . • 8-225

What is meant by type of waste? Is this referring to chemical form, or a physical description such as paper, glass, plastic? I believe the desired information should be stated more clearly.

The waste volume may not be difficult to indicate if the drum size is acceptable, e.g. 55 gallons. The requirement for mass seems unnecessary and may not be readily supplied by generators who do not possessia large scale,

The requirement for radionuclide identity, concentration and total. activity evokes the same kind of comments as were offered relative to 161.55. The identity of the radionuclides can be readily listed, assuming that mixtures of muclides are acceptable. However, how do we measure the concentration and total activity of paper wipes, empty containers of glass or plastic, or contami-nated protective devices such as gloves?

Finally (20.311 (c)(3) mandates the establishment of a quality Assurance program to assure compliance with \$161.55 and 61.56. It is our belief that such a program would not be necessary at many generator sites where the low level radioactive wastes would be generated. . . . . . . . . .

This leads me back to the "de minimus" classification concept mantioned in the preamble. It appears to me that the Commission could relieve generators of radioactive waste from unnecessary could relieve generators of radioactive waste from unnecessary paperwork by identifying a concentration of activity which would be exempted from the requirements of \$161.55, 61.56, 61.57 and 20.311. However, such an exemption must be made with the concurrence of the Department of Transportation so that transportation could be made to ordinary land-fills.

It is hoped that the observations offered in this letter will assist you in the development of a practical system of licensing requirements.

amer U. Sib James A. Gibbs

Regulatory Affairs Manager

40

S. 1. 18 1.

National Aeronautics and Soace Administration John K Kennedy Spece Center Kennedy Space Center, Flonda 32899

MD-ESB-A

Barrow We Allen Of

Mr. R. Dale Smith Chief, Low-Level Waste Licensing Branch Division of Waste Management United States Nuclear Regulatory Commission Washington, D.C. 20555

#### Dear Mr. Smith:

 $M-2 \begin{cases} \text{The Biomedical office has reviewed the subject proposed rulemaking and concurs with those portions applicable to "generating" and "waste collacting" licensees. We are, however, concerned that requirements to "conduct quality assurance programs" (reference page 38102, Sections 20.311(D))(3) and (E)(5)) beyond standard management/administrative audits could be conclude burdensees to main licensees the bardie colly minimal$ could be overly burdensome to many licensees that handle only minimal amounts of low-level wastes. It is suggested that those sections specify that minimally acceptable quality assurance programs in such instances may be comprised solely of management/administrative audits.

COCKET MUNCH PR

OCT 1 5 1981

(46 FR A

.....

Paul Sychanan, M.J.

Director, Biomedical Office

MD-ESB-A/G. M. Marmaro

17 October 1981 DOCKETEB 781 00121 P255 : OFFICE OF SEL Secretary of the Commission Ling & Nuclear Regulatory Commission BRANCH Washington, D.C. 20555

#### Northern Illinois University 💆 DeKalb, Illinois 60115

Department of Biological Sciences 815 753 1753 815 753 0433

DOCKET AULTER PROZOSED RULE T (46 FR 3

Dear Madam or Sir

MASA

DOCICETED

USNAC

20 1981

This will serve as Comment on the "Proposed Rulemaking on Land-Disposal of Low-Level Radioactive Waste" (LOCFR61).

Despite the laudatory effort to improve containment, stabilization and immobilization of low-level radioactive waste in shallow land burial, neither emerience nor current studies provide assurance of an effective sequestering of toxic chemical or radioactive waste for the necessary number of years.

The mechanism for migration of radioactive materials off dump site remains poorly understood. After only four years in one trench at Sheffield, Illinois, tritium migrated 150 feet. Jess Cleveland and Terry Rees, U. S. Geological Survey (Science 212, 1506, 1981), reported in their characterization of plutonium migration in the Maxey Flats leachates, that organic matter produces stable, nobile complexes of plutonium. Radionuclides have migrated at five of the six commercial sites in this country - sometimes after less than a decade. It is essential for the NRC to dismiss land burial as an effective means of disposal, and develop criteria for the disposal of lowlevel radioactive waste in deep mined cavities in areas of low precipitation. Although medical/miversity redicactive waste represents about 25 percent D-55-14 by volume of solid waste, it contains less than 1 percent of the radioactivity

Nonmerr Blancis University is an Equal Opportunity / Affirmative Action Employee

A-8

## Northern Illinois University



Department of Energy Washington, D.C. 20545

TI 007 22 P4 59

DOCHET MUTHER PR.

OFFICE OF SECRETARY DOCKETING & SERVICE BRANCH 28

*...* 

Mr. J. B. Martin U. S. Nuclear Regulatory Commission Vashington, D. C. 20555

Dear Mr. Martin:

The Department of Energy has partially completed a review of the proposed regulation 10 CFR 61 concerning the near-surface disposal of muclear wasts. Some comments were provided Kitty Dragonette at the Low-Level Waste Management Frogram Lawiew Committee meeting in September.

OCT 1 4 1981

Our review indicates that Table 1, The Disposal Concentration Limits By Nuclide, may need further work in that the limits provided may insdvertently eliminate volume reduction. This could occur if the limiting nuclide is carbon-14 or nickal-63 in controlling the disposal category. In order to constructively comment, it is necessary to review and assess the scenarios and methodology used to develop Table 1. This information is not currently available being contained within the Environmental Impact Statement.

Therefore, I request an extension of the time limit for comment on 10 CFR 61. It would seem appropriate that the comment period be extended to 120 days from the issuing of the draft Environmental Impact Statement.

Sincerely. x A

Sheldon Meyers Acting Deputy Assistant Secretary for Nuclear Waste Management and Fuel Cycle Programs Office of Nuclear Ebergy

Sincerely,

GEN-7 Bruce W. von Zellen Professor

-, -,

Medical/university waste must be segregated at the source, stored for a matter of months, and disposed as normal trash. In summary, the problems associated with the land burial of radioactive

waste make it imperative to vigorously develop alternatives to this unacceptable means of disposal.

disposed annually of which 98.7 percent has half-lives of 60 days or lass.

. . . . .

D-55-14



.

As a licensed Radioactive Waste Disposal Service (collector), we are very much concerned with the increased paperwork requirements contained in the new manifest requirements of 10 CFR 20.311. We feel there should be some allowance made for our type of operation in the final rule.

Santlemen:

There are two provisions which appear unnecessarily burdensome for a service company such as ourselves. These are:

M.f 10 CFR 20.311 (d)(5): This requires the generator to forward (A) to us a copy of the manifest for the waste pick up that we have just made. All of our pick ups are made in our own vehicles by NDL employees. In all cases the vehicle returns to our licensed storage location. This requirement necessitates an additional page for an already multipage form, the only purpose of which is to notify us of something we already know; namely, that we received waste from a particular generator.

Our suggestion is that the requirement in (d)(5) be waived when the person making the pick up is the same as the intended scipient. The provisions of (e)(1) would still apply.

10 CFR 20.311 (e)(2): Because the waste received from one generator (due to varying burial site requirements), may be destined to be shipped to any of the three authorized sites; and because any one trailerload of waste originating at our facility may conceivably contain waste from as many as one hundred and sixty customers; it will be extremely burdensome, if not physically impractical, to include copies of originating generator manifests along with the new manifest which will be prepared by NDL. In the first case the generator manifests will contain information not applicable to the shipment in question (e.g. containers destined for a different site or some of the containers listed on one manifest not getting on one particular shipment). In the second case the paperwork for one burial site shipment could include as many as one hundred and sixty pieces of paper in addition to the paperwork prepared for a particular shipment by NDL.

While admitting that the 160-figure is a little extreme (a worst-case scenario), in practice there probably would be fifty to sixty originating generator manifests, many of which would contain nonpertinent information.

Our suggestion is that the requirement in (e)(2) be dropped, provided that copies of originating generator manifests be retained on file and all required information be included on the manifest prepared by the disposal service (collector) for shipment for burial.

If you have any questions on the comments and/or suggestions above, please contact me.

Very truly gours, Alan Jones General Manager

AJ:52

.



## 

D-55-0

Table 1 of the proposed rule IOCFR51. section 6.55 provides a specific list of radioactive isotopes and their concentration limits for each of three classifications of radioactive wastes. Proposed 10CFR20.311 (b) provides that "The total quantity of noted isotopes identified in Table 1, Part 61 of this chapter must be shown.", implying that a specific assay must be performed for each radioactive waste shipment. We do not believe this to be a practical requirement nor a technical necessity, particularly for nuclear plant wastes. Except as discussed below, we don't believe the issue is whether or not the concentration limits are reasonable, rather we believe it is an issue over the practicality of demonstrating compliance for non-gamma emitting isotopes. Any provisions which would require radioactive waste generators to do a complete assay on each radioactive waste shipment, in accordance with Table 1 of 10CFR61 and 10CFR20. 311, will as a minimum be likely to result in significant increases in personnel radiation exposure. Increased radiation exposure would result because of requirements to collect more and larger samples and because of the increase in sample handling times in order to accommodate more sophisticated analytical techniques. In addition, the transportation of radioactive waste shipments would be impacted due to having transportation casks tied up longer while awaiting completion of an assay prior to shipment; similarly the urgency for temporary on-site radioactive waste storage would be increased.

D-55-3 In the summary of the proposed rule, that portion of section V, part C, dealing with waste characteristics and classifications, indicates that the maximum concentration for most alpha emitting transuranic nuclides has been calculated to be in the range of 10 nCi/g. Further, it is stated that the calculations were conservatively based with no credit for dilution. A higher concentration limit could have been adopted but was not, since the current concentration limit of 10 nCl/g was in the spirit of ALARA and has been demonstrated achievable. We are not certain we can agree with these assessments by the NRC. First, it is not clear if the transuranic isotope concentration limit is a cumulative limit for all transuranic isotopes (except Pu-241) or if it D-55-3 is based upon a concentration per transuranic isotope.

> Second, the assay of solid Now level wasta for alpha emitting transuranic isotopes involves sophisticated analytical techniques. Some studies have in fact indicated that the concentrations of transuranic isotopes in most waste streams is low. Guidance has even been offered for assaying transuranic isotopes using indirect measuring techniques. However, while we consider these studies to be extremely valuable as an indication that transuranic isotope concentrations in reactor plant radioactive waste streams are low, we are not certain that the conclusions are sufficiently absolute. We would urge the Commission to perform an in depth evaluation of transuranic isotopes in reactor plant radioactive waste streams prior to adopting the 10 nCi/g concentration limit, and to evaluate the practicality of techniques for demonstrating compliance. We have seen studies which suggest that the concentration limit for transuranic isotopes could be as much as fifty times higher.

OCKETED INNRC

nct 23

DCCRET NUMBER DR

PROPOSED RULE MK.

32

(46 FR 3808

October 21 JOSLCRETARY OFFICE OF SELERETARY DOCKETING & SERVICE BRANCH

WISCONSIN Electric Power Company 231 W. NICHIGAN, P.O. BOX 2046. MILWAUKEE, WI 53231

48. MILWAUKEE, WI 53203

Mr. Samuel J. Chilk Secretary of the Commission U. S. NUCLEAR REGULATORY COMMISSION Washington, D. C. 20555

Attention: Docketing and Service Branch

Dear Mr. Chilk:

M-

## - PROPOSED RULEMAKING ON LAND DISPOSAL OF LOW-LEVEL RADIOACTIVE WASTE

.

We have reviewed the proposed rulemaking on land disposal of low-level radioactive waste which was published in <u>The Federal Register</u> on July 24, 1981 and have the following comments.

Although the major impact of the proposed regulation will fall upon disposal site operators, the additional costs resulting from the additional requirements undoubtedly will be passed on to waste generators. These additional costs to the site operator include new administrative and operational costs, and the provision of financial surety.

One of the new requirements which would apply to waste generators and shippers directly concerns a new manifest system which would require waste shippers to provide more information in shipping papers and to investigate missing shipments. Existing Department of Transportation regulations pertaining to shipping papers and reporting requirements are adequate. Furthermore, it is more appropriately the carrier's responsibility to investigate shipments.

Another new requirement to be placed on the generator/ shipper concerns waste categorization through the use of Table 1 of Section 61.55. There are several difficulties with this (table: a) it is incomplete, i.e., there are many nuclides which are not included or provided for; b) the 0.1% limit for chelating agents is unnecessary and is unclear as to whether this is D-55-2 percentage by weight or volume; c) the limits on concentration D-55 are too restrictive, especially the 10 nCi/g limit for transuranics, and the technical justification for these limits has not been (demonstrated. We note in particular that the 10 nCi/g limit,

-1-

#### Mr. Samuel J. Chilk

October 21, 1981

D-55-3 {although historically implemented on an arbitrary basis, continues by be promulgated without justification. A new limit should be derived, based on technically sound and justifiable analysis.

-2-

M-4
On Page 38088 of <u>The Federal Register</u>, the NRC certifies that the proposed rule will not, if promulgated, have a substantial economic impact upon a substantial number of small entities. We do not agree with this judgment in view of the additional paper-work involved with the Manifest Tracking System and the increase in burial fees which would result when the burial site operators pass on their additional costs. The economic impact will be felt by all of the approximately 9000 licensees, and it will be felt most strongly by the "small entity" licensees who comprise all but a few hundred of the total number of licensees. In the summary, the NRC acknowledges that the economic costs of the proposed rules to "small entities" have not been quantified, and the burden to prove otherwise is placed on these small entities. Many of the small entities are physicians, medical clinics, and hospitals who are currently permitted to dispose of radioactive wastes into sanitary sewage systems. The proposed rule would greatly increase their disposal difficulties with no benefit to the public health and safety.

The new regulations are inappropriate, unnecessary, and provide no benefits for the additional costs that will be incurred.

Very truly yours,

#### C. W. Fay, Director Nuclear Power Department

. Mailing Address Alabama Power Company 600 North 18th Street Post Office Box 2641 Birmingham, Alabama 35291 Telephone 205 783-6081 DOC: NO FO Mr. Samuel J. Chilk October 13, 1981 F.L. Cleyten, Jr. U. S. Nuclear Regulatory Commission Pace 2 Senior Vice President Flintnige Building Alabama Power resource = 01,23m P1 23 an Section 61.55 Waste Classification October 13, 1981 The waste classification system should not prevent shipment of DOCKETING & SERVICE mixed classes of wastes on one truck. As it is now written, 10 CFR Part 61 only states that Class A wastes will be segregated . . at the disposal site and nothing is said about Class B and Class C Docket Nos. 50-348 wastes. It appears that mixed shipments are allowed but it is 50-364 not explicitly stated. An example of the problem caused if mixed shipments are not allowed is the following. Some waste containers are designated Class B and require a shipping cask. Since a D-52-1 shipping cask holds many waste containers, several Class A containers Mr. Samuel J. Chilk NOTES NUMBER PR. are used as fillers to help assure the safe transportation of all Secretary of the Commission the containers. If the containers are not segregated at the U. S. Nuclear Regulatory Commission RULE disposal site, then all of them would be disposed of in a Class B 1717 H Street, N.W. burial trench. This would result in unnecessary costs to the Washington, D.C. 20555 shipper and misrepresent the actual volume of Class B wastes being disposed of. Therefore, 10 CFR Part 61 should clearly Cear Mr. Chilk: stated that all waste containers will be segregated at the disposal site or that they will not be segregated, whichever is true, Re.: Joseph M. Farley Nuclear Plant - Units 1 and 2 Comments on Proposed 10 CFR Part 61 Section 61.55(d) states that waste which has a radioisotope "Licensing Requirements for Land Disposal of Radioactive Wastes" concentration in excess of Column 3 in Table 1 is not generally Lacceptable for disposal without specific Commission approval. The In response to the Federal Register Notice dated July 24, 1981, we D-55-3 value given in Column 3 for alpha-emitting transuranic radionuclides would like to offer some comments concerning the NRC proposed rule 10 CFR is 10 nCf/g." This value is based on current DOE requirements which Part 61, "Licensing Requirements for Land Disposal of Radioactive Wastes." were derived from conservative calculations. However, to our know-Although the proposed rule is directed at licensing low-level radioactive waste burial grounds, certain parts of the rule will directly impact the ledge, it has not been demonstrated that the detectable limit for alpha-emitting transuranic radionuclides is down to less than packaging and processing of low-level radioactive waste at Farley Muclear 100 nCi/g. Therefore, a 10 nCi/g limit is not measurable and thus Plant. Our comments address the packaging and processing concerns raised not enforceable. by the proposed rule. Another concern with Table 1 is a footnote which states, "Wastes "de minimis" Classification of Wastes containing chelating agents in concentrations greater than 0.12 On page 38085 of the Federal Register Notice containing proposed In page Socos of the rederal keyster Notice Containing proposed 10 CFR Part 61 (Vol. 46, No. 142 dated July 24, 1981), it is stated that Part 61 will not establish a generic "de minimis" category for waste (i.e., wastes that would be exempt from Part 61). Recently the NRC set "de minimis" limits for tritium and carbon-14 in liquid scintillation and animal carcass waste (10 CFR Part 20). are not permitted except as specifically approved by the Commission." Since cleaning fluids, decontamination cleaning compounds, etc. contain chelating agents, the 0.1% restriction would require a-D-55-2 multitude of specific approval requests for Commission review. It D-56-2 would be much better to establish a waste characteristic requirement in Section 61.56 for chelating agents such that these wastes The new limits on tritium and carbon-14 allowed a significant could be handled routinely. Since decontamination activities will reduction in the volume of low-level radioactive waste being become more and more common as nuclear power plants age, the shipped from hospitals and research facilities. Since a large chelating agent disposal issue should be addressed generically in fraction of the currently shipped low-level wastes from power 10 CFR Part 61 rather than on a case-by-case basis. reactors are not hazardous to the health and safety of the public due to their very low levels of contamination, it would be very Section 61.56 Waste Characteristics beneficial for "de minimis" limits to be specified in 10 CFR Some of the requirements in this section appear overly restrictive Part 61. Such limits could significantly reduce the volume of .D-576 -11 and could lead to significant increases in the waste volume. In low-level radioactive wastes being shipped to burial sites and particular, section 61.56(b)(1) requires that the waste form be thus reduce the need for new sites. Therefore, the NRC shouldable to withstand a compressive load of 50 psi while maintaining speed up its current schedule (i.e., two years plus) to establish "de minimis" limits for radionuclides and incorporate those limits into 10 CFR Part 61.

Mr. Samuel J. Chilk October 13, 1981 LAW REGISTERING TERTING COM U. S. Nuclear Regulatory Commission Page 3 2749 DELK ROAD, S.E. IGIA 3000 physical dimensions within 5% in the presence of moisture, 17.1.1.H.H.BER microbial activity, and with internal factors such as radiation effects and chemical changes. Since this requirement is part of D-56-9 PROPOSED RILLE TH October 20, 1981 section 61.56 which concerns stability of the waste for at least 150 years, it is implicitly assumed that the compressive load/ dimensional integrity requirement must be met for at least D-56-8 DOLTETED 150 years. The basis given for this requirement is to prevent Secretary of the Commission slumping, collapse, or other failure which could lead to water U.S. Nuclear Regulatory Commission infiltration. Washington, D.C. 20555 81 0CT 23 P1 23 Subsidence (i.e., slumping, collapse, etc.) is caused by void spaces in a burial site. Section 61.52(a)(5) addresses the void concern by requiring that the void spaces between waste packages Attention: Docketing and Service Branch, Comments on lOCFR, Part 61, Branch as Published in the <u>Federal Registar</u>, be filled with earth or other material. Therefore, the compressive Subject: load/dimensional integrity requirement is overly restrictive and could significantly add to the volume of low-level radioactive July 24, 1981 waste by requiring additional processing of the waste to meet this requirement. The following comments address limited sections of 10CFR Part 61, as solicited in the Nuclear Regulatory Commission letter, dated August 4, 1981, Subject: Proposed Rulemaking If you have any questions concerning these comments, please advise. on Land Disposal of Low-Level Radioactive Waste. • ', ' Yours very truly, i a the second enerterneti 1942red (d. 1982re) (1) ITEM: TYPOGRAPHICAL ERRORS SUMMARY - V. C., last three words in Column 1, ~~ \*\*;2:..\*.\*\* page 38085 - through "te" exclusion, should read, through "the" exclusion . . • ED-1 FLCJr/GGY/de cc: Mr. R. A. Thomas 61.53(d) "incidate", should read "indicate" . Mr. G. F. Trowbridge - , Mr. J. P. O'Reflly (2) ITEM: See Section 61.52 (a) (8), in regards to the 100ft. Mr. E. A. Reeves buffer zone. Mr. W. H. Bradford COMMENT: the requirement for a 100ft. buffer zone may be better served from a performance objective approach. What is the purpose of the 100ft. buffer, to prevent intrusion, off-site contamination, or both? If intrusion is the criterion, the 100ft. buffer is probably sufficient; if off-site contamination is the criterion, then the buffer width should be performance based (e.g., a function of transmissivity, waste classification, Rd, distance, time, etc.). If both intrusion and off-site contamination are the concerns of 61.52 (a) (8), then 1002t. should be de-fined as the minimum distance and, if necessary, an addative factor superimposed to accomodate the performance objectives. ÷

, **x** 

The second second

Secreatary of the Commission October 20, 1981 Page Two

(3) ITEM: (See Section 61.56(a), in regards to stipulating minimum requirements for all classes of waste.

COMMENT: the next section 61.56(b), does not EDspecify if any, or all classes of waste must meet the stability requirements.

(4) ITEM: See Section 61.56(b), 2nd sentence, in regards to the need for assurance that the waste will not degrade.

COMMENT: the waste will degrade. The intent of the sentence would be better served without requiring natural processes not to degrade an environmentally unstable material. One can require the waste binder or container D-56-8 to last 150 years, but not the waste itself. The waste should be allowed to degrade, the binder and/or containers could be required to withstand <u>disintegration</u>, thus avoiding soil collapse and subsequent infiltration.

Law Engineering Testing Company appreciates this opportunity to comment on this important document. We look forward to the publication of 10CFR Part 61.

Sincerely,

LAW ENGINEERING, TESTING COMPANY

X a

Louis S. Karably, P.E., P.G. Manager, Radwaste Isolation Program

LSK/mgd

cc: F. J. Steinbrenner



OOCKETED USNRC

OFFICE OF SECRETARY DOCKETING & SERVICE BRANCH

DOCKET HUMTER PR - 2 et al

'81 OCT 26 P2:55 October 22, 1981

Secretary of the Commission U.S. Nuclear Regulatory Commission Washington, D.C. 20555

Attention: Docksting and Service Branch

(46 FR 38081) Subject: Proposed Rulemaking on Land Disposal of Low-Level Radicactive Waste (46 FR 38081 July 24, 1981)

Dear Sir:

Commonwealth Edison has reviewed the subject rulemaking and offers the attached comments. We appreciate having been given the opportunity to comment.

Sincerely,

L. O. CelGeorge Director Nuclear Licensing



#### Comments - Commonwealth Edison Procosed Rulemaking On Land Disposal Of Low-Level Radioactive Waste (46 FR 38081 July 24, 1981)

#### DEVELOPMENT OF PERFORMANCE OBJECTIVES - PROTECTION OF THE INADVERTENT INTRUDER

This section lacks adecuate provision for the protection of the general public from potential releases to the environment.

For example, while terrorists cannot be accurately described as "inadvertent" intruders, we believe there should be a plan to safequard a site from terrorists. While security staffs at nuclear power plants have grown in proportion to the threat of potential covert actions of terrorists and extremist groups, there is no reference to security or anti-terrorist possibilities at the proposed land disposal sites. This could be due to the belief that site disposal security is not necessary. when, in fact, disposal sites are more suscepticle to covert acts than nuclear power sites. This is true because the nuclear waste materials are already packed in containers in preparation for transport.

1 2

. .

#### SUEPART A

1 -

C-1

### 61.7-0.4 INSTITUTIONAL CONTROL

A period of 100 years is too long to be meaningful. As a requirement, it would be impossible to meet unless the licensee could provide some kind of assurance of institutional control for 100 years. If left in the regulation, it could become a stumbling block to the licensing of low-level waste sites.

COBALT 60, the material expected to be present in any D-59-21 quantity, has a half life of 5.3 years. 100 years, as well as the SCO-year passive control, therefore appears arbitrary. It is tecnnically incorrect to assume that carbon steel containers, which are bio-degradable over the long term, will resist degradation for the 130 year period. The proposed changes to 10 CFR 61 imply a potential change in the container industry from carbon steel to nonbib-degradable materials such as high density polyethylene. This chance would be financially prohibitive.

SCEPIRT 5 - GENERAL PROVISIONS 51.13K TECHNICAL ANALYSIS

A systems approach should be used in the regulation of ic -level waste burial sites. This means that criteria for compliance at the site boundary should be established and arbitrary intermediate levels of radioactivity on packages, leach rates, etc. should not be part of the regulations.

## 61.55 WASTE CLASSIFICATION

There are several radium 226 sources used at Commonwealth D-55-4 Edison Fossil Stations for flow rate determinance. Radium 226 is (not specified in Table 1.

Chelating agents are not permitted in concentrations above 0.1% without NRC approval. This is overrestrictive, and may needlessly impact chemical cleaning operations planned for Dresden D-56-2 Unit 1. Additional comments regarding this impact may be submitted after further study.

#### SUBPART E 61.62 FINANCIAL ASSURANCES

This section contains no details with respect to the amount E-1 of financial assurance that each disposal site licensee will be required to obtain. Commonwealth Edison's concern is that prohibitive amounts will force present licensees out of business and result in a lack of low-level waste disposal sites.

#### 2710N

B-235

-3-

. . .



DOLXETED USHRC

.

.

81 0CT 26 P2:53

OFFICE OF SECRETARY DOCKETING & SERVICE BRANCH

BOCKET NULLEER PR-22+ a

(46 FR 38081

October 21, 1981

Secretary of the Commission U.S. Nuclear Regulatory Commission Washington, D.C. 20555 Attention: Docketing and Service Branch

Dear People,

re: Proposed Amendments (published 21st July 1981) to the Commission's rules regarding Licensing Requirements for Land Disposal of Radicactive Waste

The attachment contains comments on the proposed amendments.

Thank you for your attention.

Sincerely, . esan Gordon Thompson, Ph.D. Staff Scientist

Attachment

COLKETED USNRC • : - - -

81 OCT 26 P2:53

<u>Comments on</u> <u>Proposed amendments (published 21st-July 1981)</u> <u>to the NRC's rules regarding licensing</u> requirements for land disposal of radioactive waste

by

Gordon Thompson Staff Scientist Union of Concerned Scientists 1384 Massachusetts Avenue Cambridge, Massachusetts 02238

October 1931

1264 Massachusers Avenue + Dartor dge Massachuserts 22226 + Telechone (617) 547-5552

1725 : Breer N.W. + Suce SCT + Washington C C 20006 + Telectore (200) 256 4600

## -2-61.13 Technical Analyses Analyses of release pathways should be prepared in such a manner that they may be validated by data acquired from sub-77-53-1 sequent monitoring. Monitoring should be conducted with this end in view. Validation should be required at set periods. 61.41 Protection of the general population from releases of radioactivity This Section states that ground water at the nearest public drinking water\_supply must meet National Primary Drinking Water standards. That provision should be extended to all actual or potential water supplies outside the site boundary. 61.42 Protection of individuals from inadvertent intrusion This section should be expanded to include individuals and populations.

The amendments should specify maximum individual doses and maximum population doses, in the event of the specified intrusion or of more severe intrusions (see our comments on 61.7).

61.52 Land disposal facility operation and disposal site closure At (a)(6) in this section, it is stated that gamma radiation D-52-5 /must be within a few percent above background. This requirement ( should be replaced with a more precise requirement.

At (a)(7), it is stated that three permanent survey marker D-51-2 control points must be established on the site. In addition to that requirement, there should be a requirement for warning signs with a 500-year design life (see our comments on 61.7).

#### General Comments 1.

The proposed amendments in many respects provide an appropriate regulatory framework for near-surface disposal. In certain respects, the amendments are imprecise or inadequate as outlined in the following specific comments.

- 2. Specific Comments
  - Part 61

## 61.1 Purpose and Scope

The amendments should be confined to near-surface disposal until requirements for other methods are formulated. 61.7 Concepts

This section proposes ((b) (2)) a maximum site inventory D-55-10+ for certain isotopes. Criteria for determining this inventory should be specified in these amendments. Furthermore, a maximum site inventory should be determined for every isotope This section proposes ((b)(3)-(5)) provisions to guard against inadvertent intrusion. The nature of this intrusion should be specified in these amendments.

Adoption of a maximum site inventory for all isotopes would provide some protection in the event of inadvertent D-55-105 intrusion of a severity greater than that specified or in the event of other circumstances which might lead to unanticipated releases of radioactivity.

No provision is made in this section, or elsewhere in these amendments, for warning signs designed for a life equal D-51-24 to the 500-year design life of the intruder barriers. Such provision should be made.

-3-DOCTOR HUMAN PR-2 FR 38081 (46 61.53 Environmental monitoring At (a), it is required that certain data must cover at 3164 Main Street | Buffaio, New York 14214 (716) 832-9100 least a twelve month period. This period is much too short October 23, 1981 for the appropriate collection of many kinds of environmental Dear Sir, data. This section should specify the relation of monitoring to Herewith find enclosed our comments on Licensing validation of analyses (see our comments on 61.13). Requirements for Land Disposal of Radioactive Waste. These su comments are being submitted on October 23, 61.55 Waste classification A maximum site inventory should be determined for every one day after the close of the comment period October 22. I hope that they still will be taken into isotope (see our comments on 61.7). GEN-7 In Table 1, the concept "theoretical maximum specific activity" consideration and would appreciate confirmation of is used. A definition of this concept should be included in this fact from you. these amendments. Table 1 should be extended so as to stipulate maximum con-Thank you for your interest and consideration. centrations for every isotope which may be included in radioactive waste. The suggested procedure for comparison of the isotope Sincarely your: with Sr-90, Cz-137 or U-233 is imprecise. lina Hamilt 61.55 Waste characteristics Director This section should state that waste containing chelating Sierra Club Radioactive Waste 00T 26 P2 53 agents in concentrations greater than 0.1% are not permitted DOCHETER (as is now stated in the footnotes to Table 1).

D-53-1

ED-1

D-55-10

D-56-2

B-238

3164 Main Street Buffalo, New York 14214 (716) 832,9100

COLAINTS ON PART 61 "LICENSING RECURRITIONS FOR LAND DEAR BAD OF RADICACTIVE RASTE." Submitted by Mina Hamilton, Director, Sierra Club Radioarchyge Waste Campaign (2)

61.52 Facility Operation

D-52-3

4) No matter how "orderly" the emplacement package integrity cannot be (46FR38) maintained during emplacement and disposal -- steel drums and wooden crates all degrade at different times. Even if identical steel packaging were utilized degradation of the steel will occur at different rates according to type of manufacture and contents. Slumping and shifting of packages is inevitable with the consequent slumpage and caving in of trench caps.
5) Future subsidence cannot be significantly reduced by filling void spaces at emplacement or time of durping. The rotting, corroding and deteriorating of different package materials (steel, wood) will inevitably produce voids subsequently resulting in slumpage problems.

61.56 Taste characteristics

(5(b) "The requirements ... are intended to provide waste stability for 150 years." The assumption that waste will not degrade or slump for over 150 years." The assumption that waste will not degrade or slump for over 150 years." The assumption that waste will not degrade or slump for over 150 years. Steel drums 151 corrode and rust out in 30 to 60 years. Steel drums exhumed at the 152 Idaho dump site (INEX) were all seriously degraded after 13 years. The 153 symption that steel will not corrode has no basis in experience. 15 is unclear as to how the goal of stability for 150 years was arrived 25. Thuch of the material to be dumped is contaminated with design and

> strontium which will require stability for at least <u>300 years</u>. (This is presuming the industry rule of thusb of multiplying it x 10.) If prior to this time, trench collepse, since end erosion, or other events

> > slerra club radioactive waste campaign

provided an access route for water into the trenches, these hazardous isotopes could migrate off site. In the case of cesium we are particularly concerned because of its water solubility.

-2-

(2) It is highly commendable that the NRC has finally banned use of extremely 56-4 temporary cardboard and fiberboard for rackaging of materials that will be hazardous for hundreds of years.

(5) Should be expanded to include persons living downwind of the site. Although persons "transporting, handling or disposing of the waste" can be protected from gases or fumes by protective clothing and masks, citizens living or working in the vicinity need a more specific guarantee regarding harmful airborne contaminents such as radon gas or tritiated methane.

61.52 Near-surface disposal facility operation and disposal site closure. (a)(2) Apparently, the requirements for stability for Class 3 wastes will not apply to Class A waste. Given the presence of such isotopes as cesium and strontium in Class A waste, we find the lack of such requirements is inconsistent with the over-all goal of protection of public health and safety.

> (4)(5) There is no basis in experience for assumption that "orderly" or neat stacking has any effect on long-term package or trench integrity. The action of neat stacking is correlated to the expected lifetime of mood or steel. No matter how perfectly steel drums are stacked, they will eventually corrode. If provisions 61.52 (a)(4)(5) or 61.51 had a basis in reality, the trenches at West Valley would be stabile. But within 2 years of closing over 126,000 gallons of water had accumulated in one trench at West Valley. The origin of this ster is unknown.

-2012

SUPPART C - PERFCEMANCE CEJECTIVES

61.44 The goal of long-term stability and avoidance of continued, active maintenance at the site is central to Part 61 of GFR 10. The goal is, of course, enthusiastically shared by environmentalists, industry and government. But there is no basis in experience with burial sites located in regions of moderate to high rainfall that indicates this goal can be achieved by the means outlined in Part 61. Unless the recommended measures have been <u>demonstrated to work</u> at an already existing facility, then the measures are examples of wishful thinking and not concrete tactics whose supplementation will result in the desired goal. To establish regulations on the basis of a hoped-for future dream and not on the basis of actual experience is a difference process that may hull the unwary into a false sense of security.

-3-

The goal of stability supposedly will be achieved through specific site design features. For example, in 61.51(a)(4) "Covers must be designed to prevent water infiltration." Thus far, over 12 years of experimentation with increasing the size of covers from 4" to 6", with using impermeable clay as the material, with planting of special grasses at the Test Vallay dump site has not prevented water infiltration. Mater infiltration continues to be a problem at the closed-down Sheffield, Illinois, Maxey Flats, Ky sites and the still operating Barnwell, S. C. facilities. All operating connercial dump sites except  $\frac{1}{2} exc. feel}$  rainfall) have water infiltration problems. Twenty years of experimentation and we seem to be no closer to prevention of water infiltration through the covers. Perhaps concrete covers would work but this approach has not been demonstrated to work. An obvious problem with concrete would be the difficulty of maintaining the cover in place as trench contents degrade, shift and slump downwards leaving a void for the cover to collapse into. The only way to keep such a cover in place would be to have concrete walls suck into the ground to hold the roof up. Th addition, george the lifetime of this concrete would not reach the goal of 150 years.

C-6

There is an apparent misunderstanding in Part 61 regarding the simpler inevitable mechanism of slumpage of trench contents. Without this understanding--that slumpage leads to cover indentations, cracks and collapse-then it is possible to make the mistake that cover "design" can solve the problem. Likewise\_there is a fundamental misunderstanding that slumpage can be avoided by techniques of placing materials into the trench or by packaging of materials. Slumpage is intrinsic to placing materials that degrade (steel) in a matter of decades into a hole for 150 years. See page 2,1.

Unless the relationships are clearly seen between gradual deterioration of packaging, slumpage, trench cover collapse and water infiltration, no meaningful design criteria can be established.

SUPPART D - TECHNICAL REQUIREMENTS FOR LAND DISPOSAL FACILITIES (61.50 Includes a number of laudatory goals. We are disturbed however by 61.50 (a)(5) which does not absolutely exclude areas of frequent ponding and flooding, only stating the area should be "generally free" of such characteristics and which states that burial in up to a 100-year flootplain is excluded. Since the design goal is stability for 150 years, obviously

B-240

-4-

· · -- 6---5-Regular municipal landfill if there is certainty that full decay has occured. an area of more frequent floods should be excluded. We have major problems D-50-3 D-55-21 This would significantly reduce low level with the 150 year design concept (see below) and think burial should be dump site volume requirements 17-55-14 (currently 25% of total low level waste volume is provided by medical excluded in a 500 year floodplain. wastes) and involve a more rational segregation. (61.50 (a)(7) should specify the depth to the watertable which is acceptable. If anything the proposed system is going to increase volume requirements "Sufficient depth" to the watertable is so imprecise as to be useless. because the temptation of waste generators will be to package their waste so 2-55-8 D-50-2 as to fit into the category of Class A and thereby avoid more stringent (and there should be no exceptions to the specified depth regardless more expensive)burial site characteristics of Class B waste. of rates of diffusion. 61.17. The concept of segregating different wastes in order to better protect . the public health and safety is an excellent one. However, the method of classification proposed will not lead to better protection of the public. D-55-The method of classification should not be based simply on number of curies of insterial per gram. The classification should be based on the half-life of the material. To mix together cesium and structium contaminated wastes with Ht of 30 years and a necessary period of 300 years isolation from the biosphere, with isotopes such as technicium 99 M and Portine with it 2 respectively of 6 hours and " days and a sequestering period of perhaps 3 months is totally irrational. But is it possible to separate these wastes? prist of ous in, The easy to implement mechanism would be to separate these wastes at the In particular, this is feasible at medical hospitals and research institutions. If wastes of under it of 190 days were segregated, 75% of hospital wastes would fail into this category and could be stored at a warehouse on or near the D-55-2 hospital complex, until safe levels had been reached; then 755 of the volume medical D-55-1 ofyrestes going to radioactive dump sites could be eliminated and sent to a

B-241



UNION CARBIDE CORPORATION MEDICAL PRODUCTS DIVISION PO. BOX 324, TUXEDO, NEW YORK 10987

DOCKETED

## OCT 26 P2:54 CKETING & SERVICE

31

OCCNET MULISER

PACTOSED RULE MA

October 21, 1981

Secretary of the Commission U.S. Nuclear Regulatory Commission Washington, D.C. 20555

TELEPHONE: \$14-151-2131

Attn: Docketing & Service Branch

#### Gentlemen:

The following comments are submitted for your consideration to revise the proposed rulemaking regarding the Licensing Requirements for Land Disposal of Radioactive Waste, as published in the Federal Register on July 24, 1981.

## Timing

The Low Level Radicactive Waste Policy Act sets a definite time when regional compacts must be established. This time table in turn places requirements on the individual regional burial sites to make application B-3 to the NRC for approval. The Commission should commit to a timing for. response to an application in their regulations such that the applicant has some degree of assurance that the statutory deadline can be met.

### Ceminimus Levels

The treatment of low level radioactive waste problems is incomplete Without addressing deminimus levels of wasts. Part 61 should address D-55-2 this issue.

## Cecay In Storage

It is increasingly common for nuclear facilities to package recidentive GEN-3 time prior to shipment to the burial grounds. At our site we are just completing a shielded waste storage area which will allow storage of fully packaged 55 gallon drums of waste for up to one year prior to shipment. In the spirit of the ALARA concept we believe that any new packaging or labeling requirements resulting from this process rule exempt materials packaged prior to the effective date of the regulation so they may be disposed of properly.

Summary of Rule

## Para V.G

5

Any license issued by the commission should be for a period of time essentially equal to the expected life of the burial site. An inference B-5 is made in the statement of consideration to a five year license term. This is inconsistent with the nature of the facility and invites public hearings and intervention with each license renewal stage.

## Para V.H

The cost impact of the proposed regulation is not addressed quantitatively. Waste disposal is now a significant (greater than 10%) cortion of cur operating cost. Disposal costs over the past two years have incresed at a far greater rate than all other cost elements. We D-55-11 enticipate that the enactment of these regulations will cause a continuation of this trend. This issue is particularly important to waste cenerators who must compete in world markets with competitors who are not subject to the same criteria. It is an obligation of the commission to determine the potential cost impact on waste disposal sites and on waste cenerators.

Manifest System - 20.311(b) and (c)

The manifest system, when established, should be universal and acceptable to all disposal facilities, cenerators and regulatory authorities. Documentation of radioactive wastes has become a significant part of the cost of waste disposal.

## Intruder Protection - 61,42

Even after the institutional surveillance period has expired, there should remain in effect reasonable assurance that intruder protection is

in place. This protection could be provided by such basic means as costing of property, restrictions recorded on deeds, atc. Because of the residual protection remaining after the institutional surveillance period, it is overly conservative to treat the intruder with the same level of protection as a member of the general public is assured; i.e., no more than 500 mR per year. A more rational basis would be a number at least 10 times greater, that corresponding to the limiting safe scoupational exposure of 5,000 mR per year.

## <u>Site Design</u> = 61.51(a)(7)

ED-1 The wording seems too specific. We recommend; "THE DISPOSAL SITE SHALL E USED EXCLUSIVELY FOR THE DISPOSAL OF WASTES CONTAINING RADICACTIVE "STERIAL".

.. .

13434

• 1

## <u>Site Closure</u> - 61.52(a)(6)

D-52-5 Following our comment on Section 61.42, we recommend the following wording; "WASTE MUST BE PLACED AND COVERED IN A MANNER THAT LIMITS THE CAMMA RADIATION AT THE SURFACE OF THE COVER TO THE LIMITS IN 10 CFR PART 20".

Limit For Transuranics - 61.55 Table 1, Col. 3

D-55-3 This is an appropriate time to address the issue and establish a tecnnically based limit for transuranic wastes. By failing to generate a reasonable technical basis, the regulation makes a de facto endorsement of the existing limit. The justification for not addressing the issue is stated as "the lower value of 10 ncl per gram has been demonstrated as an achievable concentration to control the disposal of transuranic nuclices". This statement is not true because there have been occasions when wastes could not meet this limit and therefore are not acceptable to control.

<u>Site Criteria</u> - 61.56(b)1

D-56 - 11 The performance criteria for compaction is a function of site conditions and trench depth. We propose site specific limits for compressive strength based on local soil conditions.

D-52-1 The proposed regulation would have Class A waste segregated from Class B. It should be recognized that Class A material would form a hood shielding buffer from the Class B material which would cenerally be a more racioactive wasta form. Hy segregating the Class B material, burial trenches may be fulled with dirt simply to provice shielding whereas the Class A material could have been used for that shielding and resulted in better land use.

D-56-9 The proposed regulation requires a structural stability of Class B containers to maintain the physical dimension of a waste package within 5%. This should more appropriately be 5% of the volume, since a 5% deflection in the diameter of a drum is inconsequential.

Presently, there are few, if any, waste containers that will qualify as a stable container under the proposed rule. Until such a container has been proven we believe that the regulation should offer the alternative of percetual care to correct the effects of settling or other manifestations of instability.

In general conclusion we observe that the current trend of radicactive waste disposal is to reduce volume by compacting or condensing to higher concentrations. This trend is occasioned by regulatory quotas and economic necessities. It is apparent that this trend will continue as swidenced by the emergence of incineration of low level compatibles. The development of other technologies may follow. Considering this trend we can forsee the possibility of a large part of all low level waste approaching the class B & C categories. This becomes even more likely if local authorities decide to scale down the Federal limits. Therefore, any sound basis to relax these criterior should be advanced.

Thank you for your consideration.

Very truly yours,

James J. McGovern Business Manager Radiochemicals

JJMcG:js





Hr. R. Galle Smith \_\_\_\_\_ October 19, 1981 \_\_\_\_\_ Page 3

34J:137:11

D-51-1

F-1

8. Section 61,61(4) requires that trench covers be designed to <u>movent</u> water infiltration. While it is containly desirable to <u>administer</u> water infiltration, it is abaot impossible to completely prevent any infiltration. This same comment also applies to 63,63(b)(6), which requires elimination of contact with the water by water.

Section 51.72. Must criteria are to be used to determine whether a State or tribe is potentially affected by a proposed site? It is cortainly resonable to encourage review of a license application by the State or tribe on whose Tunks a proposed site is located. Extension of review participation beyond this could mable a neighbor or more distant State through which waste will be transported Not only to incorrence but to review and supress approval/disapproval for a site. This could create a treatmoos subjustrative burden and cost increases for both the NCC and the applicant.

Thank you for the opportunity to commant on both draft and final versions of the proposed rule. We are looking formers to the release of the final regulation and will do per state to see that it properly implemented.

Sincerely,

DIEN-HULLEAR SYSTEMS. INC.

Broca H. Johnson Chairman and President 14 Jacques Leel Road Woods Hole, Ma 02543 October 18, 1991 -Secutory Low-havel Weste Licensing Branch

Division of Waster Management Nuclear Regulation Consumision Washington DC 23555

Atention: Docketing and Service Manch Re: 4/24/31 FR Notice Proposed Reks Dear Sin:

GEN-9

COURSE MULSER PR - Jula

(46 FR 38081)

theme you for earling we acopy, without my hering to request it. I arrived while I was preparing to here I consider and response can vary consuling to works there. I have in teach and response can vary coording to whether I am here on twice. I supected to real guidely but shas I the heat of the outget in the maintiplese thes proceeding the Careful Hair, 9/21/81, Oceansgraphing, Sub-ommittee, therebent Merine and Ficheries Consultee, US Horse of Representatives).

Ou-livers satisfy one but inritude there ( es Pulis Handley while Privater of the Vational Carding of the contract of the Standy of his detention ( out a next carge) Abringer stickens), But elas! Abringer stickens), But elas! Nonetimes time counches allow no wore. Huns, I revealed to Rep. Norman E D'amours (NH), who chiesed the above-mentioned hearing, that I read the proposed hearing, that I read that it reads like a Harvard sophomore class everise. I could ascassily have said Dartmouth, or wort any other institution that anostes proves in a meaner and to an estent that hear created lethed aboundation on a monumental scale.

COLKETED USNPC

181 OCT 26 P2:54

OFFICE OF SECRETARE DOCKETING & SERVICT. BRANCH

en

.23

1

1113.22

Dundenscored large parts of pp 3 8091-6 on my copy. Since I am leaving here tomorrow to return to Deinicis, I am not yoing to escent time conventing in speafic detail. I do not object to an exercise in ideal schemes for use us a way to weasure deviation from ideal (of the ideal ace laws, I will have to suffice for me to cell your attaction to the reverse side of the paper I leve is conta way to way to
chosen for this communication . a notion that even before its founding developed on commerces and industries destructive to human health and well-being, porticularly tobacco (eq amentoring et seg ), and contimes to do so in this name fa spurious and specious vision 17 "public sefety "and "prevention 17 juvenile delinguency " will self distruct by continuing to build on chight - Nothe-hand holding - the - per (It is no accelent that in anticipation of setting down some comments to you Delected net of The Falmenthe Vaillie Library this - Prod, now-Sen. Hayabawa's book m sementics Language on thought and action ) By proundy, on crea anxionaly, committing to paper the Torm DISPOSAL, or in The case of the Wand . Lequiliture, SAFE DISPOSAL out doing with

create the reality nor the powerbility. I am not telling you once thing you don't know. Philaps you'll heply, "We must start armewhere". "The starts were all made long and, the meny awests unheilded. It is not true that "not enough was known" on "people isn't know." Hobel prizes still acent gives for putting on the brakes. Guite the exiting.

the attrocores walketing pradio active surves detectors (note no nuntion overlag that they are relisative) does more than put eyp on the NRCs collective face. It puts we in wind of a threat ascribed to the late revolutionary the Invess. He supposedly said, cinca 1968, "Will get you in your homes," He needed have troubled himself. The VS Goot which provotes the manfield application of the valid secondice sinciple of culiestened self-interest "Contid visit ports. is regulty welking it most (and increasingly moster) whither we liave technologies or don't, whither we are a democracy or arent, it. (on the way out we are likely to press through a stage alim to the annually in the leger colony described by hickness in his nord Hawkie).

Don't even ask whether its tos late for limens to surrow meamale folices. Shut it all down it will all self-detende anyway from inherent flaws. Create no more nuclear materials, with all attendent saturalizations, not for plarmo therepies, not for industrul "isting, but for basic science" not for law inforcement, certainly not for weaponry. Helter and ford will follow more readily, and human missing will diminic ens. monely. Humanity is willing to accept loss taking now. Human itself of a science of it, as in the Canter next forms. unabashed embracing of aprituelity. Twenty yours ago I called the blow bould the cheet canalizer. Now we reprogressed to the Withmate Pacifier.

this kind Juning may be wither what you want nor expect. style fing himited appearance Statimints should lead you to expectit, after all, I don't onew the world tenough 3 fingers of Scotile, vor do Istorge for any commercial interests, nor and I on an ego trip that could have me send it to Perspectives in Biologyand Weekicine. I will formally to ask that you place without further connect from we at this time, into the record in lieu of in as further comment by we the records of the hearing in Boston 9/21/81, referred to above, excerning part Sumping productes in Boston (conta nest rage)

Harbor and beyond, and alio the record of The bearing by the Falmoute (Mars) Board of Health last Hunday; Cctober 15 th, the Civil Defense" Room of Torn Hall (02540) will put flesh ruts the skeleton 7-55-5 Don't be to startled of the 7-56-15 is too jungreuous to handle bare-. . . . . . (1) com have If first strike doesn't intervine. I hey have some Illinois -triented conchrents in a bet. Semper -

Los Alamos National Laboratory Los Alamos, New Mexico 87545

Energy Division Safeguards Systems Group Q-4 ULL CCtober 15, 1981 TELEPHONE (505) 667-7777 81 OCT 285 P2 543-7777 Q.n.P DOCKETING & SERVICE יות אפניניא וויניא EHUPOSED BULE PK-

R. Dale Smith, Chief Low-Level Waste Licensing Branch Division of Waste Management Nuclear. Regulatory Commission Washington, DC 20555

Dear Mr. Smith:

D-56-5

8-249

Thank you for arranging to send me a copy of the proposed rulemaking on land disposal of low-level radioactive waste (lOCFR Part 61) and for the opportunity to provide these comments. The following comments are made with the hope that they will help in further sharpening the scope of your rule-making process.

(1) I have reservations about the open-ended definition of wastes that might be included in land disposal. Under the present definition (see Section IV under Supplementary Information), high-specific activity wastes, such as those produced presently during the cleanup operations at TMI-2, will qualify for land disposal as "Class C Intruder Waste." A better restriction on the kind of wastes that will qualify for land disposal seems appropriate. . . .

(2) The classification of wastes (Part 61.55) as "Class C Intruder Waste" is too vague and it lends itself to convenient interpretations. Some limits could be established in terms of specific activity to restrict the use of this term. The waste form criteria and packaging requirements also must be rigourous to this class of waste compared to classes A and B as defined in your proposal.

(3) Waste characteristics (Part 61.56), I feel, have several inconsistencies. It is the stated objective of the requirements to provide stability for the waste for a period of at least 150 years. Then, under item \$7, Subsection (a), you have detailed in the second second

> and the second An Equal Caportumity Emaloyer/Operated by University of California 💦 🗄 👘 🖓 👘 👘

.

0-4/81-584

the requirements for gaseous forms of wastes. These two do not appear consistent to me. Presently, there are technologies available to readily convert radioactive nuclides of halogens,  ${}^{3}_{\rm H}$ ,  ${}^{14}$ C, etc., into more manageable waste forms. The state-of-the-art in the management of radioactive noble gases D-56-54 dictates dilution and dispension as the safest method. The implied suggestion in this document to contain and bury up to 100 Ci of gaseous form of radioactivity in a container at a land-burial site and expecting stability for at least 150 years is baffling.

(4) Under "labelling" (Part 61.52), it would be desirable to D-57-1 linclude color coding to identify the three classes of wastes.

> (5) Under Part 20 ("Standards ....."), there is a requirement to conduct guality assurance and compliance verification. To

M-2 thave this system work properly, a variety of procedures have to be established and approved by the licensing authority. I am not sure whether small waste generators such as medical facilities, small industrial operations, etc., can invest resources to comply with this requirement.

(6) Under Subpart C (61.40 to 61.44), it would be desirable to add a performance objective for the site to minimize intrusion by animals such as rodents, deer, etc., which have the potential C-7 for transporting the radionuclides to the food chain. Parenthetically, I must add there are recognized studies done at Los Alamos showing the potential for a simple rodent to create havoc at a low-level waste-disposal site.

I think your approach to rule-making, stating performance objec-) tives (rather than detailed criteria), is commendable, recogniz-C - 1 ing the numerous forms of low-level radioactive wastes. This ( is also the reason why I think the definition of Class C wastes needs to be more restrictive. Please note that while preparing these comments, I did not have the benefit of your rule-making document (NUREG-0782). If I can elaborate on any of these comments, please let me hear from you again.

Sincerely yours,

KSP:ke

cy: CRMO, MS 150 (2) file

**Bechtel National**. Inc. DOCKETED Engineers - Constructors 81 OCT 26 P2:55Fifty Bears Street San Francisco, California ۵ Mex Address; P. O. Box 3965. San Francisco. CA 9418 SERVICE. October 21, 1981 Secretary of the Commission U.S. Nuclear Regulatory Commission Washington, D.C. 20535 COCRET NUMBER PR-2. EROFOSED SULE PR-2. 46 FA Attention: Docketing and Service Branch Proposed Rule for Licensing

Subject: Requirements for Land Disposal of Radioactive Waste

Gentlemen:

We appreciate the opportunity to provide comments on the proposed rule for Licensing Requirements for Land Disposal of Radioactive Waste. We feel that the proposed rule is an important step towards satisfying the need for additional low level waste disposal capacity.

In general we concur with the content of the regulations as written. However, there are several areas which we feel should be clarified to avoid potential problems during implementation. Detailed comments from our review of the proposed regulation based on our engineering and construction experience are attached.

If you have any questions please feel free to contact me directly.

Sincerely. Ashton J. O'Sennell

Vice President

RJT: sas

Attach.

B-250

# COMMENTS ON LOCER61

### LICENSING REQUIREMENTS FOR LAND DISPOSAL

COMMENTS ON THE SUPPLEMENTARY INFORMATION

The following comments refer to the issues raised in the discussion section which precedes the regulation. The toman numeral and letter following the comment number refer to the section to which the comment applies.

1) V C A classification system for low level waste is proposed containing four classes. Specific activity limits define each class. While the desire to provide a level of protection commensurate with the activity level is recognized, it is not clear why class A and B wastes need be distinguished.

D-55-1 The basis is apparently that "is is obvious that if these (unstable trashtype wastes) were disposed of with higher activity wasts, their deterioration D-55-13 could lead to failure of the (burial) system and permit water to penetrate

the disposal site and cause problems with higher activity wastes." If this is demonstrably so, would not it be better to define classes A and B solely on the basis of stability rather than on stability and activity concentration? Thus, all waste up to the class B activity limit would be separated by whether it was solidified or unsolidified.

2) V C It is stated that the Commission recognizes the need for a "de minimis" classification of wastes, that this "should be determined on a specific waste basis," and "Part 61 will not establish a generic 'de D -55-2 minimis' category for waste." The "de minimis" levels need to be established

as soon as possible to prevent disposal of large volumes of materials which, in hindsight, would have net such criteria and could have been disposed of by other means. This need is especially critical in view of the limited capacity of the existing disposal sites.

(3) V E The discussion states that active institutional controls such as periodic surveillance and controlled access cannot be relied on for more than 100 years. The implication is that governmental institutions would D-59-25 be aware of the necessity to maintain the site. However, in the following paragraph, credit is taken for passive controls. It seems that if a government institution is available to maintain land ownership and records, that same institution could maintain a fence. This would reduce the concern about the potential exposure to intruders.

COMMENTS ON THE REGULATION

The following comments address specific concerns raised by the regulation. The number following the comment number refers to the section of the regulation to which the comment is addressed.

4) 61.1 Purpose\_and\_Scope

A-1

The purpose and scope states that these regulations establish requirements upon which the commission issues licenses "for the disposal for others of radioactive wastes." While IGCFR Part 20 covers the disposal of waste by an individual licensee, the quantities are limited to very low levels. The purpose and scope should be rephrased to allow an individual licensee to operate a burial size. The words "for others" and the last sentence in 51.1(s) should be deleted.

### 5) 61.2 Definitions

a) A "near-surface disposal facility is defined to be within the upper 15-20 meters of the earth's surface'." Is this definition intended to limit the maximum depth of a trench? If not the definition should be rephrased ED-1 to clearly indicate that 15-20 meters is an approximate range.

b) The term "stability" should be defined since it is the basis for the seperation of class A and B waste.

(c) The use of the word "isolation" in the definition of "disposal" implies

GEN-1 } degree of absoluteness generally not attainable and its use in this context implies that the disposal facility is not part of the biosphere. The definition should read "'Disposal' means the placement of radioactive waste in an approved disposal facility.

6) 61.7(s)(1) This clearly limits the depth of individual disposal units ) or trenches to 20 meters. No justification for the limit has been offered

- ED-1 by NRC. At some locations it may be appropriate and economical for individual trenches to exceed 20 meters in depth. The statement should be revised accordingly.
- (7) 61.7(b)(1) The safety objectives should be rephrased to read "minimized GEN-1 the migration by any route including surface, groundwater discharge, erosion or wind erosion, and minimize the exposure to inadvertent intruders."

(3) 51.7(b) (2) According to this paragraph a maximum disposal site inventory will be established for certain isotopes based on the characteristics of

D-55-10 the disposal site. However, no criteria on which to base maximum site inventories is provided. The criteria for setting maximum inventories should be provided since this will limit the capacity of the site. The criteria should take into account the site conditions and locations. Also the criteria should reflect the fact that if a site is properly selected. La single large size may be more desirable than a series of smaller sizes.

9) 61.1(c)(3) As indicated in paragraph 61.52(a)(9) closure and stabilization Deasures must be carried out as each disposal unit is filled. Paragraph 61.7(c) (3) states that the site is in the closure phase when closure and ED-1 stabilization activities are being carried out. In order to clarify the intent of this paragraph, the work "final" should be inserted between the fifth and sixth words so that it reads, "during the period when the final Lsite...."

(10) 61.12(d) Among the specific technical information that must be included in the license application is: "(d) a description of the design basis natural D-50-9 events or phenomena and their relationship to the principal design criteria." These events require further definition. The definition should include not only the types of events bur also the time frame over which they must be considered. ÷.,

11) 61.24(h) This appears to be inconsistent with IOCFR2.105 which provides the applicant an opportunity to petition for a hearing on any additional B-R requirements or conditions. The "...or thereafter..." is particularly onerous in that it permits the staff to bypass the rules of procedures as described in Part 2 of the chapter. The Commission already has methods to require immediate action by a licensee through either an Emergency Order or a Compliance Order. This requirement should be deleted.

B-4 [12] <u>61.25(a)</u> This paragraph, which appears to be directed toward safetyrelated features at the facility, in reality prohibits any facility change without undergoing a notification and approval procedure. The first sentence should be modified to permit the normal flexibility to modify office buildings, maintenance structures, parking lots, etc. that all nuclear licensee's enjoy while holding licensee issued pursuant to this chapter.

3-7 [13] <u>61.29</u> The requirement that the licensee maintain responsibility for the disposal site for a <u>minimum</u> of five years is an open ended requirement. A specific time period should be set. As currently stated the criteria does not provide sufficient guidance to establish adequate funding. Since wastes will not be received during this period all funding must be derived from less charged during operation. It is necessary for planning purposes to know the time period over which the licensee will be responsible.

C-4 (14) 61.42 As currently written, this requirement is too absolute. The sentence should be rephrased to read "Design, operation and closure of the land disposal facility should not result in conditions...".

 $D-51-1 \begin{cases} 15) & \underline{61.51(a)(3)} \\ \text{criteria will ensure that only acceptable sites are selected. Hence, the site will meet the criteria without improvement.} \end{cases}$ 

16) 61.51(a)(4) The word "prevent" should be changed to "minimize".

(57-) 17) 61.51(a)(6) There is no way to completely "eliminate contact of water with waste." However, it is possible to minimize the likelihood of water contact and provide a water collection and treatment system which can be used as water intrusion monitoring shows that it is necessary.

(18) <u>61.51(7)</u>. This paragraph states that the disposal site shall be used exclusively for the disposal of radioactive waste. This seems to be

- D-50-40 unnecessarily restrictive. It should be acceptable to allow disposal of other vaste types as long as there is no commingling of the vaste types within a disposal facility. Once an acceptable disposal site has been found, maximum use of the site for the isolation/disposal of any environmentally dangerous materials whether they are radioactive or not should be provided.
- ED-1  $\begin{cases} 19 & 61.52(a)(9) \\ measures should be defined. \end{cases}$

D-52-1 System must be capable of providing early varning of migration of radionuclides from the disposal units before they egress the site boundary."

 $(21) \frac{61.55}{61.55}$  An alternate method of determining the waste classification should be provided. Provisions for classification by external dose determination should be made. For cases where the types of isotopes of concern are known, this method would allow adequate classification. This alternate method would D-55-64 be particularly helpful for nuclear power plant trash. Generally, trash

bas a very low specific activity compared to the class A limits. A determination of the radionuclide identity and concentration, as required by Part 20.311, would require the purchase and use of a portable spectrum analyzer. Instead, a contact dose rate measurement of the containerized trash could be made to show that the activities were below the class A limits. Also, since a radvaste classification system is already established in 10CFR71, is in possible to the 'the two systems together? 22) 61.55 Table 1

D-55-1 The logic behind the numbers selected for this table is not apparent. It would appear, for example, that carbon 14 which contains less than 0.8 microcuries per cc may be disposed of as segregated waste but that any concentration greater than 0.8, even if it is only a tiny increase, immediately requires that the disposer seek special permission from the government for disposal. The abrupt demarcation needs explaining so that the logic of it can be understood.

23) 61.55 Table 1, Footnotes

D-55-12,5a) The term "significant gamma radiation" should be defined.

D-55-4 [1] How is radium treated? A value should be established.

D-56-2 (c) The footnotes place a restriction on wastes containing chelating agents in concentrations greater than 0.1%. Is this limit intended to be 0.1% by weight or volume?

D-56-5 (24) <u>61.56(a)(5)</u> The inconsistency between paragraph 61.56(a)(5) and (7) bould be clarified. Wastes in gaseous form, allowed by paragraph 61.56(a)(7), could be considered the very waste disallowed by paragraph 61.56(a)(5).

(25) <u>61.56(b)(3)</u> This paragraph requires that "Void spaces within the waste and between the waste and its package must be reduced to the extent practicable." This requirement should be deleted as being too wague unless specific acceptance criteria can be established. For solidified liquid or slurry wastes, 10-152 freeboard in a 55 gallon drum is appropriate, but a container of dewatered

D-56-9 | resins may well contain 50% voids. If the 5% dimensional requirement is

D-56-10 applied to the steel container instead of the solidified monolith, most packages would not meet the criteria since the packages are typically only 90% filled to minimize the potential for spills. This requirement should be modified to be consistent with standard practice.

STATE OF NEW MEXICO Secretary of the Commission GOVERNOR'S CABINET October 16, 1981 SANTA FE Page 2 87503 GEORGE S. GOLDETTER PH.D. SECRETARY FOR HEALTH & DIVISIONNE Page 38097. Col. 3. Line 6 DOCKETED It would be more technically correct to have this sentence read: "Waste that has a radioisotope concentration that exceeds the numerical values shown in Column 3. D-55-12 Table 1 .... It is implicit, by definition, that the concentration of H-3, C-14 October 16, 1981 OCT 27 P3:03 and Co-60 cannot be in excess of their respective theoretical specific activities; 142 (Page 38097, 61.55 (Waste Classification) NG & SERVICE BRANCH D-55. ) It is not clear from the draft whether Ra=226 will be permitted and in what 45 Secretary of the Commission concentrations. U. S. Nuclear Regulatory Commission Washington, D. C. 20555 Page 38099, 61.70 (Scope) WELET MUNEER DD It is unclear what the role of an Agreement State would be in the licensing and Attention: Docketing and Service Branch F regulation of land burial of radioactive wastes as opposed to a non-agreement State. Gentlemen: We have reviewed the "Licensing Regulrements for Land Disposal of Radioactive Page 38099. 61.72 (a) Waste" (IOCFR 61) and offer the following comments for your consideration. Rather than relying on notification in the Federal Register, it would be more Page 38085, Col. 3 (Institutional Control) beneficial to State and tribal governments if the Director specifically inform . the appropriate State and/or tribal official if a near-surface radioactive Some provision should be made to require more obvious passive controls other than dispasal facility is being proposed on lands under their jurisdiction. cublic records. A "permanent" type of identification monument should be D-51-24 constructed describing the facility and its boundaries. Unlike individual trench ncerely. markers this should be of sufficient height above the ground surface that it will remain visible for many hundreds of years. Page 33096, 61.50 (a)(5) George S. Goldstein, Ph.D. D-50-3 Lower flood plain frequencies of 250 to 500 years should be considered since this Secretary is the time frame required for radioactive decay to innocuous levels. . . GSG:teb Pa<u>ce 38096,</u> 61.52 (a)(6) Exposure rates for gamma levels at the trench cap surfaces should be some D-52-5 specific value above background so that compliance may be readily determined. Page 38097, Table 1, footnote 3 D-55-12) This footnote does not refer to any waste class but only to Class C intruder wastes. For clarity it should read: "Maximum concentration for near surface . . . disposal." B-253

Jal Jaffer 8.0. Box 013950 Miami, FL 33101 27 207 DOCKET NUMBER PR- 2,4 + 47 DOCKETING & SERVICE (46 FR 38081) Dactating + Service Section U.S. Nuclear Regulatory Commission 1717 H Street N.W. Washington, DC 20555 A-4 GEN-6 Attn: Notice of Appoint Rules - Licensing of low-level Redisative Was TO WHOM IT MAY CONCERN - 10/22/81 RES ROADED-RULES LICENSING LOW-LEVEL RADIOACTIVE WHITES Gentlement I am greatly concerned over the issuance of the proposed rules above referenced. I lost think the Communition has any automorphy. to license the dispecel of radicactre nostes which remain trace for the license the dispecel of radicactre nostes which remain trace for (angths of time, and for proposed purposes and scope, beyond the NRC) authority. I think the action is how cardicus enough to require 76 a generic impact ctatement prepared over and a longer period of more informed comment. I think actions already taking as in some rodioactivity being released to asphaltzared roadis serials enough to require complete investigation. Please keep ne informed, Supramistically yours of the Joel JoHer, P.C. Box C: 3956 Miann, FL 3301

	41			
DOCALL NUMBER DR	-2.1.	SIGNORE		
PROPOSED RULE I IN	V. La K	SIGNOFF		
/ 46	FE 39091)	Li UMB Abbravel No. 29-	AC218	
FEDERAL ASSISTANCE		sealing for ATL 81-60-00	a a	
application	D. Date	Month P Han OI OU JU	3.5	
Ation Application	19 Year Month Dati	SEP 0 S	1981	
/Mark Notification Of Intent (Out.)			-77	
bast Report Of Federal Action Blank OC	T 1 9 1981 (	John Quina - Che	Ar.	
A Lonal Applyment/Becometric		S. Federal Employer Identidention No.	σī	
A Assistant Name II C. Novelson Description			•	
b Computation Low-Level Waste Lic	ensing Branch	&. Program		
c. Street/F.O. Box		(From a. Number 77709	10 10	
d. City Washington . Court	NY .	Federal b. Title Ilnknown		
1. State D. C. 5. ZaC	aas 20555	II S Nuclear Reg		
h. Connet Person R. Dale Smith, Chief		tom Comm		
intre a reconone no.)		tory commu		
7. Title and description of applicant's project Proposed	Rulemaking on	8. Type of soblicant/resistent		
Land Disposal of Low-Level Radibac	tive waste	To the reason the Contraction Advised Agency		
I has notice invites public comment of	a proposed A-	S-County An Indean Train		
specific requirements for licensing t	he land dispo-	- Federal Assault		
sal of radioactive wastes.	-	ISpealy :- + age Tal Agency		
		9. Type of sourcenes		
		A-Basit Grant D-Insurance		
		C-Los Enter eppropriate letter(s)		
10. Area of project impact (Names of cities, counties, states, erc.)	11. Estimated number	12. Type of application		
Shahamida Asisona	benefiting	A-New C-Revision E-Augmentation		
Statewide, Artzona	!	Enter appropriate letter 2		
13. Proposed Funding 14. Congressional District	a Oft	15. Type of change For 12c or 12c		
a. Federal S	b. Project	A-lograsie Dollars P-Other Specify:		
b. Appart	01,02,03,04	C-Increase Duration		
e. Scare	Duration	E-Cancellation Enter supro-		
d. Local 1 100 19	Months	prate unter ip		
e. Other 1 1	I CAP MONUN GALL	13, Existing federal identification number		
20. Federal settery to receive request (Name, cury, state, ste co	de	125 Bemerke added		
	•			
122. [e. To the pest of my knowledge and to. If require	a av OMB Circular A-95 t	THE ADDINGE DOR WAS SUDMITTED	Response	
The application are true and correct, the aid respon	to instructions therein, to see are attached:	appropriate clearinghouses and response	stieched	
Apolicant   document has been duly authorized   111 ATIZO	na State Cleari	nghouse 🛛	2	
That		• 0	Ē	
· · · · · · · · · · · · · · · · · · ·		<u>Li</u>	<u>.</u>	
22 Typed name and title	a. Signatura	L Date signed		
represent		10	" <sup>uy</sup>	
t 24, Agency name		Application	нын саў	
26. Orsenitational Unit	27. Administrative offic	• .28. Federal sosiication		
		- identification		
28. Address		30. Federal gram		
			<u> </u>	
31. Action taken 1 32. Funding		Type month sey 34. Year not Starting	uh dey	
a. Awargad a Sectoral is 00	33. Aetion date 19	este 19		
b. Pejected 10. Azzhcant 1 00	Neme and selection	minitermatien 136. 1827 mg Miniteri	ALA 343	
amenament in that a contract of the state		19		
a. Deferred	!	Jr. nemaras edded		
Ce. Mithorawh				
IS , is to taking above action, any commenta te	anved from siearing-	D. Fegera Agency A-35 CHic a	<u> </u>	
Federal agency Int Part 1, 2MB Cuttor A-85, 1 Pas oren o	e is ove under provisions	Neme and selephone numbers		
A-25 artise				
424-101		Standard Form 424 Page	110-751	
		Prescribed by USA. Fozeral Management Circ	147 944	



aidi tita min li

B-255

Memo to Charles F. Tedford September 14, 1981 Page 2

B-8

B-11

D-55-13 advantages. First, a more effective use of land surface area may be achieved and, secondly, by placing the Class C waste at a depth of 35-50 meters below ground surface level, covered by a 5 meter overburden layer, on top of which a layer of Class A or Class B waste is buried, will provide additional intruder protection to the Class C wastes.

> 3) The Part 51 regulations place obligations upon the license holder for certain activities to be conducted after the operational period, specifically, activities related to closure and post-closure monitoring activities. The NRC proposes certain regulatory efforts with respect to the financial affairs of the licensee to ensure that the obligations of closure and post-closure activities will be met. Authority to investigate and regulate the financial activities of licensees does not, in general, exist in State radiation control programs. Some undesirable experiences have resulted from this since, if a licensee fails to meet its obligations, the expenditure of cublic funds is necessitated to correct any and all deficiencies. Court action is the only possible method of recovering these expenses.

Since there is no profit to be gained as a result of closure and postclosure activities, it may be economically advantagous for the licensee to seek legal maneuvers, whereby its obligations may be avoided. In view of this, I recommend as an alternative that a long-term care fund be established at the time the license is issued. Monies would be paid into this fund during the operational period of the license, either from a tax or a fee structure imposed on the licensee. Obligations of the licensee would terminate with the operational license. At that time, the state, using monies contained in the long-term care fund, would take all title and responsibility to the site and ensure proper closure and post-closure activities by means of open bid contract. The former licensee would be entitled to bid on this contract, provided his performance during the operational period was acceptable. In any case, the cost of closure and post-closure monitoring activities would be paid for by the time the site had been filled.

(4) In order to further ensure adequate performance of the licensee during the operational period, I would suggest a change in the orientation of the licensing; from a single license that is generally oriented towards the operation of the site, to a license which is more specifically oriented to the burial units. Under this licensing scheme, only a limited number of burial units should be permitted to be open at any given time. The licensee would not be permitted to open new units for use unless, and until, filled burial units were adequately closed and stablized. The majority of the work of closure and stabilization should, therefore, be accomplished as an ongoing measure during the lifetime of the site. Nemo to Charles F. Tedford September 14, 1981 Page 3

D-57-12

Paragraph 61.57, "Labeling", requires that each package of waste be labeled with the words "Class A, Segregated", "Class B, Stable", or "Class C, Intruder" to identify its contents. In view of the result which labeling is intended to produce, I question whether the use of the terms "Segregated", "Stable" or "Intruder" will provide any advantage. My previous experience suggests that the use of labels specifying "Class A waste", "Class B waste", or "Class C waste" would be less confusing and more meaningful to the individuals likely to handle the packages.

58:jr



Secretary of the Commission October 23, 1981 Page Three

2. Site Inventory Limits (Section 61.75 (465) and Section 61.55)

It is more appropriate to limit concentrations rather than quantities. The 500-year period assumed as the effective life of the intruder barriers implies protection against deliberate intrusion after that time period. Limiting concentrations initially buried should provide the protection required.

3. Changes (Section 61.25)

B-4

This section should be modified to allow the operator to make changes under cartain conditions in a similar manner to that permitted by 10CFR 50.59 during the operational period of the burial facility. The requirements as listed in this section are avkward and would prevent necessary changes from being made quickly to handle a given situation. Or, as stated in Part ( $\lambda$ ) of this section, the exception categories for this specific license condition should be stated.

Stability of a Disposal Siza (Section 61.44)

C-6 The NRC should make clear that this section applies to the period of institutional control.

Disposal Sites Suitability Requirements (Section 61.50 (6))

) The meaning of the word "upstream" is not clear. It is assumed that the D-50 NRC means onsite.

6. Disposal Site Design (Section 61.51 (3))

The meaning of "must be designed" should be clarified. It would appear that additional requirements would be necessary only if the site was marginal.

GEN-1 { (6) It would be almost impossible to design it to eliminate contact with water. The statement should read "designed to minimize" instead.

D-52-4  $\begin{cases} (7) & \text{The meaning of this item is not clear. The NRC should clarify. It is assumed that it means the site shall not be used for chemical waste disposal along with the radioactive waste, but even that needs an explanation as to limitations if for example the "chemical" waste is radioactive.$ 

7. Disposal Facility Operation (Section 51.52 (4))

D-52-2 Strike the word "orderly" from the sentence. As it may not be cost beneficial to stack certain types of waste and yet they certainly can otherwise be loaded into the trench in a manner that maintains the package integrity. Secretary of the Commission October 23, 1981 Page Four

3.2.2 (5) Change sentence to read "void spaces between waste packages must be filled with earth or other material so as to reduce future subsidence within the fill.

(8. Waste Classification (Section 61.55) Table 1 Note 4

D-56-2 What is the basis for the quantity limits that enables one to make the case-by-case determination? As stated previously (item 2), it would appear that concentration limits rather than quantity limits would be appropriate to control all hazard aspects. What is the basis for the 0.1% chelating agent concentrations?

D-56-8Section 61.56 (b) (1) This section appears to imply that drums of compacted trash must be structurally stable for at least 150 years. This would probably require that such waste be solidified. The NRC should clarify the intent of this section.

9. Institutional Requirements (Section 61.59 (b))

. This item requires an environmental monitoring program for 100 years which seems to be excessive by about 95 years. The environmental monitoring program should only continue during the five-year post closure period if environmental monitoring is intended to be a period of sampling of wells and

other environmental samples for radioactivity. If environmental monitoring is deemed to be required for the period of IGO years, it should be severely limited to occasional samples of the ground water pathway.

Very cruly yours, William O. Parker, J. Ly S

SAG: scs

B-10

. ITTEL NUMBER FROFESED BULE m.T. 646 HOWARD UNIVERSITY DOLKETED USNRC on WARIENGTON, D. C. 20069 21 October 1987 01 29 P4:13 WI FICE OF SECURITY AND SAFETY SERVICES OFFICE OF SECRETARY DOCKETING & SERVICE BRANCH · · . 13 ... . . 6 ACCENT NUMSER DD Mr. R. Dale Smith, Chief PROPOSED BULE FK Low-Level Waste Licensing Branch FR 38081 Division of Waste Management 46 U.S. Nuclear Regulatory Commission Weshington, DC. 20555 Alfana Storta R. I. Bas 193.A Dear Mr. Smiths This cames in response to your letter dated August 7, 1981 concerning the Proposed Rulemaking on Land Disposal of Lou-Level Radioactive Waste. <u> 1</u> Listed below are some comments that I would like for you to consider in regard to "Transfer for Disposal and Manifests": . . . . . 20,311 Trensfer for Disposal and Manifests (b). Approximate date(s) of the waste generation and the length of the period of storage before pick-up. should, also, be shown on the manifest. This information will give some idea of the radioactive GEN M-1 decay which might have already taken place, and the real activity at the time of disposal. This is considered important information. (c) (5) Sixty (60) days is considered to be too long a period for the missing radioactive material to generate ances before being ansidered missing. It should be reduced to thirty (30) days. Your cooperation in this regard will be appreciated. Sincerely yours, 9.11 BILLY T. NOTWOOD "Lirector" ٠, Accurations by card. 1144 (17). 0 Ret. Party Elent 8-259



COCXETED USNRC

ETING & SERVICE

700 University Avenue, Toronto, Ontario MSG 1X6 81 IET 29 P.4:34

October 15, 1981

Nr. R. Dale Smith, Chief Low-level Waste Licensing Branch Division of Waste Management Office of Nuclear Material Safety and Safeguards US Nuclear Regulatory Commission Washington DC 20555 United States UCERET NUMBER PR-2.2121. OGGEGED RULE (46 FR 38081

Dear Mr. Smith:

The following are Ontario Hydro's comments on the proposed rulenaking for 10 CFR Part 61 for "Land Disposal of Radioactive Wastes."

### General Comments

The general philosophy of Part 61 is that "overall performance objectives are stated and the applicant has flexibility in choosing design features and operating practices to achieve these objectives...prescriptive requirements are stated as minimum criteria to afford some flexibility in meeting thea." This approach is supported by Ommario Hydro rather than having specific desailed criteria and standards.

The geotechnical siting requirements are sufficiently general for inclusion into the Federal Register as rules. They reflect the US preference to found disposal facilities above the groundwater table. While this is feasible in many parts of the US, it is simply impractical in Ontario, where the groundwater table is often close to the surface. In the latter case, the NRC requires that diffusion be "conclusively shown" to be the predominant means of radionaclide movement is groundwater. (Page 38096). This means that the disposal sites will be limited to those with subsurface permeabilities of less than  $10^{-6}$  cm/sec (i.e., till, clay shale, etc). Such geologic materials are, indeed, included in our investigation plans.

C-1

D-50-

Acknowledged by card.

### Specific Points

p 18083 The proposals call for a 500 mram/a maximum individual exposure limit for intrusion. Ontario Eydro supports this limit as only a few people could conceivably receive this dose.

- 2 -

p 38084 Under "complexity," the disposal site "must be capable of being investigated and analyzed." This seems vague. Will there be specific definitions derived for "complexity" or will decisions on complexity be purely subjective (i.e., will a site be ruled out simply because it is harder to analyze than other sites)?

p 38085 The rules do not attempt to establish a generic deminimus category for waste. This is unfortunate as it would undoubtedly be a greater economic advantage than case by case decisions. However, as there is not as yet a concensus on a generic deminimus level, any level chosen would be premature.

4.
p 38085 Federal or State government ownership of land used for disposal of radioactive waste is certainly an easy way of guaranteeing proper safe use of the land. However, I an uncertain of some of the implications. At what point would the overnment take over control of the land? Would the government be legally responsible for maintaining active control if the licensee failed? Would the government be legally responsible for any health problems arising from the site after active controls have been removed?

D-59-3 In the case of federal ownership, would this not stop the Nuclear Regulatory Commission from regulating the site? I understood that one department of the federal government could not regulate another.

5. p 38087 Onder disposal site closure phase, I do not understand why public hearings are held on the closure of a site after all the waste has been emplaced. The public hearings should all be done before the original license is granted. What happens to the waste if, after the site is full and ready for closure, the public reacts negatively to closung plans? This would place an unfair financial burden on the licensee.

.  $\int p 38093$  Why must the licensee sign the license under oath? I am not familiar with US legal practice.



p 38097 Under waste characteristics, why must wastes not be packaged in cardboard or fibrehoard boxes? If the waste can be segregated then these types of packaging should be acceptable.

- 3 -

Yours truly,

R.G. Tulk

R.G. Tulk Nuclear Materials Management Department

P.J. Armstrong Supervising Design Engineer Nuclear Materials Management Department

DOCKETED. USHAC

. .

222

.......

े.... - भ्रम

PROPOSED RULE

ARKANSAS POWER & LIGHT COMPANY 51 01 29 P4:35 POST OFFICE BOX 551 UTTLE ROCK ARKANSAS 72203 (501) 371-4000 October 21, 1981

Secretary of the Commission U.S. Nuclear Regulatory Commission Washington, D. C. 20555

ATTN: Docketing and Service Branch

Subject: Arkansas Núclear One - Units 1 & 2 Docket Nos. 50-313 and 50-368 License Nos. DPR-51 and NPF-6 Comments on Proposed Rules on Licensing Requirements for Land Disposal of Radioactive Waste (File: 3905, 2-3905)

Gentlemen:

ØCAN1Ø8107

The purpose of this letter is to submit Arkansas Power & Light Company's comments on the proposed rulemaking on land disposal of low-level. radioactive waste which was published in the <u>Federal Register</u> on July 24, 1981.

Comment No. 1

As Mr. R. Dale Smith indicated in his August 7, 1981 transmittal letter, waste categories based on radioisotope concentration and waste characteristics are proposed, and it would be the shipper's responsibility to determine the proper category and comply with the minimum requirements of the category (Sections 61.55 and 61.56 of Part 61).

D-55-6Due to the heterogeneity of the isotopes present in the waste, we feel it would be highly impracticable to positively determine a given activity level as being charactaristic of the presence of a respective isotope. Consequently, it would be most difficult to use Table 1 to assign a waste classification category to the waste with an acceptable degree of certainty and thus makes impracticable the proposed waste classification methodology as being a definitive approach.

WEMBER MODLE SOUTH LITILITES SYSTEM



D-55-11

B-262

Secretary of the Commission October 27, 1981 Page II

C-3 Section 61.41 - The ground water concentration limit should be established at the boundary of the disposal site. A "mearest public drinking water supply" criterion might change after establishment of the site, causing potential danger on causing retroactive design limitations.

F - 1 Section 61.70 - Participation should be limited to involved states.

Cverall the document is liveable and an important step forward in providing for continued benefits derived from use of radioactivity and radioactive materials. Further improvement can be achieved by incorporation of these comments.

· · · · ·

Sincerely,

larse a. Luit Carol A. Lively

Executive Director

CAL/smt

COLXEI LU USHRC

181 HUY -5 P5:07

THE NATIONAL ASSOCIATION OF INSURANCE BROKERS, INC.

November 5, 1981

11/9/81 eng

. . .

۱.

42.00

Secretary U.S. Nuclear Regulatory Commission Washington, D.C. 20555

Attention: Docketing and Service Branch

RI: Proposed rule amending 10 CFR Part 51 Subpart I -Financial Assurances, as published in the Federal Register, Volume 46, No. 142, Friday, July 24, 1981

Dear Sir:

· · · ·

The financial arrangements sought by the Nuclear Regulatory Commission are of sursty nature and are designed to ensure that funds are available to properly close down radioactive waste disposal sites.

Several options for providing financial assurance are specifically proposed—surety bonds, cash deposits, certificates of deposit, deposits of government securities, escrew accounts, irrevocable letters or lines of credit, and trust funds. In addition, the proposed rule would provide other options by permitting "such types of arrangements as may be approved by the Commission." The National Association of Insurance Brokers believes this flexibility is crucial if the proposed rule is to function in a reasonable manner.

The NAIB represents major commercial insurance brokers in the United States. Our members develop more than half of the mation's business-related insurance coverages. The primary function and responsibility of a commercial broker is to develop insurance programs and provide related products and services to protect the assets of clients ranging from large and small businesses to public and private institutions of all kinds.

NAIB believes that the question of whether or not these financial assurance provisions are reasonable depends on their availability and cost. For instance, today there is some question whether survey bonds would, in fact, be available as the survey market is not interested in providing bonding at prices that make it an option. Also, the Commission should note that traditionally survey is available only to financially stable firms which could qualify for any of the other options, nut is unavailable to those who may have a greater need of it.

E-1

NAIB amending 10 CFR 7art 61 page 2

E-

Insurance is not mentioned as an option here and, in the tra-ditional sense, may not be appropriate because it could be argued there is no risk transfer involved when the need for closure is a certainty. However, insurance could be a viable means of financial protection against the premature closure means of rinancial protection against the premature closure of a waste disposal site, i.e., if a site had to be closed down before its operators had planned to do so. If, however, the closure came about "as scheduled," then insurance is not a viable solution.

Today there are no insurance products on the market which would cover the type of closure which concerns the NRC in this proposed rule. In the future, however, some form of insurance may be offered, and the NAIB feels such an option should be favorably considered by the Commission.

In summary, NAIS believes that the single most important factor in the feasibility of the financial assurance provi-sions of the proposed rule is that the words "such types of arrangements as may be approved by the Commission be retained. If the Commission or staff have additional ques-tions, the WAIB would be pleased to respond.

Sizeraly.

. .

Barbara S. Haugen Government Affairs Officer

asa:bch

C. V. "Hans" Hancock Speaker Pro Tem Joe Winght Associant Previously Pro Tem Soboy H. Richardson Majority Floor Leader John M. Serry, Jr. Majority Foot Leader Eugene P. Stuart Anerity Foor Lasee Arthur L. Schmett Minerty Floor Londer LEGISLATIVE RESEARCH COMMISSION Cavid K. Karom William "Sil" Johnormoye Malormy Caucus Charman Scara Canada Frankfort, Centucky 40601 512-554-3100 Maximy Caucus Charmon Water A. Baser Nerman W. Farstiff Los Presher, Senate President Pro Tere dinamy Caucus Charman William G. Xenson, House Speaker Lowel T. Hughes Majorty White Weaty May Charment Over Middleson Minority White Vis Hoters. .... Woody Allen Minority White Sinctor SSLAT. November 2, 1981 781 NOV -6 A8:35 The Honorable Samuel J. Chilk Secretary Nuclear Regulatory Commission Maconic Building 1717-E Street, N.W. 5 Washington, D.C. 20535 (46 FR 39081) Dear Secretary Chilk: This letter is in reference to your regulations in 10 CTR. Fart 61 relating to licensing requirements for land disposal of radioactive waste. D-55-2

In 1930, the Kentucky General Assembly established a Special Advisory Committee on Nuclear Issues composed of legislators, university and industry experts, and citizens to, among other things, examine alternatives for hanging Kentucky generated low level nuclear waste. During its soudy, it was brought to the attention of the Committee that the establishment of <u>de minimus</u> standards for each radioisotope in air, soil, water, and solid waste by the Nuclear Regulatory Commission and other announters An Ell, soll, water, and solld waste by the Active Assessory Commission and other appropriate federal agencies would be of great benefit in managing low lavel nuclear waste. The Committee would like to encourage the incorporation of the de minimus concept in the current revision of 10 CFR Part 51.

Sincerely yours. Rapresentative Pata Wort Chairperson, Special Advisory Committee on Nuclear Issues

11:9:81 cmp

HOUSE MEMBERS

FW/ chis

SENATE MEMBERS



### CORRECTION NOTIFICATION

To Recipients of PR-2, et al. (46 FR 38031) - Licensing Requirements for Land Disposal of Radioactive Waste

Comment Nos. 58, 59, and 60 were miscoced. These numbers will not be used again.

Cocketing and Service Branch Office of the Secretary of the Commission

11/18/81

CORRECTION NOTIFICATION

To Recipients of PR-2, et al. (46 FR 38081) - Licensing Requirements for Land Disposal of Radioactive Waste

Comment No. 51 was miscoded. This number will not be used again.

Cocketing and Service Branch Office of the Secretary of the Commission

12/3/81

DR. HARRY LAWROSKI, PE CONSULTANT COLVETED 2375 Beimont Ave. Idago Fails, Idago 53401 Phone (208) 524-3237 '81 NIV 19 P3:23 November 13, 1981 Mr. R. Dale Smith Branch Chief, Low-Level Waste Licensing Branch U. S. Nuclear Regulatory Commission Washington, D. C. 20555 and and **F**I Dear Dale: \_ As we discussed in New Orleans, I suggest the wordings in 10 CFR. Part 61 on pages 94 and 98 be changed from "Commencement of construction prior to such conclusions shall be grounds for denial to possess and use special nuclear material in such plant or facility." to "Commencement of construction prior to such conclusions shall be at the risk of the applicant and shall not be a factor in determining the acceptability of the vanilations for linear " the application for license." Since timing for opening of disposal sites may be critical, some thought should be given to limited work authorization permits. This would allow work in select areas of a potential site. I note these words will be helpful. Very truly yours,

Hand Harry Lawroski

Acknowledged by

A-3

DEPARTMENT OF THE ARMY . T. OFFICE OF THE ASSISTANT SECRETARY WASHINGTON, D.C. 10115 E2 -2 P4:28 24 NOV 1951 Secretary of the Commission US Nuclear Regulatory Commission ATTN: Occketing and Service Branch COLLET REMOVE Washington, DC., 20555 Dear Mr. Secretary: This is to provide you with coordinated Army comments on your agency's proposed rulemaking action regarding land disposal of low level radioactive waste set forth in the Federal Register, Yol 46, No. 142, July 24, 1981. The Army has reviewed subject proposed rule for impact on the current Army radioactive waste program. In particular, only Parts 20.311, 51.55, 51.55, and 51.57 will affect the Army at this time. Part 51.53 classified low level radioactive waste into Class A, Class 3, and Class C categories. Class 3 concentrations require packaging to meet the requirements of Part 61.56. Since waste shipped IAW AR 365-11 are already required by Headquartars, US Army Armament Material Readiness Command (19800M), to meet these requirements, this part has no effect. However, Part D-55-6 51.37 requires an additional label specifying either "Class A segregates," "Class 3 stable," or Class 1 intruder" which will require changes in existing shipping forms as outlined below. Part 20.311 requires paperwork to accompany each shipment to supplement exist-ing requirements for record keeping. The current wasta disposal form used by the site contractors and ARRCOM already contain most of the required information. These items not now recorded on the form, such as solidification agents, Class A, 3, or C arconts, etc., will be included on ARACOM shipping documents and forms when required. All other requirements for the generator and/or shipper (i.e., Parts 20.311(d), 20.311(e), and 20.211(f)) are being accomplished by the ARACOM audit program and/or the ARACOM broker, Southwest M-4.5 Nuclear Coro. Sincersly, ..... Lewis D. Wather Lewis D. Valker Deputy for Environment, Safaty and Occupational Pealth OASA(ILLERM)

Acknowledged by can

. . .

181 NFC -2 A11:25

DC: METE:

Nuclear Privatery Convission Washington, D.C. 20555

LAND AUCURPTS RED TO ED RUIS -

Novembor 2h. 1981

Subject: Proposed Rulemaking on Land Disposal of Low-Level Radioactive Waste.

NOCTEL RUNSER DD

PROPERTY RULE TA

2.07

In my opinion nuclear wastes, both high and low level, -8 {should be permanently disposed of in underground geologic repositories. This problem should not be left for future generations to resolve.

As nuclear worker radiation doses rise, the long term implications are frightening, given the recent estimates of cancer risks associated with low-level radiation exposure. Therefore, offerts should be made to forge a new partnership among federal, state and lecal officials concerning all decisions of the dispecal of radioactive waste.

> Sincerely yours, Zelie M. Jewelw Milia H. Jonsen R.N. Route I Box 86A Grandview, Tennessee 37337

Union Gil Company of Caufornia SECRET & ARVEST Union Cil Center, Box 7500, Los Angeles, California 30051 Teleonone (213) 977-6810 · HOETO KONO DEPARTMENT OF PLANNING AND ECONOMIC DEVELOPMENT "61 CED 14 AND :45 Camamana Building, 250 South King St. UDIOD ----eni Ref. No. 3903 . . . . . . • • • December 2, 1981 Carleton B. Scott HORDED BULLS PR-2 November 13, 1981 Mr. Samuel J. Chilk ACTES ESTICES DE PUIS (46 FR 3808 Secretary of the Commission U.S. Nuclear Regulatory Commission COCAET AND DO U.S. Nuclear Regulatory Commission Washington, D.C. 20535 Office of Nuclear Material Safety MONDER ains MA-Washington, D.C. 20555 Att: Docketing and Service Branch Gentlemen: Dear Sir: Subject: Draft Environmental Impact Statement on 10 CFR Part 61 Union Oil Company of California (Union) takes this opportunity to present comments on the Nuclear Regulatory Commission (NRC) proposal published in the July 24, 1981 <u>Federal Register</u> (46 FR 38081) to add a new 10 CFR Part 61 to provide licensing procedures, performance objectives, and technical criteria for "Licensing Requirements for Land Disposal of Radioactive VASTAS" Thank you for the opportunity to review the subject draft EIS. Inasmuch as the proposed regulations provide ample opportunity for the expression of State concerns prior to the licensing of a disposal site, we feel that any Geastal Lone Management (CNO program concerns can be addressed at that time. Moreover, should a disposal site be proposed within the CM areas of the State, the feelaral consistency provisions of the Maticaal CM Act cuite that the licensing be subject to State review for consistency with Hawaii's federally approved CM program. licensing facilities for the land disposal of radioactive wastes. Set forth below is a summary of Union's nine major concerns with the NRC proposal. F-1 5. Existing facilities holding valid licenses for the land disposal of radioactive waste should be exempt from the additional licensing requirements of the proposed regulations. (2. A "de minimis" classification of wastes should be established to give relief to those facilities handling material whose radiation levels are sufficiently low so as to pose no threat to health or the environment. Sincerely D-55-2 Hideta Kono 3. The provision that binding interpretations of regulations can be made only by the Commission or the General Counsel is overly restrictive. Negotiations with the NRC will be A-6 hampered if all other NRC agents have only apparent authority and not actual authority. 4. Provision should be made for an "interim status" which would allow present disposal operations to continue legally until such time as licensing can be considered for them.  $\mathcal{E}$ -1  $\left\{ 5. \text{ Self-insurance should be permitted for those who qualify based on an annual submittal of financial reports.} \right\}$ 31:2170071 811202 467838081 B-268

- 2 -- 3 -6. The procedures for waste classification should be clarified. The list of radioisotopes in Table I should be expanded and detailed instructions should be given for classifying "" material not explicitly listed. establishing a "de minimis" classification of wastes. Wastes Containing the most minute amounts of radioactive material would be regulated irrespective of the level of radiation. This places an undue ourden on facilities handling material whose radiation levels are sufficiently low so as to pose no threat to health or the environment, A "de minimis" classification of 7. The proposed exposure limit of 25 mrem/yr at the boundary is restrictive and fails to take account of background radiation. The limit should be set as an increment above wastes should be established which considers two factors: (1) the volume, amount, or weight of the waste and (2) the (.-3 Concentrations of radioactive elements in the waste. background with total dose not exceeding one-third of the . . . . . Coccupational limit. . Section 61.3(b) and Section 30.33(5) D-56-6 8. Provision should be made to allow bulk shipment and disposal D-56-6 8. of low level waste. An applicant should be allowed to commence construction at his own risk prior to a final NRC determination, provided other 9. An applicant having all the required permits to construct on approvals are received. If an applicant has all required a site except an NRC license for disposal of low level waste should be allowed to commence construction with the underpermits to construct on a site except an NRC license for A-35 disposal of LLW, he should be allowed to commence construction - standing that he may not be allowed to use the site for low level waste disposal until such time as appropriate approvals for such activity have been received. with the understanding that he may not be allowed to use the site for LLW disposal until such time as appropriate approvals for such activities have been received. A - 3 . For example, assume that a company wishes to develop a new mine Union's specific, section-by-section comments are set forth and processing plant, and the plant will produce a LLW. The below. company has the option to dispose of the LLW on site or ship it in the the state of the state 1. a. 1. a. off-site for disposal. The mine site is found to be acceptable and a second for a LLW disposal site. The company has other facilities that Section 51.1 - Zurpose and scope. produce LLW and disposes of such material at commerical sites. The company chooses to consolidate all LLW for disposal at the Existing facilities holding valid licenses for the disposal of new site if approval is given by NRC. The company wishes to ... radioactive wastes should be exempt from the licensing proceed with development of the mine and other facilities (some requirements proposed in Part 61. This exemption should include disposal at licensed uranium mill tailings and low level waste disposal facilities with an approved disposal plan under 10 CFR A-1 of which will be used in processing LLW) independent of NRC approval for LLW disposal. The overall project construction should not be held up pending a Part 61 license, since if it was not approved the option exists to ship the waste to another site Section 20.302. Requiring a licensed facility to significantly amend its license under Part 51 or apply for a new license, only for disposal. serves to discourage consolidation of wastes without any benefit to safety or the environment. Section 61.5 - Interpretations. (Section 61.2 - Definitions. Under this proposed provision a person is not entitled to rely on the representations of anyone from the Nuclear Regulating ED-1 A definition of naturally occuring radionuclides should be A-6 Commission (NRC) unless the interpretation is in writing from (included. 🗠 the General Counsel or is specifically authorized by the **.**:.. Section 61.3 - Licenses required. Commission. All employees of the NRC have only apparent authority and any representations which they make in negotations are not binding. This means that the only way to negotiate with the NEC will be to negotiate directly with the Commission or As proposed, Section 61.3(a) (also 61.10 and 61.23) would regulate the land disposal of any radioactive waste containing A-1 ' with the General Counsel. This appears to be unworkable. . source, special nuclear, or by-product material without . . . .

B-269

#### Section 61.7 -Concepts.

A-1

B-1

Section 61.7(c). The procedures to obtain a license are both costly and lengthy. The analogous RCRA regulations provide for an "interim status" which allows existing disposal facilities to continue to operate under minimum standards until such time as licensing can be considered for them. Under the proposed NRC regulation, existing disposal operations will apparently become illegal overnight unless an "interim status" is provided or they are made exempt.

- 4 -

### Section 61.10 - Content of application.

Under this section an application must be filed by anyone dealing with a waste containing a source material. A material is classed as a source material if it contains more than .05 weight percent of uranium, thorium, or any combination thereof. In the case of rare earths a material is a source material if it contains in excess of 0.25 weight percent uranium, thorium, or any combination of these.

The term "containing" is overly broad. The contamination with a source material could for example be one percent. This means that some waste containing source material is covered when it contains one percent of .05 weight percent. This is only .0005 weight percent. This is probably far less than the content of most naturally occurring ores. This is an extreme example of the need for a de minimis provision.

Section 61.12 - Specific technical information.

The requirement in Section 61.12(a) to provide a description of geologic and other data for the disposal site and vicinity is unduly broad in that "vicinity" is not defined. This provision may require surveys to be made on neighboring property which may or may not be accessible to the applicant's personnel. In any event the perimeter of the survey should be more precisely described.

Note that "land disposal facility" as used in Section 61.12(f) includes the buildings. If the buildings are an integral part of the disposal process the inclusion of their description is reasonable. However detailed descriptions of auxilliary buildings such as offices, cafeteria facilities, etc., should not be required.

The Section 61.12(h) requirement should be limited to an identification of "known" natural resources. The applicant should not be forced to undertake expensive resource evaluations of an exploratory nature, such as deep drilling, etc.

Section 61.20 - Filing and distribution of applications.

- 5 -

R-13 The applicant should be required to provide only enough copies of the application and EIS to satisfy the distribution requirement.

Section 61.23 - Standards for issuance of a license.

This section applies to a radioactive wasts "containing" or "containinated" with a source material. Regardless of the

meaning "containing", "containinant" clearly suggests the mere presence of very minor amounts of source materials, such as at A-1 < the one percent level or less. This ties in with the earlier discussion (Section 61.10) that even low levels of thorium and uranium are included within the scope of these regulations.

Section 61.28 - Content of application for closure. ED-1

An explanation should be provided to define the data that are "pertinent" to the long-term confinement of emplaced radioactive waste. The requirement as stated gives undue breadth for Unterpretation.

Section 61.41 - Protection of the general population from releases of radioactivity.

The proposed limit of 25 mrem/yr at disposal site boundaries is C-3 Trather restrictive in view of the occupational limit of 5 rem/yr. In some locations the background radiation lavels may result in exposures above this proposed limit. Either a higher limit would be preferable or one which is based on a percentage of the total background level for a year, providing this does not exceed one-third of the occupational limit. The limit might be stated as an abount or percentage above background with the increment above background being set on a case-by-case basis.

> Section 61.50 - Disposal site suitability requirements for land disposal.

In Section 61.50(2), the criteria to determine if a site is D-50-1 Capable of being modeled need to be established. What role will modeling play in the licensing process? What site characteristics and events will be modeled? Do NRC approved models exist?

 $D^{-5}$  0-2. (The Section 61.50(4) areas to be avoided should be limited to areas having known natural resources. · · · ·

- 6 -- 7 -· • D-50-3 fit Section 61.50(6) is intended to provide that upstream drainage must be diverted, it should so state. Section 61.59 - Institutional requirements. The term diffusion is unclear in Section 61.50(7). There are The proposed rule requires state or federal ownership of the many types of diffusion, including convective, molecular, etc. site before a license may be issued. Provision should be made The designation of the particular type of diffusion is important for an inderstanding and enforcement of the regulation. to insure that ownership would revert to the applicant if the site is not used as a final low level vaste disposal site. It is easy to foresee occurences where, even though a site is approved for low level waste disposal, the project is abandoned prior to disposal due to adverse public opinion or changes in D-50-2 D-59-32 The Section 61.50(8) language is unclear. Note that in other regulations disposal unit's defined as the actual disposal area while "disposal site" includes the disposal area plus a economics or need. It is also very possible that a company may choose to remove low level waste from a disposal site and use buffer zone. It is not clear from the language in this section of the regulations whether no groundwater shall be discharged the land for some other purpose. Further, if the NRC subsequent to till transfer determines that use of the site will not pose a radiation hazard, the applicant should receive the right of from the disposal unit or from the disposal site. first refusal for use of the land. Section 61.53 - Environmental monitoring. Section 61.62 - Funding for disposal site closure and stabilization. The term "information" in Section 61.53(a) is unduly broad and D-53-14 is subject to interpretation. The Section 61.62(f) requirement for surety without proof of forfeiture is very unusual. Needless to say, this type of surety will be quite expensive if not prohibitive. There seems to be no justification for this provision. The surety should be In Section 61.53(b), the type of required monitoring should be specified explicitly. It is understandable that monitoring should be required during operations. It is not clear why Laonitoring is required during construction. available if there is a forfeiture, perhaps also in those cases where forfeiture is imminent. It should not otherwise be available. Section 61.55 - Waste classification. E-1 A detailed set of instructions describing the use and As under the Clean Water Act, self-insurance should be permitted for those who qualify based on an annual submittal of financial reports. Financially strong companies should not be required to furnish surety bonds underwritten by surety companies that are application of Table I should be provided. The instructions contained in the footnotes to Table I are incomplete, vague, confusing and difficult to apply. The list of radioisotopes explicitly listed in Table I should be expanded. In particular, ED-1 weaker financially than they are. Furthermore, it is our understanding that the majority of surety companies will not thorium and the daughters of uranium and thorium should be added. A Class D should be established to cover radioisotope offer bonds of the type being considered because of the long concentrations in excess of the values shown in Column 3. In term obligation which would be involved; and the other general, the procedure for determining the class of radioalternatives, i.e. cash deposits, certificates of deposit, isotopes and mixtures of radioisotopes not explicity listed in irrevocable letters of credit, etc., are quite costly. Table I needs clarification. Section 20.311 - Transfer for disposal and manifests. Section 51.56 - Waste characteristics. The requirement of a notification of receipt by the disposal facility to the shipper (generator) might be satisfied by using M-1 Section 61.56(2) should be delated. DOT and NRC transportation regulations allow certain materials to be shipped in a signed and dated copy of the manifest. Ciberboard boxes (Type A Packaging). For certain low level D-56-4 (waste use of cardboard or fiberboard boxes may be suitable for Section 20.311(d). The reference made to meeting the requirements of Section 61.36 should be qualified. Class A 36-7 <u>-aske use of cardination of intermination should be made</u> on a case-by-case basis. A section should be added to allow
D-st. (bulk shipment and disposal of low level waste based on a case-by-case evaluation, taking into consideration such factors segregated waste need only comply with Section 61.56(a). as the nature of the waste, method of disposal, cost/tenefit analysis, safety, etc. • • •

B-271

5 J 2 39 -4452 Cruzente Road Chapter Fails 1218: 543-621 December 17, 1987 46 '81 EE 22 P3:44 Secretary of Commission U.S. Nuclear Regulatory Commission Washington, CC 20555 Attn: Docketing and Service Branch Re: Proposed Rule 10 CFR 61 Gentlemen: Stock Equipment commends the Commission for the draft 10 CFR 61 ms a start toward criteria for proper discosal of radicactive wasts. Consistent criteria providing required safety for the general public

> The following comments are intended to assist the Commission in formulating well-defined regulations which promote reasonably achievable safety standards employing the best available technologies. They are intended to provide background and comment supporting your overall safety objectives.

and radiation workers will be most welcome in the industry.

1. Regardless of the ultimate regulation and waste form criteria, a key to any program is inspection and enforcement. A number of documents have been issued in the past, most of which have provided sound criteria to improve safety and the overall quality of waste management. Reasonable programs and definitions have previously been issued in Appendix I and ETS3 11.2, calling for permanent. In-plant solidification capability, solidification of all wet solids prior to ship and process control progress to provide reasonable assurance that solidification is achieved. These documents have not been seriously implemented or enforced. These requirements are incomplete with respect to the whole of the low level waste management issue, but could provide a sound, uniform basis from which the segments of transportation, interim storage and shallow land burial could be examined.' The results of failure to enforce the presently existing criteria have been the following:

a) A significant portion of the waste processing equipment Installed in nuclear power plants today cannot meet the present requirements as outlined in ETSB 11.3 and many of the waste generators, due to the uncertainty of requirements, are reluctant to improve the situation. An example of this is the recent occurence at one burial ground where inspection

Stock Equipment Company + Manna Eukong + 1422 Euclid Avenue + Cleverand, Chio 44115, U.S.A.\*+ (216) 621-3054 + Terex 98-0377 Sock scularty Council / a Unit Di General Score / 2 PDR PR 2 46FR35C31 PDI

POR

Acknowledged by card. 12 23 81. mov.

Section 30.32 - Application for specific licenses.

ED-1 On 46 FR 38103 column 3, the fourth line should read "months prior to commencement of." The word "prior" is omitted in the referenced line.

> We trust that you will take our comments into consideration in promulgating this regulation. Should you have any questions concerning Union's comments, please contact Dr. Allan A. Hirata of my staff at (213) 977-6720.

- 8 -

Sincerely, Carleton'3

D-55-B/

· ·

Stock Equipment Company

Stock Equipment Company

۰.

December 17, 1981 . Page 3 . 2151 2 December 17, 1931 D-56-14 GEN-2 (c) Amended 10 CFR 20 Section 20.311 a) Radiological Safety/Guidance for Cn Site Contingency Storage Capacity e) Leachability Test Standards and Criteria (if any) was accomplished on a number of containers of solidified waste which resulted in this particular solidification agent being withdrawn from the market. 5] A number of waste generators continue to utilize portable equipment which does not meet the ALARA concept of provide consistent process control.  $\frown$  3. We strongly support the NRC in taking the lead for basic standards and technology in the protection of the public health and occupational (c) Burial ground states have taken the lead in developing their workers during transportation and disposal. For some of the reasons own, but often inconsistent, standards for disposal. These  $\mathcal{D}$ -56-16 ) listed above, it is our belief that there is an immediate need for this programs are reactionary in nature, in response to a wide criteria and encourage the Commission to adopt same soon, rather than range of 111-regulated waste management programs at generation allowing additional work to progress over a number of years. We also sites which typically employ less than the best available suggest that after the standards are established, enforcement be immediately technologies. An example of this is the recent acceptance established to insure compliance. The criteria should be based upon of "high integrity containers," by some, but not all burlal ALARA and use the best technology available. It is our opinion that sites. (The high integrity container itself is subject to differing guidance as to whether it should provide 150 or 300 10 CFR 61 in its present form is more concerned with the long-term migration considerations rather than protection during processing, transyear service, what the transportation application might portation and disposal where the hazards are the greatest. require or whether the life requirement can be credibly oroven. D-56-16 and economic resources to evaluate new or instants and technical criteria should be delegated to the individual states. Most states have limited technical and economic resources to evaluate new or instants and the state of the s d) Designers of waste processing equipment, plant designers and the waste generators, if permanent installations are installed, establish and enforce standards and criteria. Some states also appear are forced to design extreme flexibility into the capability reluctant to act without guidance from the Commission. of the equipment installed which significantly increases the price as an attempt to be sure that the equipment will, in D-55-2  $\begin{cases} 4... In addition to the three general categories of waste established, ....$ Let us our belief that it is in the best interest of the American public that the document elso include a definition of diminimous radiation level.fact, meet the changing regulatory clicate over the life of the station. . . . . . Based upon our marketing studies it is our colpion that a 5. We agree fully that a key to overall present and long-term safety high percentage of the nuclear reactors in operation and is to place reliance on stability of disposed waste as indicated in currently building will require retrofit of weste processing Subsection 8 of Section 5. Waste fort stability must be well defined, egulpment to meet criteria in part due to the lack of consistent however, and enforced during the discosal operation. wasts form criteria and enforcement. Please note that polymer and cement waste forms in use today greatly exceed 2. The philosophy and ideas expressed in 10 CFR 51 are significant the proposed waste form strength criteria. Some containers and solidified D-52-11 steps toward obtaining the objective of consistent safe criteria. Conwaste forms in use today, conversely, will not withstand a static 50 psiversely, it is difficult to respond to the document in part due to its load. In addition, the dynamic loads due to refilling trenches, compacting D-56-0 emissions, in part due to the uncertainties the document creates in the trenches and handling containers can, in some cases, exceed the strength validity of Appendix 1 and ETSB 11.3 and in part due to other regulatory of the containers and waste forms. D-56-142 cocuments issued since Part 61 or yet to be issued. The overall waste management "puzzle," If you will, is really the sum of all the pleces, 6. Waste form and characteristic requirements as set forth in Section beginning with the generation of the waste and carrying through its 61.56A will permit packaging waste which is in a readily dispersable form treatment, handling and disposal. To reflect intently on one piece with and which contains significant volumes of liquid that will contribute to incomplete knowledge of the other pletes and their interrelationships radionuclide migration by groundwater. We suggest that 140 curies of Cobalt cannot result in an integrated approach, except by chance. Part 61 50 in a 55 gallen drum is of significant concern during the transportation, neers to interrelate with Appendix I and ETSB 11.3. It additionally must disposal and first 50 years of institutional control. There are waste account for: .: . 

Stock Equipment Company

Page 4

Jacamper 17. 1981

solidification technologies available and in use today that can con-sistently place this material into a stable waste form. It is our D-56-RS opinion that the use of these technologies can reduce the transportation and disposal problems as well as reduce the potential hazards due to -future intruders at the sits.

7. It is our cpinion that to meet the purpose of this regulation, all classes of waste should exclude the presence of liquids. The practice 2-56-13) of disposing of ion exchange media by dewatering, for example (which contains potentially, relatively high redistion content) vs. solidification of the evaporator concentrates containing a relatively low radiation content is not logical technically or politically.

> 8. Stock suggests considering in-plant handling, transportation and disposal processes that all wasts containers should be constructed of materials that will not support combustion. This term is applicable in the recently received Radiological Safety/Guidance for On Site Contingency Storage Capacity for low level dry waste storage. We concur with this application, but suggest the need is even more important when considering ion exchange resins or waste solidified in organic compussible solidification ACERTS.

Thank you for the opportunity to comment upon this document. Stock stands ready to assist in any reasonable endeavor if you should desire it. We will also comment further regarding other phases of the program when .... draft documents are available.

Sincerely.

Paul C. Selliame

PCW:re

D-56-6



Argonne National Laboratory has reviewed the Proposed Licensing Requirements for Land Disposal of Radioactive Waste (10 CFR 61) and the supporting Environmental Impact Statement (NUREG-0782). Our comments are attached.

We believe that the proposed 10 CFR 61 rule will provide a workable regulatory framework for licensing and operating new low-level radioactive disposal sites. The site requirements and criteria, operating and closure practices, and standards are conservative out in our opinion are generally practicable.

Very truly yours,

Howard Kittal, Manager Office of Wasta Management Programs

ChK:sfn Enclosure

. .

Actinowiedzed by card

3112290365 811214 508 Pe 2 40FR38081 PD PDR The University of Clauses

APCONNE L'AVERSTES ASSOCATION

### Comments\_on Proposed Licensing

Requirements for Land Disposal of Radioactive

### Waste, 10 CFR Part 61, and on Supporting Braft

### Environmental Impact Statement, NUREG-0782

### Argonne National Laboratory

I. 10 CFR 61

### December 10, 1981

### General Comment

8

Our general impression of the proposed rule 10 CFR 61 is that it is a good document. It should provide a workable regulatory framework for the successful licensing - and operation - of new low-level waste disposal sites. We do not find any serious flaws. It proposes reasonable site requirements and criteria, operating and closure practices, and standards. It implicitly and explicitly states, by virtue of its performance standards, that zero release or zero migration is not expected.

### 8. Definitions (61.2)\*

The addition of definitions and discussions of several terms which have been caltted from Section 51.2 (Definitions) might eliminate some ambiguities in interpreting the regulations. The suggested additions and the reasons for adding them are cullined below.

1. <u>"Long-Term"</u> In Supplementary Information, Section V.3, "long-term" is defined as the time after operations case (presumably the post-closure period). It is not clear that this is the intended definition to be used in the many references to "long-term" in the regulations. If so, further subdivision of the time following cassation of operations may be appropriate because the impacts and problems for different intervals of time beyond closure are quite different. For example, the problems during the period that one can rely on "passive" institutional controls (deeds, records, etc., that allow the owner and potential user to be aware of past use) are different from the problems beyond that period, and also from the problems in the period of active institutional control. A claim [Section 61.7(b)(3)] that is reasonable for a period of the order of 1000 years is that future accupation and use of the site is unlikely; it is less reasonable for a period of the order of 104 years or longer. It has not been established that the allowed concentrations of very long-lived radioisotopes are low enough to permit unrestricted use of the site (which must be considered probable after all records are lost), and there is nothing in the regulations that limits, the period of concern for public health and safety.

\* Numbers in ( ) refer to Section Nos. in 10 CFR.61.

-2-

2. "Discosal" The word "discosal is commonly interpreted to mean "permanent disposition of". If this is the intended definition, it should be so stated and noted that near-surface disposal is not necessarily a permanent

ED-1 The stated and noted that near-surface disposal is not necessarily a permanent index of disposition. Over a time period of the order of 10° years or longer, one cannot exclude the possibility (or even the likelihood) that the waste will be dispersed into the environment. The definition of "disposal" raises a legacy problem, and the implications of this for the hazards of waste with the limiting uranium and TRU concentrations need to be addressed, or at lease acknowledged, in the regulations.

D-56-0 3. "Stability" It is not clear whether the word "stability" is meant to be volume stability, so that the waste will not degrade, slump or collapse after burial, or also shape and physical stability, so that an intruder would clearly distinguish it from soil. If the former definition is allowed, then FUSAP and similar waste is stable; if the latter definition is intended, it is not. If volume, shape, physical stability are required, some time limits may be needed; it might be difficult to ensure shape and physical stability for 164 years or longer unless rather expensive means, such as those proposed for high-level wastes.

C. Protection of General Population from Releases of Redioactivity (61.41)

 $\mathcal{D}-55-9 \begin{cases} 1. \text{ The performance objectives are given in terms of radiation dose.} \\ \text{Since chemically-taxic, in addition to radiotoxic, substances may also be present in the waste, we believe that a general statement, at least, be included to the effect that releases of chemically-toxic substances shall not exceed any local or Federal standards that exist. \end{cases}$ 

2. Two sets of radiation standards have been specified - one in terms of annual dose to any member of the public (25 mmem whole body and any organ except thyroid) and one in terms of drinking water concentration. The latter standard is based on 4 mmem/year for man-made radiouclides. Although it is recognized that the former is for individuals and the latter is for <u>populations</u>, it appears there are two different sets of standards. It is conceivable that releases to the general environment may cause exposures to as many individuals as contemination of the nearest public drinking water supply.

C-3 . Recarding the statement "...at the nearest public drinking water supply...," this supply may not be the one most likely to be affected by the disposal site. The intent of this performance objective is certainly meant to apply to any water supply contaminated by waste migration, and this should be so stated.

> 4. It is possible that the last sentence in this paragraph might be misinterpreted by some to mean that the national drinking water standards are being applied to groundwater in general and not only to public drinking water supplies. We suggest that this sentence be reworded in somewhat this manner: "The waste disposal site shall not cause the National Primary Drinking Water Standards to be exceeded in any public drinking water supply." Additional clarification is needed to make the first and second sentences more compatible in terms of allowable dose, since in the first sentence drinking water could

C-3 {/ieid a dose .. 40 mmen to the whole body and still be in compliance, while in the second sentence it would not.

GEN-2 GEN-2 S. The evaluation of an annual dose to the individual requires a model which allows one to calculate dose from an environmental radioactivity concentration or source term. This model can, of course, not be given in the proposed rule, but it is presumed that guidance in this area will be provided later in Regulatory Guides. The rule could give some indication as to how this performance objective is to be met.

5 6. There is typographical error in the spelling of "radioactive" in the ED-1 Second sentence of the paragraph.

7. This Section is a general statement on standards, although not specifically directed at thesa. Standards are fixed absolute numbers, regardless of the uncertainties in the data on which they are based. Measurements and calculations made to assess performance against thesa standards are subject to uncertainties and to analytical and statistical errors. Thus, if the standard is 5 pCi/l, is a measurement of 5.1 + 0.2 pCi/l in violation? Probably yes, but is a measurement of 4.9 ± 0.2 pCi/l in violation? Probably yes, but is a measurement of 4.9 ± 0.2 pCi/l in violation? Probably yes, but is a measurement of a statistical errors. Thus, if the standard is standards could address this problem in some way. We do not have a clear answer at this time, but it is a technical rather than a legal question, and this may make it difficult to reasolve. Possibilities are (1) spacify a dose standard, e.g., 25 mrem/year, and the probability of delivering that dose, (2) specify a concentration, e.g., 5 pCi/l, and the standard deviation tolerated in a measurement to meat this standard and the method by which it was calculated.

#### D. Protection of Individuals from Inadvertent Intrusion (61.42)

1. It is cur belief that the indevertent intruder scanario is given too much weight and leads to some unreasonably low concentrations in Table 1, for example, in the case of  $9^{4}$ % (0.002 uCl/g). This may not cause any impact on waste disposal, since 54 ND is not an abuncant radionuclice, but this does establish a precedent that could be unnecessarily troublesome.

2. The inadvertent intruder scenario is tanuous at best - it requires predicting some far distant future event for which the uncertainty is large - and should not be the limiting or driving force in determining the hazards.

E. Disposal Site Suitability Requirements for Land Disposal (61.50)

D-50-2 1. We believe that the intent of this requirement is that the water table shall not cyclically rise into and fall beneath the buried waste. Burial beneath the water table could be satisfactory, if diffusion is the controlling rate (as stated in this paragraph), if the travel time is very slow, if the performance objectives can still be met, and if the water table never drops below the buried waste.

### Environmental Monitorine (51.53)

D-53-1 l. It is not clearly stated in this section that the radiological and/or nonradiological (chemical and biological) characteristics of the environment should be determined to establish baseline concentrations.

2. Should there not be a reporting requirement to demonstrate compliance with applicable standards and discuss results? This is implicitly covered in 61.80 (h) (1).

-4-

### Waste Classifications (61.55)

1. The proposed 10 CFR 61 specifically mentions two waste categories although they are outside its intended scope. These categories are: (1) wastes with radioisotope concentrations that exceed the limits in column (3) of Table I [Part 61.55d]]; and (2) wastes that might be exempted from the regulations (Supplementary Information, last paragraph of Section Y.C). On the other hand, no mention is made in the current proposed regulations of the category referred to as "low-activity bulk solid waste" although it was included in the preliminary draft of 10 CFR 51 (issued November 5, 1979). Waste from the Formerly Utilized Sites Renedial Action Program (FUSRAP) would, presumably, fall into this category. FUSRAP waste is within the scope of 10 CFR 61, but it is unclear whether this was intended or incidental. It is of considerable interest why the low-activity bulk solid waste category was eliminated and whether it may be re-introduced at some future time.

2. FUSRAP waste meets the requirements of all of the 10 CFR 51 waste classifications (except possibly with regard to dimensional stability -- see below); it is mainly soil contaminated with very long-lived radioisotopes (mostly uranium and thorium ores and processing residues) at average concentrations that are smaller than the uranium and TRU limits in Table I by a factor of 100 or more. Waste-specific requirements for Class A, B, and C wastes may not be appropriate for such wastes.

3. In raising this question regarding the fate of the low-activity bulk solid wasts category, we are aware of the recent published Branch Technical Position on Discosal or Chrite Storage of Thorium or Uranium Wastes from Past Operations (46 FR S2CG1). The question concerns the wasta identified in category 5 of the position paper, i.e., waste for which long-term disposal at a site other than a licensed disposal site will not normally be a viable option.

H. Labeling (51.57), and Tests at Land Disposal Facilities (61.31)

ED-1] 1. It is not clear where the primary responsibility lies for verifying the character of a waste shipment. Is it the responsibility of the generator (Section 61.57), or the site operator, or the Commission (Section 51.31)? What means will be adopted to provide quality assurance?

D-55-1.

7-55-2

-	13	<b>-</b> 2,	·
<u>.</u>	NU	REG	-/3782

Federal and State Responsibilities. (1.2.3) types would

be routingly measured and the concen-1. The unconstant states and the concen-2 rement states and unconstructions and the concen-rement states and unconstructions and the concent release not call a function of another and the the and the proposed states are not call a function of the states and the the action of the proposed states are not call a function of the states and the the action of the provide states and the states and the states and the the action of the provide states are not call a function of the states and the the action of the provide states are not call a function of the states and the the action of the provide states are not call a function of the states and surveillance duragethe deselogmenticom existing, dedainstitutional control phases? Samples may be taken for analysis either

from (a) the final waste form, or (b) the 2. If they at the final waste form, or (b) the 2. If they at the final waste form, or (b) the 2. If they at the final waste form, or (b) the 1. If they are the final waste form, or (b) the 2. If they at the final waste form, or (b) the 2. If they at the final waste form, or (b) the 2. If they at the final waste form, or (b) the 2. If they at the final waste form, or (b) the 2. If they at the final waste form, or (b) the 2. If they at the final waste form, or (b) the 2. If they at the final waste form, or (b) the 3. If they at the final waste form, or (b) the 3. If they at the final waste form, or (b) the 3. If they at the final waste form, or (b) the 3. If they at the final waste form, or (b) the 3. If they at the final waste form, or (b) the 3. If they at the final waste form, or (b) the 3. If they at the final waste form, or (b) the 3. If they at the final waste form, or (b) the 3. If they at the final waste form, or (b) the 3. If they at the final waste form, or (b) the 3. If they at the final waste form, or (b) the 3. If they at the final waste form, or (b) the 3. If they at the final waste form, or (b) the 3. If they at the final waste form, or (b) the 3. If they at the final waste form, or (b) the 3. If they at the final waste form, or (b) the 3. If they at the final waste form, or (b) the 3. If they at the final waste form, or (b) the 3. If the final waste form, or (b) the 3. If the final waste form, or (b) the 3. If the final waste form, or (b) the 3. If the final waste form, or (b) the 3. If the final waste form, or (b) the 3. If the final waste form, or (b) the 3. If the final waste form, or (b) the 3. If the final waste form, or (b) the 3. If the final waste form, or (b) the 3. If the final waste form, or (b) the 3. If the final waste form, or (b) the 3. If the final waste form, or (b) the 3. If the final waste form, or (b) the 3. If the final waste form, or (b) the 3. If the final waste form, or (b) the 3. If the final waste for (b) the 3. If the final waste

D-55-6

5-56-17

visued. certain action levels, then other radio-nuclides would be measured. The action
<u>Other Issuesvelsenand reversities in the radio-point construction and the second state of </u>

of being background levels. The need for such a classification was noted in theuterorreionaloroperionicaldes zevimesized elistemel by attegulation see Apo. C, Section 5.1.3). Support for a "de-minimus" or comparable classiication hes opening the period of the period of the second state of the second low-Level Waste Strategy Task Force (Ref. 1), the Conservation Foundation fiaTonce Group well Swondon Ridforeriggenester Mergensis (Ref. 2), and the itate Planning Council on Radioactive Waste Management (Ref.3). sampling and more extensive analysis be set with a Keen

-Réference Discosal Facility Costs (3.5.5) eve to practicality ...

▶ . The direct operation cost for environmental monitoring (about \$25,700 per year) shown in Table 3.5 is believed to be inadequate. We estimate that the cost of only the radiochemical analyses listed in Appendix E, page E-55 is about \$40,000 per year. In addition, the cost for sample collection, sample preparation, quality assurance, and other factors might increase this cost by a factor of two. · .

Alternatives to the Base Case (5.2.4)

The EIS mentions use of high-integrity containers, but defines "hich-intecrity" only in subjective terms: LL# shippers and site operators will need a tightening of the definition of "high-integrity" if the use of ... such containers is specified as meeting NRC technical criteria for disposal. Mill NRC provide a cuantitative definition of "high-integrity container", cr will this be left to others, such as state authorities or the private sector?

\* Numbers in ( ) refer to Sections in NUREG-0782, Vol. 2, unless stated otherwis-

- 146=

Classification of New Requirements (5.5.2) Ξ.

1. Sufficienteleabedray effectations introduced and discovering anticu-Very in numid areas. The proposed approach of requiring structural stability for higherBarnend Wasisethereared as naded for greater Thereader, from a for high resident standard in the decontainerized discosal of low-activity vista, briefly discussed on page 5-113. This option should be available to waste generators and site operators for low-activity waste such as building ruble, machinery and attained at the decontainerized for a statistic to trash. Airbone additional and a statistic control procedures.

Eschelatenzieladubilicahrozeranfrermentikasoni emerinentierentie (2003100 (6.2.1) (UCi/g) at Barnwell Table 1 Col 1 Allowable by 10 CFR 61

	-11-230	in lables	<u>b-3 15</u>	a copogespencal	error.	
96-135a	ckercund Ir	machation	(Accend	11x €242-2.7)	3.592-1	

Cs-114\* 4.62-02 4.372+2 1.052-4 I-131The pre-oper-OEtStal tritium concertSTation of 350 pCi/K.ESEEMout three times gressar than it Islandour (northern [Numish]) area. The gress-sipha and beta Semégátrations ar greggenable. 4.372-2 1.052-4 Cr-317 1.12-01 4.37E+2 2.5CE-4 He-Siderences 9.2E-01 4.37E+2 2.10E-3 4.37E+2 1.1CE-4 Sc-397. Managing 72-Level Wastes: 137752 Sc-90. (August 1520) (August 1520) 2:50-30. (August 1520) 2:50-30. (August 1520) 2:50-30. (August 1520) 2:50-30. (August 1520) Ca-141° 4.5-02 Ma-542° Towards a lasticnal Policy for Maraging Low-Level Addinactive Waste, " Ru-103° The Conservation Foundation (June 1981). 7.802-5 4.5-01 4.5-03 "Low-Layah Spicective Waste Margament: An Economy Systement," State 2.32-03 State 2.32-03 Council on Rasi costive Waste Managament (July 1981). Ru-106\* Te-993. C-14

All with half lives < 5 years</li>

\*\* Soil density 100-15/ft3

B-309 B-277

shows the average concentrations of a number of isotores which records indicate are buried at Barnwell. These concentrations are compared to the values from Col. 1, Table 1, of proposed 10 CFR Part 51. Also shown are the ratios of these average concentrations

STATEPLANNING BUREAU State Capitor AP.9 7019 Cifice of Pierre, South Cakota \$7501 COMMENTS: 10 CFR 61 PROPOSED RULES 505/773-3651 Executive Management: The South Cakota Capartment of Water and Natural Resource's Radiation Program has reviewed the summary volume of CEIS 10 CFR 61, and we recommend approval. '31 EC -9 P2:35 The following are comments relative to minor discrepancies. MOPOSED BULLE FITT December 2. 1981 1. Volume 1, page 5, section 2.1, paragraph 3. 40, 51, GI, (46 FR 70 The word "generated" should be replaced by "produced" because it may create confusion regarding fuel cycle power "generation", and the "generation" of LLW. Therefore, the last sentence should read: Institutional LLW production 38081 Low-Level Varia Lidensing Branch Division of Waste Management Diffice of Nuclear Material Safety and Safeguards will account for about 19% of the non-fuel sources. Societ Currots of Nuclear Regulatory Commission 2. Volume 1, page 9, paragraph 3, line 12. Washington, DC 20555 enorated sure CM "The most important . . . ". The use of "geometric means" should be justified RE: SAI≢ SC811020-E15, Volumes 1-4 in lieu of arithmatic means. CRAFT EIS on 10 CFR Part 61 "Licensing Requirements for Land Dispa of Radioactive Waste" 3. Volume 1, page 19, Table 5.5. The column heading "Costs & Immacts" should define units in column Dear Sirs: (i.e. dollars, etc.' Thank you for the opportunity to review and comment on your draft environ-4. Volume 1, page 27, paragraph 2, line 8. mental impact statement on 10 CFR Part 61 "Licensing Requirements for Land Disposal of Radioactive Waste." The use of the term "daughters" should be replaced by "proceny" likewise throuchout the proposal. '81 050-9 P2+35 Attached are some notations made by the South Dakota Department of Watar and Natural Resources during the course of their review. You may The following comments pertain to Volume 1, Attachment A, Proposed Rule 10 CFR 61: Licensing Requirements for Land Disposal of Radioactive Waste. want to take their comments into consideration when you prepare your final report. Yolume 1, Attachment A, page 38082, column 1, paragraph 3, line 6. Overalf, South Dakota agencies felt your impact statement deserved The word "numbr" should read "number" ED-1 recommendation for approval. Column 2, paragraph 1, line 11. "nonradio-active" should read "nonradioactive". 2. Sincerely. Volume 1, Attachment A, page 38084, Column 1, paragraph 1, item(5) "Stability - " The statement appears too generalized and vague. It should read: Stability - Stability of the disposal size over the long term (100 years) is mandatory to prevent loss of site integrity; The potential for migration and transport of wastes to offsite areas should be virtually eliminated. (As an example). D-50-9 Tony C. Metry Commissioner STATE PLANNING BUREAU S. Volume 1, Attachment A, page 38084, Column 2, line 24. The word "if" should read "it". ED-1 )s. Volume 1, Attachment A, page 38085, Column 1. The classification scheme presented covers present waste streams but contains no provisions for future waste stream developments. If a "miscellanecus" class "C" which is characterized by activity can be included it should provide a time-buffer for future amendment needs. 



Attention: Docketing and Service Branch

Final Draft: 10CFR Part 61

Dear Sir:

I am enclosing herewith various comments on the draft of 1007R61, dated June 29, 1981. I hope these comments can still be considered and incorporated before the issue of the final

version of this regulation.

Yours sincerely. Geoffrey G. Eichhold

Regents' Professor

GGE:st Enclosure

s i soged by the

AN EQUAL EQUCATION AND EMPLOYMENT OPPORTUNITY INSTITUTION

COMMENTS ON FINAL DRAFT, 10CTR61

dated June 29, 1981

Geoffrey G. Zichholz Georgia Institute of Technology

### General Remarks

This proposed regulation has gone through so many drafts and public comments that it may seem invidious to introduce further criticism. For that reason in the following emphasis is given to major points of philosophy of regulation, operational criteria and anticipated problems in administering the regulations. It had been my impression that 10CRF61 would deal specifically with <u>lowlevel</u> waste, but that distinction does not appear in the final draft (compare 43FR49811). There clearly is a distinction in emphasis in dealing with nearsurface disposal of low-specific-activity and deep disposal of high-level waste in mined cavities. Subsequent comments address themselves primarily to lowlevel waste. This point is, in fact, covered indirectly in f61.2, Definition of "Waste".

I do not find the argument in Section IV for dropping the LLW designation a compelling one. Somewhere else, possibly in Parts 30 or 40, provision may have to be made for performance criteria and licensing of LLW incinerators and other volume reduction schemes.

### Specific Comments

**(**1.

6-3

I feel the description of performance objectives is good, and it is important to stress these over "prescriptive requirements" that necessarily reflect the current state of the art. However, the phrasing of \$61.41 may be too concise to permit ready implementation. In particular, it is not clear how such annual doses will be estimated, whether they will be based on

· · ·

ED-1

on unrealistic accuracy of measurement. Is it meaningful to have a actual observations or on predictive models, incorporating "realization -1-55-11 lest 2650 acory Commission "January 13 1982 "January 13 1982 It is not clear what constitutes capability of being more to Hill and checking to the line of the can be. 1 4.50 and decempinize provide promersical equidance caefining wasserfeel criterica. stability, the Commission has proposed to characterize a struc-., tura lar staller waste forst as grey with "di langtain its spysical estimate dimensions within 32 and its form, under the expected disposal con-ditions of a signation e.4 7-50-2 20 Specifically, shis definition excludes many thermoplastic binders; e.g. asphalt, from conformance sug to the fact that a chertoplastic = well-Лc. =eć monolith will, gradually creep or flow uncer a 50 psi compressive "TGachigs chief amplication" a tase compressive supergrampargae states draf D-56-5 a significant role in soil subsidence. Experience shows that sub-1/51. Stente : reletsoniempiocestery uf Miesofaultichiemfarting isatalyen a time the manner in which waste packages are placed in the ground and D-5 scalesoid bachfildeger WastCompissiepostoogniztsfeltkeinedathateresis paragraph 61.52(a), (4) and (5) in which requirements for orderly placement bi-S(1),(v) and (v) in which redurements in other y picturement ins these was bainside set blicks do all interevoids are actablished activity werner a Pfleiderar contants that with the precautions astablished activity meas Armer's Printerer's contends that with the predutions statistics to the prevent voids in both the ground and in the waste package; is, Paragraphs 31.52(a),(4) and (3) and artso(b);(3), follower (contended to the compressibility of the monolith is a more appropriate measure of "statistic the monolith is a more appropriate measure of "statistic the monolith is a more appropriate measure of "statistic the monolith is a more appropriate measure of "statistic the monolith is a more appropriate measure of "statistic the monolith is a more appropriate measure of "statistic the monolith is a more appropriate measure of "statistic the monolith is a more appropriate measure of VOL Tep D-55-2 will behave in much the same manner as an incompressible component seiption theusigroundengesser. stressersesten bliezeen band bankeilling produces an environment more akin to a hydrostatic environment in 5. S 41 Angraidareer Larsidarer are intaista " easibier juna coused an perticularly foregoing rationale and the fact that thermoplastic binders offer un cratumificaer permanence, isproverunts, over antication for a solid antication of the solid antication for the solid a ED-1 Les certinision which recontizes volumetric incompressibility of mono-ED-1 729171130159H141.66454, Mayietsoosticsmanne for H-3 and C-14 according to thetheseternase Solbiglil establiquids form of a disposal container as a means to provide stability; i.e. confinement and isolation capability. Claik-Source assumed filesustatics is a continue and isolation capability. Claik-Source assumed filesustatics is a continue and isolation capability of poration assumes that this is a tacit recognition of the solity of chese anion dispatity.copsiles a provide isolation, capability, a linguage this is a construe from previous Commission off isology, seemer a of affecterer forporation previous Commission off isology, seemer a of a file cerer forporation previous Commission off isology, seemer a of the potential to provide Congressing commission of the solfficient and the potential to provide Congressing commission of the commission of the solfficient and the potential to provide Congressing commission of the commiss D-59-3 ap, nowever, have some serious reservations regarding high integrity shou containers as they are turrently used. For example, high integrity containers are structurally designed to withstand normal conditions Some of chaspentension specifies al manifim de frenition test navitared musticiantegrity containers are routinely subjected to transport and mandling vion,sizuezions apere/anformentei-ispacture/ercs=apuacerrantsportidistr implaing cased the casign basis by a considerable margin. This combined with ED-1 the fact that high integrity containers are predominately used to contain extremely dispersible waste forms with potentially nigh NORMAL STREET

U.S. Nuclear Regulatory Commission as for 1-129, Cu-125, No-29 January 13, 1982 when this implies that the quantity stated is both minimum and much Class 7. specific atteitesteat statiquestion the wiscon of the currently established design criteria. In the past, the Commission Considera recognized inte beokinisticationisticatise sectices the locantie | chip, for dispersion of radioactive materials due to handling accidents. Strata -2: 40 EREtiene to belderes snass to tedra sadensis -22 assent tale safetuine guard to discersion of radioactive materials in fight of the nonconservative design criteria imposed on high integrity confizures. tainers. Werner & Pfleiderer Corporation requests the Commission Sel. Statut reexamine, the design criteria for nich integrity containers דבלב סאפר באמר בסאמנשי בפרחק באייניה בהפרבמכה בהו אני של שליט שליט אפרבומהופה בארפהצי ומיים proposed rulemaking and look forward to the Commission's consideration of atmo feragoing Amycerisposal or storage of large quantities of radican 'w noble gases should be considered Mary State ontonext and is probable --less desirable disposal option. df2053.0091512529.0009200000 chemaines contravene the provisions of b(1) and (2) Javid M. Enegess. (11. I support the early development of usings "de minimis" classification "AZARCOUS Waste Treatment Systems (though not that term) to avoid filling up expensive ficilities with suse. CNE/WJK/cs tially nonhazardous waste. This should be done as an early data to until t the provisions of \$ 20.306 into Part 61. 12. \$ 51.5%: Whose responsibility is it to provide sorting and labeling ... waste packages? The main value of this would seem to lie in escablishing compliance with transport requirements, Title 49 and 1007871, and a. classifying the waste on site. 13 / 1 61.55 Land Comership: Although continuity in repsonsibility is clear. desirable, it is less clear why the land has to be actually publicate summer as long as some lies is maintained to provide for official acc. ; moritoring and inspection is needed. Does this paragraph rule out reviewless owned lands?

14. (161.2, 61.59, 61.62: The term "custodial agancy" must be defined.

B-312 B-280

15. State and Tribal Governments: PL95-573 and \$61.70 are the first instances Wal as the section of the sector of a par with what extent Federal regulations have overriding ..... JANUART 12, 1982 and participarty dribit is, or when cases relevant tribal law as quoted in \$ 51.72. In case of conflict between (10) 223 tribal positions does either prevail automatication and a prevaire TI : LIUR ELICOONS Secretary of the Commission Jeale-Sumaser jadgalait derefasatienda (46 FR 51776 Jashington, D.C. 20555 (16. Manifest Tracking System: § 20.311 is liable to greenerer pruntain of parts Attention: Docketing and Service Stanch work if not properly applied. Provision should be made for--detal (46 FR 38081 Subject: DOW COMMENTS, PROPOSED RULINAXING 10 CFR bi (46 F & JOUX) autorisations to ship low-level miterial of a recurring type, subject to The subject proposed rulensking was published in the Federal a subject proposed rulensking was published in the Federal and the subject of the subject of the subject will lis on on October. 22, 1981. NUBEG-0732, a draft environmental inthe supplicants a meeta specific scients and so Sguidhatbia should be subject 4. 2 support to 10 CFR 51, however, NURIG-0781 has not been published theckBoy the mittenses of NEWShopeelanski THIS Chair Schi relevance to are dated September 18, 1951, and were submitted to the ( A. | pros letter sister infrank alls frankt all freike soonet then after ( A. | pros letter sister that additional commands will be sub-The Completion has extended the inment period for 13 GFR 51 17 to 3 20 might 1 19 Forment Dow received NURIC-0732 on October 14. - 1931, and a draft Branch Technical Position (172) on Waste 13705-3042242) Decords Saragraph, <u>ifeseppitather</u>utechienteremisory licenses, 10110431 should state the type of information required, that a limited environmental ED-I SUIIS-0782 GEN-Treport must be filed, or alternatively under what conditions the Commission On pages 4, 5, 5, and 7 of the comments dates September 13, 1931. Drygtated concern that NVRIC-0732 may be placing too nuch explains on planary analysis and overlooking the condents of ALARA and best available technology and ignoring the needs and objectives of assuring protection of the vorkers, the general population, and the anvironment during the operation of the disposal facility. Review of NURIG-0/32 has confirmed that the concern is valid and justified. STP on Waste Fors Dow connects dated November 15, 1981, were submitted to Mr. Robert I. Browning, Deputy Director Division of Vaste Managament. Copies of the 317 and Dow comments are attached. Please consider then as part of the Dow comments on 10 072 pl. IN SPREATING DATE OF THE DOW CHENCAL COM

2 2 4

·31 550 Fanuery 12. 1982 Grossville. TH 38555 December 21, 1981 Secretary of appre Corristian opportunity to consent on 10 CF2 61. ATTIL Pockeping and Bestore and technical criterian and property at the state of th RE: 10 CFR Parts 2, 19, 20, 21, 30, 40, 51, 61, 70, 73, and 170 LicSASSADe Proquirements for Land Disposal of Radioactive Waste The lack of coverage of socio-economic impacts of low level radioactive Refailities upon a community is most disappointing. Cur wasta (11 county recently whent through the experience of being considered for a lirw favores a Solidification Services a greater insight into the current practice of licensing procedures. As TH is one of the 26 agreement.stateset it would be responsible for regulating such a facility, but it became clear that the criteria being developed by the state with the assistance of COAG Idaho DOE also did not include socio-economic impacts either.

Sumberland Sounty is a rural county, so it seems it is automatically a prime candidate for a waste facility. But most of its income over the past ten years has come increasingly from tourism and second home development. Even the slightest suggestion that the county was being considered as a dump site for linw created an unfavorable climate for potential investors to consider the county for further development.

It seems that certain land uses within a rural area are not compatible with a llrw facility. For our county's experience, it is certain that tourism and second home development are not enhanced by such a facility. I am sure that there are other land uses that conflict with siting a ling in the vicinity: recreation, wilderness areas, prime agricultur l Acknowledged by send. JZL30/51 - 54

B-313 261

Rt 6 Dex 526

land, mineral reserves, among others. Sensitivity to the goals of a community being considered for such a facility will go far towards finding a community willing to host such a facility.

SEN-4

On the reverse side, some current land uses seem to be more appropriate for the placement of a linw facility: sites near or on existing nuclear facilities (nuclear plants; DOE reservations - Cak Ridge, Hanford, and Savannah River, among others.) The concentration of such multiple nuclear uses, if other technical criteria are met, certainly seem a practical way to go as the surrounding community already made an earlier commitment to a nuclear facility and might be more willing to welcome another phase of it.

The other area I would like to comment on is the long term care of facilities once they are closed. The U.S. Ecology Corp. has a rather messy history of abandoning line facilities in one state and opening others elsewhere. In the case of XY and Illinois, these states are left with perpetual care of dumps that are becoming each year a clearer public menace. It seems rather shortsighted for the government to allow a private company to make profits off of a inadequate operation then for the state to assume perpetual care for it once the situation becomes too hot for the company to handle.

I believe that such facilities should be operated by a quasi-public corporation that does not have profit making as its primary motive. Perhaps the level of competency would be higher than that experienced in the past. Perhaps not. But the issue of perpetual care would continue to be the responsibility of this public corporation rather than Sit being turned over to a state that is generally unable to adquately .D-59-3 fund cleanup operatiions as needed.

4

I appreciate the opportunity to present these comments.

Sincerely,

(rue granf lo

Louise Gorenflo

3

### CORRECTION NOTICE

TO: Recipients of PR-2, 19, 20, 21, 30, 40, 51, 61, 70, 73, 170 (46 FR 38081)

### Comment Number 72 of the above rule has been miscoded. Please remove it from your file. Number 72 will not be used again.

## Docketing and Service Branch 1/7/82

. . .

Osvald U. Anders SOl Linwood Dr. Midland, Mich. 48640

December 30, 1981

(46 FR

### 122 JUL -4 P5:09

Secretary of the Commission 22 divided Py U.S. Muclear Regulatory Commission attn: Docketing and Service Branch Washington, D.C. 20555

Sir:

This letter is in reply to the actice published in the Federal Register Vol. 46 No. 142 of July 24, 1981, inviting public comment on proposed anmendments to the Nuclear Regulatory Commission's rules to provide specific requirements for licensing of land disposal of radioactive vastes. I understand that the comment period has been extended to December 31, 1981 and I hasten to submit these comments with the postmark of that day.

The proposed rule making stands to become a landmark governing the disposal of low-level radioactive vastes as it, together with its detailled L-volume Draft Environmental Impact Statement, represents the fruit of considerable and long-time effort of the Commission's staff, embedying their philosophy of government and experience with controlling the nuclear community.

As a member of that community I have personally felt the heavy hand of this control, its slow and inscrutable working, which, with all the good intentions and byzantine ritual of decision making, discourages and kills private initiative due to the mismatch of relaxation times between legalistic government action and the mismatch is bility of the "regulated" to survive long enough to benefit from the action.

The present document substitutes recordsceping and usny layers of surveillance for technology brailable yet poorly understood by both regulators and the lay public. It substitutes "common sense" and proof by delay and survival for sound technical understanding and will invariably succeed to kill what it purpotes to regulate and make safe.

I shall direct my comments toward two areas: The licensing process and the Waste disposal technology, both subject of the proposed rulemaking which is with detailled emphasis overdoing the former what it lacks with respect to the latter.

The establishment of a disposal site and the operation of a disposal facility for low-level radioactive vastes is a serious enterprist requiring responsible people. The requirements for establishing such facilities [urgently needed by . GEN-44 the nuclear utility industry and others, as laid down by the proposed document, will de facto eliminate any private enterprise from the contention and reserve it to utility consortia and branches of the "government" which can proove fimancial capabilities for the long times contemplated and can efford the financine of the legal circus surely to develop upon the promulgation of the propo-" -sed document. Indeed much emphasis is placed on the establishment of this fimancial capability which is clearly seen as needed to keep the hordes of lawyers and "experts" in business who will migrate, like a minstral show from hearing to hearing as site after sits in endless revolution attempts to attain the re-8-9 quired 5-yearly extension of its existence permit. Personal experience regarding to these types of "public hearings" carried out presently by the Commission encourages me to nothing but scepticism, as such hearings typically drag on

Acknewledged by care. 115 \$2 maly
- 2 -- 3 -Mr. Robert E. Beswhiag - - hr. Robert E. Browning Novamber 25, 1981 ----November 23. 1941 not for months but years, consult "experts" who make a carreer of it or house terion in technical terms, for which the nuclear community has been waiting vives and people not at-all familiar with the issues, who however qualify is for more than a decade. It leaves the interpretation of this thrase forever concerned citizens and discuss subjects far afield. 3. "Wirferes vith Testigling Sechanical strength based on the second strength the second strength the strength of the second strength of the second strength strength strength for at least a decade per site. The two understok for second to fill make pases that Wirfer 1993, he facto this must be cleared with the "eventual output as the side for site, with the second strength should be second of the strength tight side strength strength sold contained is the second of the the en-tioner as the side for side with the second strength strength strength the second of the side of the second strength strength side of the s Onde tagain laryers presedures of also pays with will prove the adainiting during cornets varians parie Maringscommeres encust stertant datauto set such D-55-1327 elest indiaste et den Douthan I begettien rentissienis andest one such. I propose some fraction of the specific activity of some comunity mined uranium ore, 3.5. Singereine equivalent to the highest-specific-activity topsoil accessible to the public within the continental United States, whereby the equivalency can be B-9 defined specifically for each isotope and mixture of isotopes. Such a simple basymitics would settle the question once and for all and put some floor under Arthuisenesions, aimittedly at the expense of the legal community. Grouinsade: Salicontive vaste can be made to defend itself and can be stored in such a way unified the party to the biosphere are small enough as to be of no consequence. S1 - fighting the present document places the emphasis on site and site structure. Vore effectively the waste form itself could be made, and should be made in Pfimetonabe insuares office agents suspis relitiones givings would be beld aburmalOgThurner.2. Interuppiersian is yars sougrange and the preparation for theibersimsserbin destrieving saeder stegestery incurentation etc. by fithe largers of 1406 abtervences, the legal extensions granted during the "discovery place" etc. drag out the process for months, say acthing about the a way so as to localize the radioactivity and not release it, no matter what fa 6 the disposal site hydrology and disposal practice. The method cited in the datica properties should be considered. NRC 7-8-8 decument is typically the most unreliable and obsolete method of solidifying Stilfernient? to the isquence to the issues a la condition of the is con-Ricuids i.e. solidification with urea formaldehyde, a technique long recogstrated and the transferred to and the state of the state mized as unsatisfactory by the technical community. Solidification of acueous in the intense, the inclusive for attrials incluses is currently yyears, the isotness would be required to submit a license reneval application.... A public backful tolding of offering is and sharing the the facility would be dis-currently enough the top store up and big with the facility would be dis-currently indicating the years! I have a submer to the facility would be dis-currently applied to years! I have a submer to the facility would be dis-currently applied to years! I have a submer to the facility the facility could an enough the submer to the facility is a submer to the facility of the facility is a submer to the facility could an enough to the submer to the facility the facility is a submer to the facility the facility is a submer to the facility the facility to the facility to the facility to the facility the facility to the facility Wastes by incorporation into plastic matrixes, renders the radioactivity solid, 3. virtually immobilized and unleschable, requiring less stringert site criteria. 1 D-56-19 If formed in large enough blocks, typically up to 50 ou ft and larger, they would also discourage the indiverting intruder, as it would require considerable effort to nove or break up such waste forms. Waste treatment by volume reduc-(assuming the trupt bed is aloud by TU feet alovernating the operation, Openhe histories first first there is an alover for the second tion and subsequent solidification into plastic can effectively deal with 14quid vastes. It can also accounciate vastes composed of pulverized or granular solids. Consideration must be given to sufficient cross linking of the plastic to give the structure sufficient rigidity and comesive strength to counter osactic pressure and swelling which can destroy the effectiveness of asphalts and other weak matrixes. Absorbents as mentioned in 61.56(a)(3) are definitely They "gampetos" with the Reastey withly firewryes by required to finance the dis-D-56-/ \unsatisfactory, representing obsolete technique. Leaching water will readily possigni trisconstmi diffethe severas farm rip. frasher, pays will have to keep liberate the radioactive material from such matrixes and render them mobile. the CERcetible 3 Wil businesseeing messiof Hispersip Lessee, will become so excritiant as to greatly discourage the "generator" to send anything for dis-B-6:4 The proposed rules place some weight onto the container of the waste, 51.36(1)(1). possilisie an I father store his source this con site till be dies. The typical container for low-level wastes are steel barrels. In the discosal I site environment these containers are prome to rust, even in the desert if buried, Anyone in any way involved in this activity, has to be prepared to face the The which he succession non-compliances which are associated with byzantice D-56-8 D. lapi and thus cannot really be relied on to not leak or give structural strength The viscille bureaucratic non-compliances which are associated with by antice ow rules of record keeping that no human can fulfill. Whether warranted because tion. Of public starts of the one of the second fulfill. Whether warranted because tion. Of public starts of the one of the second fulfill. The second should be the tion of the second to the vestepile in the burial trench. Mon-deformability of the veste form itself and efficient backfilling of the voids in the pile will avoid sinkhole formation causing rainwater and surface-water to be funnelled into the filled trenches. I am very much in support of segregation of wastes by relative toxicity and hazard. I believe that a somewhat more sophisticated, potentially computerised, scheme, than that suggested in the document, can simplify procedures, save land and render less of it in need of perpetual care. For classification Now, why does the regulatory agency find it necessary to propose these pro-D-55- 125 the amount per package and halflife as well as type of radiation emitted would cedures? It clearly sters from lack of self-confidence of the agency relative he the important parameters. With segregation goes the requirement of stability to the technical matter, its paramoid fear of the public and inherent distrust, potentially justified, of anybody crazy enough to try to submit himself and ageability. For solidification products the compressive strength, leachability, radiation resistance and biodogradability should be established prior to into these rules. terment to provide for proper classification. Caly unsuitably packaged vestes Now let us look at the other side of the picture. I am more qualified there need to be placed into expensive vaults. The price would discourage the generator from sending such to the disposal facility. as radiochemist.

. .

C-4 { I believe the document places too much consideration on the indivertent intruier.

8-316

(The document states: .... remaining radioactivity is at a level that does not

D-55-123 pose a danger to public health and safety." It doliberately avoids the cri-

B-284

CONFEDERATED TRIBES AND BANDS SALABUSHED BY THE TREATY OF THE A 1938 GENERAL COLNER. Vakin:a Indian Nation Els meeds are readily taken care of by placing the required cap onto the burial TRIBAL COUNCIL \*\*\* ; \* \* \*\*\* trench thick enough to divert surface water and remain unaffected by the freeze POST OFFICE BOX 161 . . . they cycle. No truly "inadvertent" intruder yould burrey to more than 7 ft under Secretary of the Commission "Carting was made as a the ground without considerable deliberation and knowing what he is doing, Ee January 13, 1992 December 30, 1981 ?ace -2-22 JUL-E 23:11 site and its proper use e.g. as a golf course etg. 1942 7-59-2 ens R. Dale Smith, Chief Low Level Waste Licensing Branch I have that the above comments can still be considered for making the proposed Division of Waste Management addition to 10 CFR a workable document with which free enterprise, and the nuclear establishment can live. Safety and (74) ``-22±0 . . . . . . . Secretary of the Commission (46 FR 3808) U. Very'uninky Pepristory Corrission 11.8 (Arton we brieve finds this requirement annot be interest function for the interest 51.81 (A), an environmental nonitoring program must be rain any voltage the interest interest 12.21 (A), an environmental nonitoring program must be rain any voltage of the interest 12.21 (A), an environmental nonitoring program must be rain any voltage of the interest extensive reports that must be maintained at the site, the (Mighter A), lists ain the (A) extensive reports that must be maintained at the site, the (Mighter A), and (Mi Unshington, 7. C. 20535 Artington, 1. (. 2000) Arting Dicketing and Service Branch Arting Concerning the Ticensing B-5 Cavald U. Andersiperants for Land Disposal of Padipactive Waste" 201 Linwood Dr. Dean Signa Mich These connents are submitted on behalf of Hiddle South Services. Inc. (153) in conmercian with the above subject matter. "TS is a technical services and support company for the widdle South Utilities System ("SU), which serves the electrical energy remitments of approximately 1,600,000 customers in portions of Arkanses, Lousiers, Mississippi the rule making. and Missouri. The "SU System has five nuclear units and, therefore, is concerned with the development and implementation of the proposed rules. faste Treating the Yakimas have a Treaty with the Federal Covernment, since the yaximas have a Treaty with the Federal Covernment, since (1995 estimated and the state of the state 1 MS5 has raintained a continuing interest in the development of 1007761 over the last few years. The current rule, as proposed, incomportees a number of inconvenents over DC-52 earlier drafts, and MSS wishes to contend the VPC on its continuing efforts towards further inprovement. Rowever, MSS continues to have a fundamental concern over the lack D-55-2 )of a "de minimus" waste classification and the operability of waste separation according ( so the proposed Table 1. We also believe that other provisions of the proposed rule can be inproved. The purpose of this letter therefore is to notify The purpose of this letter therefore is to dothly evaluate these rules so that we may present to you our publications commu-words which is serving the community of the first of the list operators. Not tration would place an overly the consulty and serving our constants. Not Oue on the unavailability of resources, we were not able to movide the detailed or ments on the proposed rule that we would have liked to provide. In this repart we have GEN-77 carricipated in comments generated by the Atomic Industrial Forum and Stilling Nuclear Wasne Management Stroup and support the comments that are to be submitted by them. The following specific comments are submitted with regard to the propsed regulations. wastes are composed of a minume of various increase a varying concentrations and it would be enumerally difficult if not increased as separate and categorite our wastes Russeit Jen according to Table 1. "Te Minimus" Waste Classification . -Russell Jim, Councilman MSS agrees with the Commission that there is a need for a "de minimus" classification of wastes that would be exempt from the 10 GPR Part 61 regulations. We believe that this D-55-20 Yakima Indian Nation is an issue of major importance, and we unve the Commission on establish this classi-Maxian in as short a tire frame as possible. To date, two types of wastes, Wydrocen-1 and Carbon-14, have been recoved from regulation as redicactive vestes, unter certain conditions, in 10 CTR 20.306. We believe that this regulation establishes a mod rodel for the classification. It would be impractical to shorten the merating hifs of existing or a newly licensed size with wastes that could be discosed of at a municipal solid waste discosal facility without enfancering haran realth or the environment. 1/22/82 ener Action view set by card ... 1.7 82. maly ... LUCHT COMPANY - VISSISSIPHI POMER & LOHT COMPANY - NEW CRUZANS FUBLIC SERVICE NO. B-317 B-285

يا يُد التو

. .....

U. S. Nuclear Regulatory Commission Attention: R. Dale Smith Pace Two December 29, 1981 THE UNIVERSITY OF TEXAS MEDICAL BRANCH GALVESTON, TEXAS 77550 Section 61.56(a)(8) places an unnecessary and \*2 JUX-5 P3:12 prohibitively expensive burden on licensees who use biological material, particularly medical December 29, 1981 Office of Environmental Health & Safety users. Known infectious material should certainly . Room 106 Basic Science Building be treated, but it is unreasonable to require Area Code 713/765-4190 treatment of all biological specimens. Area Code 713/7654190 1 . . . . . . Low Level Waste Licensing Branch, Division of Waste Management 7. It is not clear whether or not Section 61.56 will permit continued land disposal of liquid scintilla-Utision of waste Management Office of Nuclear Material Safety & Safeguard U. S. Nuclear Regulatory Commission Washington DC 20555 tion fluid. If not, this will present a problem to many licensees, particularly those in agreement states that still require that liquid scintillation fluid be disposed of as radioactive waste. 40, 51, 61, 70, 73, 170 (46 FR' 38081) Attention: - R. Dale Smith ..... I hope these comments will be of benefit to you in promulgation of these regulations. In general, I feel the approach taken is Gentlemen: reasonable and workable. Thank you for the opportunity to comment. Sincerely, I understand that you are still receiving comments on the pro-posed Part 61 and associated amendments to other parts of the regulations. Please consider the following comments: Leo Wade, Ph.D. Director 1. Knowing the history of licensing delays within the Nuclear Regulatory Commission and particularly those associated with major licensing actions, the regula-LW/ar cc: Bureau of Radiation Control Texas Department of Health Att: Mr. David Lacker 1100 W. Act. et tions should require the NRC to either approve or deny license applications within a specified period ~B-3 of time. Applicants who have options on land or own land that could be used for other purposes may incur substantial financial losses due to unreasonable delays. Therefore, they should be compensated for Att: Mr. David Lacker 1100 W. 49th St. Austin TX 78756 Mr. Saul Harris Nuclear Program Mgr. Ediscn Electric Institute 1111 19th St. N. Washington DC 20036 delays. Therefore, they should be compensated for
those losses if the Nuclear Regulatory Commission
unduly delays licensing action.
2. Section 61.4. It would probably be more appropriate
to express limits in terms of dose commitment rather than annual dose, particularly in cases where releases EDmay be intermittent or one-time only. mashington DC 20036 Concentrations listed in Table I of Section 61.55 are unduly restrictive for beta emitters not specifically . . . . listed. listed.
Section 20.311(d)(1)(2) unnecessarily duplicates the
requirements of Section 20.311(e)(3) and 4. There is
no need for the licensee to perform these activities
if they are to be performed by the processor.
Section 20.311(d)(3)(e)(5) should be more specific
about the items to be included in the Quality Assurance
Program.
Addnoweiged by card...[1]32.mdV... 1-1 A state of the sta B-286 ŧ

30 December 1951 30 December 1981 182 JR -7 92:14 Para 51.52(a)(3) "A buffer zone of land must be maintained between any buried waste and disposal site boundary. The buffer zone shall extend at least 100 feet outward from the outermost waste disposal site." Comment. A buffer zone is Secretary of the Commission required for safe operation however the small distance of 100 feet is questioned. US Nuclear Regulatory Commission DICKET BURGER Presumably, from descriptions given in other portions of the proposed rule, the Washington, DC 20555 PROPERTY RULE LAN disposal site will be located in a sparsely populated area so land acquisition is not a problem. Therefore, it is suggested that a buffer zone of at least 30,40, 51, 61, 300 feet and possibly as much as 500 feet be mandated. This increased distance will be advantageous for working space and remedial activities should an Sir: accident occur and will prevent close-in development of property by adjacent property owners during periods of operation, closure, and institutional control. The following comments are made in response to the Proposed Rules describing Licensing Requirements for Land Disposal of Radioactive Waste (Ref.: Federal Sincerely. Register/Vol. 46/No 142/Friday, July 24, 1981, pg. 38081 ff).

Para 51-51(a)(b) "The disposal site must be designed to eliminate the contact of water with waste during storage, the contact of standing water with waste during disposal and the contact of percolating or standing water with wastes after disposal." Comment. The word eliminate is an absolute term and if taken literally requires that the soil around the burial pits and the backfill or cover have zero water content. Even desert soils and soils which have been heated to above the boiling point of water and then stored in the open have finite water contents. As written, this paragraph precludes any permanent contact between soil and waste because all soils contain the compound H.O. This restriction surely is not the intention of the writers of this draft regulation. Additionally, the elimination of contact with percolating water after disposal requires a perfect hydraulic barrier between the ground surface (where it rains) and the waste. There is no such thing. Even the best built ship leaks. All soils have finite permeability so it is inevitable that there will be some contact of percolating water with waste if soils are used as the barrier. The intent should be to reduce the amount of percolation to very small levels, not to eliminate it. If Para 61.51(a)(6) remains as written, it Will never be satisfied by the developer of a LLW storage facility and therefore unless this provision is modified or waived there will be no licenses granted. The terr "eliminate" is not necessary for the preservation of public health and safety. "Minimize" would be an appropriate word to substitute for eliminate.

GEN-1

D-50-2

Para b1.51(a)(å) "Any groundwater discharge to the surface within the discosal site must not originate within the hydrogeologic unit used for disposal." <u>Comment</u>. If discussion on this point is correctly understood from the Site Suitability Symposium (Washington, 8-9 December 1931) and NRC Technical Position Paper on "Site Suitability and Site Characterization" undated but distributed 9 December 1931, it is the intent that no surface-vater features sustained by groundwater discharge be present within the site boundaries and further, if groundwater does discharge from the receiving hydrogeologic unit beyond the site, the travel time must be so slow to allow radioactive decay to protect the individual users and general population. It is suggested that this paragraph be rephrased to be more specific.

Acknowledged by card. 1/7/82 mdy

. .

*.*....

PAUL F. HADALA

DON C. BANKS

905 Santa Rosa Drive

Vicksburg, MS 39180

302 Enchanted Drive

Vicksburg, MS 39180

B-287

-4-FTALIC FARTICIPATION IN LICENSING DEDISIONS -A MEGESSARY PARTMER FOR TECHNICAL SITE SYLTABILITY "...For most of the alpha emitting transuranic nuclides, the maximum January 6, 1981 allowable concentrations Werencelculated to be in the range of 10 nanocuries peregrantcementi? Depotetby disposal facilities. These calculationstrerestosservativelyshesed, in the that did not Secretary of the Contission The Cholest Regulatence femaning sion Mashington, 2003. 200305 LEAST NURSE allow credfullanadilucios tongther walter. If this factor were היוז ביין בייין changed, the values would increase somewhat. A decision was made not to recalculate in order to come up with higher values. This Paragraph 61.7(c)(2) states: decision is based on WSINGTE. First, in the spirit of the decision is based on WSSTRIGT. First, in the spirit of the ALRA (as Low as Reasonably Achievable) concept, the lower value of The new rate of the decision of the set of the spirit of the spirit of the The new rate of the decision of the spirit of the spirit of the concept spirit her decision of the spirit of the spirit of the concept spirit her decision of the spirit of the spirit of the concept spirit her decision of the spirit of the spirit of the spirit concept spirit her decision of the spirit of the spirit of the spirit of the concept spirit her decision of the spirit of the spirit of the spirit of the concept spirit of the concept spirit of the concept spirit of the concept spirit of the concept spirit of the concept spirit of the concept spirit of the concept spirit of the concept spirit of the concept spirit of the concept spirit of the spirit of the spirit of the spirit spirit of the concept spirit of the spirit of the spirit of the spirit spirit of the concept spirit of the spirit of the spirit of the spirit spir ALARA (as Low as Reasonably Achievable) concept, the lower value of RE: Liderstrigodieallygetherauthdrit/Lindrobiuctoshi abive surface opera-Radiour aud weintersynatermiseis Bulisptit Aufspese reneval, at Rephicie zig6384 praring history will be reviewed and a decision made to permit or deny continued operation ... " lier Secretarie understand the above requirement to mean that the disposal facility operating license must be renewed periodically, at five year intervals. Assonie deutes approariate stiges the roperation of the disposed facility is D-55-3 in Died podeguse should be for the operational lifetime of the facility as is the current licensing practice for operating plants. This would require a Thank logg-term commitment from the facility operator while at the same time prevent a possible periodic disruption of service. Furthermore, periodic license renevals are unnecessary as NRC has authority to perform inspections under Paragraph 61.82 and to take appropriste action in instances of violation unisr (faragraph 61.24 and 61.33 of this rule, including revoking or suspending any license. (Additional comments on public hearings associated wine license, renevals are 270-vided under item 5.) Feorgia Yuan Transuranic Limit 3. Paragraph V.B. states: timstranicialitance istat reading achievablath alt mases and could 72 taufe needless hartship and larpenes. Theztelasomette Aldati-therefore, "... The Commission is applying a 500 area/yr maximum individual 1= "13-justfileasion"forVascablichases at mistarbitraryrligit is a blatant exposure limit for this unusual case. (intrusion) This limit is based on ICRP recommendations for dose limits to individuals and is misuse of this concept. The "Reasonably Achievable" concept within a lavel that is recognized as providing adequate protection. Since ALARA is just as important as the concept of "As low as ... only one, or at most a few, persons would be involved, it is not 7-55-3 necessary to consider a population dose. This limit is then used We also call your attention to the fact that the House Science Committee, during a recent mark-up session on HR 5016, voted to expand the to determine the allowable concentrations of nuclides in each class definition of transuranic wasta from 10 to 100 mCi/g. of weste. (See Paragraph 61.42) ..." As stated above, Paragraph 61.42 provides a 500 mrem/yr accidental Paragraph 61.55 - Table 1 instrusion dose limit for the isotopes listed in Table 1, including At present the burial sites in South Carolina and Nevada will not accept transuranics. Yet this stated criteria (SCO mrem/yr) was apparently ignored in the development of the maximum allowable concentration for any waste that has transuranics above 10 nCi/g. However, alpha-emitting transuranics that are found in nuclear power plant radioactive waste are alpha-emitting transuranic isotopes for Class & weste. Rather, Paragraph 1/12/82 - pro not the isotopes of major concern. Therefore, we recommend that Columns V.C. states: Presented and 2.86 Table de start thursen of Alphersenter a consumning incorpet 3. Start 2.86 Table destriction incorpet have been identified and rationed 7. Start 1.20 Starting out ing incorpet, is also regimmend the rationed 7. Start 1.20 Starting batter is cooper in also regimmend the rationed the Cat hyage Made the start of a start is also be also the start of the start is and the Cat hyage Made the start of the start of the start is also be also be also be also be also be also be also the Cat hyage Made the start of the s

·: .\*

-5-

#### 1977.000077263

D-55 - 3 ) hazard) is the basis for the TZU limit, Cm-242, which has a 163 day (for TRU wastes. Fointies tubic opinion. Even though the disposal of low-level Paragraph 61.53(d) Additivitie (waste has been practiced for nearly 20 years, it

The paragraph indicates that radioactive wastes with concentrations that excenditheregiues inton inecontais, ale hoi gealraid interiote for D-35-5 D-35-5 near-surface disposal and shall not be disposed of without specific .... disposed Compassions approved parmine Toenbackioncel.332: Shoc Yequirecent would clearly create problems for spent resin shipments from our nuclear

Tents anticup facilizies. Arestarve recordence ber she linthante apectivotie criteria the Commission intends to use in authorizing disposal of wastes

int frim Congrbich enter thestaite for class Greistesifipettibdeinizaten 3 of Table 1. ;

Tottor Act (fabii De Minime Companyationer on States to be "rescons-

inter for distanterby Gostates redicactive waste generated

borden The Commission regardizes the seed for the figure finites inclusion of wastes, wastes that would be exempt from Part 61 and vissis speid would be considered of no regulatory concernation Commission believes, Sourver, as the recerci Radiation Policy Council has cites ocupled

problem have orea concerned of that such argunting should be determined on a spe-concerned of the such argunting should be determined on a spe-concerned of the second such an exception in a concerning (46 73 16220) established such an exception in a concerning (46 73 16220) established such an exception in a concerning (46 73 16220) for certain fand available d-levels of tribing and carbon-14 contained in liquid scintillation

Levels of tritium and carbon-16 contained in liquic scintillation and animal carpase yastes, Other wastes may also readily lend hesthemsetives for treatment in this finner." The doministion will be working over the next 2 years to define these wates and appyide clear definitional exceptions as appropriate? This, fart of Will not establish a seneric "de minimic reasory for years." and (2) public acceptance of clasposal facilitations. While these

Northeast Utilities supports the "de minimus" concept and encourages the two require/sprospratsolization and the television further al Sit spectrum to us that broader use of the "de minimus" classification would result in conserva-pisgue rad parchaver waste disparative arealunger Saintaining Rockston for the

health and safety of the public. The AIT's National Environmental opening of restudiec states and a Nuspers Tente States To Min 2015 Called trations

Of Redicouclides in Solid Wastes" which should be reviewed by the Commission. (NFC) has amended its licensing regulations for low-level radio-

. . . . . . . . .

.

active waste disposal.(1) Unfortunately, these amendments meither define the characteristics of a good site nor provide mechanisms

for increasing public acceptance of disposal facilities. The

discussion below reviews the problems with the Commission's

amendments and the consequences for radioactive waste disposal.

-6-

#### SITE SUITABILITY REQUIRENENTS

Among its licensing amendments, the NRC has introduced 5. Public Hearings requirements for size suitability", an expression

"reshnica¥ : Paragraph V.G. states: · · · · · · . . . . . . in lying

1 recipe or guide for choosing locations which will "...The life of a typical facility can be broken into 3 phases: sight confine recebberaldeds describbelitelowity, paseridsore distruction, and

institutional control...at intervals specified in the license, (the tire. Sith suitmontal mera for materialy Electron in commently Surveys) the licensee

would be required to submit a license reneval application (S. 61.27). fattaion, ifer ohle Chiantiationthe dispossivatie micetrengtaniant funding arequirements

would be updated and financial arrangements for assurance of adequate suitability so stunding, seviewed, A gabito hearing would be affered is"

Bing, Saterovisiors for the five phases for the five phases that sate up the life

ourate a statut The first public hearing is provided for subsequent me docketing the license application.

expects that site characteristics combined with environmed 2. During operation, public heatings are provided for at each of the S barriers and 2 optiliz hearings would be held at 5, 10, 12 and 20 years after

meating the performance of actives. This approach provides

greater flexibility public hearing is provided for at the time of site closure, i.e.,

ission ifr low-lawstinging is provided for sette time of license transfer, The utilization of both technical requirements for s.

Thus, for a disposal facility with a 25 year operating life, the regula-suitability scole scole for a schir or seath and is main and a size of a con-

1745 Mere dienzaged wienerte Berzitian wighere the konegatos Statte-perticipation, and believes that all pertiment issues should be addressed prior

ment (hereintef tementement offensementeme die bieve beisn Bussenedution plateructionist tactics some parties have utilized at public hearings and the

of the ligendistrimentificance these have had on cost and schedule of a project.

"PeiforManprovisionations.mattines.entryblig hearingsneithreheir permisious in estetiles pongerer aid predistable construction porspin, closure and inth fireniferidt fidenswigrospates autoseenia concernatothet co private and inturiar will be willing to out our thense has to shie Metree of regulatory discossfreerifing-level radioactive waste]. ... Performance objective requirements , however, require more effort and time in development as well as in licensing of specific facilities due to the large number of factors that must be considered to determine compliance. ...

It would be easy for an applicant or licensee to المتحد المتحد والمتحد والمحاد و

B-289

.. •. \*

٠.

-1-

demonstrate compliance with prescriptive requirements...since engineering limits are established which can be readily measured or calculated and the specific requirements for the design and operation of a [low-level radiozotive waste] disposal facility would be clearly defined and readily apparent to an applicant or licensee". (2)

C -

Although the performance objectives achieve their soll of providing flexibility, the technical site suitability requirements are not true prescriptive requirements as indicated in the Draft EIS. In fact the technical site suitability requirements will be just as difficult to apply as the performance objectives since in most cases the requirements rely on compliance with the performance objectives to define suitability. The Draft EIS further stated that the site suitability requirements are merely "common sense". Rather than defining suitable characteristics, the NRC plans to eliminate bad sites with the application of its technical requirements reserving judgement on what is suitable for the lighting process:

"NRC has set out what are believed to be common sense site suitability requirements that can be consistently applied throughout the country. ... The requirements are intended to eliminate,... sertain characteristics that are known to or have potential to lead to long-term problems." (2)

Thus, the NRC has not defined a suitable site but has underscored some characteristics of an unsuitable site. However, the Commission's attempt to define <u>unsuitability</u> is too general to be very useful in identifying sites which carnot be considered for licensing.

D-50-1 A brief review of some of the technical requirements illustrates this point. One requirement is: "The site shall be capable of being characterized, modeled, analyzed and monitored." This requirement loes not significantly narrow the spectrum of sites which the NRC could judge as suitable. Presumably, the NRC is implying that it favors sites which are geologically and hydrologically "simple" so that models can be relied upon to predict and monitor the transport of radionuclides from the site. However, this requirement places no bounds on how simple a site must be nor on how well characterized, reliably modeled, thoroughly analyzed or carefully monitored. Furthermore, this requirement merely restates the need to comply with the performance objectives since compliance with the objectives already requires utilization of predictive transport models which themselves must be derived from accurate characterization, analysis and monitoring of the site.

Another requirement is: "Upstream drainage areas must be minimized to decrease the amount of runoff which could erode or inundate the disposal cells." This requirement, aptly characterized as common sense, relies on the NRC interpretation of "minimize" and does not constitute a truly prescriptive requirement since no limit is set on either the size of the drainage basin or the anticipated erosion rate at the site. Until the NRC reviews a specific site the applicant does not know whether runoff has been sufficiently minimized for the purposes of obtaining a license.

Another requirement states that: "Areas must be avoided D-50-55 where tectonic processes such as faulting, folding, seismic activity, or vulcanism may occur with such frequency and extent

-5-

D-50-1

D-50-3

-6-

D-50-5

C-1

to significantly affect the ability of the disposal site to meet the performance objectives...or may preclude defensible modeling and prediction of long-term impacts". This requirement relies on application of the performance objectives to determine if tectonic processes at the site make the site unsuitable. Reliance on the performance objectives makes the requirement difficult to apply and redundant. The requirement only guides site selection if the NRC can describe tectonic conditions which deem the site unsuitable independent of compliance with the performance objectives. Otherwise, the NRC is only providing a glimpse at which site characteristics are important to the licensing decision.

Finally, the NRC requires that: "The disposal site must be generally well drained and free of areas of flooding or frequent ponding. Waste disposal shall not take place in a 100-year flood plain, coastal high-hazard area or wetland." The latter part of this requirement is prescriptive in that it states clearly a condition which is not suitable for waste disposal. However, the first part of the requirement again sets no standard for what is acceptable. The frequency of allowable ponding is

The technical requirements, including those discussed briefly above, in practice warn applicants to be wary of certain site characteristics which the Commission has highlighted but, do not provide a clear definition of suitable site characteristics. This need to maintain technical requirements which are general in nature results directly from the "systems approach" which

not specified and well drained is not defined.

emphasizes the interactions and strengths of all the compoments in a disposal system (site, engineered barriers, waste form etc.) to contribute to the overall goal of waste disposalconfining radionuclides. (3) Site suitability defined in this context allows each site to be judged in light of the entire disposal system being proposed rather than representing an ideal site. The necessity of this approach is supported most clearly by considering its alternative. If the NRC issued technical requirements, which rigidly defined an ideal site it would severely limit the possible number of acceptable sites in the country. Future sites might be concentrated by necessity in arid regions of the western U.S. Since most low-level radioactive waste is generated in the eastern half of the country. this siting pattern would result in increased transportation distances, costs and risks from accidents. In addition, as recognized in the Draft EIS. prescriptive requirements focus on components of the disposal system deemphasizing the importance of the system as a whole.

-7-

However, although more flexibility in locating sites for radioactive waste disposal is attained by emphasizing the entire disposal system and identifying only general technical requirements, other problems are created. Since site suitability for radioactive waste disposal cannot be more specifically stated using this approach, controversy over its meaning will likely surface during the licensing process. Understanding the political context in which the licensing decision will be made sheds light on this problem.

- - - -

C-1



F-1

#### AMERICAN NUCLEAR SOCIETY 555 North Kensington Avenue, La Grange Park, Illinois 60525 USA Telephere (312) 332-4413 EHE POLITICAL JOUTERT Telev 334433 EHE POLITICAL JOUTERT TELEV 32 JUL 15 FJ 31

Radioactive waste disposal is a politically charged social issue. Low-level radicactive waste discreal like all nuclear issues is a problem with national and local tensions. At the national level. it is inextricably linked to the debate over nuclear power. This debate enters low-level radioactive waste decisions because 50% of all how religion of the state of the OF THE power reactors. A receptively ministry 333) in the State of Wash-CN PROPOSED ALLS ington noted this fact and attempted to ban the disposal of FOR low-level radiossistereners dates itestors while allowing OF RADICACTIVE WASTE the disposal of non-reactor waste. At the local level, the public 46 Fed. Reg. 38089 (July 14, 1981) is concerned about ban het Haibaor Histe because the major impacts from disposal activities will be born by citizens living near disposal sites. Increased traffic, potential leakage of radionuclides, the dedication of land in perpetuity, these are all consequences a local population bears. The strength of the sentiments felt by State government in dealing with these problems was seen in October, 19 Attorney for ne possible closure of all three disposal facilities ways for in fundation for the second secon of Washington, Nevada and South Garphane Gash Hamaless packaging and transportation practices as threatening the safety of waste discosal in their States. Temporary closure of two of these disposal operations created concern that nuclear activities might have to cease until disposal capacity could be made available again. Limited storage capacity at hospitals and universities received widespread media attention. South Carolina has since halved the amount of waste it will allow buried in the State each

year and the State of Nev<u>acchostonegene</u>t in court for the right to close down the site at Beatty, Nevada. Finally, the passage of the Low-Level Radioactive Waste Policy Act last year, focused national attention on State responsibility and rights in providing The enclosed comments were prepared and reviewed by waste distribution Successformers of the Fuel Cycle and Waste Management Division.

-9-

#### THE PROBLEM

The NRC licensing process addresses this complex social issue by emphasizing technical requirements and providing highly formal and limited opportunities for public participation. The licensing process does not take into account heightened public awareness of the political and health questions related to radioactive waste disposal. Thus, the major problem with relying on the licensing process to apply loosely defined technical requirements in the regulations is the inevitable controversy that will result when the public challenges the acceptability of radioactive waste disposal and asserts its rights to be part of the licensing process. Interpretations of the general language used in the requirements are easily challenged, hence the determinations of site suitability based on those requirements are likewise easily challenged. Controversy will shape the debate over site suitability.

The licensing proceeding for the Diablo Canyon nuclear power plant in California serves as an example of this problem. The major technical issue is whether the plant's location 2.5 miles from a fault could cause failure of essential safety mechanisms in the reactor during a major earthquake. Experts in active faulting and seismology have argued on both sides while the Pacific Gas

B-324

F-1

B-292

and Electric Company, owner of the plant, has weathered 8 years of construction, 36 federal hearings, 4 state hearings, 3 licensing appeals and 2 Congressional inquiries.(4) Resolving the seismic issue is complicated by the incomplete development of the science and the discovery of the problem after more than 75% of the reactor had been built.(4) The late recognition of the problem and reliance on a licensing process which utilizes formal hearings to resolve controversial issues only contributed difficulties to an already intractable problem. The uncertainties in the technical data fuel the controversy and the NRC has yet to devise a way of increasing public trust in-its decisions or judgement.

F-1

F-

#### PUBLIC ACCEPTANCE

. . . . . . . .

Lessening public opposition to agency activities is often the primary motivation for including the public in decision making. The NRC recognized the need to provide public participation opportunities in its licensing process and attempted to channel public concern into aiding the licensing decision. The licensing process has been amended to include several opportunities for formal hearings and a mechanism for States and Tribes to participate in the technical review of the site proposed for waste disposal. However, these mechanisms do not address public acceptance of waste disposal and are not adequate for conflict resolution or identification at any stage of the licensing proceeding. A brief summary of the NRC provisions for State and Tribal participation provides some insight to the weaknesses of these mechanisms.

Opportunities for State and Tribal participation begin after

Submission of an application for a license. Following formal notification in the <u>Federal Register</u>, a State or Tribe has 120 days to submit a proposal for participation in the review of the license. The proposal must include:

- \* a description of how the State or Tribe wants to participate;
- \* a description of the material the State or Tribe wishes to submit to the NRC for inclusion in the review;
- \* a description of the work the State or Tribe Wishes to perform for the NRC; and
- \* a preliminary estimate of the types and extent of the impacts the State or Tribe anticipates as a result of the waste disposal activities. (1)

This provision invites States or Tribes to assist the NRC in its licensing decision but does not give them any decision making power, nor does it guarentee them any influence over the HRC's decision. It is not clear what standard the NRC will use for granting these opportunities and it is not clear how the NRC will weigh the information gathered in this manner in the licensing decision. In addition to these disadvantages, this mechanism for participation appears particularly weak in contrast to the opportunity to participate in the Agreement State program. In this program; a qualified State can regulate radioactive waste disposal itself and discretion over the issuance of licenses is transferred from the NRC to the appropriate State agency. The possible advantages of merely assisting in an NRC decision do not clearly outweigh the option of attaining Agreement State status. Although the NRC amendment allows participatory rights to Tribes which currently can not attain Agreement State status, the type of participation proposed by the

.. .

12

-10-

#### -12-

Commission remains a weak and uncertain mechanism for influencing the decision making process.

F-1

The public participation methods proposed by the NRC reinforce the existing pattern in nuclear facility licensing, relying on formal public hearings to both gather and disseminate information and initiating formal contact with the public only after an application for a license has been made. Public hearings offer only limited one-way communication. The agency presents its proposal and receives formal testimony that tends to summarize positions rather than resolve issues.(5) This highly formal process frustrates the public which may be hearing about the proposal for the first time and does not provide the agency with the type of information it can use in making a decision. The public's comments often seen unsophisticated and ill-informed, a direct consequence of their limited opportunities to learn about and take part in planning up to the hearing stage.

A recent Congressional investigation of Federal regulations pointed out two major hinderances to effective public participation in the existing regulatory process related to the timing of public participation long after the beginning of informal discussions between the applicant and the NRC:

" By the time notice is posted in the Federal Register, the staff of the Nuclear Regulatory Commission typically will have worked with the applicant...for a year or two on the technical details of the application. Not only does this process give the agency staff vested interest in the application as it stands, but the public is usually shut out of the early, and often determinative, stages of the process. ... By the time the public can get involved in a decision, so much money has usually been spent by the [applicant]..in planning and studying the site that it becomes uneconomical to change the course of action."(6) The WRS ameniments continue to provide too few opportunities too late for meaningful public participation. The meet for public participation and the lack of public acceptance of agency decisions stem in part from lack of communication between the agency and the public:

-13-

#### "Concern for participation arises almost entirely in the context of real or imagined failure of government to respond appropriately to the more competitive needs and demands of citizens, some of whom feel that the response would have been more satisfactory had their values been given an assured fair hearing." (7)

If the NRC is to encourage public participation, it must develop with the public a relationship akin to the one it has with the industry it regulates. The NRC should begin informing the public about radioactive waste disposal plans when it begins discussing those plans informally with the applicant.

The Commission should require the applicant to identify local concerns and to inform the public and its representatives in government about the type of facility which is being proposed. The NRC recognized the potential for early public participation in its regulations when it considered an option requiring a notice of intent to file an application 3 to 6 months prior to the actual filing. This requirement encourages early participation and could result in early identification and factoring in of public concerns in the applicant's proposal. The NRC rejected this option for the following reasons:

"...(1)it added an administrative burden on the applicant; (2)from a practical standpoint, it is probably not meeded to assure early state input; and (3)its purpose can be accomplished by other means." (2)

These reasons for rejection must be balanced by the tenefits of

- 4

of the requirement. Although the applicant may already have reasons for seeking State input, there is little incentive to

seek local or citizen input. Droumentation of how the applicant sought public participation prior to submitting an application and reasons for incorporating or rejecting these concerns are important to building a credible record with the public and in gaining acceptance of the facility. The NRC should reconsider this option in a form that would go beyond public hearings and formal testimony to one which would incorporate an interactive approach for gathering and disseminating information about the proposal.

F-

B-295

F-1

Beyond the formal requirements and review of specific aspects of a license there are other considerations in increasing public participation in agency decisions. How the public contributes, who should represent the public and how the NRC plans to use the information it receives should be primary considerations in the development of a relationship with the public. Of course, it cannot be expected that increased public participation will necessarily result in easier decision making. More likely, the NRC will find itself responding to increased pressures to decide issues in a greater variety of ways while being pulled in conflicting directions by new constituencies. However, Congress recognized that relying on a more varied constituency and a larger information base can yield positive results for the agency:

"Specifically the presentation of alternative viewpoints which is a consequence of broader participation checks possible imbalance in several ways. First, agency decision makers are provided with a greater range of alternatives and information. Second, participation promotes agency autonomy by widening the official perspective of agencies and providing an alternative basis of support." (6) Finally, it should be emphasized that greater public participation although commonly viewed as a cause of delay in the licensing process, must be also viewed as a necessary means to seeking acceptance of waste disposal. In some cases public participation may actually speed up controversial proceedings by avoiding the lengthy alternatives of litigation and appeals:

-15-

"[In the view of the Senate Committee on Governmental Affairs], the fact that an additional party participates in an...agency proceeding does not mean that the proceeding will be delayed. ...Ultimately...the over all time elapsed may in fact be lessened, since if all relevant issues are resolved in the initial proceeding, the likelihood of a subsequent court reversal to consider relevant issues is substantially reduced and along with it the risk that the agency will simply have to go through its paces all over again." (6)

With these caveats in mind, suggestions for constructing a successful public participation program can be made. The NRC should identify its primary goals in seeking information from the public, for example:

- \* identification of public opposition or support and its causes;
- \* identification of local or State preferences for locations within the State or region for radioactive waste disposal;
- \* identification of preferences for State or Federal ownership after disposal operations have ceased; and
- \* identification of the need to compensate the local population for increased risks resulting from waste disposal.

In addition, the NRC should initiate educational opportunities

for local, State or Tribal governments so that they are better

equipped to understand information in the licensing process.

-

.

-1ć-

This information exchange will improve bitizen participation by providing a clearer concept of the disposal facility and its potential problems.

E-1

Who represents the public has always been a problem in designing public participation programs. The NRC has restricted its public participation to State governments and Trical representatives These representatives of the public contribute a limited parspective on public conterns and must themselves incorporate balancing decisions in order to fairly represent the geographic regions which define their constituencies. While State concerns may be focused more on financial problems, long-term monitoring responsibilities and ownership, the concerns of local citizens may tend to emphasize impacts on the community from construction, the influx of labor, demands on housing and social services and increased health hazards. The NRC should expect to include a variety of representatives of the public so that the perspectives and interests of those both itractly and indirectly affected by radicative waste disposal can be heard in the licensing process.

Finally, perhaps the most significant aspect of seeking public participation is how the NRC uses public opinion in formulating its licensing decisions. The history of public involvement in federal agency decision making is so infested with failure to consider public opinion that researchers have coined the term "cooptation" for the usual outcome of the heavily relied on public hearing :

"A public hearing serves a cooptation function when the goal of the hearing is to let imate citizens and interest groups let off steam and complain about the project. The posture of the decision makers may be one of responsiveness. While it is implicit that public input will have no impact on the program or on policy, people are formally given a thanke to have a say so they may not take the agency to court for failure to provide public involvement. By attending and presenting their case to an unresponsive agency, the opposition has been unwittingly coorted into serving the goals of the agency."(3)

Until the Commission is able to gain public trust in its judgement and until the public accepts radioactive waste disposal, the NRC may find it impossible to license a radioactive waste disposal site without coopting or appearing to coopt the public. The NRC is afterall a regulatory agency mandated to regulate in the public interest. Definition of the public interest requires a broad disinterested view of the public opinion and welfare. The Commission must seek a more interactive relationship with a broad spectrum of the public before it can begin to define the public interest and rely on it in its licensing decisions.

#### CONCLUSIONS

The NRC licensing process ensures that low-level radicactive waste disposal sites can be considered for licensing in almost any area of the country. The technical siting requirements eliminate only extreme consistions which common sense would dictate unsuitable. Yet the Commission has overlooked a key problem in its licensing process - likely public opposition to decisions which are based on technical adequacy with no meaningful opportunities for public expression of acceptance or opposition and no provisions for conflict resolution. The remedy lies in equal attention to the political and social context in which the questions of technical adequacy are answered. Fublic concerns over nontechnical issues as well as public understanding of what site

-17-

-13-

. . .

suitability means must influence the litensing decision if lowlevel radioactive waste disposal is to continue.

- · · ·
  - · ·

#### REFERENCES

- 1. 45 Federal Register 38031-38105 (July 24,1981).
- U.S. Nuclear Regulatory Commission, <u>Draft Environmental Impact</u> <u>Statement on 10CFR Part 51, "Licensing Requirements for Land</u> <u>Disposal of Radioactive Waste"</u>, NUREG-0762, 3Volumes (September, 1961).
- U.S. Department of Energy, <u>Report to the President by the</u> <u>Interagency Review Group on Nuclear Masse Management</u>, TID-29442, UC-70 (Marcn, 1979).
- 4. R.J. Smith, "A Seismological Shoot-out at Diablo Canyon," <u>Science</u> 214, 528-529 (October, 1981).
- A.B. Bishop, M. McKee, R.D. Hansen, <u>Public Participation in</u> <u>Public Policy Information: A State-of-the-Art Report</u>, Y/OWI/SUB-77/22335 (1977).
- U.S. Senate, Committee on Governmental Affairs, <u>Study on</u> <u>Federal Regulations prepared pursuant to Senate Resolution 71</u>, 6 volumes (September, 1979).
- J.A. Riedel, "Citizen Participation: Myths and Realities", <u>Public Administration Review 32</u>, 211-220 (May/June, 1972).
- T.A. Heberlein, "Some Observations on Alternative Mechanisms for Public Involvement: The Hearing, Public Opinion Poll, the Workshop and the Quasi-Experiment", <u>Natural Resources</u> Journal 16, 197-212 (January, 1976).



- - - -



B-297



### THE UNIVERSITY OF ARIZONA

TUCSON. ARIZONA 83721

COLLEGE OF ENGINEERING DEPARTMENT OF NUCLEAR ENGINEERING

January 7, 1982

IC RESERVER TRACE

1020204

4 . . 4 . .

Secretary of the Commission U.S. Nuclear Regulatory Commission Washington, D.C. 20555

Attention: Docketing and Service Branch

Subject: Proposed Rule 100FR61

Dear Sir:

It is my understanding that the comment period for Proposed Rule 10CFR61 has been extended until January 14, 1932. In view of that I would like to make the following general comment:

In addition to the Proposed Rule 10CF861 there have been generated by other agencies of the federal government a number of documents pertaining to veste disposal. It has cone to my attention and undoubtedly others that there are inconsistencies in definitions of key words in these documents. For example, in Part 61 the definition of "Engineered Barrier" is "a mannade structure or device that is intended to protect an intruder from inadvertent exposure to radiation from certain vastes." In the DOE Site Design Criteria for the Shallow Land Disposal of Solid Low-Level Radioactive Waste the term "Engineered Barrier" is defined as "an addition to the geologic environment which has been designed and enplaced to minimize or praclude radicnuclide transport, or human intrusion, or to preserve the integrity of the disposal site." There are many other examples of inconsistant terms that could be cited.' Therefore. I strongly recommend that an interagency agreement be

GEN-2 { concluded on a glossary of terms pertaining to waste management and that the terms utilized in Proposed Rule 10CTR61 be consistent with this glossary.

Sincerely. Nuclear Fuel Cycle Research Program

JGM:mg

1 200 1 200 1 200 1 200 1/12/82 PMP

#### 07M/11D 20:41 GMT DHEC BRH COLUMBIA S CAROLINA



BQARD William M. Wilson, Chairman J. Lorin Mason, Jr., M.D., Vice-Cheirman Leonard W. Dougias, M.D., Screvery Oren L. Brady, Jr. Moses H., Clarxoon, Jr. Garald A. Kaynard Barbara P. Nussele

(46 FR 38081

1/22/82 emp

COMMISSIONFR Robert S. Jackson, M.D. 2000 Bull Street Columbia, S.C. 29201

January 8, 1982

Mr. R. Dale Smith, Chief Low-Level Waste Licensing Branch Division of Waste Management U.S. Nuclear Regulatory Commission Washington, D.C. 20555

Dear Dale:

We have reviewed the proposed rule, 10 CFR Part 61, Licensing Requirements for Land Disposal of Radioactive Waste and herein offer our comments and recommendations. We commend the Low-Level Wasta Licensing Branch for their efforts in developing these regulations. We recognize the tremendous undertaking on your part.

As the regulatory authority in the State of South Carolina with the responsibility to adopt the regulation when issued, we support promulgation of this needec rule, and concur with the general concepts. However, we will reserve implementation of certain aspects of the regulation which may conflict with existing regulatory requirements and criteria now in place at the Barnwell, South Carolina facility. We understand the Commission's intent that all future disposal activities at existing sites comply with Part 61; however, this will be discussed with the Commission's staff in the future.

Since much of the outlined regulations are general in nature, we strongly recommend the development and issuance of Regulatory Guides which will identify and explain specific requirements for licensing. In addition,

GEN-2 dentify and explain specific requirements for licensing. In addition, we support and recommend development and issuance of Regulatory Guides which incorporate standards for waste form packaging, high integrity containers, solidification media, leachability standards, test methods, and approval guidelines and processes. We offer our assistance in developing these needed guidelines.

Our specific comments and recommendations are outlined as follows:

1. Paragraph 61.12 Specific Technical Information.

a. We recommend addition of a requirement for contingency plans for storage of waste in the event of an unplanned occurrence or natural phenomena which may interrupt site operations and timely disposal of the waste when received.

B-298

07M/11D 20:39 GMT DHEC BRH COLUMBIA S CAROLINA Mr. K. Dale Smith, Chief Page Two January 8, 1982 b. Provisions should be required for the inspection of waste forms to insure that license requirements are being met. These provi-sions should require periodic inspection of waste packages, administrative procedures, or waste inspection and preparation procedures submitted by the waste generators utilizing the burial facility. 2. Paragraph 61.13 Technical Analysis . An analysis of the impact on the immediate area at and adjacent to the burial facility from the transportation of waste should be con-sidered. Satellite activities such as trucking terminals could have significant radiological impacts. D-52-4 *G*. Paragraph 61.41 Protection of the General Population from Releases of Radioactivity. 6-3 It is not clear in this section if the EPA National Primary Drinking Water Standards (40 CFR Part 141) apply for a private drinking water supply such as a private well, or whether the annual whole body dose of 25mRem applies in such cases. This should be clarified. Paragraph 61.50 Disposal Site Suitability Requirements for Land 4. Disposal. D-50-25 a. Criterion 4 regarding economically significant resources should also emphasize the necessity for considering major aquifers and ground water recharge zones as economically significant resources. · . . b. Criterion 5 eddresses the possibility that a site located within the hundred year flood plain could be submerged within the time e 11 the hundred year flood plain could be submerged within the time that the waste still required isolation. By ruling out sites within the hundred year flood plain, this possibility is reduced. However, there are sites within river basins which would be above the hundred year flood plain but within the area that would be flooded if the valley were impounded for irrigation, flood control, or hydroelectric power. Since suitable sites can and usually have been identified, it should not be difficult to consider this nearthal for the suitable sites can and usually have D-50-3 possibility. C. Criterion 11 requiring that no nearby facility interfere with the environmental monitoring program should not be construed so strictly that location of burial facilities adjacent to existing 1-50-4 Federal facilities is precluded. (d. A new criterion addressing the mechanical and physical properties of the site soil is needed. This should require that the soil be: (1) Strong enough to support heavy equipment (2) suitable for com-paction to form trench caps (or provisions for borrow dirt or engineered caps) (3) amenable to control of surface/subsurface runoff (4) amenable to remedial measures in the event of migration, and (5) capable of characterization.

· · · ·

07M/11D 20:38 GMT DHEC BRH COLUMBIA 5 CAROLINA

# Page Three an e sa integra den e an e se Ce. A proposed site must be amenable to modeling and sufficiently simple so that reasonable numbers of monitoring wells adequately determine flow patterns; that is, small areas of significantly increased permeability should not be present, nor should other small-scale discontinuties which require precise positioning of D-50-1 monitoring wells.

Сь.

f. The preoperational ground water characterization should be comprehensive enough to determine the behavior of each zone underlying the site. For example, water levels in each saturated zone should be recorded for some minimum period long enough to determine the seasonal variations. Mass belance (e.g., perculation versus evapotranspiration) should be determined in the unsaturated zone and subsurface flow patterns and rates determined. It is particulated to determine the long enough to determine the source of the long enough to determine the modifier this long enough to determine the seasonal variations.

changes such as changes in land use would cause.

5. Paragraph 61.52 Land Disposal Facility Operations and Disposal Site Closure.

/i. Segregation of waste based on waste classification is considered a Segregation of wasta based on wasta classification is considered a viable option to further assure that performance objectives are met in the long term, but it has certain short term drawbacks. Operationally, wasta packages with low radiation levels have been used as shielding for higher radiation levels have been objected in trenches, thus minimizing accurational amount trenches and additional handling equipment. Radiation levels at the trench boundary will be elevated and significantly increase operational experiment. occupational exposure.

Subparagraph (a)(5) states that "void spaces must be filled with earth or other material to reduce future subsidence within the fill." The term "other material" should be specified.

Subparagraph (a)(6) states that radiation at the surface of the trench cover be within a few percent above natural background levels at the site. A "few percent" should be qualified.

d. Subparagraph (a)(9). In addition to adequate closure and stabilization measures being carried out, erosion control measures should be added.

6. Paragraph 61.55 Waste Characteristics.

Subparagraph (a)(3) should be qualified to indicate that the addition of absorbent material applies only to those institutionally generated, aqueous and biological waste forms. It should be also stipulated that organic, toxic, and reactive wasta forms are prohibited from land burial unless mitigating measures are taken to stabilize the waste and make it environmentally acceptable. We will continue to reserve the right to ban certain wate forms that, in our opinion, are not environmentally acceptable at the Barnwell, South Carolina facility, irregardless of federal regulations allowing such.

B-299

## 07M/11D 20:37 GMT DHEC BRH COLUMBIA S CAROLINA

::

Page Four

b.

Subparagraph (b)(2) establishes a free-standing, non-corrosive liquid contant of 1.02 of the volume of waste. We have of course established a more restrictive criterion of 0.55 in regular waste containers and a limit of 12 in high integrity containers. The industry has accepted and can meet these requirements which are soundly established. We feel that the more restrictive criterion should be codified in Part 61.

We appreciate this opportunity to comment on the proposed rule. We will be most receptive to meeting with you and your staff if you feel further discussion is warranted. If we can be of any assistance, please do not hesitate to contact us.

Yery truly yours.

Heyward G. Straly, Chief Bureau of Ratiological Health

#### HGS:kn

cc: Hr. G. Wayne Kerr

BIRMINGHAM AUDUBON SOCIETY

POST OFFICE BOX 314 . BIRMINGHAM, ALABAMA 33201

:. ·

"82 J"12 P3:26

January 4, 1982

Secretary of the Commission U. S. Nuclear Regulatory Commission Washington, D. C. 20533 Attn: Docketing and Services Branch

- **:** `

Gentlemen/Ladies:

OFFICERS ward M. Einspain Provident

David A. Larson Vice President

Ann C. Tate Vice President, Construction

> Mary A. Williams Reneway Services

Mades & Scharter

Arthur Tyson Treasurer

Guine Arnest

Donald B. Sweeney, Je. Director, Wilalife Film Ser

Lois Anne Brever

COMMUTTEES

Ans C. Tate

Can C. Hoiling

Pat Emery

Joint Findley III Public Sciences

Holes H. Kittings Speakers Burna

Jack A. Johnston Don L. Williams Field True

Thomas A. Impol Burd Council

J. Russed Bauloy. Jr.

John F. Harsh William L. McGowan Parcy L. Thispan

Composite Barbare W. Never

DIRECTORS AT LARGE Waiter F. Come Ruth R. Horney Marcarette K. Persons Resert R. Rud. Jr. Harriett H. Wright

PAST PRESIDENTS James V. Pervy, Jr. Eberta G. Red Attached are our comments on your July 24, 1981, proposed rule entitled, "Licensing Requirements for Land Disposal of Radicactive Waste."

While our comments address many specific points within the processed rule, it is fair to say that as a whole the rule has sarious shortcomings. Despite some strong points, the rule needs considerable revision.

Of particular concern is the last item on our comment sheet. Although your summary states that the processed rule "does not deal with disposal by individual licensees by burial of their own wastes," clearly there should be regulations governing such activities. Parraps this already is the case.

Mention of that one specific point is not to downplay our other concerns, but to assure you that from first to last our comments are offered with serious appreciation of how important this processed rule is.

We are grataful for the occortunity to comment. Also, we ask to be placed on your list(s) for rulemaking notices and notices of other Commission activities of public interest.

Sincerely John Northrep

Copies: John Northrop Sen. Jeremian Denton Memcer, Conservation Committee Sen. Howell Perlin U. S. Rep. Albert Lee Smith Mr. Aubrey Godwin 7.1000 June 1. 1/22/82 0000

For the Conservation and Appreciation of Widlife and Widemess, Natural Resources and Natural Beauty

Birmingham Audubon Comments--page 2

. . . .

:2

......

after notice to the Governor and congressional delegation of the affected state and a public hearing in the state's largest city. **Birmingham Audubon** Comments on Proposed Rule: 6. 61.41--Radiation limits are too high. Limits should be 10 millirems to the whole body, 10 millirems to the thyroid, 10 "Licensing Requirements for Land Disposal of Radioactive Waste c-3 millirems to any other organ. STRONG APPROVAL 7. 61.42--Limit too high. Instead of 500 millirem per year, ) it should be no more than 10 percent of background per year. 1. 61.3(b) -- The requirement for license before commencing: construction of a land disposal facility. D-52.53 percent. 8. 61.52(a)(6) -- Change "within a few percent" to "within one D-50-6ments are not likely to affect performance objectives. D-52-38 9. 61.52(a)(8) -- Instead of 100 feet, 100 vards. Cear-surface 10. 61.55(d) -- There should be no exemptions in disposal -7 3 prohibitions against the higher level wastes. 3. 61.51(a)(7) -- The requirement that the disposal site be D-52-4/used exclusively for the disposal of radioactive wastes. D-50-9  $\begin{cases} 11. 61.56(b)$ --Stability should be guaranteed at the site for D-50-9 as long as the waste materials remain dangerous, not just for D-56-8 LISO years. 4. 61.7(c)(1) -- The requirement that the State or Federal government own the proposed disposal site before the Commission (issues a license. • . • . . · · · · · ... 12. Subpart B-Licenses lacks adequate specificity. Some suggested laprovements: DESIRABLE CHANGES 6111(b)(2) -- List specific qualifications for personnel. R - 1 61.12(b) -- List specific minimum standards. 1. The proposed rule endorses 100-year limits on institutional 61.12(1) -- List specific minimum measures. safeguards, despite allowing disposal of wastes which will remain 61.13(b) -- Give minimum standards for "adequate." D-35-5 dangerous for up to 500 years. If 100-year safeguard limits are retained, wastes remaining dangerous for longer periods should be سل 13. 61.62(b)(g) -- Licensees should be required to place funds discosed of elsewhere, where safequards can be maintained for as D-59-21 in escrow to cover costs of decontemination, closure and stabilization. long as the wastes remain dangerous. S In addition, Licensees should be required to place in escrow funds ...... necessary to cover costs of institutional safeguards for the duration 2. The proposed rule offers provisions for disposal of wastes of those safeguards. which may be subject to ."criticality." Materials subject to criticality should not be disposed of in sites covered by the 14. 61.71--". . . the Director must make available Commission proposed rule. Separate sites should be required, and a separate staff . ...\* rule covering such sites should be issued. 15. 61.28--A provision should be added allowing the Commission to require alterations to the site or the arrangement of its contents 3. The term "reasonable assurance" is used repeatedly through-D-52-6 out the proposed rule. The phrase "a conclusive showing" should before closure, if necessary to protect public health and welfare. be substituted in the following sections: 61.13(d) 16. There should be criminal penalties of up to 25 years for violating the regulations. 61.23(5) (c) (d) (e) GEN-1 61.30(1)(2) ulations. . 61.40 A-1 61.51(a)(2) 61.54 17. Although this proposed rule "does not deal with disposal by individual • licensees by burial of their own wastes" (Summary, p. 38081), such activities Also, in 61.52(a), the term "assurances" should be replaced by the term should be governed by regulations as stringent as those for other disposal "proof. sites. 1/ 4. 61.7(b)(5)--There should be no exceptions to the prohibition A-7 < than 300 years. D-55.(, 5. 51.5-Any exemptions to these regulations should come only . . . . 

Utility Nuclear Waste Management Group 1111 19th Street, N.W. + Washington, D.C. 20036 + (202) 828-7669

#### NEMBERS utersta Parver Compan Anone Public Senace

Company Antimore Gas & Electric

Company Baston Eduan Company Carolins Power & Light

Company The Cleveland Electric

Huminading Company Commonwealth Educe

Completent Ediant Ca. of flew Torth, Inc.

The Detroit Edison -

Company Duce Power Company Duceurane Light Company Plantas Power Corporation Plantas Power & Light

Company

184

Carsonacon Tentheast Utilities

Northern Status Power Company

Philanesena Decuric Company

Conseny Power Auctionity of the

Cas Company

USING District SHOPPS nurra puthers California Edizos Compeny

Contoracion Hacanana Caccole Power

Company Tange Alemie Sectors Company

Sectoments Prestor

centana forer 2 Light Company

January 14, 1982



emp

Attn: Docksting and Service Branch

U. S. Nuclear Regulatory Commission

Washington, D.C. 20555

Notice of Proposed Rulemaking, 10 CFR Part 61; Re: Licensing Requirements for Land Disposal of Radioactive Waste (46 Fed. Reg. 38,081)

Dear Sir:

Secretary

Company Georgia Power Company Guil States Utilions Company These comments are submitted on behalf of the Utility Nuclear Waste Management Group (UNWMG) in connection with toustan Lighting & Power the above-referenced matter.

Conserv Illinois Forer Company The UNWMG has maintained an interest in the developtime flarme liest # Tonic Company Long Island Lighting Company Los Angeles, Department ment of 10 CFR Part 61 for scae time. In particular, the ENNMMG provided, on December 22, 1978, detailed comments in response to the Commission's October 1978 Advance of Natar & Forest Notice of Proposed Rulemaking (43 Fed. Reg. 49,811): In Middle South Services. addition, members of the Group and its consultants Capital Public From responded to the notice, published last year, announcing the availability of a preliminary draft regulation for Tuntari Notana Front public review and comment (45 Fed. Reg. 13,104).

With respect to the rule itself, the present version -- as a proposed rule -- represents a significant improvement over earlier drafts. Rowever, as stated in Statement on Part 61, the UNWMG seriously questions the Portiand General Electric desirability of many of the specific requirements embodied in the proposed rule. In addition, as detailed State of free York Public Service Electric & in the Attachment to this letter, certain aspects of the proposed rule are of particular concern from a practical standpoint, and will require modification to provide a fully workable basis for the land disposal of radioactive waste.

Received Company In particular, UNWMG is troubled by the fact that the fact the fact that the fact the fact that the fact the fact

A Program Administered by Edison Electric Institute

DWWNG

Secretary of the Commission January 14, 1982 Pace Two

provide a practical basis for classification. To address this difficulty, UNWMG recommends the adoption of a "key  $\mathcal{D}$ -55-6 )isotope" approach to waste classification. Such an approach should offer a reasonable solution to the classification problem and is discussed in detail in the comments.

In addition, the UNWMG is of the view that the 10 nCi/g limitation on alpha-emitting transuranic nuclides not only presents difficulties from the standpoint of detectability, but is otherwise unjustified. Accordingly, the matter of establishing an alternative limit is discussed in the D-55 - 3 (Attachment, and a recommended approach described.

> Finally, in concluding this letter the UNWAG would like to make one additional observation. The current version of Part 61 appears to contain nothing that would prevent the location of a land disposal facility at a reactor site. The location of such a facility at a reactor site could have advantages in certain circumstances. Accordingly, the Commission may wish to acknowledge this fact in promulgating the final rule:

Sincerely, Stan Program Manager

....

Enclosure

ATTACHMENT

••••

-----

D-55-3

Comments of the UNWMG Re Notice of Proposed Rulemaking, 10 CFR Part 61; Licensing Requirements for Land Disposal of Radioactive Waste \_\_\_\_\_\_(46 Fed. Reg. 33,081)

## Comments on "Supplementary Information"

(1) The maximum allowable concentration of alpha-emitting -p. 38,084transuranic nuclides, specified as 10 nanocuries per gram 85. Col. 3 & 1. (10 nCi/g) in the proposed rule, should be increased by at respectively least a factor of ten. The prescribed level of 10 nCi/c is too low to be technically justifiable and, in addition, is below the range of completely practical detection and D-55-3 . . measurement. 715 . S. S. Calculations support a concentration value for the

· · . 3

· ·

trench which is nearly an order of magnitude higher than the proposed limit. See, <u>e.g.</u>, Leddicotte, G., <u>et al.</u>, "Suggested Concentration Limits for Shallow Land Burial of Radionuclides" (prepared under the sponsorship of the Utility Waste Management Group, a predecessor of the UNWMG, and presented at the Symposium on Waste Management, Tucson, Arizona, Mar. 7, 1978). The specification of limits in terms of maximum concentrations causes difficulties. This point has been addressed by others. A maximum-toaverage ratio of 10 can be postulated on the basis of

.

NUREG-0456. Conservatively, a factor of 10 for dilution due to the waste form itself, as well as burial in the trench, can be used to arrive at a more reasonable upper limit on TRU waste concentration of 100 nCi/g.<sup>2</sup>/ Even without considering dilution, however, the

- 2 -

10 nCi/g limitation appears to be excessively low. Earlier works (Adam, J. and Rogers, V., "A Classification System for Radioactive Waste Disposal -- What Waste Goes Where?", NUREG-0456, Jun. 1978; Rogers, V., "A Radioactive Waste Disposal Classification System," NUREG/CR-1005,

Gamma spectrometry techniques can be used to identify radionuclides in a homogenous waste sample. Even the weak gammas can be monitored with appreciable resolution, thus efficiently categorizing the waste components. Most of the waste generally brought to the disposal site will have lower concentrations than the maximum limit for burial. In a survey of five major DOE sites by Healy and Rodgers (Healy, J. and Rodgers, "Limits for the Burial of the Department of Energy Transuranic Wastes," Los Alamos Scientific Laboratory, LA-UR-79-100, Jan. 1979), it is indicated that 97% of the waste disposed of at these sites is only lightly radioactive or is suspected of being radioactive because of the place at which it was generated. The authors cite a 9-month data base regarding the TRU content of trash obtained from the Plutonium Research and Development Facility at Los Alamos Scientific Laboratory. From these data they estimate that, for a limit of 10 nCi/g, a dilution factor of 20-60 could be expected for these wastes. An average dilution factor of 40 for the waste contents can be used without much argument. Combined with dilution in the trench, a total effective dilution factor of more than 100 can be assigned for the TRU waste. Conservatively, using a factor of 10 dilution, the TRU waste disposal limit can be easily raised to 100 mCi/c.

• •

. . . .

8-303

· .

.

Sept. 1979) suggest that, for shallow land burial, a 10 nCi/g limitation could be raised to 100 nCi/cc or 60 nCi/g. The proposed higher limit has been criticized. however, on the grounds that it was derived on the basis of the ICRP-2 lung model; the ICRP-30 lung model is believed to be more realistic and, therefore, ensure adequate conservatism in the dose conversion factors. A comparison of the inhalation dose factors based on ICRP-2 and ICRP-30 models, however, yields some interesting results. In a recent publication by Dunning and Killaugh (Dunning, D. and Killaugh, A., \*A Comparison of Effective Dose Equivalents from Three Major Internal Dose Compilations," Radiation Protection Dosimetry, Vol. 1, No. 1, pp. 3-9, 1981) estimates of effective dose equivalents based on ICRP-2 and ICRP-30, by the inhalation pathway. are cited. For most transuranics the inhalation dose factors are more conservative on the basis of the old ICRP-2 model than the revised ICRP-30 lung model, with the exception of Am-241 for which the ICRP-30 dose factor is a factor of 2 hicker than the corresponding ICRP-2 dose factor. For all other members of this group, the ICRP-2 numbers are consistently higher. Therefore, conclusions derived on the basis of the ICRP-2 lung model are valid for the transuranics, and the higher than 10 nCi/g limit succested for TRU waste is justified.

In sum, UNWMG believes that the proposed limit of 10 nCi/g for TRU waste is unduly conservative and should D-55-3 be raised to at least 100 nCi/g; especially when potenttially significant economic gains are considered. In its discussion of waste characteristics and classification, the Commission states that it "recognizes -p. 38,085, the need for a 'de minimus' classification of wastes, wastes that would be exempt from Part 61 and would be considered of no regulatory concern," and notes that it will be working over the next two years to further define these wastes and to provide for additional "de minimus" D-55-2

exceptions from Part 61 requirements.

- 4 -

UNWMG believes that the establishment of a "de minimus" category of low-level wasts (LLW), whether upon a generic or a case-by-case basis, would be extremely useful and would result in considerable savings of time, money and valuable burial space at disposal sites without any corresponding increase in risk to the public health and safety.

We stand ready to offer the Commission whatever assistance we can to facilitate and expedite the

• • • • •

- 1 -

(2)

col. 2

Arguments to the effect that an increase in the limit would be negated by current limitations in routine measurement techniques are not correct. Few utility wastes will exceed 10 nCi/g of TRU, but proving this generically may be quite difficult. On the other hand, it can likely be shown that no utility wastes exceed 100 nCi/g. Thus, increasing the limit by a factor of even 10 will considerably expedite the qualification of utility waste streams on a generic waste source basis.



tion, we would like to call attention two reports which should aid the Commission's effort. The first is entitled 'Suggested Concentration Limits for Shallow Land Burial of Radionuclides" by G. W. Leddicotte, et al., and is cited above in connection with the discussion of limiting TRU waste concentration. The other report is AIF/NESP-016, "de minimus Concentrations of Radionuclides in Solid Wastes," which was prepared for the National Invironmental Studies Project of the Atomic Industrial

. . . . .

- 5 -

an applicant who, perhaps through no more than inadvertence, does something at a site which is later considered to be "commencing construction," before a license has been granted. An adequate sanction would be to require that the site be restored to its "pre-construction" condition. Accordingly, the last sentence of § 61.3(b) should be eliminated. . . .

- 6 -

for monitoring the licensee's performance and taking

e.g., § 61.24(b)) -- without the necessity of a formal

license renewal every five years.

Comments on substantive requirements

phases. Thus, the Commission will have ample opportunity

appropriate action -- including license revocation (see,

Denial of a license is too strong a sanction against

(2) -p. 38,091, cpl. 1 4 2

D-56-8

A-5

(1)

2. 38,090, col. 3

A-3

necessary characteristic of certain Class & segregated wasts. From the overall context of this section it is clear that the requirement refers to structural, rather than chemical, stability. However, the intent of this section should be specified further to avoid any possible confusion.

Section 61.7(b)(2) states that "stability" is a

Requirements with respect to chemical stability appear elsewhere, e.g. \$ 61.56(a)(4).

(3) .p. 38,092, col. 3 B-2 The statement in § 61.13(a) that, "[f]or near-surface disposal, the groundwater pathway will generally be the most significant in terms of releases of radioactivity" may not be correct and -- in any event -- adds nothing substantive to the rule here. Accordingly, it should be deleted.

- 7 -

(4) -p. 38,093, col. 1

B-

Section 61.13(b) provides that "(a)nalyses of the protection of individuals from inadvertent intrusion must include <u>demonstration</u> that the waste classification and segregation requirements will be met and that adequate barriers to inadvertent intrusion will be provided." (Emphasis added.) What is required for "demonstration," however, should be specified in terms of analysis resulting in reasonable assurance that waste classification, segregation and barrier requirements will be zet.

In this general regard, the addition to Part 61 of a section similar to proposed 5 60.101(a)(2) (46 Fed. Reg. 35,280, 35,288), concerning the purpose and nature of findings, would be helpful. Such a section could explain that, while requirements may be stated in unqualified terms, it is not expected that complete assurance that B-1 they will be met can be presented. Reasonable assurance is the general standard required.

- 9 -

Again, the words "demonstration" and "demonstrated"

-p. 38,094, col. 1 are used in §§ 61.23(f), (i) and (j). The requirements for such demonstration should be specified consistent with the approach discussed in comment number (4), above. (6) The absolute prohibition against changing either the facility or procedures, as contained in § 61.25(a), is too restrictive. The minimum requirement providing that, in all cases, the Commission he provided with at least 60 days prior notice of any changes does not accommodate the day-to-day needs of a land burial operation.

> This section should be modified to allow modification, under certain circumstances, without prior notification. In this connection, the provisions of 10 C.F.R. § 50.59, pertaining to production and utilization facilities, provide a good pattern for an approach.

(7) Here, again, as in § 61.7(b) (2), the intended meaning
 -p. 38,095, col. 3
 D-56-0
 (See substantive comment number (2) above.) In addition, the term "long-term stability" requires additional

(5)

- 9 explanation. If, as it appears, the intended reference D-56-B is to the period of institutional control, this should be stated. Further, since "long-term" is used a number of S times in the rule, it might be desirable to include it ED-1 in the definition section. Section 61.50(a)(6) provides simply that "[u]pstream (8) -p. 38,096, col. l drainage areas must be minimized to decrease the amount of runoff which could erode or innundate waste disposal units." As stated, the requirement is overly vague and D-50-3 should be rewritten to limit "upstream drainage areas" of concern to those which contribute to ceneral surface water runoff over the site. (9) Sections 61.51(a)(4) and (6) require that covers be "designed to prevent water infiltration" and that the -p. 38,096 col. 2 disposal site be "designed to eliminate the contact of water with waste during storage, the contact of standing water with waste during disposal, and the contact of percolating or standing water with wastes after disposal." GEN-(Emphasis added.) As now worded, these requirements would be difficult -- or, perhaps, impossible -- to meet. While contact of LLW with water can (and should) be minimized, there is currently no existing technology by which it can be proven that all contact with water has been completely eliminated. Accordingly, we suggest that this provision be modified to require designs such

. :

GEN- | S that contact with water is minimized. Section 61.51(a)(7) provides that "[t]he disposal site (10) -p. 38,096, col. 2 shall be used exclusively for the disposal of radioactive wastes." We assume that the restriction which this section imposes concerns the type of waste (i.e., radioactive D-50-4 waste and not hazardous chemical waste) which may be disposed of at the site. However, this provision can also be read to prohibit any activity other than waste disposal at the site. We suggest that the section be clarified so as to specifically provide that the disposal site, as defined, shall be used exclusively for the disposal of radioactive wastes: and not for non-radioactive chemical or other waste material.

- 10 -

(11) -p. 38,096, col. 2D-52-1

certain types of waste be placed in units "sufficiently separated from other units so that there is no interaction between them," should be clarified. Specifically, UNWMG suggests that the prohibition on interaction be limited to that which could reasonably be expected to lead to premature near-surface land disposal system failure.

The meaning of \$-61.52(a)(1); which requires that

(12) Section 61.52(a) (4) currently provides that, "Wastes -p. 38,096, col. 3 must be emplaced in an orderly manner that maintains the package integrity during emplacement and disposal." The D-52-2 word "orderly" should be deleted from this provision.

It is not essential to the meaning of the sentence and the

intent of the provision is equally clear without it. Moreover, if the word "orderly" is not deleted, greater physical contact with the packaged waste by employees might be required to arrange the containers in an "orderly" manner. Increased handling of the containers would, of course, increase radiation exposure, resulting in a risk which -- although small -- is not necessary in order to maintain package integrity:

- 11 -

(13) -p. 38,096, col. 3

わー5えース

As it is currently worded, § 61.52(a)(5), which provides that "[v]oid spaces between waste packages must be filled with earth or other material to reduce future subsidence within the fill," is unnecessarily proscriptive as to the specific method to be used to minimize subsidence. It can probably be expected that reasonable efforts will be made to reduce void spaces. Ecwever, this provision should be reworded to make clear that its primary intent is to reduce future subsidence and, thus, permit greater flexibility in the procedures which may be used to achieve this goal.

(14) -7. 38,097, col. 263 ED-1 to achieve this goal. Section 61.53 establishes three classes of radioactive waste: Class A segregated waste; Class 3 stable waste; and Class C intruder waste. To avoid possible confusion with Department of Transportation nomenclature, we suggest that these categories be designated Class X, Y and 2, rather than A, B, and C, respectively. In addition, the nuclide concentration limits contained in Table 1 do not provide a completely practical basis for classification. Difficulties with respect to limitations on TRU waste are discussed in "supplementary information" comment (1), above. Further, in many other cases, the required measurements would be difficult and, in some instances, direct quantification may be impossible. (See, <u>e.q.</u>, "Draft Environmental Impact Statement on 10 CFR Part 61 Licensing Requirements for Land Disposal of Radioactive Waste', NUREG-0782, Main Report, pp. 7-23 to -24 Sept. 1981 [hereinafter referred to as Part 61 DEIS].)

The NRC has noted that

One solution could be to routinely measure only those radionuclides that can be reasonably and accurately measured without terribly expensive and sophisticated techniques. Concentrations of other radionuclides would be scaled to the measured radionuclides based upon existing or generator-specific data. Additional measurements would be performed to determine concentrations of other radionuclides if the measured radionuclide concentrations exceed given action levels. A more detailed set of measurements could be performed periodically (e.g., annually or semiannually) or after a significant process change to upgrade the scaling factors and the action levels.

... Two radionuclides which are present in LWR waste streams and can be readily measured by Ge(Li) carma spectroscopy are Co-60 and Cs-137. In

- 12 -

D-55-6

- 13 -

. .

. . . .

the procedure, these two isotopes would be routinely measured and the concentration of other radionuclides estimated based upon scaling factors developed from either data specific to the facility or from a set of reference scaling factors developed from existing data. Samples may be taken for analysis either from (a) the final waste form, or (b) the waste after any and all volume reduction but prior to solidification. If the concentrations of Co-60 or Cs-137 exceed certain action levels, then other radionuclides would be measured. The action levels used may also be either based upon data specific to the facility or from a set of reference action levels based upon existing data. If the concentrations of Co-60 and Cs-137 do not exceed the action levels, then other radionuclides would not need to be analyzed.

D-55-6

Part 61 DEIS, p. 7-23. We strongly support the concept of using key isotopes which can be measured externally and without opening the waste package, but caution that "action levels" which will trigger the necessity of sampling and more extensive analysis be set with a keen eye to practicality.

· · · · · · · · · · · ·

من به

. .

12

. .

The table below is taken in part from a report on the Barnwell site prepared in 1980 for TVA. This table

- 14 -

#### Average Waste Concentrations Buried at Barnwell Compared to Table 1 Col 1

	Isctope	Average Concentration (uCi/g) at Barnwell	Freposed 10 GFR 61 Table 1 Col 1 • (uCi/g)**	Ratio of Actual to Allowable by 10 CFR 61
D-55-4	Cs-137 Cs-134* I-131* Co-60 Co-50* Cr-51* Fe-55* Sr-69 Sr-69 Sr-90 Zr-95* Ce-141* Mn-54* Ru-103* Ru-106*	2.35-01 4.62-02 2.05-01 1.4E-01 4.62-02 1.1E-01 9.22-01 4.62-02 2.3E-03 4.62-02 3.452-02 3.42-02 3.42-02 3.42-02	6.24E-1 4.37E+2 4.37E+2 4.37E+2 4.37E+2 4.37E+2 4.37E+2 4.37E+2 4.37E+2 4.37E+2 4.37E+2 4.37E+2 4.37E+2 4.37E+2 4.37E+2 4.37E+2 4.37E+2 4.37E+2 4.37E+2	3.695-1 1.052-4 4.605-4 3.202-4 1.055-4 2.505-4 2.505-4 2.105-3 1.105-4 1.205-3 1.105-4 1.205-3 1.105-4 1.105-5 7.805-5 1.105-5
	C-14	2.32-04	5.002-1	1.80E-2 4.60E-4

\* All with half lives < 5 years

\*\* Soil density 100-1b/ft3

shows the average concentrations of a number of isotopes which records indicate are buried at Barnwell. These concentrations are compared to the values from Col. 1, Table 1, of proposed 10 CFR Part 51. Also shown are the ratios of these average concentrations

. . .

- 15 -

to the Part 61 concentrations. It can be seen that the only isotope which approaches a Col. 1 limit is Cs-137 (37%). The next nearest isotope is TC-99 (2%), an isotope that is difficult to measure. No other isotope exceeds 0.25% of the Table 1 limit, and most (including Co-60) are less than 0.1%. This strongly suggests that the only key could be Cs-137. UNWMG urges that a "key isotope" approach to classification be developed, and stands ready to offer the Commission whatever assistance it Can.

> Further, one of the footnotes to Table 1 states that, "Wastes containing chelating agents in concentrations greater than 0.1% are not permitted except as specifically approved by the Commission." The basis for the 0.1%

D-56-2 limitation, however, is unclear and may be so low as to present significant difficulties in certain cases; such is those which arise in connection with the disposal of waste resulting from reactor system decontamination projects. This limit should be either justified or increased.

(15) D-54-85 The use of the term "stability" in § 61.36(b) should be clarified by referring specifically to structural stability -00. 38.097-98, (see substantive comment (2), above). In addition, UNNAG col. 3 & 1, respectively is concerned that the 34 deformation limit may be too restrictive; especially when one considers that waste will be containerized to at least some extent, and will also be sup-D-56-9 ported by surrounding waste and/or earth. Further. asphalt, which is generally considered a satisfactory solidification medium, may not meet the 5% limitation. If the 54 restriction would have the effect of precluding the use of asphalt as a waste form, it should be specifically excluded from the requirement.

(16) -p. 38,098, col. 1 £ 2

D-59-2

Section 61.59(b) states that the "active institutional control program" which the landowner is to carry out must include "an environmental monitoring program at the disposal site," among other, things. The rule should provide an indication as to what activities are to be included in such an "environmental monitoring program." In addition, this section should make it clear that, during the period of institutional control, the land could be used for other purposes not inconsistent with the public health and safety and would not disturb the integrity of the site.

۰.,

- 17 -

•

(17) -p. 38,098, col. 3

2

E-1

· · ·

of time for which a license may be obtained under Part 61 (see substantive comment (3), above), we suggest that the phrase "before the next license renewal" be deleted from the final sentence of 5 el.62(d). This sentence would then reads "This will yield a surety that is at least sufficient at all times to cover the costs of closure of the disposal units that are expected to be used."

Consistent with suggested changes as to the length

(13)-p. 38,099,col. 1 4 2F - J

the disposal units that are expected to be used. CNWMG assumes that the provisions of \$ 61.72 pertain to participation by <u>potentially affected</u> States and tribes, and not States and tribes in general. This limitation, however, is not completely clear. Accordingly, the section should be modified to remove any uncertainty.

72 20115 01:00 enp

January 13, 1982

Secretary of the Commission U.S. Nuclear Regulatory Commission 1717 H Street N.M. Washington, D.C. 20555

Attention: Docketing and Service Branch

Gentlemen:

The Federal Register; Volume 46, Number 142; dated July 24, 1981, invites public comment to proposed rules regarding Licensing Requirements for Land Disposal of Radioactive Waste (ICCFR61). As an interested party. Werner & Ffleiderer Corporation hereby submits to the Commission detailed comments regarding the aforementioned rulemaking.

Werner & Pfleiderer is a U.S. corporation which supplies radioactive waste volume reduction and solidification systems to the North American market. Our system's Topical Report has been reviewed and accepted by the USNRC for reference in license applications for commercial nuclear power plants. To date, we have sold six such systems to domestic nuclear utilities and one to the state-cwmed utility of Mexico. Several other similar systems have been sold worldwide by a European affiliate. This business is the focal point of our corporate activities in the radioactive waste processing market and is the reason for our interest in the proposed rulemaking.

Werner 1 Pfleiderer Corporation recognizes the need for regulatory action to establish technical standards for land disposal of radioactive waste and welcomes this opportunity to participate in the regulatory process. We believe that the proposed standards established by IGCFRGI are a rational and viable means of protecting the health and safety of the public. We do, however, have the fallowing specific comments concerning elements proposed in the regulation:

ome.58[20]1 and the second second

U.S. Nuclear Regulatory Commission Page 2 January 13, 1982

#### 1. Paragraph 61.56(b),(1) Waste Form

In an attempt to provide numerical guidance defining waste form stability, the Commission has proposed to characterize a structurally stable waste form as one which "will maintain its physical dimensions within 52 and its form, under the excected discosal conditions of 50 psi .....\*. Werner & Pfleiderer Corporation finds this means of defining stability to be inappropriately restrictive. Specifically, this definition excludes many thermoplastic binders: e.g. asphalt, from conformance due to the fact that a thermoplastic monolith will gradually creep or flow under a 50 psi compressive loading. The implication is that compressive strength per se plays a significant role in soil subsidence. Experience shows that subsidence results from a variety of factors, including predominataly the manner in which wasta packages are placed in the ground and soil backfilled. The Commission recognizes this in Paragraphs 61.52(a), (4) and (5) in which requirements for orderly placement of the waste and backfilling to eliminate voids are established. Werner & Pfleiderer contends that with the precautions astablished to prevent voids in both the ground and in the waste package; i.e. Paragraphs  $\overline{o}1.52(a)$ , (4) and (5) and  $\overline{o}1.5\overline{o}(b)$ , (3), volumetric incompressibility of the monolith is a more appropriate measure of stability than pure compressive strength. As such, the wasta form will behave in much the same manner as an incompressible component of the surrounding soil. Proper waste placement and backfilling produces an environment more akin to a hydrostatic environment in which forces (pressures) are in virtual equilibrium. Based on the foregoing rationale and the fact that thermoplastic binders offer significant performance improvements over more traditional binders; e.g. cement, Wesner & Pfleiderer Corporation requests the Commission to reexamine the rationals supporting the currently proposed numerical guidance used to define waste form stability and acopt a stability definition which recognizes volumetric incompressibility of monolithic waste forms.

### 2. Paragraph 51.56(b), (1) Disposal Container

Paragraph 61.56(b).(1) establishes use of a disposal container as a means to provide stability; i.e. confinement and isolation capability. Although no specific guidance is provided, Werner & Pflaiderer Gorporation assumes that this is a tact: recognition of the solitly. Although this is a departure from previous Commission philosouny, Werner & Pfleiderer Corporation recognizes that a high integrity container to provide isolation capability. Although this is a departure from previous commission philosouny, Werner & Pfleiderer Corporation recognizes that a high integrity container has the potential to provide long-term confinement capability. We do, however, have some serious reservations regarding high integrity containers as they are currently used. For example, high integrity containers are structurally designed to withstand normal conditions of transport which specifies a maximum 4 foot drop test. High integrity containers are potential impact and/or drop accident could exceed the design basis by a considerable margin. This complex with the fact that high integrity used to

U.S. Nuclear Regulatory Commission Page 3 January 13, 1982

D-56-12 We continue to believe that solidification requests the Commission nonconservative design criteria. In the past, the Commission recognized that solidification of waste reduces the potential for dispersion of radioactive materials due to handling accidents. We continue to believe that solidification is an essential safeguard to dispersion of radioactive materials in light of the nonconservative design criteria imposed on high integrity containers. Werner & Pfleiderer Corporation requests the Commission to reexamine the design criteria for high integrity containers for use with highly dispersible waste forms.

We appreciate being given the opportunity to provide comments on the procosed rulemaking and look forward to the Commission's consideration of the foregoing concerns.

Very truly yours,

WERNER & PFLEIDERER CORPORATION

David N. Enegess Business Manager Hazardous Waste Treatment Systems

CNE/WJK/gs

7-56-12

D-56-9

B-312



Daw

#### DOW CHEMICAL U.S.A.

November 25, 1981

LARKUN LABORATORY 1891 N. SWEDE RO. HIDLANG, NOCHIGAN 48840

Mr. Robert E. Browning Deputy Director Division of Waste Management Office of Nuclear Material Safety and Safeguards U. S. Nuclear Regulatory Commission Washington, D.C. 20555

Subject: DRAFT BRANCE TECHNICAL POSITION ON WASYE FORM

#### Dear Mr. Browning:

Dow has reviewed the subject draft Brauch Technical Position (STP) dated October 30, 1981. Dow is in complete agreement that there is a need for the Nuclear Regulatory Commission (NRC) to provide guidance on acceptable forms for stabilized wastes. NRC should expedite the actions necessary to establish and enforce regulatory standards and tecnnical criteria for proper disposal of wastes that are being generated and disposed today. Dow appreciates the opportunity to review and comment on the STP prior to its being issued.

#### Specific commence are as follows:

#### Cover Letter

It is noted that April 1982 is the target date for completing the evaluation of the need and basis for including a minimum acceptable leach rate. Dow suggests that an additional target date of no later than July 1982 be established for including such criteria in the BaP.

#### A. Introduction and B. Backgrouna

These sections contain statements or philosophy, policy, or historical fact that Dow understands. Specific conments are limited to the details in the following sactions.

#### C. Regulatory Position

1.4. Minimum compressive strength of 00 psi appears to be reasonable when the normal anticipated depth of the trench, density of wastes, density or the

AN OPERATING UNIT OF THE DOW CHEMICAL COMPANY

. .

Mr. Robert E. Browning -1-.

November 25, 1981

fill, movement of equipment, and similar factors are considered. ASTM C39 is a tecnnically sound and generally accepted test for determining compressive strength of concrets. However, cement wasta solidification technology normally produces a product with compressive strength of a few hundred psi and occasionally a few thousand psi and Dow solidification technology produces a product with compressive strength in the range of 1,000 to 3,000 psi. Merely stating a minimum strength of 50 psi for either of toase technologies could result in a unusual and tuere.ore possibly unstable waste form.

Dow suggests that compressive strength be specified as  $\pm 102$  with a miniaum of 50 psi when turke or more samples are tested in accordance with ASTM C39.

1.5. Exposure of the samples to a minimum of 10<sup>8</sup> Rays is realistic when compared to the wastes being generated and disposed today. Resistance to degradation caused by radiation effects can only be measured by determining the compressive strength before and after exposure.

> Dow suggests that the 3TP specify that compressive strength of three or more samples is to be determinen before and after irredistion. Variation in compressive strength should not exceed <u>-13X</u> with a minimum of 50 psi.

- 1.c. ASTM G21 and ASTM G22 are tecnnically sound and generally accepted tests for determining growth or fungi and bacteria. By stating "no indication or culture growth should be visible" NRC has assumed that if there is no culture growth there will be no biodegradation. Dow agrees with this assumption and has no further comment.
- 1.d. Dow has no objections to the use of ANS 10.1 for leach testing. As previously stated, Dow suggests that NRC expedite establishing a minimum acceptable leach race and including such criteria in the BTP.
- 1.e. Resistance to degradation by intersion in water can only be dessured by determining the compressive strength before and after intersion.

November 15, 1981

Dow suggests that the BTP speciry that compressive strength of three or more samples is to be determined before and after immersion in leathants specified in ANS 16.1. Variation in compressive strength should not exceed  $\pm 252$  with a minimum of 30 ps1.

1.f. Telephone discussion with Hr. Finothy C. Johnson of your office has confirmed that the reference to ASTM 3335 was in error and should have been ASTM 3533.

- 1-

The number of temperature cycles is not derined in ASTM 8553. Stating a number of cycles to represent a specific waste would obviously depend upon the geographic location of the generating and disposal sites as well as the specific storage and transport times and conditions.

As with exposure to other conditions degradation can be measured only by determining the compressive strength before and after exposure. Dow suggests that MRC select which ASTM 3553 Service Condition is to be used and specify that three or more samples are to be tested before and after exposure to five temperature cycles as defined in ASTM 3553 and that variations in compressive strength should not exceed <u>+</u>23T with a minimum of 50 psi.

1.g. Dow agrees with the method of measuring free liquid as specified in ANS 53.1, Appendix 2.

> "Dow recognizés that the wording in Appendix 2, "There shall be no free liquid visible flowing or dripping from the breach," results in a problem or definition and interpretation, however, the NRU attempt to define "no free liquid" as less than one percent is excessive, unnecessary, and can possibly result in acceptance of unstable waste forms. Cement and Dow solidification technologies have demonstrated the ability to routinely produce solidified products with such less than one percent free liquid.

Dow suggests that MRC include the reference to Appendix 2 in the BTF and specify the limit on free liquid in terms of both percent and a maximum volume per container (one percent not to exceed one gallon per container is not dirricult to consistently achieve with either caneat or Dow technology).

.

. . . .

. .

.

Mr. Robert E. Browning

. .

November 25, 1981

1.h. Dow agrees with the philosophy expressed by this paragraph but suggests that there are methods to obtain samples of the largest anticipated fullscale waste products other than sectioning or coring.

-4-

Dow suggests that this paragraph be expanded to include obtaining representative samples by sectioning or coring after solidification or by obtaining representative samples after the addition of all solidification components but prior to solidification.

- 1.1. Dow agrees with this paragraph but suggests the pravious wording that compressive strength be specified as +102 with a minimum of 50 psi.
- Dow suggests that NRC specify how radiation degradation is to be measured. The test method should specify limits on factors such as gas evolution, compressive strength of the ion exchange beads, and release of the radionuclides.
- 3.s. Dow agrees with the minimum design liretime of 300 years.
- 3.b. Dow agrees that corrosive effects or both the waste contents and the burial ground environment should be considered. NRC should make it clear that the design must demonstrate that thus effects must not shorten the minimum lifetime or 300 years.
- 3.c. Mechanical strength to withstand a load equivalent
- 4 d. to 35 feet of material having 4 density of 120 lbs/ft<sup>3</sup> placed directly on the top surface of the container is roughly equivalent to the minimum 50 psi compressive strength of the solidified waste form. High integrity containers containing unsolidified waste mixed with containers of waste that have been solidified with either cenent or Dow solidification technologies will nave less compressive strength and therefore be less stable that the surrounding wastes.

NRC should specify mechanical strength equivalent to or greater than vastes solidified with either cement or Dow technologies and the method to be used to confirm this design strength.

. .

Mr. Robert E. Browning

November 25, 1981

3.e. Dow agrees with designing mechanical strength based on creep test data. MRU should specify the method to be used to confirm this design strength.

-5-

3.f. Dow sgrees that thermal loads from processing, storage, transporting, and burial should be considered. NRC should define these thermal loads and the method to be used to confirm that the aign integrity container will meet or exceed them.

> Fires have occurred during transportation and in burial grounds. Dow suggests that NRC specify the containers must be tested and neet the criteria of 49 CFR 173.398(a).

- 3.g. Dow agrees that radiation stability and biodegra-
- 6 b. dation properties should be considered. NRC should specify the limits and cest methods for these characteristics as was discussed in paragraphs G.I.b. and G.I.c.
- 3.1. High integrity containers must be capable of meeting the requirements of ab CFR 173.39d(o). The free drop test in 10 CFR 71 Appendix A. is a maximum of four feet while the night integrity container may be dropped a minimum of 10 feet (assuming the truck bed is about four feet above the ground and the shipping cask is six feet call) and possibly a maximum of about of feet to the bottom of the empty outfal trench.

Dow suggests that NRC should specify that high integrity containers must meet the requirements or 49 CFR 173.398(a) including the 30 foot drop test.

3.j. Dow agrees with these conditions. 4 k.

#### D. Implementation

Dow agrees with the philosophy and policy of this section. As previously stated NRC should expedite issuing the 3TP and inmediately start and expedite actions accessary to establish and enforce regulatory standards and technical criteria for proper disposal or wastes that are being generated and disposed cody. Mr. Robert E. Browning -6-

November 25, 1981

. . . .

Once again, Dow appreciates the opportunity to review and comment on the STP. Please contact me is discussion of our comments is desired or Dow can be of further assistance.

Sincerely,

/J. B. Oven Group Leader • Nuclear & Solidification Services 517-636-3388

٤o

MIDDLE SOUTH SERVICES.INC/BOX ELCOO/NEW ORLEANS, LA.70161/(504) 529-6252

January 13, 1982

Secretary of the Consission U. S. Nuclear Regulatory Consiston Mashington, O. C. 20535 ATTN: Docketing and Service Franch

. .

"De Minima" Maste Classification -

. . . . .

SUBJECT: Comments on the Proposed Rule Concerning the "Licensing Paquirements for Land Disposal of Padioactive Nasts"

Cear Sir: These comments are submitted on behalf of Middle South Services, Inc. (MSS) in connection with the above subject matter. MSS is a technical services and support company for the Middle South Utilities System (MSU), which serves the electrical energy requirements of approximately 1,600,000 customers in portions of Arkansas, Lousiana, Mississippi and Missouri. The MSU System has five nuclear units and, therefore, is concerned with the development and implementation of the proposed rules.

MSS has maintained a continuing interest in the development of LOCTRGL over the last few years. The current rule, as proposed, incomparises a number of increases over earlier drafts, and MSS wishes to comment the VRC on its continuing efforts towards D = 52. D-55-2 of a "de minimus" waste classification and the comminity of waste securation according to the proposed Table 1. We also believe that other provisions of the proposed rule can be improved.

> Due to the unavailability of resources, we were not able to provide the detailed comments on the proposed rule that we would have liked to provide. In this repart we have corridicated in comments generated by the Atomic Industrial form and Villity Nuclear Maste Management forum and support the comments that are to be submitted by them. The D-55-4following specific comments are submitted with repart to the proposed reputations.

D-55-2 MSS agrees with the Commission that there is a new for a "de minimus" classification of wastes that would be exempt from the 10 CTR Part 61 repulations. We believe that this is an issue of major importance, and we unve the Commission to establish this classification in as short a time frame as possible. To fate, two types of wastes, Writtown-3 and Carbon-14, have been rerowed from repulation as religactive wastes, under our in conditions, in 10 CTR 20.305. We believe that this repulation establishes a prof nodel for the classification. It would be impractical to shorten the operating life of existing or a newly licensed size with wastes that could be disposed of a simulated solid waste disposal facility without endangering human health or the environment.

SERVING: MICELE SELTH LITLITES, INC. + ARKANSAS SEWER & LIGHT CEMPANY + LELISIANA PEWER & LIGHT CEMPANY + MISSISSIPEI FEWER & LIGHT CEMPANY + NEW ÉPLEANS PUBLIC SERVICE INC. Secretary of the Commission January 13, 1992 7808 - 2-

### License Renewal

B-5

Section 61.27 of the proposed regulations discusses the explication for a litense renewal. In the preamble (45 PR 30097), it is suggested that licenses be renewed every five years. We believe that this requirement cannot be justified. According to Section 61.51 (6), an environmental monitoring program must be maintained during the land discosal facility site construction and operation. Also the Oraft 515, Volume 2, lists a number of extensive reports that must be maintained at the site, these includes: (1) Personnel exposures; (2) Waste receipt and discosal records; (3) Personnel training records; (4) Records from the Quality Assurance program; (5) Environmental conitoring (p. 3-35). Once a licensee has obtained his license has deconstrated compliance with all the requirements in 10 CTR Part 61. The Commission retains authority over the licensee Arring operation and post-operation, and through its inspection, and enforcement branch is able to remitter the activities at the disposal site. Therefore, a five year licensee renewal is unjustly burdenseew. We recomment that full-term licenses to issued.

Maste Treatment

Section 61.51 (a)(7) states that, "(b)he dismosal site shall be used exclusively for the disposal of radioactive wastes." Although it ancears that the exclusion refers to the type of wastes (i.e., hazarfors vs. radioactive), it could also be interpreted as mening that only disposal and not treatment of waste could occur on site. This would prevent the installation of infinerators or other volume refurction equipment at the disposal site. We recorrect that the regulations be changed to accurroiate waste treatment.

Table 1 MSS is concerned that the or

55 is concerned that the processed categorization of vests by redicisorance concentration would place an overly time-consuming and costly burien on LLRT generators. Next wattes are composed of a minute of various isotores at varying concentrations and it would be extremely difficult if not impossible to separate and categorize our wastes according to Table 1. Secretary of the Commission January 13, 1992 Page +3-

#### 561.3(b)

The requirements set forth in Section 61.3(b) are unnecessary. We are not advocating that the construction of these sites be rushed to completion without complying with all of the pertiment requilations. However, we think it should be at the discretion of the prospective licenses whether to commence construction before receipt of the final operating license. We therefore recommend that S61.3(b) be called in its entirety.

#### Stability .

Since the stability of the wastes and waste site are very important in the classification system of these regulations, we feel that the term "stability" should be better defined. In many instances the term "structural stability" is used. Does this team chemical structure, physical structure, or both? In the preamble (46 PR 3904), a defimition for stability is given as follows:

D-56-84

B-4

A-3

"It (the waste) should maintain its configuration and consistency under the conditions it was exposed to after disposal."

This is a very ambiguous definition since there are a number of conditions (physical, phenical, and biological) that the waste may be exposed to after burial.

Also, in Section 61.7(b)(2) where Class A "errestated "Astes (consisting primarily of unstable, trash type wastes) are discussed, it is stated that "if mixed with higher activity waste, their decerioration could lead to failure of the system and mentic water to penetrate the disposal unit..." Once again, does the term "deterioration" mean physical compaction, cherical decay, or a combination of both? We think it would be beloful if these sections were clarified.

#### Additional Comments

1. Section 61.25(a) does not allow any changes in the land discosal facility or procedures described in the license application without prior notification to the Condssion. We believe that changes and rodifications should be allowed with notification to the condsthe Condision in a timely manuer.

GEN-1 Section 61.51(a)(9) states that "(t)he disposal site must be designed to elimithate the contact of water with waste during storage.... disposal, and ... after disposal." Unless the word "eliminate" is deleted and the word "minimize" is substituted, it would be virtually impossible to demonstrate that the design of a disposal site would rest these requirements. Secretary of the Commission January 13, 1992 Page -4-

#### Conclusion

We would once again like to errored the Cormission for the work they have done thus far in striving to provide definitive regulations for low-level radioactive waste disposal facilities. Generally, we support the proposed regulations. We believe that it is very important that the Cormission establish a "de minimus" classification of wastes that are exempt from 10 GTR Part 51 as scop as possible. In surrary, the most significant changes we would like to see made in the proposed regulations include: full-term licenses for disposal facilities; the inclusion of waste treatment; the elimination or improvement of Table 1 which categorizes waste according to radioisotopic concentrations; and the easement of current requirements that do not allow any construction activity prior to the issuance of an operating license.

We wish to thank the Commission for the opportunity to submit these comments and we hope that they will be helpful in you further deliberations concerning the processed regulations.

Sincerely,

Felix M. Killar Manager, Fuel Supply

**R**K1885

co: Gr. D. G. Gibbs Dr. T. V. Schwatz Mr. J. D. Patterson File: 941-01 ÷...



#### Waste Stability Requirement

Paragraph 61.7(b)(2):

As it is not possible to reduce water access to zero the phrase "eliminated or" (line 6) should be deleted. Furthermore, "stability of the waste and the disposal size" needs to be clarified as to whether stability of the disposal site refers to its operational phase or the stabilization for site closure, the latter of which, according to paragraph 61.7(c)(2), would not be required until disposal operatious are about to cease.

1/22/82 amp

B-319

The term "cover" should be clarified as to whether it includes an impervious cap.

. . . . . .
-7-

B-9

E-1

Northeast Utilities believes that the conditions for facility construction, operation, closure and license transfer must be specified and agreed upon at the outset by the licensee. A public hearing will be held at that time, if requested, for public input to the process. Subsequent to license issuance, the NEC should monitor and inspect the activities at the facility to ensure they are in conformance with the license. Unforeseen events can be accommodated by amending the facility license. Further participation by the public is not necessary to ensure the facility is being operated properly. The prospect of repeated public hearings every five years to consider tontinued operation of the facility is totally unnecessary and must be eliminated from the final rule. A more appropriate and effective oversight would be provided by assigning a full time NEC inspector to a disposal site.

## Additional Comments on 100FR61

Title of Proposed Rule: .

ED-1 It is essential for the public to begin to distinguish between high level nuclear vastas which will require disposal in geologic repositories and low-lavel vaste which, under 10 CF3 61, will be permitted to be disposed of in shallow land burial sites. Furthermore, the act F.L.96-373, which gives authority to states to establish regional compacts and for which 10 CF3 61 will be a major guideline is, is thild the "Low Level Radioactive Waste Policy Act" (emphases added). Therefore, we believe that it is entirely appropriate to change the title of the proposed rule to "Licensing Requirements for Lend Disposal of Low-Level Badioactive Wastes."

# (Paragraph 61.59(g): -

D-59-25 The 100 year institutional control period should be extended for as long as the governing body exists. This would extend the surveillance period and protect against site intrusion until the governing body determines the site could be reopened to the public.

## Paragraph 61.62(g):

"Pay as you go" surety requirements for closure should be permitted, as opposed to surety bonding for an entire site.

General Comment on Draft Environmental Impact Statement (NUREG-0782)

The Commission should indicate how the adequacy of quality-scaling factors (used to estimate nuclides not readily identifiable) will be determined.

General Comments on 10 CFR 20 Proposed Changes

GEN-Z Notice should be provided as to when the Regulatory Guide on classification of radioactive waste will be available. In the interim, guidance should be provided regarding classification of wastes as required by 20.311(c).

-8-

Paragraph 20.311(d)(3) and (f)(5):

M-2 The degree of implementation and criteria for the quality assurance programs, required under these paragraphs should be indicated. We would also recommend that the tarm, "quality assurance" be changed to "quality control" so as not to be confused with the quality assurance requirements of 10 CFR 50, Appendix B.

By:

Should you have any questions regarding our comments, please feel free to contact us.

Very truly yours,

## NORTHEAST UTILITIES SERVICE COMPANY

G. Council

Senior Vice President

J. P.

J. P/Cagnetta Vice President Suclear and Environmental Engineering 

# AMERICAN NUCLEAR SOCIETY

FUEL CYCLE AND WASTE MANAGEMENT DIVISION

Peadquarters; 388 Norin Kousiagtan Aunua LaGrange Port, Illinois 66538 USA

January 12, 1982

С.

Samuel J. Chilk, Secretary United States Nuclear Regulatory Commission Washington, DC 20555

Re: Licensing Requirements for Land Disposal of Low-Level Radioactivé Waste 46 Fed. Reg. 38081, 7/24/81

Dear Secretary Chilk:

Enclosed for your consideration in the above referenced matter is the Statement of the American Nuclear Society's Fuel Cycle and Waste Management Division regarding its comments and recommendations therein.

The Commission is to be commended for its further improvements in this area and it is our hope that full consideration will be given to these additional recommendations.

Very truly yours. Junter of Setts

Winston W. Little, PhD Chairman

WWL:ez xc: C. Rickard L. Manning Muntring M. Norman L. Oyan C. Z. Anderson O. J. Dulemple J. Seelan ANS Executive Committee

12 11:15 24 inf

(46 FR 39081)

Nuclear Monitoring Systems & Management Corporation

January 11, 1982

Secretary of the Commission U.S. Nuclear Regulatory Commission 1717 H Street N.W. Washington, D.C. 20555

#### Dear Sir:

A-8

Inclosed is a document prepared by this company which addresses:

1) our concerns over

#### and

and

and and the second by the

2) our recommendations for changes in

the proposed rules and regulations of the N.R.C. (F.R. Title 10, Part 61, dated July 24, 1981).

In this document, we propose a "systems" solution to the problem of "lowlevel" radioactive waste disposal. This "systems" solution includes:

> a landfill in which the vast majority of "low-level" wastes would be disposed of via shallow land burial.

 a monitored, retrievable storage system (such as our proposed S/M/R "Low-Level" Facility) for the small percentage of those "low-level" vastes which would not qualify for shallow land burial under the final rules and regulations of the N.R.C.

Except for the specific references to Texas, the contents apply to every state and, for this reason, we thought you might appreciate the opportunity to examine and evaluate this document from your vantage point.

> Sincerely, NUCLEAR MONITORING SYSTEMS 6 MANAGEMENT, CORPORATION

Louis A. Galloway, III An.D., 7.2. Executive Vice President Corporate Technology

1+00 3/V Preaway + Sure 3/0 + Houston Texas 77007 + 713, P60-0168

B-323



.

#### AMERICAN NUCLEAR SOCIETY 555 North Kensington Avenue, LaGrange Park, Illinois 60525 USA Telephone (312) 352-4611 Teles 234633

.

. . . .

#### ACKNOWLEDGEMENT

. . . .

.

•

The enclosed comments were prepared and reviewed by American Nuclear Society members of the Fuel Cycle and Waste Management Division.

-1-

COMMENTS OF THE FUEL CYCLE AND WASTE MANAGEMENT DIVISION OF THE AMERICAN SUCLEAR SOCIETY ON FROPOSED RULE

FOR

LICENSING REQUIREMENTS FOR LAND DISPOSAL OF RADIOACTIVE WASTE

46 Fed. Reg. 38089 (July 24, 1981) (to be codified at 10 C.F.R. Fart 61)

> JAMES R. HEELAN Attorney for American Suclear Society 535 North Kensington Avenue La Grange Park, Illinois 60523 Telephone: (312) 352-6611

## TABLE OF CONTENTS

- I. 1 The American Nuclear Society ٨. 1 Scope of American Nuclear Society Review . . . . . 1 3. . COMMENTS AND RECOMMENDATIONS 2 II. III. CONCLUSION 10
  - · · · ·
- - - - 11 -

1

Pate

## I. INTRODUCTION

A. The American Nuclear Society

The American Nuclear Society, an international organization of engineers and scientists, now in its 28th year, is a nonprofit scientific, technical, and educational organization. ANS currently has an individual nechership of over 13,000 and is governed by its officers and a Board of Directors elected by the individual membership.

.

...

To carry out its purposes, ANS has 17 separate technical divisions. The objective of each division is to provide means for furthering the science, engineering, and art of that branch of scientific discipline. The disciplines range from those related to nuclear power — such as nuclear fuel cycles, waste management, radiation protection and shielding, reactor safety, and reactor operations to other disciplines, such as controlled nuclear fusion, isotopes and radiation, environmental sciences, and alternative energy technologies and systems.

B. Scope of American Nuclear Society Review

These comments are in response to NRC's proposed rule on "Licensing Requirements for Land Disposal of Radioactive Waste," 46 Fed. Reg. 38081 (July 24, 1981) (to be codified at 10 C.F.3. Part 61, Nuclear Regulatory Commission). ANS has actively followed the progress of this proposed rule with a technical support committee of interested and technically qualified members of the Fuel Cycle and Waste Management Division. Based on a technical review of the Federal Register notice by division members, there have been developed and formulated the enclosed comments on the proposed rule.

-1 -

## B-325

~

11.

#### COMMENTS ON PROPOSED RULEWARING ON LAND DISPOSAL OF LOW-LEVEL RADICACTIVE WASTE 10 CFR 61

Federal Register Vol. 46, No. 142, Friday, July 24, 1981 Proposal Rules

The present version of the rule, 4s proposed, presents a significant improvement over earlier drafts and the Commission should be commended on its work thus far. However, members of the ANS Fuel Cycle and Waste Management Division present the following comments for your consideration.

1) V. Summary of Rule

C. Minimum Technical Requirements

(1) A classification system for low level vasts is proposed containing four classes. Specific activity limits define each class. While the desire to provide a level of protection commensurate with the activity level is recognized, it is not clear why class A and 3 wastes need be distinguised.<sup>2</sup> The basis is apparently that "it is obvious that if these (unstable trashtype wastes) were disposed of with higher activity waste, their deterioration could lead to failure of the (burial) system and permit water to penetrate the disposal site and cause problems with higher activity wastes." If this is demonstrably so, would it not be better to define classes A and B solely on the basis of stability rather than on stability and activity concentration? Thus, all waste up to the class B activity limit would be separated by whether if was solidified.

(2) It is stated that the Commission recognized the need for a "de minimis" classification of vastes, that this "should be determined on a specific vaste basis," and "Part 61 will not establish a generic 'de minimis' category for waste."<sup>3</sup> The "de minimis" levels caed to be established as soon as possible to prevent disposal of large volumes of materials which, in hindsight, would have met such criteria and could have been disposed of by other means. This meed is especially critical in view of the limited capacity of the existing disposal sites. The AIT report "De Minimis Concentration of Madionuclides in Solid Wastes, AIT/MESP-OLS," might be of some use to you.

2) V. Summary of Rule

1. Institutional Control

D-59-2 The discussion states that active institutional controls such as periodic surveillance and controlled access cannot be relied on for more than 100 years.<sup>3</sup> The implication is that governmental institutions would be swars of the necessity to maintain the site. However, in the following paragraph, credit is taken for passive controls. It seems that if a government institution is available to maintain land ownership and records, that same institution could maintain a fance. This would reduce the concern about the potentional exposure to intruders.

46 Fe<sup>4</sup>. Reg. 38083, et seq.
561.55; 38084, et seq.
38085
38085

3) 61.1 Purpose and Scope

The purpose and scope states that these regulations establish requiremente upon which the commission issues licenses "for the disposal for others of radioactive vastes." While 10 CFR Part 20 covers the disposal of vaste by an individual licensee, the quantities are limited to very low levels. The purpose and scope should be rephrased to allow an individual licensee to operate a burial site. The words "for others" and the last sentence in 61.1(a) should be deleted.

ED-|

GEN-

A - 1

a) A "near-surface disposal facility"<sup>3</sup> is defined to be "within the upper 15-20 matars of the earth's surface." Is this definition intended to limit the maximum depth of a trench? If not, the definition should be rephrased to clearly indicate that 15-20 maters is an approximate range.

5) The use of the word "isolation" in the definition of "disposal"<sup>9</sup> implies a degree of absoluteness generally not attainable and its use in this context implies that the disposal facility is not part of the biosphera. The definition should read "Disposal means the placement of radioactive waste in an approved disposal facility."

$$\mathcal{D}$$
-56-8 $\begin{cases} c \end{cases}$  The term "stability" should be defined since it is the tasis to the separation of class A and 3 waster.

<u>51.7(a)(1)</u> This clearly limits the depth of individual disposal units of tranches to 20 meters. No justification for the limit has been offered by NRC. At some locations it may be appropriate and economical for individual tranches to exceed 20 meters in depth. The statement should be revised accordingly.

<u>61.7(b)(1)</u> The safety objectives should be rephrased to read "minimize the migration by any route including surface, groundwater discharge, erosion or wind erosion, and minimize the exposure to inadvertent incruders."

- 3 -

- (c) <u>61.7(b)(2)</u> According to this paragraph, a maximum disposal site inventory will be established for certain isotopes based on the characteristics of the disposal site. However, no criteris on which to base maximum inventories is provided. The criteris for setting maximum inventories should be provided since this will lisit the capacity of the site. The criteris should take into account the site conditions and locations. Also, the criteris should reflect the fact that if a site is properly selected, a single large site may be more desirable than a series of smaller sites.
  - 6. 38089
  - 7. 38089
  - 8. 38090
  - 9. 38090
  - 10. 38090 et saq.

$$S_{1} = \begin{cases} 1 \text{ difference is a subject of a strategy is difference of a strategy is$$

..

•.

24 - 12 - 14

**!** !

B-327

, . . . . .

13) 51.55 Waste classification 20

D-55-6

D-55-1

D-55-

51.55 An alternate mathod of determining the vaste classification should be provided. Provisions for classification by external dose determination should be made. For cases where the types of isotopes of concern are known, this method would allow adequate classification. This alternate method would be particularly helpful for nuclear power plant trash. Generally, trash has a very low specific activity compared to the class A limits. A determination of the radionuclide identity and concentration, as required by Fart 20.311, would require the purchase and use of a portable spectrum analyzer. Instead, a contact doas rate measurement of the containerised trash could be use to show that the activities were below the class A limits. Also, since a radwaste classification systems together?

61.55 - Table 1 The waste classifications scheme presented here with the associated concentration limits presented in Table 1 would have a substantial impact on the nuclear power industry's waste disposal costs and bence, upon the cost of nuclear generated power.

The concentrations given in this table are much more limiting than necessary. In the study prepared for the USNRC by Ford, Bacon & Davis Utah, Inc.. "A Radicactive Wasta Disposal Classification System." SUREG/CRICOS, conservative radioaccivity limits for various waste classes were established through detailed harards analysis. The limits recommended in NUREG/CRICOS should be incorporated into 10 CFR 61 in place of the arbitrary values in Table 1.

The logic behind the numbers selected for this table is not apparent. It would appear, for example, that tarbon 16 which contains less than 0.8 microcuries per ct may be disposed of as segregated vasce but that any concentration greater than 0.3, even if it is only a tiny increase. immediately requires that the disposer seek special permission from the government for disposel. The abrupt demartation needs explaining so that the logic of it can be understood.

(c) De Minimis dassification of vasces

<u>61.55 Table 1</u> should consider a "de minimis" classification of vastes (i.e., vastes that would be considered of non-regulatory concern); we believe strongly that this should be addressed in the proposed 10 CFR 61 regulation. De minimis levels for unantum, technetium, pluconium, and meptimium should be stated. A de minimis or lower acceptable level for matural and depleted unanium should be stated; we recommend that a value atural and depleted unanium depleted unanium be set as a lower limit in Section 61.55, Table 1. Baccant information received from Nuclear Regulatory Commission staff members reveals that proposals concerning de minimis levels for unanium are being prepared by the NRC staff that would establish multi-tired acceptable levels for shallow-land burial of unanium vastas. One level proposed by the NRC would permit disposal of uranium

20. 38097

- 6 -

D-55-2 wastes in an unlicensed burial ground (i.e., sanitary type). A second proposal would permit disposal in a shallow-land licensed burial site, and a third proposal lavel would permit disposal in a shallow-land licensed burial site which has a covenant in the title on the property. These multi-tiered NRC proposals are consistent with our recommended levels. We recommend that de minimis levels consistent with those proposed by NRC relative to <sup>99</sup> Tc and low-enriched uranium as residual contamination in smelted alloys (Addendum to 10 GFR Parts 30, 32, 70, and 150) also be stated. In this regard, we recommend that serious consideration be given to establishment of de minimis levels of 3.5 ppm<sup>233</sup>U, 5 ppm Tc, 0.01 ppb Fu, and 1 ppb Np.

## d) Alpha-emitting transuranic isocopes (Waste Classification)

In Section 61.55, Waste Classification, Table 1, we believe strongly that the maximum concentration for alpha-emitting transuranic isotopes should be increased from the 10 nCi/g limit presently proposed to 100 nCi/g. Fart of the rationale behind the 10 nCi/g limit is stated to be that this value has been imposed by DOE: however, DOE is at the present time-seriously considering revision of DOE Manual Chapter 0511 to raise this limit to 100 nCi/g dated 7/30/81. The 10 nCi/g value is also inconsistant with the value of 100 nCi/g used by the Invironzent Protection Agency in their proposed regulation 40 CTR Part 91 for the disposal of spent fuel, high-level, and transuranic (TRU) wastes. This regulation states that TAU westes containing more than 100 nC1/g of alpha-emitting TAU isotopes must have the same controls as are required for high-level wastes. We recommend the same controls as are required for high-level wastes. We recommend that the 100 aCi/z limit be reflected in each of the columns 1-3 in Table 1. We also recommand that a limit of 100 pCi/cm<sup>2</sup> for transferable surface contamination of alpha-emitting transuranic isotopes (not natural) (or depleted uranium isotopes) be imposed, consistent with the proposed revision to DCE Manual Chapter 0511.

Additionally, the value of 10 nC1/g is based on naturally occurring tadium deposits. Eadium is significantly more hazardous than  $\frac{1}{2}$  or the transuranium muchides when dissolved in water, as the WC's for the soluble forms of these nuclides are about 100 times (1000 times for  $\frac{1}{2}$ ) greater than that of radium. Thus, it would appear reasonable to set activity limits for alpha-anitting transuranic isotopes at 100 times greater.

The supplementary information in the NRC document also states that there is no need to increase this limit from the standpoint of achievability. Much of the waste presently stored as transuranic vaste is segregated from Low-leval waste on the basis of waste origin since the 10 nGi/g limit is too low for accurate measurement and certification. However, segregation according to the 100 nGi/g limit could be achieved, eliminating expensive retrievable storage and deep geologic disposed of "suspect" transuranic waste.

Another concern is the footnote (#4) to Table 1 that refers to isotopes contained in matals, matal alloys, or permanently fixed on metal as containation. The footnote, which states that "the values showe may

- 7 -

D-55-3

waste containing 5 nCi/g would convert a low-level waste into a transuranic waste at 150 nCi/g. However, the residual ashes could be incorporated into concrete, glass, metal, etc. The leach rate of transursaic isocopes from these materials is very low (i.e., many orders of magnitude lower than the untreated waste form). Thus, a combination of leach rate and transuranic content could be used to determine the disposal options for these wasta forms. D-56-02 We are troubled with Table 1, as many of the nuclide concentrations limits may not provide a practical basis for classification. In many - cases, the measurements are difficult and some are almost impossible. Perhaps the Commission would specify practical analytical methods Acceptable for determining nuclide concentrations. e) <u>51.55 Table 1. Footnore s</u>
ED-1 {(1) The term "significant games radiation" should be defined.
D-55-4 {(2) Ecw is radium treated? A value should be established.
(3) The footnote places a restriction on wastes containing chelating agents in concentrations greater than 0.11. Is this limit intended to be 0.12 by weight or volume? This limit is too low, many agents were developed to decontainate prime and equipment to reduce radiation levels to workers. A restriction on the solidified product of 0.12 might cause potential users to forego utilisation of them because of the restriction en disposal and then let radiation levels rise. One manufacturer has even developed a solidification seent for their chelating 51.55 Table 1, Footnote 4 even developed a solidification agent for their chelating STORE. 15) 61.56 Waste characteristics21 D-56-5 (i)  $\frac{51.56(a)(51-(7)}{(1)}$  The incrusistancy between paragraphs 51.56(a)(3) and (i) snould be clarified. Wastes in gaseous form, allowed by para-graph 61.56(a)(7), could be considered the very waste disallowed by paragraph 61.56(a)(3). (j)  $\frac{51.56(b)(1)}{(1)}$  "Stability for 150 years" needs to be modified to indicate what forms of proof are acceptable. Some matal, wooden, and concrete structures can be shown to have maintained their "stability" for 150 years past. Very few of these are applicable to waste packaging. There is no way that deformation alone of the

- 8 - 1

be increased by a factor of ten" should be modified to include concrete and other media that exhibit low leach-rate behavior. An incentive should be provided to reduce the volume of wastes by incineration or

metal smelting. These treatments may normally be avoided by waste generators since they would convert some low-level wastes into transuranic

wastes. For example, a volume reduction of 30 by incineration of a

21. 38097. at seq.

D-56-8 waste form can be a hazard to the public. The key requirement is to keep the waste from being dispersed which is little affected by "slumping" or a "51" deformation.

 $D-56-13 \begin{cases} \frac{61.56(b)(2)}{2} & \text{Non-corrosive liquid" should perhaps be changed to} \\ \frac{61.56(b)(2)}{2} & \text{Non-corrosive liquid" should perhaps be changed to} \\ \frac{1}{2} & \frac{$ 

(d) <u>31.35(5)(3)</u> This paragraph requires that "Told spaces within the vaste and between the vaste and its package must be reduced to the extent practicable." This requirement should be deleted as being too vague unless specific acceptance criteria can be established. For solidified liquid or slurry wastes, 10-15% freeboard in a 35 gallon drum is normally appropriate.

Does this mean that filler material must be added to packages containing irregularly shaped solid objects? Or is it the intere of this article that all such objects should somehow be chopped, melted, or otherwise compacted? What forms are acceptable, i.e., ash, pellets, compressed trash?

- #1

G-1 2 51.32 Eliminate radioactive waste already disposed of and covered, from NRC inspection requirement.

D-50-4 The current version of Part 61 appears to contain nothing that would preclude the location of a land disposal facility at the site of another licensed activity. Such a location could be quite desirable since the facility has been dedicated to a nuclear application.



#### III. CONCLUSION

The recommendations made throughout this commentary need not be be repeated here; however, it is worthy of note that those who prepared this review are multi-disciplined professionals whose known dedication to and expertise in this field counsel an indepth appraisal of these recommendations.

The Chairman and members of the American Nuclear Society's Fuel Cycle and Waste Management Division appreciate the opportunity to comment on the proposed rule, stand ready to assist in any further deliberations, and avait the release of the final regulation.

- 10 -

Respectfully submitted,

JAMES R. HEELAN American Nuclear Society 555 North Kensington Avenue La Grange Park, Ellinois 60525 Telephone: (312) 352-6611

.

Accorney for the American Nuclear Society

Law Ciffices of VALORZ. MCALLISTER. WESTMORELAND & VESPER A ROTHING CONVIN

CALL I VALCAI ICONT N. NOULISTER LC VEITOGAILAND THOMAS I VIIPER MARIE I CARDONI MARIE CARDONI MARIE CARDONI MARIE CARDONI MARIE CARDONI MARIE CARDONI MARIE I SALAN MARIEN I SALAN MARIEN I SALAN MARIEN I SALAN Standah oppica Manland Profisional Plata Baster M Daater M Notifipila, N.: (1923) (1901 64-18 The Comcorde Morl-200468-1601

ATLANTIC CITY OFFICE GUARANTES TRUE DISCOME ATLANTIC CITY, N.J. OBAGI (608) 344-6339

CF COUNSEL

HALLS ALCH

CFFICE

CH5 FR 3808

(46 FR 51776

12 3815 91:53

ant

. .

January 13, 1982

Secretary of the Commission U. S. Nuclear Regulatory Commission Washington, D.C. 20535

Attantion: Docketing & Service Branch

Gentlemen:

Enclosed please find an original and one copy of Comments of the Township of Lower Alloways Creek to Proposed Rule for Licensing Requirements for Land Disposal of Radioactive Waste (10 CFR Part 61).

Yours very cruly,

Carl J. Valore Special Nuclear Counsel for the Township of Lower Alloways Creek

Inc. (2)

CJV:17

1122 82 amp

: WASTE TO: Secretary of the Commission U.S. Nuclear Regulatory Commission Washington, D. C., 20555

NUCLEAR REGULATORY COMMISSION

UNITED STATES OF AMERICA

COMMENTS OF THE TOWNSHIP OF

LOWER ALLOWAYS CREEK TO PROPOSED RULE FOR LICENSING REQUIREMENTS

FOR LAND DISPOSAL OF RADIOACTIVE

Attention: Docketing & Service Branch

10 CFR PART 61, et al

F-Z

1. <u>Background</u>: The Township of Lower Alloways Creek (TOLAC) is a community located in Salem County; New Jersey. Within its territorial borders are Salem 1 and 2 Nuclear Generating Stations (operating licenses) and Hope Creek 1 and 2 Nuclear Generating Stations (construction licenses). Salem 1 and 2 have been permitted license amendments and/or changes to the plant design to permit increased spent fuel storage in the spent fuel pools from 284 spent fuel rods in each spent fuel pool to 1,170.

2. <u>Site Selection and Characterization Phase</u>: The proposed rule and the supporting draft environmental impact statement discuss the applicants coordinating with the NRC, State and local governments as to site selection and characterization. (Draft environmental impact statement-NUREG-0782 Volume 3 E-5) (Part 2- 2.101(b) (1) (i))

The procedure as contemplated by the rule is that the applicant would establish a region of interest and then develop a most-favorite site for evaluation of information. After establishing a most-favorite site the applicant would file an application which would be announced in the Federal Register and a hearing could potentially be held in conformity with Part 2 of the Commission regulations.

-2-

(a) The applicant should be required to notify the community wherein the site is proposed and any adjoining communities of an intent to file an application with the NRC at least sixty days prior to the filing of the application. This would enable local communities to participate in a meaningful way in the licensing process. As a practical matter local communities do not have the socio-economic-political flexibility to respond to licensing procedures as parties or intervenors unlass they are given adequate lead time in reference to the proposed action.

(b) The environmental impact statement required under 10 CFR Part 61 for a waste disposal facility and the environmental report submitted by the applicant should consider the cumulative effects of putting a waste disposal fability within a 25-mile radius of an operating nuclear power plant and/or storage of spent nuclear fuel. The existence of flood plains and road networks should be carefully considered as well as the cumulative effects of the increasing radiation hazards by accregating nuclear land uses in reasonably close proximity.

J. <u>Institutional Control</u>: The concept that the Federal or State land owner will carry out custodial care of the site in such activities as site security and monitoring of the environment is not entirely satisfactory.

(a) The applicant and the government land owner should be required to enter into a contract with the local community that is host to the site. That contract would require the applicant and government entity to post bond in an adequate

E

amount for maintaining site security and monitoring the environment. In the past State regulation of sanitary landfills has not always been consistent with the safety and health of the public. Many hazardous wastes have been placed in sanitary landfills and have endangered local communities and residents. In such instances local communities and private land owners have found inadequate legal remedies and an inability to obtain immediate relief where water supply and other beneficial attributes of land ownership have been interfered with. Much of this hardship on local communities and private land owners could be eliminated if the applicant and the State or Federal government as land owner was required to enter into a contract with the local community whereby specific contractual obligations would be undertaken in the event of pollution of ground water, soil or air as a result of the operation or custodial care of the site. The form and execution of such a contract should be approved by the Nuclear Regulatory Commission and made part of the requirements of approval for a license.

F-2

-3-

(b) The local community that is host to the site should be included in approving any funding mechanisms which are created for the long-term care of the facility.

4. <u>Record-keeping</u>: An accurate record of any and all wastes disposed of at the site should be filed with the clerk for the local community. Such a record will promote public involvement and awareness and reduce unfounded fears and suspicions.

(a) The proposed rule apparently assumes that arrangements on record-keeping can be worked-out between the State and local Community. This is unsatisfactory and may not take place. It should be the Nuclear Regulatory Commission's responsibility to provide a proposed rule that full records regarding the type of materials, their composition, and their location within the site; the volumes of material by type; the identity of the shipper and the materials shipped are all supplied to the

-4-

local community as disposal is taking place.

.

5. <u>Closure</u>: It is not clear whether an application for closure is an amendment to the license which would require a hearing pursuant to Part 2 of the Rules of Practice. Any application for closure should be considered a significant amendment to the license and should be served on appropriate Federal, State and local officials and if a hearing is requested then the NRC should appoint an atomic safaty and licensing board to review the amendment to the application to permit closure.

5. Local Reparations: It should be recognized that potentially hazardous uses of land are local undesirable land uses. The Nuclear Regulatory Commission should require that the applicant or State or Federal land owner or combination of the same provide economic reparation to the local community for the diminution in property value and partial loss of tax ratables in respect to property values that may be affected for a reasonable radius surrounding the site. The process could be established in the rule and if agreed to by the local community or individual property owners be in lieu of any statutory or common law remedies the local community or property owners would have predicated upon inverse condemnation.



## GENERAL 🎯 ELECTRIC

Secretary of the Commission January 11, 1982

General Electric Company Wilmington Manufacturing Department Wilmington, N. C.

• •

#### COMMENTS ON PROPOSED 10 CFR 61

#### OVERALL COMMENTS

(1) The approach used by the Commission to emphasize performance specifications, supplemented only as necessary by technical specifications, is to be commended. This will assure protection of the public health and safety and at the same time capitalize on the benefits of specific site characteristics. This approach also removes the ultraconservatism that would be necessary with generic technical specifications applicable to all conceivable locations.

(2) The Commission has established a manageable approach to the complex issue of waste classification and should be commended for this. The documentation of the waste categories, the specific criteria for these categories and the rationale for arriving at them, provide an excellent and workable basis for arriving at final regulations.

#### SPECIFIC COMMENTS

#### Page 38085 (Supplemental Information) -Waste Characteristics 6 Classification

. . . .

We support the concept of not developing a waste classification system based on total hazard. While concerns related to both the chemical and radiological components of a waste must be evaluated to determine the proper disposal method, the establishment of a total hazard classification system would be a redundant effort. The U. S. Environmental Protection Agency already addresses chemically-related concerns through the comprehensive regulations associated with the implementation of the Resource Conservation is Recovery Act (RCRA).

# GENERAL 🌀 ELECTRIC

Secretary of the Commission January 11, 1982 Page 2

 $\mathcal{D}$ -55-7 The proposed regulations properly address necessary radiation protection considerations and provide adequate guidelines for judging proper treatment or exclusion of nuclear of waste components. This avoids the need to develop a very complex classification system that somehow interrelates radiological and nonradiological concerns on a detailed basis.

> We also support the continuing work of the Commission to identify and define specific wastes which can be exempted from Part 61. This activity can result in conservation of the limited low level

D-55-2 This activity can result in conservation of the provide the

#### Page 38086 - Manifest Tracking System

The need for a manifest system to assure traceability of waste shipments from a generator through the transporter and finally to the disposal site, is recognized. We question the need for a copy of the manifest preceding the shipment for the following reasons:

- - 2) The expressed concern that a missing or delayed shipment would not be detected can be reconciled by other methods such as an independent transmittal of the manifest at the time of shipment or by telephone notification to the receiving facility at the time of shipment.
  - 3) The requirement for the manifest to precede the shipment implies that the shipment should not leave until notificaton has been received by the shipper that the receiver has received said copy. This can result in a shipment sitting for an extended period while these notices go back and forth.

## Page 38087 - Operational Phase

 $\overline{B-5}$   $\lambda$  requirement that the disposal facility license be periodically renewed for continued operation is noted in this paragraph as is a 5 year normal term for material licenses. We suggest the concept that it not be necessary to renew the license. Paragraphs 51.25 and 62.26 in the proposed regulation provide adequate assurance

## GENERAL 🍰 ELECTRIC

1

Secretary of the Commission January 11, 1982 Page 3

B-5 that licensee-originated changes will receive review by the Commission. New-found issues of national concern that are independently identified by the Commission can be applied on a national basis to the sites as such issues are identified and are not dependent on waiting for a reneval application.

Paragraph 61.2 - Definitions

"Near: surface" disposal facility means land disposal facility in which radioactive waste is disposed or in or within the upper 15-20 meters of the earth's surface.

It is suggested that this definition be changed to read as follows:

Near surface disposal facility" means disposal facility in which radioactive waste is disposed on in or within the upper 15-20 meters of the earth's surface or to whatever greater depth can be demonstrated as capable of meeting the required performance criteria and technical specifications.

Rationale - The restriction in or within the upper 15-20 meters could prevent utilization of greater. depths at locations where hydrogeological conditions and waste stability characteristics would allow this. The criteria of the proposed regulation are established to prevent exposure to the public by transmittal through ground water flow and to prevent exposure to the intruder. The establishment of an allowable depth should be made on a sitespecific basis and with the objective that the criteria will be met. The unsubstantiated establishment of a nationwide depth limit is not in keeping with the logic used throughout the rest of the proposed regulation. . . .

.

61.3(a)

:

ED-

No person may receive, possess, and dispose of radioactive waste or byproduct material at a land disposal facility unless authorized by a license issued by the Commission pursuant to this part.

- -

GENERAL 🎯 ELECTRIC

Secretary of the Commission January 11, 1982 Page 4

It is suggested that this paragraph be changed to read as follows:

. .

.7

---3

27

No person may receive, possess and dispose of radioactive waste containing source, special nuclear, or byproduct material at a land disposal facility unless authorized by a license issued by the Commission pursuant to this part or unless exemption has been granted by the Commission under Paragraph 61.6.

Rationale - Paragraph 61.3(a) as written would prohibit transfer for land disposal of any radioactive waste to a nonlicensed person. This is overly restrictive and would force the shipment (to a licensed facility) of radioactive wastes that are not of a health or safety concern. The suggested addition to paragraph 61.3(a) would allow determinations to be made by the Commission on a case-by-case basis where it could be demonstrated that health and safety concerns could be met by alternate disposal methods.

61.24(5)

The licensee shall submit written statements under oath upon request of the Commission, at any time before termination of the license, to enable the Commission to determine whether or not the license should be modified, suspended or revoked.

It is suggested that this paragraph be deleted.

Rationale - The paragraph is very vague as to intent and method of implementation. It is not clear under what circumstances such an oath would be required and has a direct implication that the licensees are untrustworthy. There are certainly sufficient written transmittals required in other paragraphs of the proposed regulation to obtain necessary documentation of deliberate falsification of information. . . .

51.25(a) - Changes Except as provided for in specific license conditions, the licensee shall not make changes in the land . . . • •

· · • . .

ED-1

- B-12

# GENERAL DE ELECTRIC

Secretary of the Commission January 11, 1982 Page 5

B-4

described in the license application. The license will include conditions restricting subsequent changes to the facilly and the procedures authorized. These restrictions will fall into three categories of descending importance to public health and safety as follows: (1) those features and procedures which may not be changed without (i) 60 days prior notice to the Commission, (ii) 30 days notice of opportunity for a prior hearing, and (iii) prior Commission approval; (2) those features and procedures which may not be changed without (i) 60 days prior hearing, and (iii) prior Commission approval; (2) those features and procedures which may not be changed without (i) 60 days prior notice to the Commission, and (ii) prior Commission approval; and (3) those features and procedures which may not be changed without 60 days prior notice to the Commission. Features and procedures falling in paragraph (a)(3) of this section may not be changed without prior Commission approval if the Commission, after having received the required notice, so orders.

disposal facility or procedures

It is suggested that this paragraph be changed to read:

... the required notice, so orders. Changes that do not impact public health and safety Can be made immediately with subsequent notificaton to the Commission within 90 days.

Rationale - The necessity to make minor changes that do not impact on the public health and safety occurs routinely during the operation of a facility. The requirements in Paragraph 61.25(a) that no changes can be made without prior notice to the Commission, are overly restrictive.

# GENERAL 🏵 ELECTRIC

Secretary of the Commission January 11, 1982 Page 6

61.56(a)(1)

The waste must be packaged and the waste form and packaging must meet all applicable transportation requirements of the Commission set forth in 10 CFR Part 71 and of the Department of Transportation set forth in 49 CFR Parts 171-179, as applicable.

D-56-6 It is suggested that this paragraph be changed to read:

... and of the Department of Transportation set forth in 49 CFR Parts 171-179, as applicable. In the case of unpackaged (bulk) shipments, these must meet the requirements of 49 CFR 173.392.

Rationale - The proposed regulations should provide for the shipment of bulk (unpackaged) wastes under conditions that comply with Department of Transportation requirements for such wastes and that the wastes can meet the proposed Part 61 criteria when disposed of at the burial facility. A requirement on packaging would serve no useful purpose under these circumstances and should not be imposed.

61.56(5)(1)

Waste must have structural stability. A structurally stable waste form will maintain its physical dimensions within 54 of its form, under the expected disposal conditions of compressive load of 50 psi, and factors such as the presence of moisture, and microbial activity, and internal factors such as radiation effects and chemical changes. Structural stability can be provided by the waste form itself, processing the waste to a stable form, or placing the waste in a disposal container or structure that provides stability after disposal.

D-56-11

## GENERAL 🎲 ELECTRIC

Secretary of the Commission January 11, 1982 Page 7

It is suggested that this paragraph read as follows:

Waste must have structural stability. A structurally stable waste form will maintain its general physical dimensions and form under the expected disposal conditions and factors such as the presence of moisture and microbial activity, and internal factors such as ...

Rationale - The requirement of withstanding a compressive load of 50-psi (more than 7,000 pounds per square foot) appears to be a very rigorous loading requirement and is above that available from many soils. If it is still felt that a numerical value is necessary, then consider one of these approaches:

- Make the compressive load requirement for the waste when buried no more than that of the surrounding soils at the site under consideration.
- Evaluate a structural approach recognizing that the waste is constrained by surrounding soil and other wastes. This could conceivably reduce the compressive load requirement by a tenfold magnitude and still have an adequate safety factor.

Similarly, requiring that the waste retain its form within 5% under loading is a very rigorous requirement and using the second potential approach listed above, could be eased or eliminated.

51.56(5)(3)

D-56-1

D-56-10

Void spaces within the waste and between the waste and its package must be reduced to the extent practicable.

It is suggested that this paragraph be deleted.

Rationale - This is a very subjective statement and open to varying degrees of interpretation. The goal of reducing the void spaces in a waste package is desirable and will be attained because of aconomic incentive independent of regulations.

ibav

Secretary of the Commission 62 1915 P4:54 U.S. Aucleur hegulatory Commission 62 1915 P4:54 Ashington, D.C. 2033 ATL: Docketing and Service Branch gg.

Deur Sir/Hs.:

I would like to comment on the proposed rule, "Licensin, A GFR 5/776) Acquirements for Land Disposal of Radioactive Waste" (45 FR 38081), and on the Draft Invironmental Impact Statement, WREG-0702, supporting that proposed rule.

In light of the troubled distory of land disposal of redioactive wants, this proposed rule is an a initiable attempt to regulate such facilities. The provisions which would subject these facilities to the MRC nearing process, thereby encouraging public dents, are especially commendeals. Hereby encouraging insufficient consideration has deen given to the long-term consequences of shallow hand burish of radioactive wastes.

. realistic extrapolation of past and present societal trands would reveal some flaws in the analysis presented in MURIG-0782. Addicactive wastes remain dangerous for centuries; their safe disposal is dependent upon a stable social structure for at least io) years. Mistory (and current events) clearly shows that such stability cunnot be predicted, much less insured. It is questionable whether the United States, as we now know it, will exist 500 years from now. The stability of the financial world is even more uncertain. The financial assurances proposed in MURG-0782 are not depression-proof. Should this Action be intext in the future, its economy most surely would not have remained static. The effects of war have not been considered; even a conventional war fought near a waste disposal site would release large quantities of realise the to the environment.

C-4 The DEIS analysis is particularly deficient in its assessment of future land use. Present trends in population growth, soil erosion, and water resources indicate that in the future land and water will be extremely precious commodities. There is already a mater shortage in many parts of the country. As the population increases and more and more farmland is lost to erosion and uroan growth, arable land will be in short supply. It may be that the "intruder scenarios" proposed in the DIS will not be insivertant, as is now assumed; the need for land and food may be so severe that the use of waste ourial sites for agriculture may be permitted. This is especially likely since the site characteristics (topography, soil permeability, meteorology) most favonice for waste disposel are also the dest for farming or residential purposes. The water orisis may be so serious that the use of contaminated while will be allowed.

> In example of an analysis in the DIN that is deficient. even in the context of today's society is that of exposure culculutions for the intruder-agriculture scenaric, appendix 3, Section

. . .

. . .

Samewiedered by card 122 82 energy

January 4, 1982

JANTS I.

2000000

RECEIPTING SULL

---

3.4.2. The intruder is assumed to be an individual who lives in a house on the waste purial site and consumes food grown in a garden on the site. The intruder is assumed to work at a regular job during the day and spend only about half his the at home. This scenario does not consider a more plausible family-intruder, in which the following factors would alter the risk shalysis: (1) scena members of the family not work or may work in the home, thus spending much more time at the site; (2) some family members hay be children or pregnant water, who are nore subsidire to radiation. This analysis is indicate for present-day society; its applied blick to the fature is even more uncertain.

The DEIS also neglects the vulnerability of shallow land ourial to acts of terrorism or sabotage.

I fear that the proposed rule, if adopted, would encourage the proliferation of what disposal sites. This is undesirable for the reasons detailed acove. Instead, the problem of rudioactive waste disposal should be addressed from another perspective, that of waste reduction. Acther than continuing to produce wastes when there is no satisfactory method to dispose of them, a moratorium on waste production from those waste sources for which there exist alternative means of obtaining the same benefits is appropriate. The use of nuclear fission to generate electricity is an obvious candidate. Anote perspectives exist for the production of electricity; solar, wind, coal, menetohydrodynamics, etc. The costs, of which waste generation is one. Far cutweight the operating plants will decrease the volume of waste to as disposed of up about 65% (according to figures on p. 5-66 of .ULAD-0782). This would permit the use of existing waste facilities while alternative meterials. This alternative, which was not considered in the DITS, is clearly the most reasonable solution to the problem of redire the waste the volume of which was not considered in the DITS, is clearly the most reasonable solution

> Sincerely, Amy S. Hubbard Box 5636 Cleveland, Chio 44101

Florida Power C7/12 January 13, 1987 3F-0132-12 File: 3-0-3-a-10 Secretary of the Commission **U.S. Nuclear Regulatory Commission** Washington, D.C. 20555 Attn: Docketing and Service Branch Subject: Crystal River Unit 3 Docket No. 50-302 Operating License No. DPR-72 Proposed Rulemaking on Land Disposal of Low-Level Radioactive Waste Dear Sir: Florida Power Corporation wishes to offer the following comments regarding the proposed rulemaking on land discosal of low-level radioactive waste as published in the July 24. 1981 Federal Register. First, the requirements concerning transportation and packaging appear to have much more impact upon waste generators/shippers than upon licensed land disposal facilities. These requirements may be inappropriately placed in a regulation

D-55-6 Second, the three class identifications specified in the proposed rule appear to require additional analysis, handling, and, therefore, increased exposure to individuals for no apparent benefit. State-of-the-art methods now available do not lend themselves to performance of the specified isotopic analysis. Aside from the methodology question, personnel resources now available are unable to support the performance of the extensive isotopic analysis specified in the proposed regulation.

D-55-11 Finally, the cost/benefit aspect of classifying waste when compared to the safety aspect is not apparent. The additional risk and costs associated with additional handling and analysis must prove beneficial either in reduced risk to the public or in reduced cost of disposal in order to be a sound business practice.

Florida Power Corporation appreciates the opportunity to provide comments regarding this proposed rule and respectfully requests that these comments be considered in the final rule.

Very truly yours,

concerning land disposal facilities.

David G. Mardis Acting Manager Nuclear Licensing

General Office 3201 "hmv-loum Sirest South + P.O. Box 14042, St. Petersburg, Forica 33732 + 813-466-8151

(-4)

## POWER AUTHORITY OF THE STATE OF NEW YORK 10 COLUMBUS CIRCLE NEW YORK, N. Y. 10019 (212) 397-6200 32 J.H. 15 P.S. China H S Len JOSEPH A. SCHMIEDER EBERUTIVE VILE PREMERT & EMILP ENGINEER LERGY W. SINCLAIR SEMME VICE PERMENU & CHEF FINANCIAL SPRICER JPN-82-8 THOMAS A. FREY PERMON VICE PRES BERT & SEREAL COUNTEL · • . Secretary of the Commission 92 U. S. Nuclear Regulatory Commission Washington, D. C. 20555 '-<sup>'4</sup>'7 . . . . . .

\*\*\*\* \* \* <sub>10</sub> \* .

•

-2etal

Attention: Docketing and Service Branch

TRUSTEES

--------

-----

-----

RICHARD M. PLYNN

ROBERT & MILLONZI

-----

Subject: Indian Point 3 Nuclear Power Plant Docket No. 50-286 - CH6 FR 38081 James A. FitzPatrick Nuclear Power Plant Docket No. 50-333 Comments on Proposed Rule - 10 CFR Part 61, "Licensing Requirements for Land Disposal of Radioactive Waste"

Dear Sir:

Attachment A to this letter provides the Authority's comments on proposed 10 CFR Part 61, "Licensing Requirements for Land Disposal of Radioactive Waste."

· · · · · . . . . . . . Very truly yours, . ` 5... 1.627 · • • .... in second J. P. Bayne 5 · · • · · · Senior Vice President -Nuclear Generation . . • . • • •

cc: Resident Inspector -Indian Point Unit 3 U. S.: Nuclear Regulatory Commission P. O. Box 33 Buchanan, New York 10511 Mr. J. Linville . Resident Inspector U. S. Nuclear Regulatory Commission P. O. Box 136 Lycoming, New York 13093

Adjo 2010 2010 2010 2017.

. .

ATTACHMENT A

.

.

Comments on Proposed Rule - 10 CFR Part 61, "Licensing Requirements for Land Disposal of Radioactive Waste."

The Authority has the following specific comments on 2 .4:15 P5:10 proposed 10 CFR Part 61;

Page/Paragraph	Comments	
p. 38097 para. 61.55 (a) D-56-2	Chelating Agents - Table 1 has a eliminating wastes containing che in concentrations greater than 0.1 specifically approved by the Comm requirement eliminates most routin tion techniques used to reduce oc exposure and therefore it is not with the ALARA philosophy. One d already has a 1% by volume limit of factor of 10 greater than this re- this criterion must be met, accept for packaging of these agents must	footnote lating agents & except as ission. This ne decontamina- cupational in conformance isposal site which is a gulation. If able methods t be developed.
p. 38097 para. 61.55 D-55-6	Table 1 - This list of radioisoto waste into three classifications. of the concentration limits shown demonstrating compliance with man emitters would be extremely diffi analyses of isotopes such as Ni-5 TC-99, I-129, and CS-135 would be difficult to perform for every sh or yearly analyses such as those p Transuranics (TRU) by research fa	pes divides While some are reasonable, y of the beta cult Accurate 9, Ni-63, Nb-94, extremely ipment. Batch erformed on clities would
D-55-3	be much easier to implement. Thi preferable because the measuremen nanocuries/gm), in the presence of radionuclides, is very difficult t house. Detection is easy if TRU enly enes present. Additionally, crease in the use of volume reduc the concentration listed are very	s method is t of TRU (>10 other interfering o perform in- isotopes are the with the in- tion techniques low and may
D <i>-55-8</i>	become very cumbersome and outdat future. The limits in Table 1 sh to determine their ability to be realistic situation and in a cost	ed in the near- ould be reevaluated achieved in a -effective manner.
p. 38098 para. 61.56 (b) (1) D-56- 9	By specifying a structurally stab (<5%) the rule may be unintention. high integrity containers for all Experience has shown that drums an normally only be filled to 80% of Further, the 50 psi criterion show the specific burial site rather the	le waste form ally mandating wasta forms. nd. liners can total capacity. uld be left to han it being
וי-זכיע	La generic criterion.	

. .

. \*. . . . . . · · • · · · · : . . . . . الميان المحاجمة . . . . .

٠-,

1

• • .



B-340

R. Dale Smith

F-1

January 12, 1982

'between the states and the federal government. Rather there should be full cooperation between state government and federal agencies in all phases of low-level waste management. This rulemaking as drafted does not lay the groundwork for that cooperation. For example, the state proposal for participation required under Part 61.72(b) and (c) calls for a submission by the state of various specific items of information at a time so early in the process that all the state's concerns may not yet be apparent because of lack of information. While we understand the Commission's desire to avoid an unnecessarily protracted participation by a hostile state, nonetheless the legitimate interests of state governments should be accompodated in a more thorough and flexible manner. The regulations as drafted do not accomplish this. Note also in Part 61.71 the statement that "upon request of a state or federal government body, the director may make available Commission staff to discuss with representatives of state." (emphasis added). As a first step in the right direction, surely the word "may" should be replaced with "shall." This section should be completely revised to facilitate collegiality between the federal government and the states.

-3-

6) Part 61.32 -- Commission inspections of land disposal facilities. We thoroughly endorse the notion that the Commission should be afforded an opportunity at all reasonable times to inspect radioactive wastes and the premises, equipment, etc. An explicit provision should be added that host states enjoy a similar right.

 $P_{art 2.} \underbrace{ \begin{array}{c} \text{Subcart 2:764 (a) (b) (e).} \\ \text{The intent and consequences of these parts are unclear. They appear to authorize an initial decision by the Commission that would preclude effective appeal by either a concerned party or state. Immediate effectiveness, as it has been implemented by the Commission in rector licensing, has had the effect of denying states effective participation, discouraging cooperative efforts between state and federal governments, and rendering state's appeals ineffectual, since a facility would commence operation before appeals had run their course. The implications of these sections should be clarified. \\ \end{array}}$ 

Below you will find additional detailed comments of various State agencies on the proposed rulemaking.

#### THE RESOURCES AGENCY

···· , ··· ··

With regard to site suitability described in Section 61.50(a)(5), the criteria should be changed to require a lower risk of flocding. Currently, the section would allow waste disposal in a flocdplain that is likely to be flooded less than once every 100 years. The current ratings of flood risks are crude at best. For example, in California we have had floods rated as a 100-year flood and as a 300-year flood, both within the last 25

D-50-35

years. This experience has led many people to suggest that our estimates of flooding potential are much too low.

Based on the California experience, we would suggest that the 100-year flocdplain discussed in the regulation should be increased to at least a 300-year floodplain and, preferably, to a 500-year floodplain. Where the purpose of the disposal site is to keep the wastes isolated for a period of at least 500 years, surface flooding of that site should be avoided within our best estimates of what would be likely to happen within that 500-year period. The experience at Maxey Flats, Keneucky, should convince people that flooding of the disposal site should be avoided.

Third, the performance objective in Section 61.2 concerning protection of individuals from inadvertent intrusion should be strengthened. Some kind of permanent sign or warning device should be in place at the perimeter of the site. The warning sign or symbol should be designed to last 500

-1

. .::

D-51-2 Cevice should be in place at the perimeter of the site. The warning sign or symbol should be designed to last 500 years and to remain effective as a communicator, even if the language spoken in the area changes within that period. An example could be a combination of the skull and crossbones and the symbol for nuclear radiation.

D-55-8 Fourth, although the regulations describe minimum requirements for waste characteristics to be accepted at a disposal site, the regulations do not appear to require some kind of checking of the condition of the materials at the site.  $\lambda$  site could experience the problems found in the past when sealed steel drums were delivered for disposal and no one knew what the drums of the drum, either at the site of origin or at the disposal site, the requirements for waste characteristics may well be ignored by many of the waste generators.

#### OFFICE OF EMERGENCY SERVICES

F-1

F-Z

B-\_ e

E-1

Following closure, the draft assumes the State becomes the site owner (pp. 3-36 of NUREG-0782, V-2.). However, paragraph 61.59 of Part 61 states either the state or the federal government shall become the site owner. Governmental ownership is certainly desirable; however, the apparent conflict should be clarified.

Paragraphs 61.1 and 61.3 indicate that licenses will be issued by the NRC. In paragraph 61.70 through 61.73, provision is made for a state or tribal government to participate in the licensing process, yet it is, quite clear the NRC retains sole authority to issue the license. This suggests that a local jurisdiction has neither a voice in determining whether or not a site is established in their locale nor the conditions under which it is established and operated. The NRC should take steps to facilitate participation by affected local governments, including consideration of funding such participation.

If the State government has little or no real control during the functional life of the site, there is some question whether it would wish to assume responsibility for the site when it was closed. This would be especially questionable if the new site owner (i.e., the State) was expected to fund the cost for maintenance and monitoring.

Although several methods are mentioned for providing funds to the institutional authority, the rule makes no provision for it. In fact, the Commission admits it has no authority to "...require land disposal facility licensees to provide financial responsibility for activities occurring after the original licensee's responsibilities have caused and the license has been transferred to another party." We would suggest the Commission ask Congress for authority to require financial assurances for licensees for the active institutional control period.

For additional comments please see Part 7.2 below of the comments on the environmental impact statement.

F- | The California Department of Conservation (CDC) has reviewed the subject document for its geotechnical and procedural aspects. We. . .feel Section 61.72 is very important, providing for State participation in the review of any license application that affects the State. These procedures are very important to assure a real opportunity for the states, and thereby any affected local government, to have an effective input in the lowlevel waste (LLW) disposal process and specific site decisions which inevitably will impact all "host" states.

-6-

However, we believe that there is a significant defect in Subpart D, Subsection 61.50, <u>Disposal site suitability</u> requirements for land disposal. These requirements will not provide adequate protection to usable groundwater or to the environment from radionuclides that could be transported from the site by groundwater.

None of the stipulations in the disposal site criteria refer specifically to preventing migration of radionuclides into usable groundwater. Item (7) in Subsection 61.50 states, "The disposal site must provide sufficient depth to the water table that groundwater intrusion, perennial or otherwise, into the waste, will not occur. The Commission will consider exceptions to this requirement if it can be conclusively shown that disposal site characteristics will result in diffusion being the predominant means of radionuclide movement and the rate of movement will result in the performance chjectives of Subpart C of this part being met."

Our concern is that the above-quoted stipulation is concerned only with groundwater intrusion into the facility and, furthermore, would allow diffusion of radionuclides in groundwater as an acceptable concept in the disposal of waste.

What is lacking in these criteria is the fail-safe approach to planning and design. The uncertainties inherent in geologic, design, and operational factors for any LLW site cast serious doubt on the assumption that the wastes can be guaranteed to be isolated for the prescribed time. If radionuclides should prematurely escape from their confinement at the site, it would be difficult and expensive, if not impossible, to prevent their contaminating the groundwater. Therefore, CDC recommends that Item (7) in Subpart D, Section 61.50 be rewritten as follows:

D-50-2

## •

-5-

. . .

. . .

## Department of Conservation (continued)

D-50-2 The disposal site must not be located 1/ within basins containing usable groundwater or their recharge areas, or 2/ within geologic formations which will permit the diffusion of radionuclides to the environment, or their transport by groundwater to a degree exceeding the performance objectives of Subpart C.

> We recognize that the adoption of this recommendation will have the effect of decreasing the number and size of the search areas which would be eligible for consideration as potential LLW sites. Nevertheless, we believe that the seriousness of the risk of any radioactive contamination of groundwater warrants this degree of effort to assure that even if radionuclides were to escape, they could not contaminate any usable aquifer.

The regulations also fail to specify in Subpart G, Subsection 61.81 the nature and extent of Records, Reports, Tests and Inspections which will be required to ensure compliance with Subpart D - Technical Requirements for Land Disposal Facilities. Greater specificity is necessary regarding geologic, hydrologic, and other types of surveys and/or research to determine that potential sites comply fully with the regulations.

GEN-2-

DRAFT ENVIRONMENTAL IMPACT STATEMENT ON 10 CTR PART 51 "LICENSING REQUIREMENTS FOR LAND DISPOSAL OF RADIOACTIVE WASTE"

••••

. . .

The Draft Environmental Impact Statement is an important accompanying document, without which the proposal rulemaking would be difficult to assess. Comments from several state agencies are included below. First, however, I would like to make a few additional points.

1) Part 2.3 -- Alternative Disposal Methods. The EIS discusses briefly ocean disposal of low-level wastes. Although this disposal alternative is not addressed in detail within the EIS, we want to express our opposition to the use of the oceans for disposal of low-level wastes.

2) Part 4.6.1 -- Institutional Control Requirements. We support the concept of permitting disposal of low-level wastes only on land owned by the federal government or by the states; since the need for control of near-surface disposal facilities will last, in some cases, for several hundred years.

3) Part 5.5.1.2(2) -- Site Characteristics. With regard to the location of future sites, we believe the criteria should be changed to require a lower risk of flooding. We reiterate our comment (see comments above from The Resources Agency) that the 100-year floodplain may not be conservative enough. We suggest that a, 300-year floodplain or, preferably, a 500-year floodplain be required to avoid surface flooding of a site.

4) Part 5.5.1.3(2) -- Design and Operations. We share the view that prior to any license application, the applicant shall gather information concerning "the ecology, meteorology, climate, hydrology; geology, and seisnicity of the site." However, we disagree with the requirement that "for those characteristics that are subject to seasonal variation, data shall cover at least one full year." We believe this should be strengthened. Any locale's susceptibility to changing environmental factors requires that an attempt be made to gather historical data so as to try to accurately reflect how a proposed site has changed over time. We suggest that this section be amended to require collection of historical data going back a reasonable period of time, to the degree such collection is feasible.

3) Part 7.2 -- Waste Classification Based Upon Consideration of a Potential Inadvertent Intruder. The discussion of financial requirements during the operation of the

هيد يون د د م

However, there is a lack of depth to the analysis, and adequate solutions are not suggested for problems that have been identified. For example, per our comments above, it is clear that a sinking fund or some similar financial assurance mechanism would be the most preferable alternative for ensuring that necessary funds will be available for the lifetime of the site (i.e., including postclosure lifetime). Steps should be taken by the Commission to seek the authority to explicitly require that a sinking fund be established. Instead, the document endorses less satisfactory alternatives while at the same time the Commission recognizes the shortcomings of this approach. Additionally, the EIS, like the draft rulemaking, fails to account for the possibility of serious problems occurring at the site. It does not make contingencies for such problems or for the costs which a state would no doubt incur if such problems occurred. This is a major failing of the document and should be rectified. Costs and cost estimates should reflect the possibility of a serious failure of the site -- a failure of greater consequence than those that have already occurred at existing sites.

6) <u>Part 7.2.6 -- Transuranic Isotores</u>. We support the retention of the 10 nancourie per gram limit for surface disposal of low-level waste. We believe that wastes that exceed this limit should not be considered low-level waste and should not be buried at commercial low-level waste disposal sites.

7) Part 9.4 -- State, Tribal, and Public Participation. We would like to reiterate our point made earlier in Part 61.71 of the proposed rulemaking that there should be full cooperation between the state and federal covernments in all phases of low-level waste management. " This cooperation will strangthen the working relationship between the states and the federal government and thereby facilitate the safe establishment of necessary new disposal sites.

Below you will find additional detailed comments of several state agencies on the draft Environmental Empact Statement.

. • •

Note: No pase 10. Mis numbered.

-9-

## STATE WATER RESOURCES CONTROL BOARD

## General Comments:

. . .

1. In California, disposal to land of all but very low level radicactive wastes is prohibited by state law (California Administrative Code, Title 17, Section 30288, attached).

-11-

• \* \* j

2. The entire document fails to emphasize the need to prevent significant movement of pollutants from the disposal site to underlying ground water. The placement of an impervious cap over the waste will not preclude gravity drainage of liquid pollutants through a pervious trench bottom. Further, if the trench walls contain pervious beds (even lenses or "stringers"), water from precipitation or other nearby sources can move laterally into the trench, leach out pollutants, and then percolate vertically to underlying ground water. These ground water pollution threats can be essentially precluded by requiring disposal trenches to have impervious bottoms and sides. An engineered impervious barrier such as a clay liner could be racuired for each disposal trench. Better yet, the trench.site should be in an area having a substantial thickness of clay. (See Class I Disposal Site Criteria, California Administrative Code, Title 23, Section 2510.)

# Specific Comments:

- 1. Summary, Page 11. The abbreviations, "FWR" and "BWR" should be interpreted (re Report Page 3-10, bottom).
- Report, Pages 10-6 and 13 et seg. The southwest "hypothetical regional site" is described as serving 2. the western half of the country. The "High Plains" location, however, is far from the significant con-centration of nuclear generating facilities on the west coast. It would be more appropriate for the western hypothetical sits to be located near the west coast facilities.

3. Proposed Rule 10 CFR Part 61:

C = 7 f  $\lambda$ . Section 61:40 sets standards to avoid excessive exposure to humans. Excessive exposure to animal life should be avoided also.

D-50-2 B. Section 61:50 should include criteria requiring impervious material (natural or "engineered") beneath and along the siles of all disposal trenches.

### Attachment

### SUBCHAPTER 15. WASTE DISPUSAL TO LAND

#### Article 1. General Provisions

Attick 1. Central FLORIDA 2000, Definition of Terms, (a) "Disposal site" means any place next for the disposal of while or liquid waster. It does not include any part of a swape treatment plant or point of discharge of swape efficient out or Land damage from pipes or disclose into waters of the state. (b) "Disposal area" is that partition of the site which has received or is revealing waters. (c) "Landwate" is dratange from the nexts on fluid resulting from the percolation of liquid through a waste substance. (d) "Usable," ground on surface water includes potentially realist water.

(d) "Heilde" ground or surface water menors programmy commuter. (c) "Hydraulic continuity" is a condition existing when fluid occupy-ing an interview of a saturated material is able to move under a local differential to adjoining interview or surface channels containing fluid. (f) "Capillary fringe" is the parity saturated zone immediately above the water table in which water is held by capillary forces. (e) "Toric" means lefted, informate, in damaging to man on other fixing segminist including plants, dimension and inclusive stabiliza-tion of decomposing waste in a sile such that leachate or the rate of segmentation of gases is no longer a threat to water (mainly, "Your: botherie each Section 103, 1310, 1000, Water Case These I Vec Sofelagee 11 (1) EUR 2018, 1000, Water Case Theore I Vec Sofelagee 13 (1) EUR 2018, 1000, Water Case Theore I Vec Sofelagee 13 (1) EUR 2018, 1000, Water Case Theore I Vec Sofelagee 13 (1) EUR 2018, 1000, Water Case Theore I Vec Sofelagee 13 (1) EUR 2018, 1000, Water Case Theore I Vec Sofelagee 13 (1) EUR 2018, 1000, Water Case Theore I Vec Sofelagee 13 (1) EUR 2018, 1000, Water Case Theore I Vec Sofelagee 13 (1) EUR 2018, 1000, Water Case Theore I Vec Sofelagee 13 (1) EUR 2018, 1000, Water Case Theore I Vec Sofelagee 13 (1) EUR 2018, 2018, 2018, 2019, 2018, 2019, 20

# Article 2. Classification of Waste Disposal Sites

Article 2. Classification of Waste Disposal Sites 2510. Class I Disposal Sites. Class I disposal sites are those at which recepter postertion is provided for all time for the quality of ground and water waters from all waters deposited there is an all gainst hazard to public health and wildlife resources. The following criteria nort be mer to quality a site as Class I. (a) Coological conditions are naturally capable of preventing verti-cal fordogical conditions are naturally capable of preventing term the site and walls between liquids and gases remaining from the work in the site and walls to face or groundwaters. In Coological conditions are naturally capable of preventing lateral in the site and walls sufface or groundwaters. (c) Chalerlying geological formations which contain rock fractures of the disposal area bar barrier to the movement of liquids or gases from the disposal site to make permethility, must be permanently scaled to provide a comparison barrier to the movement of liquids or gases from the disposal site to make water. (d) Invalation of disposal areas shall on occur until the site is closed in accordance with requirements of the regional bard.

¢n

#### RADIATION 1111.1. 17 sherteter 16, No. 12-318 161

:301

(b) Lacreta from bulls thush undergoing medical diagonsis or thera-ny with salisawitive material shall be earmpt from any limitations tra-taneed in this section, provided that the user provides for appropriate rainedgeal monitoring whenever any waste loss in the user's insta-tion which may carry such exercts is opened.

then which may carry such exercises opened 20289. Disposed by Burial in Soil. (2) No user shall dispose of su-location material by burial in soil uniese: (1) the total quantity of radiancine material buried at any :\*\* for anoman's specified on Section 2035, Appendix 1, out (1, 1020 tr \*\*) the anoman's specified on Section 2035, Appendix 1, out (2) havial to at a minimum depth of four feet; and (3) successive burials are separated by distances of at least sis for; and not more than 12 fourials are made in any pres-tion down of the specified on the sources are madiention for herence re-

the the department will not approve any application for beense :: receive adjustive material from other persons for departation line ...; received by the Federal or State Concentration.

30759. Treatment of Disposal by Incineration. No user shall the or dispose of tadioactive material by incineration except as specifica-approved by the department pursuant to Section 2035 Note Authoride effect. Sections 107, 200, 200, 2011, fleibh and Salers Cala Human - ). New wettuen filed 9 22-11, effective showed day threesters (Regime No 20).

Article 6. Records, Reports and Notifications

Article 6. Records, Reports and Notifications REZIT. Records. (a) Each user shall maintain accurate and con-plete written records, as follows: (1) The results of each required calibration, survey and test. (2) Each receipt, transfer, and disposal of a source of radiation. (3) Badiation expressives of all hericiduals for show personner monitoring for required matter Section 20216. Exposure records that, be kept on department Form IIII 2005 of in a manner which inclures all the applicable information required on said form. Each entry shall be for a period of time not exceeding one extendar quarter. (4) Results of medical examinations and bio-assays pursuant at Section 20277.

(b) Each required record of dose received by individuals and re-inclueal examination and bioastay shall be preserved buildinilely re-until the Department buthorizes their disposal. Each other require-tectord shall be preserved for a period of three years following the date of the occurrence that is the subject of such record.

Sent by Bill Fraderen, Port. Atath, 5-093+-3-2757

(c) Dispersibilities shall not be subject to wadowit.

Leachate and autoutace flow into the dispersibilities shall be cructained within the site unless other dispersition is made in accordance with requirements of the regional board.
Site shall not be tested over zones of active faulting or where the former of active shall be to acted over zones of active faulting or where the former of ever zones of active faulting or where the former of average would impair the competence of natural formers of publical barriers which prevent continuity with usable over zone.

waters. (b) Sites made suitable for use by man-made physical barriers dual (b) Sites made suitable for use by man-made physical barriers dual not be brated where hoppoper operation or multitenance of met stanc-tures could permit the water, leadnate, or gass to evolve turable promet or surface water, (could evolve a start, barrier, e.g., g, and b bot would be subject (could be which comply with u, b, c, c, f, g, and b bot would be subject to invadition by a tick; or a Bool of preater than (R)-year frequency may be considered by the regional board as a limited Class 1 disposa-site.

ver. Alutary: 1: Amendment of unbrechain (a), (b) and (c) blevt 12.29.12, effective thirti-eth-day thereafter (Register 12, No. 33).



. r • . . . į.,

.

. Bulinas lite	Muna-	Reimarles	Nines- entire	Rolin getale	¥.
	.01	H~13	10	Sha M	
1 + +++++++ + + + + + + + + + + + + + +	147	Bedres FBL	je :	SPH IN.	
* Status 7 17 1		Prime 111		Julius. 313.	
Action 15	in the second se	from \$1		and and the second second	
Autor 11	. 14	ball.	30	Suration 45	
	M	Berghalt	the l	Flight a 34	
Nerman 121		A second second second		The set of	
Parian (11	- <b>#</b>	1 -nd 719		trates If	
		atefenge 111	hen .	6-4, ber. 16	
and the		Plant and the St		Toutolum 167	1 . 1
Bearson #7	n ni -	blangener.bt		Testerburger when	
adates Mit	10	Herring tota,	110	3-sharten #1	l i
4 Backer 119m		Moreney Wi	1ee	festartent ben	
et mer f	10 1	Distant of the second of		Address 11 a	
	10	1-m-11-mm 11/	140	felleren tim	
wine #1	14	S	110	1.8 min. 271	
00 1000 111		N 1017	144	1 ************************************	
# mm (11	· · · · ·	Sel 441	i iii	Televise (No.	1 1
131	1 100	Iretune that	14	Triane Ill	
		<del>(*</del>	1 <u>11</u> - 1	Test rem the sussesses a sure	١.
mine 111		All and a state of the second state of the sec		Theffe and 199	
	- W -	franken Pilin		12mlma 202	
wines fol	10	france HI	100	11 uffange 201	[ ·
Maine 10	<b>`</b> !!	Tallada (0)		Thursday Jacksrufy	
Auge une \$1	1.00	freden ber	1.0	The are the	
a-H.S.a	90 .	3 37		1= #12 the second second	
Mall St		Therease Proven		[ ] = 173	
anges 64	100	Theirse [1]		1 mantes 143	1
19 mar 113	10	hateran 10 100	100 s	I summer but an an an an	j i
11 yearson 103,	204	Alarma Williams	100	Frances burters a drates	l I
Aires  11		Tutone )10		Patrane 111 - Landard 111	
	914	Trappine St	H	Torneto per til.	
and up (2) (3 have		Tour freine filt .	211	3 1/14	
W-100-153	1 m -	Frankleren Sil.		Arten 17	( ;
10	1.010	(namelan 113,	· •	Burrhaum \$15	Į i
ling design as 13 1, and and	5 <b></b>	Minahan 194	· • • •	A series of a series a series and	ł
Callana, 32	10	HArene W1.	1.11	Simine #	
	1 914	Bi -frens 18 hes	84	Stringer Bl	5
444 999	1 11 1	81	he		1.
Exform 181		But-tran M		1	ι.,
H	114	R-4 min \$1,	. hv .	Farries 13	1 ``
febaten ?	11:00	Hellengene \$87		1	)
and and States	1 22 /	Notice (1), and	1 4 1	Terreinen M.	1
-ture 118 -		Freminer Hit		and the state of the state of the state	1
m1	1 14	Franklas #11	tivi *	alphe aw Mirry of unknows a seen	1
	1 1 5	Fagedreer 11	1.2	here and the second sec	1
		Contrast In contrast	1.22	Beg telserglate attas plan phy ;	٢.
	1.1	*	1 16	1 shows or montants at firing parts.	1.
	1 ° V	bdess 37	1 100	hes of as Lores spage aders .	L
Jul + 4 877,		l i	1	1 .	L

As the basics denotes the two of a set, we (c), pr(c) and (r) M share there is an obtain a point and order to be a set of the set o

B-345

54

#### -15-Department of Conservation (continued)

-14-

## RESOURCES AGENCY, DEPARTMENT OF CONSERVATION

The California Department of Conservation, based on review by the Division of Mines and Geology, has considered the Draft Environmental Impact Statement with respect to geotechnical aspects and procedural requirements.

In the DEIS, NRC discusses the use of high-integrity disposal package containers with extended containment life (approximately 300 years) for use in the disposal of highconcentrations wastes, as a waste processing option (DEIS, Ch. 5.2.4.8, App. D.4.3). This section also discusses potential use of similar containers for lower concentration wastes, but usage of this type of containerized disposal is not required by the proposed regulations. Also, for less concentrated wastes, the proposed regulations appear to require that the disposal package containers maintain their integrity only during the operational phase of the disposal site tranches (DEIS, App. D.4.3). However, we feel that because the less concentrated wastes could still release radionuclides similar to, or even the same as, those contained in the waste packages for high-concentration wastes, container integrity is essential to preventing the release of radionuclides into groundwater (prior to adequate confined decay time) to insure that the resultant activity level is low enough to not cose a danger to public health and safety.

As discussed in the DEIS, the proposed regulations in 10 GFR Part 61 assume that in the event of early release of radionuclides from disposal containers, or from decontainerized disposal, the site design, including the geologic setting, should be capable of preventing radionuclide migration out of the disposal trenches and into the surrounding groundwater and environment. However, the proposed regulations provide no fail-safe assurance that this will be the case.

Even if the wastes were to be segregated according to the active life of the different radionuclides and disposed of in containers which could maintain their integrity for the necessary containment time of each of the different classes of radionuclides, there does not appear to be adequate provisions in the proposed regulations for enforcement of this degree of detailed inspection during wate processing. We feel that the potential for migration of radionuclides from the disposal site and subsequent contamination of groundwater in the vicinity of the disposal sites could, coupled with adequate site planning and design, be minimized by containerized disposal of wastes in containeers capable of maintaining their integrity for a minimum confinement period of 100 years [10 CFR Part 61, Subpart A, 61.7 (4)]. However, due to the lack of provisions in the regulations to require containerized disposal of all wastes,

• • •

along with uncertainty in the capability for adequate p-56-17 enforcement of the regulations relative to proper packaging and disposal, we recommend that item 7 in Subpart D, Section 61.50 of [the proposed rulemaking for] 10 CFR Part 61 (be rewritten as recommended above.

D-56-17

## -17-Department of Realth Services (continued)

٠<u>:</u>

common design and performance standard, the Congress might accept such a responsibility. That uniformity might, however, require some special handling as was done for uranium mill tailings.

To summarize, the EIS should contain a section specifically developed for informing the Congress on the impact of its impending action as authorized in PL 96-537. That section would chart waste disposal needs and costs by regions as they actually exist or are planned by the states. The conclusion of such a piece might well be that the implementation date of 1986 is too early in terms of waste volume, and unaffordable. Additionally, given the amount of time necessary to bring new sites into operation (4-7 years), the 1986 date in PL 96-537 may be premature, if safe management and disposal of these materials is to be assured.

• •

۰.

- . . . . .

# DEPARTMENT OF HEALTH SERVICES

We continue to be troubled by the cost issues and their presentation in this EIS draft.

. ..

-16-

First, we are discouraged to find NRC using their own regions for the waste data bases. The states have been working for more than a year now with regions and waste volume projections based on U.S. Department of Energy (USDOE) studies. Comparison, then, with the USDOE data becomes difficult or impossible. However, because we know something of USDOE's efforts, their strengths and weaknesses; there is a need for careful comparison of data and conclusions on such an important matter as this. The final EIS should facilitate those comparisons.

Secondly, costs are based on the 20-year period from 1980 to 2000. We think it important that costs be shown by year from 1986 (when exclusionary authority may be conferred by Congress) through 2000. For some regions (as defined by current state actions, or the USDOE), initial costs may verge on prohibitive. A review of USDOE data indicates that by 1986 only three regions would generate the volume of waste on which the EIS was based. One, Region 5 (USDCE), would not have the wasts volume by the year 2000 (see Table 1).

Given the history of some existing disposal sites, one key concern should be the assurance of adequate financial resources on the part of the applicant to construct and operate a disposal facility and to provide adequate financial provisions for site closure and long-term care.

The EIS, although it cites no specific cost figures, appears to underestimate the short- and long-term costs of operating and maintaining a low-level waste disposal site, and fails to recognize the problems small companies (as identified in the EIS) have in meeting financial requirements in operating a waste disposal site. It seems likely that few small companies can raise the necessary capital for plant development, set aside trust funds, cash deposits, purchase surety bonds against short-term financial needs and further set aside additional money for 100-year care costs within the life span of the disposal site. The most careful attention should therefore be paid to the financial resources of any applicant who seeks to develop and/or operate a new site.

The "unanticipated contingencies" not addressed by the EIS (i.e., problems occurring at a site) should, we believe, be explicitly addressed either by the NRC or the Congress. To the extent that all national sites meet or exceed a

# R. Dale Smith

Thank you for affording us the opportunity to comment of these documents. This is a most timely issue, and one which we are sure will benefit from the careful attention and input provided by all interested parties.

.

Sincerely, Mary a. Freeday

Phillip A. Greenberg Assistant to the Governor for Energy and Environment

ł

.

## -18-

• ·

# TABLE I

fraction of Representative Site Waste Volume by Region\*

	_1986	2000
Region 1	1201	230%
Region 2	46%	971
Region 3	- 1403	2901
Region 4	871	160%
Region 5	198	404
Region 6	521	84 %

.

.

\*Peference: Low-Level Radioactive Wasts Management Report, prepared by the U. S. Department of Energy, March 13, 1991.



ØCANØ182Ø4

January 14, 1982

D-55-3Additionally, the actual measurement of TRU in the 10 nanocurie per gram range while in the presence of other interferring radionuclides would be very difficult with today's technology. We do recognize that detection of 10 nanocuries per gram can be readily accomplished, however, if transuranic isotopes are the only ones present. Furthermore, it should be recognized that the present policy of volume reduction does increase the concentration of radionuclides in the waste and could cause the waste to exceed the Table 1 concentration limits. For these reasons, we feel that the concentration limits in Table 1 should be reevaluated to determine their ability to be achieved in a realistic situation and in a cost effective manner. Simply put, implementable technology does not exist at this time to realistically determine the concentrations characteristic of a given isotope, especially in dry trash.

2

COMMENT NO. 11 - SUPPORT OF AIF COMMENTS

Arkansas Power and Light Company endorses the comments prepared by the Atomic Industrial Forum Working Group on 10CFR61. Proposed Rulemaking on Licensing Requirements for Land Disposal of Radioactive Waste and on the Environmental Impact Statement - NUREG 0782. The AIF comments are hereby incorporated into ours by reference.

Sincerely,

David C. Trink

David C. Trimble Manager, Licensing

OCT: DET: s1

STONE & WEBSTER ENGINEERING CORPORATION

243 SUMMER STREET. BOSTON. MASSACHUSETTS ADDRESS ALL COARESPONDENCE TO P G. BOS 2322-5005TON. MASS. 02107 W U.TESZ P-DOOT PADOT SUMMER STATUME SUMMER SUMMER STATUME SUMMER SUMMER STATUME SUMMER SUMMER STATUME SUMMER SUMMER STATUME SUMMER SUMMER STATUME SUMMER SUMMER STATUME SUMMER SUMMER STATUME SUMMER SUMMER STATUME SUMMER STATUME SUMMER STATUME SUMMER STATUM

Secretary of the Commission Attention Docketing and Service Branch U. S. Nuclear Regulatory Commission Washington, DC 20555

Dear Sir:

LICENSING REQUIREMENTS FOR LAND DISPOSAL OF RADIOACTIVE WASTE, 10CFR61 46FR38081, JULY 24, 1981 46FR31776, OCTOBER 22, 1981

imposed by various disposal sites.

46FR51776, OCTOBER 22, 1981 We are pleased to submit our comments on the subject proposed rule. We generally concur with the philosophy and recommended approach embodied in the proposed rule and believe the proposed rule to be appropriate in that it will serve to increase the consistency of practices and requirements

January 12, 1982

V. Summary of Rule. On page 467R38087, in the paragraph entitled "Operational Phase," it is stated that a license reneval application would be submitted every 3 yr after issuance of the license. We suggest that the first 5-yr interval should commence after the Commission authorizes the licensee to receive waste, since the license is issued prior to construction and the operational phase does not begin until construction is complete.

Section 61.24(2). We suggest this paragraph be replaced with the following:

Prior to completion of the construction of the facility, the licenses will make available for Commission inspection any information necessary to assure the Commission that the facility has been constructed in accordance with the applicable requirements established in the application. At or about the time of completion of construction, the applicant will submit an attestation to the Commission that the facility meets the applicable requirements of the license.

The attestation shall state the name of the applicant, the name and location of the facility, the time when the facility is expected to be ready to commence operation, and shall contain a statement that the facility meets the applicable requirements of and conforms to the application for a license for such facility.

B-5

B-4

S of the C

-56-9

January 12, 1982

Within 10 days of receipt by the Commission of such streststion, notice thereof shall be published in the Federal Register. Thirty days following receipt by the Commission of such attestation, the licensee may commence waste disposal activities unless the Commission issues an order probibiling or limiting such actions and explaining the reasons therefor.

Section 61.26(h). The Commission should not make any change to a license unless that change is first justified wis the performance of a value/impact analysis.

2

Section 61.36(a)(1). This section requires that the waste package presented for disposal comply with NRC and DOT transportation regulations. This implies that the disposed package could or must be a Type A, Type B, or Large Quantity package including all related shielding and other transportation-related requirements. While it is unlikely that this is NRC's intent, the wording of the paragraph can be interpreted in this manner. NRC should clarify and reword this requirement.

Section 61.56(b)(1). The requirement that waste packages presented for disposal retain 95 percent dimensional stability after burial is inconsistent with the capability of most solidification processes. A solidification processes cannot fill a container 95 percent full. Therefore, when the container degrades, the waste form can compress to less than 95 percent of the original package volume.

The 95 percent stability requirement as written, therefore, implies or requires the use of high integrity containers for solidified materials. The option of using a high integrity container in lieu of solidification then becomes no option at all.

Section 2.764(e). The initial issuance of the license and any amendments to the license should be justified via the performance of value/impact statements.

Section 20.311(b). The shipmunt manifest should also indicate the "radiation level" of the waste container to be buried as well as the other waste characteristics moted.

We appreciate the opportunity to assist in the development of this rule, and hope that the above comments will assist you in its finalization.

Very truly yours,

R. B. Bradbury Chief Licensing Engineer

R3:sed

HEALTH PHYSICS SOCIETY 152210218 21:14 Brockneven National Laboratory Upton, New York 11973 Telephone: (\$16) 282-4209 January 18, 1982 -Secretary U.S. Nuclear Regulatory Commission Washington, D.C. 20555 Attention: Docketing and Service Branch Re: Licensing Requirements for Land Disposal of Radioactive Waste: Proposed Rule 10 CTR Parts 2, 19, 20, 21, 30, 40, 51, 61, 70. 73. and 170; 46 FR 38081, July 24. 1981 Gentlemen: ç.....

The Health Physics Society is a National organization of approximately 5,000 scientists and professionals engaged in the practice of radiation protection. The Society's primary objective is the development of scientific knowledge and practical means for the protection of man and his environment from the harmful effects of radiation while encouraging the optimum utilization for the benefit of mankind. It is in the sense of this objective that we offer, for your consideration and action, the following comments on the proposed rules referenced above.

These comments were developed by the Society's Committee on State and Tederal Legislation and are offered on behalf of the Society by the Committee and the Society's Officers. We wish to acknowledge the cooperation of NRC's staff in providing some initial information on the proposed rules changes and the Invironmental Impact Statement.

In general, we agree with the efforts by the Commission to develop the new Part 61 specifically addressing land disposal of radioactive vastes.

A common aspect of the proposed rules is the matter of how such radioactive wastes are classified (for use by shippers, i.e., waste generators; by processors, who consolidate shipmants; and by the recipients, i.e., licensed land disposal facilities). The Health Physics Society marbers would be most affected by the proposed rules which spply primarily to generators, although we are also interested in the health physics practices expected of facility licensees to protect the employees, the "intruder," and the general public during several phases of the life cycle of the facility.

We charafore believe and recommend that NRC clearly separate the rules with which the generator must comply in separating, identifying, classifying, packaging, labeling, and shipping wastas from rules which are to be uniquely met by the licensed disposal facility. There is an obvious overlap in the system for classifying wastes as shipped and as received, to provide primarily for long-term

B-351

- 0

NRC from Mainhold

January 18, 1982

P-SS-W protection from the operation of the facility. But the burdens of certification of packaging, atc. which fall on the shipper (or processor) should be clearly spelled out from those which fall on the facility licensee. We have attempted to address this fundamental issue in our comments below.

### Waste Classification

. .

. .

Under the proposed rules, wastes are to be classified as either Class A Segregated Waste; Class B Stable Waste; and Class C Intruder Waste. Implied is a classification below Class A, a de minimis category, and a classification above Class C, for wastes which have concentrations in excess of Class C. Essentially. there are these five categories that both the generator, the processor, and the facility operator need to consider.

-2-

The Health Physics Society applauds the effort by the NRC to develop, at least for these proposed rules, a de minimis classification that would be exempt from Part 61 (and other parts of 10 CFR that relate to low level radioactive vastes). D-55-2 These vastes would be considered to be a negligible hazard of no regulatory concern and could be disposed of perhaps in a sanitary landfill. We encourage the use of such exemptions such as was done in 46 FR 16230 for specified waste forms and concentrations, and recommend that an exemption (or de minimis) category be included in the Part 61 classification system.

- We are concerned that Table 1 (46 FR 38097) has too many requirements crammed in the columns and footnotes, setting forth both the classification requirements and waivers and other key provisions in one table. We urge NRC to dismantle this ED- | {and valvers and other key provisions in one classification in both tabular and table and express the requirements for each classification in both tabular and narrative form, with a separate table for each class (from exemptions through above Class C). Each separate table should define a range of concentrations or quantities that fall within that classification, and include the waivers that may apply to that D-56-2 group. Cartainly, the matter of disposing of radioactive wastes containing chelating D-56-2 group. Cartainly, the matter of disposing of radioactive wastes containing chelating

Similarly, beta-emitting auchides with little or no gamma radiation, beta-emitting nuclides with significant gamma radiation, and alpha-emitting isotopes other than radium should have been clearly listed in the table, and not buried in a footnote. It appears that the concentration limits for such beta emitters, those not specifically listed, are unduly restrictive. Since the classification system has great ED-1 impact on the shipper in the packaging and also the use of the manifest system

(see below), the separate subparts of 10CF361 must be easily understood side by side. The current conglomerate shown as Table 1 cannot permit an easy understanding as currently written.

To properly use the classification system proposed within Table 1, and the various footnotes and waivers, the generator will be required to perform monitoring and analysis of each container during waste separation and packaging. A major sampling problem could result if the generator-licensee had to prove that each 2-55-6 container pet the classification requirements stated on the labels or in the manifest. Recognition of the limits of instrumentation would assist NRC in the final development of suitable concentrations within each classification in the recommended expansion of Table 1 and footnotes. The instruments limit the ability of generators (of such wastes to carry out the classification process. We understand that detection D-55-3 of TAU at 10 aCi/g is possible when only TAU isotopes are present. We suggest that the figure for Class C limits for TAU wastes of 10 aCl/g be increased to perhaps

NEC from Meinhold

-3-

January 15, 1982

- - - -----

D-56-3 \$100 nCi/g, which would still provide protection for the health and safety of vorkers, intruders, or the public.

#### Waste Characteristics (61.56)

Paragraph (a) states that the requirements are intended to facilitate handling "at the disposal site" and "provide protection of health and safety" presumably all the way from the generator's facilities via transportation to the disposal facility. Yet, subparagraphs (1), (2), (3), (6), and (7) emphasize packaging requirements D-56-6 while subparagraph (8) refers to treatment of biological. etc. wastes.... Subparagraphs (4) and (5) also apply primarily to the generator of the waste, who has the most control over toxicity or explosive problems. These sections place a heavy burden on licensees disposing of radioactively-contaminated biological or infectious mat-Verial, particularly medical licensees, if the infectious concentrations are not known.

Paragraph (b) translates these requirements (which are primarily within the 'D-56-0 ) control and responsibility of the generator) into requirements for handling at the Laite to provide stability of the vaste for 150 years or more. Hence the structural (stability requirement becomes the long-term responsibility of the generator. D-56-10 as does the requirements for keeping liquids low and void spaces to a minimum. Does "practicable" in 61.36(b)(3) imply compaction or solidification? Would continued

land disposal of liquid scintillation fluids be permitted?

We are concerned that a generator will be held responsible for certifying that his waste, at the time of shipment, has been packaged so as to meet the D -56-8 stability requirements (of maintaining physical dimensions within 57 and its form under 50 psi and other factors for over 150 years). A separate rulemaking on the D-56-16 stability requirements for containers, as shipped by generators, would be useful. Ferhaps these requirements could be identified clearly as shipper responsibilities and included in the revision of 10 CFR 71.

Labeling (61.57)

The labeling of each container is the responsibility of the wasta generator and requires an understanding and use of the classification system lumped into Table 1. Does NRC intend to require standards labels, varning signs, or other D-57-1markings to supplement or replace current DOT labels? A clear set of classification markings would facilitate the land disposal operations and assist radiological emergency personnel who may have to respond to accidents involving low level waste en route to disposal facilities. Generators have some correct concerns over the precision with which they are expected to analysis the nuclide content. This is virtually impossible to do in a quantitative manner so the "less than" statements are usually employed.

#### Protection -(51.41, 51.42, 61.43)

• • •

The Society recognizes the need to assure radiation protection for employees, possible intruders, and the general public during land disposal facility operations C-7 and beyond. We recommend that each of these groups be properly protected using limits in 10 CFR 20, as currently written or as proposed, rather than proposing a system of new and unique limits related to waste disposal. The establishment of a unique system of dose limits for a separate area of the nuclear fuel cycle seems unnecessary and may imply a special need for protection in the minds of the public; thereby exacerbating an already confused public perception of radioactive waste disposal issues.

NRC from Meinhold

B-1

M-1

M-4

January 18, 1982

#### NRC from Meinhold

January 18, 1982

4

.

-:\*

1 1

.71

## Specific Technical Information and Technical Analysis (61.12, 61.13)

We concur that applicants for a facility license should include in the standard technical specifications a radiation protection program (item k) and an environmental monitoring program (item 1). Much of the future success of a maticnal land disposal operation for radioactive waste lies in achieving proper radiation protection for employees, intruders, and the public. All would be adequately safeguarded by emphasis on thoroughly planned and implemented radiation protection and environmental control programs. Obviously, these programs require proper analysis of instrumentation readings, maintenance of disposal logs, etc. to verify meeting of applicable requirements of Farts 20 and 51. Monitoring and analytical requirements for occupational exposures should be consistent with the criteria for ALARA.

-4-

## Transfer for Disposal and Manifests (20.311)

The purpose of this section is described as being designed to control transfer of waste and establish a manifest tracking system. On the face of it, the meed for a manifest system follows automatically from the increased emphasis on a new classification system for waste, on new packaging and labeling requirements, and on placing responsibility for compliance with these new requirements on the waste generator.

We cherefore recommend that the proposed manifest system requirements be included in revision to 10 CFR 71, Packaging of Radioactive Material for Transport ... and be applicable to the wide range of radioactive material shipments for the sake of consistency. The manifest system for radioactive wastes should be identical to the manifest system for any other radioactive material - or to any other hazardous material during transport.

There is clearly a need to establish a labeling system for packages intended for land disposal of radioactive wastes that is consistent with labels for packages containing other radioactive material - or other hazardous material. The labels and manifest information must obviously be identical and consistent with any label or manifest for any shipment of radioactive material.

With regard to the specific requirements of the proposed manifest system, we concur that, to be meaningful, there must be clear identification of each pack-

As indicated above, the entire responsibility for classification and conformance to waste characteristics falls upon the generator not just during the initial transportation but for the next 150 years (61.56(b)(1). We suggest that such a warranty be either specifically required in writing on each manifest or that a system of shared responsibility between generator and facility licensee be permitted. The generator will warranty that, to the best of his knowledge and efforts; the packages being shipped will meet the long-term, 150-year criteria at the time and place of shipment. The facility licensee will warranty to NKC or the State that, to the best of his knowledge and efforts, the packages as buried will meet the same criteria. The form of quality assurance programs by both the generator or processor to meet these criteria may need to be spelled out in more detail.

#### Regulatory Invact

The NRC sets forth the basis for the proposed rules and refers interested parties to NUREG-0782, the draft EIS, Volumes 1,2,3 and 4, issued September 1981. In the preparation of these comments, no detailed study of NUREG-0782 has been attempted. The following comments are based on discussions with persons who have made that effort and in response to issues raised in 45 TR 38088.

The proposed rule changes will have an impact on significant numbers of persons including organizations licensed by NRC and agreement States and users of non-licensed radioactive material who offer such material as wastes for final land disposal. Much of this impact will be positive providing clear and comsistent criteria for the safe disposal of such wastes and thereby encouraging the optimum utilization of radiation for the benefit of mankind, as a result of resolving the land disposal issue. Proper standards for licensing land disposal facilities are urgently needed to permit additional sites to be developed and offered to generators.

The Eastch Physics Society Counities on State and Tederal Legislation has not specifically addressed the process through which future land disposal facility applicants will have to go to achieve a License. We note briefly, with concern, that the opportunities for hearings, through Atomic Safety and Licensing Boards, with further appeals, etc., may result in excessive delays in achieving new sites unless issues that might be contended are limited to those identified in the proposed rule and discussed in the accompanying draft EIS, including the

## NRC from Meinhold

-6-

January 18, 1982

- M-1 rules and the EIS would be that a major long-term public and occupational health problem is created by the operation of land disposal facilities for low level radioactive wastes. Ence, the NEC has proposed rules which, according to the Federal Register notice, are compatible with, and possibly duplicative of, existing Federal regulations, but with an increasing intensity and specificity to a perceived unique problem. Not only are rules proposed for the disposal facility, but more and different NEC rules are proposed for the packaging and labeling of each container; presumably on the argument that if each container meets the classification requirements, the sum of all containers (the disposal facility contents) will meet health and safety criteries for at least 150 years.
- D.56-8 The scope of the new rules imply a regulatory deficiency that requires a program of 150-year stability, to be cartified in advance as being able to meet health physics objectives over the future. We believe the need for land disposal facilities is imperative and that the new proposed rules should reflect a less deficient and less desparate current practice.

Sincerely yours,

Charles B. Meinhold President



Don't Waste Washington Legal Defense Foundation Suite 654 Colman Building

S11 First Avenue Seattle, Washington 98104

- 723 J.112 P1543

January 12, 1982

Secretary of the Commission U.S. Nuclear Regulatory Commission Washington, D.C. 20555



Attn: Docksting and Service Branch

Re: Proposed Rule 10 CFR Part 61, Licensing Requirements of Land Disposal of Radioactive Waste, 46 <u>Fed. Reg</u>. 38081 (July 24, 1981)

Dear Mr. Secretary:

This comment is submitted on behalf of the Don't Waste Washington Legal Defense Foundation.

The Don't Waste Washington Legal Defense Foundation is a Washington not-for-profit corporation concerned generally with safe radioactive waste disposal. We are committed to educating the public regarding the need for development and implementation of safe radioactive waste disposal technologies and systems, and to participating in decisionmaking which promotes the safe and orderly transportation and disposal of radioactive waste.

Don't Waste Washington shares the concerns of the Commission regarding the need for careful monitoring and control of radioactive waste disposal in order to protect the public health, safety, and welfare. We velocine the Commission's recognition that radioactive waste materials must be isolated and monitored for long periods of time. We therefore believe that language reflecting this recognition should be included in the final rule. For example, \$61.27(a) states, in part: "Failure to renew the license in no way relieves the licensee of responsibility for carrying out site closure, postclosure observation and transfar of the license to the site owner." The principle of long term fixed responsibility embodied in this language is critical to the protection of health and to the minimization of danger to life and property required by the Atomic Energy Act.

- Second, we agree with the proposed regulation's statement D-55-2 that there should be no generic "de minimis" category for waste. Third, we support the Commission's position that adequate financial assurances are essential for all aspects of waste disposal-construction, operation, site closure, E-1 postclosure observation and maintenance, and institutional
  - (control. Such strong assurances are necessary to discharge

Acknow 2000 1/22/82 and

B-5

our responsibility to future generations, and to assure that the public will not bear costs which should be borne by the users of the facility. Self-insurance does not satisfy the surety requirements for the reason stated in §61.62(g). We agree that the amount of surety liability should change with changes in cost estimates. This amount should be E- ) subject to increase at the time of license renewal (if not annually), especially because the purpose of reviewing the license periodically is economic rather than technologic. Subpart E should be amended to state more clearly that the Commission will review the applicant/licensee's financial assurances periodically, and that the burden of proving adequate financial qualification is on the applicant/ SUBPART A F-1 licensee. 61.7(c)(1) We have some question about the proposed rule's justification for relying primarily on performance objectives in an area where the concern for public health and safety is so great. It is not sufficiently clear to us that the technology is not sufficiently advanced to warrant use of more prescrip-C - Iformance objectives are to be used, the rule should empha-61.7(c)(2) size and confirm that the standards of the Atomic Energy Act are applicable, and that the burden is upon the applicant/licensee to show that it has net these standards of protecting health and minimizing danger to life and property. SUBPART B We believe that \$61.25 must be rewritten in such a way 61.23 that the right to notice and opportunity to be heard is Letter protected. As this section stands now, it is not clear that the public, a state; or Indian tribe will have input into any decision on amendments in license conditions other than those specifically listed in subsection (b) 61.24(a) This may become important where, for example, the new conditions contained in an amended or renewed license are

January 12, 1982 Page 2

Don't Waste Washington also wishes to comment on the issue of state authority and state participation in the procedures for license application, amendment, renewal, and termination. Don't Waste Washington believes that the states have and will continue to assert strong leadership roles in management of radioactive waste. Congress has concurred in this judgment in the Low-Level Radioactive

inconsistent, with those imposed by a state lessor in the

Secretary of the Commission

lease itself.

Waste Policy Act.

F-1

At the outset, at least with respect to state cwnership of the land, it appears that the state has a sufficient degree of control because it can make the initial determination whether to take ownership of the land selected by the applicant for a waste disposal site, and on what terms to · · ·

· . . .

• • . •

Secretary of the Commission

14. <u>1</u>. 14. 14

January 12, 1982 Page 3

. . . \_::

<del>\_\_</del>;

: 22 273

- 27

lease the facility. The state must not subsequently lose this control because of federal policies that restrict state participation during operation of the facility. We believe that, with respect to state-owned land, a license to construct; and operate a waste disposal site must not be granted, amended, renewed, or terminated without the explicit concurrence of that state. The applicant/licensee must comply with all applicable state laws that are consistent with the requirements of the Commission.

We suggest the following specific changes:

- Change the word "may" to "shall" in the application is in accordance with established administrative procedures and may involve participation by affected State covernments or Indian tribes."
- Add to the end of the first sentence an and regulations that are consistent with Commission regulations."
  - Add a subsection which says, "Where the land is owned by a state, the applicant has met the state's requirements.'
- Add to the end of the sentence, "and, if the land is owned by a state, the state has Given its concurrence."
- Add to the end of the sentence, "and, if 51.24(c) the land is owned by a state, the state requirements have been met.
- 61.24(d) Add to the end of the first sentence, "and if the land is owned by a state, to all state laws and regulations consistent with Commission regulations."
- Add to the end of the sentence, "or any state law or regulation consistent with Commission 61.24(e) regulations, where the land is owned by a state.

Secretary of the Commission January 12, 1982 61.24 (h) Add after the words "additional requirements," an explicit statement, "including state requirements, where the land is owned by a state." F-1 61.28(a) Add before the second sentence, "Where the land is owned by a state, this closure application must include a certification from the state that the applicant has complied Mr. Samuel J. Chilk with all state requirements. SUBPART E Washington, D.C. 20555 61.61 Add a sentence, "Where the land is owned by a state, this showing shall be supplemental Cear Mr. Chilk: to any assurances required by the state. SUBPART F The wording of this Subpart should be brought into conformity with its counterpart, Subpart C in Final Rule 10 CFR Part 60, Disposal of High-Level Radioactive Wastes in Geo-Logic Repositories: Licensing Procedures, 46 Fed. Reg. 13971 (February 25, 1981). Primarily, the use of the word "may" must be replaced by the word "shall." F-1 61.71 Second line--change "may" to "shall" Fifth line from the end--change "will" to "shall" Third line--change "will" to "shall" 61.73(a) Second line--chance "may" to "shall" 61.73(b)

•

Thank you for the opportunity to comment on the Proposed Rule 10 CFR Part 61. Con't Waste Washington appreciates the opportunity to participate in this rulemaking proceeding, and is prepared to contribute further views or any other assistance that the Commission may require. Please send us notice of any further proceedings, including further comment periods and promulgation of a final rule.

Very truly yours, Lydia Gold by mult Lydia Gold Multure Michael W. Gendler

LG, MWG/mm

WISCONSIN Electric POWER COMPANY 122 11 11 11 23 231 W. MICHIGAN, P.O. BOX 2046, WILWAUKEE, WI 52201 emp January 11, 1982 -2eta Secretary of the Commission U.S. NUCLEAR REGULATORY COMMISSION

#### PROPOSED RULEMAKING ON LAND DISPOSAL OF LOW-LEVEL RADIOACTIVE WASTE

Since the comment period for the proposed rulemaking on land discosal of low-level radicactive waste has been extended, we employ this coportunity to supplement our earlier comments which were contained in our letter to you of October 21, 1981.

There are two additional problems with Table 1 in 10CFR61.55.

 $D-55-8 \begin{cases} 1. & \text{It may be necessary to dilute wastes in order to comply} \\ \text{with the proposed concentration units. This is undesirable,} \\ \text{and it would be contrary to NAC's policy on volume reduction} \\ \text{as published in 45F351L00 on October 15, 1581. This problem} \\ \text{would be most severe with transuranic waste; the 10 n Ci/g} \\ \text{limit should be increased to a level commensurate with the} \\ \text{actual risks.} \end{cases}$ 

D-55-2 There is no provision for the establishment of minimum concentrations of nuclides in waste below which the waste is acceptable for disposal through ordinary means such as sanitary land fills.

The additional radiation dose to personnel which would result from the required additional sampling and analysis of radwaste should be considered in the context of ALAA. This actual dose to personnel would be difficult to justify when compared with the small hypothetical dose calculated to result in the environs of a disposal site. In addition, there will be additional costs of labor and equipment associated with the additional ( sampling and analysis requirements. Certain types of analyses are difficult

-1-

1/22/82 emp

## Mr. Samuel J. Chilk

C. M. Fay

.

January 11, 1982

D-55-6 by nature. Nuclides which decay by beta emission and electron capture such as Nickel-59 and Iodine-129 are in this category. Time consuming chemical separations will be required as part of these analyses. This will not be cost-effective, especially for the small licensee.

Finally, we recommend the elimination of reporting requirements for nuclides which constitute insignificant fractions of radwaste packages.

-2-

Yery truly you<del>rs</del>,

Assistant Vice President

:

STATE OF NEW YORK Department of Law Alsany, N.Y. 1224 S13-474-4318

January 14, 1982

US Nuclear Regulatory Commission Secretary's office Washington, D.C. 20555

To Whom It Concerns:

ATTERNET GENERAL

I have enclosed two copies of the Attorney General's Technical Comments on the proposed Part 61 regulations for management of low level radioactive waste. We appreciate the opportunity to respond to your proposals and look forward to any response from you in the future.

2.1.1.25

Encl.

Environmental Enginee

· · ·
#### 

#### TECHNICAL CONTINUES OF THE NEW YORK STATE ATTORNEY GENERAL'S OFFICE 12 12 11 27 ON THE PROPOSED 10 CFR PART 61 -LICENSING REQUIREMENTS FOR LAND DISPOSAL OF RADIOACTIVE WASTE

.

#### January 14, 1982

#### Introduction

The Attorney General's office has been actively involved with nuclear waste management for over a decade, participating in radioactive waste management policy making from uranium tailings to high level waste repositories. Members of and experts for the office have testified in numerous proceedings before Congress, and federal and state agencies. The office has maintained close scrutiny of the ongoing efforts to cleanup radioactive waste accumulations at West Valley, Model City, and elsewhere in the state. The office has also participated in numerous hazardous waste disposal cleanups and landfill design and siting efforts. It is on the basis of this experience we make the following comments and recommendations to the CS Nuclear Regulatory Commission on their proposed regulations for 10 CFR Part 61 to better control the land disposal of radioactive wasta.

D-55-8 Our comments should not be taken as an endorsement of land burial of radicactive wastes in general. Rather, we believe that source reduction, precompaction and numerous other tachnologies exist to reduce the need for reliance on land disposal of wastes as it is practiced today or as it will be practiced under Part 61 when promulgated. In addition, we believe the USNRC should give careful consideration to emerging techniques of waste management such as encapsulation and above ground dry storage similar to methods employed at the Idaho Engineering Laboratory to isolate transuranic wastes. These and other approaches to waste management, when fully exercised by wasts generators, may significantly reduce the need for burial of waste; especially those wastes which play a large role in the deterioration of the land burial facilities.

#### History of Land Disposal of Waste

Land disposal of waste has been practised for millenia. Archeologists rejoice when they come upon dumps of previous societies because therein lies artifacts which greatly help to describe the way of life of earlier peoples. Artifacts of bygone societies, however, are very different from the wastes of today. Few wastes were toxic and certainly none were particularly radioactive. Performance objectives of such facilities did not reflect concern that intruders be kept out or infiltration be limited. Earlier sociaties merely practised the objective of "out of sight - out of mind."

Modern day performance objectives informally evolved from this objective into an objective of waste isolation. Facility engineering changed little, but the waste spectra certainly did. Hazardous chemical wastes greew in volume, toxicity and persistence and a new group of wastes, nuclear wastes, began to appear. Although gigantic amounts of scientific effort went into creating these exotic wastes, little or no effort went into waste site suitability considerations or facility engineering. The result of this lack of vision has been such notable anvironmental disasters as Love Canal in New York and Montague in Michigan. Although radioactive waste dumps cannot be considered environmental disasters yet, some of them have certainly not succeeded in isolating wastes from the biosphere. The waste facilities of West Valley, Maxey Flats, Sheffield are good examples of such troubled sites.

These three sites have failed to completely isolate the waste materials dumped for a variety of reasons. Chief among them are soil cover erosion and deterioration. Once deteriorated, these covers fail to prevent infiltration of rainfall, runon and snowmelt. Infiltration, in turn, causes leachate to percolate the wastes and mobilize contaminants and generates leachate, often in millions of gallons in each disposal unit.

Since past performance objectives required sites with tight soil to preclude groundwater migration, leachate would accumulate there, only to later migrate in transmissive horizons or overflow the boundaries of the disposal units into surface drainage systems. Site operators and responsible governmental units have been forced to deal with these leachate accumulations by very expensive pumping and treatment efforts. Expensive studies have been undertaken by scientific teams after the fact determine whether the site was appropriate in the first place.

Other studies have focussed on the post-closure performance of the disposal unit itself to determine the dynamics of internal consolidation and soil cover deterioration. These studies have conclusively shown that stabilization of low level waste disposal units is a long one. Materials inside degrade, containers rist away, losing their structural strength, and consolidation goes forward inexorably to some ultimate, but unpredictable point. This inevitable process of internal consolidation leads inescapably to a loss of structural support for the soil cover, often to different degrees in different parts of the disposal unit. This loss of support then leads to differential subsidence of the soil cover, cracks and holes in the cover and finally infiltration of water.

3.

NUREG/CR-2101, "Evaluation of Trench Subsidence and Stabilizaton at Sheffield Low-Level Radicactive Waste Disposal Facility." discusses these phenomena in detail and concludes:

"All trenches have a potential for some future subsidence due to piping of soil, natural soil consolidation, and waste container deterioration. The location, sizes and extent of this subsidence is quantitatively indeterminant due to the unknown void sizes and their numbers and locations within the trenches.....Increased incidences of subsidence may 000117

from 10 to 14 years after waste disposal when metal drums, cardboard, and wood containers will have experienced some deterioration."

This report and others like it for other similar facilities do not give one much confidence that the environment or the finances of site owners have been well protected by this type of waste disposal system. It is within this framework that we have considered the USNRC proposed Part 61 rules and regulations.

Subpart C...Perfomance Objectives

· · · · · · ·

••• •••

un (4234

Subpart C is the heart of the proposed Part 61 regulations. All other subparts measure achievement in terms of the objectives listed in Subpart C. Consequently, we have considered that part of the regulations first.

From review of Subpart C, it appears that the USNRC has made an in-depth review of the problems of low level waste disposal. We are impressed with the innovative approach to discosal . regulation through protection of various populations at risk; general populations, intruders, and employees. We particularly support the notion of site stability as a criterion for successful disposal. If the site is stable over long periods of time, the likelihood of excessive long term maintenance costs is significantly lower. As history has shown, site owners (States in this case) have been and probably will be saddled with unrecoverable multi-million dollar bills each decade to maintain facility integrity. ÷ ,

We are, however, disappointed by the limits proposed by the USNRC for the protection of the general population. Subpart C in § 61.41 proposes that the facility will be operating

satisfactorily as long as the population outside the site boundary is receiving no more than 25 mrem whole body dose, 75 mrem thyroid dose, and 25 mrem to any other organ. In addition, the groundwater at the nearest public water supply must meet the National Primary Drinking Water Standards. As long as the effluents from the facility contaminate the environment such that the groundwater does not get too contaminated and the. biologic pathways to man do not transmit doses in excess of the limits above, the facility is operating or has operated in compliance with the regulations. In short, the facility can leak as long as the leaks do not contaminate the area too much.

4.

4

47

-14

This regulatory approach tries to provide reasonable population protection through inherently uncertain health based risk assessment. In actuality, however, use of this approach for land burial is tantamount to an admission that burial of low level radioactive wastes in the forms required and at the sites found suitable by Subpart D of the regulations cannot achieve long term isolation of radioactivity.

We believe this kind of a regulatory approach is unacceptable for protection of the general public. First, this approach relies on a conceptual framework similar to the EPA's water quality considerations for NPDES permit limitations based on a receiving water body's assimilitive capacity. Permit limitations based on assimilative capacity are very difficult to set and impossible to verify as adequate for environmental protection. Instead, notions of "best available technology," "best practicable technology," and other technology and cost based standards for permit limitations have been developed. Considering the difficulty of monitoring and determining assimilative capacity predictions unique to groundwater compared to similar efforts for surface waters, use of the concepts embodied in \$ 61.41 seem most ill-advised.

Permitted radioactivity releases from other nuclearer facilities are not based on assimilitive capacity. Rather their limits are set by USNRC regulations which are not site specific. 1-5 In addition to the maxima set forth in these regulations, the notion of "As Low As Reasonably Achievable" (ALARA) requires extra efforts by licensees to reduce releases and personnel exposure even lower where reasonable. The proposed limits under 5 61.41 are not consistent with this existing defensible and workable industry guideline.

Another basic goal mandated by Congress is the notion of

and and a second se

C-3 ("zero discharge." A well known goal of the Clean Water Act of 1972 is that discharges of wastes into navigable waters be

C-3

reduced to zero by 1985. Although this goal has engendered a. good deal of controversy and may not be reasonably achievable by certain industries and old manufacturing facilities, there is little disagreement that the goal itself is appropriate and generally achievable. 5 61.41 flies directly in the face of such a goal. Rather than basing performance objectives on "zero discharge," this section proposes instead as much discharge as the environment can handle as long as the population does not get overdosed: "

5.

5 - 5

D-50-2

"the term "reasonably achievable" provides basic guidance for a variety of environmental protection regulations. Reliance on this approach for low level waste disposal seems appropriate. As other sections of our comments point out, affordable engineering systems exist today to assure that environmental releases from disposal sites can be reduced to de-minimus levels. Certainly, then, application of ALARA notions to LLW disposal is defensible from a regulatory and cost point of view. In addition, we are confident that application of ALARA concepts will easily justify the application of "zero discharge" limitations to ground and surface waters.

On the basis of the foregoing arguments, we recommend that the language of § 61.41 be amended so that "zero discharge" be the performance objective for ground and surface waters in perpetuity and that ALARA guide all site activites, including Subpart C sections 61.41 through 61.44. The proposed population doses and maximum contaminant limitations may remain as absolute maxima. Releases of radicactivity which cause such doses or maximum contaminant levels, however, should certainly not be considered satisfactory facility performance under any the second circumstances. As such, even the totally unacceptable performance of the West Valley, Sheffield and Maxey Flats sites would be satisfactory under at least \$ 61.41.

الإسروا الجم ويدار عبيها Subpart D .... Technical Requirements . . . .

Subpart D sets forth a number of technical requirements for land disposal facilities which will supposedly assure that Subpart C performance objectives are met, by the facility. This section features numerous defensible and innovative regulatory ... approaches to facility integrity and safety. . Some of the proposed rules, however, should be changed to assure that discharges from the site will be zero or at least as low as reasonably acheivable, performance objectives we have proposed in cur comments above.

Site suitability requirements

6-3

<u>(-></u>

A second state is the second state of the seco n na serie de la companya de la comp Nome de la companya d Nome de la companya d

§ 61.50 of Subpart D sets forth some basic site suitability requirements no one can argue with. Unfortunately, these requirements do not require siting the facility in a soil environment which will preclude or even reduce leachate migration. The only subsection dealing with leachate even tangentially is \$ 61.50(a) 7 which requires that the water table be sufficiently below the site to preclude groundwater intrusion. This section, as written, implies that if groundwater is kept out, no liquid will enter the fill and the previous impermeable soil requirements are not necessary to protect the public. · . . .

6.

Unfortunately, groundwater has not been the only way copious 1 . . . quantities of leachate have been generated in past landfills. Rather, surface infiltration through subsidence cracks and holes has caused much of the leachate generation. In addition, liquids in wastes, rainwater and snowmelt entering the trenches during . operations and internal decomposition add to the volume of wastewater which either accumulates or migrates from the site. Because of the high likelihood that substantial quantities of leachate will be generated by future facilities, site suitability criteria must address this issue, not avoid it.

Some professionals have argued that a leaky site is appropriate. They point to the lack of leachate at the Barnwell facility as an example of good performance. Any liquids which enter the trenches through the subsidence and shrinkage cracks and other entry points drains right cut into the porous soil of the site. Voila'.... no problem. If this stratzgem may ever, in fact, be effective, it is only by accident and as long as the soil beneath the facility can attenuate the radionuclides sufficiently to protect the environment. Luck or wishful, thinking are not appropriate for sound management stratagems.

Under the proposed regulations, in order to assure that the soil" can perform its required function in perpetuity, it must be shown to possess appropriate radionuclide attenuation characteristics such that eventual doses to humans are acceptable under 5 61.41. This showing can be easily made, based on alluring scientific experiments and projections ... Unfortunately, these experiments and projections are still crude indeed and their predictive value very limited. It is impossible to verify these predictions with real life conditions because these ... experiments and calculations must be based on leachate characteristics and waste spectra which are, at any time, nearly impossible to predict. Consequently, although such showings are easily made, their validity is doubtful indeed and would provide regulators and the public a false sense of security about the long term integrity of the facility. , .

• . •

•. • • •

7.

D-50-2]

65N-2-1

D-52-1

Optimistic "scientific" showings have formed the bases for both radioactive and hazardous waste dump licenses in the past. States are now suffering the inevitable consequences of the lack of validity of these showings. Therefore, basing new licensing regulations on scientific demonstrations of questionable predictive capability is unacceptable.

We recommend, instead, that the USNRC develop site suitability requirements vis a' vis soil horizon characteristics based on the assumption that substantial quantities of leachate will enter or be generated by the waste mass at one or more times during the hazardous life of the waste. Perhaps the appropriate regulatory framework should specify porous soils. Or perhaps, tight soils with adequate leachate collection and treatment facilities are the answer. Whatever the case, failure to specify soil horizon characteristics fails to deal with the very real problems of leachate generation experienced by almost all "secure" landfills in humid climates.

#### Facility Design

D-57-1 S 61.51 discusses minimum engineering features of a land disposal facility. This section fails as does the previous section, to squarely deal with the difficult problem of substantial leachate generation. Although subsections deal with engineering features which may reduce the likelihood of external infiltration, they certainly do not eliminate it. In some ways, this section provides little more guidance than has been applied to the closed landfills for radicactive waste which have failed to perform well. Thought should be given to specific guidance for engineering features which effectively respond to leachate management, subsidence control, and aqueous and aeolian erosion.

:

••

Facility Operation

5 61.52 discusses general facility operational guidelines. These rules also demonstrate USNRC recognition of the problems encountered at waste sites in the past. Many of the requirements here will lead to significant improvement in environmental protection. Subsections (a) 1, 2 and 3 preclude "interaction" of radioactive wastes of Classes A, B, and C. In so doing, USNRC seems to be making an effort to achieve a consistent level of environmental and intruder protection for different danger level wastes. The rules, however, do not define the term "interaction" and are therefore vague and difficult to enforce.

Environmental segregation of waste of different classes of danger does make good sense. If the segregation and the pretreatment of the wastes assure that the risk potential is equal among the classes and also an acceptable level, then the management scheme is defensible. We have problems, however, with the level of pretreatment accorded Class A waste. According to personal discussions on several occasions with USNRC staff, Class A waste comprises a whopping 60% of the volume of all low level wastes covered by the proposed Part 61 regulations. This waste will be dumped in the trenches without pretreatment for stability or leachability reduction as detailed in 5 61.56. As such, Class A waste will be managed in much the same way as it has been for

**.**...

غدف فالتشابية والمعاقد

8.

leachate production experience of the past will not recur in the future at Class  $\lambda$  waste dumps.

Most of the low level wastes which are compressible and degradable fall within the limits of low activity Class A waste. These characteristics lead inexorably to collapse of the cover over the trench. Collapse of the cover (differential subsidence and holes) leads inexorably to surface water infiltration and leachate generation. These physical problems require institutional responses for engineering and construction which end up costing taxpayers millions of dollars. We have no reason to believe Class A waste will behave in any different way underground than it has in the past. Therefore, we conclude that tomorrow's Class A waste dumps licensed under 10 CFR Part 61 as proposed will suffer the same performance difficulties their predecessors have with similar financial impacts on their owners.

We must not forget, however, that the relaxed performance objectives under § 61.41 will still be met by a dump whose covers fail just as long as the population does not get overdosed and the groundwate? at the nearest public well does not get too contaminated. Although these objectives may be satisfactory to the USNRC, it is unlikely that a Class A landfill whose covers admit copious quantities of water which forms a large radioactive plume will be satisfactory to local authorities or the citizenry. These governmental units and people will generate significant political pressure to close such a facility. Then as now, when half the nation's wate sites are closed due to performance difficulties, adequate space will not be available for the disposal of wastes. Such a dump crunch will be particularly troublescme when compacted facilities have moved to limit dumping by out of compact generators.

Subsidence, leachate generation, and costly repair and maintenance requirements will become important questions in

federal and state licensing proceedings. As such, they will become rallying points for those opposed to nuclear power and waste management in particular localities. They form a reasonable hasis for denial of licenses and revocation of licenses once operations have commenced. They will inevitably exacerbate the same disposal availability problems from which we now suffer. Consequently, we believe that Part 61 rules should be written to preclude the recurrence of these phenomena.

Precluding subsidence and its inevitable ramifications is relatively easy and cost effective when long term maintenance and repair costs are included in the equation. Higher activity Class B and C vastes will be stabilized in such a way under 5 61.56 that they will "...not degrade and promote slumping, collapse, or other failure of the disposal unit and thereby lead to water infiltration." The encapsulation matrix or container must withstand a compressive load of at least 50 psi. In this way, these Class B and C wastes will not be responsible for disposal unit cover collapse. If 40% of the mation's low level wastes can already be affordably pretreated in this fashion, then we submit that the remaining portion of the wastes should receive a similar degree of pretreatment.

USNRC has argued that because Class A waste decays radioactively within the period of institutional control, the inevitable subsidence and leachate phenomena associated with Class A waste dumps do not present unacceptable risks to the public. This may be true if the site successfully attenuates the plume of radioactive leachate and if the radionuclides are indeed short lived. Both assumptions are very doubtful, however, given experience at all the low level waste dumps in the U.S. to date. In any case, the bad press and public criticism such dumps will engender if and when a radioactive pluma is discovered moving away from the site will undo all public relations dividends of the carefully stabilized Class B and C waste dumps.

The USNRC should recognize these very raal siting and operational considerations and reconsider pretreatment for Class A waste. Although more costly in the short run, stabilization pretreatment will yield substantial benefits during and after the 100 year institutional control period, both in terms of environmental protection and in terms of public acceptability.

#### Environmental Monitoring

D-52-2

 $\mathcal{D}$ -53-1  $\left( \begin{array}{c} 5 & 61.53 \\ \text{environmental monitoring.} \end{array} \right)$  This subsection is very vague compared

to the lavel of specificity which could be justified. After only a decade of operation, the existing low level waste sites have been found to require substantial exploration for remedial purposes. USGS, USEPA, and numerous other state and federal agencies have spent millions of dollars in the last few years developing monitoring nets around these facilities and evaluating the data these nets have generated. These efforts should provide USNRC with a great deal of guidance to specify particular types of and sensitivity for monitoring nets. All the proposed of providing early warning of rigration of radionuclides from the disposal site." This description provides no spatial or time specifications and thereby allows licensees to determine specifications which are in their best interest; namely far field and long time period monitoring systems.

Such insensitive monitoring systems are not in the best interests of either environmental protection or the finances of the site owner. We, therefore, recommend that the USNRC rewrite Part 61 regulations to reflect appropropriate spatial and temporal specifications. These should include several monitoring wells both upgradient and downgradiant, screened at various depths in both the saturated and unsaturated zones to permit stratified sampling if desired. Such a well system should include wells below as well as beside the waste mass. Similar monitoring systems should be emplaced for gaseous emissions as well. Surface runoff and acolian particulates should receive monitoring scutiny as well.

Groundwater monitoring systems remain the most important part of the environmental protection scheme for the facility. Although such systems may provide effective early warning systems, engineering design requirements may be acceptable as well. One approach used in the hazardous waste landfill field is the "telltale" layer beneath the waste mass. This confined porous layer conducts early migratory liquids to a point where samples can be obtained. These samples will provide an accurate representation of the source term for environmental impact assessments. In addition, it can trigger early mitigation efforts and other institutional responses. We recommend that USNRC consider the appropriateness of "telltale" layers and rewrite § 61.51 to reflect the availability of this effective and affordable environmental protection measure.

Subpart D, Financial Assurances

-) { 55 61.62 and 61.63 discuss minimum financial arrangements needed to obtain low level waste licenses. These sections go

9.

D-53-1

. . .

11.

• •

. . .

- 19

.\*

新田田

- >

12.

5-1

B-1

much further than earlier regulations to protect the financial interests of site owners. Although innovative and comprehensive, we remain concerned that the post-closure maintenance, leachate collection and treatment, cover repair and other likely costs will be underestimated by applicants and accepted by both site owners and the USNRC due to the pressing need for waste disposal site availability. In the long run, as today, it will be state taxpayers, rather than waste generators, who will eventually bear the excessive costs of site maintenance and repair. We recommend that the USNRC study carefully the likelihood and costs of major facility repair and maintenance generically by climatological region prior to the consideration of any application for waste site construction. Such a study will highlight the high financial risk associated with burial of wastes such as contemplated for Class A and help site owners and regulators determine adequate lease arrangements prior to the licensing proceeding.

Subparts E and B both discuss various applicant qualifications necessary for granting a license. We were surprised to find no discussion of the past experience and professional qualifications of applicants. We feel applicants which have had a bad track record managing their enterprises in the past represent poor choices for new facilities. We believe, however, that problems of the past may be proloque to the future and should be an important part of the application review Tocess. ... Comment Summary 2012 and a Comment Summary 2012 and and a second seco

## The USNRC proposed regulations for Part 61 represent a solid step in the right direction. We do feel, however, that the performance objectives and several other parts of the regulations fall short of the intention of the USINC "...to assure the protection of the public health and safety." In order to achieve

this goal, burial facility licensees must be able to conclusively . prove that releases of waste will not exceed certain limits. We feel these proposed limits are far too lax and fail to recognize that burial sites fail due to waste consolidation and then generate copious quantities of contaminated leachate.

We also believe that Class A waste deserves adequate pretreatment to preclude consolidation. Such pretreatment would limit the physical, financial and public criticism ramifications associated with subsidence and leachate generation and/or accumulation. In fact, we believe that the USWRC should consider the effect of the regulations on facility siting. More stringent requirements for Class A waste pretreatment and zero discharge " " ... Nº Sal

> 1999 A. 17 A. 7 C. 13196

will better convince the public and regional governmental units that the site will be a good neighbor.

A set and a set and a set and a set and for the set and a set and set and set and set and a set and a set and set and set and set and set and set and a set and a set and set and set and set and set and set and a set and a set and set and set and set and set and set and a set and a set and set and set and set and set and set and a set and a set and set and set and set and set and set and a set and a set and set and set and set and set and set and a set and s

// A contract of the second seco

والمحاج المراجع والمراجع و

B-363

•••••••

· . . .

Atomis Industrial Forum, Inc. 7101 Wisconso Avenus Weshington, D.C. 20014 Tisphona: 13011 854-9280 TWX 7108249602 ATCMIC FOR DC	<u>ני אניני</u>	
January 18, 1982		Comments on the Proposed NRC Rulemaking: "Licensing Requirements for Land Disposal of Radioactive Waste", 10 CFR Part 61, and on the "Draft Environmental Impact Statement", NUREG 0782 prepared by
	(46 FR 38081)	The Air Subcommittee on Low-Level Radioactive Waste and the AIF Subcommittee on Solidification of Low Level Reactor Radwaste
Secretary of the Commission U.S. Nuclear Regulatory Commission Mashington, D. C. 20555		Comments on "Supplementary Information":
Attention: Docketing and Service Branch	(46FR 51776)	p. 38084 (The first paragraph in this section under "Disposal Para V,C (Site Design, Land Disposal Facility Operation and Dispos-
Re: Notice on Proposed Rulemaking "Licen: for Land Disposal of Radioactive Wast 61 (46 Federal Register 38081) and "I mental Impact Statement," NUREG-0782	sing Requirements te" 10 CFR Part Draft Environ-	ED- K site Closure Requirements" calls for operations and design which will result in the elimination of ongoing active maintenance after closure, requiring only minor custodial care. "Active maintenance" is not clearly de- fined here or even in the definition in Para. 61.2 "Defi- nitions". In the definition, active maintenance is des-
Dear Sir:		cribed as a "significant remedial" action. It is sug- gested that the postclosure maintenance requirement
The enclosed comments on the proposed rul. EIS were prepared by two AIF subcommittee: radioactive waste.	e and the draft s on low level'	should be size specific and incorporated in the license conditions
The consensus of the subcommittees is that rule appears reasonable; however, some su some exceptions to the proposed rule and were made and are reflected in the enclose	t the proposed ggestions and to the draft BIS ures.	p. 18084 (The first paragraph under "Waste Characteristics and Para V,C (Classification" points out the desirability of the physi- D-55-12 (cal integrity of the waste and the site lasting until D-56-8 (cal integrity of the waste and the site an
John H. MacM Chairpan, AI	HAL illan F	p. 38085 In the section "Naste Characteristics and Classifica- Para V,C tion", there is a discussion of a "de minimus" clas- sification for wastes which would be exempt from 10 CFR Part 61. It is noted that the NRC in the next two years will work to define these wastes and "to provide for additional ware exemptions as empropriate " Ma
Fuel Cycle	cue nuclear	support the need for a "de minimus" concept and encourage the expeditious astablishment of suitable criteria for
JHM:gpg Enclosures		D-55-2 this concept. A "de minimus" classification would result in the conservation of valuable disposal site burial space while at the same time protecting the health and safety of the public. In this connection, the Commission is urged to permit case by case reviews of requests for specific applications of the "de minimus" concept during the period criteria are being developed. We will be pleased to be of assistance in the establishment of "de
ا_ دار		minimus" levels and would like to call attention to the

.

.

.

-2-

(report by the Forum's National Environmental Studies Pro-ject entitled "De Minimus Concentration of Radionuclides in Solid Wastes", AIF/NESP-016. Also, the Utility Nu-clear Waste Management Group sponsored a study, "Sug-gested Concentration Limits for Shallow Land Burial of Radionuclides", which should be of value.

- p. 38086 [In reference to the "Manifest Tracking System" we urge Para F M-1 [Angle Angle Angle

The five phases of the life-cycle for a typical land p. 38086 Para G disposal facility discussed in the proposed rule are: preoperational, operational, closure, post-closure obser-vation, and institutional control. Some discussion is needed to indicate that several of these phases may proceed simultaneously at the same site as part of the normal disposal site operation. B-3

There should be a sixth phase in the life cycle for a land disposal facility identified in the regulation; namely - release for uncontrolled use. This phase, which occurs after the radioactive contents of the landfill have decayed, should be stressed. It is further imporcal forms out of landfills intended for radioactive waste. Chemical wastes are present forever and the land used must be restricted forever. Radioactivity decays away in time and land use can be recovered.

p. 38087 (The proposed rule requires a license renewal every five Para G (years. For a facility of this type, a five-year license

years. For a facility of this type, a five-year license renewal policy appears unreasonably short. Disposal sites should be provided with a full term license, subject to appropriate review. The fiscal basis for site B-5 operation and monitoring assumes a reasonable operating life. Licensing similar to that under 10 CFR 50 would appear to be more appropriate.

(The proposed rule states that the Institutional Control p. 38087 Para G Board has a responsibility to "keep people off the site". This approach may be unnecessarily restrictive.

- D-59-2 Limited use of the land may be desirable in the future. A more appropriate action might be to control access to the site. The control board should have the flexibility to institute suitable control options depending on the particular condition existing at the site.

#### Comments on Part 61:

p. 38089 (See our comments on Para. V.C. p. 38084 on "active main-Para 61.2 \tenance".

-3-

Para 01.2 (cenance -ED-) p. 38090 In the definition of "Disposal" isolation of radioactive Para 61.2 wastes from the biosphere should be clarified. A better definition would be: "Confinement of the wastes with no

- p. 38091 Para 61.7 The primary safety objectives for a near-surface disposal facility should be redefined in a manner that will (a) (b)(1) keep the site personnel dose as low as reasonably achiev- (1) Reep the site personnel dose is low is reasonably inlev-able and (b) keep the environmental impact and population dose below specified limits. In the existing statement of objectives the word "prevention" should be replaced
   D-57-1 with "minimize". To minimize migration of radionucludes is to provide a means towards achieving the primary.

  - GEN-1 Safety objectives. "Prevention of exposure to "inadver-tent intruders" is a special case of (b) above, and should be regarded as a secondary objective.

p. 38091 In the first sentence the word "eliminated" should be para 61.7 modified or omitted. The requirement may not be possible. (b)(2)

.1

7 . 5

B-1 B-1 Para 61.13 or how analysis will be accomplished. This section (b) should be clarified. This comment applies to Paras. 61.23(f),(i) and (j).

p. 33094 The section on "changes" is too restrictive. The Para 61.25 licensee should be allowed to make changes when deemed necessary, providing they do not decrease the level of protection to the public and provided they are brought to the attention of the Commission in a timely manner (i.e., Subpara.(d) provisions for changes similar to those in 10 GFR 70.32 (d)). 8-4

P. 38095 The closure period should be included in the closure Para 61.29 plan rather than the regulation. The required period of observation should be a function of specific site charac-teristics.

B-365

.

(Blank Page)

-4-

p. 38095 The requirement for transfer of the license to Federal Para 61.30 or State custodial care should be part of the closure agreement and not the regulation. Transfer should be contingent upon licensee satisfaction of required conditions. Transfer "when the agency is prepared" leaves an open-ended commitment by the licensee which (5) R-8 is not warranted.

p. 38096 Para 61.50 (a)(3) D-50-6 Projections of population growth should be limited to useful demographics. Projection to 100 years or even 50 could be a useless exercise or worse, could rule out an acceptable site.

p. 38096 p.y., Replace "prevent" with "minimize". Para 61.51 cm.) (2)(4)

p. 38096 Ges. Replace "no interaction" with "no significant inter-Para 61.52 ? action". (a)(1)

p. 38096 D-s7-1 Replace "eliminate" with "minimize". Para 61.51  $(a){6}$ 

p. 38096 Para 61.52 (z)(4)

D-52-57 Para (a)

D-56-2

(7)

(2)

38096

p. 38097 Para 61.55

The word "orderly" should be deleted or explained.

"Accurately located" depends on the state of the art. The word "accurately" should be defined or not included.

Table 1 has a footnote eliminating wastes containing chelating agents in concentrations greater than 0.1% except as specifically approved by the Commission. This requirement eliminates most routine decontamination techniques to reduce occupational exposures and thus adversely affects ALARA programs. It is not clear why this restriction is being imposed. Guidance on acceptable packaging and disposal techniques for these agents is needed.

. . . .

÷

. . .

p. 38097 Table 1 is a specific list of radioisotopes with their respective concentration limits for three clas-Para 61.55 sifications of wastes, Class A segregated waste, Class B stable waste, and Class C intruder waste. While some of the concentration limits shown are reasonable, D-55-6 demonstrating compliance for others would be difficult (for example Ni-59 and Ni-63) because of problems in sampling and long delay periods for off-site transport which consequently result in additional increases in personnel radiation exposures. Additionally, the measurement of IRU in the 10 nanocurie/gram range in the presence of other interfering radionuclides is D-55-3 D-55-8 reduction increases the concentration of radionuclides D-55-8 reduction increases the concentration of radionuclides in the waste and in some cases may cause the waste to (exceed the Table 1 concentration limits. For these reasons, the concentration limits in Table 1 should be reevaluated to determine their ability to be achieved 7)-55-1 in a realistic situation and in a cost-effective man-Lner... Change to read ... "that does not significantly exceed atmosphere at 20 degrees C". If this requirement is p. 38097 Para 61.56 not modified, waste generators may be required to package gases under reduced pressure or elevated temperatures. The basis for the 100 curies should be (1)(7) D-58-5 / provided. p. 38098 By specifying a 5% limitation on the physical waste for specifying a st finitation on the physical waste form, the proposed rule may well be unintentionally mandating a high integrity container for Class B wastes. Experience has shown that drums and liners can normally be filled to about 80% of their volume. Demonstrating compliance with a 95% criteria could be difficult. Also, the 50 psi compressive load criteria may eliminate bitumen media as a waste stabilization -Para 61.56 (b)(1) D-56-9 process. The compressive load criteria may also be -imore appropriately related to individual disposal site D-56-11 overburden characteristics rather than specified as a generic criteria.

. . . . .

Lausville, Kentucky 40207 522 418-7160 12 ( 17 27:17 -6-Eliminate radicactive waste already disposed of and covered from NRC inspection requirement. It is not understood how this paragraph applies to 10 CFR 61. USEcology ΙÔΙ Regulatory requirements suggest rewording section to "conduct operations in a manner which assures com-pliance with Paras. 61.55 and 61.56 of this chap-وعين تدريد (46 FR 38081) January 13, 1982 ter." Existing wording implies that a separate qual-ity assurance program will be used. Same comments as suggested for 20.311(d)(3). R. Dale Smith, Chief Low-Level Waste Licensing Branch Civision of Waste Management U.S. Nuclear Regulatory Commission Washington, D.C. 20555 Dear Mr. Smith: US Ecology has reviewed the proposed NRC - rules for the land disposal of radioactive wastas (10 GTR 61) published in the Federal Registar on July 24, 1981, and comment as follows: Disposal Site Design, Land Disposal Facility Operation, and Disposal Site Closure Requirements. Page 38084, Col. 1. GEN-1 This section as well as others in the proposed regulation refer to "eliminating the need for active maintenance after site closure." It is impossible to ensure that the need for no active maintenance can be achieved, so we would suggest that the word "minimize" be used in place of "eliminate." 2. Waste Characteristics and Classification. A. Page 38084, Col. 2. This section states that the stability of the waste and the disposal site should last long enough for the radioisotopes to decay to levels where they are no longer of concern from a D-55-12 migration standpoint. Without any reference to appropriate standards it is the regulatory agency's interpretation < 2 D-56-B which must be used as a standard. References to applicable standards should be given. (5. Page 38084, Col. 3. The limit of 10 nanocurie per gram for TRU waste is excessively conservative and is counter productive. Limits of this nature may dissuade waste generators from practicing volume reduction. D-55-3 22/82 cmp

US Ecclogy, Inc. \$200 Energywire Agad, Suite 526 P.O. Bax 7246

B-368

) p. 38100 ) Para 61.82 )(a)

Partz-1 5 p. 38100 Partz - 1 5 Para 2.103

 $M-2 \begin{cases} p. 38102 \\ Para 20.311 \\ (d)(3) \end{cases}$ 

and (f)(5)

G-1

- 2 -- 1 -Page 38085, Col, 2. Decreased burial efficiencies will be experienced in burial facilities if the Commission fails in the establishment of a lower limit Class A segregated waste concentration. It is 61.5 Interpretations 7age 38090, Col. 3. suggested that de minimus quantities be established to properly . categorize those vastes that are suitable for alternate disposal By placing the responsibility for all regulatory interpretations upon the General Counsel and preventing other Cormission personnel A-6 methods. (e.g. sanitary landfills). from making interpretations, the Commission could significantly disrupt operations by slowing the regulatory process. To avoid J. Financial Assurances. Page 38085, Col. 3. this, the regulations should establish a period, for example 10 days, within which the General Counsel must respond to requests for The Commission indicates that certain types of surety mechanisms interpretation. are acceptable. It is recommended that representatives of the commercial insurance industry be given definitive criteria against E-3. 61.7 (c) (2): Page 38091, Col. 3. which to base any bonds in order to ascertain whether or not underwriters will accept the risk and default provisions set forth in Since closure plans are to be submitted with the initial license these regulations. . . . application and periodically updated until final closure occurs, the licenses should not have to apply for a license amendment to Manifest System. Page 38086, Col. 2 and Pages 38102, 38103, Section 20.311. close the site. The closure plan will have been periodically reviewed by the licensing agency throughout the disposal site The manifest tracking system provides for three (3) copies of the operational phase, therefore, closure of the disposal site in manifest. It has been the experience of this company that a greater accordance with the final updated and approved closure plan should number of copies is needed. A total of 5 identical manifests would be sufficient. provide a copy for the generator, carrier, an advanced copy to the intended recipient, one copy with the shipment and one copy as a 9. 51.13. 7458 38092. receipt for the wasts. • • • B- 1 The definition of what is required for "demonstration" or how analysis 5 Aug 1 1 44 5. Regulatory Flexibility Act. Fage 38088, Col. 1. will be accomplished should be provided. The comments apply to The lack of a de minimus quantity as well as any lover limits on the (61.2 (f) (i) (j). . Class & segregated waste concentrations conflicts with the Department 10. 61.24 Conditions of Licenses. of Transportation Regulations governing the transport of radicactive • • • • material. 49CFR section 173.389 defines radioactive material as 61.24 (f). Page 38094, Col. 2. "any material or combination of materials which spontaneously emit ED-I This section refers to possession of radioactive materials and should ionizing radiation. Material in which the estimated specific be clarified to indicate that buried waste is no longer in the activity is not greater than 0.002 microcurries per gram of material, possession of the licensee. and in which the radioactivity is essentially uniformly distributed is not classed as Radioactive material." 11. 61.25 Changes. ..... 8. / 61.2 Definitions . . . . 61.25 (a). 7290 38094, Col. 3. This section on changes is too restrictive. The licensee should be 2ace 38090, Col. 1. allowed to make changes when deemed necessary providing they do not "Disposal" is defined as isolation of radioactive wastes from decrease the level of protection to the public and provided they ED-I the biosphere. This is not possible as the "biosphere" can be are brought to the attention of the Commission prior to implementation. interpreted as meaning part of the world in which life can exist. Many microorganisms will exist in and around all the waste. It is 12. 81.29 Post-Closure Observation and Maintenance. Fage 18095, Col. 1. suggested that disposal be defined as isolation from the biospheres inhabited by man or his food chain. If the Commission or licensing agency closely ponitors the closure B-7 . per 61.29 and is familiar with the site, the minimum period of five years for post-closure observation and maintenance could be shortened commensurate with the condition of the disposal facility as closed. Provisions should be made for a period of less than five years if conditions warrant upon request of the licensee. 

## 61.30 Transfer of License. 61.30 (a) (5). Page 38095, Col. 2. This section will allow the long-term custodial care agency even though prior commitments have been made to delay in accepting its responsibility. The use of the phrase "when the agency is prepared" in effect negates all prior commitments of long-term custodial care agency. 14. 61.50 Disposal Site Suitability Requirements. ( 61.50 (a) (2). Page 38096, Col. 1. ... D-50-1 GEN-2 Color of the transformation of the second transformation of transfoD-D-50-6 for a (3). Page 38096, Col. 1. D-50-6 for a time frame for projecting population growth and future developmentis needed. Is this time frame the 100 year custodial care period or( the 500 year angineered barrier life span? D-50-3 $\begin{cases} 61.50 (a) (5). 7ace 38096, Col. 1. \\ Clarification is varianted as to the requirement for vell-drained and free of areas of flooding and frequent ponding. Is this a selection criteria based on "as is" conditions or "as can be$ engineered." D-50-4 $\begin{cases} 61.50 (a) (11). Page 18096, Col. 1. \\ 12 the environmental monitoring program is masked then it is not going to be a proper indication of the stability of the site.$ therefore, the use of the phrase "significantly masked" is not appropriate. 7-56-11 15. 61.51 Disposal Size Design for Land Disposal. 61.51 (a) (4). Page 38096, Col. 2. No cover will totally prevent all water infiltration. This section should require that covers be designed to minimize water infiltration. D-51-1 61.51 (a) (5). Fage 38096, Col. 2. In a humid area some minor erosion will occur. This is one of the criteria which must be taken into account when selecting and closing a disposal site. It is not possible in a humid area to ensure that ercsion will not result or that active maintenance will not be necessary in the future.

- 4 -

. . .

16. 61.52 Land Disposal Facility Operation and Discosal Site Closure

- 5 -

D-52 ~2 (a) (4) Page 36096, Col. 2. D-52 ~2 The word "orderly" should be deleted or defined. Orderly placement is often inconsistent with the principles of ALARA.

52.52 (a) (7) Page 38096, Col. 3. The requirement to accurately locate each disposal unit is reasonable if an allowed tolerance is indicated.

(17. Table I Page 38097.

18. 61.56 Waste Characteristics

D-56-5 (a) (7) Page 18097, Col. 1, D-56-5 (a) (7) Page 18097, Col. 1, packaged in greater quantities of radioactive gases to be packaged in greater quantities than 100 curies. This is inconsistent ( with the requirements of these proposed regulations.

(61.36 (b) Fage 39098, Col. 1.

By specifying a 54 limitation on the physical waste form, the

D-56-9 (proposed File may well be unitantionally mandating a high integrity container for Class B wastes. Experience has shown that drums and liners can normally be filled to about 80% of their volume. Demonstrating compliance with a 95% criteria is imrealistic as a practicle matter. Also, the 50 psi compressive load (criteria may eliminate bitume media as & waste stabilization process.

The compressive load criteria would be more appropriately related to individual disposal site overburden characteristics rather than (specified as a generic criteria. . . . . .

Craft MIS Comments

In general, the Environmental Impact Statement (IIS) states current practices and also cites alternate methods. - It is unclear whether the Commission accepts either case or prefers the alternates. It is important for the Commission not to specify a method as any proposed site for disposal will be unique.

1. Pace 5-106

Vibratory compactors are recommended by the EIS. Such compactors are good for granular soils but are less effective for conesive soils. The Commission should specify a degree of compaction or required permeability rather than & method of compaction.

2. Page 3-35, Section 3.6.3.6 Security

Full time security personnal have not been used at the present commercial burial facilities. The full time security provisions such as fences, signs and periodic law enforcement visits have provided effective security and have not resulted in any security violations that would warrant the employment of full-time security personnel. The use of radio communication to contact emergency and law enforcement agencies is not warranted as the fastest and easiest method is the telephone. Many years of experience has shown that the use of telephone commucation in emergency situations has proved reliable.

- 6 -

J. Pace Z-13, Section 2.1.

It appears that the Commission is basing site selection on the ability of the locations to fit their computer model. Jue to projected site complexity, it is not realistic to require a location to fit present computer models.

4. Page Q-16 and 17, Section 4.1.1, Table Q-7.

The cost of building demolition (\$300,000) seems high in view of the fact that the three buildings to be demolished cost only \$423,850 to build (See Q-3) and about half of that would have been materials.

5. Q-16, Section 4.1.2.

Vibratory compaction equipment will be effective only in granular fill.

6. Page X-45, Section 5.1. The last paragraph of this section factually states that this fund is inadequate to pay for long-term care of the site but does not properly state that monies were placed in the State's general fund and not earmarked for the use intended.

7. Page X-50, Section 5.5.

The statement that the fund is inadequate is not based on current facts. The surcharge has increased by 10% in 1981 and the proper investment of the funds will take care of the effects of inflation. Further increases can and are being implemented which will provide substantial funding to the State.

.

ert a state that

Provident

JJS: db



NUCLEAR ENGINEERING DIVISION

January 11, 1982

Secretary of the Commission U.S. Nuclear Regulatory Commission Washington, D.C. 20555

Attention: Docketing and Service Branch

Dear Sir:

The Nuclear Engineering Division (NED) of the American Institute of Chemical Engineers appreciates this opportunity to comment on the proposed regulation related to land disposal of radioactive waste (10 CFR Part 61) published at 46 FR 38081 on July 24, 1981.

The American Institute of Chemical Engineers is a professional organization representing over 50,000 chemical engineers, many of whom are members of the Nuclear Engineering Division. A number of these persons have had decades of experience in most (if not all) of the technologies related to land disposal of radioactive waste.

NED heartily concurs with the urgent need for regulations codifying requirements for land disposal of radioactive waste. However, we believe that Subpart D of the proposed regulations is so basically flawed that it should be deleted and the proposed regulation rewritten and republished for comment.

Generally, the regulation should present not "technical requirements" but "performance objectives." If technical requirements persist in the regulation:

- Overconservatism now embodied in the proposed technical requirements should be considerably reduced.
- Cost/benefit considerations should be based on the entire fuel cycle as now defined by the Administration and recognized by the Commissioners of the NRC.
- The definition of transuranic waste should be made realistic recognizing the entire fuel cycle with a rational application of the ALARA principle.
- Fall recognition of the beneficial consequences of "layered"
  - 4 18 2 11 1 12 13 8 82 8 14 19 19 19 19 19 19 19

C-1

Secretary of the Commission January 11, 1982 Fage Two

disposal should be reflected.

#### PERFORMANCE OBJECTIVES

C -

The primary flaw in the proposed regulation consists of the aportive attempt by the NRC to formulate generic limits for disposal. This will deprive a potential disposal facility operator of the opportunity to either:

- Take advantage of natural or engineered features of his operation to permit him to accept materials of above-normal radionuclide content and dispose of them with quite adequate protection of the public in both the short and long term, or
- Limit his receipts to less contaminated material to permit him to avoid some of the more costly proposed NRC requirements, still with adequate safety, to be able to offer a service less costly to his customers (and, therefore, the public).

An unacceptable reason given by the NRC not to adopt a performance objective appears in Section 2.2.1 of NUREG-0782 (hereinafter, the NUREG), where it is stated, "In addition, it may not be totally clear to an applicant or interested person how to design and operate a disposal facility to meet the general objectives." We submit that anyone so naive or lacking in thorough understanding of all technical aspects of waste disposal should not be involved in a responsible technical position related to a land disposal operation. The role of the regulating agency should be that of regulation, not the establishment of design bases.

We have felt repeatedly in past years that NRC regulations resembled more and more a "cook book" approach. As with other nuclear operations, waste disposal cannot be left to unqualified persons to be carried out simply by rote. It must be done under the supervision of highly qualified management.

Accordingly, there should be no reason why the NRC should not adopt only performance objectives for waste disposal rather than the limits and criteria proposed. To the contrary, it appears that the former would offer worthwhile benefits. Granted, Subpart D does allow departure from its requirements if it can be shown that there will be compliance with Subpart C; however, the very existence of the requirements of Subpart D will inevitably present a major obstacle to an applicant endeavoring to benefit from an especially good site or operational provisions.

UNDUE CONSERVATISM

 $D_{-5} = 1$  We feel that there is gross overconservatism in the NUREG. The proposed

Secretary of the Commission January 11, 1982 Page Three

requirements are, to a great extent, not cost-effective and extend far beyond ALARA requirements. There are many examples of this, including:

- Both NUREG-0456 and NUREG/CR-1005 are frequently referenced in the NUREG. Indeed the conclusions of all three documents bear many similarities. However, in the two referenced documents no credit was taken for the waste package nor for waste stabilization or solidification. The proposed regulation carries with it requirements for all of these. That all documents are similar at the bottom line but so very different in basic premises shows the extreme conservatism in the NUREG.
- It is well known that the mixing of <sup>129</sup>I with stable (<sup>127</sup>I) iodine results in directly proportionate reduction in iodine related dose to the thyroid and other organs. This is recognized in the NUREG (for example, on pages 4-16 and 5-73) but then ignored in consideration of <sup>129</sup>I limits.
- As addressed below, the only scenarios for transport of radionuclides from a disposal trench when the waste is emplaced at moderate depths (say 10 meters) below grade are the water pattways. Sections 61.51(a)(4) and (6) require that the disposal site design be such as to prevent water infiltration and to eliminate the contact of water with waste. If the proposed regulation is adopted, it seems fair to presume that a prospective disposal site operator would be required by the NRG to give reasonable proof that it is probable that these requirements can be and will be met. If so, then there exists no credible water pathway, making limits proposed quite irrational.

#### SOURCES OF WASTE

The NUREG pointedly ignores significant sources of waste. It references what is now known to be an interim federal policy against recovery of valuable, irreplaceable energy resources by reprocessing spent fuel. This is no longer the policy, as was recognized by the Commissioners in their Second Prehearing Memorandum and Order dated November 6, 1981 in the waste disposal confidence rulemaking (PR-50.51). In that document, the Commissioners stated, "On October 8, 1981, the President issued a statement outlining a policy favoring commercial reprocessing." It is well known that the operation of a seprocessing plant generates sizeable quantities of "low level" waste. One must include in such quantities the waste also discharged from the operation of the plutonium and "high level" waste solidification facilities.

VFurther, it is the policy of the current Administration that the nation should

7-55-3

D-55-

Secretary of the Commission January 11, 1982 Page Four

proceed with a breeder program. This will require the processing of plutonium for fuel, generating more waste not addressed in the NUREG.

Indeed, the NRC in its proposed 10GFR Part 60 recognizes the possibility of reprocessing in the definition of "high level radioactive waste." (60.2)

The nature and quantity of reprocessing, high level waste solidification. and olutonium processing wastes mandates that they be included in consideration of land disposal regulation.

#### TRANSURANIC (TRU) WASTES

D-55-3

The definition of the waste form called "transuranic" must be redone for at least several reasons:

- . The quantity of wastes from those portions of the entire fuel cycle ingored in the NUREC, much of which would likely be in the 10 to 100 nCl/cm<sup>3</sup> range, must be taken into account in assessing the cost-benefit balance on the ALARA principle. One of the more significant components of these wastes is, for instance, leached fuel cladding hulls. It is likely that the transuranium content of hulls will be below 100 nCi/cm<sup>3</sup>. To unnecessarily identify them as being excluded from land burial could result in large and wasteful expenditures of money. Other components of these wastes likely will fall in the same range.
- . It appears that, in tying the 10 nCl limit to lung dose, the inhalation pathway has been used. With layered disposal (see below) this pathway for transuranics should not be applicable. The water pathway would lead to, first, only an ingestion ontway, and second, only to the requirement for an inventory limit rather than a concentration limit .-
- . The "fact" that the 10 nCi limit is readily complied with now loses its meaning when all real (even if not now generated) wastes are taken into account and the principle of ALARA is sincerely applied.

VALIDITY OF PROPOSED LIMITS

----

· · · · D-55-1 { As noted above, the NUREG frequently references NUREG-0456 and ~~ NUREG-CR/1005. Both of these documents (with deep involvement of

Secretary of the Commission Janaury 11, 1982 Page Five

NRC personnel) concluded:

· Intruder-construction and intruder-agriculture scenarios would require concentration limits to be imposed on waste to be disposed of by land burial.

. . . .

- The potential for other intruders and the potential (likely or not) man-rem consequences to a few individuals lead to the conclusion that such a scenario is unimportant in the future.
- Water migration scenarios would require not concentration limits but total site inventory limits to be imposed.
- . None of these scenarios is expected to take place while there is institutional control (taken in the NUREG to be 100 years).
- . The "layered" mode of disposal, as described in the NUREG. obviates the need to address any intruder scenario for the higher activity wastes. ·····
- . There is no need for special packaging or stabilization of wastes to make the above conclusions valid (the referenced NURECs assumed no such requirement).

Therefore, the limits proposed in the draft 10CFR 61.55 must be recomputed. There should be a clear recognition that waste buried at the bottom (say 10 meters deep) of an operation is only subject to sitespecific investory limits (a performance objective). Further, credit should be given for compliance with the requirements of 61.31 "disposal site design for land disposal."

Also, the consideration of subsidence should be redone. We can see no valid reason why, during the period of institutional control with monitoring by the NRC, any subsidence could not and would not be promotly corrected. In line with this, we believe that even without the proposed requirements D-56-8 to prevent or minimize subsidence, all significant subsidence could be expected to take place well before the expiration of the 100 year period (after site closure) of institutional control. Accordingly, the costly requirements proposed in Part 61 are unnecessary both during the priod of institutional control and after.

> Finally, througout the NUREG, numbers on concentrations, doses, etc., are expressed to as many as three or four significant figures. Such a presentation is very misleading in that it implies that our knowledge is that precise (which it is not) or that such accuracy is needed (which it is not). • . ..

(We do note with approval that the discussion on 46 FR 38085 indicates that

Secretary of the Commission January 11, 1982 Page Six

de minimis waste classifications will be established. Hence, potentially lower cost disposal options may develop for materials at lower concentration D-55-2 levels. We encourage establishment of such de minimis limits outside the scope of Part 61. This approach should allow consideration of both the form and type of waste.

In conclusion, 10GFR Part 61 should be rewritten based on simple performance objectives. Any competent applicant has available all the necessary tools (hydrological, geological, climatological, etc., data, codes, etc.) to support appropriate site-specific limits for his proposed operation. We urge that he be given the latitude to take advantage of site-specific benefits or to search for an improved site which would give him an opportunity to offer a better service at a reduced cost.

We request the opportunity to answer any negative response by the NRC to any aspect of this discussion.

Sincerely,

R. O. numm

R.I. Newman Chairman, Nuclear Waste Task Force Past Chairman, Nuclear Engineering Division



CONFERENCE OF RADIATION CONTROL PROGRAM DIRECTORS. INC. P. O. Box 148 Concord, New Hampshire:03301 UI 20 PI:12

(103) January 15, 1982 Entry MORDED RULE PR-2 et al (46 FR 38081) RECEIPER -(46 FR 38081) RECEIPER -(46 FR 38081)

Mr. R. Dale Smith, Chief Low-Level Waste Licensing Branch United States Nuclear Regulatory Commission Washington, D.C. 20555

Dear Dale:

The following comments relate to the Proposed Rule for a new Part 61, and other related amendments, as published in Volume 46, No. 142 of the Federal Register, dated July 24, 1981.

The Nuclear Regulatory Commission is to be commended on this proposed regulation. This new proposed rule provides a vast improvement in guidance and requirements for the control of land disposal of radioactive waste. The existing regulation on the disposal of radioactive waste contained in Part 20 ' provides only a limited guidance that has resulted in problems at both NRC and state licensed burial sites.

The Conference has taken the position since the late sixties that more federal guidance and criteria is needed in the area of shallow land disposal of radioactive waste. The Conference has passed various resolutions addressing this concern. Such guidance is imperative for uniform management of the country's radioactive waste.

Many of the improvements in the proposed Part 61 are responsive to some of the recommendations made by the Conference. Froposed Part 61 also establishes, in the NRC regulatory system, many "state of the art" improvements that have been developed by the states in the operation and regulation of low-level radioactive and hazardous waste burial sites.

The Conference concurs and supports the following proposals contained in the new Proposed Part 61:

Solution: Several categories based on hazard evaluation.

D-55-3 2. Confirmation of the definition of transuranic wastes as recommended by the Conference several years ago. Mr. R. Dale Smith January 15, 1982 Page 2

E-1

3. Technical requirements for burial that become more stringent based on the increasing hazard of the radionuclide concentration in the waste.

5-8  $\{4. \text{ Technical requirements on stability of waste packaging.}$ 

5. Technical requirements on burial site operations that minimize voids in trenches with emplaced waste.

6. Technical and financial requirements associated with the site closure phase and the post-closure observation and maintenance phase.

 $\mathcal{D}$ -59-2 7. Defining a period of time for institutional control, and relating the classification and disposal of waste to this time frame.

For design purposes of new proposed sites, and until specific recommendations are forthcoming from the Environmental Protection Agency, the Conference supports the objective in Proposed Part 61 that any movement of radioactivity should not result in calculated doses in excess of 25 mrem/year to an individual at the site boundary, and support the application of the EPA drinking water standard to the nearest public drinking water supply? We also recommend the application of these dose limits as guidelines for existing sites. Of course, the application of ALARA should be applied to a near surface burial site, as with other licensees.

We strongly support the proposed amended requirements to Part 20 for the certification and use of shipping manifests to track waste shipments.

We offer the following specific comments on the proposals.

(1) 61.2 Definitions:

a. Should include a definition for "minor custodial care."

. .

ED-1, b. "Disposal" - As stated, the temporary storage of waste could meet the definition. Disposal into land generally connotes long term or permanent removal of the waste from the biosphere. Possibly the words "long term" should be added before the word "isolation."

> C. "Waste" - The definition should include levels of concentrations of radioactive materials specific to a particular waste stream below which regulatory control is no longer required. This position was previously

Mr. R. Dale Smith January 15, 1982 Page 3

)-55-2 Stransmitted to the Commission in the form of Resolution III which was adopted at the thirteenth annual Conference meeting.

Although the definition for waste may not be the area in which the following concerns should be addressed, we strongly urge the Commission to consider the total hazard, both radiological and chemical, in the classification of acceptable waste for a radioactive burial site. Although this concern is somewhat addressed under 61.55, criteria or guidelines are needed which specifically consider the appropriateness and/or procedures of combining waste which is highly toxic, chemically, but low in radioactivity, with chemically nontoxic radioactive material.

Consideration should be given to a definition of "toxic chemical/radioactive waste" which may require different handling and burial requirements. This concern was expressed to the Commission in the form of Resolution II adopted by the Conference.

· · · · · · · · · · ·

128

-

•••

2.55

1

104

#### (2) 61.7 Concepts: 2

Under section (c)(4), "The Licensing Process," the concept of license transfer to a state or federal agency after finding of satisfactory disposal site closure is discussed. Under this concept, if transferred to a state, the NRC would be licensing a state government until institutional care is not required. This concept for a low-level waste site may need further discussion and refinement before implementation. For example, what criteria will be used to judge the adequacy of the state government licensee? If "state government" violates the conditions of the license, what enforcement actions would be taken? "Additionally, why would the license be terminated if transferred to the Department of Energy, but not terminated if transferred to a state? Experience and history has shown that states have been as effective as the federal government in assuming responsibility for long-term care of existing sites. Therefore, consideration should be given to license termination after transfer to a state government.

#### (3) 61.12 Specific Technical Information:

Section (d) would require a description of the design basis natural events or phenomena. Requirements should be

en al antiga de la companya de la co

R-8

Nr. A. Dale Saith  
yang 4  

$$\begin{cases}
\theta^{-1} \\
\theta^{-1} \\$$

GE

Mr. R. Dale Smith January 15, 1982 Page 5

· ·

standards of the new Proposal. We recommend immediate consideration be given to the following:

- Standards for acceptable solidifying agents for liquid low-level waste.
- 2. Acceptable testing procedures to determine if solidified liquids meet the above standard.
- 3. Comprehensive licensing guidelines for waste reduction methods such as incineration and compacting units.
- 4. Guidelines for types of acceptable or optimum geological formations for the land disposal of low-level radioactive waste.
- Standards or criteria for the optimum design of a site, considering dry vs. humid climates.
- 6. Guidelines as to acceptable or optimum burial practices.
- 7. Guidelines relating to the minimum acceptable health physics program for a near surface burial facility.
- Guidelines relating to the minimum acceptable neuradiological occupational protection program for a near surface burial facility.
- 9. Guidelines relating to the specific areas that must be considered for emergency planning for a near surface burial facility.
- 10. Guidelines on environmental contamination trend analysis, and recommended protective actions based on potential increasing environmental levels of specific radionuclides.
- Standards or criteria which specifies what constitutes stabilization and a decommissioned site.
- 12. Guidelines which identify the minimum acceptable activities to be performed by a government agency after closure.
- Guidelines on methods of determining financial needs for long-term care.
- 14. Guidelines as to the minimum acceptable environmental monitoring programs for a near surface disposal facility.
- Guidelines on the application of ALARA for near surface facilities.
- 16. Guidelines on ground water modeling for near surface facilities.

Mr. R. Dale Smith January 15, 1982 Page 6

Again, I would like to commend the Commission for the excellent work contained in the new Proposed Part 61. Such regulations have been needed for many years. We appreciate the opportunity to comment on this very vital need to our country.

Yours very truly,

John R. Stanton, Chairman Conference of Radiation Control Program Directors, Inc.

JRS/CMH/pch

cc: Board Federal Lisisons Executive Secretary G. Wayne Kerr

SCHOOL OF HEALTH SCIENCESD 1 113 ang

NORT EVERENT

78032 -----

.. .

<u>\_\_</u>

January 15, 1932

Samuel Chilk Secretary, Nuclear Regulatory Commission Attention: Docketing and Service Branch Washington, DC 20555

Dear Mr. Chilk:

I am writing you as Chairman of the Program Review Committee, an independent review committee comprising representatives from industry, institutions. environmental groups, and state and federal government.

EG&G Idaho, Inc., as Manager of the Department of Energy's Low-Level Maste Manacement Program, has constituted this review committee. The Committee has reviewed your proposed regulation 10 CFR 61. Most of our comments have been included with the Department of Energy comments through EG&G Idaho. However, there is one matter that we consider merits special comment.

From our review and discussions with NRC, DOE, and EGIG Idaho staff, we find that the "inadvertent intruder" is the dominant or controlling feature of the regulation. The waste categories, radionuclide concentration limits, and other matters are set by the calculated dose to the inadvertent intruder. In our judgment the HRC has given too much weight to the intruder scenario in the development of the regulation. We do not feel that safety regulations for the general population should be dictated by the hypothetical number actions of a very small number of individuals.

Our judgment is that the inadvertent intruder scenario is more enalgous to an accidental exposure pathway than to a chronic exposure pathway. Action to mitigate the consequences of the event and to lessen the likelihood of the event seems more appropriate than using the event as a basis for operational regulation.

We agree that the "inadvertent intruder" should and must be considered in the regulation. However, more emphasis should be placed on requirements to reduce the likelihood of intrustion. One such requirement might be to use durable monuments large enough to warn potential intruders. The orcoosed waste form and container requirements for structural stability may also serve to mitigate the consequences of an inadvertent intrusion.

Pharmacy Building TE Stress Lafavette, Indiana 47907

C-4

Samuel Chilk January 15, 1982 Page 2

The above discussion represents the general feelings and consensus of the committee. There was unanimity among the Committee members that this guestion and our judgments should be brought to your attention. The Program Review Committee would be happy to elaborate on these comments or answer any further questions concerning our conclusions. Please accept our compliments on your efforts and our thanks for the help and assistance received from the NRC staff, in particular, Ms. Kitty Dragonette, who was very helpful and attentive.

Sincarely.

Paul L. Ziemer, Chairman Low-Level Waste Management Program Review Committee

PLZ:mhf

cc: Cale Smith Division of Waste Management U.S. Nuclear Regulatory Commission Washington, DC 20555



#### THE UNIVERSITY OF TEXAS SYSTEM CANCER CENTER

M. D. Anderson Hospital and Tumor Institute2() 93:32 Texas Medical Conser Houston, Texas 77030

Department of Physics



Secretary of the Commission U.S. Nuclear Regulatory Commission Washington, D.C. 20555 Attention: Docketing and Service Branch

Cear Sir:

I appreciate this opportunity to comment on Proposed Licensing Requirements for Land Disposal of Radioactive Waste. There definitely needs to be a single, coherent set of licensing requirements regarding burial sites and I am glad to see these regulations formally presented for review. I am particularly gratified to see the waste classification scheme, Part 61.55.

The University of Texas System Cancer Center generates radioactive waste in the course of medical therapy, diagnosis, and research activities and may be directly affected by parts 61.55, 61.56 and 20.311. I will confine my comments to these parts.

Part 61.55

The waste classification of isotopes and activities is acceptable. All radicactive waste generated here would fall under Class A segregated waste with regard to activity.

Part 61.56

My uncertainty with portions of this section is NRC interpretation in applying these requirements to medical research waste containing organic solvents such as toluene, xylene, dioxane, etc. used in liquid scintillation counting. As I understand this part, crums containing liquid scintillation vials, which in turn contain organic solvents, would be unacceptable due to parts (a) (5), (a) (6) and primarily (b) (2). It would be worthwhile for the NRC to refer specifically to liquid scintillation vial waste. Due to the low activity and small volume of solvents, I think part (a) (3) is sufficient for liquid scintillation vial waste. Complying with part (b) (2) might be practically impossible due to the usually unknown mix of solvents in scintillation cocktail.

Actorowiedzed by and 1. 25/82 . and

Rehabilitation Center The University Cancer Foundation The Anderson May fair Extraineral Programs Division: knowners and iscore rack is bridge simultaneous dral Institutions - Collaboration Studies Subassions

D-56-7

Secretary of the Commission Page 2 January 13, 1982

### Part 20.311

M-4

M-4

It seems to me that certain parts of this section put an undue burden on generators of low level medical radioactive wastes in terms of increased cost and manpower required to implement these rules. According to this section all waste generators are considered shippers and are responsible for meeting all of the implementation and record keeping requirements of this section regardless of arrangements made with other parties (i.e. collectors, reprocessors) for shipment of wastes. The minimum requirement for medical low level waste generators should be to accurately indicate radionuclide identity, activity, physical forms, and MEC waste classification for each drum or container.

If other parties are willing to take responsibility for wastes leaving the generator's facility, they should be permitted to do so. Otherwise this is yet another requirement forced upon medical institutions which could better utilize their money and personnel on other more pressing safety problems.

I have the following specific comments on this section:

(b) As stated above, for medical low level radioactive waste I see no need for information provided by the generator other than radionuclide identity, activity, physical form and NRC waste classification for each container.

(c) The generator should as a minimum certify only the proper classification, description and packaging of wastes. Other parties (i.e. commercial firms) should be able to certify the remaining information if they contractually agree to this.

The generator institution should be able to allow individual authorized users of radioactive materials to certify the above information.

M-2 (d)(3) Other parties should be allowed to perform this assurance under contract. In any case the NRC should specify what a quality assurance and management audit program should entail. For Class A waste this should be minimal.

(d)(4) See comments on (b) and (c).

(d)(5) This is unnecessary when another party is receiving the waste at the generator's institution.

(d)(3) Generators should be exempted form this requirement when another party receives the waste at the generator's institution.

Secretary of the Commission Page 3 January 13, 1982

> The licensed waste processor should be allowed to perform duties for the generator as commented upon above. This will also eliminate unnecessary duplication of effort by the generator.

> > c 🚓

(h) See (d) (8).

Sincerely M. E. Norten

Radiation Safety Officer

cc: Edgar Bailev. Texas Radiation Control Bureau

MEN:eg

Carolina Power & Light Company

January 14, 1982 J. 20 73:32

Secretary of the Commission U. S. Nuclear Regulatory Commission Washington, D.C. 20555

Attention Docketing and Service Branch

KINE RUXER DR. 2.2 al FROMOSED RULE DR. 2.2 al (46 FR 38081) D-55-6

Dear Sir:

As requested by the notice published in the July 24, 1981 Federal Register, Carolina Power & Light Company (CP&L) hereby submits comments on the proposed 10CFR Part 61. As generators of low level waste, CP&L has concerned itself primarily with the portions of the proposed 10CFR61 which address waste classification. However, in regard So the entire regulation, we express our support for the concept of stating overall performance objectives for low level waste disposal supplemented by prescriptive design requirements only where necessary.

COMMENTS TO PROPOSED IGGTR PART 61

This approach provides needed flexibility, particularly in landfill design where local conditions vary.

Comments on specific sections of LCCFR Part 61 are as follows:

#### Section 61.27, Application for Peneval or Closure

This paragraph discusses the license renewal requirements. On page 38087 of the preamble, NRC indicates that the renewal period will be five years. We think this is inappropriate and unnecessary. Section 61.25 <u>Changes</u>, already requires Commission approval before changes can be made to the site. Section 61.24(a) already provides authority to revoke modify or suspend a license. Therefore, the automatic five (5) year renewal is not needed. Also, a nuclear power plant is licensed for the life of the plant (normally 40 years). Since the degree of public risk is very low for a low level waste disposal facility, the license should be issued for the life of the facility, similar to a nuclear power plant license. In this way, a private operator of a low level waste facility can be assured that his initial investment would not be lost after five (5) years, should his license not be renewed.

#### Section 61.55, Waste Classification

D-55-6

B-5

The waste classification system as outlined in Table 1 is unworkable. The Draft Environmental Impact Statement (DEIS) Summay, pages 43 and 44, addresses the difficulty of measuring

Antonovisione by and 1/2 1/8 2 414

such isotopes and proposes the use of scaling factors based on measured levels of indicator isotopes. This wording must be included in 100FK61 to avoid future enforcement problems. The standard error associated with this technique must also be recognized and allowed for enforcement. Language must be introduced into Part 61 recognizing that the specific activity of isotopes need not be measured directly and permitting the use of indirect methods for <u>reasonably estimating</u> them.

An attractive and reasonable alternative which would avoid these isotopic analysis problems and still accomplish the Commissions objective of having an <u>inventory</u> of all waste at a site at time of closure, would be to entitle each column in Table 1 as <u>typical wastes</u> or <u>waste sources</u> (similar to the list shown on page 35085 of the preamable). For enforcement purposes, a simplified indirect measurement of indicator isotopes in each package would then suffice to categorize waste packages as falling within the "typical" levels shown for each column of Table 1.

- We strongly support the NRC's efforts in establishing a "de minimis" classification which would essentially be exempt from Part 61. For clarity, we would suggest a column for the "de

-55-2. A Part 61. For clarity, we would suggest a column for the "de ninimis" levels be added to Table 1. We would also suggest a column be established to cover all low level wastes that exceed Class C wastes, so that it is clear what is to be done with these wastes as well.

In establishing a limit of 0.1% by weight for chelating agents in waste, the NRC is not taking into account the stability requirements of Class B and C wastes. We feel that applying a 0.1% chelating agent limit to these wastes will discourage the use of present and developing decontamination techniques useful in reducing occupational exposures. If the NRC believes it is necessary to establish additional protection, guidance for proper and reasonable packaging should be developed as opposed to concentration limits. General performance guidelines could achieve stated geals without requiring the extreme efforts necessary to routinely demonstrate compliance.

D-55-3 We recommend a 100 nCi/gm limit for transuranic vaste. The proposed 10nCi/gm is an arbitrary level and is very conservative from a health and safety standpoint. Although 10nCi/gm is achievable during normal power plant operations, there are documented reports of levels occasionally reaching the 10-100 nCi/gm range (EPRI Project 613, August, 1950). These events ware associated with unusual fuel performance. Raising the limit would accomplish two things. First, the vary small amount of power plant waste falling in the 10-100 nCi/gm range -3-

D-55-3 Can be readily disposed of safely. Second and more importantly, all waste from operating power reactors could safely be assumed to fall below the 100 nCi/gm level, thereby achieving compliance by definition and making it unnecessary to perform direct or indirect measurements which are technically very difficult.

. . . . . .

Table 1 is designed to provide a limit of 500 mRem/year exposure to an inadvertent intruder 100 years after closure of the site. We feel the 5 Rem/year limit currently applied to radiation workers represents an acceptably low risk and should be the standard used in this intruder scenario. The low probability of intrusion and the limited number of individuals und would ever be involved in such an event would justify calculating risk based on individual exposure and not population exposure. This change would increase allowable capacity at the disposal sites and would reduce the total number of required sites in the future.

The limits established in Table 1 assume an intrusion 100 years after the waste is placed. From our review, it appears that no credit has been allowed for the decay of the predominately short-lived isotopes that would occur during the operational phase. If so, higher activities could be allowed per vaste class in the early years and still meet the intrusion criteria after 100 years.

Due to the expected increase in paperwork associated with lOCTR 61, we encourage that existing forms and documents be used where possible to avoid unnecessary additional administrative burden.

A general comment concerning Table 1 is that it contains far too much information to be limited to the space given. We would suggest that narrative be provided to explain the intent of each classification and footnote.

Of final concern is that Table 1 and the waste classification scheme in general would appear to discourage waste volume reduction which is contrary to the NRC Policy Statement of October 16, 1981, concerning volume reduction, as well as CPLL policy of promoting total waste volume reduction. If volume reduction is to be encuraged, a waste classification system must be developed in the final 10CFR61 whereby users are rewarded in some way for using waste volume reduction techniques.

Section 51.56

D-56-11 The prescriptive requirement of 50 psi compressive load for waste form stated in 61.56 (5)(1) should be replaced by a performance objective allowing specific requirements to be developed on a site-bysite basis to avoid subsidance at that specific site configuration. It is cur understanding that to maintain the waste stability to within 52 would require filling the waste container to less than 952. This is D-S6-7 impractical from a large operations standpoint, and should be changed to a performance objective which would require filling the container as close to capacity as reasonable.

#### Agreement State Status

Within the proposed IOCFR51, the status of agreement states is ambiguous. There are several places in the regulations where it would be appropriate to explain how IOCFR Part 61 would apply in states with agreement status. Specific language is suggested as follows:

#### Section 61.1(b)

"Except in Agreement States where a compatible waste disposal program is in place or as provided in Section 61.5 'Exemptions'...."

#### Section 61.70

"This subpart describes mechanisms through which the Commission will implement a formal request from a non-Agreement State, an Agreement State without a compatible disposal program, or Tribal government to participate in the review...."

#### Draft Environmental Impact Statement (EIS)

DEIS is unreasonably voluminous based on the small degree of environmental or safety risk from low level waste. The disposal of wastes has received significant public attention recently and has become a very sensitive topic. A DEIS of this volume tends to inappropriately legitimize these overstated concerns.

Should you have questions concerning our comments, please let me know. We appreciate this opportunity to comment on the proposed regulation.

Very truly yours,

EParitck %. Howe Vice President Technical Services

PWH:dcj

A -

**Nuclear Engineering Division** (Founded in 1954)

EXECUTIVE COMMITTEE 1931-1882

Chairman REMCO P. WASZINK Westinghouse Electric Corporation P.O. Sox 19218 Tampa, PL 32686 (\$13) \$37-7273 Vice Charmen BURTON D. ZIELS Diamond Power Specify Corporation P.O. Box 418 Lincaster, OH 43130 (814) 687-4069 ERENT A. SHELTON Commonwealth Ealson P.O. Bes 757 Chicago, IL 60690 (312) 294-2837 AIRSUS RHOL Hoimes and Narver, Inc. 700 Kingerkamack Road Oracell, NJ C7549 (201 263-2000 Ext. 2120) Secretary VILLIAM E. McCOWN Vissinghouse Hanford Ca. R.O. Box 1970, W/8-23 Richard, 4:4 89352 (503) 378-4209

ADMINISTRATIVE COMMITTEE CHAIRMEN

Papers Review 8. D. ZIELS Program B.R. SHELTON

Honors B. D. ZIELS S. M. HOOLUND

Publications T. W. EDWARDS

Liembership JOHN SUSNIR Public Issues L.A. SAAYER, JR.

LIAISON REPRESENTATIVES

5. D. 218 LS (to énergy Convertion Group) L. C. OVEI: (a American Nuclear Society) M. N. HCGLUDD (to American Power Conference) V. 20mULT-EIS (to Nuclear Power Code) & Signation Committee) L.A. 164Y 83-, R. (Sovernment Relations Committee) Enclosure Contact

GPERATING COMMITTEE CHAIRMEN Advanced Rescions VI. J. RCVIAN

Nuclear Heat Exchanger S. M. CHO

Prozest Servicing

Pressure Vessel & P.p.np 15. D. BERNSTEIN

Ascuarie Svrieme WA REITER

Fare Suiters S. L. ROSEN

Mambar . Anierican Association of Engineering Societies . Accessitiation Speed for Engineering and Technology

Please address realy to: General Electric Company 175 Curtner Avenue San Jose, CA 95125 Mail Code 772 U \$364.... 1223 \*\*

. . . .

U.S. Nuclear Regulatory Commission

Attention: Docketing & Service Branch

12 01122 20:55

en

.....

The American Society of Mechanical Engineers

January 21, 1982

Secretary of the Commission

Washington, D. C. 20555

Subject: Transmittal of Comments to ICCFR Part 61, "Licensing Requirement for Land Disposal of Radioactive Waste"

We are pleased to take this opportunity to comment on the proposed rule, 10CFR Part 61, "Licensing Requirements for Land Disposal of Radioactive Waste"

The attached comments are those of the Radwaste Systems Committee (ASME), chaired by Larry Reiter.

Finil 8. Hel

Kenneth E. Kolb Member, Radwaste Systems Committe of Nuclear Engineering Division of American Society of Mechanical Engineers

Lananastrea

122 JUN 22 97:55

-----

REVIEW COMMENTS OF PROPOSED RULEMAKING

ON LAND DISPOSAL OF LOW-LEVEL RADIOACTIVE WASTE - 10CFR61

SUBMITTED BY

CODES AND STANDARDS SUBCOMMITTEE RADWASTE SYSTEMS COMMITTEE NUCLEAR ENGINEERING DIVISION OF AMERICAN SOCIETY OF MECHANICAL ENGINEERS

## PAGE 38084 - PARA. V. C - WASTE CHARACTERISTICS AND CLASSIFICATION "Stability should last long enough for the radioisotopes to decay to levels where they are no longer of concern from the migration standpoint." What criteria determines the length of time migration concern? 5.. PAGE 38059 - PARA. 61.1(a) -PAGE 38087 - PARA. G - OPERATIONAL PHASE "At intervals specified in the license (the normal term for materials license is currently five years) the licensee would be required to submit a license renewal application." A-Suggest that the site be provided a full term license with subsequent periodic reviews not subject to public hearing. The review should ascertain that the conditions supporting the full term license have not changed significantly. Paragraphs 61.25 and 62.26 in the proposed regulation provide adequate assurance that licensee-originated changes will receive review by the Commission. New-found issues of national concern that are independently identified by the Commission can be applied on a national basis to the sites as such issues are identified and are not dependent on waiting for a renewal application. PAGE 38087 - PARA. G - INSTITUTIONAL CONTROL BOARD "...surveillance to keep people off the site..." The Institutional Control Board should have the prerogative to deter-EDmine the extent of site access on a site specific basis. PAGE 38086 - PARA. V. F - MANIFEST TRACKING SYSTEM "...to provide copies of the manifest to proceed and accompany shipments..." - . . The need for a manifest system to assure traceability of waste shipments from a generator through the transporter and finally to the disposal site, is recognized. We question the need for a copy of the manifest preceding the shipment for the following reasons: 1) The copy of the manifest accompanying the shipment will allow the transporter and disposal facility to verify the shipment content. 2) The expressed concern that a missing or delayed shipment would not be detected can be reconciled by other methods such as an independent transmittal of the manifest at the time of shipment or by telephone notification to the receiving facility at the time of shipment.

B-5

M-1

B-383

"...Commission issues licenses, for the disposal for others of radioactive wastes ... set forth in Part 20 of this chapter." While IOCFR Part 20 covers the disposal of waste by an individual

l'icensee, the quantities are limited to very low levels. The purpose and scope should be rephrased to allow an individual licensee to operate a burial site. The words "for others" and the last sentence to 51.1(a) should be deleted.

3) The requirement for the manifest to precede the shipment implies that the shipment should not leave until notification has been received by the shipper that the receiver has received said copy. This can result in a shipment sitting at the initiating site for

an extended period while these notices go back and forth.

PAGE 38090 - PARA. 61.2 -

"Disposal" means ... facility."

Suggest: "Disposal" the placement of waste in a licensed land disposal facility for radioactive waste.

PAGE 38090 - DEFINITIONS

"Near surface" disposal facility means land disposal facility in which radioactive waste is disposed of in or within the upper 15-20 meters of the earth's surface."

It is suggested that this definition be chanced to read as follows:

"'Near surface disposal facility' means disposal facility in which radioactive waste is disposed of in or within the upper 15-20 meters of the earth's surface or to whatever greater depth can be demonstrated as capable of meeting the required performance criteria and technical specification."

Rationale: The restriction in or within the upper 15-20 meters could prevent utilization of greater depths at locations where hydrogeological conditions and waste stability characteristics would allow this. The -criteria of the proposed regulation are established to prevent exposure to the public by transmittal through ground water flow and to prevent ` exposure to the intruder. The establishment of an allowable depth should be made on a site-specific basis and with the objective that the criteria will be met. The unsubstantiated establishment of a nationwide depth limit is not in keeping with the logic used throughout the rest of the proposed regulation.

2

# PAGE 38090 - PARA. 61.2 "Earth's surface" should be defined. This could be the final surface elevation of the disposal site as used in the site closure and stabilization plan.

### 9. PAGE 38090 - PARA. 61.2 -

"Stability" should be defined.

It'is a basis for separation of Class A and B waste.

10. PAGE 38090 - PARA. 61.3(a) -

ED-

Change: (...issued by the Commission pursuant to this part.)

to: (...issued by the Commission pursuant to this part or unless exemption has been granted by the Commission under Paragraph 61.6.)

Rationale: Paragraph 61.3(a) as written would prohibit transfer for land disposal of any radioactive waste to a nonlicensed person. This is overly restrictive and would force the shipment (to a licensed facility) of radioactive wastes that are not of a health or safety concern. The suggested addition to Paragraph 61.3(a) would allow determinations to be made by the Commission on a case-by-case basis where it could be demonstrated that health and safety concerns could be met by alternate disposal methods.

PAGE 38091 - PARA. 61.7(a)(1) -

"...uppermost 15 to 20 meters of the earth."

Suggest: Addition of sentence:

(Surface burial deeper than 20 meters may also be satisfactory.)

Rationale: Deeper surface burial may prove satisfactory relative to protection of the public and economics.

#### PAGE 38091 - PARA. 51.7(b)(1) -

2.

"(b) Waste Classification and Near-Surface Disposal. (1) Disposal of radioactive waste in near-surface disposal facilities has two primary safety objectives: prevention of migration of radionuclides, primarily through groundwater; and prevention of exposure to inadvertent intruders."

A paragraph change to include the following is proposed: "...has the following safety objectives: 1) Minimize migration by surface and groundwater, and EDwind effects. 2) Keep personnel dose ALARA. 3) Keep environmental impact within specified limits." 13. PAGE 38091 - PARA. 61.7(5)(2) -This paragraph states that for certain isotopes a maximum disposal site inventory will be established based on the characteristics of the disposal site. D-55-10 Because this rule is site capacity and size limiting, criteria such CEN-2 as the isotopes, their maximum permissible inventory, and inventory function site characteristics should be established. PAGE 38091 - PARA. 61.7(c)(3) -14. Succest that: ED-"During the period when the site closure..." be chanced to: "During the period when the final site closure...". 15. PAGE 38092 - PARA. 61.13 -It is not apparent what is required for "demonstration" or how analysis B-1 will be accomplished. This section should be clarified. This comment also applies to Paragraph 61.2(f)(i)(j). Once buried, the waste is no longer in the "possession" of the licensee. 16. PAGE 36094 - PARA. 61.24(b) -"The licensee shall submit written statements under oath

"The licensee shall submit written statements under oath upon request of the Commission, at any time before termination of the license, to enable the Commission to determine whether or not the license should be modified, suspended or revoked."

It is suggested that this paragraph be deleted.

Rationale: The paragraph is very vague as to intent and method of implementation. It is not clear under what circumstances such an oath would be required and has a direct implication that the licensees are untrustworthy. There are certainly sufficient written transmittals required in

B-12



 $\mathcal{D}\text{-}55\text{-}1 \begin{cases} \text{The concentrations given in this Table are much more limiting than is} \\ \text{necessary. In the study prepared for the USKRC by Ford, Bacon, & Davis} \\ \text{Utah, Inc., "A Radioactive Waste Disposal Classification System,"} \\ \text{NUREG/CR1005, conservative radioactivity limits for various waste classes} \\ \text{were established through detailed hazards analysis. The limits recommanded} \\ \text{in NUREG/CR1005 should be incorporated into locFR51 in place of the arbitrary values in Table 1.} \end{cases}$ 

D-55-6 An alternative method of determining the waste classification should be provided. Provisions for classification by external dose determination should be made. For cases where the types of isotopes of concern are known, this method would allow adequate classification. This alternate method would be particularly helpful for nuclear power plant trash. Generally, trash has a very low specific activity compared to the Class A limits. A determination of the radionuclide identity and concentration, as required by Part 20.311, would require the purchase and use of a portable spectrum analyzer. Instead, a contact dose rate measurement of the containerized trash could be made to show that the activities were below the Class A limits. Also, since a radwaste classification system is already established in 10CFR71, is it possible to tie the two systems together?

 $\mathcal{D}^{-55-1} \begin{cases} \text{The logic behind the numbers selected for this table is not apparent. It would appear, for example, that carbon 14 which contains less than 0.8 microcuries per cc may be disposed of as segregated waste but that any concentration greater than 0.3, even if it is only a tiny increase, immediately requires that the disposer seek special permission from the government for disposal. The abrupt demarcation needs explaining so that the logic of it can be understood.$ 

#### De Minimis Classification of Wastes

Section 61.55. Table 1: should consider a "de minimis" classification of wastes (i.e., wastes that would be considered of non-regulatory concern); we believe strongly that this should be addressed in the proposed IOCFR61 regulation. De minimis levels for uranium, technetium, plutonium, and neptunium should be stated. A de minimis or lower acceptable level for D-55-2 natural and depleted uranium should be stated; we recommend that a value of 0.035 cercent natural and depleted uranium be set as a lower limit in Section 61.55, Table 1. Recent information received from Nuclear Regulatory Commission staff members reveals that proposals concerning de minimis levels for uranium are being prepared by the NAC staff that would establish multi-tiered acceptable levels for shallow-land burial of uranium wastes. One level proposed by the NRC would permit disposal of uranium wastes in an unlicensed burial ground (i.e., sanitary type). A second proposal would permit discosal in a shallow-land licensed burial site, and a third prosesed level would permit disposal in a shallow-land licensed burial site which has a covenant in the title on the property. These multi-tiered NRC processals are consistent with out recommended levels. We recommend that de minimis levels consistent with those proposed by NRC relative to "To and low-enriched uranium as residual contamination in smelted alloys

D-55-2 (Addendum to 10CFR Parts 30, 32, 70, and 150) also be stated. In this regard, we recommend that serious consideration be given to establishment of de minimis levels of 3.5 ppm <sup>233</sup>U, 5 ppm Tc, 0.01 ppb Pu, and 1 ppb Np.

In Section 61.55, Waste Classification, Table 1, we believe that the maximum concentration for alpha-emitting transuranic isotopes should be increased from the 10 nCi/g limit presently proposed to 100 nCi/g. Part of the rationale behind the 10 nCi/g limit is stated to be that this value has been imposed by DDE; however, DDE is at the present time seriously considering revision of DDE Manual Chapter 0511 to raise this limit to

**D-55-3** Considering revision or DUE Manual Chapter USI to raise this limit to 100 nCi/g dated 7/30/81). The 10 nCi/g value is also inconsistent with the value of 100 nCi/g used by the Environmental Protection Agency in their proposed regulation 40CFR Part 91 for the disposal of spent fuel, high-level, and transuranic (TAU) wastes. This regulation states that TRU wastes containing more than 100 nCi/g of alpha-emitting TRU isotopas must have the same controls as are required for high-level wastes. We recommend that the 100 nCi/g limit be reflected in each of the columns 1-3 in Table 1. We also recommend that a limit of 100 pCi/cm<sup>2</sup> for transferrable surface contamination of alpha-emitting transuranic isotopes (not natural or depleted uranium isotopes) be imposed, consistent with the proposed revision to DOE Manual Chapter 0511.

> Additionally, the value of 10 nCi/g is based on naturally occurring radium deposits. Radium is significantly more hazardous than <sup>21</sup>JU or the transuranium nuclides when dissolved in water, as the MPC's for the soluble forms of these nuclides are about 100 times (1000 times for <sup>21JU</sup>) greater than that radium. Thus, it would appear reasonable to set activity limits for alpha-emitting transuranic isotopes at 100 times greater.

> The supplementary information in the NRC document also states that there is no need to increase this limit from the standpoint of achievability. Much of the waste presently stored as transuranic waste is segregated from low-level waste on the basis of waste origin since the 10 nCf/g limit is too low for accurate measurement and certification. However, segregation according to the 100 nCf/g limit could be achieved, eliminating expensive retrievable storage and deep geologic disposal of "suspect" transuranic waste.

> Another concern is the footnote to Table 1 that refers to isotopes contained in metals, metal alloys, or permanently fixed on metal as contamination. The footnote, which states that "the values above may be increased by a factor of ten," should be modified to include concrete and other media that exhibit low leach rate behavior. An incentive should be provided to reduce the volume of wastes by incineration or metal smalling. These treatments may:normally be avoided by waste generators since they would convert some low-level wastes into transuranic wastes. For example, a volume reduction of 30 by incineration of a waste containing 5 nCi/g would convert a low-level waste into a transuranic waste at 150 nCi/g. However, the residual ashes could be incorporated into concrete, glass, metal, etc. The leach rate of transuranic isotopes from these materials

> > 8

7

is very low (i.e., many orders of magnitude lower than the untreated waste form). Thus, a combination of leach rate and transuranic content D-56-65 (can be shown to have maintained their "stability" for 150 years past. D-56-65 (very few of these are applicable to waste packaging. There is no way that deformation alone of the waste form can be a hazard to the public. could be used to determine the disposal options for these waste forms. The key requirement is to keep the waste from being dispersed, which is little affected by "slumping" or a "55" deformation. Many of the nuclide concentrations limits may not provide a practical basis for classification. In many cases, the measurements are difficult and some are almost impossible. Perhaps the Commission would specify It is succested that Paragraph 61.55(b)(1) be changed to read as follows: practical analytical methods acceptable for determining nuclide concen-"Waste must have structural stability. A structurally tration. stable waste form will maintain its general physical. dimensions and form under the expected disposal condi-Table 1, Footnote 4 tions and factors such as the presence of moisture and D-55-12 (a) The term "significant gamma radiation" should be defined. microbial activity, and internal factors such as ... ! D-55-4 fb) How is radium treated? A value should be established. D-56-11 Rationale: The requirement of withstanding a compressive load of 50 psi (more than 7,000 pounds per square foot) appears to be a very rigorous C) The footnotes place a restriction on wastes containing chelating agents in concentrations greater than 0.1%. Is this limit intended to be 0.1% by weight or volume? This limit is too low, many agents were developed to decontaminate piping and equipment to reduce radiation levels to workers. A restriction on the solidified product of 0.1% might cause utilities to not use them because of the restriction on disposal and then let radiation Boading requirement and is above that available from many soils. If it is still felt that a numerical value is necessary, then consider one of these approaches: Make the compressive load requirement for the waste when buried no more than that of the surrounding soils at the site under consideration. levels rise. Evaluate a structural approach recognizing that the waste is constrained by surrounding soil and other wastes. This could conceivably reduce the 29. PAGE 38097 - PARA. 61.58(a)(1) compressive load requirement by a tenfold magnitude and still have an It is suggested that this paragraph be changed to read: adequate safety factor. "...and of the Department of Transcortation set forth in Similarly, requiring that the waste retain its form within 5% under load-49CFR Parts 171-179, as applicable. In the case of unoackaged ing is a very rigorous requirement and using the second potential approach (bulk) shipments, these must meet the requirements of 49CFR173.392." listed above, could be eased or eliminated. Rationale: The proposed regulations should provide for the shipment of The 5% limitation on physical waste form is too restrictive for stable, solidified, structurally strong waste handling. If applied to the drum-container-liner outer dimensions rather than to the solidified waste bulk (unpackaged) wastes under conditions that comply with Department of Transportation requirements for such wastes and that the wastes can meet itself. In this case, drums-containers-liners which are typically filled to 80% to avoid spills/splashing during the filling procedureprovide handling appurtenances and a clean surface for filling-transportationthe procosed Part 51 criteria when disposed of at the burial facility. A requirement on packaging would serve no useful purpose under these -circumstances and should not be imposed. burial actions. Once in the ground, these may be breached or oxidized in 30. PAGE 38097 - PARA. 51.55(2)(7) time, leaving the contained solidified and stable waste without effect on safety to the public or environs. Suggest that this paragraph be changed to read 32. PAGE 33099 - PARA. 61.56(b)(2) -"...that does not significantly exceed atmosphere at 20 degress C." "Non-corrosive liquid" should perhaps be changed to "liquid". If the intent of this article is to minimize corrosivity it should say so instead PAGE 38093 - PARA. 61.56(b) of setting arbitrary limits. Is the radioactivity in the liquid of any "Stability for 150 years" needs to be modified to indicate what forms of proof are acceptable. Some metal, wooden, and concrete structures concern? Is there intent to limit the amount of "clean" water in the container? 10

.

723 ------

23

-11 -13

B-387

33. PAGE 33098 - PARA. 61.56(b)(3) -

It is suggested that this paragraph be deleted.

Rationale: This is a very subjective statement and open to varying degrees of interpretation. Does this mean that filler material must be added to packages containing irregularly shaped solid objects? Or is it the intent of this article that all such objects should somehow be chopped, melted, or otherwise compacted? What forms are acceptable, i.e., ash, pellets, compressed trash? The goal of reducing the void spaces in a waste package is desirable and will be attained because of economic incentive independent of regulations. independent of regulations.

(34. PAGE\_38098 - PARA. 61.57 -

D-56-10

D-57-1

These labeling requirements should be expanded, clarified, and made more specific. G - 1 S35. PAGE 38100 - PARA. 61.82

Eliminate radioactive waste already disposed of and covered from NRC inspection requirement.



Nuclear Diagnostic Laboratories, Inc.

Serving Laboratories, Hospitals & Industry

· Port Office Box 791 Peekskill, New York 10566 914 - 737 - 7330

162 JEH 25 P2:43

108

January 19, 1982

Secretary of the Commission United States Nuclear Regulatory Commission Washington, D.C. 20555

## Attention: Docketing & Services Branch

Gentlemen:

This is to further comment on the proposed changes to Part 20, as published in the Federal Register, Volume 46, No. 142, Licensing Requirements for Land Disposal of Radicactive Waste.

NDL is a radioactive waste disposal service (collector). We pick up prepackaged various-sized containers from some 150 customers. When an NDL serviceman goes in to pick up waste containers from a customer, the drums are logged on an NDL Radioactive Manifest Record. The Manifest Record has spaces for twenty-five containers.

When the serviceman returns to NDL's facility, the various containers are unloaded into our storage area and separated according to the following waste categories and sizes:

M-45	5 Gallon(D) (Dry Waste)	Destined	for	land	Durial	in	South Carolina
	5 Gallon (L) (Absorbed Waste)	Destined	for	land	burial	in	Washington/Nevada
	<pre>\$ Gallon (S) (Scintillation Wasta)</pre>	Destined	for	land	burial	in	Nashington/Navada
	5 Gallon (R) (Biological Waste)	Destined	for	land	burial	in	South Carolina
(	5 Gallon (W) (Dry Waste)	Destined	for	land	burial	źn	Washington/Nevada
	/						

125. 173 EN

Secretary of the Commission USNRC Page Two

M-4

30 Gallon (D) (Dry Waste)	Destined for land burial in South Carolina
30 Gallon (W) (Dry Waste)	Destined for land burial in Washington/Nevada
30 Gallon (S) (Scintillation Waste)	Destined for land burial in Washington/Nevada
30 Gallon (L) (Absorbed Liquid)	Destined for land burial in Washington/Nevada
55 Gallon (D) (Drv Waste)	Destined for land burial in South Carolina
55 Gallon (W) (Dry Waste)	Destined for land burial in Washington/Nevada
55 Gallon (S) (Scintillation Waste)	Destined for land burial in Washington/Neveda
55 Gallon (L) (Absorbed Liquid)	Destined for Land Durial in Washington/Nevada
55 Gallon (V) (Vials non-scintillation)	Destined for land burial in Washington/Nevada
55 Gallon (R) (Biological Waste)	Destined for land burial in South Carolina
83 Sallon (R) (Biological Waste)	Destined for land burial in Jouth Carolina

n

Conceivably we could make a pick up from one customer with every one of the above containers listed on one Radipactive Hamifest Record. And because of the four container sizes and the three separate destinations for the different types of wastes; it is also conceivable that the above sixteen containers would be shipped in sixteen separate trailer-load shipments to the three burial sites within the course of a year.

With the above in mind one may understand that the rule, proposing to include copies of the originating generator manifests of all containers along with the new burial trailer-load shipment manifest, would be the creation of additional paperwork in the extreme; and entirely unnecessary, inasmuch as every container on a trailer-load shipment is on file at NDL.

As a concrete example, on January 8, 1992, this company shipped 1,931 five-gallon containers, one ten-gallon container, fifteen thirtygallon containers and eight fifty-five gallon containers in one trailerload to the Richland, Washington site. These containers represent Secretary of the Commission USNRC Page Three

pick ups from a number of customers. If the proposed rule was in effect at the time this particular shipment was made, the paperwork would, to put it mildly, be indeed voluminous. As a burial site employee aptly put it "You would need a van to travel behind the shipment, just to carry the paperwork".

. .

D

We cannot emphasize enough the severe burden this rule would place upon this company and others with similar operations. In addition we find this rule highly impractical and totally unnecessary inasmuch as all data can readily be obtained from the files. It is merely duplicating paperwork.

With our current practice every container is assigned a unique number and is readily identifiable. When a container is picked up, its unique number with the other necessary data is logged on our manifest. When that drum is loaded for shipment to an authorized burial site, that same information, including the unique identifying number, is transferred to the shipment manifest, while the originating customer manifest is kept on permanent file at NDL. Should someone have need of any additional information regarding a particular container (which is unlikely for the disposal manifests are quite complete), it can be readily made available.

We thank you for your consideration of the above. If you have any questions or desire clarification, please contact me or Alan Jones.

Very truly yours, Pastore President

PJP:sz



.

SARAH T. MORROW, M.D., M.P.H.

SECRETARY

A-2

D-53-1

-----

TELEPHONE AC 919/737-3313/2301

1 1 TASO PULS PREZ de

46 FR 32021

CONCET NUMBER

## STATE OF NORTH CAROLINA C2 JEI 26 P1:31 DEPARTMENT OF HUMAN RESOURCES The North Carolina Radiation Protection Commission :

JAMES B. HUNT JA.

CINALBALAN AAYMOND L. MURRAY, PLD. Remish VICE-SHARLAN JAMES & WATSON, JR., MLD. Ghasal Pret MICHAEL SFIC GILSTRAP Graansta PATRICK W. HOWE Sala-en JANICE KEENE MS. A.T. Chapel Hut STERAS A. WATTESCH. D.D.S. Charal Hull POSENT W. MICCHNELL M.D Greanwille" A WEELET CLOMAN, O.P.M. 3-10-1122/8 CHARLES PATTOR LA. C.C. A134mPA VARCH P SANDERS, Ph.O. Juran

January 21, 1982

Secretary of The Commission U.S. Nuclear Regulatory Commission Attention: Docketing and Service Branch Washington, D.C. 20555

P. O. EOX 12200

Gentlemen:

The North Carolina Radiation Protection Commission is responsible for promulgating, adopting, amending and repealing statewide regulations governing the possession, use and disposal of radiation sources in North Carolina. Since North Carolina is an Agreement State of the N R C, we have a particular interest in your proposed licensing requirements for land disposal of radioactive wastes published in the Federal Register, Vol. 46, No. 142, Friday, July 24, 1981.

RALEIGH 27605

We are cognizant of the importance of having adequate radioactive waste disposal facilities available and of establishing, for protection of the public and the environment, adequate regulations and criteria for the siting, construction, operation and closure of such facilities. In this regard, we believe that the above referenced proposed rules include considerable and commendable improvements over earlier drafts. At the same time, I have attached for your consideration a number of comments and suggestions for change which are intended to further improve your proposed rules. It is requested that the Nuclear Regulatory Commission give these comments and suggestions favorable consideration despite this technically late submission due to recent adverse weather conditions in our area.

We appreciate having the opportunity to provide our comments on these proposed rules and would support your efforts toward their further improvement and early adoption.

Sincerely yours,

Raymond I Mussay

Raymond L. Murray, Ph.D. Chairman

Attachment

Asknowledged by eard. 1/27/8.2.1.1

Comments on N R C Proposed 10 CFR Part 61, Licensing Requirements for Lano Disposal of Radioactive Waste, as published in the Federal Register, Vol. 46, No. 142, Friday, July 24, 1981 -

1. On page 38037 in item H, Other Considerations, the Nuclear Regulatory Commission (N R C) expresses its intent for all radioactive waste disposal facilities to comply with the provisions of Part 61, even though such facilities may be licensed by one of the 26 Agreement States. This statement implies that all aspects of Part 61, both administrative and technical provisions, would have to be adopted by Agreement States.

From the perspective of N R C responsibility under the Atomic Energy Act, it does not seem either necessary or important for most administrative provisions to be uniformly adopted by each Agreement State. It is agreed that most, if not all, of the technical requirements related to protection of public health and the environment should be imposed by such states. Accordingly, it is recommended that the stated intent in item H should be elaborated more fully in order to clarify its impost upon the preromatives which will be left to the Agreement States

clarify its impact upon the prerogatives which will be left to the Agreement States. We feel that the N R C should leave as much latitude as possible to Agreement States in their implementation of programs for siting, licensing and regulating disposal facilities, consistent with N R C responsibilities under the Atomic Energy Act. In so doing, the N R C would, to some extent, allow individual Agreement State programs to minimize the rather major burdens which will be imposed upon them by Part 61.

2. It is recommended that the 12-month preoperational monitoring requirement in Part 61.53(a) be revised to require that the applicant shall have implemented such preoperational monitoring by the time a licensee application is submitted. The current language would have the effect of arbitrarily adding 12 months to the length of time required to site, build and initiate operation of a new disposal site.

Since Part 61.3(b) already prohibits construction until after license issuance and N R C review will probably take about 18 months, it seems clear that the purpose of 12-month preoperational monitoring would not be compromised by the suggested change. Considering the important January 1986 exclusionary date in the National Low Level Radioactive Waste Policy Act and the probable four to five years required for opening a new facility under the proposed Part 61, it would seem justified and prudent to shorten the process by up to 12 months through this change.

3. We endorse in principle the concept of a waste classification system asproposed by the Nuclear Regulatory Commission. This is a significant step in the right direction; however, the three-class approach in Part 61.55 does not address those important wastes lying at the two extremes of the Class A, B and C system.

First, the proposed concentration limit for Class C wastes does place "higher D-55-5 level" wastes in the position of not being suitable for near surface disposal. Such wastes must be addressed at the earliest possible date and not left hanging as is presently the case. Dilution or volume expansion to make wastes disposable at near surface burial sites under Part 61 appears undesirable and counterproductive, as well as inconsistent with the N R C position which favors volume reduction. Second, the proposal does not address the admittedly difficult question of concentrations of radioactivity.in some wastes that they should not be considered as radioactive for the purpose of disposal. Such wastes should not be required by regulations to consume valuable space in a near surface burial facility.

(Page 1)

Comments on N R C Proposed 10 CFR 61 -Page two

D-55-8

In this regard we support the N R C's stated intention to systematically D-55-22 develop "de minimis" classifications for specific radionuclides and waste streams; however, this process should be accelerated as much as possible and not allowed to "bog down".

While we endorse the N R C policy of encouraging volume reduction, we are concerned that the proposed Part 61 does not appear to include any incentives 4. for volume reduction. In many respects the proposed rules may provide disincentives

D-55-B \for example: - excluding wastes with higher activity than the limits for Class C from near surface burial sites, without providing a viable alternative for disposal, may encourage dilution in order to gain access to such

D-55-3 = arbitrarily, limiting alpha particle emitting transuranic radionuclides to 10 nGi per gram, thereby discouraging volume reduction in order to ensure an available disposal option; and - the added expense and complexity for disposal of wastes as Class C

rather than B, or as Class B rather than A, and the lack of disposal options for higher level wastes than Class C will have the effect of discouraging volume reduction when it would raise waste from a lower to a higher class.

While part of this dilemma may be unavoidable, it does support the concept D-56-Nof carefully restudying the more expensive criteria applied to higher level wastes D-56-96 to make sure that they are necessary (e.g. the 5% dimensional tolerance and 50 psi compressive load requirements in Part 61.56 and the limit of 10 nCi per Gram for alpha particle emitting transuranics without regard to actual environmental Sum dealth risks).

5. Table 1 of proposed Part 61.55 includes a multitude of footnotes which have major impact upon interpretation of values in the table. The brevity of the footnotes and their interrelationships result in deficient clarity. Since the ED-1 footnotes and their interrelationships result in devide of major importance, it is subjects of the footnotes are regulatory in nature and of major importance, it is recommended that they should be expanded and clarified, possibly in the form of

( regulatory text rather than footnotes.

 $\begin{array}{c} \textbf{U}_{\text{regulatory text rather than routholes.}}\\ \text{With respect to these footnotes, questions remain as to}\\ \textbf{D}_{\text{respect}} = \textbf{A}_{\text{respect}} = \textbf{A}_{\text{r$ 

. \*

Tor various containers and to routine analytical techniques for evaluating container contents; D-56-2 {wastes containing specific chelating agents which do not promote migration through the soil; and the implication that the N R C might approve special additional limits for an individual licensee; hence raising a question with respect to justification of special treatment of one licensee over another.

6. The language of proposed Parts 61.55 and 61.56(a) could be construed to require elaborate and sometimes impossible assay and analysis (chemical and radio-logical) of every individual container of waste. Such an interpretation could be constructed by the second sec make it impracticable to impossible for many generators to adequately declare their wastes for disposal at a near surface burial site. 

Comments on N R C Proposed 10 CFR 61 -Page three

It is suggested that the N R C proposal be revised to provide for detailed D-55-62 licensee characterization (chemical and radiological) of their radioactive waste )streams and development of more practicable routine assay procedures for individual containers of waste from each waste stream.

> The N R C's stated concept of employing a 5-year license issuance with potential requirements for public hearings on each renewal may well have the effect of discouraging private industry from risking entry into the waste disposal business. This could result in failure to establish needed new regional burial sites.

It is recommended that the N R C consider a concept of issuing long-term, life-of-site licenses and reserve subsequent public hearing for requested license amendments which would have the effect of expanding the disposal site or the license authorizations. Since the N R C already proposes to require licensees to keep their applications fully current, make various reports to the N R C and obtain prior approval in specific instances, and the N R C will have modification, revocation, suspension and civil penalty authority; the five-year expiration, public hearing and renewal process appears to serve no purpose other than to arbitrarily place the facility's license in jeopardy.

As referenced by the N R C on page 38083 under "Protection of the Environment," the E P A has not yet fulfilled its responsibility to prepare a standard that will

C-Z set limits for releases of radioactivity to the general environment from disposal facilities. This has led the N R C to necessarily develop proposed rules based upon other existing standards in anticipation of what the E P A may eventually do. I upon other existing standards in anticipation of what the E P A may eventually of It is unfortunate that the "cart has had to be put ahead of the horse". We would urge the E P A to assign high priority to meeting its important responsi-bilities in this area at the earliest possible time.

------:12 1

9. The word "orderly" in Part 61.52(a)(4) implies careful stacking in a rectangular array. Since this would probably result in higher worker radiation exposure inconsistent with ALARA principles and less efficient use of disposal ひーらえース units (trenches), it is suggested that the N R C consider replacing the words an orderly" in Part 61.52(a)(4) with the word "a". It would appear that this change should not compromise this rule's intent of "maintaining package integrity during emplacement and disposal" and would reduce worker exposure, promote efficient site utilization and lower disposal cost.

> 10. We recognize the difficulty in using regulatory language which is always specific and subject to uniform literal interpretation; however, it is recommended that the proposed Part 61 be carefully reviewed giving special attention to terms

- such as: significant, prevent, eliminate, few and adequate. For example: "significant" in Part 61.50(a)(4), referring to natural resources, does not lend itself to literal interpretation and would be subject to a wider range of individual options;
  - "prevent" in Part 61.51(a)(4) would appear to imply "zero" or "none" when perhaps "minimize to some low level" may be the intention; - "eliminate" in Part 61.51(a)(6) also appears to imply "zero";

1

- "a few percent" in Part 61.52(a)(6) would result in a wide range of defensible individual opinions; and
- similarly, "adequate" in Part 51.52(a)(9) is not directly interpretable and would be largely a matter of opinion.

B-391

. •.

GEN-1

New England Nuclear

# Comments on N R C Proposed 10 CFR 61 - Page four

GEN-I To the maximum extent feasible, use of such terms should be minimized. For example, in Part 61.52(a)(9) it might be more pala table to require the licensee to carry out the closure and stabilization procedures specified in his application and approved by the Nuclear Regulatory Commission. There are adequate provisions elsewhere to require revision of procedures, correction of problems, et cetera, when previously approved procedures later prove to be indequate.

Raymond Lillarray

Raymond L. Murray, Ph.D., Chairman North Carolina Radiation Protection Commission January 21, 1982

## 132 JU 27 02:56

46PR5

6:

JULKET INJUSER

PROPOSED RULE

BROFOSED RULE PIN

January 21, 1982

R. Dale Smith, Chief Low-Level Waste Licensing Branch Jivision of Waste Management Nuclear Regulatory Commission Washington, D. C. 20555

Dear Mr. Smith:

I enclose this letter to alert you to the fact that New England Nuclear chaired the NELRAD 10 CFR 61 Subcommittee and was actively involved in the development of the detailed comments. As such New England Nuclear endorses the report and recommends due consideration be given to it.

Sincerely,

NEW ENGLAND NUCLEAR slum C. Brantley J.

VVP. Administration, NENC

JCB/da

Association

549 Albany Street, Boston, Massachusetts 02118 Telephone 617-482-9595 Telex 94-0996

# NELELD

182 JE 27 92 E

Jamuary 21, 1982

7. Dale Smith, Chief low-Level Maste Licensing Branch Livision of Waste Management Suclear Regulatory Commission Vashington, D. C. 20555

lear Mr. Smith:

I enclose comments on 10 CFR Part 61 and the accompanying Draft Environzental Impact Statement. These comments are written from the generator's rerspective, compiled from safety officer input from several waste genersting firms in New England.

These collective comments are a product of NELRAD activity. NELRAD is a consortium of New England firms and institutions who use radioactive interials and have a common need for a nearby low-level radioactive waste iisposal facility. Our group was formally organized in 1981 to support the efforts of the six New England states in complying with the Low-Level the efforts of the six and Tradicactive Naste Policy Act.

We are pleased to be a part of the process that allows review of proposed regulations. In general, we approve the intent of 10 CFR Part 61 and encourage expeditious progress.

Sincerely,

in in L. Dullat

/Jamis D. Stelluto Ixecutive Director, SELRAD

.35/da Inc.

2. 0. BOX 1267 CONCORD, MA 01742-1267 Tel. (617) 371-0358

то:	NELRAD Committee
FROM:	NELRAD 10 CFR 61 Subcommittee
SUBJECT:	Comments on Proposed Rulemaking on Land Disposal of
	Low-Level Radioactive Waste.
DATE:	1/20/82

The enclosed report presents detailed comments on 10 CFR 51, the Draft Environmental Impact Statement (NUREG-0782) and the accompanying summary of the proposed rule. The detailed comments are preceded by a summary of the main points.

Members of the subcommittee which prepared this report are:

F.N. Brenneman (to Dec. 1981)	Northeast Utilitles P.O. Box 270 Hartford, CT 06101
M. Galanik	77 Main St. Room 208, 238, MIT Cambridge, MA 02139
D. Gomer	Nuclear Metals, Inc. 2229 Main St. Concord, MA 01742
L.R. Smith (Chairperson)	New England Nuclear Corp, DuPont 549 Albany St. Boston, MA 02118
J. Stelluto	NELRAD

549 Albany St. Boston, MA 02118

Yankes Atomic Electric Company 1671 Worcester Rd. Framingham, MA 01702

E. Tarnuzzer

Thanks are due to K. Bennert, J.D. Bernardy, J.C. Brantley and C.8. Killian for their helpful comments and to K. Thomas for processing this report.

L.R. Smith
#### SUMMARY OF COMMENTS ON THE PROPOSED RULEMAKING ON LAND DISPOSAL OF LOW-LEVEL RADIOACTIVE WASTE: PREPARED BY THE NELRAD 10 CFR 61 SUBCOMMITTEE.

#### 10 CFR 16 Scope.

We believe that the development of performance standards in 10 CFR 61 is the best approach to establishing licensing requirements for land-disposal of low-level radioactive waste. We agree that only essential generic prescriptive requirements should be included in the regulations and all site specific requirements should be incorporated in individual facility licenses.

#### Site Licenses and Inspection.

The scope and sequence of activities in establishing safe operations and ensuring proper closure of a facility appear realistic. We recommend that safeguards be strengthened by:

granting disposal facilities a full term license with appropriate review instead of subjecting a license to the public hearing process every five years.

assigning a full-time NRC inspector to each LLW site during the operational phase.

encouraging active monitoring and review of site records by state authorities.

# Probability of Inadvertent Intrusion.

The method used to establish generic prescriptive requirements to protect inadvertent intruders is appropriate for estimating the lower boundaries of concentration limits. However, these limits are unnecessarily conservative because the probability of intruders encountering radioactivity has not been factored into the calculations.

We believe that better estimates of maximum permissable concentrations can be made if the following considerations are included in the calculations:-

- The probability of inadvertent intruders encountering critical waste forms.
- The concentration of radicactivity in waste sent to a site exhibits ь. a log-normal distribution with an average concentration at least an order of magnitude lower than the maximum permissable concentration.

If these probability factors are included in the calculations, maximum permissable radioactivity concentrations are expected to be at least an order of magnitude higher than those presented in 10 CFR 61 and will still provide sufficient protection to the inadvertent intruder.

#### Site Selection and Utilization.

The site could be better utilized if credit was given for the decay of short lived radionuclides during the operational period.

It is clear that local resources and demographic developments are important considerations when selecting a suitable site. However, demographic predictions can be unreliable therefore we recommend that the NRC consider zoning requirements to restrict activities that may adversely affect the site Lhydrology and environment.

The potential impact of changes in natural radioactivity in ground water due to site excavations was not conspicutusly covered in the DEIS. We recommend that these effects be considered.

### Clarification of the Intent of 10 CFR 61.

We have indicated several instances where the intent of the regulations is not clear. The following improvements are recommended:



ED-

- replace absolute statements by achieveable practical ones.
- ь. use units and terminology recommended by scientific standard setting organizations (eg. ICRU and ICRP).
- clarify performance objectives by specifying internal and external c. dose equivalent limits to individual crgans as suggested by the ICRP.

#### Waste Concentration.

Waste generators are concerned that the difficulty in accurately assaying radicactivity in individual containers will cause overly conservative values to be assigned to shipments resulting in poor utilization of the site. we recommend that the NRC consider relaxing concentration limits on individual containers and accept inventory methods designating average concentrations in waste shipments. Using inventory averaging methods would also enhance the generator's ability to determine if waste concentrations

twere below "de minimis", levels. The establishment of "de minimis" levels.

D-55-2 for radionuclides and waste forms should be encouraged for better site utilization.

The 100 Ci per container limit appears excassively conservative. We recommend that DOT limits be adopted since the most restrictive potential impact scenario appears to be individual exposure from accidential breach of containment during transportation to the site.

## Use of ALARA Concept.

In several instances the ALARA concept is improperly used to justify excessive restriction. We recommend that optimum levels be defined at which an operation could be described to be #LARA. Imposing further restrictions yielding small benefits at great cost is not ALARA. For example, D-55-3 the proposed 10 nCi/g limit for TRU contaminated waste is not ALARA

Simply because industry has complied with this regulation. The EPA has D-55-3 suggested that 100 nCl/g is an appropriately conservative limit. We recommend that the EPA's suggestion be adopted unless a better limit is derived. · . · .

• ,

3

# Manifest Tracking.

M-1

We recommend that a reasonable procedure be developed to integrate. enforcement agencies into the control or supervision of the manifest tracking systems.

. . . Conclusion,

In general, we recommend that 10 CFR 61 should not duplicate existing regulations but should reference them and be compatible with them.

Our final conclusions is that 10 CFR 61 will provide a reasonable and necessary regulatory frame-work for low-level radioactive waste disposal. We submit these comments in the hope that they shall improve both safety and cost effectiveness.

\_ ·

#### SPECIFIC COMMENTS ON THE PROPOSED RULEMAKING ON LAND DISPOSAL OF LOW-LEVEL RADIOACTIVE WASTE: PREPARED BY THE NELRAD 10 CFR 61 SUBCOMMITTEE

.

. . Page 38084, Col. 2. . (\*

This section specifies that waste stability should be sufficient to  $\mathcal{D}$ -55 -12 ensure that the residual radioactivity is "no longer of concern from the migration standpoint". We recommend that the quoted phrase be replaced by appropriate dose limits.

2. /Page 38084, Col. 3.

ь.

The 10 nCl/g limit for transuranic waste is too conservative. The EPA suggests that 100 nCi/g is a conservative limit. Overly conservative limits may dissuade waste generators from practicing volume reduction of waste potentially contaminated with TRU.

Although industry has been able to comply with the 10 nCi/g limit for TRU waste we do not agree that this limit is ALARA. On page 7-13 of the DEIS and in several other instances the ALARA concept is misapplied to justify excessive restrictions. Operations should only be said to be ALARA when the cost to reduce impacts from these operations is justified by the benefits accrued and when further costs to reduce impacts are not justified. Compliance with an excessive restriction or achieving a lower level of impact are not necessarily ALARA." Reduction of environmental impacts 2-3 orders of magnitude below comparable impacts from other conventional industries is not reasonable and therefore not ALARA.

Page 38085, Ccl. 2.

The establishment of de minimis levels for other waste streams and radionuclides should be encouraged since this should lead to improved utilization of disposal sites.

Page 38087, Col. 2.

Disposal sites should be provided full term license with appropriate review. The financial planning necessary for long term site monitoring assumes a reasonable operating life. Hence the license should not be subject to the public hearing process every five years with the possibility that renewal may not occur.

Page 38087, Col. 2.

"Keep people off the site" should be replaced by "control access to the site" to allow maintenance, surveillance and other appropriate activities. ۰.

B-5

```
In the following sections the words "assurance" and "assure"
should be replaced by "ensurance" and "ensure" respectively. This
is to indicate that positive action should be taken to achieve an
objective rather than merely persuading that an objective can be
achieved.
```

```
ED-14
```

C-1

GEN-1

 Page 38089, Col. 3, line 5, 61.2.

 Page 38090, Col. 1, line 51, 61.2.

 Page 38091, Col. 1, line 14, 61.7 (b), (3).

 Page 38093, Col. 3, 61.23 (b), (c), (d).

 Page 38094, Col. 1, 61.23 (b), (c), (d).

 Page 38095, Col. 1, 61.23 (b).

 Page 38095, Col. 2, 61.30 (a), (2), (5).

 Page 38095, Col. 2, 61.50 (a), (1).

 Page 38096, Col. 2, 61.51 (a), (2).

 Page 38097, Col. 2, 61.55 (b), (c), (1).

 Page 38097, Col. 3, 61.55 (b).

```
    Page 38090, Col. 2, line 16.
    Page 38091, Col. 1, line 3, 61.7 (a), (1)
```

Instead of "15-20" meters a single value should be used. "15-20" may be confused as meaning burial below 15 meters and above 20. meters from the ground surface.

- 8. Page 38090, 61.7 (a), (1).
  - a. We agree that both performance objectives and prescriptive requirements are necessary. Performance objectives should be limited to occupational and environmental impact concentrations and should be specified in the regulations.

b. Generic prescriptive requirements are appropriate to limit LLW concentrations and to protect inadvartent intruders. These and prescriptive requirements which provide financial surety should also be incorporated in the regulations. In the cases where prescriptive requirements are adopted from other existing or proposed regulations these should be referenced or incorporated in 10 CFR 61.

C. Other prescriptive requirements which limit site inventory or which protect against excessive migration of radionuclides are site specific and should be incorporated in site licenses. 10 CFR 61 should specify that site licenses will incorporate these site specific prescriptive requirements.

9. Page 38091, Col. 1, 51.7 (b), (1).

- a. We recommend that the primary objectives for disposal of LLW are: To isolate LLW from the biosphere in a manner that maintains:
  - personnel dose equivalent commitments ALARA;
     ii. environmental impact and personnel dose equivalent commitments below specified limits.

"Prevention of migration of radionuclides" is an important strategy for achieving those primary safety objectives. "Prevention of exposure to inadvertent intruders" is a special case of (a) (ii) and should be called a secondary objective. Page 38091, Col. 1, 61.7 (b), (2). Omit "eliminated or" since it is not possible to reduce water access to zero. This paragraph should be written more clearly. In particular it should be clear when "stability" refers to trench structure or the waste itself. The generic term "radionuclide" should replace "isotope" and "radioistope" in the following sections: Page 38091, Col. 2, line 1, 61.7 (b), (2). Page 38097, Table 1 - 61.55 (a), (1). Page 38097, Col. 2, 12. Page 38091, Col. 2, 61.7 (b), (3). Replace "would" by "could" since intruder risk has a statistical basis. 13. Page 38091, Col. 2, 61.7 (5), (4). The first sentence is ambiguous. It may be rewritten thus: a., "Institutional control of access to the site is required for at least 100 years after closure of the burial site: Page 4-49 of the DEIS indicates user consensus of opinion expects EDъ. that the institutional control period may reasonably range from 100 to-300 years. Since this parameter is somewhat arbitrary it should be the last parameter selected in the equation for determining prescriptive requirements. 14. Page 38091, Col. 3, 61.7 (c), (1). "Established administrative procedures" should be cross referenced to enable recognition of specific procedures. 15. Page 38093 Col. 1, 61.13 (b). "Demonstration" should be replaced by "reasonable indication" since it is not possible to demonstrate the achievement of performance objectives until long after a site has been closed. Page 38094, Col. 1, 61.23 (a). 16.

Replace "should" by "to".

$$\begin{aligned} & for the set of the set of$$

;

•

.

.

.

.

.

э

product of a large number of conservative, unreasonably conservative even if individual mildly conservative. A better method for ers is to use the best estimates of each parameter ertainty errors to generate upper and lower ries. A simplified version of this approach values for each parameter (eg. as used in the is preferable to compounding conservative should be given for improving critical wasta ant uptake in the intruder-agriculture scenario. I in the DEIS which indicate that the average intration in waste can be expected to be from maximum concentration. Hence concentration elaxed by at least one order of magnitude and dequate intruder protection.

i u

mits are to be included in the regulations we velopment of concentration limits for other compounds. However a "generic nonsite-specific system" will be too conservative. Waste normally only need to consider one site to ley should normally only need to be conversant ion system specific to that site. A site specific system should allow optimum site utilization.

difficulty in economically assaying waste to plies to the conditions of a particular category.

- to generators assigning conservative estimates centrations and consequential under utilization а.
- of regulatory authorities to assay containers lers control by assay unenforcable.
- factors recommended to simplify waste analysis icable to industries making a wide range of cts.

. .

th and safety is being referred to.

er limits appear excessively conservative and he DEIS.

- For waste disposed as received the DOT limits should apply 1. since airborne release and non-occupational exposure is the controlling factor.
- п. For waste processed on site the limits for individual containers should be 10 times the DOT limit since it is occupational exposure which provides the limiting scenario.

# tan tanàn amin'ny tanàna 🛄 kaominina mpikambana

38. Page 38097, 61.56 (a), (5).
Add "normally" to read "waste must not contain or be normally capable of generating..." This is to ensure that plastics with high ignition points but which are capable of emiting toxic burn products are not unnecessively excluded from the waste.
39. Page 38097, 61.56 (a), (7).

a. Add "significantly" to read: "... at a pressure that does not significantly exceed one atmosphere at 20°C." If this sentence is not modified, waste generators may be constrained to packaging gases under reduced pressure or elevated temperatures.

D-56-5

It is not clear whether the 100 Ci limit applies only to gases. This should be clarified. 100 Ci limit per package is too restrictive for certain radioactive gases. eg.  $^{14}CO_2$  and  $^{3}H$ .

Although there is provision for exceptions to the proposed limits on a case by case basis, calculations should be included to show the impact expected from radioactive gas. Also the wide range in toxicity of labeled compounds should be addressed in the DEIS.

40. Page 38098, 61.56 (b), (1).

The intent of "within 5%" should be clarified.

41. Page 38098, 61.56 (b), (2).

"Non corrosive liquid" should be defined as it is in the DEIS, ie., "pH between 4 and 10 and incapable of significant galvanize and chemical reaction".

42. Page 38098, 61.58



Provision should be made to allow waste generators to categorize waste by an inventory process. The quantities of waste generated in a year or present in an individual shipment could be determined with greater accuracy than by making separate determinations for individual containers. This comment is particularly relevant to very low contamination levels and radionuclides which are restricted by the ground water migration scenario.

Scenarios assume that all waste is placed just before the site was closed. However in practice waste will accululate over a 20 to 60 year period and a considerable fraction shall have decayed before site closure. Hence a relaxation in concentration limits can be applied to short lived radionuclides received during the initial period of burial activities.

12

# 43. Page 38100, 61.82.

- We recommend that a full-time NRC inspector or agreement state agency inspector be assigned to a LLW site during the operational phase.
- b. The State authorities should be encouraged to munitor the disposal site and review site records.

44. Page 38102, 20.311

From the language in the proposed ruls, it is not clear how enforcement agencies would be involved. One possible procedure would require the site operator to return a receipted copy of the manifest system to the generator vice merely notifying him of receipt. If the generator were then required to maintain a file of all shipment manifests and backrouted receipts, the enforcement agency could check for compliance at each generator's place of business. Such a system closes the loop on the process and allows one to readily check for compliance during a regular facility inspection. It has the further advantage of not burdening the enforcement agency with volumes of manifests. Additionally, generators could reserve making payment to shippers until the backroute is received. This would provide a strong economic incentive to comply with the manifest tracking system.

In any event, some reasonable procedure should be developed to integrate enforcement agencies into the control or supervision of the manifest tracking system.

• •\* \* • •

M-



Covernor

D-55-13

D-52-1

TO:

ALAN I

Barbara Ritchie Page 2 January 13, 1982

DEPARTMENT OF SOCIAL AND HEALTH SERVICES

STATE OF WASHINGTON

Olympia, Washington 98504

January 13, 1982

Barbara Ritchie Department of Ecology PV-11

FRCM: AMANCY P. Kirner, Supervisor Radioactive Materials Unit Radiation Control Section

SUBJECT: COMMENTS ON PROPOSED 10 CFR 61 ENVIRONMENTAL IMPACT STATEMENT

The environmental impact statement for 10 CFR 61, "Licensing Requirements for Land Disposal of Radioactive Maste," appears to be well written and it adequately supports appropriate regulation of a radioactive waste disposal site having more than 10 inches of rain each year. The environmental impact statement and its proposed regulation, however, fail to accurately address realistic concerns and place realistic conditions on the operation of a radioactive waste disposal site at an arid location. Among the major problems of an arid site are slumping and wind erosion. Nore emphasis should be placed on mitigating these two impacts, absent infiltration of ground and surface water.

While it is conceded that the stability of the waste form enhances safety, the strong reliance placed upon scenarios involving groundwater infiltration appears inappropriate for an arid site such as the low-level waste disposal site located near Richland, Washington. Without the liklihood of groundwater or surface water transport of radionuclides, segregation of class A and class B wastes seems to be unnecessary when weighed against the burden of operating separate disposal units. For the arid site, a case can even be made to allow. co-mingling of waste classes in an attempt to lower the average concontration of the most hazardous wastes. The same logic can likewise be used for class C wastes at an arid site, provided class C wastes are placed at deeper depths and solidified in'a relatively leachfree matrix or otherwise segregated for their hazardous lifetime. The proposed regulation and its supporting environmental impact statement do not appear, however, to make a convincing enough argument for the establishment of a third disposal unit to handle only class C

 A second sec second sec D-55-13 (wastes at an arid site. Washington does not believe that the segre-D-52-1 (gation of waste by classification should become a matter of compatibility.

> Since the draft environmental impact statement on 10 CFR 61 was written, the state of Washington has increased its perpetual care and maintenance funding. Descriptions of Washington's funding mechanisms should be revised on pages 9-6 and K-50 to reflect these recent changes as follows:

- a. The Perpetual Care and Maintenance (PCSM) Fund contribution by U S Ecology, currently at \$0.275 per cubic foot, will increase to \$1.75 per cubic foot. This renegotiated contribution will provide a total fund of approximately 36 million by July 1, 1985. The total accumulation will be determined by the actual volume of waste disposed.
- b. In addition, US Ecology will contribute \$0.25 per cubic foot of waste disposed to a newly established Contingency Closure Fund. A contribution at this rate will yield a fund of \$300,000 by July 1935, assuring that the state of Mashington will be able to adequately close the radioactive waste disposal operations conducted at the site if and when needed.
- c. On Jaunary 15, 1982, the company will post a surety bond in the amount of \$500,000 for a period of one year to protect the state if the company should leave the site without meeting closure conditions as stated in the license.

Thank you for the opportunity to incorporate our concerns.

NPK/db



Morris, R. 62450 (815) \$42-2820

Papers Annon B. D. ZIELS

Program

MOTOTS P W JANNE A. M. HOGLUND

Publications T. W. EDWARDS

S. R. SHELTON

Secretary WILLIAM E. McCOWN Westinghouse manford Ce. P C. Bea 1970, W/E-23 Tichuand, WA 99352 28) 376-4203

CHINISTRATIVE COMMITTEE CHAIRMEN

The American Society of Mechanical Engineers CZ FE3 -1 P Bring somes **Auclear Engineering Division** (Founded in 1954) (FOUTGED MT 1934) EXECUTIVE COMMITTEE Charmen Nichard D, Riefe Public Fornes Sector and Gas Company 80 Per Roce Homere, No Official (201) 430-4029 M. M. Thomas Black & Vestch P. O. Box 8405 Kansas City, HO 64114 Vice Cherman REMCO P. WASSINK Westinghouse Electric Corporation P.O. Bes 18218 Tanos - FL 3366 (513) 837-7203 Subject: Proposed Rule 10 CFR 61 Supplemental Comments Prepared by Radwasta Systems Committee BURTON O. ZIELS Dumond Power Speciality Corporation P O. Box 415 Loncaster, Ort 43130 (814) 653-6540, aut 370 CHEFR 38081 Secretary of the Commission K. W. McMillAN Bechie Power Corporation Pity Beard Street P O. Ben 3965 San Francisca, CA \$4118 (415) 758-0567 U. S. Nuclear Regulatory Commission Washington, D.C. 20555 Attention: Docketing and Service Branch BRENT & SHELTON Commonwealth Edison Dreston Nuclear Power Station Dear Sir:

The Radwaste Systems Committee of the Nuclear Engineering Division of ASME recently submitted comments on the proposed Rule 10 CFR 61. Subsequent to the preparation of that submittal, additional comments were received. We request that these additional comments to accepted by the Commission as input to their deliberations on the final rula.

Member + American Association of Engineering Societies + Accreditation Board for Engineering and Technology

Both sets of comments have been submitted to ASME for review and approval, but neither as yet represents the formal position of ASHE. They do, however, represent the considered opinion of a large number of individuals intimately involved in the area of low-level radioactive waste.

Acknowledged by ars. A/4/82. Ant

Sincerely, Mahones

Michael M. Thomas, Chairman Codes and Standards Subcommittee of the Radvaste Systems Committee

Supplemental Review Comments of Proposed Rulemaking on Land Disposal of Low-Level Radioactive Waste - 10 CFR 61 . •

Submitted by

Codes and Standards Subcommittee Radvaste Systems Committee Nuclear Engineering Division of American Society of Mechanical Engineers

bg Attachment

Nuclear Heat Exchanger S. M. CHO Mechanisms & Procion Someong T. L. ERION

UAISON REPRESENTATIVES

R. P. WASZINK (IG Power Ceeartmen L.C. OTEN (IO American Nuclear Society) B. M. HCGLUND (IO American Power

Conterences W. SCHULTHEIS (to Nusiver Power Codes & Standards Committee)

OPERATING COMMITTEE CHAIRMEN

Pressure Vessel & Piping M. D. BERNSTEIN

And waste Sustama A. REITER

Advanced Reactors W. J. ROWAN

Mant Svalama M. M. DeLONG

B-402

SUFPLEMENTAL COMMENTS ON PROPOSED AMEROENTS TO 10 CFR Parts 2, 19, 20, 21, 30, 40, 50, 51, 61, 70, 73, AND 170 LICENSING REQUIREMENTS FOR LAND DISPOSAL OF RADIOACTIVE WASTE

# General Comments

D-56-14

spproach for disposal of low level radioactive wasta espoused by many groups and other government agencies.
While some of the technical requirements (e.g., limits or concerntrations and quantities of various isotopes) may prove to be a problem in the future. The NRC's approach is generally compatible with current utility and disposal site operating practices. The proposed regulation seems appropriate in light of some past problems with burial sites and the lack of consistent practices and requirements disposal sites.

1. The proposed regulation is consistent with the philosophy and recommended

2. How will 10 CFR 61 affect NRC Branch Technical Position 11.3 regarding mandatory solidification?

GEW-1 {3. Are there acceptance criteria which verify prescribed criteria after completion of the facility?

ED-12 The terms "short term" and "long term" should be defined clearly when used in 10 CFR 61.

GEN-1 S. The draft 10 CFR 61 regulations require a "quality assurance" program without establishing what program to follow. There are no National standards on such a program. A "quality control" program probably is more appropriate and could be implemented at less cost.

6. The draft 10 GFR 61 regulations discourage volume reduction of radioactive wastes. This conflicts with a recent NRC policy statement which promotes volume reduction methods.

Specific Comments

1 21. Page 38083, first paragraph, second column - The wording "coming in contact with the waste" should be changed to "becoming exposed to the wastes."

2. Page 38087, last paragraph, second column - The NRC should define what requirements are necessary for a period of postclosure observation and maintenance. This would preclude opening this issue up in a licensing effort and prevent the need for such a plan.

**1** 

Page 38090, last paragraph in column 1 - This paragraph rigidly defines the "Land disposal facility" associated with the disposal of radioactive wastes "into the subsurface of the land." This should be rewritten to include aboveground, engineered structures and should be discussed in section 6.1.7(a)l of the same chapter.

P-52-4 Continuing into the next paragraph - Reference should be made relative to the licensing of "onsite storage" facilities.

4. Page 38091, the first sentence, column 2 - It is not clear why for D-55-10 based on the characteristics of the disposal site.

D-59-2 In the same column, the second paragraph states that "institutional control is relied on for periods up to 100 years ....." Row was this amount of time arrived at?

5. Page 18091, first paragraph, last column - The last sentence of this paragraph refers to a "final review of licensee's site closure and stabilization plan." It is not clear if an initial review was made or when it was made. Without establishing a plan initially, how will the public ever accept that the licensee will meet the criteria. This should be discussed in detail in part (c) of this section, "The Licensing Process."

6. Page 38092, Section 61.11(c)(4), second column - This sentence is not clear as to whether the plans are relative to before, during, or after usesge as a land disposal facility.

 Page 38092, fourth paragraph, third column - The quality assurance program description requirements should be discussed in further detail. Reference should be made to where it is documented that such a statement exists.

8. Page 38093, third paragraph, first column - What is meant by the term "mass wasting?"

9. Page 38095, Section 61.42, third column - It is not clear as to what time frame should be applied to the inadvertent intrusion necessary to evaluate the 500 mR/year.

10 Page 38097, Section 61.55, Table 1 - This table serves no purpose other than to possibly restrict the utilities' radwaste disposal efforts without consideration to individual waste inputs. This table should be deleted and/or revised to represent typical comprehensive radwaste inputs which possibly have some deviations from the radioisotopic concentration ranges given in Table 1.

Other reviewers commented that most of the isotopes listed are primarily beta-emitters which would be difficult or impossible to detect with

B-1

D-55-6



B-404

#### Secretary of the Commission

 | { regulatory policy should also be coordinated with BIA's Office of Trust Responsibilities with respect to locating, licensing, operating and maintaining commercially operated disposal sites on Indian lands. Because of variations and changes in Indian land ownership, BIA's involvement will be essential.

Specific comments on the proposed rule and the environmental impact statement are attached separately.

We hope these comments will help you in the preparation of a final statement.

Sincerely ruce Blanchard, Director **Environmental Project Review** 

21

Enclosures

# PROPOSED RULE

The following more specific comments are addressed to the rule itself but would also apply to corresponding sections of the Summary and the environmental statement.

# Section 61.2, Definitions

D-52-3 Buffer Zone. The buffer zone definition should include depth as well as lateral boundaries, and should be described as a three-dimensional zone. The performance standards might then apply to releases beyond the boundary of the buffer zone. Unrestricted use of land and resources beyond the buffer zone (laterally or at depth) would then be allowable during and after site operations.

## 61.50 Disposal Site Sultability Requirements for Land Disposal

D-50-1 (a)(2) We suggest being more specific in the "modeling" requirements. Do you mean physical scale model? numerical ground water/solute transport model? conceptual model? (There are many kinds of possible models.)

 $D-5v - \lambda$  {(a)(7) We endorse this option but suggest using "molecular diffusion" in place of "diffusing"  $D-5v - \lambda$  and/or defining maximum hydraulic conductivity allowable such as 10-<sup>6</sup> cm/sec.

## 61.51 Disposal Site Design for Land Disposal

GEN- [] (a)(4) It is impossible to totally "prevent" infiltration; suggest using "minimize" in place of "prevent."

(aX6) This requirement appears inconsistent with 61.50(aX7). The option of disposing in the saturated zone should be mentioned again.

2

### 51.52 Land Disposal Facility Operation and Disposal Site Closure

D-52-3 (aX8) We believe the location of the buffer zone should be determined on the basis of site performance. The zone ideally would be enclosed within a three-dimensional surface surrounding and underlying the burial site. Our concept of the buffer zone is a zone that provides a controlled/restricted-access volume of earth material around and under the site, beyond which unrestricted use of land and resources, surface or subsurface, could be allowed during and after site operation. The 100 foot lateral extent listed in the rule appears somewhat arbitrary.

## (51.53 Environmental Monitoring

D-53-[ (a) We believe that "geochemistry" should be listed with the other subjects (ecology, meteorology, climate, hydrology, etc.). Although geochemistry is often an implied aspect of hydrology, we believe it deserves specific mention because it plays such an important part in radionuclide mobility in ground water.

# 61.55 Waste Classification

D-55-9 [General: We believe waste should be classified according to total toxicity as described above under "General Comments."

A-8 { (d) It is not clear what the disposition will be of wastes which exceed Class C concentrations. What type of disposal is envisioned by the Commission for those wastes?

# 6156 Weste Characteristics

D-55-72 General: The non-radiological toxicity of the waste needs to be considered here, we believe. It is apparently ignored.





CFFICE OF SECRET ... DOCKETHING & SERVICE BRANCH

Comments of Comments Karr-McGee Corporation TOPE Kerr-McGee Nuclear Corporation and Kerr-McGee Chemical Corporation on Proposed 10 CFR Part 61 (46 Fed. Reg. 38081 (July 24, 1981))

DOCNET NUMBER 2202 SUCCEST RELEASEN PE

January, 1982

Acknowlad and by card . 1/26/17.61

B-406

2

LOVINGTON & BURLING		KERR-MCGEE CORPORATION		
721894048 (202) 682-6000	P. O. BOX 7586 WASHINGTON, D. C. 20044	785; 710 833-0005 (C0 #84) 10152; 80-553 (C01,140 #54)	ENVEROMMENT AND MEAL TH MANAGEWONT DRIVENOM	UCOUT WOMER DO
	January 18, 1982 62 JN 22 P	4:22	January 14, 1982 CERTIFIED MAIL, RETURN RECZ	FORTER RULE <u>FR-20tal</u> (46 FR 380 LIFT REQUESTED
Secretary of Nuclear Regul Washington, I Attention: 1	the Commission Latory Commission D.C. 20555 Docketing and Service Branch	146 P. 1 170	Secretary of the Commission U.S. Nuclear Regulatory Com Washington, DC 20555	Dio
Re Gentlemen:	900011 No. Proposed 10 CFR Part 61 PROPOSED	RULE PR-20tal. (46 FR 3808	Rs: Proposed Rule 10 CFR P Licensing Requirements	art 61 for Land Disposal of Radios

Enclosed please find Kerr-McGee's comments on the above-referenced rulemaking proceeding. It is our under-standing (from a telephone call to Mr. Smith's office) that the comments will be considered timely so long as postmarked today (Monday) in view of the storm that paralyzed the D.C. area (where the comments were finalized) last week. 19 - 191 - 1<u>1</u> 1.1

. •

۰, Very truly yours,

. . .

Charles H. Montance Attorney for Kerr-McGee

3 - - - -

~

.

cc: Mr. R. Dale Smith (w/encls.)

: • ۰. . • .

. . . .. Asknowledged by conflictence and and

52, 5°2° υ... JXX 22 P4:22 F2072150 2389 URN RECEIPT REQUESTED

10 CFR Part 61 frements for Land Disposal of Radioactive Waste

# Dear Sir:

The Nuclear Regulatory Commission (NRC) recently published an invitation for comments in connection with the proposed Rule 10 CFR Part 61 concerning Licensing Requirements for Shallow Land Disposal of Radioactive Waste. (See 46 Federal Register 38081, July 24, 1981.) The attached comments of Kerr-McGee are in response to that invitation.

Sincerely,

W. J. Shelley, Vice President Nuclear Licensing and Regulation

bs/bec . ... Enclosure

. .

. . . . . .

• .

#### Comments on Proposed 10 CFR Part 61

These comments, submitted on behalf of Kerr-McGee Corporation, Kerr-McGee Nuclear Corporation, and Kerr-McGee Chemical Corporation (Kerr-McGee), are addressed the Licensing Requirements for Land Disposal of Radioactive Waste, proposed by the Nuclear Regulatory Commission (NRC) at 46 Fed. Reg. 38081 (July 24, 1981). A number of the standards proposed in 10 CFR Part 61 are premature and beyond the authority of NRC. In addition, many of the standards and other requirements are unreasonably stringent and are not supported by the rationale offered in the underlying documentation. Kerr-McGee's specific objections are detailed below.

 The Exposure and Emission Standards Proposed for Part 51 Are Premature and Beyond the Agency's Authority to the Extent that They Are Not Already Embodied in 10 CFR Part 20

transferred to the Administrator of the Environmental Protection Agency (EPA) the authority under the Atomic Energy Act to set generally applicable radiation standards.<sup>1</sup> Assuming <u>arguendo</u> the validity of the Plan,<sup>1</sup> the EPA Administrator, not NRC, is

Reorganization Plan No. 3 of 1970 by its terms.

(-2)

C - 2 Fresponsible for issuing standards relating to exposure to ionizing radiation from a low level radioactive waste disposal site during and after its operation. NRC in fact appears to admit that EPA is responsible for issuing the pertinent standards. EPA, however, has not issued standards governing waste disposal of the sort involved here. In the absence of EPA standards, NRC is limited to the existing standards found in 10 CFR Part 20.\*

> The 10 CTR Part 20 regulations limit public exposure to 500 mrem per year, and specify stringent limits on the release of about two hundred radionuclides by air or water to unrestricted areas. A number of the standards specified in proposed Part 61, however, are not currently contained in 10 CTR Part 20. For example, proposed § 61.41 limits release of radionuclides such that no member of the public may be exposed to a dose of more than 25 mrem to the whole body, 75 mrem to the thyroid, and 25 mrem to any other organ. No such standard appears in 10 CTR Part 20 for waste disposal operations. NRC accordingly lacks authority to adopt it here. To take another example, neither the purported concentration limits for radionuclides in proposed § 61.41 nor the 10 pCi/1 uranium/thorium

5ee 46 Fed. Reg. 38083.

- 2 -

<sup>&</sup>lt;sup>1</sup> See section 2(a)(6) of the Plan, 35 Fed. Reg. 15623, 84 Stat. 2086, 42 U.S.C. § 4231 note.

Kerr-McGee has challenged, in a lawsuit currently pending in the Tenth Circuit, the validity of the plan insofar as it purports to transfer authority over radiation standards. This section of Kerr-McGee's comments apply in the event that NRC is correct and that the validity of the plan is upheld.

With one exception, the Part 20 standards, insofar as they deal with exposure of the public, were issued by the Atomic Energy Commission prior to loss of standard-setting authority to EPA pursuant to Reorganization Plan No. 3. The one exception is that portion of Part 20 incorporating EPA's 10 CFR Part 190.

standard is contained in any regulation under 10 CFR Part 20. It is therefore invalid as beyond NRC's power. All portions of proposed Part 61 predicated on these invalid standards must be withdrawn pending final action by EPA on pertinent standards.

- 1 -

II. Various Standards and Requirements Specified in Proposed Part 61 Are Unduly Stringent and Unsupported.

(-2

6-3

A. The 25 mrem Rule Set Forth in Proposed § 61.41 Is Unsupported

Proposed § 61.41 limits release of radioactive material from a low level waste facility such that the release will "not result in an annual dose exceeding an equivalent of 25 millirems to the whole body, 75 millirems to the thyroid, and 25 millirems to any other organ of any member of the public." This proposed limit appears to be derived from EPA's 40 CFR Part 190. This derivation is incorrect for a number of reasons. First, it is incorrect for a purely legal reason: EPA specifically excluded operations at waste disposal sites from Part

190's coverage." NRC may not promulgate its own standard in the absence of action by EFA.

Second, the 25 mrem standard specified in the regulation is too low. 25 mrem is a mere one-rwentieth the level (500 mrem) specified in 10 CFR i 20.105 for releases to unrestricted areas.<sup>4</sup> There is no evidence that complicance

5ee also 25 Fed. Reg. 4402 (May 19, 1960) (Radiation Protection Guidance for Federal Agencies).

. .

with the SCO mrem standard has resulted in any untoward results. The 25 mrem restriction is demonstrably too low for another reason. Average natural background radiation (cosmic, terrestrial, and internal body radiation) varies from state to state in this country between 90 mrem and 180 mrem average exposure per year. Variation is caused primarily by different altitudes above sea level and by natural rock formations. Living near a granite rock formation, for example, may result in 25 to 100 mrem addition exposure per year. 25 mrem is thus only a small fraction of natural background exposures. 25 mrem is also only about half the exposure incurred in an ordinary chest x-ray.' Epidemiological evidence unanimously indicates that increased exposures discernible adverse health effects.' Leading

· .

6-3

<sup>40</sup> CER § 190 02(b).

See Biological Effects of Radiation, 15 Encyclopedia Britannica 382 (1979) (gives 50 mrem for chest x-ray and 100-150 mrem for natural background); <u>Low-Level</u> ionizing Radiation, <u>Hearings before the Subpommittees on Energy Research and Production and Natural Resources and Environment of the House Committee on Science and Technology</u>, 96th Cong., 1st Sess., at 3-9 (1979).

See, e.g., Hickey, et al. Low level Ionizing Radiation and <u>Human Mortality: Multi-Regional Encemiological Studies</u>,
 40 Health Physics 625 (1961) (Exhibit A); Frigerio, et al.,
 The Argonne Radiological Impact Program (ARIP)-1. Carcinogenic Hazard from Low-level, Low-rate Radiation (Argonne Nat'1 Lab. Report. ANL/IS-26, Part 1) (1973) (Exhibit 3); High Background Radiation Research Group (China), <u>Health Surrey in High</u> Background Radiation Areas in China, 209 Science 877 (1980) (Exhibit C); Gopal-Ayengar, et al., <u>Evaluation of the Long Term</u> <u>Effects of High Background Radiation in Selected Population</u> <u>Groups on the Kerala Coast in Peaceful Uses of Atomic Energy</u>, pp. 31-51 (1971) (Exhibit D); Cullen, et al., <u>Dosimetric And</u> <u>Cytogenetic Studies in Brazilian Areas of High Natural Natural</u> <u>Actuvity</u>, 19 Health Physics 165 (1970) (Exhibit 2).

- 5 -

authorities believe that low level radiation is not hazardous. As Dr. Evans recently testified, "say we double the background radiation [to] us, what will that do to us? The answer is nothing . . . [T]here are no radiobiological effects."

Third, the 25 mram restriction deviates in a significant way from that specified in EFA's 40 CFR Part 190. EFA excluded radon and its daughters from Fart 190's coverage. The proposed NRC regulation does not contain a similar exclusion. It is possible that certain radioactive ors residues (a.g., some uranium or thorium wastes and some rare earth wastes) may be disposed in low level repositories covered by proposed Part 51. In order to attain a 25 mram standard as applied to radon, it may be necessary to cover the wastes in question with many meters of earth cover. The benefits from this action are speculative and slight. Secause of the importance of the issue, the purported risk from radon is discussed in datail below.

6-3

**...** 

NRC has elsewhere indicated that the primary hazard posed by disposal of uranium mill wastes is emanation of radon-222, a radioactive, inert gas. NRC in the past has attempted to vindicate stringent radon controls on the basis of (1) risks to persons living at some distance from the

Testimony of Dr. Robley Evans before the New Mexico EI3, reprinted in Uranium Ore Residues Hearings. Potential Harard Disposition, Hearings before the Procurement and Military Nuclear Systems Subcomm. of the House Comm. on Armed Services, 97th Cong., 1st Sess., at 456 (1981). See also T. Suckey, Radiation Hormasis (1980). - 6 -

uranium wastes in question and (2) risks to persons living near the wastes. Neither justification has merit.

Radon-222 emanation from uranium wastes does not pose a significant risk to persons living outside the nearvicinity of the wastes in question. The amount of radon likely to result from the wastes is miniscule compared to the amount of radon released naturally from soils in the United States and from such accepted economic activities as agriculture.<sup>14</sup> Moreover, increased radon from uranium wastes cannot be detected more than a short distance from such wastes, even in a downwind direction.<sup>11</sup> The effect of the wastes is wastly overborne by natural discharges.

> Stringent controls for radon emanation similarly cannot be justified on the basis of risk to nearby residents. NRC acknowledges that there are no discernible adverse health effects from exposure to radon from uranium wastes.<sup>11</sup> In the

See. e.g., Transcript of Bearing before the New Mexico EI3 at 461-62 & 470 (testimony of Dr. Evans) reprinted in Uranium Ore Residues: Potential Hazards and Disposition. Bearings before the Procurement and Military Nuclear Systems Subcomm. of the House Armed Service Comm., 97th Cong., lat Sess. 453 (June 24 & 25, 1981).

<sup>&</sup>lt;sup>11</sup> Set. E.G., Shearer & Sill, <u>Evaluation of Atmospheric</u> <u>Radiation in the Vicinity of Uranium Mili Tailings</u>, 17 Eesith Physics 77 (1969) (Attachment F); Letter, Greenleigh (DOE) to Selander (EPA); dated July 15, 1981, at 2 (DOE comments on EPA inactive site standards) (Attachment G); Uranium Ore Residues Hearings, <u>subra</u>, at 457; L27 Cong. Rec. S 12984 (daily ed. Nov. 5, 1981).

<sup>&</sup>lt;sup>11</sup> NRC, Generic EIS on Uranium Milling (GEIS) at A-35 ("We (NRC] know of no data or studies which indicate definitively that health effects do or do not occur at the low levels of exposure that are anticipated to result from operation of uranium mills").

absence of data, NRC apparently assumes that some health effects may occur on the basis of the "linear non-threshold model." This model has been aptly criticized by sumerous authorities. For example, the model is based on the assumption that tumor induction is a straightforward "one-hit" process in which any unit of exposure will result in some carcinogenic activity. But "the bulk of the evidence" argues against the hypothesis that neoplastic transformation is a linear function of dose.13 Additionally, many prominent health physicists and other experts believe that the body is capable of repairing damage caused by low-level radiation." Moreover, the linear non-threshold model is contrary to direct evidence that "scologically realistic, low-level radiation is biologically stimulatory, and presumably beneficial."15 This effect, known as "radiation hormesis," is a restatement of the Arndt-Schulz law that small doses of substances are stimulatory (i.e., beneficial) even if large doses are harmful. As one biochemist recently explained,

- 7 -

"[r]adiation hormesis denies the validity of straight line extrapolation from known harmful

13 Anderson, Pathology 347 (7th ed. 1977).

6-3

See, e.g., Robbins & Cotran, Pathologic Basis of Disease 552 (2d ed. 1979); Transcript of New Mexico IIB hearing at 495-96 (testimony of Dr. Evans); In the Matter of Duke Power Co. (Perkins Nuclear Station, Units 1, 2 and 3), 8 NRC 87. [1975-78 Transfer Binder] Nuclear Reg. Rep. (CCE) 1 30,312 at p. 23,669 (1978).

<sup>13</sup> Eickey, <u>Letter: Cancer and Concensus</u>, Chemical & Engineering News 65 & 75 (Sept. 14, 1981) (Attachment E).

. . .

a la ser a altre de la

doses to zero. The argument that low doses give harmful effects in proportion to dosage is invalid."<sup>16</sup> Finally, pertinent epidemiological studies are consistent with the proposition that low-level radiation is not harmful. Indeed, those studies indicate that people in high background radiation areas enjoy <u>better</u> health than people in low background radiation areas.<sup>17</sup> For example, people living on the Colorado plateau in the United States have significantly lower death rates from cancer and chronic disease that people in the

- 8 -

<sup>14</sup> T. Luckey, Radiation Hormesis 162 (CRC 1980). <u>See also</u> Luckey, <u>Latter: Hormesis</u>, Nuclear News 48 (Dec. 1981); Bickey, <u>Letter: Hormesis</u>, Nuclear News, at p. 54 (Dec. 1981). In any event, even 12 carcinogenesis is linearly related to dose, there may also be a second beneficial effect at low doses, and these benefits may cutweigh the risks. Such a relationship would appear graphically as follows:

'n



See Sagan, Some Thoughts on Dose-Response, Hormesis and All That, Nuclear News, at p. 82 (Cct. 1981).

<sup>17</sup> See, e.g.; Eickey, et al, Low Level Ionizing Radiation and Ruman Mortality: Multi-Ragional Epicenological Studies, 40 Health Physics 525 (1981); Frigerio, et al, The Argone Radiological Impact Program (AGT9)-1. Carcinegenic Eazard from Low-level, Low-rate Radiation (Argone Nat'l Lab. Report ANL/25-25, Part 1) (1973); High Background Radiation Research Group (China), Health Survey in High Background Radiation Areas in China, 209 Science 877 (1980); Gopal-Ayengar, et al, <u>Evaluation of Long-Term Effects of High Background Radia-</u> tion on Selected Population Groups on the Kerala Coast in Feaceful Uses of Atomic Energy, Vol. II, Proc. 43th Int. Conf. Peaceful Uses of Atomic Energy, pp. 31-51 (1971).

the state of the s

C-3

eastern costal plains where natural ionizing radiation is much lower. Another striking example involves the people of Kerala,

- 9 -

India.

(-3

"The people of Kerala are reported to have the highest literary rate and the best health status in India; yet their expenditure on health care is not appreciably above average, and they have the lowest food intake and the least adequate diets of people of all states of India. A partial explanation for this paradox is that radiation from the unusually high radium and thorium in costal and river rocks exposes many residents to 10 times more terrestrial radiation than the U.S. average."

NRC admits that the linear nonthreshold model may be erroneous.<sup>13</sup> In fact, as one statistician has charged, the purported general regulatory "acceptance" of the model is "in large part [a] misuse of statistics.<sup>434</sup> Exclusive or even primary reliance on the linear non-threshold model is an over-simplication and may result in an over-investment in controls to avert hypothetical risks.<sup>31</sup> This translates into

-

In the words of Professor David Okrant of UCLA, "Resources for the reduction of risks to the public are not infinite. At some point, a greater improvement in health and safety is to be expected from a more stable and viable economy than from a reduction in pollution or the rate of accidents." Okrent, <u>Comment on Societal Risk</u>, 208 Science 372, 374 (1980). NRC should also recognize that stringent controls aimed at reducing exposure to radiation from atomic energy activities can easily have a reverse affect. For example, new and costly NRC controls will result in increased energy costs to consumers.

(footmote cont'd)

- 10 -

an unvarranted burden on atomic energy activities and a waste of society's resources. NRC should re-evaluate and modulate its reliance on the linear non-threshold model for purposes of assessing the risk from exposure to low-level radiation. In particular, NRC should give greater weight to the possibility (indeed, likelihood) that the ionizing radiation, particularly from naturally occurring concentrations of radionuclides such as are involved here, is not harmful and may in fact be beneficial.

Even assuming that the linear non-threshold model is

appropriate, the risk which it projects in the situations

# (fostnote cont'd)

This will induce increased reliance on insulation. This means fewer air changes in residences and businesses. This in turn leads to increased exposure to naturally occuring indeor radon. EZA estimates (under the linear non-threshold model) that the risk posed by indoor radon is about 1 in 300 lifetime (approximately 1 in 21,CCO per year). IPA, Draft EIS for Remedial Action Standards for Inactive Uranium Processing Sites at pp. 4-20 & 21. This means that, under government estimates, about 10,000 people perish per year from natural indoor radon exposure in this country alone. This is far more than the maximum number of deaths (6) projected by NRC from radon emanating from three times the amount of uranium mill tailings now in existence. Increased insulation, prompted by rising energy costs, already is leading to about 1,000 additional hypothetical deaths from indbor radon per year and will eventually lead to about 10,000 additional hypothetical deaths. Cohen, <u>Health Effects of Radon from Insulation of Buildings</u>, 39 Health Physics 937, 940 (1980). Less costly uranium fuel cycle regulatory requirements can make a major contribution in decreasing hypothetical radiation fatalities in this country because less costly regulation will result in less expensive electricity. This in turn will lead to less reliance on conservation measures which are 100 to 1000 times more hazardous radiologically speaking than radon releases from the uranium fuel cycle.

C-3

<sup>&</sup>lt;sup>19</sup> Luckey, <u>Letter: Hormesis</u>, Nuclear Nevs 52 & 54 (Jec. 1981) (emphasis added).

<sup>&</sup>quot; GEIS U-4.

<sup>&</sup>quot; Hickey, subra note 15.

- 11 -

involved here is insubstantial. NRC calculates that the maximum risk posed by radon-222 from even totally <u>unregulated</u> mills is 1 in 70,000,000 for three times the number of mills now in existence.<sup>11</sup> This maximum risk, which is only hypothesized to exist anyway, is insignificant and de minimis. NRC has admitted that even if it actually eventuates, it is "about equal" only to the risks posed by "a few puffs on a cigarette, a few sips of wine, driving the family car about 5 blocks, flying about 2 miles, canceing for 3 seconds, or being a man aged 60 for 11 seconds.<sup>211</sup> Indeed, the risk perceived by NRC is far less than many risks commonly and ordinarily accepted in our society.<sup>14</sup>

38. See, e.g., CZIS at 19.

(-3)

- <sup>23</sup> 46 Fed. Reg. 15167 (March 4, 1981). <u>See also</u> 127 Cong. Rec. S 12984 (daily ed. Nov. 5, 1981).
- The following table sets forth many commonly and ordinarily accepted risks in our society.

Cause	Individual risk/year
smoker	1/300
agricultural employment	1/1,700
motor vehicle - Total (1975)	1/4,500
air pollution - sulphates	1/6,700
covernment employment	1/9,100
truck driving employment	1/10,000
falls	1/13,000
alcohol	1/20,000
living for one year downstream from & da	am 1/20,000
motor vehicle - pedestrian (1975)	1/25,000
drowning (from recreational activities)	1/53,000
inhalation and ingestion of objects	1/71,000
home accidents (1975)	1/93,000
hicycling	1/100.000
nerson in room with smoker	1/100,000
one pint of milk per day (aflatoxin)	1/100,000

. .

(footnote cont'd)

- 12 -

Kerr-McGee understands that NRC staff may take the position that this <u>de minimis</u> risk is not dispositive because the risk to people living next to, or on top of, uranium wastes, is greater. To Kerr-McGee s knowledge, NRC has not provided a detailed quantification of this alleged risk for public review. However, the risk -- which is purely hypothetical to begin with -- cannot be large in comparison to many customarily accepted risks. This view is strongly supported by the Commission's admission that radon exposures at the edge of uranium wastes stabilized in accordance with the agency's now suspended Uranium Mill Licensing Requirements will result in exposures which are only "a small fraction of any reasonable health protection limit."<sup>21</sup> Moreover, purported fears for persons living atop such wastes are groundless in view of requirements such as these in proposed Part 51 and in the Uranium

(footmote cont'd)

. . . . .

Cause

• • •

. . . . .

1-3

Individual risk/year
Individual risk/year

accidental poisoning - solids and liquids	1/170,000
electrocution	1/200,000
vaccination for small pox (per occasion)	1/330,000
air travel - one transcontinental flight/yr	1/330,000

Source: CSEA Testimony of Professor Richard Wilson (Ixhibit I) reprinted in Eutt, <u>Unresolved Issues in Conflict</u> Between Individual Freedom and Covernment Control of Food Safety, 33 FDC L.J. 558, 564-66 and 568 (1378).

<sup>15</sup> GEIS at p. 12-15 explains that "[e]xposures as close in as a fencepost near the edge of the pile would be about 1.1 x 10-4 WL above background levels. . ., which is a small fraction of any reasonable individual health protection limit (1% of the Surgeon General's guidelines)."

- 13 -

**`**:

Mill Tailings Radiation Control Act, that ownership of the disposal site generally be transferred to the government for actual tailings.<sup>14</sup> Chviously, Congress presumed that the government can and will keep people from occupying the disposal area.<sup>27</sup>

NRC's risk estimates under the linear non-threshold model, which already predict insubstantial hazards, are in fact excessive because the agency has relied on erroneously high risk estimators. The latest study by a panel of eminent scientists from EPA, the Department of Energy (DOE), Germany England and Canada indicates that the <u>maximum</u> hypothetical risk from radon-222 can be no greater than 1/3 that employed by NRC per unit exposure and <u>may in fact be zero</u>.<sup>24</sup> Moreover Professor Cohen, after analysis of risk estimates for radoninduced lung cancer employed in the BEIR-III report, concluded that the risk estimators advocated in that report overstated the risk from low-level exposure by a factor of twenty to

1-3

<sup>27</sup> NRC is authorized to waive government ownership requirements with respect to byproduct material as defined in section lle.(2) pursuant to 42 U.S.C. § 2113(b). Government ownership is unnecessary for purposes of assuring that people do not intrude on properly disposed uranium and thorium wasts. Deed covenants barring such intrusion are sufficient for this purpose. NRC's Branch Technical Position, published at 46 Fed. Reg. 52501 (Oct. 23, 1981), properly recognizes this fact in adopting a policy of relying on deed covenants in options.3 and 4.

<sup>11</sup> Evans, et al., <u>Istimate of Risk from Environmental Exposure to Radon-222 and its Decay Products</u>, 390 Nature 98 (March 12, 1981) (attachment J). Significantly, DOE has expressly endorsed reliance on the study by Ivans, et al., for standard-setting purposes. Uranium Cre Residues Hearings, supra, at 176.

forty.<sup>13</sup> To the extent that NRC purports to rely on a risk of exposure through an injestion pathway, that risk estimate is similarly excessive. Professor Evans has specifically criticized risk estimates for injestion of low-levels of radium as unduly conservative and as unsupported by the evidence.<sup>34</sup>

- 14 -

The 25 mrem rule may also apply to thorium-bearing ores or thorium wastes. It is therefore pertiment to examine problems arising from application of the rule to such wastes. Thorium wastes do not pose a significant radom problem. The radom decay product (known as thoron) in the thorium decay chain has a much shorter half life (only 55.6 seconds) in comparison to radom-222 and accordingly cannot be expected to diffuse in sigmificant amounts through even nominal cover. It furthermore cannot be expected to travel any significant distance off-site. Moreover, the decay products of thoron, with one exception, are short-lived in comparison to decay products of radom-222.<sup>21</sup> The primary hazard NRC identifies with respect to thorium is direct gamma radiation.

See also Cohen, Failures and Critique of the BEIR-III Lung Cancer Risk Estimate, \_\_\_\_\_ Health Physics \_\_\_\_ (in publication) (attachment K).

(-3)

<sup>&</sup>lt;sup>14</sup> 42 U.S.C. § 2113(b).

See R. Evans, et al., in Radiobiology of Plutonium (Stover & Jee, eds. 1972) at pp. 431-68 (practical threshold for humans); Raube, et al., <u>Scne Cancer from Radium: Canine Dose</u> <u>Response Explains Data for Mice and Humans</u>, 208 Science 61 (1980).

<sup>&</sup>lt;sup>31</sup> The properties of the thorium decay series are described in Rundo, <u>The Radioactive Properties and Biological Behavior</u> of <sup>33</sup>Ra (Th X) and its Daughters, 35 Health Physics 13 (1978).

- 15 -

. . .

NRC's risk estimates for low-level camma radiation are even more suspect than NRC's risk estimates for exposure to low-levels of radon-222. NRC's estimates, insofar as they exist at all, are traceable to exclusive reliance on a linear non-threshold model. All the criticisms already made with respect to the linear non-threshold model apply with equal or . greater force. Indeed, the objections are even more telling, because the epidemiological studies in India. Srazil and China on high-background radiation were in thorium-rich areas. None of these epidemiological studies detected adverse effects. They are instead consistent with the view that higher exposure to natural radiation may be beneficial. Moreover, various advisory groups have acknowledged the lower likelihood of gamma radiation to result in harm. The ICRP has indicated that camma and beta radiation (termed "low-LET") is an order of magnitude less effective in doing biological damage than high-LET radiation.<sup>31</sup> The BEIR-III Committee have also admitted that the linear nonthreshold model "probably leads to overestimates of the risk of most cancer . . . for exposure to low-LET radiation at low doses." The BEIR-III, report in fact recommends the use of a range of estimates for the risk of such radiation, and notes that the risk may be zero. NRC

6-3

. . .

and ZPA have improperly relied exclusively on a linear nonthreshold model for projecting the risk from gamma radiation from thorium wastes.

- 15 -

In sum, the 25 mrem rule is unduly stringent as applied either to uranium or to thorium wastes. The costs for control of exposure to uranium or thorium wastes at a level of 25 mrem are totally out-of-line with the speculative benefits such control would produce. The American Mining Congress (AMC) has estimated that the cost to comply with a 25 mrem rule (excluding radon) at uranium mills would approach \$1,000,000,000 (one billion) per hypothetical health effect avert. The cost per health effect would obviously be even greater if the restriction applied to radon:<sup>36</sup> The ratio of low-level waste repositories would not be more favorable. NRC should exclude radon and its daughters from the 25 mrem standard proposed in § 61.41.

# The 10 pCi/1 Uranium/Thorium Standard Proposed in Section 51.41 Is Unduly Stringent

2

NRC's justification for the 10 pCi/l uranium and thorium limitation appears to be based on a Suggested Action Guidance (ERA-SAG) entitled "Advisory Opinion for Uranium" issued in draft form on July 15, 1981. This EPA document is labeled "do not quote or cite," indicating the extremely praliminary nature of its conclusions. A critique of the EPA

<sup>34</sup> Uranium Ore Residues Hearings, <u>supra</u>, at pp. 255-88 <u>et sec.</u> (hereby incorporated by reference).

<sup>&</sup>lt;sup>32</sup> International Commission on Radiological Protection, Report of Committee II on Permissible Dose for Internal Radiation (1959).

<sup>&</sup>lt;sup>33</sup> See National Academy of Science. The Effects on Population of Exposure to Low Levels of Ionizing Radiation 4 (1980) (SIER-III Report).

document is attached (Ixhibit L). The critique explains the deficiencies in the rationale for the 10 pCi/1 uranium limit and applies equally to that limit as proposed by NRC in § 61.41. The critique also applies to the proposed 10 pCi/1 thorium limit. The subject uranium/thorium limit should be deleted.

> C. The Provision Relating to Protection of Individuals from Inadvertent Intrusion Are Arbitrary and Unreasonable.

Proposed § 61.42 requires that a low-level radioactive wasts facility be designed, operated, and closed such that no "individual inadvertently intruding into the disposal site and occupying the site or contacting the waste after active institutional controls over the disposal site are removed, could receive a dose to the whole body in excess of 500 millirem per year." For purposes of proposed Part 61, NRC hypothesizes active institutional controls to last 100 years.

6-4<

Kerr-McGee objects to the restriction embodied in § 61.42. The proposed regulations provide that the low level waste repository must be located "on land owned in fee by the Federal or a State government." Proposed § 61.59(a). NRC's "inadvertent intruder" hypothesis is predicated on the notion that the government will somehow fail, allowing people to occupy the site of the low-level waste repository for dwellings and agricultural purposes. NRC's assumption that the government will fail is inconsistent with the concept of a national government and the firmly established principle that the United States is a Union of States established for perpetuity.<sup>34</sup> It is arbitrary and unreasonable for NRC to propose a standard based upon a notion antithetical to our Constitution. This is doubly the case since the proposed regulations authorize NRC to require a low-level waste repository licensee to provide funding for institutional controls of indefinite duration. Proposed § 61.63. Similar objections apply to NRC's argument in support of its preference against any disposal which may require long-term maintenance.

> D. The Concentration Limits for Alpha-Emitting Transuranic Isotopes Specified in Column 3 of Table 1 Are Unreasonably Low.

D-55-3 Table I to Part 61 proposes a 10 nCi/qm limit on alpha-emitting transuranic (TRU) waste for purposes of disposal as low-level waste. The 10 nCi/qm limit for TRU waste is totally arbitrary. It was not established on the basis of health effects

"A disruption of the federal Union heretofore only menaced, is now formidably attempted.

I hold, that in contemplation of universal law, and of the Constitution, the Union of these States is perpetual. Perpetuity is implied, if not expressed, in the fundamental law of all national governments. Is is safe to assert that no government proper, ever had a provision in its organic law for its own termination." IV <u>Collected Works of Abraham</u> <u>Lincoln</u> 254 (R.F. Sasler, ed., 1953).

- 17 -

•

<sup>&</sup>lt;sup>33</sup> Gur Constitution contains no provision for the dissolution of our government. <u>Marbury v. Madison</u>, 5 U.S. 1 Cranch) 137 (1903). As Chief Justice Marshall explained, "[t]he [Constitution's] principles . . so established, are deened fundamental. And as the authority from which they proceed is supreme, and can seldem act, they are designed to be permanent." <u>Id</u>. A Civil War was fought over this issue. As President Lincoln observed in his First Inaugural Address (March 4, 1861):

for hazards. Instead, it was "derived from the upper range of concentrations of radium-226 in the earth and is subject to modification based on long-term studies of nuclide migration in soil." AEC Manual, Chapter 511; at p. 51 (Sept. 19, 1973). Kerr-McGee is aware of no health or safety reason for maintaining the 10 nCi/gm TRU limit. The limit may readily be raised by at least a factor of ten without posing any serious health or safety harard under the conditions NRC applies to. Class C "intruder" waste.

- 19 -

D-55-3^

NRC's Tationale for retaining the limit at 10 nCi/gm is three-fold. First, NRC asserts that past experience has indicated that the 10 nCi/gm limit is "achievable." See Draft HIS on 10 CFR Part 61 at 7-13. The notion of "achievability" as used by NRC apparently means that NRC believes that vastes can be limited for low level disposal to 10 nCi/gm. The fact that 10 nCi/gm is "achievable" in the sense that some vastes meet that limit does not mean that the standard is defensible or desirable. To the contrary, 10 nCi/gm is neither defensible nor desirable. The bulk of the TRU waste in the hands of NRC licensees exceeds 10 nCi/gm in concentration. It cannot realistically be reduced in concentration. Moreover, most of this waste results from decommissioning activities." There are currently no facilities available for disposal of the TRU

See cenerally Rockwell Int'I (Rocky Flats), Transuranic (TRU) Waste Management Program: U.S. Department of Energy Acceptance of Commercial Transuranic Waste (Feb. 1980 (revised)) (describes amount of TRU waste held by NRC-Licensees).

. ...

• . •

vaste in question. This essentially precludes prompt decommissioning of the NRC-licensed facilities.<sup>37</sup> This is contrary to the public interest in timely decommissioning and is wasteful of resources. The 10 nCi/gm standard is therefore not "achievable" if viewed in the context of the TRU-waste disposal problem.

Second, NRC contends that the 10 nCi/gm TRU limit is warranted by radiation dose calculations. This contention is not supported in NRC's rationale. According to the Draft EIS, the 10 nCi/gm limit was supposedly derived on the basis of an intruder/construction scenario and an intruder/agriculture scenario. Assuming arguendo that intruder hypotheses are appropriate bases for setting concentration limits,<sup>31</sup> neither intruder scenario envisioned by NRC warrants the 10 nCi/gm limit on TRU waste suggested in proposed 10 CFR Part 61. Under the intruder/construction scenario, NRC assumed that institutional controls would fail one-hundred years after closure of the repository and that a person would erect a dwelling atop the disposal trenches, excavating three meters down from the surface in doing so. See Draft EIS on to 10 CFR Part 61 at G-58. Under the intruder/agriculture scenario, NRC

a second a second s

··· · · · ·

· · · · · · ·

D-55-3

- 20 -

<sup>&</sup>lt;sup>17</sup> Kerr-McGee estimates that it has 10,000 to 30,000 cubic feet of TRU waste from decommissioning activities at its former small mixed oxide plant in Cimarron, Cklahoma. This waste exceeds 10 nCl/gm in TRU concentration. No commercial facility is available for disposal of this waste. Kerr-McGee cannot complete decommissioning until a repository is available for this TRU waste.

- 21 -

.....

D-55-3

assumed that some of the soil excavated in the intruder/construction scenario is scattered about the ground around the house and that vecetables are grown in it. See id. at G-61. Neither of these scenarios would result in any exposure to TRU waste. Proposed section 61.52(a)(3) requires that Class C waste (which includes all TRU) be so disposed that the top of the waste is a minimum of 5 meters below the surface of the cover." (Emphasis added.) Even if three heters of soil were excavated for a house, two meters would of soil would remain separating the basement slab from the TRU. The construction intruder accordingly cannot come into contact with the TRU. Similarly, no TRU would be brought to the surface to result in exposure in the acticultural scenario. in sum. NRC's calculations of intruder exposure to TRU are in error because the agency failed to give credit for five meters of cover."

The agency's intruder acenario calculations are incorrect for an additional reason. NRC calculated airborne - 22 -

Tradioactive concentration for the intruder scenarios using Equation G-11, which can be represented by Tsa =  $X/C_3 = (Tsa)_0 (10/V) (5/30) (50/PE)^2$ where Tsa = ratio of airborne concentration to soil density, dimensionless.

X = average airborne concentration, Ci/m<sup>3</sup>.

Cs = average soil concentration,  $Ci/m^3$ .

(Tsa)<sub>0</sub> = reference airborne to soil radioactive concentration ratio, dimensionless.

 $= 2.53 \times 10^{-10}$ 

V = mean wind speed, m/sec.

= 3.61 m/sec.

10 = reference mean wind speed, m/sec.

S = Soil silt content, percent.

= SO

30 = reference-soil silt content, percent.

PE = Thornwaite's precipitation - evaporation index, dimensionless.

50 = reference 2E index, dimensionless.

Equation G-11 essentially states that the airborne radioactive concentration due to wind resuspension is inversely proportional to the mean wind speed and the square of the moisture content. However, if this were true, wind resuspension would increase as the mean wind speed decreases, which cannot be.

Wind resuspension source terms have been shown by field measurements to be proportional to the cube of the wind

D-55-3

<sup>&</sup>lt;sup>35</sup> The intruder scenarios envisioned by NRC are in any event unduly conservative. It is unreasonable to assume that a residence will be constructed atop a government owned repository and that people will occupy the site for a significant period of time. It is also important to note that NRC cannot alter the intruder scenarios so that exposure to TRU waste disposed in accordance with the regulations can be realistically projected. Under the proposed regulations the repository will be sited, among other things, so as to be protected from a 100 year flood and the waste will be buried. Under these circumstances, the principal means of erosion will be sheet erosion. Many thousands of years will be required to erode five meters of cover under the applicable universal soil loss equation. See Conservation Agronomy Technical Notes, Note No. 231 (June 9, 1981) (Exhibit M).

- 23 -

speed above a threshold value and inversely proportional to the square of the moisture content," or

 $Q = V^3/75^2$  where

D-55-3

Q = wind resuspension source term, Ci/sec. The corresponding airborne radioactive concentration can be represented by X ~ Q/V indicating that the greater the wind speed the more dilution. Hence, substituting  $Q = V^3/2\Sigma^2$ into X ~ Q/V, we derive X -  $V^2/2\Sigma^2$ . In sum, the term (10/V) in Equation G-11 should be inverted and then squared. If the term is merely inverted, the value for Tsa becomes 4.60 x 10<sup>-11</sup> instead of the 3.53 x 10<sup>-10</sup> calculated by NRC. This would result in a reduction in inhalation doses by a factor of 7.7 in the intruder scenarios.

Third, NRC argues that most of the economic gain from raising the 10 nCi/gm limit would be negated by measurement problems. In many instances, measurement may be expensive: However, Kerr-McGee believes that reasonably economic measurement techniques involving alpha-counting may be devised, at least for its TRU waste from decommissioning. Moreover, no commercial facility is currently available for disposal of TRU wastes contaminated in excess of 10 nCi/gm. NRC should permit its licensees the option of disposal of at least some of this higher concentration of TRU by near-surface methods since

\*\* P. Baçnold, The Physics of Blown Sand and Desert Dunes (1941); Momeni, et al., The Uranium Dispersion and Dosinetry (UDAD) Code (Nureg/cr-0553) (May 1979).

· · · · ·

A . . . . .

D-55-3 there are no alternative repositories available and since there are no health or safety reasons to retain the current stringent disposal limit.

In conclusion, NRC's rationals for maintaining the current 10 nCi/gm limit for TRU is insufficient. The agency should raise the limit by at least one order of magnitude.

> F. The Surety Requirements Set Forth in the Proposed Rule Are Beyond NRC's Authority and Improper

Proposed § 51.52 requires financial surety for monitoring and closure requirements. The only surety authority enjoyed by NRC, beyond NRC's power to examine financial status in reviewing license applications, is with respect to uranium mills. 42 U.S.C. § 2210(x). That authority was specifically conferred by Congress in section 203 of the Uranium Mill Tailings Radiation Control Act (UMTRC Act) of 1978. That specific provision would be superfluous if, as NRC now contends, the agency all along possessed general authority to require sureties. It is elementary that a statute should not be interpreted so as to render any of its provisions a nullity. See, e.c., United States V. Menasche, 346 U.S. 528, 538-39 (1945); Abaracor, Inc. v. United States, 571 F.2d 552, 557 (Ct. C1. 1978); Kenneth V. Schmoll, 482 F.2d 90,94 (10th Cir. 1972); Tabor v. Ulloa, 323 F.2d 823, 824 (9th Cir. 1963). NRC's assertion of authority to impose a surety requirement absent express statutory authority is unlawful.

Proposed § 61.52 is arbitrary for a wholly different reason: it bars the use of self-insurance. NRC's rejection

B-419

E-

- 24 -

- 25 -

of self-insurance alternatives is also arbitrary and unsupported. Many of the Commission's licensees are large, financially sound corporations, such as public utilities and energy companies. These licensees are at least as financially sound and stable as the bonding companies, insurance companies, and banks which would be the sources of the surety which NRC seeks to require. It is simply an unnecessary tie up of working capital and an unwarranted subsidy to the banking and insurance industries to require operators of nuclear facilities to purchase surety bonds or insurance from outside organizations. Moreover, there is no evidence that alternatives to self-insurance are available. Indeed, Kerr-McGee has determined that surety honds are generally not available for stabilization of uranium mill tailings.<sup>41</sup>

E-1

NRC accepts self-insurance to fulfill financial protection requirements for operators of nuclear reactors and certain other nuclear facilities. See 10 CFR §§ 140.11, 140.12, 140.14, 140.15(b). Other agencies likewise endorse self-insurance. See, e.g., 30 CFR § 800.5, 800.12, 806.11(b), modified, 45 Fed. Reg. 52306 (Aug. 6, 1980) (Office of Surface Mining authorizing self-bonding for final reclamation); 33 CFR §§ 135.203, 205 & 213 (offshore oil); 46 CFR §§ 540.5(d), 542.8(a)(3), 543.8 & 544.8 (Trans-Alaska Pipeine); 49 CFR §§ 1043.5(a) & 1084.7 (motor carriers and freight forwarders). These provisions demonstrate the propriety and workability of self-insurance surety mechanisms.

Self insurance accompanied by a financial test (e.g., a demonstration of the financial well-being of the licensee on an annual basis) will satisfy any legitimate health and safety concerns while at the same time costing licensees much less. The additional expense attendant upon the financial arrangements preferred by NRC is not counterbalanced by any benefits. It is "arbitrary and capricious for the Agency to reject obviously less burdensome but equally effective controls in favor of more expensive or onerous ones." <u>South Terminal Corp.</u> v. <u>EPA</u>, 504 F.2d 646, 676 (1st Cir. 1974). NRC should at the very least permit self insurance accompanied by a financial test.

III. <u>Miscellaneous</u>

1. <u>Uranium and alpha-emitting radionuclides</u>. The table limits low-level disposal of U-235 (enriched uranium) to 40 nCi/cm<sup>3</sup> and natural or depleted uranium to 50 nCi/cm<sup>3</sup>. The notes to table 1 propose to restrict alpha-emitting radioisotopes other than radium to the values for J-235 (40 nCi/cm<sup>3</sup>) for purposes of low level waste disposal. In the event that the limit for alpha-emitting transuranics is raised to 100 nCi/gm (or more) for purposes of Class C intruder waste, the limits for U-235, natural or depleted uranium, and alphaemitting radionuclides should be raised to a similar level for Class C disposal. NRC's basic rationale for limiting disposal

D-55-1

E-1

- 26 -

<sup>&</sup>lt;sup>31</sup> <u>See, e.g.</u>, Letter, Mr. Provost (Sursty Ass'n of America) to Mr. Lingo (Kerr-McGee) (Exhibit N).

- 27 -

55-1 (of uranium and alpha-emitter's is (as a general matter) no more applicable to these radionuclides than it is to alpha-emitting transurances.

2. Land disposal other than near surface. The proposed Part 61 regulations at several points reserve subsections for requirements applicable to land disposal other than near-surface. <u>See</u>, e.g., proposed § 61.51(b). Kerr-McGee supports the concept of an intermediate-level waste facility permitting land disposal of higher-concentration TRU wasts than is permitted at low-level waste facilities. The Company believes that NRC, EPA and the Department of Energy should promptly finalize plans and requirements for such a facility and that it should be made available for permanent disposal of TRU-waste resulting, among other things, from decommissioning activities. Such a facility would fully protect the public health and safety and be far more economic that a deep geologic repository for high level waste.<sup>12</sup>

67- I

A-8

is no § 61.60.

3. Proposed § 61.23(g) refers to a § 51.60. There

\*\* See Winograd, Radioactive Waste Disposal in Thick Unsaturated Zones, 212 Science 1457 (1981) (Exhibit O).

#### TENNESSEE VALLEY AUTHORITY CHATTANOOGA. TENNESSEE 37401 400 Chestnut Street Tower II

February 4, 1982

LONGSED RUL

182 FEC-9 P#:00

(46 FR 51176

46 FR 3808

STORDED RULE PR-6.

Secretary of the Commission U.S. Nuclear Regulatory Commission Washington, DC 20555

Attention: Docketing and Service Branch

Dear Sir:

The Tennessee Valley Authority (TVA) is pleased to provide comments on the proposed amendments to 10 CFR Parts 2, 19, 20, 21, 30, 40, 50, 51, 61, 70, 73, and 170 concerning licensing requirements for land disposal of radicactive waste as noticed in the July 24, 1981 <u>Federal Register</u> notice (46 FR 38081-38105) and comments on NUREG-0782.

We appreciate the opportunity to comment and our specific comments are enclosed.

Very truly yours,

TENNESSEE VALLEY AUTHORITY

L. M. Mills, Marager Nuclear Regulation and Safety



#### ENCLOSURE

1. E-K	Page 38086, last paragraph in column 1This paragraph would require a licensee (disposal facility operator) to provide financial responsibility for a disposal site even after it has been turned over to a Government agency for long-term institutional care and monitoring. Since the institutional care period could last as long as 100 years, a licensee's money could be tied up for 100 years. We believe it would be preferable to have the licensee turn over the site and any required money at the time that the license is transferred to the Government.	
ED-1 {2.	10 CFR 61.41, first sentence—We believe the term "general environment" should be defined and suggest the definition should be similar to the same term in 40 CFR Part 190.	M-I
GEN-1 {3.	10 CFR 61.51(a) (4)—We suggest changing "prevent" to "minimize" and, in 10 CFR 61.51(a) (6), changing "eliminate" to "minimize."	
D-52-4	<u>10 CFR 61.51(a) (7)</u> —We believe this item should be deleted since many activities usually take place at a disposal facility other than disposal of radicactive waste. Some facilities are used for transportation equipment storage, empty container storage, and a center for radwaste services to nuclear plants. This regulation could have a significant impact on present disposal sites and conflicts with 10 CFR 61.11(c) (4).	
ED-i{	<u>10 CFR 61.52(a) (4)-(10)</u> —It appears that these regulations apply to Class A, B, and C waste. However, 10 CFR 61.52(a) (2) says that these regulations apply to Class B waste. We believe this should be clarified.	
D-52-3	10 CFR 61.52(a) (8)—We believe the designation of a minimum distance of 100 feet for the buffer zone does not appear to be necessary. We support the need for a buffer zone but the distance criteria should be determined by site specific and other factors of the disposal facility to meet the performance objectives of Subpart C of this part.	
ED-1 2.	10 CFR 61.55, Table 1—The limits for some of the isotopes in this table are the same for all three classes of waste. We suggest the regulations should contain statements that any waste containing these isotopes in concentrations exceeding these limits cannot be disposed of at a near-surface disposal facility.	
D-st. 8	10 CFR 61.55(b) (2) and 61.56(b) — The stability requirement of at least 150 years for Class B stable wasta appears inconsistent with the institutional control period of up to 100 years (see paragraph 61.59(b), page 38097). As discussed in paragraph 61.7(b)(4), page 38091, Class B stable waste contains the " types and quantities of radioisotopes that will decay during the 100-year period to levels that do not pose a danger to public health and safety." Therefore, for waste in this classifica- cation, there does not seem to be a need for stability requirements beyond 100 years.	

D-54-6 Waste is given as at least 150 years. This waste classification specifies a maximum concentration of radionuclides so that at the end of 500 years, the remaining radioactivity is at a level that does not pose a danger to public health and safety (see paragraph 61.7(b) (5), page 38091). Therefore, for this waste in this classification, the stability requirements should be for 500 years.

9. <u>10 CFR 20.311(d) (5)</u>—It appears this regulation requires that a copy of the shipping manifest be sent to the intended recipient (possibly by mail) at the time of shipment. If shipments are sent by truck it is very likely that the truck will get to the disposal facility before the advance copy of the manifest. Therefore, this method would not provide a way of providing prior notification of the intended shipment.

We believe the substance of the information required by this regulation is already required by the Department of Transportation (DCT). We suggest the procedure be revised to require only the submittal of copies of the material sent to DOT. This requirement should be placed in 10 Part CTR 61 or 10 CFR Part 71 and not 10 CFR Part 20.

10. NURSE-0782, Volume 3, Table E.2, Page E-21-We believe the concentration should be in µ3/m<sup>-1</sup> instead of m3/m<sup>2</sup>. Also the annual average secondary standard of 60 µ3/m<sup>3</sup> for sulfur dioxide was revoked in 1973. The current annual average primary sulfur dioxide standard is 30 µ3/m<sup>3</sup>. Also, this table includes only four of the seven pollutants for which standards have been developed. We suggest inclusion of the remaining three—carbon monoxide, ozone, and lead.

. •





Robert Bernstein, M.D., F.A.C.P. Commissioner 1100 West 49th Street Austin, Texas 78756 (512) 458-7111

 Robert A. MacLean, M.D. Deputy Commissioner Professional Services Hermas L. Miller Deputy Commissioner Management and Administration

. .

NALET ASSESSOD

ECTOR SULE

February 1, 1982

Secretary of the Commission U.S. Nuclear Regulatory Commission Washington, D.C. 20555

Attn: Docketing & Service Branch

Dear Sir:

We appreciate the opportunity to comment on Proposed CFR Part 61, Licensing Requirements for Land Disposal of Radioactive Waste (46 FR 38081). As an Agreement State radiation control agency, the development of these rules is important to us since in all likelihood they will be items of compatibility. It is our opinion that these rules are essential to the future of the safe handling and disposal of low-level radioactive waste.

Comments were solicited from our staff and from radioactive material licensees of our Agency. A list of the major comments received is enclosed.

Yours truly.

Maul K.O David K. Lacker. Chief

David K. Lacker, Chief Bureau of Radiation Control

Enclosure



:

COMMENTS ON PROPOSED 10\_CFR PART 61

61.2 - Definitions

ED- 1 The definition of disposal indicates that the waste is isolated from the biosphere. In some areas the minimum covering of five meters of soil would not be adequate to completely isolate the waste from certain plant root systems or burrowing animals.

61.51 - Disposal Site Design for Land Disposal

 $D-5^{-1}$  It is recommended that preferential consideration be given to a progressive slope design for burial of radioactive waste. The reason is to prevent vertical walls which contain fractures, bedding planes and joints as avenues for exit of liquids.

61.55 - Waste Classification

H Nore consideration should be given to separation by half-life prior to burial so that the material with shorter half-lives could be stored for several years and retrieved rather than this material taking up permanent space in a disposal site.

Part 20 - Manifest System

 $\mathcal{M}_{-1}$  It is felt that generators shipping to intermediate processors should not have to package and classify the material as Class A, B, or C. To package Class B or C waste in packages which will last 150 years seems overly costly and unnecessary for short shipments which are to be repackaged.

L'Énergie Atomique Atomic Energy A of Canada Limited du Canada, Limitée Société de Recherche Research Company Chalk River Nuclear Laporatories Laboratoires Nucléaires de Chait River MEMORANDUM FF 181982#January 21 Chalk River Environmental Authority Dr Paul Lohaus لانتهط Low-Level Waste Licensing Branch U.S. Nuclear Regulatory Commissio 718 WASHINGTON, D.C. 20555 U.S.A. HOLE PR FOR COLOR AND AND A TA

As discussed with you by telephone (82/1/11), I would like to make further comments on the draft regulation 10 CFR 61 as given in Volume 1 of NUREG-0782.

Section 61.56, section (a) makes it clear that the requirements in that section are minimum requirements for all three classes of waste. However section (b) does not make it clear that it only applies to Classes B and C as was the intent of the regulation. This could be corrected by adding "in Classes B and C" after "waste" in the third line of 61.56 (b).

Also it appears inconsistent to allow "natural specific activity" for natural or depleted uranium metal while only allowing 0.05 uCi/cm<sup>3</sup> for other forms of U such as  $UO_2$ , since uranium can oxidize in room temperature water to form  $UO_2$ . Also the product of this reaction, a flocculant uranium oxide powder, has a high

 $D^{-55-1}$  of this reaction, a flocculant uranium oxide<sup>2</sup> powder, has a high surface area and is much more amenable to dispersion than  $UO_2$  pellets.

As written, Table 1 of 10 CFR-61 allows up to 30 kg of UO, in a drum but excludes a single UO, pellet under the 0.05 uCi/cm maximum concentration regulation. I understand that the intent of the regulation was not to exclude quantities of natural UO<sub>2</sub> and the regulation will be revised.

I understand that the alpha emitting transuranic isotopes or Pu-241 are allowed under Columns 1 and 2 of Table 1 at the same level as under Column 3 and that the Table will be rewritten to show this.

I still don't quite understand the concentration limits stated in terms of theoretical maximum specific activity (TMSA) which is defined as the radioactivity of a nuclide per unit mass of that nuclide e.g. Gi of Co-60 per g of Co-60. The TMSA is about 1.1 x 10  $\mu$ Ci/g of Co-60. Over what volume of waste is this activity to be averaged, one cm? =D-| last line. There is no section 61.60.

- 2 -

I hope these comments will be of some use to you.

Yours sincerely,

E

ind

MAF:sa

M.A. Feraday

CRNL-47 (Rev. 3/74)

Dear Paul:

ED-1

ED-1



Enclosure A

## COMMENTS ON PROPOSED REGULATION 10 CFR 61

## 1. General Comments

*k*.

D-55-2

D-55-

Institutional Control - The Department believes that the duration of institutional control significantly affects the cost of waste disposal. Disposal of Class B waste would be significantly more expensive than Class A waste. For example, raising the institutional control period from 100 to 300 years would increase the Class A limits for Co-60, Ni-63, Sr-90, Cs-137, and H-3. The increase of a factor of 100 for Sr-90 and a factor of 44 for Cs-137 are particularly significant since this would allow larger volumes of mixed fission product waste to be classified as less expensive Class A waste.

Therefore, the Commission's basis for limiting institutional control to 100 years should be reexamined. Such controls have existed in this country for more than 100 years. Loss of institutional controls would generate more serious public concerns than those presented by a low-level waste disposal site.

b. Intruder Scenario - The inadvertent intruder scenarios are similar in many respects to scenarios in Paragraph 259 of NCRP report No. 39 for accident scenarios in less urgent emergencies in that a limited number of members of the general public may become inadvertently exposed to radiation. An annual dose connotes chronic exposure and seems unwarranted and inappropriate to use for an intruder. (For additional information see the discussion paper in Enclosure B.)

. De Ninimus Levels - Values and/or criteria for wastes and waste streams which could be exempt from 10 CFR 61 should be defined in establishing de minimus levels. We recommend that serious consideration be given to the establishment of de minimus levels for radionuclides such as 2350, 99 Tc, Pu, and Np in consultation with the Environmental Protection Agency.

C. Definition of Transuranic Waste - The transuranic nuclide concentration limit established should be consistent with health and safety limits using realistic pathway analyses as performed for other radionuclides in Table 1. The Department is reassessing the limit it established for its operations in 1970 using environmental, health, and safety considerations. The Commission should consider recent NCRP and EPA studies in recommending the establishment of higher transuranic concentration limits. Waste streams may be processed which would result in TRU or low-level waste according to the level of alpha activity, and disposed of accordingly. A concentration limit that is presently being considered by the Department is 100 nCi/g of alpha activity from radionuclides with half-lives greater than 20 years.

Quality Assurance - Future regulatory guides should address quality assurance with respect to site characterization, designs, construction, tests, measurements, etc. Practical, specific quality assurance program requirements should be created for land disposal activities.

2

A. Requirements for detailed waste data - The proposed waste classification system implies detailed radioisotopic information for all constituents of the waste. Such a requirement would be unnecessary, costly, and could lead to increased occupational exposure. The Commission's intentions regarding this area should be made clear.

9. Yolume Reduction - Some of the radionuclide concentration limits in Table 1 may inhibit volume reduction. This could occur when a specified radionuclide reaches its limit. For example, C-14 could limit reactor waste although it is not the most prevalent or most hazardous radionuclide. The Commissioners have issued a policy statement supporting volume reduction and the regulations should be consistent.



# 2. Summary of Rule -

 $\square$  { Section G. - The Commission should state whether a safety evaluation report is to be prepared by staff prior to the issuance of a license.

A - 1Section H. - The summary states "it would be the Commission's intent that all future disposal would be expected to comply with the provisions of Part 61." The NRC should recognize that in Agreement States regulations must be promulgated by state authorities.

D-1 The summary should acknowledge that future regulatory guides will address the Commission's thinking in certain areas, e.g., site suitability.

Section 61.1 (b) - This section states ". . . the regulations in this part apply to all persons in the United States." An additional clause should be added to this sentence to read ". . .States where the Commission retains authority."

 $\begin{array}{c|c} & \underbrace{Section \ 61.2}_{used \ in \ Sections \ 61.59 \ and \ 61.62. \end{array}$ 

Section 61.2 - To differentiate between intruder barrier and engineered barrier, "Engineered barrier" should be defined as "a man-made structure or device intended for the purpose of confining the waste and limiting the migration of the waste components or to protect an inadvertent intruder from direct exposure to the waste."

5. Section 61.6 - The title should be changed to read "Specific Exemptions."

ଚ୍ଚ-ା

	3		4
D-55-10 F	Section 61.7(b)(2) - In view of our comments on Table 1, the Commission should indicate which radionuclides could serve in establishing the maximum disposal site inventory.		Section $61.50(a)(5)$ - This paragraph requires the disposal site to be "generally well drained." This is difficult to quantify and is adequately covered by the remainder of the sentence. Delete "generally well drained."
B-7 5	<u>Section 61.7(c)(3)</u> - We question the basis for applying the 5 year post- closure observation period for all cases. We suggest that in some cases a shorter period may be justified, for example, if a facility has operated for 30 years without major problems. On the other hand, a	J-50-32	defined. The Commission should establish procedures for an applicant to propose mitigation measures which would allow for exemptions to specific technical requirements.
2	longer period may be justified when facilities have operated for shorter periods or have extenuating circumstances.	(19.	Section 61.50(a)(6) - The term "upstream drainage area" needs to be explained.
ED-1	Section $61.7(b)(4)$ - Delete line 1D and replace with "an unacceptable hazard to an intruder or to".	(20.	Section $61.50(a)(7)$ - Sufficient depth to the water table should be more specifically defined as that depth completely above (or below) the
210.	Section 61.7(b)(5) - Line 17, insert after "pose", "an unacceptable hazard to an intruder or" and delete "a danger".	D-50-2	transection zone between the Saturated and unsaturated zones. The transection zone is the zone over which the water table and the capillary fringe cap fluctuate.
B-1 { <sup>i1.</sup>	<u>Section 61.23(a)</u> - This section states that the applicant should be qualified by reason of training <u>and experience</u> . This might tend to rule out new companies and would lead to monopolies by present companies. It is suggested that "and" be replaced with "or".		We believe that the intent of this requirement is that the water table shall not cyclically rise into and fall beneath the buried waste. Burial beneath the water table could be satisfactory, if diffusion is the controlling rate (as stated in this paragraph), if the travel time
(12.	Section $61.23(g)$ - There is no Section $61.60$ as stated in this section.	Ĺ	is very slow, it the performance objectives can still be met, and if the water table never drops below the buried waste.
€D-()13.	<u>Section 61.24(h)</u> - It is suggested that the phrase "or thereafter" be deleted. Changes after the issuing of a license should be limited to those needed to protect health and safety or should be negotiated with the licensee.	D-50-5	<u>Section <math>61.50(a)(9)</math></u> - It would be sufficient to say that active seismic faults or volcanic sites are considered unsuitable. Tectonic processes such as faulting or folding occur on a time scale so much longer than the disposal site lifetime to make this paragraph meaningless. Seismic
B-5 { <sup>14.</sup>	<u>Section 61.27(d)</u> - This section discusses license renewal, although no prior section discusses license duration. The Commission should indicate the amount of time for which a license is issued.		overriding direct impacts as to make any effect on the disposal site inconsequential.
B-8 { <sup>15.</sup>	Section 61.31 - The Commission should specify more clearly in the regu- lation the custodial agency license conditions. We assume that the license conditions will change as a result of the license transfer from	ED-1 {22.	<u>Section 61.51(a)(1)</u> - The term "long-term isolation" needs to be clarified. Adding "after site active operations cease" at the end of the statement would help.
(	the operational phase to the custodial phase. Section $61.41$ - The basis for the drinking water limits should be provided.	23.	<u>Section 61.51(a)(6)</u> - This section establishes technical requirements which are impossible to meet and demonstrate because all soils contain
c-32"	Section 61.50(a)(2) - It is not clear what constitutes a capability	D-51-1	some moisture. Our understanding is that the objective of this section is to minimize contact between water and waste. The following paragraph is a suggested replacement:
D-50-1	or being modeled, analyzed, or monitored. A more specific statement, such as "the site shall be located in areas where hydrogeologic conditions allow reliable, technically conservative performance evaluations through	GEN-1	"The disposal site must be designed so that storage areas for waste are well drained and protected from the weather. Disposal areas
	cnaracterization, modeling, analyses, and monitoring. That is, the site must be able to be described using basic model assumptions, and the input parameters required for a model simulation must be measurable, attainable, and distributed in a definable manner."		should be designed to minimize the contact time between percolating water and waste and to eliminate standing water contact with the waste."
-		m-572-1 {24.	<u>Section 61.52(a)(1)</u> - This paragraph states that Class A wastes "must be segregated from other wastes by placing them in disposal units which are sufficiently separated from other units so that there is no interaction

•

.

.

D-55-13 2

A-7

D-52-1 Detween them." It is not clear what kinds of interactions (e.g. chemical, physical) are to be avoided.

Section 61.52(a)(4) - It is not clear whether paragraphs (a)(4) through (10) of this section apply <u>only</u> to Class B stable waste, as stated in 61.52(a)(2), or also to other classes of waste.

 $\begin{cases} 26. \quad \underline{Section \ 61.52(a)(4) \ and \ (5)}_{requirements, \ which \ would \ increase \ disposal \ costs, \ are needed \ if \ the \ overall \ performance \ objectives \ established \ elsewhere \ in \ the \ regulation \ were \ met. \end{cases}$ 

27. Section 61.52(a)(6) - The phrase "a few percent" is vague. The level of gamma radiation at the surface of a trench cover should be specific. "Within a few percent" should be replaced with "no more than 10 percent" if measuring within 10 percent is feasible. An alternative approach would be to use survey meters in order to take corrective action following trench covering should the surface radiation limit exceed 1 mrem/hr, as this has been an operating practice used by the Department.

28. Section 61.52(a)(9) - No additional site closure requirements that would reduce the likelihood of inadvertent intrusion are discussed. Such requirements should be added. One example is requiring that permanent monuments be provided for the site.

29. Section 61.54 - It is our understanding that this section allows applicant to propose site specific requirements for waste segregation and disposal and facility design and operation. We strongly support this flexibility and recommend that the Commission stress this feature of the regulation in the future.

30. Section 61.55, Table 1 - This table gives specific radionuclide concentration limits for three proposed classes of waste. This table could imply that all radionuclide concentrations have to be quantitatively measured in order to classify the waste. It would be costly, by as much as a factor of two or three, and impractical to quantitatively measure the concentration of each radionuclide in each waste container. Many wastes contain a large number of radionuclides, but only a few would be significant in determining the potential hazard and, therefore, the category of the waste. The regulations should specifically state that radionuclide concentrations can be estimated by using techniques such as: consideration of external radiation levels, ratios of known radionuclide concentrations, or the source of the waste.

31. Section 61.55, Table 1 - A footnote states that wastes containing chelating agents in concentrations greater than 0.1 percent by volume cannot be buried without specific Commission approval. Strong decontamination solutions used in the nuclear industry can exceed this limit. Wastes containing chelating agents which are solidified with cement or

other acceptable media should be accepted without limit, other than radionuclide concentration, at disposal site. As a minimum, low activity Class A wastes should not have this restriction since the limits for Class A wastes are based on the intruder scenario and not on groundwater migration.

6

33. Section 61.55, Table 1 - The concentration limits for all radionuclides except transuranic nuclides are expressed in terms of microcuries per cubic centimeter, and averaging over the volume of the package is specifically permitted. The limit for transuranic nuclides is in terms of nanocuries per gram, but the regulations do not say whether this can be averaged over the weight of the package. Averaging over the weight of the package should be permitted.

34.  $\frac{\text{Section 61.55, Table 1}}{\text{permitted for krypton-85 in a Class B waste would be that of Cs-137, 44 Cl/m<sup>3</sup>, or 440 Cl/m<sup>3</sup> if contained in metal. This eliminates disposal of immobilized krypton-85 in dry wells. Disposal of krypton-85 immobilized by section by section wells.$ 

ized by zeolite encapsulation or ion implantation into a metal may be the most effective means of management. The rules should be written so as not to eliminate this mode of disposal. This may be done by categorizing the maximum specific activity (for 7% krypton-B5 in stable krypton) as a Class B waste (Column 2 in Table 1). Since for the waste forms under consideration, the maximum concentration in the closed container would be less than 5 x 10<sup>6</sup> Ci/m<sup>3</sup>. the specified contained concentration should be increased to 5 x 10<sup>6</sup> Ci/m<sup>3</sup>.

77-5

5

ED-

36. Section 61.55(d) - The term "near surface disposal" is used and therefore implies a non-near surface disposal concept may be suitable for certain wastes. There is a need to define the non-near surface concept.

ED-1

D-55-1

ED-1

D-56-9

D-56-11

17 Section 61.55(d) - It would be technically more correct to have this sentence read: "Waste that has a radioisotope concentration that exceeds the numerical values shown in Column 3, Table 1 ...." It is implicit, by definition, that the concentration of H-3, C-14, and Co-60 cannot be in excess of their respective theoretical specific activities.

38. Section 61.55(d) - This paragraph and the limits in Table 1 would prohibit the disposal of waste such as uranium tetrafluoride (UF<sub>4</sub>) even though the disposal of natural or depleted uranium metal is permitted by a Table 1 footnote. In Table 1, Column 3, the radioactivity limit for natural or depleted uranium waste is 0.05 uCi/cm<sup>2</sup>. UF<sub>4</sub> has properties similar to uranium metal and its radioactivity disposal limit should be increased. Otherwise packages or drums of UF<sub>4</sub> may not be placed in the facility unless the Commission gave specific approval. In the preamble, UF<sub>6</sub> process waste is listed as Class A. Segregated Waste. UF<sub>4</sub> is considered as a UF<sub>6</sub> process waste.

(39. Section 61.56(a) - The health and safety statement should be more specific by referring to the general public and/or the personnel at the disposal site to clarify the intent of the protective measures.

Section 61.56(a)(1) - The requirement should only apply to receipt of waste at the site, not to actual disposal. Shipping containers are frequently designed for reuse and can be costly. The regulation should not require the disposal of costly reusable shipping containers. Further, the regulations provide no credit for the container in terms of waste containment. It is suggested that "The waste ..." be replaced with "All waste accepted at the site."

<u>Section 61.56(a)(2)</u> - This section precludes the use of cardboard and fiberboard boxes presumably to protect the operational staff at the disposal facility. These containers are permitted by transportation regulations and are being used by Department facilities without difficulty. We recommend deleting this section as it is an undesirable and unnecessary generic restriction. Individual sites may still choose not to accept such containers.

42. <u>Section 61.56(b)(1)</u> - Wastes falling into the intermediate category are required to retain structural stability in the ground under a compressive load of 50 psi in order to mitigate the burial trench subsidence. Since large component packages would only be placed in a single layer in a trench, the weight on top of these packages would be due to the earth covering the package. Therefore, for large packages, the 50 psi compression requirement is unnecessary. It is recommended that the compressive load requirement be related to the depth from the earth's surface to the top of the waste and to the covering material rather than specifying a single value for all cases. Section 61.56 states that 150 years of structural stability is required.
 The Commission should clarify what structural stability pertains to; the disposal site, the waste package, or the waste form. The Department suggests that waste form rather than waste package should pertain to the structural requirement. The 5 percent tolerance on physical dimension of waste packages will causé expensive high integrity containers to be used. Experience has shown that most drums are only filled to about 80 percent of their theoretical volume rather than the 95 percent required.

8

:4

rv

2

 $\mathbf{N}$ 

- 43. Section 61.56(b)(3) This section states that "void spaces within the waste and between the waste and its package must be reduced to the extent practicable." This requirement is vague and should not be needed if the stability requirements of Section 61.56 are met. If interpreted literally, this regulation would require that voids in packages containing large activated components be filled in. Such an operation would increase worker radiation exposure and increase the package weight. "Increased weight could have a significant impact on the ability to ship and handle these packages. It is our understanding that it is the Commission's intent that this section would require operators to compact the soil covering of waste. We believe that this is adequately treated in the definition of stability and should be deleted.
  - 44. Section 61.57 There should be clarification on the responsibility for sorting and labeling of waste packages. This would assist in establishing compliance with transport requirements, Title 49 and 10 CFR 71, and in classifying the waste on site. Presumably, the shipper is responsible for labeling the package. Since extensive labeling is required for transportation purposes, these sets of requirements should be compatible.

45. <u>Section 61.59(a)</u> - Does this paragraph rule out tribally-owned lands? This question should be clarified.

46. Section 61.62(a) - Replace the word "aliminated" with "minimized". "Elimination" of ongoing active maintenance may be the goal but it probably will not be met. This fact is recognized in 61.63(a) by the words "any required maintenance".

- 47. Section 61.70 It is unclear what the role of an Agreement State would be in the licensing and regulation of land burial of radioactive wastes as opposed tr a non-Agreement State. Our impression is that this section applies only to non-Agreement States.
- (48. <u>Section 61.71</u> It is recommended that the word "may" be replaced by "shall". This function should be an obligation of the Director.
- $G = 1 \begin{cases} 49. \\ \frac{\text{Section 61.80(b)}}{\text{specified.}} &= \text{To whom the records will be transferred should be} \end{cases}$
- $\overline{L}^{-2}$  50. <u>Section 61.80(g)</u> Delete the words "if any". Licensees should be required to furnish an annual financial report.

B-429

D-51-12 GEN=1

7
# ED-1,

. . . .

# Editorial Comments

. . . .

- The Environmental Impact Statement indicates the Department of Energy (Department) is responsible for disposal of all Federal government low-level waste. At present, only the Department waste and classified. defense-related waste is accepted at the Department's facilities.
- 61.1(b) For consistency, either delete the paragraph reference for Part 40, or add paragraph references for Parts 60 and 20.
- 3. 61.2 Definitions

"Active Haintenance" by current understanding is improperly defined - what is actually defined is Remedial Action. That which is excluded is active maintenance.

"Active Maintenance" 15th line typo "revegetation".

"Near Surface Disposal" add "a" after means and delete "in or"

- 4. 61.4 4th line change "should" to "shall"
- 5. 61.4 4th line from end change "at" to "to"
- 61.7
  2nd line delete word "intended"
- 7. 61.7(b)(3) 10th line add "disposal operations cease,"
- 8. 61.7(b)(4) 1st line add an "a" after "for" and make "period" singular
- 9. 61.7(b)(5) Next to the last line delete "recognizable". There is no reason to believe an intruder would recognize the problem regardless of form.
- 10. 61.7(c)(4) 4th line change "agency" to "government" and change "that" to "whichever". Delete next sentence. No Federal agency can own land.
- 11. 61.10(a) 2nd line change "use" to "handle"
- 12. 61.22(o) 1st line change "its" to "the"
- 13. 61.23(g) 4th line insert "are met" after section
- 14. 61.24(b) 2nd line put commas before and after "under oath"
- 15. 61.50(a)(4) Change "must" to "should"

- 16. 61.52(a)(4) Add "operations" after "disposal"
- 17. 61.53(d) 3rd line typo "indicate"
- 18. 61.55 Ist line insert "Low-level" before "radioactive"
- 19. 61.55(b)(1) 3rd line insert after Column 1 "but does not exceed the values shown in Column 2; and,"
- 20. 61.55(c)(1) 2nd line insert after Column 2 "but does not exceed " the values shown in Column 3:..."
- 21. 61.56(b)(1) 9th line delete one "as"
- 22. 61.72(a) 7th line insert "by" after "affected"
- 23. 61.72(c)(5) Delete "be" after "should"
- 24. 61.80(d) Replace "takes" with "shall take"

10

ED-1

#### Enclosure B

. 1

## DOSE LIMITS UNDER THE INTRUDER SCENARIO

Subject

Dose criteria for intruder scenarios.

#### Department of Energy Position

The design objectives for disposal sites are appropriately and properly expressed as dose criteria. However, the intruder dose limit should be similar to limits used for accidents, i.e., single event exposures to a limited segment of the public.

Discussion The Department takes issue with the NRC's assertion that a 500 mrem/yr intruder limit does not significantly increase costs compared to a higher limit. Nearly all of the radionuclide concentration limits in Table I are set by the intruder scenario. Significant volumes of waste, particularly large metallic components from decommissioning of reactor plants, would exceed the Class C limits. Disposing of this waste by any means other than shallow land burial would greatly increase costs. Also, the intruder dose limit affects the limits for Class A wastes. Since Class B and C wastes are much more expensive to dispose of, increasing Class A limits would reduce costs. Therefore, setting an unnecessarily low intruder dose limit does significantly affect waste disposal costs.

man and the second of Section 61.42 proposes a whole body annual dose of 500 mr for the inadvertent intruder and the waste classification limits of 61.55 are controlled by the doses received from this scenario. The limit in 10 CFR 20 is based on potential release pathways and not from intrusion into the waste. Based on release pathways, an individual exposure limit of 500 mr/year is reasonable and achievable, in the context of the intent of Part 20, i.e., normal operations. We believe that human intrusion is a credible event and should be carefully evaluated, but to a separate criterion.

A person who inadvertently intrudes into the waste and who has bypassed warning markers is subject to the maximum radiation level potential available at that site. .. In the context of this application, human intrusion is not a normal routine exposure scenario and warrants separate treatment similar to an accident situation or an unplanned release. In NCRP Report, No. 39, "Basic Radiation Protection Criteria" Paragraph 259, the limit on accidental exposures in less urgent emergencies is 25 Rem. As we have previously indicated in 10 CFR 60, we believe that warning markers can be developed which will reduce the probability of intrusion. Additionally, we are not convinced that discounting institutional controls after 100 years is a proper approach. We continue to believe that sufficient institutional memory will survive most probable events to reduce the probability of intrusion.

· . ·

Based on the above, an inadvertent intruder should be allowed a higher radiation dose than the exposure limits applicable to routine operations. 2

200

•

NON-NEAR SURFACE DISPOSAL A - B

# Subject

The need and desirability of requiring non-near surface disposal of some low-level wastes.

# Department of Energy Position

An implication in 10 CFR 61 is that similar regulations governing the use of non-near surface facilities for the disposal of low-level waste will be issued. The Department does not consider that a need for such facilities has been established. Our recommendation is that the Commission only specify design objectives at this time. Design and analytical guidance may also be needed.

# Discussion

As a result of the waste classification system development by the Commission (draft 10 CFR 61), quantities of low-level radioactive waste may be unacceptable in near surface disposal facilities and may require greater confinement in, presumably, non-near surface disposal facilities. Requiring non-near surface disposal will result in a variety of impacts, beyond those of low-level waste, including social, political, economic, and transportation impacts.

Several social, political, and economic impacts could arise if non-near surface disposal were to be required:

- The volumes of waste, estimated to be up to 50,000 ft<sup>3</sup>/year, could only justify one such expensive facility nationally. The institutional and political questions associated with establishing such a facility could be similar to those associated with a high-level waste disposal facility. Such a situation is not particularly attractive.
- 2. The Department is presently evaluating the need for non-near surface disposal facilities and the technology required for such a facility. Several alternatives exist for non-near surface disposal facilities and other methods for achieving greater isolation than afforded by normal near surface disposal, including providing longer lasting packages or engineered barriers. Focusing on non-near surface disposal now may prematurely foreclose other options.
- 3. Reprocessing of spent fuel may become more prevalent. This could create a new commercial waste stream beyond Class C wastes requiring greater isolation than near surface disposal, however, since the NRC has not considered reprocessing in this rule it appears premature for the Commission to establish a requirement for non-near surface disposal facilities.

#### SPECIFIC NUCLIDE CONCENTRATION LIMITS

4

# Subject

3

The radionuclide concentration limits established in 10 CFR 61.

# Department of Energy Position

Some of the values in the regulation seem unduly conservative, some of the nuclides seem unnecessary and inappropriate, and sufficient information is not provided for some nuclides. It is recommended that only those nuclides for which there is a demonstrated hazard protection need and for which sufficient information exists be included in the regulation. The Department recommends that the limit for carbon-14 be raised by a factor of 8, for the nickel isotopes, a factor of at least thousand, and for nicbium a factor of 12.

# Discussion

While it has not been possible to thoroughly review the Environmental Impact Statement (NUREG 0782) or the Data Base (NUREG/CR-1759), it appears that the NRC has used overly conservative assumptions in calculating the concentration licits. Our preliminary review of the calculation of the Commission for "C, nickel and niobium isotopes, and the analysis is included in this paper. We agree that the methods used by the NRC provide conservative bounds, but using a series of conservative assumptions greatly compounds the conservatism and may lead to the imposition of unrealistic limits.

# Carbon-14 Discussion

Three principal pathways of movement of  $^{14}$ C into vegetation should be considered:

- A. Direct uptake of lower molecular weight <sup>14</sup>C species by root systems which have penetrated the burial zone.
- B. Uptake of  $14^{14}$ CO<sub>2</sub> as bicarbonate following dissolution of microbially-released  $1^{14}$ CO<sub>2</sub> by plant foliage following diffusion from decomposing wastes.
- C. Photosynthetic uptake of  $^{14}\mathrm{CO}_2$  by plant foliage following diffusion from decomposing wastes.

The rate of release and relative importance of each pathway will depend on type and depth of buried wastes; adequacy of isolation of buried zone from wetting by soil water and upward mass movement, total amount and specific activity of organic wastes, permeability of the soil, base organic content of soil, soil pH, and vegetation type. Since most carbon in plants is absorbed from the air, we consider pathway C in lieu of pathway A as used by the Commission to be the main pathway under most conditions. The rate of transfer to vegetation can be roughly estimated as follows:

- Calculate rate of decomposition An approximate turnover time of 1000 years for decomposition of the approximate 4000g/mj of organic carbon in an eastern forest soil is a reasonable start.
- Using the release rate and diffusivity of CO, in soil calculate the partial pressure of CO<sub>2</sub> in soil and the transfer of CO<sub>2</sub> to air from soil.
- 3. Calculate <sup>14</sup>C uptake from soil solution by plants using concentration of <sup>4</sup>CO<sub>2</sub> in soil water and rates of transpiration of water by plant leaves (any physiology text). This uptake will depend on soil temperature, Henry's constant, partial pressure of <sup>14</sup>CO<sub>2</sub> in soil air, solubility of CO<sub>2</sub> in water, and transpiration rates in liters of H<sub>2</sub>O/m<sup>2</sup>. For these calculations assume that this <sup>47</sup>CO<sub>2</sub> would be fixed photosynthetically in plant leaves.
- 4. Calculate photosynthetic uptake from air based on diffusion of  ${}^{14}\text{CD}_2$ from the soil surface, and content of 640 micrograms of  ${}^{12}\text{CD}_2$  per m<sup>3</sup> of air, an exchange rate of 1 yolume per minute, and proportioned uptake of  ${}^{12}\text{CO}_2$  with respect to  ${}^{12}\text{CO}_2$  from this air.

Using such a procedure should yield an increase of greater than a factor of 8 in the carbon-14 concentration limit.

# Nickel Isotope Discussion -

As a consequence of Table 1, two land disposal waste classifications (A segregated and unacceptable) pertain to Ni-59; and three classifications (A segregated, B stable, and unacceptable) pertain to Ni-63. The tabulated waste concentrations, in uCi/cm<sup>3</sup>, may be increased by a factor of ten for isotopes contained in metals and may be averaged over the package volume (1.e., 200,000 cm<sup>3</sup> for a 55 gallon drum). The tabulated nickel concentrations in subpart D of 10 CFR 61 which categorize the radioactive wastes are 2.2  $\mu$ Ci/cm<sup>3</sup> (Ni-59) and 3.5 70  $\mu$ Ci/cm<sup>3</sup> (Ni-63).

The low-level wastes anticipated for disposal during the time period 1980 to 2000 were divided into four general groups based upon common characteristics (see Table 3.1 in NUREG 0782). These groups are: (1) light-water reactor process wastes, (2) trash, (3) low specific activity wastes, and (4) wastes. having unique special characteristics. By combining the isotopic concentrations (see Table 3.3 in NUREG 0782) and total projected waste volumes (see Table 3.4 in NUREG 0782), the accumulated radioactivity for land disposal will be 6.300 and 830,000 Ci for Ni-59 and Ni-63, respectively. It should be noted that 92.0 percent of the Ni-59 activity and 96.6 percent of the Ni-53 activity are

and the second second

· · ·

projected for the group 4 waste category which comprises only 1.6 percent of the total low-level waste volume. Despite the high nickel concentrations in group 4, all of the projected waste streams of Ni-59 would be classified as class A segregated, and only the four Ni-63 waste streams in group 4 would be restricted to a class B stable classification. It should be noted that the two streams designated as LWR nonfuel reactor components and sealed sources would require the factor of ten increase in Ni-63 concentration permitted for metals to qualify for near surface disposal. Even with recycled uranium fuels, nickel isotopes are not a problem in gaseous diffusion cascades.

6

1.17

Although the projected nickel isotope waste streams are acceptable for land burial as defined by the concentrations in Table 1, certain nuclear reactor hardware may be classified as unacceptable. The BWR end fittings, which 3 contain inconel expansion springs, may contain up to 50 and 7,000  $\mu$ Ci/Cm<sup>3</sup> of Ni-59 and Ni-53, respectively, after a burnup of 27,500 MWd/MI.<sup>3</sup> These values, which exceed the maximum concentrations of 22 and 700  $\mu$ Ci/Cm<sup>3</sup> in Table 1, would not permit near-surface disposal of the BWR end fittings. Due to the corrosion resistance of the activated metal hardware, the BWR in Subpart C (see 61.58 in 10 CFR 61). To evaluate this possible alternative, the respective nickel isotope concentrations in the BWR end fittings are about 100 and 15 ppm (by weight) for Ni-59 and Ni-63.

A reasonable pathway analysis can be performed by starting with Reference Man.<sup>6</sup> The nickel content of the body is 10.0 mg with 5.3 mg distributed in the soft tissues (60,000 g) and 5.0 mg distributed in the skeleton (10,000 g). A daily nickel intake in food and fluids amounts to 0.4 mg/d, and only 0.6 ug of nickel enters the body daily by inhalation. About 98.5 percent of the nickel entering the body either goes directly to excretion or is retained with a biological half-life ranging from 0.1 to 0.2 days. The remaining nickel entering the body, 0.006 mg/d, is assumed to be uniformly distributed throughout all organs and tissues and retained with a biological half-life of 1200 days. This halflife is compatible with the fractional absorption, daily intake, and total body content of nickel (10.0 mg) given for Reference Man.

With the given information, the quantities of Ni-59 and Ni-63 can be calculated which will give an annual dose of 25 millirems to the whole body (see 61.41 in 10 CFR 61). A 25 millirem annual dose to a 10,000g skeleton containing 5 mg of nickel will require isotopic concentrations of either 5,000 ppm of Ni-59 or 2.8 ppm of Ni-63. It should be noted that the same annual whole body dose to 60,000 g of soft tissue (5.3 mg nickel) would require higher isotopic nickel concentrations. Comparing these isotopic concentrations with the BWR end fittings indicates that the potential for an overexposure due to nickel activity is difficult to conceive. Since naturally occurring nickel is widely distributed in the earth's crust with an estimated average content of 0.019 percent, the radioactive nickel isotope concentrations will be diluted as migration proceeds from the burial site. Even without isotopic dilution, the Ni-63 concentration of 120 ppm would only require a decay period of 540 years to reach the 2.8 ppm level.

8-433

.

7

An intruder may receive 500 mrem/year thus increasing exposure by a factor of 20. This results in concentrations of Ni-63 of about 100 ppm for class A waste and 2500 ppm for class B and C wastes.

If all of the nickel in an intruder's body was derived from the waste, the Hi-59 and Ni-63 concentration limits resulting in an exposure of 500 mr/year could be set as follows:

N1-59 N1-63	Waste Class	Weight percent A	Ni-isotope/I B	otal Ni C
		10 0.012	10 0.25	10 0.25

For decommissioning metallic wastes, core barrels, and reactor vessels, this would seem to be a preferred way to express the limits in Table 1.

# Nicolum Discussion

If the Department's recommendations on carbon-14 and nickel isotopes are adopted, niobium-94 would become the limiting isotope for gon-fuel activated reactor components. The proposed NRC limit of 0.02  $\mu$ Ci/cm<sup>3</sup> for Nb-94 in metal would be exceeded by many components. In a study by Battelle, the average Nb-94 concentration in the stainless steel core barrel of a reference PWR was calculated to be 0.17  $\mu$ Ci/cm<sup>3</sup>. Thus, this metal would be unacceptable for burial unless it could be diluted by a factor of nine. The actual concentration of Nb-94 in activated steel is not known since the Nb-94 is masked by the much larger initial concentrations of other isotopes. It is calculated by assuming a conservatively high concentration of target Nb-93 atoms in the steel.

One way to assess the hazard to a potential intruder from the ND-94 in waste is to assume that all of the niobium in the intruder's body comes from the waste. As in the case of the nickel isotopes, this is a conservative assumption since niobium is not concentrated by the body. Niobium is present in many common foodstuffs and the daily intake of 620  $\mu$ gm is balanced by an excretion rate of 620  $\mu$ gm per day. 110 mg of niobium is present in the soft tissue of the body.

What would the dose to the intruder be if the nicbium in his body came from the core barrel in the Battelle study? For stainless steel with a nicbium content of 0.016 weight percent and 0.17 µCi/cm of Nb-94, the specific isotopic concentration of Nb-94 is 678 ppm (mg Nb-94 per Kg nicbium). The limiting organs for a total body source of Nb-94 are the small and lower large intestine walls. The dose to these organs if the intruder's 110 mg of nicbium contained 678 ppm Nb-94 would be 3.4 rem. If 5 rem per year were the limit to the intryder, then the appropriate Nb-94 limit would be about 985 pm or 0.24 µCl/cm. Since inhalation of Nb-94 was found to be more limiting than direct exposure in the DEIS (NUREG 0782), it is useful to consider how much metal would have to be inhaled by an intruder to receive a dose of 5 rem to the lungs. Assuming the niobium corrosion products are still a trace part of the steel corrosion products, the intruder would have to inhale the corrosion products of 1073 grams of steel with a specific Nb-94 concentration of 985 ppm. It is difficult to postulate a realistic scenario where this amount of rust could be inhaled.

8

. .

To further put the hazard from small amounts of Nb-94 in stainless steels in perspective, one should consider the nonradiological hazards of this material. Stainless steels contain abut 10 percent by weight of nickel. Nonradioactive nickel is known to have a severe acute systemic toxicity when inhaled and severe chronic systemic toxicity when either inhaled or ingested. Thus, nickel metal and insoluble compounds have been assigned a threshold limit value of 1 mg/m<sup>3</sup> in the workplace environment.

Thus, one can compare the radiological and nonradiological hazard of breathing rust from activated stainless steel with 985 ppm Nb-94. If an intruder were breathing air with a Nb-94 concentration at the derived air concentration in limit of  $_3 \times 10^{\circ}$  Bq/m<sup>3</sup>, the nickel concentration in the air would be about 270 mg/m<sup>3</sup>. The nonradiological hazard is far more limiting. Thus, since disposal of nonradioactive stainless steel is not generally considered to be hazardous, it seems reasonable to increase the limit for Nb-94 to 985 ppm Nb-94 or 0.24  $\mu$ Cl/cm<sup>3</sup>.

# REFERENCES

- D. E. Reichle, <u>Dynamic Properties of Forest Ecosystems</u>, Cambridge Press University, International Biological Program - Vol. 23, 1981.
- 2. W. Stumm and J. J. Morgan, Aquatic Chemistry, Wiley Interscience, 1981.
- Personal communication S. D. Clinton, Chemical Technology Division, Oak Rige National Laboratory, with E. J. Barber, Enrichment Operations, Oak Ridge Gaseous Diffusion Plant, on December 11, 1981.
- A. G. Croff, et. al., <u>Revised Uranium Plutonium Cycle PWR and BWR</u> Models for the ORIGEN Computer Code, ORNL/TH 6051, 1978.
- Personal communication S. D. Clinton, ORNL, with C. W. Alexander, Chem-Technology Division, ORNL, on December 2, 1981.
- W. S., Snyder, et. al., Report of the Task Group on Reference Man Pergamon Press, New York, 1975.
- Personal communication S. D. Clinton with M. R. Ford, Health and Safety Research Division, DRML, on December 8, 1981, concerning data proposed for publication in ICRP 30.
- C. O. Smith, Notes on Reactor Materials Engineering for CRSORT Students Part 3, Oak Ridge National Laboratory, 1958, p. 733.
- NUREG/CR-013D, "Technology, Safety and Costs of Decommissioning a Reference Pressurized Water Reactor Power Station" by R. I. Smith, G. J. Konzek, and W. E. Kennedy, Jr.; June 1978.
- ICRP Publication 23, "Task Group Report on Reference Man," Pergamon Press, Oxford, 1975,
- ICRP Publication 30, "Limits for Intakes of Radionuclides by Workers", Supplement to Part I, 1979.
- N. Irving Sax, "Dangerous Properties of Industrial Materials", Fourth Edition, Van Nostrand Reinhold Company, 1975.

# USDpumerd@agy

· · · · ·

8-435

9



Mr. G. W. Roles U.S. Nuclear Regulatory Commission Mail Stop SS-697 Washington, D. C. 20555 Mr. Roles:



8203250139 820226

PDR

PDR PR 2 46FR38081

Ancione Universities Association

Thank you for sending me a copy of the letter from Catherine Quigg containing her comments on the proposed rule 10 CFR 61, in which Ms. Quigg quotes her understanding of statements made by me to her in a telephone discussion she stated was held on August 4, 1980. Some of the quotations attributed to me differ significantly with views. I held then (and hold now). I therefore called Ms. Quigg on February 25, 1982, but we were unsuccessful in determining why her account of our discussion differs from mine.

In any case, I would wish to state my personal views on nuclear waste management directly to the NRC rather than to have them interpreted by a third party. Therefore, please note the following corrections in the quotations Ms. Quigg attributes to me in her letter containing comments to the NRC on 10 CFR 61:

( 1. Page 2. Last sentence

Contrary to the alleged quotation, I do not necessarily believe that TRU "waste should be treated the same as high-level waste". Wastes can contain TRU concentrations above the present 10 nCi/g limit and yet not present any significant health hazard. Such waste, of course, need not be treated the same as high-level waste.

2. Page 4, Paragraph 2

D -56-20 Contrary to the alleged quotation, "There is no way to reduce resin volume", I am well aware of incineration and other methods being investigated to reduce the volume of spent resins.

THE UNITED NO CHURCH

Page 4, Paragraph 3

D-55-3 Contrary to the alleged quotation, "I strongly favor the consideration of spent resins and cladding hulls as intermediate wastes", I do not favor consideration of an "intermediate waste": classification for these or any other-nuclear wastes. I believe that any such need is provided for by the waste classification system proposed in 10 CFR 61, along with DOE's Greater Confinement Disposal Studies Under their Low-Level Waste Technology Program.

Mr. G. W. Roles

February 26, 1982

I would appreciate your assistance in bringing the above corrections to the attention of any other NRC staff to whom Ms. Quigg's letter may have been distributed.

-2-

. . .

Very truly yours,

J. Howard Kitte: Manager Office of Waste Management Programs

JHK:pf

cc: C. Quigg R. D. Smith

US Dependent of Every 7700 South Cass Avenue Arcon Elinois 60439

46 PR 38081 (WM-4) and

DODES 1

PROFORED

DOCKET ANILLED

PROPOSED RIN

46 FR 517

Electrone 372/972-6677

Mr. G. W. Roles U. S. Nuclear Regulatory Commission Mail Stop SS-197 Washington, D. C. 20555

Dear Mr. Roles:

Subject: Comments by Argonne National Laboratory on Environmental Monitoring Costs in Draft EIS on 10 CFR Part 61

March 4, 1982

Reference: Ltr. J. H. Kittel to R. Dale Smith, "Comments by Argonne National Laboratory on NRC Proposed Licensing Requirements for Land Disposal of Radioactive Waste (10 CFR 61), and supporting Environmental Impact Statement (NUREG 0782)" December 14, 1981.

In response to your recent telephone inquiry, attached are the environmental monitoring cost estimates developed by Argonne National Laboratory for a reference LLW disposal site. These estimates formed the basis for our comments in the reference letter that the direct operation cost for-environmental monitoring shown in Table 3.6 of NUREG 0782 is believed to be inadequate.

Please let me know if further information is needed.

Yery truly yours,

J. Howard Kittel, Manager Office of Waste Management Programs

JHK:pf cc: S. A. Mann, DOE-CH

8203290301 820304 PDR PR 2 46FR38081 PDR

The University of Chicago

ARCOME UNVERSITIES ASSOCIATION

.D-53-

# DRAFT EIS ON 10 CFR PART 61

. .

# COMMENTS ON ENVIRONMENTAL MONITORING COSTS

The environmental monitoring costs given for the Reference Site in Table 3.6 (page 3-39) are 5543,000 over 20 years, or \$26,700 per year. The environmental monitoring program for the Reference Facility is discussed in Appendix E, Section 5.2.6 (pages E-55 and E-58). A preliminary estimate of the costs can be obtained from the sampling and analysis schedule in Table E-10 (page E-57) and is given in the following table. For this cost estimate, we have assumed that 1) the particulate air sample to be analyzed daily for gross beta-gamma activity is counted on-site as part of routine operations, 2) the other samples are sent to an outside commercial firm for analysis, and 3) the cost per analysis are average charges by firms performing this work, expressed in 1981 dollars.

Annual Environmental Monitoring Costs

Sample	No. of Locations	Frequency of Analysis	No. of Analysis	Cost per Analysis	Total Cost
External Gamma-TLD	50	Quarterly*	200*	\$25	\$ 5,000
Atmosphere - Particulate	<b>1</b> .	Weekly	52	80 (Y-spec)	4,160
Atmosphere - Charcoal	1	Weekly	52	20 (1311)	1,040
Soil and Vegetation	10	Quarterly	40	20 (β-γ) 20 (α) 40 (tritium)	800 800 1,600
Off-Site Wells	5	Semi-annually	10	80 (Y-spec) 20 (a) 40 (tritium)	800 200 400
Site Boundar, Wells	7 10	Semi-annually	20	80 (γ-spec) 20 (α) 40 (tritium)	1,600 400 800
Disposal Are Wells	<b>15</b>	Querterly	60	80 (γ-spec) 20 (α) 40 (tritium)	4,800 1,200 2,400
Trench Sumps	58	Monthly	70**	80 (γ-spec) 20 (α) 40 (tritium)	5,600 1,400 2,800
		· .		· · ·	\$35,800

"Paragraph 5.2.6.4 (page E-58) states 12 locations are to be analyzed monthly, for a total of 296 analyses. The number in this table was taken from Table E-10

Assuming water was present 102 of the time.

 2

In addition to the purely analytical costs, expenses of sample collection, sample preparation, quality control, maintenance of sampling equipment, record keeping, supplies, and other expenses may increase the analytical costs by a factor of up to two. If the analyses were performed in-house by personnel who also perform other work, such as the sample collection and preparation, some cost saving would result. We believe a more realistic estimate is \$60,000 per year for the total program.



Radiation Control Act of 1978. EPA wishes to minimize duplicative reporting requirements for releases reported to other agencies. EPA intends to work with NRC to minimize duplicative reporting requirements to the extent possible.

EPA has regulatory responsibility for the disposal of hazardous wastes under the Resource Conservation and Recovery Act, as amended (RCRA). RCRA, Section 1004(27), specifically exempts "source, special nuclear, or byproduct material as defined in the Atomic Energy Act of 1954, as amended. " Additionally, Section 1006(a) of RCRA states that "Nothing in this Act shall be construed to apply to ... any activity or substance which is subject to the ... Atomic Energy Act of 1954 (42 U.S.C. 2011 and following) except to the extent that such application (or regulation) is not inconsistent with the requirements of such Acts." RCRA does not address the issue of hazardous chemicals mixed with radioactive materials. We believe the most positive way for NRC and EPA to establish jurisdication over these wastes is in a memorandum of understanding. Such a memorandum would enable both agencies to avoid excessive costs and duplicative licensing of wastes. Furthermore, close coordination of EPA's RCRA and NRC's nuclear waste requirements is necessary in areas such as manifest tracking systems, groundwater protection, technical requirements, and financial assurances, since some NRC licensed wastes may be disposed of in EPA permitted facilities. A memorandum of understanding could serve as a vehicle for accomplishing this.

Hazardous and toxic chemicals are frequently present in these nuclear wastes. EPA is particularly concerned that these hazardous and toxic non-radioactive chemicals and their health impact are not considered in this proposed rule and EIS. We consider the rule and EIS deficient in this regard. Accordingly, EPA has rated this draft EIS ER-2 (environmental reservations and additional information

EPA has divided its enclosed comments into major and minor comments on both the proposed regulations (10 CFR 61) and the supporting draft Environmental Impact Statement (EIS). Should you have any questions on our comments, please call Dr. W. Alexander Williams (755-0790) of my staff.

Sincerely yours, Faul C. Cahill

requested).

Director Office of Federal Activities

. ....



#### Major Comments on 10 CFR 61

 The 500 millirem per year performance objective for an inadvertant intruder limit is not appropriate as a "regulatory limit." It is not a regulatory limit which will be monitored against for compliance. Nor is it a triggering level for an action such as an accident-related Protective Action Guide. The 500 millirem per year as applied in Part 61, is the design basis for the waste classification system. Wa believe that if "As Low As Reasonably Achievable" (ALARA) principles were applied, the exposure to the inadvertant intruder would be lower than 500 millirem per year. This is evident from NRC's own analyses in the regional case studies, which use realistic assumptions and the proposed prescriptive requirements. 1.2

5

-7

٠*ـ*ـــ

2. Setting an individual exposure limit at the site boundary is appropriate. The 25 millirem per year is in the correct range of values if, as the NRC indicates, that range includes 1 to 25 millirem per year. We are assured by the NRC analyses and their own statements in the DEIS that they should be able to establish a regulatory limit compatible with any future EPA standard established using ALARA~ principles.

3. It is not appropriate to adopt the contaminant level of 4 millirem per year from the National Interim Primary Drinking Water Regulations as a performance objective for contamination of public drinking water supplies from LLW disposal. The National Interim Primary Drinking Water Regulations were established in the context of levels of contamination, which if exceeded, would require mitigating action by the water supply authority. It is also not appropriate to allow one group of radioactive materials users to contaminate a water supply to a limit which would preclude other releases from nuclear power plants. hospitals, and other users. It does not appear from the NRC analysis that the LLW disposal requirements are so sensitive to this limit that a lower value would be difficult to meet. It should also be noted that the 10 picocurie per liter value for uranium and thorium is not part of the National Interim Primary Drinking Water Regulations. EPA would not object to the use of the National Interim Primary Drinking Water Regulations for protection of groundwater. In forthcoming RCRA land disposal regulations we expect to use the drinking water regulations as one aspect of groundwater protection. . . .

4. The NRC's intentions in applying 10 CFR 20 to the operational phase of LLW disposal is unclear. This arises out of the lack of specificity in Section 61.43 in view of the requirement of Section 61.41. This ambiguity is enhanced in the DEIS: see Vol. 1, Section 5.1.3 (page 35); Section 5.2 (bottom of page 38); Vol. 2, Section 6.3 and Section 6.4 (page 6-13); and Section 6.6 (page 6-18).

B-439

6-3

Detailed Comments of the U.S. Environmental Protection Agency (EPA) on the U.S. Nuclear Regulatory Commission's proposed licensing Requirements for Land Disposal of Radioactive Waste (10 CFR 61) and draft Environmental Impact Statement (NUREG-0782)

It is our understanding, after talking to NRC staff, that the offsite exposures during operation of the disposal site (approximately 20 years) should be held to the same limits as the long-term offsite exposures. To make this clear, it should be stated that Section 61.43 (use of 10 CFR 20 limits) applies to occupational exposures only. Section 61.41 should be broadened to include direct radiation exposure.

2

5. Section 61.59.b makes it clear that active institutional controls may not be relied on for more than 100 years following transfer of control of the disposal site to the owner. This is appropriate for

D-59-2 risk assessment and as a basis for design criteria. However, the regulations should contain a positive requirement that active institutional controls should be established for this period, since this is the design basis of the facility. It is also clear from the (DEIS (Vol. II, page 4-69) that institutional maintenance of records of the nature of the hazard is desirable over a longer period of time.

(J This should also be made a positive requirement of the license transfer to the site owner, although it is realized that it cannot be depended upon.

5. Class A segregated wastes will be put in separate trenches from the stable Class B wastes and will have potentially permeable trench covers. The active maintenance of such trenches can be expected to be extensive for many years. NRC should indicate how they plan to develop long-term stability of the Class A trenches. Although Class A wastes may not present a serious radiological public health hazard, they may, in the eastern United States, under certain hydrogeological conditions, cause site instability problems, pose a "public nuisance", and, more important, the non-radioactive chemical portion of Class A wastes may cause significant ground and surface water pollution just as sanitary landfills may.

7. We urge the NRC to give a high priority to fulfilling its pledge to conduct performance, safety, and cost/benefit analyses for other readily available disposal methods such as hydrofracture, deep well injection and disposal in a mined cavity. These alternatives could show lower intruder impacts. Promulgation of 10 CFR 61 should not be delayed to permit consideration of these alternatives, however.

8. Because it is based solely on the intruder scenario, Table 1 presents some practical incongruities. For example, tritium and some of the short halflife isotopes would have heat rates that Clearly would be impractical (i.e., self boiling of tritiated water and, decomposition of the solidifying medium). Also, it does not appear that some of these high specific activities exist in any actual waste stream.

A-8

3

9. Part 61 shows no consideration of hazards from other chemicals and toxic substances that may be associated with the waste. This is a particularly significant omission because some of these materials may nave essentially infinite lives compared to many of the radioactive constituents. As a minimum, Part 61 regulations should indicate that these materials must be handled in a manner compatible with RCRA requirements.

(10. The NRC proposes to develop a manifest tracking system which is "somewhat similar" to the EPA's hazardous waste manifest system. NRC

M- | { and EPA should seriously consider coordination and possible integration or the two systems. At the least, the two systems should be compatible because of the possible future need to transfer wastes from NRC regulatory authority to EPA authority, or vice versa (i.e., slightly realoactive LLW whose primary hazard is from non-radioactive substances) and for future interaction between NRC and EPA on the ocean olsposal of LLW.

# Minor Comments on 10 CFR\_61

1. (Sect. 61.52.a.6): It states that, "Waste must be placed and covered in a manner that limits the gamma radiation at the surface of the cover to levels that are a few percent above the background levels of the site." This would be difficult to enforce in its present form due to the ambiguity of "a few percent" and the variability of background levels.

-1  $\begin{cases} 2. (Sect. 61.55, Table 1): The table should have a title and appropriate labels for the columns. \end{cases}$ 

3. Methane, Carbon dioxide, and other waste decomposition gases generated within a shallow disposal trench can build up sufficient pressures to directly affect the stability of any engineered trench covers, particularly if the trench covers are impervious and not properly vented. Gases can also be generated from the decomposition of wastes in arid disposal sites, even in the absence of significant precipitation.

Gases from sanitary landfills have travelled underground for hundreds of feet. Gases from Class A trenches, therefore, have the possibility of affecting the stability of the Class B trenches if proper precautions are not taken. Gas generation is a long continuing process which commonly extends for 50 years or more, requiring active repair and maintenance work on the trenches. Part 61 does not indicate any consideration of this phenomena. 4. We recommend that the following additional or similar requirement be added to Subpart D, "a disposal medium with a permeablility sufficiently low to cause the accumulation of water in the trench should not be used, especially for Class B wastes, unless compensating measures are taken to prevent or reduce the leaching of radioactive materials from the waste such as (1) installing an impermeable trench cover to keep water out of the trench or (2) preparing wastes in a low-leachable form."

7 - 15 C

٠.

75. Part bi or its explanatory preamble should contain a clear statement that, "The intent of site selection is to insure that off-site migration or releases of radioactivity from waste disposed in a facility shall not exceed specified health and safety limits. Although it is reasonable to expect success of a properly sited facility in providing the necessary confinement, this cannot absolutely be guaranteed in advance of actual use."

۵

# Major Comments on DEIS for 10 CFR 61

ED-1

1. The need for cooperation between EPA and NRC in the development of a final 10 CFR 61 is indicated by the intention of NRC to include "Specific concentration limits for the disposal of important naturally occuring and accelerator produced nuclides" in a planned regulatory guide on the classification of waste. (See page 42 of DEIS Vol. 1 and page 7-22 of Vol. 2). Natural redionuclides, particularly radium, are expected to be included in EPA regulations under the Resource Conservation and Recovery Act. Positive steps should be taken to ensure compatibility under these two jurisdictions.

2. In the presmble of the regulation (p. 38091, column 2) and in the DEIS, it is indicated that engineered barriers such as concrete covers are assumed to have an effective life of 500 years. Outside of comments received at public hearings, there is no indication of any basis for such an assumed lifetime. NRC should discuss data which confirm the lifetime of engineered barriers and criteria for approving barrier designs.

3. Monitoring at LLW disposal facilities will be an important activity and will require a regulatory guide from NRC. This should be added to NRC's commitment to prepare regulatory guides in the future. (Vol. 2, Section 2.2 p. 2-4). This guide should establish "action levels" for elevated levels of radioactive materials in the environment to indicate when increased monitoring and corrective actions should begin.

4. The NRC calls for Class B wastes to be stable for 150 years. We believe that criteria should be given that will reasonably assure a 150 year waste stability.

с, **л**., **л**., **н**., **н** 

5. Actual experience at existing sites has shown that under certain hydrogeological conditions, such as outlined in Base Case 3, the direct overflow of contaminated water to land surface has been a very important, if not the dominant, pathway. The "direct to land surface overflow" pathway also has a very short travel path and does not give the benefits of delay for decay afforded by the longer travel paths and slower travel times of the ground-water pathway. The short- and long-term impacts of the "overflow to land surface" pathway should be evaluated and compared with the ground-water pathway for Base Case 3 and for all similar test cases.

6. (Vol. 4, Append. H, Sect. 3): The DEIS should make estimates of maximum individual and population health risks, projected over time, for each of the pathways evaluated. As a part of this evaluation, the integrated activity moving through each pathway and its resultant population dose should be presented.

7. (Vol. 2, Sect. 3.8, p. 3-48, and Append. G, p. G-98) ): The doses for transportation are based on a 1972 USAEC report on the transportation of LLW from nuclear power plants. We believe that the quantities of wastes and level of radioactivity in them are considerably higher in actual experience per shipment than was predicted in the 1972 estimates. Therefore, estimated dose levels should be adjusted accordingly. Also, Table 5.5 and the discussion on p. 5-15 should be adjusted accordingly.

4

. . .

8. The DEIS considers the time value of money in the estimation of the postoperational (closure and institutional control) costs but does not consider the time value for the design and operational costs. The operational costs which occur in the years 1 through 20 should also be discounted. Otherwise, combining the operational and postoperational costs to represent the total disposal costs for each alternative may result in an incorrect cost ranking of alternatives.

9. The cost estimates for institutional control need to be reexamined. After the several manipulations involving constant dollars, inflation and a nominal interest rate are made. It is unclear on what basis the final institutional control costs are stated...

10. The DEIS is deficient throughout in its failure to account for and assess the potential environmental impact and health risk from the non-radioactive chemical, hazardous, and toxic materials in the LLW. The contamination of ground and surface waters and risk to inadvertant intruders could be significant from non-radioactive materials.

#### Minor Comments on DEIS for 10 CFR 61

1

1. The short-term impacts of LLW disposal have been underestimated. In the Summary, in Chapters 3 and 4 of Volume 2, and in Appendices D and G, the radionuclides considered are only those with long half-lives

B-441

6

or those occurring in "significant" quantities in LLW. For short-term impacts, such as radiation doses delivered during the operational phase of a waste disposal facility (including occupational, population and accident doses), the source term should include the presence of shorter-lived, gamma-emitting radionuclides:

2. The ground-water pathway is not significant for many arid zone sites. In arid regions, attention should be given to the upward translocation of radionuclides by plants and animals and by the upward "wicking" effects of the strong evaporation potential on capillary water in the soil. Transport, and possibly erosion, by wind should also be considered.

3. The EIS does not give any specific distances for separating the Class A and Class B trenches. This distance could be significant in assuring overall site stability.

4. It is difficult to follow the projections of waste volumes as given in Tables D.25-D.26, which are based on Tables D.9 and D.11-14. For example D.9 gives untreated waste volumes by "region" while Tables D.25-26 list wastes by "spectrum".

5. Carbon-14 from Light Water Reactors (LWR) would more appropriately be scaled to Co-60. In the LWR, C-14 is produced both in the fuel and coolant. Except in cases of gross fuel failure, C-14 in the fuel stays there. Consequently, C-14 that ends up in LLW originates from activation in the reactor coolant. Therefore, it would seem more approportiate to scale C-14 to an activation product such as Co-60.

6. (Vol. 2, p. 6-7, last sentence); An explanation or reference should be given for the assumed release fraction 0.1 of the radioactivity within the waste packages involved in a fire.

7. (Vol. 2, Chap. 6, Sect. 6-2): Accident probability numbers would be helpful in placing the accidents and potential consequences in perspective. Data is available to at least make reasonable probability estimates.

9. (Append. E, Sect. 2.2): Consideration No. 1 should be expanded to include "and other discontinuities in the geologic media which increase the permeability significantly." For example, sand lenses or layers and desictation cracks, as well as other unexpected features may be encountered, in addition to those listed.

9. (Append. E., Sect. 3.2.2): Texture and mineral composition are critical parameters for radionuclide retention by a geologic medium. It would be helpful to plot the composition of the Schwinn Formation on a United States Department of Agriculture (USDA) Textural Classification Chart in addition to the descriptive terminology of sandy loam and loamy sand. The advantage of using the USDA chart is that it shows the percent clay, silt and sand fractions of the medium which are textural parameters correlative with permeability and the sorption potential of a disposal medium.

7

10. (Appendix J): Insufficient data are presented to make reasonable predictions about the sorption capabilities of the geologic media. For example, only the percentage of the silt-size fraction was given when the clay-size fraction of the medium is more important in assessing its capabilities to retain radionuclides. The Eh and pH are not presented either.

11. (Appendix J, Sect. 1.1.1): In the first sentence, change "underlain" to "overlain."

12. (Append. J. Sect. 1.1.3): The factors that contribute to increased permeability should be briefly described in this section. For example, in glacial tills, this includes sand lenses and desiccation cracks.

13. (Append. J., Sect. 1.1.9, 1.2.9 and 1.3.9): In a detailed site evaluation, the following additional information is needed: texture of the medium described in percentage of sand, silt, and clay-size fractions; mineral composition and organics described quantatively for each size fraction and a weighted average for total sample; cation exchange capacity correlated to clay mineral and organics content; Eh and pH of medium; chemistry of the groundwater; and naturally occurring radionuclides in the medium.

14. Executive Order 12291 requires government agencies to use a 10 percent real discount rate in developing their regulatory impact analyses. The DEIS uses an "implied" real discount rate of approximately 1 percent in the cost analyses. We believe the use of 1 percent in the DEIS should be reconciled with the requirements of the executive order. In doing so, the DEIS should present a sensitivity analysis of alternative discount rates, possibly using 1, 5 and 10 percent.

15. (Appendix Q): The methodology for calculation of capital, operational, closure and institutional costs appears reasonable. What appears to be lacking is a consistent treatment of the date attached to each cost. For example, capital costs are presented in 1980 dollars, operational costs are provided as sums of money appropriate to the 20 year operating life of the site but no particular date is attached to this amount, and closure costs are presented as 1980 costs inflated to the end of site closurs. In order to evaluate disposal costs appropriately, a consistent time treatment of money is needed.

16. The last term in the long equation for postoperational costs on p, Q-44 should be corrected.

17. In the discussion of financial arrangements, the license applicant is to prepare an estimate of the amount of money required for closure and long-term care. We urge the use of a very careful review process on these cost estimates because past experience has shown that the licensee, with one exception where the regulator took an active part, has traditionally underestimated closure and long-term costs.

18. (<u>Appendix G. pp. G-12/13</u>): The dilution factor Q was taken to be equal to its pumping rate in the intruder well pathway. This dilution factor should be taken as the total groundwater flow within the plume of contamination rather than the well pumping rate.

19. The methodology used to develop trench infiltration should be presented somewhere in the text or appendices. The reference cited in Appendix G on p. 68, paragraph 2, is insufficient. There is no indication that the "externally produced" value is ever allowed to vary within the analysis.

20. A clearer explanation of the basis for the indices used in calculating the interaction factors in Appendix G is needed.

21. (Append. G, p. G-68): The rate of infiltration was calculated by multiplying by the number of days for which the precipitation exceeds 0.01 in/day. This method is very simple but; in our opinion; does not meet the state of the art for estimating infiltration rates. Our technical staff is available to discuss other possible methods for calculating infiltration.

22. (Vol. 2, p. 5-22): The dose rates through the population well and surface water pathways for Case 1A (Table 5-6) is calculated to be approximately 10 times lower than for Case 1 (Table 5-3) simply due to replacing the backfill material on moderately permeable soil used in Case 1 with more permeable soil. This fact does not agree with current knowledge of the leaching process. This phenomena was explained by the vaste-water contact time being longer in Case 1 than in Case 1A and a resultant higher rate of leaching. In reality, the pellicular water in the waste, which contains dissolved radionuclides, will be maintained in between the interstices of the water all of the time. Therefore, any adde percolation of gravity water will be mixed with the pellicular water during the course of percolation and will be independent of the true velocity of the water movement. Therefore, the rate of radionculide release should change very little because of changes in the permeability of the backfill material.

23. (Vol. 1, p. 19, Table S.5): The values listed under Body & Bone need to be explained in a footnote as "the number of years at which this impact level exists."

24. At several points in the DEIS (e.g. Vol. 1, p. 23), the comment is made that "the potential hazard quickly drops to about 1000 millirem per year to bone at about 500 years following facility closure." The term "quickly" does not seem compatible with a 500 year time period.

9

25. (Vol. 1, p. 20): The statement, "Maximum annual thyroid doses are in the range of 850 mrem at the intruder and population wells, 270 mrem at the population well, and 12 mrem at the surface water body" is ambiguous. This leaves in doubt what the dose is at the population well.

26. (Vol. 1, p. 26, Table S.7 & Vol. 2, p. 4-41, Table 4.19): Several of these costs (those with double asteriak) are not comparable with the others because they are based only on the 10% volume of waste employing that technique. To make the costs comparable, costs for the other 90% of the wastes should be figured and a weighted average presented.

27. (Appendix N): EPA's authority under the Marine Protection, Research, and Sanctuaries Act of 1972 for ocean disposal of radioactive waste should be listed.

28. (<u>Appendix N</u>): EPA's proposed guidance for Occupational Exposures should be discussed. (Federal Register, Vol. 46, No. 15, Friday, January 23, 1981, page 7836).

29. (Appendix N, p. N-5): EPA's schedule for its Low-Level Waste Standard is currently under review and may be revised. We will notify NRC of any such revision.

30. (Appendix N, p. N-19): National Primary Drinking Water Regulations is incorrect, it should be Interim National Primary Drinking Water Regulations.

31. (Appendix G): Tables G-3 through G-10 are poorly titled and measurement units are not well identified.

S ....



Dear Sir or Madama

12.5

The following are comments on the U.S. Nuclear Regulatory Commission's proposed licensing requirements for land disposal of radioactive waste contained in 10 CFR part 61 and on the Draft Environmental Impact Statement NUREG-0782 prepared as a decision document for the performance objectives and technical and financial criteria set out in Part 61. We understand that the commission intends to accept comments submitted after that date and to consider that to the extent possible. Because our review of the docket indicates that very few comments were submitted on closure and post-closure care and because we consider these topics to be extremely critical aspects of the regulatory program, we are submitting the following comments. We have addressed the most major concerns expressed in the docket.

The International Research and Technology Operation of the General Research Corporation (formerly the International Research and Technology Corporation) has been involved since 1977 in supporting the U.S. Environmental Protection Agency in developing standards for closure and post-closure care; cost-estimation, and financial responsibility of the owners and operators of hazardous wasts treatment, storage and disposal facilities. Hany of the issues concerning the scope and content of Part 61 and NUREG-0782 are similar to issues addressed in the analysis of how to protect human health and the environment from hazardous wasts. In particular, the long-term nature of the threat to human health and the environment posed by both hazardous and radioactive waste makes it extremely important that effective regulatory structures be developed.

The comments which follow discuss the technical and procedural requirements established by 10 GFR Part 61. Major problems associated with most of the phases in the life cycle of a typical land disposal facility for radioactive waste are identified by phase. No attempt has been made, however, to analyze issues particularly associated with the preoperational phase, such as disposal site design or the effects of different waste characteristics and classifications. Major emphasis has been placed on issues associated with the closure plan and cost estimates and the performance of the actions they require.

A Subsidiary of Flow General Inc. An Equal Opportunity Employer M/F

## Technical and Procedural Issues a. Preoperational and Operational Phases

Although the closure of the disposal facility is extremely important to the overall assurance of protection of public health and safety, Part 61 may not establish sufficiently specific requirements for the closure plan, nor may it establish adequate procedures for revising the closure plan during the operational phase of the facility.

-2-

The initial closure plan must be prepared as a part of the special technical information contained in the license application. Section 61.12(b) requires that this information include a description of design features related to disposal site closure and stabilization, and Section 61.12(g) sets a broad performance requirement that closure should "eliminate the need for ongoing active waintenance" (emphasis added), which is defined in turn in Section 61.2 to mean such "significant remedial activity" as pumping or treatment of water from a disposal unit. Closure is, in general, expected to achieve a "reasonable assurance" that exposures to humans are within the limits established in specific performance objectives set in Sections 61.41 to 61.44. Two potential problems arise from these provisions as drafted, however. First, it may not be possible or desirable to eliminate the need for active maintenance, if that maintance includes such activities as leachate monitoring and pumping. In some cases, an environmentally sound plan for closure may still require periodic leachate pumping to ensure adequate protection. Since the commission anticipates that from five to fifteen years may be an appropriate post-closure observation and maintance period, a requirement that closure activities minimize to the extent possible the need for future active maintenance may be a more appropriate performance objective.

A second potential problem is due to the lack of procedural specificity in the closure plan regulations. Given that the Commission's intent is to allow a licenses to tailor his plan to meet the specific conditions of his site, it is important to indicate explicitly the procedures which must be followed and sufficient details about the contents of the plan to ensure that the requirement can be enforced. Issues that require classification include the following:

> To what extent does the closure plan account for activities to be carried out during the post-closure observation and maintenance period?

B-444

D-52-6

# -3-

 Is the plan based on the assumption that closure will occur when intended under ideal site conditions or should it account for activities that may be necessary under a worstcase scenario (or something in between)?

- What kinds of activities are expected to be accounted for in the closure plan? For example, what level of inventory if any, should be assumed?
- What kind of documentation is expected?
- Closure plans must be approved prior to closure. How long prior to closure must they be submitted? What are the provisions of this review process? Involvement of the public? What happens if the plan is not approved?

Detailed guidance may be sufficient to clarify certain kinds of questions, e.g., the level of documentation expected in the plan. Other issues must be clarified in the regulation, however, if the Commission is to be assured that a licensee's closure plan reflects the needs of the site and can be used as a basis for financial responsibility requirements.

The purpose of the closure plan is to ensure that adequate preparations have been made for site closure and to serve as the basis for the level of financial responsibility required to ensure that adequate funds are available for closure. Unless the closure plan adequately reflects actual site conditions these goals will not be met. Provisions in the regulations for revising the plan during the operational phase may not overcome these problems. Part 61 seems to require that closure plans be revised, if necessary, as part of the closure application procedures. Although the provisions of Section 61.25 implicitly include revisions in the closure plan, the requirement is not clear. In addition, the requirement does not specifically state the circumstances under which the plan must be revised.

If regulations do not explicitly state when the plan must be revised, two problems could arise. First, an amergency situation could occur which did not allow for an extended period of planning before closure operations had to begin. In such a situation, the existing closure plan could be found to be seriously inadequate. Second, because the cost estimates are tied to the closure plan, infrequent revisions in the closure plan could lead to inaccurate cost estimates and inadequate financial assurance. Although the financial assurance mechanism will be revised annually, such a review can only ensure that the mechanism is adequate to pay for the activities called for by the closure plan. An inadequate plan will lead directly to inadequate assurance.

In addition to problems with the content of the closure plan, the precise relationship between the closure plan and the closure cost estimate is left uncertain. First, the activities that are intended to be included in the estimate and the assumptions to be used in estimating costs are left unclear (e.g., it is not clear if estimates are to be

-4-

based on a worst-case scenario). The EPA, for example, in its cost estimating regulations requires that cost estimates be based on the costs of closure of the maximum extent of site operation. Second, if the cost estimate and financial responsibility are to include the postclosure observation period, it is essential to determine the length of that observation period. If the cost estimate is not prepared assuming the longest likely post-closure period, then insufficient funds may be available if needed. For financial planning purposes it is also necessary to know the length of the post-closure period and the extent of funds that will be required.

b. Closure Phase and Post-Closure Phase

The Part 61 regulations do not completely resolve the question of when the closure phase is terminated. Section 61.30 states that the Commission will determine that closure has been made "in conformance with the licensee's disposal site closure plan, as amended and approved as part of the license". However, Section 61.29 provides that even following "completion of closure" the licensee will be required to perform observation, monitoring, maintenance and repairs "until the site closure is complete" and the license is transferred. The Commission estimates that post-closure observational maintenance will last from five to fifteen years. Despite the separation of the closure phase from the post-closure observation and maintenance phase, therefore, it appears that in practice the two phases must be treated together. The closure plan will have to include plans for post-closure observation and maintenance, or those activities will not be properly reflected in the cost estimates and financial assurances. Such estimates will be difficult to achieve because the duration of the post-closure phase is not precisely defined.

10 - E

Determination by the Commission that the required closure or postclosure activities have been completed satisfactorily and that the license may be transferred may also be difficult, in the absence of any certification procedures. Similar regulations in other related areas such as hazardous waste frequently require that the licensee obtain a certification from an engineer that activities required by the closure plan have been carried out.

A third problem associated with the closure and post-closure phases will arise if certain forms of active maintenance cannot be terminated. For example, if leachate pumping cannot be discontinued, or if such pumping can only be avoided by massive expenditures not provided

E-1

R-B

Werner & Pfleiderer Corporation

NOCKET RUNDER

PROPOSED RULE

• . .

-5-

B-B for in the financial assurance obtained by the licensee, the Commission may either be forced to undertake legal action, including bankruptcy proceedings, to obtain additional funds from the licensee, or be compelled to transfer the site. Such situations could lead to conflicts between the Commission, the states, and the licensee over transfers of responsibility.

c. Institutional Control Phase

The key issues to be resolved regarding requirements during this phase involve the scope of required activities and the allocation of responsibilities between the Commission and the Agent responsible during the institutional control period. The intent of closure is to ensure that no active maintenance is needed during the institutional control period. Section 61.63, however, stipulates that financial assurance must be available during the institutional control period to cover "any required maintenance...". It is unclear who has the responsibility of performing major maintenance activities, if necessary, and how disputes lover that responsibility will be resolved.

Second, the contents of the agreements are unclear and the criteria to be used in evaluating these agreements are not indicated. It is also unclear if these leasing agreements will be reviewed during the operational phase and who will take responsibility for the institutional control period if the leasing agreement is deemed no longer adequate.

> Sincerely, Bolin Bodensky Severn Econonic and Regulatory Analysis Department INTERNATIONAL RESEARCH AND TECHNOLOGY OPERATIONS

RRS/sb

E-1

'22 1217 31 P?:07 Harch 24, 1982

Mr. Tim Johnson U.S. Nuclear Regulatory Commission Silver Spring, MD

SUBJECT: 10CFR61/Waste Form Stability

Dear Tim:

We wish to thank Paul Lohaus, Dale Smith and yourself for the time spent with us to discuss IOCFR61 and our comments concerning waste form stability. We can appreciate the difficulty you face in developing a simple yet comprehensive statement defining structural stability of buried waste forms. Per our discussion, we have given the matter some thought and have the following suggestions regarding alternative wording:

"A structurally stable waste form shall remain a monolithic solid without volume change under the expected disposal conditions."

Or, if there is a need to distinguish thermoplastics from other waste forms;

"Waste forms which can undergo plastic deformation under a compressive stress equivalent to the burial environment shall be required to behave as a monolithic, volumetrically incompressible component of the surrounding soil; i.e. experience no appreciable volume changes or crumbling to form a discontinuous solid."

We trust that you will find the foregoing suggestions helpful as you attempt to incorporate solicited comments into the final regulation. If we can be of any further assistance, please do not hesitate to contact MS.

Very truly yours,

Billste

William J. Klein Manager, Product Development Hazardous Waste Treatment Systems

WJK/gs

D-56-9



Gentlemen:

Enclosed please find supplemental comments by the Township of Lower Alloways Creek to proposed rule for licensing requirements for land disposal of radioactive waste.

Very truly yours, CARL J. VALORE

CJV/sgp

ver 6 8-71 4/8/82

8204160485 820405 PDR PR 2 45FR38081 PDR

UNITED.STATES OF AMERICA NUCLEAR REGULATORY COMMISSION Supplemental Comments of the :Township of Lower Alloways Creek 10 CFR Part 61. et seg. to Proposed Rule for Licensing Requirements for Land Disposal of Radioactive Waste TO: Secretary of the Commission U.S. Nuclear Regulatory Commission Washington, D.C., 20555 Attention: Docketing & Service Branch The Township of Lower Alloways Creek hereby supplements the comments it filed on January 12, 1982 as follows: The classification established in the DEIS, Vol. 2, part 2.4.3.1, is based principally on radiotoxicity. The Township of Lower Alloways Creek is of the opinion that the suggestion to classify on the basis of total hazard-chemical, biological and physical as well as radiological, can be implemented. Tehnical meetings between the Environmental Protection Agency and the Nuclear Regulatory Commission should be productive of a classification system for non-radiological hazards at low-level waste disposal sites. The failure of the Nuclear Regulatory Commission to address the chemical, biological and physical hazards of material to be placed at low-level waste disposal sites constitutes a regulatory gap and a minimization of the risks involved. Respectfully CARL J. VALORE, Special Nuclear Counsel for the Township of Lower Alloways Creek April 5, 1982

D-55-9

NRC FORM 335 (7-77) U.S. NUCLEAR REGULATORY COMMISSION BIBLIOGRAPHIC DATA SHEET	1. REPORT NUMBER (Assigned by DDC) NUREG-0945, Vol. 2					
4. TITLE AND SUBTITLE (Add Volume No., if appropriate) Final Environmental Impact Statement on 10 CFR Pa "Licensing Requirements for Land Disposal of Radi Waste" Vol. 2: Appendices A-B	2. (Leave blank) 3. RECIPIENT'S ACC	ESSION NO.				
7. AUTHORIS) NRC Staff		5. DATE REPORT COMPLETED MONTH May YEAR 1982				
9. PERFORMING ORGANIZATION NAME AND MAILING ADDRESS (Include . Low-Level Waste Licensing Branch, Division of Was ment, Office of Nuclear Material Safety and Safeg Nuclear Regulatory Commission, Washington, DC 20	DATE REPORT IS MONTH 6. (Leave blank) 8. (Leave blank)	SUED VEAR 1982				
Same as item 9.	10. PROJECT/TASK/V	VORK UNIT NO.				
13. TYPE OF REPORT Final Environmental Impact Statement	PERIOD COVERE	L RD (Inclusive dates)				
15. SUPPLEMENTARY NOTES	<u> </u>	14. (Leave plank)				
The three-volume final environmental impact statement (FEIS) is prepared to guide and support publication of a final regulation, 10 CFR Part 61, for the land disposal of low- level radioactive waste. The FEIS is prepared in response to public comments received on the draft environmental impact statement (DEIS) on the proposed Part 61 regulation. The DEIS was published in September 1981 as NUREG-0782. Public comments received on the proposed Part 61 regulation separate from the DEIS are also considered in the FEIS. The FEIS is not a rewritten version of the DEIS, which contains an exhaustive and detailed analysis of alternatives, but rather references the DEIS and presents the final decision bases and conclusions (costs and impacts) which are reflected in the Part 61 requirements. Four cases are specifically considered in the FEIS representing the following: past disposal practice, existing disposal practice, Part 61 requirements, and an upper bound example.						
17. KEY WORDS AND DOCUMENT ANALYSISlow-level wastefinancial assurancesland disposalinstitutional controlssocial commitmentradioactive wastegroundwater migrationdisposal technologiesinadvertent intrusioncost-benefit analysis10 CFR Part 61waste form	7a. DESCRIPTORS					
176. IDENTIFIERS/OPEN-ENDED TERMS						
18. AVAILABILITY STATEMENT	19. SECURITY Unclassi	CLASS (This report)	21 NO OF PAGES 613			
Unlimited	20. SECURITY Unclassi	CLASS (This page) fied	22 PRICE S			

•

c /